# YEARBOOK OF THE INTERNATIONAL LAW COMMISSION 1985

Volume II
Part One

Documents of the thirty-seventh session (Addendum)



# YEARBOOK OF THE INTERNATIONAL LAW COMMISSION

1985

Volume II
Part One

Documents of the thirty-seventh session (Addendum)



#### NOTE

Symbols of United Nations documents are composed of capital letters combined with figures. Mention of such a symbol indicates a reference to a United Nations document.

References to the Yearbook of the International Law Commission are abbreviated to Yearbook..., followed by the year (for example, Yearbook... 1980).

The Yearbook for each session of the International Law Commission comprises two volumes:

Volume I: summary records of the meetings of the session;

Volume II (Part One): reports of special rapporteurs and other documents considered during the session;

Volume II (Part Two): report of the Commission to the General Assembly.

All references to these works and quotations from them relate to the final printed texts of the volumes of the Yearbook issued as United Nations publications.

\* \*

The two studies prepared by the Secretariat, which were originally issued in mimeographed form, are reproduced in the present volume with the editorial changes required for the presentation of the final texts.

A/CN.4/SER.A/1985/Add.1 (Part 1/Add.1)

UNITED NATIONS PUBLICATION

Sales No. E.86.V.9 (Part 1/Add.1)

ISBN 92-1-133324-5 ISSN 0082-8289

#### CONTENTS

|  | D    |
|--|------|
| International liability for injurious consequences arising out of acts not prohibited by international law (agenda item 8)   | Page |
| Document A/CN.4/384. Survey of State practice relevant to international liability for injurious consequences arising out of acts not prohibited by international law, prepared by the Secretariat  | 1    |
| Relations between States and international organizations (second part of the topic) (agenda item 9)  |      |
| Document A/CN.4/L.383 and Add.1-3. Practice of the United Nations, the specialized agencies and the International Atomic Energy Agency concerning their status, privileges and immunities: supplementary study prepared by the Secretariat | 145  |

# INTERNATIONAL LIABILITY FOR INJURIOUS CONSEQUENCES ARISING OUT OF ACTS NOT PROHIBITED BY INTERNATIONAL LAW

#### [Agenda item 8]

#### **DOCUMENT A/CN.4/384\***

Survey of State practice relevant to international liability for injurious consequences arising out of acts not prohibited by international law, prepared by the Secretariat

[Original: English] [16 October 1984]

#### CONTENTS

|            |   |            | Page |
|------------|---|------------|------|
| Abbreviati | ions  |            | 3    |
| Note conce | erning quotations   |            | 3    |
|            |   |            |      |
|            |   | Paragraphs |      |
| Introduct  | TION  | 1-19       | 4    |
| Chapter    |   |            |      |
| l. Acti    | VITIES CAUSING INJURIES BEYOND THE TERRITORIAL JURISDICTION OR CONTROL OF THE               |            |      |
| STAT       | TE WHERE THEY ARE CONDUCTED   | 20-54      | 6    |
| Α.         | Forms of activities   | 20-36      | 6    |
|            | (a) Multilateral agreements   | 23-27      | 7    |
|            | (b) Bilateral agreements  | 28-31      | 8    |
|            | (c) Judicial decisions and State practice other than agreements                             | 32-36      | 8    |
| В.         | Location of origin of activities  | 37-54      | 8    |
|            | 1. Activities conducted within the territorial jurisdiction or under the control of the ac- |            |      |
|            | ting State  | 39-49      | 9    |
|            | (a) Multilateral agreements   | 40-43      | 9    |
|            | (b) Bilateral agreements  | 44-46      | 9    |
|            | (c) Judicial decisions and State practice other than agreements                             | 47-49      | 10   |
|            | 2. Activities conducted outside the territorial jurisdiction or control of the acting State | 50-54      | 10   |
|            | (a) Multilateral agreements   | 50-52      | 10   |
|            | (b) Bilateral agreements  | 53         | 11   |
|            | (c) Judicial decisions and State practice other than agreements                             | 54         | 11   |
| ll. Asse   | SSMENT OF ACTIVITIES FOR THEIR INJURIOUS IMPACT   | 55-248     | 11   |
|            | (a) Multilateral agreements   | 56-60      | 12   |
|            | (b) Bilateral agreements  | 61-64      | 13   |
|            | (c) Judicial decisions and State practice other than agreements                             | 65-68      | 13   |
| Α.         | Data collection   | 69-96      | 14   |
|            | (a) Multilateral agreements   | 70-75      | 14   |
|            | (b) Bilateral agreements  | 76-83      | 16   |
|            | (c) Judicial decisions and State practice other than agreements                             | 84-96      | 19   |
| В.         | Prior negotiation and consultation  | 97-195     | 22   |
|            | (a) Multilateral agreements   | 99-103     | 22   |
|            | (b) Bilateral agreements  | 104-107    | 23   |
|            | (c) Judicial decisions and State practice other than agreements                             | 108-113    | 24   |
|            | 1. Definition of harm   | 114-167    | 26   |
|            | (a) Multilateral agreements   | 116-135    | 27   |
|            | (b) Bilateral agreements  | 136-150    | 32   |
|            | (c) Judicial decisions and State practice other than agreements                             | 151-167    | 35   |

<sup>\*</sup> Originally issued under the symbol ST/LEG/15 and Corr.1.

| Chapt | er. |   | Paragraphs | Page |
|-------|-----|---|------------|------|
|       |     | 2. Competence to decide what constitutes harm                   | 168-195    | 40   |
|       |     | (a) Multilateral agreements                                     | 170-175    | 40   |
|       |     | (b) Bilateral agreements  | 176-184    | 41   |
|       |     | (c) Judicial decisions and State practice other than agreements | 185-195    | 43   |
|       | C.  | Balancing of interests  | 196-240    | 46   |
|       |     | (a) Multilateral agreements                                     | 198-207    | 46   |
|       |     | (b) Bilateral agreements  | 208-215    | 49   |
|       |     | (c) Judicial decisions and State practice other than agreements | 216-240    | 51   |
|       | D.  | Exoneration from the duty of prior negotiation                  | 241-248    | 56   |
|       |     | (a) Multilateral agreements                                     | 242-244    | 56   |
|       |     | (b) Bilateral agreements  | 245-246    | 57   |
|       |     | (c) Judicial decisions and State practice other than agreements | 247-248    | 57   |
| 111.  | Pri | VENTIVE MEASURES  | 249-340    | 58   |
|       | Α.  |   | 252-299    | 58   |
|       | Α.  | (a) Multilateral agreements                                     | 252-270    | 58   |
|       |     | (b) Bilateral agreements  | 271-286    | 63   |
|       |     | (c) Judicial decisions and State practice other than agreements | 287-299    | 66   |
|       | В.  | Provisions for prevention of harm                               | 300-340    | 68   |
|       | В.  | (a) Multilateral agreements                                     | 301-309    | 68   |
|       |     | (b) Bilateral agreements  | 310-318    | 69   |
|       |     | (c) Judicial decisions and State practice other than agreements | 319-340    | 71   |
|       |     | (c) Judicial accisions and state practice other than agreements | 317-340    |      |
| IV.   | Gu. | ARANTEES OF COMPENSATION FOR INJURIES                           | 341-357    | 74   |
|       |     | (a) Multilateral agreements                                     | 342-349    | 74   |
|       |     | (b) Bilateral agreements  | 350-352    | 76   |
|       |     | (c) Judicial decisions and State practice other than agreements | 353-357    | 76   |
| V.    | Lia | BILITY  | 358-508    | 77   |
|       |     | (a) Multilateral agreements                                     | 384-389    | 82   |
|       |     | (b) Bilateral agreements  | 390-394    | 83   |
|       |     | (c) Judicial decisions and State practice other than agreements | 395-404    | 84   |
|       | A.  | Balancing of interests  | 405-412    | 86   |
|       |     | (a) Multilateral agreements                                     | 407-410    | 86   |
|       |     | (b) Bilateral agreements  | 411-412    | 87   |
|       | В.  | Operator's liability  | 413-437    | 87   |
|       |     | (a) Multilateral agreements                                     | 415-430    | 88   |
|       |     | (b) Bilateral agreements  | 431-434    | 93   |
|       |     | (c) Judicial decisions and State practice other than agreements | 435-437    | 93   |
|       | C.  | State liability   | 438-483    | 94   |
|       |     | (a) Multilateral agreements                                     | 439-445    | 94   |
|       |     | (b) Bilateral agreements  | 446-462    | 95   |
|       |     | (c) Judicial decisions and State practice other than agreements | 463-483    | 98   |
|       | D.  | Exoneration from liability                                      | 484-508    | 101  |
|       |     | (a) Multilateral agreements                                     | 485-504    | 101  |
|       |     | (b) Bilateral agreements  | 505-507    | 105  |
|       |     | (c) Judicial decisions and State practice other than agreements | 508        | 106  |
| VI.   | Co  | MPENSATION  | 509-621    | 106  |
|       | _   | Content   | 510-567    | 106  |
|       |     | 1. Compensable injuries   | 510-544    | 106  |
|       |     | (a) Multilateral agreements                                     | 510-516    | 106  |
|       |     | (b) Bilateral agreements  | 517-521    | 107  |
|       |     | (c) Judicial decisions and State practice other than agreements | 522-544    | 108  |
|       |     | 2. Forms of compensation  | 545-555    | 111  |
|       |     | (a) Multilateral agreements                                     | 546-549    | 111  |
|       |     | (b) Bilateral agreements  | 550-552    | 112  |
|       |     | (c) Judicial decisions and State practice other than agreements | 553-555    | 112  |
|       |     | 3. Limitation on compensation                                   | 556-567    | 113  |
|       |     | (a) Multilateral agreements                                     | 557-563    | 113  |
|       |     | (b) Bilateral agreements  | 564-566    | 114  |
|       |     | (c) Judicial decisions and State practice other than agreements | 567        | 115  |

|      |   | Paragraphs | Page |  |
|------|---|------------|------|--|
|      | B. Authorities competent to award compensation                    | 568-606    | 115  |  |
|      | 1. Local courts and authorities                                   | 570-584    | 115  |  |
|      | (a) Multilateral agreements                                       | 570-580    | 115  |  |
|      | (b) Bilateral agreements  | 581-583    | 117  |  |
|      | (c) Judicial decisions and State practice other than agreements   | 584        | 118  |  |
|      | 2. International courts, arbitral tribunals and joint commissions | 585-589    | 118  |  |
|      | (a) Multilateral agreements                                       | 585-586    | 118  |  |
|      | (b) Bilateral agreements  | 587-588    | 122  |  |
|      | (c) Judicial decisions and State practice other than agreements   | 589        | 122  |  |
|      | 3. Applicable law   | 590-606    | 122  |  |
|      | (a) Multilateral agreements                                       | 590-594    | 122  |  |
|      | (b) Bilateral agreements  | 595-601    | 123  |  |
|      | (c) Judicial decisions and State practice other than agreements   |            | 123  |  |
|      | C. Enforcement of judgments                                       | 607-621    | 124  |  |
|      | (a) Multilateral agreements                                       | 608-617    | 124  |  |
|      | (b) Bilateral agreements  | 618-620    | 126  |  |
|      | (c) Judicial decisions and State practice other than agreements   | 621        | 127  |  |
|      | ANNEXES   |            | Page |  |
| T    | Multilatoral traction   |            | _    |  |
| Ι.   | Multilateral treaties   |            | 128  |  |
| II.  | Bilateral treaties  |            |      |  |
| III. | Judicial decisions and State practice other than agreements 138   |            |      |  |

#### **ABBREVIATIONS**

| ECE   | Economic Commission for Europe                                  |
|-------|---|
| IAEA  | International Atomic Energy Agency                              |
| ICJ   | International Court of Justice                                  |
| IMCO  | Inter-Governmental Maritime Consultative Organization (now IMO) |
| IMO   | International Maritime Organization                             |
| INTAL | Inter-American Institute of International Legal Studies         |
| OAS   | Organization of American States                                 |
| OECD  | Organisation for Economic Co-operation and Development          |
| PCIJ  | Permanent Court of International Justice                        |
| UNEP  | United Nations Environment Programme                            |
|       | •   |
|       |   |

| I.C.J. Reports       | ICJ, Reports of Judgments, Advisory Opinions and Orders                       |
|----------------------|---|
| P.C.I.J., Series A   | PCIJ, Collection of Advisory Opinions (Nos. 1-24, up to and including 1930)   |
| P.C.I.J., Series A/B | PCIJ, Judgments, Orders and Advisory Opinions (Nos. 40-80: beginning in 1931) |

#### NOTE CONCERNING QUOTATIONS

In quotations, words or passages in italics followed by an asterisk were not italicized in the original text.

Unless otherwise indicated, quotations from works in languages other than English have been translated by the Secretariat.

#### Introduction

- 1. At its thirty-fifth session, in 1983, the International Law Commission requested that three parts of the study prepared by the Secretariat on multilateral and bilateral agreements and judicial decisions and State practice other than agreements relating to the question of international liability for injurious consequences arising out of acts not prohibited by international law be made widely available. The three parts of that study were prepared between 1982 and 1983 for the use of the late Robert Q. Quentin-Baxter, the Special Rapporteur for that topic. At the request of the Commission, the Secretariat has now updated them and combined them in a single volume.
- It is not the purpose of this study to define, alter or in any way affect the scope or the framework of the subject under consideration by the Commission. The outline of the study and the individual papers were prepared when the Commission was still at the preliminary stage of examining the scope and the framework of the topic. On the basis of prior preliminary studies and taking into account the reports of the Special Rapporteur as well as the Commission's reports on the question, the Secretariat examined the subject, as well as State practice, in a factual context. The presentation of material and information in this document does not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations concerning its content nor on the positions that States may have adopted regarding specific cases or agreements referred to therein.
- Briefly, the factual context of the subject is marked by the increasingly intensive use in many different forms of the resources of the planet for economic, industrial or scientific purposes. Because of their economic and ecological interdependence, activities occurring within or beyond the territorial control or jurisdiction of States may have an injurious impact on other States or their nationals. To cite only one example, at this stage of the technological revolution, the activities of productive plants may cause harmful consequences which may cross boundaries, causing atmospheric changes through "acid rain" or through river and coastal waters. Furthermore, the scarcity of natural resources, the need for the more efficient use of resources and the creation of substitute resources have led to innovative production methods, sometimes with unpredictable consequences.<sup>2</sup> This factual aspect of global interdependence has been demonstrated by events that have frequently resulted in injuries beyond the territorial jurisdiction or control of the acting State.3

- 4. Activities having an extraterritorial injurious impact are also conducted by private entities. These entities operate within the territorial jurisdiction or under the control of the acting State, within the shared domain or within the territorial jurisdiction or under the control of the injured State. For economic reasons, private entities transfer from one State to another hazardous and heavily polluting industries such as those producing steel, aluminium, asbestos and certain toxic chemicals. The injuries may be considerable. Reports indicate that these injuries are not limited to the recipient State, but sometimes cross into neighbouring States, and indeed occasionally even to the original exporting State.
- 5. Acts with extraterritorial injurious impact have always been met with in international relations and have been of concern to international law. States appear to have recognized that, in the exercise of their exclusive authority within or beyond their territories, over their ships, for example, they are expected to have due regard for the interests of other States that may adversely be affected. The present study reviews a number of examples

The States members of OECD have attempted to provide for environmental protection in "Guidelines for multinational enterprises", which the OECD Council was to review in 1984. See also OECD, Economic and Ecological Interdependence (Paris, 1982), p. 66.

In 1983, the Inter-Governmental Working Group on Transnational Corporations prepared a comprehensive draft code of conduct for transnational corporations. The draft includes a section on environmental protection that provides for measures to avoid and remedy environmental damage, to supply relevant information to developing countries about the potential hazards involved in certain industrial activities, etc. The principles of environmental protection of the code of conduct provide as follows:

<sup>&</sup>lt;sup>1</sup> Yearbook . . . 1983, vol. II (Part Two), p. 83, para. 286.

<sup>&</sup>lt;sup>2</sup> It is not the purpose of this study to describe the factual instances of global interdependence. The above brief description is intended merely to indicate the reasons for the choice of materials on State practice.

<sup>&</sup>lt;sup>3</sup> The "acting State", in this study, refers to the State within whose territorial jurisdiction or under whose control an activity has taken place that has caused or may cause injuries beyond its territorial jurisdiction or control to other States or their nationals.

<sup>&#</sup>x27;The 'injured State' or 'affected State' refers to the State that has suffered or may suffer injuries as a result of an activity by the acting State. The injuries may be to the State's property or the private property of its nationals.

<sup>&</sup>lt;sup>5</sup> These polluting industries are sometimes transferred from developed to developing countries, where labour and production costs are lower and standards of environmental regulations are looser or less strictly enforced. See B. Castleman, *The Export of Hazardous Factories to Developing Nations* (1978).

<sup>&</sup>lt;sup>6</sup> See North-South: a Programme for Survival, report of the Independent Commission on International Development Issues under the chairmanship of Willy Brandt (London, Pan Books, 1980).

<sup>&</sup>quot;Environmental protection

<sup>&</sup>quot;44. Transnational corporations, in carrying out their production activities, shall comply with national policies, laws and regulations of the countries in which they operate with regard to preservation of the environment. They shall take steps to improve the environment and make efforts to develop and apply adequate technologies for this purpose.

<sup>&</sup>quot;45. Transnational corporations shall supply to the authorities of the countries in which they operate all relevant information concerning:

<sup>&</sup>quot;(a) Features of their products or processes which may harm the environment and the measures and costs required to avoid harmful effects;

<sup>&</sup>quot;(b) Prohibitions, restrictions, warnings and other regulatory measures imposed in other countries, on grounds of protection of the environment, on products and processes which they have introduced or intend to introduce in the countries concerned.

<sup>&</sup>quot;46. Transnational corporations shall be responsive to requests from Governments of the countries in which they operate and be prepared where appropriate to co-operate with international organizations in their efforts to develop and promote national and international standards for the protection of the environment." (E/C.10/1983/S/4, pp. 11-12.)

of State co-operation, evidenced by treaties, in which the parties have agreed on procedures under which certain activities may be conducted. The substance of these agreements reveals certain procedural and substantive principles by which the parties have accommodated their conflicting interests: "good-neighbourliness", "due care", "equitable principles".

- The materials examined in this study have been selected and analysed on the basis of their relevance to the concepts of good-neighbourliness, due care, equitable principles, prior negotiation and consultation, balance of interests and prevention and minimization of injuries to others in the undertaking of activities within or beyond the territorial jurisdiction or control of States. It is not suggested that every example of State practice examined deals only and directly with acts "not prohibited by international law". Their selection was dictated by their relevance to the topic of liability or by the pertinence of the activities examined, whether or not they were wrongful. It is therefore pertinent to consider the handling of some disputes in which there was no general agreement as to the lawfulness or unlawfulness of the acts or omissions giving rise to injurious consequences; for example, the hydrogen bomb tests in the atmosphere in the 1950s and 1970s generated debate concerning their lawfulness among jurists. Similarly, there was extensive discussion among jurists concerning the lawfulness of the hydrogen bomb tests conducted by the United States of America on the high seas in 1954, although this matter was never submitted for judicial decision.7 The French nuclear tests gave rise to the same debate. That matter was submitted to the International Court of Justice for a decision, but the judgment was not rendered on the merits of the case.8
- Although certain treaties deal with matters that may be characterized as "wrongful acts", they relate to problems relevant to the topic of international liability and have been included for that reason. These treaties demonstrate the relevance of the concepts of due care, good-neighbourliness, etc., and the forms in which they have been used. A detailed examination of these treaties reveals how a particular activity with potentially injurious impact has been undertaken under some form of supervision, what preventive measures are required in order to avoid or at least to minimize injuries to other international actors, what kind of injuries are involved and at what point they become unacceptable and entail liability and, finally, what are the remedies. Since this is a survey of past trends, treaties have been selected whether or not they are still in force.
- A large number of bilateral agreements have applied the concepts of good-neighbourliness, due

care, etc., in the utilization of shared rivers. Since most of those agreements were examined by Mr. Schwebel in his third report on the law of the non-navigational uses of international watercourses, only some of them are examined in this study.

- Judicial decisions of domestic courts, of international courts and of arbitral tribunals involving efforts by third-party decision-makers are relevant for the substantive principles they examine and sometimes for the factors they balance against one another. Documents exchanged between foreign ministries and government officials are important sources of State practice, as are settlements of disputes through nonjudicial methods. Although they are not products of conventional judicial procedure, they may represent a pattern in trends regarding substantive issues in dispute. Statements made by the State officials involved as well as the content of the actual settlement of disputes will be examined for their possible relevance to the substantive principles of liability.
- 10. This study has not ignored the difficulties of evaluating a particular instance as "evidence" of State practice. 10 Different policies may motivate the conclusion of treaties or decisions. Some may be compromises accommodations for extraneous reasons. But repeated instances of State practice, when they follow and promote similar policies, may create expectations about the authoritativeness of those policies in future behaviour. Even though some of the policies may not have been explicitly stated in connection with the relevant events, or may purposely and explicitly have been left undecided, continuous similar behaviour may lead to the creation of a customary norm. Whether or not the materials examined here are established as customary law, they demonstrate a trend in expectations and may contribute to the clarification of policies concerning some detailed principles of the international liability topic. Practice also demonstrates ways in which competing principles, such as "State sovereignty" and "domestic jurisdiction", are to be reconciled with the new norms.
- 11. In referring to State practice, caution must be exercised in extrapolating principles, for the more general

However, in its judgment of 6 April 1955 in the Nottebohm case (second phase), the Court relied on State restraint as evidence of the existence of an international norm restricting freedom of action

(I.C.J. Reports 1955, pp. 21-22).

<sup>&</sup>lt;sup>7</sup> See e.g. E. Margolis, "The hydrogen bomb experiments and international law", The Yale Law Journal, New Haven, Conn., vol. 64, 1955, p. 629; M. S. McDougal and N. A. Schlei, "The hydrogen bomb tests in perspective: lawful measures for security", ibid., p. 648; M. S. McDougal, "The hydrogen bomb tests and the international law of the sea", The American Journal of International Law, Washington, D.C., vol. 49, 1955, p. 356; and H. J. Taubenfeld, "Nuclear testing and international law", Southwestern Law Journal, Dallas, Tex., vol. 16, p. 365.

Nuclear Tests (Australia v. France; New Zealand v. France), judgments of 20 December 1974, I.C.J. Reports 1974, pp. 253 and 457.

<sup>&#</sup>x27; Yearbook . . . 1982, vol. II (Part One), p. 65, document A/ CN.4/348.

<sup>&</sup>lt;sup>10</sup> For example, abstention by States from engaging in activities which, although lawful, may cause injuries beyond their territorial jurisdiction, may or may not be relevant to creating customary behaviour. The Permanent Court of International Justice and its successor, the International Court of Justice, have observed that the mere fact of abstention, without careful consideration of the motivating factors, is insufficient proof of the existence of an international legal custom. Abstention by States from acting in a certain way may have a number of reasons, not all of which have legal significance. See the judgment rendered on 7 September 1927 by the Permanent Court of International Justice in the Lotus case (P.C.I.J., Series A, No. 10, p. 28). A similar point was made by the International Court of Justice in its judgment of 20 November 1950 in the Asylum case (I.C.J. Reports 1950, p. 286), and in its judgment of 20 February 1969 relating to the North Sea Continental Shelf cases (I.C.J. Reports 1969, p. 44, para. 77). See also C. Parry, The Sources and Evidences of International Law (Manchester University Press, 1965), pp. 34-64.

expectations about the degree of tolerance concerning the injurious impact of activities can vary from activity to activity. For example, the general expectations about appropriate behaviour concerning economic and monetary activities may differ, as far as their extraterritorial injurious impact is concerned, from those regarding experimental or industrial activities, or activities relating to self-defence, self-help, the environment, etc.

- 12. The materials examined in this study are not, of course, exhaustive. They relate primarily to activities concerning the physical use and management of the environment, for State practice in regulating activities causing injuries to other States has been developed more extensively in this area. The format of the study is also designed to be a useful source material; hence, relevant extracts from treaties, judicial decisions and official correspondence are also cited.
- 13. The outline of the study has been formulated on the basis of *functional* problems which may appear relevant to the topic of international liability. Since the focus of the topic appears to be on the continuing flow of activities from the stage of initiation to that of completion, the study follows a similar chronological order.
- 14. Chapter I describes activities which have been regulated for their possible extraterritorial injurious impact in terms of both their nature and their location of origin.
- 15. Chapter II examines the process of initiation of activities that may entail extraterritorial injurious impact. It points out different stages of this process in which the acting State, prior to undertaking the activity, attempts to assess the impact of the activity on other States and international actors. State practice demonstrates the existence of a rather complex procedure for assessing the impact of activities, such as the collection of data by the acting States about the activities and their possible impact, negotiation with potentially affected (injured) States and the balancing of the interests involved by cor-

relating the benefits of carrying out the activity with its cost, etc.

- 16. Chapter III examines the procedure by which attempts are made to prevent or at least minimize extraterritorial injuries, and reviews the monitoring system provided in treaties and recommended in State practice. It also points out the types of or recommended changes in activities in order to prevent or minimize their injurious impact. It appears that monitoring systems may involve co-operation among the acting and the injured States, or may be entrusted to an independent nongovernmental body, etc.
- 17. Chapter IV examines the requirements of guarantees for payment of compensation in relation to activities with strong potential extraterritorial injurious impact whose performance has been agreed upon by the acting and the injured States.
- 18. Chapter V examines the issue of liability for extraterritorial injurious impact. Despite compliance with procedural requirements designed to prevent or minimize damage, injuries may be suffered by other States and their nationals. This chapter examines the issue of liability. It points out that in determining the liability of the acting State a balance is struck between the interests of the parties and those of the larger community. It also examines the extent to which the operator of the activity or the State in whose territory, or under whose control, the activity has taken place, is liable. Chapter V also examines the circumstances that preclude the liability of the acting State.
- 19. Finally, chapter VI examines the issue of compensation and damages. It reviews the relevant treaty provisions and forms of State practice concerning compensable injuries and other forms of compensation. It points out that some treaties provide limitations on compensation. It examines the authorities recognized in State practice as competent to decide on compensation and reviews the enforceability of judgments awarding compensation.

#### CHAPTER I

# Activities causing injuries beyond the territorial jurisdiction or control of the State where they are conducted

#### A. Forms of activities

20. Activities causing injuries beyond the territorial jurisdiction or control of the acting State vary. They may include use of airspace, nuclear or industrial activities, conservation and utilization of economically important resources, and even communication and broadcasting. Some of these activities may cause more substantial injuries than others, and the injuries may sometimes be devastating. State practice appears to demonstrate no significant relationship between the forms of activities and the substantive or procedural requirements regulating them. However, a relationship exists between the injury or harm those activities may

cause and the substantive and procedural requirements applicable thereto, that is, the rules authorizing or not such activities and the conditions governing their performance. Ensuring compliance with the norms of procedure and observance of the concepts of due care, good-neighbourliness, etc., becomes more complex as the possibility and extent of injury resulting from the activities become more substantial. Activities causing injuries that could be devastating may be banned outright. Some nuclear testing may be included in this category, as illustrated by the 1963 Treaty Banning Nuclear Weapon Tests in the Atmosphere, in Outer Space and Under Water. Similarly, the emplacement of nuclear weapons and other weapons of mass destruction on the

sea-bed and the ocean floor and in the subsoil thereof, 11 as well as military or any hostile use of environmental modification techniques,12 have been prohibited by multilateral treaties. The treaties dealing with the two latter activities provide for monitoring or "verification" of compliance with treaty obligations by its signatories. Therefore, regardless of the similarity in the "form" of these three activities and their regulation by treaties, the actual reason for banning them is the extent of their harmful consequences, which has led to a policy decision by their signatories to ban them altogether. Sometimes the extent of injuries may not lead to a total banning of an activity, but to partial or temporary banning or to substantial revision of the form in which the activity may be carried out, as for instance in the Trail Smelter case.

21. At a very general level, injuries caused by activities beyond the territorial jurisdiction or control of the acting State may be divided into three categories. The first category covers injuries generally considered minor and expected to be tolerated among States without compensation. The second category is not generally expected to be tolerated, unless with the consent of the injured State, or against payment of compensation. The third category comprises injuries that are devastating and not generally expected to be tolerated at all. State practice shows that it is extremely difficult to identify the thresholds separating the three categories of injuries. It may be easier to pinpoint activities leading to the third category of injuries; these activities are normally banned. Treaties banning some of them refer in the preamble to the more general expectations of the community, the promotion of peace and security and other principles of the United Nations Charter. The main difficulties arise in identifying the threshold between the first and second categories of injuries, that is, in identifying the types of activities and the types of injuries concerning which the acting State has to consult the potentially injured State, or take measures to prevent the injury. It appears, so far, that State practice has not dealt with this question categorically and in a single formula. Sometimes it is the nature of the resources being used that is taken as the point of departure, such as shared rivers, the high seas or airspace. At other times, the attempt has been made to determine the expectations shared by the parties. Expectations are embodied in treaties, official correspondence and general relations between States. At the most general level, State practice, both in treaties and in judicial decisions, has referred to the concepts of good-neighbourliness, due care, equitable principles, etc., as guidelines to distinguish activities with tolerable injuries from those resulting from the second category of injuries.

22. It is not only activities that may cause extraterritorial injuries; inactivity may also lead to injuries. The 1949 judgment of the International Court of Justice in the Corfu Channel case leads to this conclusion. The decision of the German Constitutional Law Court in the

Donauversinkung case (1927), involving the States of Württemberg and Prussia versus the State of Baden and regarding their rights in the flow of the waters of the Danube, bears on the question of inactivity. Holding that "Baden must desist from injuring her neighbour" the Court further stated that Baden did "not need to eliminate the natural loss of water that would occur in the storage area even if the dam were not there, but only the augmented seepage caused by the dam". As to the prohibition by the State of Baden of measures to render it possible for the waters of the Danube to flow onwards rather than to run off to the Aach, the Court considered that Baden could not justify it on the grounds that "in this way she is only maintaining the natural conditions with respect to the water", and that while a State "is not obliged to interfere, in the interests of another State, with the natural processes affecting an international river", the action of Baden in that particular case amounted to "the neglect of any orderly work of maintenance" along that stretch of the river. The Court held that the State of Baden was "therefore required to eliminate the increased seepage caused by her inactivity"\*.13

#### (a) Multilateral agreements

- 23. Many activities with possible extraterritorial injurious consequences have been regulated by multilateral treaties. They include the use of nuclear materials, industrial activities, disposal of wastes, etc. Multilateral treaties regulating nuclear activities include the 1963 Vienna Convention on Civil Liability for Nuclear Damage, the 1962 Convention on the Liability of Operators of Nuclear Ships, the 1960 Convention on Third Party Liability in the Field of Nuclear Energy and the 1963 Treaty Banning Nuclear Weapon Tests in the Atmosphere, in Outer Space and Under Water.
- 24. The 1952 Convention on Damage caused by Foreign Aircraft to Third Parties on the Surface covers some space activities, while the 1972 Convention on International Liability for Damage caused by Space Objects deals with outer space activities.
- Some polluting activities are covered by the 1960 Convention concerning the Protection of Lake Constance against Pollution, the 1969 International Convention on Civil Liability for Oil Pollution Damage, the 1973 International Convention for the Prevention of Pollution from Ships and the 1972 Convention on the Prevention of Marine Pollution by Dumping of Wastes and other Matter. The 1982 United Nations Convention on the Law of the Sea provides in its article 195 that States "shall act so as not to transfer, directly or indirectly, damage or hazards from one area to another or transform one type of pollution into another". Article 196 of the Convention refers to pollution resulting from the use of technologies or the intentional or accidental introduction of alien or new species in a particular part of the marine environment where they might cause significant and harmful changes.

Treaty of 11 February 1971 on the Prohibition of the Emplacement of Nuclear Weapons and other Weapons of Mass Destruction on the Sea-bed and the Ocean Floor and in the Subsoil thereof.

<sup>&</sup>lt;sup>12</sup> Convention of 10 December 1976 on the Prohibition of Military or any other Hostile Use of Environment Modification Techniques.

<sup>11</sup> Württemburg and Prussia v. Baden (1927) (Entscheidungen des Reichsgerichts in Zivilsachen (Berlin, 1927), vol. 116, appendix, p. 18; Annual Digest of Public International Law Cases, 1927-1928 (London, 1931), vol. 4, p. 128, case No. 86). See also G. H. Hackworth, ed., Digest of International Law (Washington, D.C., 1940), vol. I, pp. 596-599.

- 26. Among the conventions relating to the conservation of economically important fish stocks, the 1966 International Convention for the Conservation of Atlantic Tunas and the 1949 International Convention for the Northwest Atlantic Fisheries may be named.
- 27. Conventions dealing with communications and broadcasting include the 1927 International Radiotelegraph Convention, with general regulations and additional regulations, the 1932 International Telecommunication Convention and the 1936 International Convention concerning the Use of Broadcasting in the Cause of Peace.

#### (b) Bilateral agreements

- 28. A great number of bilateral agreements relate to the utilization of lakes or rivers shared by the contracting States. Bilateral agreements may also relate to nuclear activities and materials. For example, the 1966 Convention between Belgium and France radiological protection with regard to the installations of the Ardennes nuclear power station is concerned with radiological protection in connection with the nuclear power plant at Chooz, belonging to the Société d'énergie nucléaire franco-belge des Ardennes, a jointstock company created by France and Belgium and operating in French territory near the Belgian border. Similarly, the exchange of letters of 16 July 1976 constituting an agreement between France and the USSR concerning the prevention of accidental or unauthorized use of nuclear weapons relates to the use of nuclear materials that may cause injuries to the other contracting party.
- 29. Bilateral agreements have been concluded to regulate the transport of hazardous substances and the conduct of activities affecting climate and weather. The first matter was the subject of the 1973 Agreement between Norway and the United Kingdom relating to the transmission of petroleum by pipeline from the Ekofisk Field and neighbouring areas to the United Kingdom. A similar agreement was concluded between the Federal Republic of Germany and Norway, 4 while Canada and the United States of America concluded an agreement concerning weather modification activities. 15
- 30. Some bilateral agreements deal with any activities that may have harmful consequences in the neighbouring State across the border. The most recent agreement of this kind was signed between the United States of America and Mexico on 14 August 1983.16 The preamble to that Agreement recognizes the importance of a "healthful" environment for the long-term economic and social well-being of present and future generations of each country as well as of the global community. Article 2 of the Agreement provides that the parties shall

- adopt appropriate measures to prevent, reduce and eliminate sources of pollution in their own territories which affect the border areas of the other.
- 31. Some bilateral agreements deal with the use of land close to frontier areas, as for instance the 1973 Agreement between the Federal Republic of Germany and Austria concerning regional planning.

# (c) Judicial decisions and State practice other than agreements

- 32. State practice in the regulation of pollution caused by industrial activities is evidenced, for example, by the *Trail Smelter* case, by the correspondence between the United States of America and Mexico concerning their dispute relating to the *Peyton Packing Company* and the *Casuco Company*, and the decision rendered in the *Georgia* v. *Tennessee Copper Company* case (1907).
- 33. The Nuclear Tests case, brought before the International Court of Justice, the claims made against the United States of America following the Eniwetok Atoll tests and against the United Kingdom following the Christmas Island tests, as well as the Canadian claim against the USSR for damage caused by the Soviet satellite Cosmos 954, deal with nuclear activities.
- 34. Some of the judicial decisions dealing with utilization of international rivers are the Lake Lanoux case, the Société d'énergie électrique du littoral méditerranéen v. Compania Imprese Elettriche Ligure (the Roya case) (1939), and the Missouri v. Illinois (1906) and Kansas v. Colorado (1907) cases.
- 35. The Fisheries case (United Kingdom v. Norway) and the Fisheries Jurisdiction cases (United Kingdom v. Iceland; Federal Republic of Germany v. Iceland) deal with fishery activities, whereas the North Sea Continental Shelf cases, the 1959 arbitration between Petroleum Development (Trucial Coast) Ltd. and the Sheikh of Abu Dhabi, and the Continental Shelf case (Tunisia v. Libyan Arab Jamahiriya) relate to use of the ocean subsoil by the coastal States.
- 36. Other activities forming the subject matter of State practice that may be cited are the development of power plants (the *Roya* case), counterfeiting (*United States* v. *Arjona*), and highway construction (diplomatic correspondence between the United States of America and Mexico concerning the *Smugglers* and *Goat Canyons*).

#### B. Location of origin of activities

37. Activities conducted by the acting State or its nationals with injurious consequences for other States and their nationals may occur within or beyond the territorial jurisdiction or control of the acting State. They may occur in the shared domain, but cause injury to another State or its nationals either in the shared domain or within the territorial jurisdiction or under the control of the injured State. Activities may also occur within the territorial jurisdiction or under the control of the injured State itself. Although the location of activities with injurious impact is relevant, it is not the key factor in regulating them. The location of activities appears to bring into play other relevant and competing in-

<sup>&#</sup>x27;Agreement of 16 January 1974 between the Federal Republic of Germany and Norway relating to the transmission of petroleum by pipeline from the Ekofisk Field and neighbouring areas to the Federal Republic of Germany.

<sup>&</sup>lt;sup>15</sup> Agreement of 26 March 1975 between Canada and the United States of America relating to the exchange of information on weather modification activities.

<sup>&</sup>lt;sup>16</sup> Agreement of 14 August 1983 between the United States of America and Mexico on co-operation for the protection and improvement of the environment in the border area.

terests and relevant principles of international law; for example, if an activity occurs on the high seas, the interests and rights of the acting State concern the utilization of resources, including the waters of the high seas, and the relevant principle of international law is that of freedom of the high seas. The location of origin of activities may also determine the question of jurisdiction over any possible dispute regarding the consequences of the activities. State practice demonstrates that the key issue in regulating, substantively or procedurally, an activity with injurious consequences is the extent and kind of injury it causes and its impact on the functioning of relations between States, regardless of the location of origin of the activity.

- 38. The present section is primarily descriptive. It recapitulates the relevant parts of treaties and judicial decisions bearing on activities occurring within or beyond the territorial jurisdiction or control of the acting State, but causing injuries to other States or their nationals.
  - 1. ACTIVITIES CONDUCTED WITHIN THE TERRITORIAL JURISDICTION OR UNDER THE CONTROL OF THE ACTING STATE
- 39. Most of the activities occurring within the territorial jurisdiction or under the control of one State and causing injuries to neighbouring States relate to the use of resources shared by two or more neighbouring States or to activities close to the frontier.

#### (a) Multilateral agreements

- 40. The 1960 Convention concerning the Protection of Lake Constance against Pollution deals with shared resources. Under article 1, paragraph 2, of the Convention, the riparian States are to take the necessary measures in their respective territories to prevent any increase in the pollution of Lake Constance and, so far as possible, to improve the quality of its waters. To that end, the riparian States are to apply strictly, in respect of Lake Constance and its affluents, all the provisions on water protection that are in force in their territories.
- The 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden is a treaty between neighbouring States, but relates to a wider group of activities. Article 1 of the Convention defines "environmentally harmful" activities as being the discharge from the soil or from buildings or installations of solid or liquid waste, gas or any other substance into watercourses, lakes or the sea and the use of land, the sea-bed, buildings or installations in any other way which entails, or may entail, environmental nuisance by water pollution or any other effect on water conditions, sand drift, air pollution, noise, vibration, changes in temperature, radiation, light, etc. The Protocol to the Convention states that discharge from the soil or from buildings or installations of solid or liquid waste, gases or other substances into watercourses, lakes or the sea shall be regarded as environmentally harmful activities only if the discharge entails or may entail a nuisance to the surroundings. Therefore the mere discharge of "polluting" substances is *not* sufficient to bring it under the régime of the Convention.

- 42. Acts covered by the 1960 Convention on Third Party Liability in the Field of Nuclear Energy and the 1963 Vienna Convention on Civil Liability for Nuclear Damage may also be included in the category of activities occurring, most probably, within the territorial jurisdiction or under the control of a State but causing extraterritorial harmful effects. The parties to these conventions, however, are not neighbouring States of a particular region; the conventions are open to all States.
- 43. The language used in the 1982 United Nations Convention on the Law of the Sea is more ambiguous on the location of origin of activities. Article 195 of the Convention provides that, "in taking measures to prevent, reduce and control pollution of the marine environment", States shall behave in certain ways. Thus the location of origin of polluting activities may be within or beyond the territorial jurisdiction or control of States.

#### (b) Bilateral agreements

44. The location of most of the activities regulated by bilateral agreements is within the territorial jurisdiction or under the control of the States parties to these agreements. A number of bilateral agreements relate to the use of a resource shared by two States, such as rivers. In this group of agreements, activities may occur in the section of the shared resource which is within the territorial jurisdiction of either State or within the section of the resource shared by both States. Most bilateral agreements, however, deal with activities occurring within the territorial jurisdiction or under the control of one State. For example, in the 1949 Agreement between Norway and the Soviet Union, '' the parties agree not to exploit the mineral deposits near their frontiers in a way that may harm their respective territories. Thus they agree, in order to safeguard the frontier line, to have a belt 20 metres wide on either side within which no such activity may take place, unless in exceptional cases and by agreement between the parties. Article 18 of the Agreement provides:

#### Article 18

- 1. Mineral deposits near the frontier line may not be so prospected or worked as to harm the territory of the other Party.
- 2. In order to safeguard the frontier line, there shall be a belt 20 metres wide on either side thereof in which the work referred to in paragraph 1 of this article shall ordinarily be prohibited and shall be permitted only in exceptional cases by agreement between the competent authorities of the Contracting Parties.
- 3. If in any particular case it is not expedient to observe the belts referred to in paragraph 2 of this article, the competent authorities of the Contracting Parties shall agree on other measures necessary to safeguard the frontier line.
- 45. Some bilateral agreements deal with activities occurring within the territorial jurisdiction of the "injured" State. For example, under the 1967 Agreement between the Federal Republic of Germany and Austria, 18 the Federal Republic of Germany agrees to

<sup>&</sup>lt;sup>17</sup> Agreement of 29 December 1949 between Norway and the USSR concerning the regime of the Norwegian-Soviet frontier and procedure for settlement of frontier disputes and incidents.

<sup>&</sup>lt;sup>18</sup> Agreement of 19 December 1967 between the Federal Republic of Germany and Austria concerning the effects on the territory of the Federal Republic of Germany of construction and operation of the Salzburg Airport.

establish a safety zone in its own territory for an airport to be established in Salzburg (Austria). Hence, activities that may cause injuries in the territory of the Federal Republic of Germany may be caused in that territory but not necessarily by nationals of that State. The injury may be the result of the operations of the Austrian airport.

46. The passage of nuclear ships to or from foreign ports is the subject of bilateral agreements for the prevention of nuclear or other kinds of damage. These treaties approach the question of territorial jurisdiction or control functionally. Accordingly, they are relevant to nuclear damage occurring within the territory of the host State if the nuclear incident has occurred within that territory. For example, under article 20 of the 1970 Treaty between Liberia and the Federal Republic of Germany, 19 liability under the Treaty shall apply to nuclear damage occurring within Liberian territory or Liberian waters if the nuclear incident has occurred within Liberian territory or Liberian waters. And article VIII of the 1964 Agreement between the United States of America and Italy on the use of Italian ports by the N.S. Savannah20 stipulates that the United States is liable for "any damage to people or goods deriving from a nuclear incident in which the N.S. Savannah may be involved within Italian territorial waters". The agreements concluded by the United States of America with Ireland21 and with the Netherlands22 contain similar provisions.

# (c) Judicial decisions and State practice other than agreements

47. Judicial decisions and official correspondence relating to this group of activities stem from conflicts primarily between neighbouring States in relation to the use of resources shared by them, such as rivers and airspace. The sources point to a broad range of activities taking place in the territory of the acting State or under its control which may cause injury to other States and

#### "Article 7

their nationals. For example, the tribunal in the Lake Lanoux case stated that pollution of waters, changed chemical composition or temperature of waters, and diminution of the volume of water flow resulting from the use by one State of international waters within its borders could violate the rights of the affected State and give rise to a "duty of care" in carrying out the activity. In even broader language, the tribunal in the Trail Smelter case stated that

... no State has the right to use or permit the use of its territory in such a manner as to cause injury by fumes in or to the territory of another or the properties or persons therein.<sup>23</sup>

Even more generally, the International Court of Justice, in its judgment of 19 April 1949 in the *Corfu Channel* case, stated that it was "every State's obligation not to allow knowingly its territory to be used for acts contrary to the rights of other States".<sup>24</sup>

- 48. An activity originating within the territory of the acting State but not relating to the use of resources shared by two neighbouring States is the launching of satellites. For example, in a note addressed to the Soviet Union in January 1979, Canada argued the liability of that country following the crash of a Soviet nuclear-powered satellite, Cosmos 954, on Canadian soil.
- 49. In the Alabama case (1872), the United States of America sought compensation for injuries resulting from the building and outfitting, in British ports, of Confederate ships which were permitted to leave those ports in breach of Britain's duty of neutrality.

# 2. ACTIVITIES CONDUCTED OUTSIDE THE TERRITORIAL JURISDICTION OR CONTROL OF THE ACTING STATE

#### (a) Multilateral agreements

50. A number of multilateral agreements regulate activities occurring beyond the territorial jurisdiction or control of acting States, but causing injuries to other States and their nationals either in the shared domain or within the territorial jurisdiction of the injured State. Several of the treaties cited in this section deal with nuclear materials. Article XIII of the 1962 Convention on the Liability of Operators of Nuclear Ships states that the Convention applies to nuclear damage caused by a nuclear incident occurring in any part of the world and involving the nuclear fuel of or radioactive products or waste produced in a nuclear ship flying the flag of a contracting State. Thus, according to this broad definition, a damage-causing nuclear incident may occur within or beyond the territorial jurisdiction or control of States. Article XI, paragraph 2, of the 1963 Vienna Convention on Civil Liability for Nuclear Damage, in an attempt to specify the authority competent to decide on the issue of liability, refers to the location of origin of the activity. It states that, where the nuclear incident occurs outside the territory of any contracting party, or where the place of the nuclear incident cannot be determined with certainty, jurisdiction over such actions

<sup>&</sup>lt;sup>19</sup> Treaty of 27 May 1970 between Liberia and the Federal Republic of Germany on the use of Liberian waters and ports by N.S. [nuclear ship] Otto Hahn.

<sup>&</sup>lt;sup>20</sup> Agreement of 23 November 1964 between the United States of America and Italy on the use of Italian ports by the N.S. [nuclear ship] Savannah. See also the exchange of notes of 16 December 1965 constituting an agreement between the United States of America and Italy concerning liability during private operation of the N.S. Savannah.

<sup>&</sup>lt;sup>21</sup> Exchange of notes of 18 June 1964 constituting an agreement between the United States of America and Ireland relating to public liability for damage caused by the N.S. Savannah. The agreement provides (note I):

"(1) The United States Government shall provide compensation

<sup>&</sup>quot;(1) The United States Government shall provide compensation for all loss, damage, death or injury in Ireland (including Irish territorial seas) arising out of or resulting from the operating of N.S. Savannah to the extent that the United States Government, the United States Maritime Administration or a person indemnified under the Indemnification Agreement is liable for public liability in respect of such loss, damage, death or injury."

<sup>&</sup>lt;sup>22</sup> Agreement of 6 February 1963 between the Netherlands and the United States of America on public liability for damage caused by the N.S. Savannah, which provides:

<sup>&</sup>quot;This Agreement relates only to a nuclear incident occurring during a voyage of the N.S. Savannah to or from the Netherlands or its presence in Netherlands waters."

See also the Operational Agreement of 20 May 1963 on arrangements for a visit of the N.S. Savannah to the Netherlands.

<sup>&</sup>lt;sup>23</sup> United Nations, Reports of International Arbitral Awards, vol. III, p. 1965.

<sup>24</sup> I.C.J. Reports 1949, p. 22.

shall lie with the courts of the State in which the liable operator is established.

- 51. The 1972 Convention on the Prevention of Marine Pollution by Dumping of Wastes and other matter regulates certain aspects of activities relating to the use of the sea on the assumption that such uses, if not regulated, will cause injury to a number of coastal States. Sometimes particular activities, including the use of resources beyond the territorial jurisdiction or control of States, have a noticeable economic impact on other States and their nationals. These activities have also been regulated by multilateral conventions. For example, the exploitation of certain resources of the sea may fall into this category. Some of the conventions dealing with the exploitation of sea resources bear on the conservation of certain fishery resources which have strong economic implications. Thus they differ from conventions relating to general conservation; they deal with resources that affect the interests of coastal States in a much more quantitative, tangible, immediate and economic form. In the preamble to the 1966 International Convention for the Conservation of Atlantic Tunas, the parties explicitly recognize their "mutual interest" in the populations of tuna and tuna-like fish found in the Atlantic Ocean, and in maintaining those populations at levels that will permit the maximum sustainable catch for nutritional and other purposes. Similarly, in the preamble to the 1949 International Convention for the Northwest Atlantic Fisheries, the parties explicitly recognize their shared interest in the conservation of the fishery resources of the north-west Atlantic Ocean.
- The 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties provides in article I that parties to the Convention may take such measures on the high seas as may be necessary to prevent, mitigate or eliminate grave and imminent danger to their coastlines or related interests from pollution or threat of pollution of the sea by oil, following upon a maritime casualty. Such incidents on the high seas nearly always cause injuries to at least the flag State. Article VI of the Convention provides that, if the measures taken by the coastal State go beyond what is necessary to prevent the injury, that State shall be obliged to pay compensation to the extent of the damage caused by measures exceeding those that are reasonably necessary. In considering whether the measures are proportionate to the damage, article V provides that account shall be taken of: (a) the extent and probability of imminent damage if those measures are not taken; (b) the likelihood of those measures being

effective; (c) the extent of the damage that may be caused by such measures. Hence any party which takes measures in contravention of these requirements and causes damage to others shall be obliged to pay compensation. Article 1 of the 1963 Treaty Banning Nuclear Weapon Tests in the Atmosphere, in Outer Space and Under Water prohibits nuclear explosions at any place if such explosion causes radioactive debris to be present outside the territorial limits of the State under whose jurisdiction or control such explosion is conducted. In that connection, the 1971 Treaty on the Prohibition of the Emplacement of Nuclear Weapons and Other Weapons of Mass Destruction on the Sea-bed and the Ocean Floor and in the Subsoil thereof should also be mentioned. The 1976 Convention on the Prohibition of Military or any other Hostile Use of Environmental Modification Techniques deals with techniques of this type which might occur either within or beyond the territorial jurisdiction or control of the acting State.

#### (b) Bilateral agreements

53. In agreements regarding the use of foreign ports by nuclear ships, the State whose nuclear ship is visiting the foreign ports has accepted liability for injuries its ships may cause outside the territory of the host State during a passage to or from its port if the damage is caused to the host State or to ships of the host State registry.<sup>25</sup>

# (c) Judicial decisions and State practice other than agreements

54. Although almost all the judicial decisions and official correspondence dealing with questions of extraterritorial injuries surveyed in this study relate to activities occurring within the territorial jurisdiction or under the control of a State, at least one decision bears on activities occurring in the shared domain. In the *Fisheries Jurisdiction* cases,<sup>26</sup> the United Kingdom and the Federal Republic of Germany objected to the unilateral expansion of the fishery zone by Iceland, which, they claimed, had been extended to the high seas.

#### CHAPTER II

#### Assessment of activities for their injurious impact

55. The "assessment of activities for their injurious impact" referred to in this study involves a continuous process that begins prior to, but may continue during the performance of activities with potentially injurious impact in order to prevent or minimize injuries to other

States and their nationals. Such assessment comprises different stages in which a variety of interests are evaluated and accommodated, and choices and changes made. Although the expression "assessment of activities" is used in this study, the content and conduct of the process are to be found under other headings in

<sup>&</sup>lt;sup>23</sup> See e.g. the Treaty of 27 May 1970 between Liberia and the Federal Republic of Germany, which provides, in article 20, that the Federal Republic of Germany shall be liable for injuries its nuclear ship may cause "outside Liberian territory or Liberian waters during a passage to or from a Liberian port or to or from Liberian waters". See also footnotes 20 and 22 above and the similar agreements concluded by the United States of America with Italy and the Netherlands concerning the N.S. Savannah.

<sup>26</sup> I.C.J. Reports 1974, pp. 3 and 175.

many treaties, judicial decisions and official correspondence between States, although not always systematically and step by step. The unsystematic references to the procedures and stages of the assessment of activities in treaties or judicial decisions are primarily determined by the main purposes of the treaty or by the questions posed for judicial decision. Sometimes one or more aspects of assessment procedures may be irrelevant to a particular activity. For example, in the case of the prohibition of emplacement of nuclear weapons on the high seas, or of the hostile use of environmental modification techniques, assessment procedures such as collection of data, exchange of information and consultation, etc. are totally irrelevant. The only stage of assessment that may be relevant and is stipulated in the two treaties dealing with these two activities is that of monitoring. Sometimes the procedural requirements for assessing activities for their injurious impact prior to or during their undertaking have been eliminated in agreements. States have made a policy decision that the performance of these activities is essential regardless of their harmful impact, as is apparent from most treaties dealing with shipping. The basic thrust of these treaties is to determine liability and to provide compensation for injuries these activities may cause.

#### (a) Multilateral agreements

56. The requirement that States assess the injurious impact of their activities is reflected in article 192 of the 1982 United Nations Convention on the Law of the Sea, which provides: "States have the obligation to protect and preserve the marine environment." The language of article 194, paragraph 2, is more explicit. It requires States to take "all measures" necessary to prevent damage resulting from activities under their jurisdiction or control to other States and their environment:

## Article 194. Measures to prevent, reduce and control pollution of the marine environment

- 2. States shall take all measures necessary to ensure that activities under their jurisdiction or control are so conducted as not to cause damage by pollution to other States and their environment, and that pollution arising from incidents or activities under their jurisdiction or control does not spread beyond the areas where they exercise sovereign rights in accordance with this Convention.
- 57. With regard to activities concerning resource deposits in the Area which extend beyond the limits of national jurisdiction, article 142 of the Convention requires that the acting State, when exploiting the deposits, take due account of the rights and interests of the coastal State. Paragraph 1 of this article reads:

#### Article 142. Rights and legitimate interests of coastal States

1. Activities in the Area, with respect to resource deposits in the Area which lie across limits of national jurisdiction, shall be conducted with due regard to the rights and legitimate interests of any coastal State\* across whose jurisdiction such deposits lie.

Moreover, part XII of the Convention elaborates on the requirement of assessing the injurious impact of activities. Sections 1 to 4 of part XII in particular deal primarily with the detailed steps of impact assessment as set forth in this study.

58. Two multilateral treaties regarding communications systems require their signatories to use their communications installations in ways that will not interfere with the facilities of other States parties. Article 10, paragraph 2, of the 1927 International Radiotelegraph Convention requires the parties to the Convention to operate stations in such a manner as not to interfere with the radioelectric communications of other contracting States or of persons authorized by those Governments:

Article 10. Conditions to be observed by stations. Interference

- 2. All stations, whatever their object may be, must, so far as possible, be established and operated in such manner as not to interfere with the radioelectric communications or services of other contracting Governments and of individual persons or private enterprises authorized by those contracting Governments to conduct a public radiocommunication service.
- 59. The 1932 International Telecommunication Convention contains a similar requirement:

#### Article 35. Interference

- 1. All stations, whatever their object may be, must, so far as possible, be established and operated in such manner as not to interfere with the radioelectric communications or services of other Contracting Governments, or of private enterprises recognized by those Contracting Governments or other duly authorized enterprises which conduct a radiocommunication service.
- 2. Each of the Contracting Governments not itself operating systems of radiocommunication undertakes to require private enterprises which it recognizes and other enterprises duly authorized for that purpose to observe the provisions of paragraph 1 above.

Again, the 1936 International Convention concerning the Use of Broadcasting in the Cause of Peace prohibits the broadcasting to another State of material designed to incite the population to act in a manner incompatible with the internal order and security of that State. It provides:

#### Article 1

The High Contracting Parties mutually undertake to prohibit and, if occasion arises, to stop without delay the broadcasting within their respective territories of any transmission which to the detriment of good international understanding is of such a character as to incite the population of any territory to acts incompatible with the internal order or the security of a territory of a High Contracting Party.

60. Article 12 of the 1983 Convention for the Protection and Development of the Marine Environment of the Wider Caribbean Region requires the contracting parties to develop technical and other guidelines to assist them in assessing the environmental impact of their development projects upon the area covered by the Convention. The assessment should bear in particular on the effects of those projects upon coastal areas. Under this article, each contracting State shall, when requested, submit information concerning its development programme and the potential consequences thereof. Where appropriate, a State may engage in consultations with other contracting States which may be affected by the impact of its activities. This article reads:

#### Article 12. Environmental impact assessment

1. As part of their environmental management policies the Contracting Parties undertake to develop technical and other guidelines to assist the planning of their major development projects in such a way as to prevent or minimize harmful impacts on the Convention area.

- 2. Each Contracting Party shall assess within its capabilities, or ensure the assessment of, the potential effects of such projects on the marine environment, particularly in coastal areas, so that appropriate measures may be taken to prevent any substantial pollution of, or significant and harmful changes to, the Convention area.
- 3. With respect to the assessments referred to in paragraph 2, each Contracting Party shall, with the assistance of the Organization when requested, develop procedures for the dissemination of information and may, where appropriate, invite other Contracting Parties which may be affected to consult with it and to submit comments.

#### (b) Bilateral agreements

- 61. Since bilateral agreements are primarily directed to the specific use of a particular resource, their provisions, including those relating to impact assessment, also appear to be more specific. For example, they may simply prohibit certain specific activities. Nevertheless, these provisions are designed to protect the interests of both parties in security, economic or social matters. Thus article 3 of the 1922 Convention between Finland and the RSFSR<sup>27</sup> prohibits the diversion of certain watercourses, the erection of constructions or the adoption of measures that might affect the flow of water by altering the existing depth or condition of the parts of the watercourse situated in the territory of the other contracting State, thereby damaging the fairway or encroaching upon channels used for navigation or timberfloating, except by special agreement between the contracting States. This article does not prohibit certain activities, but specific results irrespective of the activities themselves.
- 62. Occasionally, the provisions relating to impact assessment may be more general, not relating to any specific activity or outcome. Thus, under article 28, paragraph 1, of the 1963 Treaty between Hungary and Romania, 28 the contracting parties are required to undertake forestry activities in the vicinity of their frontiers in such a way as not to impair the forest economy of the other party:

#### Article 28

1. Each Contracting Party shall so conduct its forestry operations in the vicinity of the frontier as not to impair the forest economy of the other Party.

Again, article 1 of the 1973 Agreement between the Federal Republic of Germany and Austria<sup>29</sup> establishes a German-Austrian Land Use Commission to facilitate co-operation in matters of land use, particularly in areas adjacent to their common frontier:

#### Article I

With a view to furthering and facilitating co-operation in matters of land use, particularly as regards areas adjacent to the common frontier, there shall be established a German-Austrian Land Use Commission (hereinafter referred to as "the Commission").

<sup>27</sup> Convention of 28 October 1922 between Finland and the Russian Socialist Federal Soviet Republic concerning the maintenance of river channels and the regulation of fishing on watercourses forming part of the frontier between Finland and Russia.

Such co-operation would obviously entail consultation between the parties or through the Commission regarding land use in the frontier areas.

63. Sometimes the entire bilateral agreement may focus on the assessment of the impact of any activity that has transboundary effects. The 1983 Agreement between Mexico and the United States of America<sup>30</sup> may be cited as an example. In addition to the provisions of the preamble referred to above (para. 30), article 1 of the Agreement provides that co-operation among the parties shall be based on equality, reciprocity and mutual benefit:

#### Article I

The United States of America and the United Mexican States, hereinafter referred to as the Parties, agree to co-operate in the field of environmental protection in the border area on the basis of equality, reciprocity and mutual benefit. The objectives of the present Agreement are to establish the basis for co-operation between the Parties for the protection, improvement and conservation of the environment and the problems which affect it, as well as to agree on necessary measures to prevent and control pollution in the border area, and to provide the framework for development of a system of notification for emergency situations. Such objectives shall be pursued without prejudice to the co-operation which the Parties may agree to undertake outside the border area.

64. Bilateral agreements have also been concluded for the safeguard of frontier lines and the protection of the security interests of the parties. For example, article 18 of the 1949 Agreement between Norway and the Soviet Union requires the parties to maintain a belt 20 metres wide on either side of their frontier within which no activity for exploitation of mineral deposits may take place unless by agreement between the two States (see para. 44 above).

# (c) Judicial decisions and State practice other than agreements

- 65. The general requirement that States must assess the injurious impact of activities undertaken by them or by persons under their control was stated in the *Trail Smelter* case. The tribunal observed that "no State has the right to use or permit the use of its territory in such a manner as to cause injury by fumes in or to the territory of another or the properties or persons therein" (see para. 47 above). The tribunal established a rather precise and comprehensive régime, which included assessment of the injurious impact of smelting activities occurring within the acting State but causing extraterritorial injuries.
- 66. A more exacting requirement of State assessment of activities conducted under a State's territorial control was laid down in the judgment of 9 April 1949 of the International Court of Justice in the Corfu Channel case (merits). In that case, the United Kingdom sought indemnity for damage to one of its ships which had struck a mine in the Corfu Channel. The author of the mine-laying remained unknown. None the less, the Court found that Albania was responsible for the damage occurring within its territorial waters:

From all the facts and observations mentioned above, the Court draws the conclusion that the laying of the minefield which caused the explosions on October 22nd, 1946, could not have been accomplished without the knowledge of the Albanian Government.

<sup>&</sup>lt;sup>28</sup> Treaty of 13 June 1963 between Hungary and Romania concerning the régime of the Hungarian-Romanian State frontier and cooperation in frontier matters.

<sup>&</sup>lt;sup>29</sup> Agreement of 11 December 1973 between the Federal Republic of Germany and Austria concerning co-operation with respect to land use

<sup>30</sup> See footnote 16 above.

The obligations resulting for Albania from this knowledge are not disputed between the Parties.\* Counsel for the Albanian Government expressly recognized that "if Albania had been informed of the operation before the incidents of October 22nd, and in time to warn the British vessels and shipping in general of the existence of mines in the Corfu Channel, her responsibility would be involved . . . ".

The obligations incumbent upon the Albanian authorities consisted in notifying, for the benefit of shipping in general, the existence of a minefield in Albanian territorial waters and in warning the approaching British warships of the imminent danger to which the minefield exposed them. Such obligations are based, not on the Hague Convention of 1907, No. VIII, which is applicable in time of war, but on certain general and well-recognized principles, namely: elementary considerations of humanity, even more exacting in peace than in war: the principle of the freedom of maritime communication; and every State's obligation not to allow knowingly its territory to be used for acts contrary to the rights of other States.\*

67. From the language of the judgment it appears that the standard of "due care" which a State must maintain as regards activities by other international actors on its territory is *at least* that of *non-negligence* in the assessment of injurious impact:

It is clear that knowledge of the minelaying cannot be imputed to the Albanian Government by reason merely of the fact that a minefield discovered in Albanian territorial waters caused the explosions of which the British warships were the victims. It is true, as international practice shows, that a State on whose territory or in whose waters an act contrary to international law has occurred, may be called upon to give an explanation. It is also true that a State cannot evade such a request by limiting itself to a reply that it is ignorant of the circumstances of the act and of its authors.\* The State may, up to a certain point, be bound to supply particulars of the use made by it of the means of information and inquiry at its disposal. . . . <sup>32</sup>

The Court recognized that, from the mere fact of the control exercised by a State over its territory and waters, it could not be concluded that that State had known or ought to have known of any wrongdoing perpetrated therein. That fact, the Court concluded, by itself and apart from other circumstances, did not prima facie involve responsibility, nor did it shift the burden of proof. On the other hand, the Court recognized that the exclusive control by a State over its territory had a bearing upon the methods of proof available to establish the knowledge by the State of such events. By reason of this exclusive control, the injured State was often unable to furnish direct proof of facts giving rise to responsibility. The injured State should therefore be allowed "a more liberal recourse to inferences of fact and circumstantial evidence". "According to the Court, this form of evidence was admitted in all systems of law and was recognized by international law. It should further be regarded "as of special weight when it is based on a series of facts linked together and leading logically to a single conclusion".34 Recourse to a very liberal interpretation and acceptance of evidence regarding the knowledge by the State of injurious acts carried out by other entities appears to have been recognized.

#### A. Data collection

69. Collecting data on the possible effect of activities with potentially injurious consequences is the first step

in the impact assessment process. It requires serious consideration, in good faith, of the interests of others. This early stage of assessment includes gathering scientific information about the kind and extent of injuries which an activity may cause to other States or their nationals. Collection of data may be undertaken by the acting State alone, by a joint commission or by a group of States. Thus collection of data may be required with respect to the impact of activities on shared domains, and to the level of possible injuries to other States and their nationals.

#### (a) Multilateral agreements

70. Some multilateral agreements provide that the data may be collected by States individually. Article XI of the 1978 Kuwait Regional Convention for Cooperation on the Protection of the Marine Environment from Pollution explicitly requires States to assess the potential injuries to the marine environment that any of their activities undertaken within their territory may cause:

#### Article XI. Environmental assessment

- (a) Each Contracting State shall endeavour to include an assessment\* of the potential environmental effects in any\* planning activity\* entailing projects within its territory, particularly in the coastal areas, which may cause significant risks of pollution in the Sea Area:
- (b) The Contracting States may, in consultation with the secretariat, develop procedures for dissemination of information of the assessment of the activities referred to in paragraph (a) above;
- (c) The Contracting States undertake to develop, individually or jointly, technical and other guidelines in accordance with standard scientific practice to assist the planning of their development projects in such a way as to minimize their harmful impact on the marine environment. In this regard international standards may be used where appropriate.

This article does not seem to be concerned about injuries to a specific State, but rather about injuries to a designated area in the Gulf waters (sea area) shared by the contracting States.

71. The 1979 Convention on Long-range Transboundary Air Pollution provides for research and exchange of information and an examination of the impact of activities undertaken by the parties to the Convention. This Convention is primarily concerned with the *prevention* and *minimizing* of injury; it is not concerned with the question of liability. Articles 3 and 4 of the Convention provide:

#### Article 3

The Contracting Parties, within the framework of the present Convention, shall by means of exchanges of information, consultation, research and monitoring, develop without undue delay policies and strategies which shall serve as a means of combating the discharge of air pollutants, taking into account efforts already made at national and international levels.

#### Article 4

The Contracting Parties shall exchange information on and review their policies, scientific activities and technical measures aimed at combating, as far as possible, the discharge of air pollutants which may have adverse effects, thereby contributing to the reduction of air pollution including long-range transboundary air pollution.

Article 7 of the Convention deals with co-operation among member States in research and development of methods for reducing air pollution and its long-range transmission:

<sup>31</sup> I.C.J. Reports 1949, p. 22.

<sup>32</sup> *Ibid.*, p. 18.

<sup>33</sup> Ibid.

<sup>34</sup> Ibid.

#### Article 7

The Contracting Parties, as appropriate to their needs, shall initiate and co-operate in the conduct of research into and/or development of:

- (a) existing and proposed technologies for reducing emissions of sulphur compounds and other major air pollutants, including technical and economic feasibility, and environmental consequences;
- (b) instrumentation and other techniques for monitoring and measuring emission rates and ambient concentrations of air pollutants;
- (c) improved models for a better understanding of the transmission of long-range transboundary air pollutants;
- (d) the effects of sulphur compounds and other major air pollutants on human health and the environment, including agriculture, forestry, materials, aquatic and other natural ecosystems and visibility, with a view to establishing a scientific basis for dose/effect relationships designed to protect the environment;
- (e) the economic, social and environmental assessment of alternative measures for attaining environmental objectives including the reduction of long-range transboundary air pollution;
- (f) education and training programmes related to the environmental aspects of pollution by sulphur compounds and other major air pollutants.

Under article 8 of the Convention, the parties are required to exchange data and information on emissions of pollutants at agreed intervals, on major changes in national policies and in industrial development, with their potential impact, and on meteorological and physico-chemical factors:

#### Article 8

The Contracting Parties, within the framework of the Executive Body referred to in article 10 and bilaterally, shall, in their common interests, exchange available information on:

- (a) data on emissions at periods of time to be agreed upon, of agreed air pollutants, starting with sulphur dioxide, coming from gridunits of agreed size; or on the fluxes of agreed air pollutants, starting with sulphur dioxide, across national borders, at distances and at periods of time to be agreed upon;
- (b) major changes in national policies and in general industrial development, and their potential impact, which would be likely to cause significant changes in long-range transboundary air pollution;
- (c) control technologies for reducing air pollution relevant to long-range transboundary air pollution;
- (d) the projected cost of the emission control of sulphur compounds and other major air pollutants on a national scale;
- (e) meteorological and physico-chemical data relating to the processes during transmission;
- (f) physico-chemical and biological data relating to the effects of long-range transboundary air pollution and the extent of the damage\* which these data indicate can be attributed to long-range transboundary air pollution;
- (g) national, subregional and regional policies and strategies for the control of sulphur compounds and other major air pollutants.
- The present Convention does not contain a rule on State liability for damage.

Subparagraphs (e), (f), (g) and (h) of article 9 of the Convention again deal with data collection and exchange of information:

(e) the need to exchange data on emissions at periods of time to be agreed upon, of agreed air pollutants, starting with sulphur dioxide, coming from grid-units of agreed size; or on the fluxes of agreed air pollutants, starting with sulphur dioxide, across national borders, at distances and at periods of time to be agreed upon. The method, including the model, used to determine the fluxes, as well as the method, including the model, used to determine the transmission of air pollutants based on the emissions per grid-unit, shall be made available and periodically reviewed, in order to improve the methods and the models:

- (f) their willingness to continue the exchange and periodic updating of national data on total emissions of agreed air pollutants, starting with sulphur dioxide;
- (g) the need to provide meteorological and physico-chemical data relating to processes during transmission;
- (h) the need to monitor chemical components in other media such as water, soil and vegetation, as well as a similar monitoring programme to record effects on health and environment.
- 72. Some multilateral agreements have established commissions designed, among other things, to carry out research and collect data. Thus article III of the 1966 International Convention for the Conservation of Atlantic Tunas establishes a Commission whose duties include the study of the effect of human and natural factors on the abundance of tuna and tuna-like fish in the areas covered by the Convention. In undertaking such a study, the Commission is not obliged to use only information supplied by member States; it may conduct its independent research studies and use the research conducted by and the services of private organizations or individuals. Article IV of the Convention, defining the functions of the Commission, reads:

#### Article IV

- 1. In order to carry out the objectives of this Convention the Commission shall be responsible for the study of the populations of tuna and tuna-like fishes (the Scombriformes with the exception of the families Trichiuridae and Gempylidae and the genus Scomber) and such other species of fishes exploited in tuna fishing in the Convention area as are not under investigation by another international fishery organization. Such study shall include research on the abundance, biometry and ecology of the fishes; the oceanography of their environment; and the effects of natural and human factors upon their abundance. The Commission, in carrying out these responsibilities shall, insofar as feasible, utilize the technical and scientific services of, and information from, official agencies of the Contracting Parties and their political subdivisions and may, when desirable, utilize the available services and information of any public or private institution, organization or individual, and may undertake within the limits of its budget independent research to supplement the research work being done by governments, national institutions or other international organizations.
- 2. The carrying out of the provisions in paragraph 1 of this article shall include:
- (a) collecting and analysing statistical information relating to the current conditions and trends of the tuna fishery resources of the Convention area;
- (b) studying and appraising information concerning measures and methods to ensure maintenance of the populations of tuna and tunalike fishes in the Convention area at levels which will permit the maximum sustainable catch and which will ensure the effective exploitation of these fishes in a manner consistent with this catch;
- (c) recommending studies and investigations to the Contracting Parties;
- (d) publishing and otherwise disseminating reports of its findings and statistical, biological and other scientific information relative to the tuna fisheries of the Convention area.

Similar responsibilities were envisaged for the Commission established under the 1949 International Convention for the Northwest Atlantic Fisheries. Article VI of the Convention reads:

#### Article VI

1. The Commission shall be responsible in the field of scientific investigation for obtaining and collating the information necessary for maintaining those stocks of fish which support international fisheries in the Convention area and the Commission may, through or in collaboration with agencies of the Contracting Governments or other public or private agencies and organizations or, when necessary, independently:

(a) make such investigations as it finds necessary into the abundance, life history and ecology of any species of aquatic life in any part of the Northwest Atlantic Ocean;

(b) collect and analyse statistical information relating to the current conditions and trends of the fishery resources of the Northwest Atlantic Ocean:

(c) study and appraise information concerning the methods for maintaining and increasing stocks of fish in the Northwest Atlantic Ocean:

(d) hold or arrange such hearings as may be useful or essential in connection with the development of complete factual information necessary to carry out the provisions of this Convention;

(e) conduct fishing operations in the Convention area at any time for purposes of scientific investigation;

(f) publish and otherwise disseminate reports of its findings and statistical, scientific and other information relating to the fisheries of the Northwest Atlantic Ocean as well as such other reports as fall within the scope of this Convention.

- 2. Upon the unanimous recommendation of each Panel affected, the Commission may alter the boundaries of the sub-areas set out in the annex. Any such alteration shall forthwith be reported to the Depositary Government which shall inform the Contracting Governments, and the sub-areas defined in the annex shall be altered accordingly.
- 3. The Contracting Governments shall furnish to the Commission, at such time and in such form as may be required by the Commission, the statistical information referred to in paragraph 1 (b) of this article.
- vith the assessment of the activities of their signatories which might affect the utilization of fishing resources of a certain area of the shared domain. These resources, although in the shared domain, are economically important to the parties to the conventions. Hence the States voluntarily limit their unilateral activities within that domain. The extent of their co-operation under the two conventions is limited to assessment and monitoring. Compliance with these regulations appears to be voluntary. Nevertheless, the conventions and compliance by their signatories with the recommendations of the commissions have created certain expectations concerning the *regulatory* nature of such recommendations.
- 74. The 1960 Convention on the Protection of Lake Constance against Pollution establishes a Commission with the responsibility of carrying out research to determine the quality of the waters of the lake and the causes of pollution. Article 4 of the Convention reads:

#### Article 4

The Commission shall:

- (a) determine the condition of Lake Constance and the causes of its pollution:
  - (b) regularly verify the quality of the waters of Lake Constance;
- (c) discuss measures for remedying existing pollution and preventing all future pollution of Lake Constance and recommend them to the riparian States;
- (d) discuss measures which any riparian State proposes to take in accordance with article 1, paragraph 3, above;
- (e) study the possibility of instituting regulations to preserve Lake Constance from pollution; consider the possible content of such regulations which shall, if appropriate, form the subject of another convention between the riparian States;
- (f) concern itself with all other questions relating to control of pollution of Lake Constance.

The joint technical sub-committee established by the Tripartite Standing Committee on Polluted Waters created under the 1950 Protocol adopted by France, Belgium and Luxembourg, in addition to defining

polluting factors (industrial or commercial origin, degrees of intensity, etc.), is required to collect any appropriate technical opinions concerning the pollution.

75. Articles 200 and 201 of the 1982 United Nations Convention on the Law of the Sea provide that States shall undertake research and studies individually or collectively through competent international organizations to assess the nature and extent of pollution of the marine environment. The area covered for the purposes of such research and studies is referred to as the "marine environment". The purpose of research and study is to assist States in reaching agreement on the formulation of certain rules, standards and recommended practices which would affect the utilization of the shared domain by the contracting States. Articles 200 and 201 read:

### Article 200. Studies, research programmes and exchange of information and data

States shall co-operate, directly or through competent international organizations, for the purpose of promoting studies, undertaking programmes of scientific research and encouraging the exchange of information and data acquired about pollution of the marine environment. They shall endeavour to participate actively in regional and global programmes to acquire knowledge for the assessment of the nature and extent of pollution, exposure to it, and its pathways, risks and remedies.

#### Article 201. Scientific criteria and regulations

In the light of the information and data acquired pursuant to article 200, States shall co-operate, directly or through competent international organizations, in establishing appropriate scientific criteria for the formulation and elaboration of rules, standards and recommended practices and procedures for the prevention, reduction and control of pollution of the marine environment.

#### (b) Bilateral agreements

76. In a number of multilateral agreements, the collection and exchange of information relate to a broad range of activities. By contrast, bilateral agreements, owing to the greater precision of their subject matter, generally require the collection and exchange of information concerning more specific types of activities using particular resources with certain results. In bilateral agreements concerning shared resources, for example, these requirements relate only to the use of the shared waters. The 1909 Treaty relating to the boundary waters between the United States of America and Canada<sup>35</sup> appears, in article III, to require an assessment of any activity in the boundary waters that might be undertaken either by the United States or by Canada within their respective jurisdictions to ensure that such activities "do not materially affect the level or flow of the boundary waters on the other, nor are such provisions intended to interfere with the ordinary use of such waters for domestic and sanitary purposes". Thus, before undertaking any activity, one party should assess the impact of its conduct on the other. Such an assessment requires the collection and study of data and information concerning the injurious impact of the projects to be undertaken.

<sup>33</sup> Treaty of 11 January 1909 between Great Britain and the United States of America relating to boundary waters and boundary questions concerning the boundary between Canada and the United States.

77. Norway and Sweden agreed in the 1929 Convention<sup>36</sup> relating to their shared watercourses that each State might ask the other's competent authorities for the information necessary to enable it to determine the effects a particular undertaking might have in the other's territory. Article 16 of the Convention reads:

#### REQUESTS FOR INFORMATION

#### Article 16

Each State may ask the competent authority in the other country for the information necessary to enable it to determine what effects the undertaking will produce in the former country.

Accordingly, information may be provided on the basis of the request of the potentially injured State.

78. Not all bilateral agreements deal with specific activities. The 1983 Agreement between the United States of America and Mexico<sup>37</sup> provides in article 7 that the parties shall assess, in accordance with their national laws, regulations and policies, the projects that may have significant impact on the environment of the border area. Article 6 of the Agreement enumerates, among the forms of co-operation among the parties, "impact assessment" and "periodic exchanges of information and data" on the likely sources of pollution in their respective territories. Articles 6 and 7 of the Agreement read:

#### Article 6

To implement this Agreement, the Parties shall consider and, as appropriate, pursue in a co-ordinated manner practical, legal, institutional and technical measures for protecting the quality of the environment in the border area. Forms of co-operation may include: co-ordination of national programmes; scientific and educational exchanges; environmental monitoring; environmental impact assessment; and periodic exchanges of information and data on likely sources of pollution in their respective territory which may produce environmentally polluting incidents, as defined in an annex to this Agreement.

#### Article 7

The Parties shall assess, as appropriate, in accordance with their respective national laws, regulations and policies, projects that may have significant impacts on the environment of the border area, so that appropriate measures may be considered to avoid or mitigate adverse environmental effects.

To co-ordinate this process, each party is required, under article 8 of the Agreement, to designate a national co-ordinator with the principal function of co-ordinating and monitoring the implementation of the Agreement and making recommendations to the parties. In respect of matters to be examined jointly, the national co-ordinators may invite representatives of federal, State and municipal governments to participate in meetings. By mutual agreement, they may also invite representatives of international governmental or non-governmental organizations which may be able to contribute information on the problems. Articles 8 and 9 of the Agreement read:

#### Article 8

Each Party designates a national co-ordinator whose principal functions will be to co-ordinate and monitor implementation of this Agreement, make recommendations to the Parties, and organize the annual meetings referred to in article 10, and the meetings of the experts referred to in article 11. Additional responsibilities of the national coordinators may be agreed to in an annex to this Agreement.

In the case of the United States of America the national coordinator shall be the Environmental Protection Agency, and in the case of Mexico it shall be the Secretaría de Desarrollo Urbano y Ecología, through the Subsecretaría de Ecología.

#### Article 9

Taking into account the subjects to be examined jointly, the national co-ordinators may invite, as appropriate, representatives of federal, State and municipal governments to participate in the meetings provided for in this Agreement. By mutual agreement they may also invite representatives of international governmental or non-governmental organizations who may be able to contribute some element of expertise on problems to be solved.

The national co-ordinators will determine by mutual agreement the form and manner of participation of non-governmental entities.

Each contracting party is required, under this Agreement, to facilitate the entry of the equipment and personnel needed for the implementation of the Agreement (presumably in order to gather information and study likely sources of pollution), subject to the laws and regulations of the receiving State. Article 15 reads:

#### Article 15

The parties shall facilitate the entry of equipment and personnel related to this Agreement, subject to the laws and regulations of the receiving country.

However, all technical information obtained under the Agreement shall be available to both parties and to third parties by mutual agreement of the contracting States. Article 16 reads:

#### Article 16

All technical information obtained through the implementation of this Agreement will be available to both parties. Such information may be made available to third parties by the mutual agreement of the parties to this Agreement.

79. Some bilateral agreements provide that a *joint commission* shall supply information concerning the use of a resource shared by the parties. Article 1 of the 1959 Agreement between Yugoslavia and Greece<sup>38</sup> establishes a Permanent Yugoslav-Greek Hydro-economic Commission to study the hydro-economic problems and projects jointly submitted to it by the parties:

#### Article 1

A Permanent Yugoslav-Greek Hydro-economic Commission shall be established to study the hydro-economic problems and projects jointly submitted to it by the Contracting Parties.

The functions of the Commission shall, inter alia, include cooperation in the study of problems relating to the Vardar (Axius) River with a view to the future regulation of watercourses in the basin of that river, the regulation of streams in the border area, improvement schemes, hydro-economic problems concerning Lake Doiran and Lake Prespa, fishing in those two lakes, the exchange of hydrometeorological data, and any other hydro-economic problems which may arise and which may be jointly referred to the permanent Commission by the Contracting States.

The composition, functions and procedure of the Permanent Yugoslav-Greek Hydro-economic Commission shall be as laid down in the regulations annexed to this Agreement and forming an integral part thereof.

80. Occasionally, the arrangements in bilateral agreements for the exchange of information aim at averting a danger to a State. The danger may be caused by natural phenomena in the territory of another State.

<sup>&</sup>lt;sup>36</sup> Convention of 11 May 1929 between Norway and Sweden on certain questions relating to the law on watercourses.

<sup>37</sup> See footnote 16 above.

<sup>&</sup>lt;sup>38</sup> Agreement of 18 June 1959 between Yugoslavia and Greece concerning hydro-economic questions.

Under article 19 of the 1948 Agreement between Poland and the Soviet Union concerning their frontier area, 39 the parties agree that their competent authorities shall exchange information concerning the level and volume of water and ice conditions in frontier waters if such information may help to avert the dangers created by floods or floating ice:

#### Article 19

- 1. The competent authorities of the Contracting Parties shall exchange information concerning the level and volume of water and ice conditions on frontier waters, if such information may help to avert the dangers created by floods or floating ice. If necessary, the said authorities shall also agree upon a regular system of signals in times of flood or floating ice. Delays in communicating or failure to communicate such information may not constitute grounds for claiming compensation in respect of damage caused by flood or floating ice.
- 81. It should be noted that, according to the above article, delays or failure to communicate such information may not constitute grounds for claiming compensation in respect of damage by flood or floating ice. An identical provision is included in article 19 of the 1950 Treaty between Hungary and the Soviet Union:<sup>40</sup>

#### Article 19

The competent authorities of the Contracting Parties shall exchange information concerning the level of rivers with which the Contracting Parties are concerned, and concerning ice conditions in such rivers, if this information may help to avert danger from floods or from drifting ice. The said authorities shall also agree upon a regular system of signals to be used during periods of high water or drifting ice. Delay in communicating, or failure to communicate, such information shall not constitute ground for a claim to compensation for damage caused by flooding or drifting ice.

Article 3 of the 1968 Agreement between Bulgaria and Turkey<sup>41</sup> provides for exchange of information between the parties concerning floods and floating ice as quickly as possible. Furthermore, the parties agree to exchange hydrological and meteorological data concerning their frontier rivers. Article 3 reads:

#### Article 3

The two Contracting Parties agree to exchange information concerning floods and floating ice by the most expeditious means possible.

In addition, the Contracting Parties agree to exchange hydrological and meteorological data concerning the rivers which flow through the territory of both countries.

The procedure for mutual reporting and the exchange of data on these matters shall be laid down in technical protocols to be concluded between the two Contracting Parties.

82. Bilateral ageements dealing with activities involving nuclear materials appear to be more precise, with more regulatory provisions regarding the collection of data and exchange of information. For example, article 2 of the 1966 Convention between Belgium and France concerning the establishment of a nuclear power station<sup>42</sup> requires the parties to inform each other, by "all

39 Agreement of 8 July 1948 between Poland and the USSR concerning the régime of the Polish-Soviet State frontier. appropriate means", regarding the studies carried out before the installations are put into service, during the operation of the station and also of the occurrence of anything in the station that might affect public health. This provision combines two stages of impact assessment: collection of data and exchange of information prior to the installation of the nuclear plant and monitoring during its operation. Article 2 reads:

#### Article 2

The Contracting Parties undertake to keep each other informed, by all appropriate means, regarding the studies carried out before the installations are put into service, the operation of the installations and the occurrence there of anything which might affect public health.

The company installing the station is a Franco-Belgian joint-stock company. The station is installed in France near its frontier with Belgium. It is not quite clear from the Convention whether the co-operation established between the States results from their partnership in the company installing the station or from the closeness of the station to the Belgian frontier.

The requirement concerning the collection and exchange of information becomes even more necessary and detailed once the activity involving the use of nuclear materials is carried out by the acting State in the territory of the potentially injured State. In such situations the collection and exchange of information are intended to ensure that the activity has met the safety measures and standards accepted between the parties or by the international community. For example, the United States of America, in its 1964 Agreement with Italy regarding the entrance and passage of the N.S. Savannah,43 the United States nuclear ship, to and from Italian ports, agrees to submit to the Italian Government the safety report prepared in accordance with the 1960 Convention on the Safety of Life at Sea, in order to enable the latter to give approval for the entry of the N.S. Savannah. Article II of the Agreement provides:

#### Article II. Safety report

(a) To enable the Italian Government to give its approval for the entry of the ship into Italian ports and the use thereof, the Government of the United States shall submit a safety report prepared in accordance with regulation 7 of chapter VIII of the 1960 Convention on the Safety of Life at Sea and in accordance with recommendation 9 of annex C mentioned above.

Similar provisions exist in the 1963 Operational Agreement between the Netherlands and the United States of America concerning the entry of the N.S. Savannah into Netherlands ports:<sup>44</sup>

#### SAFETY ASSESSMENT AND OPERATING MANUAL

#### Article 7

To enable the Netherlands Government to decide whether or not approval shall be given for the ship's entry into Netherlands waters and the ship's use of the port area of Rotterdam, the Government of the United States shall provide a safety assessment prepared in accordance with regulation 7 of chapter VIII of the International Convention for the Safety of Life at Sea, 1960, and in accordance with recommendation 9 of annex C to the Final Act of the International Conference on Safety of Life at Sea, 1960.

<sup>&</sup>lt;sup>40</sup> Treaty of 24 February 1950 between Hungary and the USSR concerning the régime of the Soviet-Hungarian State frontier.

<sup>&</sup>lt;sup>41</sup> Agreement of 23 October 1968 between Bulgaria and Turkey concerning co-operation in the use of the waters of rivers flowing through the territory of both countries.

<sup>&</sup>lt;sup>42</sup> Convention of 23 September 1966 between Belgium and France on radiological protection with regard to the installations of the Ardennes nuclear power station.

<sup>&</sup>lt;sup>43</sup> See footnote 20 above.

<sup>&</sup>quot; See footnote 22 above.

#### Article 8

As soon after receipt of the *safety assessment* as is practicable the Netherlands Government shall notify the Government of the United States of its decision as to the acceptance of the ship.

#### Article 9

An operating manual prepared in accordance with regulation 8 of chapter VIII of the International Convention for the Safety of Life at Sea, 1960, and with recommendation 8 of annex C to the Final Act of the International Conference on Safety of Life at Sea, 1960, shall be kept on board the ship and shall be kept up to date.

# (c) Judicial decisions and State practice other than agreements

84. Prior consideration of the interests of others has been explicitly recognized and referred to in some judicial decisions, diplomatic correspondence and relations between States. Sometimes assessments have been made unilaterally. For example, the United States of America made a *unilateral determination* to collect data prior to instituting nuclear tests to determine which area of the ocean would be the least likely to cause injuries to other international interests:

Eniwetok Atoll was selected as the site for the proving grounds after the careful consideration of all available Pacific islands. Bikini is not suitable as the site since it lacks sufficient land surface for the instrumentation necessary to the scientific observations which must be made. Of other possible sites, Eniwetok has the fewest inhabitants to be cared for, approximately 145, and, what is more important from a radiological standpoint, it is isolated and there are hundreds of miles of open seas in the direction in which winds might carry radioactive particles.

Construction will be supported through the Hawaiian islands, Johnston Island and Kwajalein Island.

The permanent transfer elsewhere of the island people now living on Aomon and Biijiri islands in Eniwetok Atoll will be necessary. They are not now living in their original ancestral homes but in temporary structures provided for them on the two foregoing islands to which they were moved by United States forces during the war in the Pacific, after they had scattered throughout the atoll to avoid being pressed into labour service by the Japanese and for protection against military operations. The sites for the new homes of the local inhabitants will be selected by them. The inhabitants concerned will be reimbursed for lands utilized and will be given every assistance and care in their move to, and re-establishment at their new location. Measures will be taken to ensure that none of the inhabitants of the area are subject to danger; also that those few inhabitants who will move will undergo the minimum of inconvenience."

85. The assessment was claimed to have been designed to minimize injury to the interests of other international actors:

Protection of health and safety is a primary consideration in the conduct of the HARDTACK series of nuclear weapons tests at the Eniwetok proving ground in the Pacific.

As announced previously, the test series will advance the development of weapons for defence against aggression whether airborne, missile-borne or otherwise mounted. Information on the effects of weapons will be obtained for military and civilian defence use. As in the past, test operations will be conducted in a manner designed to keep to as low as possible the public exposure to radiation arising from the detonation of nuclear weapons.

An important objective of the tests is the further development of nuclear weapons with greatly reduced radioactive fallout in that the area of radiation hazard may be kept as small as possible. This principle was first proved in the Eniwetok test series of 1956.

Various precautions have been taken to keep significant radioactive fallout within the confines of the danger area in the Pacific which was announced on February 14, 1958. With the exception of Joint Task Force facilities, there are no inhabited places within the danger area.

<sup>43</sup> See M. M. Whiteman, ed., Digest of International Law (Washington, D.C.), vol. 4, p. 555.

Extensive systems have been established to detect and measure radioactivity in the vicinity of the proving ground, in the United States, and in other parts of the world. Radiological monitoring and sampling will be conducted by several networks of stations extending from the proving ground to locations around the world. In addition marine surveys will be conducted to measure radioactivity in sea water and marine organisms.<sup>40</sup>

86. Attempts were made by the United States to predict fallout based on weather patterns and meteorological models:

#### Fallout predictions

Tests will be conducted only when the forecast pattern of significant fallout is entirely within the danger area. In forecasting fallout patterns, scientists will make use of improved methods of collecting and evaluating data which have been developed as a result of intensive study of the problem of predicting fallout in the vicinity of the proving ground.

Fallout predictions are dependent upon weather information. Experience has shown that weather data normally available in the Pacific Ocean area are inadequate for the needs of testing. Therefore for nuclear tests in the Pacific special arrangements are made to obtain additional data. For the 1958 tests thirteen special United States weather stations, located within several hundred miles of the proving ground, will participate in a weather network reporting to a central station. These stations will be staffed by military and civilian meteorologists. Weather reconnaissance will be carried on employing aircraft, ships, balloons and rockets.

Research has been conducted in the special field of tropical meteorology, and weather observers and forecasters have been instructed in the latest methods of forecasting which have been developed as a result of these studies.

Trained personnel have been organized into a fallout prediction unit. To assist in predicting fallout patterns they will utilize fallout computers which mechanize most of the mathematical procedures involved. Use of the computers will make possible rapid forecasts. Models of the clouds produced by previous large-scale nuclear detonations have been developed, and these also are expected to improve fallout predictions.<sup>47</sup>

87. In addition, a *danger area was declared* based on information regarding the width of the fallout area and the inhabitants therein:

#### Danger area

The danger area is generally rectangular in shape and comprises roughly 390,000 square nautical miles. It is approximately the same size as the area used in the 1956 test series, but its east and west boundaries have been shifted approximately 120 nautical miles to the west. Except for the test personnel, there are no inhabitants within the area.

All ships, aircraft and personnel have been cautioned to remain clear of the area which is bounded by a line joining the following geographic co-ordinates:

> 18° 30' N, 156° 00' E 18° 30' N, 170° 00' E 11° 30' N, 170° 00' E 11° 30' N, 166° 16' E 10° 15' N, 166° 16' E 10° 15' N, 156° 00' E

Notices have been given the widest possible distribution through marine, aviation and international organizations.

Regular air and sea searches of the area will be conducted in advance of the start of operations. Before each shot, the patrol of the danger area will be intensified, particularly in the area where fallout is forecast.

The Atomic Energy Commission has issued regulations which prohibit entry into the danger area of U.S. citizens and all other persons subject to the jurisdiction of the United States, its territories and possessions.

<sup>46</sup> Ibid., p. 588.

<sup>47</sup> Ibid., pp. 588-589.

The regulations effective from April 11, 1958 until the HARD-TACK test series is completed prohibit entry, attempted entry or conspiracy to enter the danger area.<sup>44</sup>

- 88. It appears that the United States Atomic Energy Commission and the Department of Defense were investigating and making predictions on the effects of radiation. It is unclear whether United States or foreign private scientific institutions were permitted to participate in any form in those investigations and predictions. Owing to the nature of the activity and the particular security interest of the United States Government in maintaining exclusive control over the area where the tests were to be conducted, the collection of data by private or foreign government agencies was virtually impossible.<sup>49</sup>
- 89. According to United States forecasts, based on data analysed using scientific methods, no significant fallout or radioactivity would occur in inhabited areas:

#### Radiation monitoring in proving ground region

Radiological safety personnel, equipped with radiation detection and measuring instruments and two-way radios to enable them to communicate with the central Task Force Radiological Safety Office, will be stationed on nearby inhabited atolls, and at weather stations of the weather reporting network. In the unlikely event of significant fallout in an inhabited area, the monitors would warn the inhabitants and advise and assist them in taking safety measures. The monitors also have trained Marshallese medical practitioners and health aides in basic emergency measures.

#### Radiation surveys of sea and marine life

Outside of the testing area, the detonations are not expected to add enough radioactive material to natural levels of radioactivity in the ocean to be harmful to marine life. Experience shows that outside the testing area, resulting quantities of radioactivity in edible sea foods will result in exposures which will be very small compared with the limits for public exposure recommended by the United States National Committee for Radiation Protection and Measurement.

As in the past there will be a programme of study to explore the ultimate destination and behaviour of radioactivity in the sea water and in marine organisms. Sweeps by U.S. Navy Vessels both during and after the test series will include such measures as taking continuous readings of radioactivity in surface water, sampling of water at various depths, making tows to gather plankton—the tiny marine organisms which tend to concentrate radioactive materials in their tissues—and catching of fish for analysis for radioactivity.

In addition to these investigations, land and marine biological surveys again will be conducted at Eniwetok and Bikini and other atolls nearby. Samples of water and of plants and animals living in the lagoons and on the reefs and islands of the atolls will be collected and analysed for radioactivity.

#### Fallout monitoring in United States

The heavier particles fall out of the radioactive cloud at early times after a detonation, while their radioactivity is still high. Therefore, the highest levels of radioactivity occur over a local area downwind from the point of detonation. The area of significant fallout is expected to occur entirely within the uninhabited danger area surrounding the Eniwetok proving ground.

As the radioactive cloud is transported away from the point of detonation, it is widely dispersed by air currents and diluted by normal air. Its radioactivity also decreases rapidly because of the normal process of radioactive decay. [Radioactive fallout consists of a mixture of radioisotopes, with varying half-lives. The mixture as a whole decreases in radioactivity in such a way that for every sevenfold increase in age, the total radioactivity is decreased 10-fold. Thus, the radioactivity at seven hours after H+1 hour is only one tenth that at H+1 hour, and in 49 hours is one hundredth, etc.] By the time the cloud from a detonation in the Eniwetok proving ground has travelled across a vast expanse of ocean, it will have become thoroughly dispersed into the air and will have lost most of its original radioactivity.

As a result, the exposures to radioactivity in the United States from the Eniwetok tests are expected to be low. Although levels of many times the normal background may be reached in some localities, these increases will be temporary and will not greatly increase the total exposure to radiation. Average exposures of residents of the United States to radiation from weapons tests during the past five years has been much less than the average exposure to radiation from natural sources during the same period. <sup>50</sup>

90. The same types of *predictions* were made by the British based on scientific, geographical and meteorological information on hand, *unilaterally*:

The tests will be high air bursts which will not involve heavy fallout. Extensive safety precautions have been taken. A danger area has been declared for the period 1st March to 1st August [1957] and all shipping and aircraft have been warned to keep clear of this area. The warning has been issued far in advance so that people should be clearly aware of the position. No permanently inhabited island lies within the danger area. Weather stations, weather ships and meteorological reconnaissance flights by aircraft will provide continuous meteorological information during the period of the tests. Provided persons stay outside the danger area they have nothing to fear. The temporary use of areas outside territorial waters for gunnery or bombing practice has, as such, never been considered a violation of the principles of freedom of navigation on the high seas. The present site has been carefully chosen because it lies far from inhabited islands and avoids as far as possible shipping and air routes. It is incidentally some 4,000 miles from Japan.51

91. As to the effect of the radiation on health, the British Government asked an independent committee under the auspices of the Medical Research Council to examine the matter:

As regards the general effects of radioactivity resulting from nuclear test explosions I am to state that before proceeding with their plans to develop and test weapons in the megaton range, Her Majesty's Government went most carefully into the question of potential hazards to health and asked an independent committee under the auspices of the Medical Research Council to examine the subject.

The Medical Research Council's report *The Hazards to Man of Nuclear and Allied Radiations* which was compiled by the leading authorities in the United Kingdom on this subject was published in June, 1956. The Prime Minister, Mr. Macmillan, told the House of Commons on 5th March:

"I understand that the Medical Research Council have no evidence that the amount of strontium-90 and other radioactive particles released by hydrogen bomb explosions which may become sources of internal radiation has reached a potentially dangerous level. The present and foreseeable hazards, including genetic effects, from the external radiation due to fall-out from the explosions of nuclear weapons fired at the present rate and in the present proportion of the different kinds, are considered to be negligible: accordingly I am not prepared to postpone the forthcoming test in the Pacific."

This statement was based on up-to-date advice from the Medical Research Council and the British Prime Minister, in reply to a further question in the House of Commons on 12 March, stated that the Medical Research Council was keeping the hazards to man from all sources of radiation under continuous review.<sup>52</sup>

<sup>48</sup> Ibid., p. 589.

<sup>&</sup>lt;sup>49</sup> Normally, when a danger area was established no one was permitted to enter it without permission from the Atomic Energy Commission or the Department of Defense. For example, the regulations issued by the Atomic Energy Agency on 9 April 1958 in connection with the series of nuclear tests to be conducted at Eniwetok in 1958 provided in section 112.4:

<sup>&</sup>quot;No United States citizen or other person . . . shall enter, attempt to enter or conspire to enter the danger area during the continuation of the HARDTACK test series, except with the express approval of appropriate officials of the Atomic Energy Commission or the Department of Defense." (Federal Register (Washington, D.C.), vol. 23, No. 73, 12 April 1958, p. 2401.)

<sup>50</sup> Whiteman, op. cit., pp. 589-591.

<sup>51</sup> Ibid., p. 598.

<sup>52</sup> Ibid., p. 599.

- 92. Although it seems that States give priority to their own security interests over the interests of other States, at least in two instances of H-bomb tests the acting States made efforts to collect and publish data concerning the effect of their activities and to demonstrate that they had given some attention to the interests of other States. Such gathering of information, of course, was carried out unilaterally by the acting States themselves.
- 93. The potentially injured State has also taken the initiative in suggesting that data be collected or studies made prior to initiation of an activity. The United States of America, in correspondence with Mexico concerning highway construction which, according to the United States, could result in an unnatural accumulation of waters and cause injury to United States citizens and their properties in the event of heavy rains, made the suggestion to study the situation and develop remedial plans. On 20 May 1957, the United States Commissioner on the International Boundary and Water Commission wrote to the Mexican Commissioner as follows:

"In view of the aforedescribed situation, I will appreciate an examination of the problem by your Section, and, if the conditions found are as reported to me, that appropriate arrangements be made with the proper authorities in Mexico to take such remedial measures as required to eliminate this threat to interests in my country."...

For two years thereafter, the United States Section of the Commission acted in this matter exclusively in an engineering advisory capacity to the Department of State and the American Consulate at Mexicali in their informal discussion of the projects and safety precautions considered essential with officials of the State Government of Baja California who were connected with the projects above described. In discussions with officials and engineers of the State Government of Baja California, United States engineers sought to avoid any implication that the State Government was obligated to obtain United States consent or approval of the U.S. Section of the International Boundary and Water Commission for its specific construction plans, and that in passing on them, the United States engineers were representing the views of their Government. A plan for culverts which was considered inadequate by United States engineers was finally abandoned by the State Government. A new set of plans drawn up by the State Government was sent to the American Consulate at Mexicali which, in turn, forwarded the plans to the United-States Section of the Commission for a statement of its views. The United States Section replied that the plans appeared adequate with certain suggested modifications which were transmitted in a letter addressed to the Chief of the Department of Highways and Communications of Baja California (Rendon) by the United States Consul at Mexicali (M. W. Boyd) on October 24, 1958 . . . .

94. The potentially injured State may suggest the study and collection of data to be made by a joint commission, as was suggested to Mexico by the United States during the *Rose Street Canal* correspondence:

The present unfortunate situation appears to have developed from the expansion of the city of Agua Prieta toward and beyond the flood arroyo. With the simultaneous expansion of the city of Douglas, the existing drainage canals have become inadequate and represent a matter of concern to both cities. As a consequence the *International Boundary and Water Commission undertook informal studies and surveys in 1949 and 1950*,\* and the results suggest the desirability of constructing new flood control works in each of our two countries.

My Government agrees that the International Boundary and Water Commission should continue its studies with the intention of bringing them to a conclusion and of submitting a joint report as early as possible in this year. This report might include recommendations not only concerning remedial measures but also with respect to an equitable division of costs between our Governments . . . . 34

95. Where the activity is in the nature of protective measures, such as flood control in light of imminent rains, the acting State may postpone the collection of data, considering that the need for the protective works outweighs the obligation of impact assessment. The acting State nevertheless usually informs the other State of activities which it intends to undertake. That procedure was followed by Mexico when it found itself compelled to take the necessary measures to avoid flooding:

Study of the new protective works has been practically at a standstill for the last two years, owing to the fact that the United States Section has declared that it must first carry out a series of investigations and make topographical studies.

My Government sincerely desires to reach an agreement with Your Excellency's Government on this question, but in view of the damage which the lack of a solution is causing the city of Agua Prieta and the fact that the rainy season is approaching, the Government of Mexico finds itself compelled to take the necessary measures sufficiently in advance, so that the floods may not be repeated this year. Consequently, the Mexican authorities will, on May 1 next, begin building certain protective works to prevent the entry into Agua Prieta of rain water collected by the Rose Street Canal in Douglas.

I take the liberty of bringing the foregoing to Your Excellency's attention to the end that the proper authorities of your Government may take such measures as they consider advisable to prevent consequences which the return of such water might have in the city of Douglas.<sup>55</sup>

96. In its award in the *Trail Smelter* case, the tribunal briefly described and highly commended the comprehensive and long-term experiments and collections of data analysed in order to develop a permanent régime fulfilling the duty of care required of the Canadian smelter. The tests had been carried out over a period of three years under the supervision of what the tribunal called "well-established and known scientists" in chemistry, plant physiology, meteorology and the like, for the purpose of collecting data on the pollution caused by the smelter and on the damage to United States interests. In the opinion of the tribunal, the study was "probably the most thorough [one] ever made of any area subject to atmospheric pollution by industrial smoke". " Some of the factors considered had been used for the first time in evaluating smoke control. The methods successfully used in testing eventually became embodied in the régime adopted by the tribunal:

The foregoing paragraphs are the result of an extended investigation of meteorological and other conditions which have been found to be of significance in smoke behaviour and control in the Trail area. The attempt made to solve the sulphur dioxide problem presented to the tribunal has finally found expression in a régime which is now prescribed as a measure of control.

The investigations made during the past three years on the application of meteorological observations to the solution of this problem at Trail have built up a fund of significant and important facts. This is probably the most thorough study ever made of any area subject to atmospheric pollution by industrial smoke. Some factors, such as atmospheric turbulence and the movement of the upper air currents have been applied for the first time to the question of smoke control. All factors of possible significance, including wind directions and velocity, atmospheric temperatures, lapse rates, turbulence, geostrophic winds, barometric pressures, sunlight and humidity, along with atmospheric sulphur dioxide concentrations, have been studied. As said above, many observations have been made on the movements and sulphur dioxide concentrations of the air at higher levels by means of pilot and captive balloons and by airplane, by night and by day. Progress has been made in breaking up the long winter fumigations and in reducing their intensity. In carrying finally over to

<sup>33</sup> Whiteman, op. cit. (footnote 45 above), vol. 6, pp. 260-261.

<sup>&</sup>lt;sup>14</sup> Ibid., p. 264.

<sup>55</sup> Ibid.

<sup>&</sup>lt;sup>16</sup> United Nations, Reports of International Arbitral Awards, vol. III, p. 1973.

the non-growing season with a few minor modifications a régime of demonstrated efficiency for the growing season, there is a sound basis for confidence that the winter fumigations will be kept under control at a level well below the threshold of possible injury to vegetation. Likewise, for the growing season a régime has been formulated which should throttle at the source the expected diurnal fumigations to a point where they will not yield concentrations below the international boundary sufficient to cause injury to plant life. This is a goal which this tribunal has set out to accomplish.<sup>37</sup>

#### B. Prior negotiation and consultation

- The object of negotiations and consultations with the potentially injured State prior to the commencement or during the performance of activities may be to exchange scientific data on the projects, to consider the views of the potentially injured State and those of the acting State regarding the potential transboundary effects of activities, or to solicit the consent of the potentially injured State regarding the undertaking of activities with whatever consequences they may have. Such prior negotiations and consultations may relate to a variety of subjects, such as the nature of the injury (material, non-material, potential), and who is to decide what constitutes harm and in accordance with what criteria and procedure. Thus prior negotiation is a procedure by which the parties may agree upon the resolution of their conflicting interests.
- Furthermore, paragraph 1 of Article 33 of the United Nations Charter recognizes negotiation and conciliation as preferable means of resolving conflicts among States. Under this Article, States which are parties to any dispute . . . the continuance of which is likely to endanger the maintenance of international peace and security, shall, first of all, seek a solution by negotiation, inquiry, mediation, conciliation, arbitration, judicial settlement, resort to regional agencies or arrangements, or other peaceful means of their own choice. This is not to claim that all disputes relating to activities that may cause transboundary injuries are likely to threaten international peace and security. The point is that negotiation has been recognized as an important first step in peacefully reconciling conflicting interests.

#### (a) Multilateral agreements

99. A more general requirement of prior consultation is embodied in article 5 of the 1979 Convention on Long-range Transboundary Air Pollution:

#### Article 5

Consultations shall be held, upon request, at an early stage between, on the one hand, Contracting Parties which are actually affected by or exposed to a significant risk of long-range transboundary air pollution and, on the other hand, Contracting Parties within which and subject to whose jurisdiction a significant contribution to long-range transboundary air pollution originates, or could originate, in connection with activities carried on or contemplated therein.

Under this article, consultation shall take place "upon request" of either the acting or the potentially injured State, when there is a "significant" risk of air pollution. The word "significant" has not been defined; it will presumably be decided between the States involved. The reference to the acting State in this article is also signifi-

cant. It imposes the obligation of consultation equally upon the State under whose jurisdiction the injurious activity has taken place and upon the State under whose jurisdiction the injurious activity "could" take place. Thus reasonable grounds for causality may be sufficient.

100. The 1974 Convention for the Prevention of Marine Pollution from Land-based Sources uses equally broad language. It provides in article 9, paragraph 1, for consultation upon request of either the acting or the injured State when the activity of the acting State is "likely to prejudice the interests" of the other State:

#### Article 9

1. When pollution from land-based sources originating from the territory of a Contracting Party by substances not listed in part 1 of annex A to the present Convention is likely to prejudice the interests of one or more of the other Parties to the Convention, the Contracting Parties concerned undertake to enter into consultation, at the request of any one of them, with a view to negotiating a co-operation agreement.

101. The 1982 United Nations Convention on the Law of the Sea provides in article 142, paragraph 2, that the State involved in the exploitation of mineral deposits of the sea-bed across the limits of the national jurisdiction of a coastal State must consult that State and maintain a system of prior notification:

Article 142. Rights and legitimate interests of coastal States

2. Consultations, including a system of prior notification, shall be maintained with the State concerned, with a view to avoiding infringement of such rights and interests. In cases where activities in the Area may result in the exploitation of resources lying within national jurisdiction, the prior consent of the coastal State concerned shall be required.

The régime here prescribed appears as a more systematic and institutionalized form of prior notification. Thus article 206 of the Convention requires notification by the acting State when it has reasonable grounds for believing that activities to be undertaken within its jurisdiction may cause injuries to others. That notification should be in accordance with the procedures stipulated in article 205:

#### Article 205. Publication of reports

States shall publish reports of the results obtained pursuant to article 204 or provide such reports at appropriate intervals to the competent international organizations, which should make them available to all States.

102. Even in connection with activities of self-help, the acting State may be required to consult the potentially injured States. For example, article III of the 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties requires a coastal State, before taking any measures, to consult with other States affected by the maritime casualty, particularly with the flag State, and give notice of the measures which it intends to take. The coastal State may also consult with independent experts before any measure is taken. The relevant paragraphs of article III read:

#### Article III

When a coastal State is exercising the right to take measures in accordance with article I, the following provisions shall apply:

<sup>57</sup> Ibid., pp. 1973-1974.

- (a) before taking any measures, a coastal State shall proceed to consultations with other States affected by the maritime casualty, particularly with the flag State or States;
- (b) the coastal State shall notify without delay the proposed measures to any persons physical or corporate known to the coastal State, or made known to it during the consultations, to have interests which can reasonably be expected to be affected by those measures. The coastal State shall take into account any views they may submit;
- (c) before any measure is taken, the coastal State may proceed to a consultation with independent experts, whose names shall be chosen from a list maintained by the Organization;
- (f) measures which have been taken in application of article I shall be notified without delay to the States and to the known physical or corporate persons concerned, as well as to the Secretary-General of the Organization.
- 103. Article 12, paragraph 3, of the 1983 Convention for the Protection and Development of the Marine Environment of the Wider Caribbean Region provides that, when appropriate, the contracting States may consult other States which may be affected by their activities (see para. 60 above). The requirement of prior negotiation and consultation in this Convention is not obligatory; it appears to be based on the principle of cooperation.

#### (b) Bilateral agreements

104. In bilateral agreements, consultation and prior negotiation appear to be envisaged on the basis of, among other things, the *spirit of co-operation* between neighbouring States, or on the basis of their uncertainty as to the legal effects of their conduct if such conduct causes extraterritorial injuries. The 1975 Agreement between Canada and the United States of America relating to weather modification activities<sup>58</sup> combines the two aforesaid elements. The preamble to the Agreement states that, because of the geographic proximity of the two States, the effects of weather modification activities carried out by either party or its nationals may affect the territory of the other. It states further that a prompt exchange of information regarding the nature and extent of weather modification activities may facilitate development of the technology of weather modification. The preamble also refers to the "traditions" of prior notification and consultation and the close cooperation between the two States, and stresses the "desirability of the development of international law\* relating to weather modification activities having transboundary effects". The relevant paragraphs of the preamble read:

Aware, because of their geographic proximity, that the effects of weather modification activities carried out by either Party or its nationals may affect the territory of the other;

Taking into particular consideration the special traditions of prior notification and consultation and the close co-operation that have historically characterized their relations;

Believing that a prompt exchange of pertinent information regarding the nature and extent of weather modification activities of mutual interest may facilitate the development of the technology of weather modification for their mutual benefit:

Article II of the Agreement requires the responsible agencies of the contracting parties to transmit information relating to weather modification activities of mutual interest. Such information, whenever possible, is to be transmitted prior to the commencement of the

activities. The article anticipates that such information will be transmitted within five working days of its receipt by a responsible agency. Article II, paragraph 1, reads:

#### Article II

1. Information relating to weather modification activities of mutual interest acquired by a responsible agency through its reporting requirements or otherwise, shall be transmitted as soon as practicable to the responsible agency of the other Party. Whenever possible, this information shall be transmitted prior to the commencement of such activities. It is anticipated that such information will be transmitted within five working days of its receipt by a responsible agency.

Each contracting party agrees, under article IV of the Agreement, to *notify* and *fully inform* the other of any weather modification activity of mutual interest prior to the commencement of such activities. Thus every effort is to be made to provide such notice as *far in advance* of such activities as possible. The article provides:

#### Article IV

In addition to the exchange of information pursuant to article II of this Agreement, each Party agrees to notify and to fully inform the other concerning any weather modification activities of mutual interest conducted by it prior to the commencement of such activities Every effort shall be made to provide such notice as far in advance\* of such activities as may be possible, bearing in mind the provisions of article V of this Agreement.

Under article V of the Agreement, the parties agree to consult each other on weather modification activities in the light of their domestic laws and administrative regulations:

#### Article V

The Parties agree to consult, at the request of either Party, regarding particular weather modification activities of mutual interest. Such consultations shall be initiated promptly on the request of a Party, and in cases of urgency may be undertaken through telephonic or other rapid means of communication. Consultations shall be carried out in the light of the Parties' laws, regulations, and administrative practices regarding weather modification.

105. An elaborate procedure for notification prior to an undertaking is provided for in the 1922 Treaty between Germany and Denmark concerning their frontier waters, 59 article 2 of which establishes a Frontier Water Commission to deal with all questions relating to the frontier waters specified in the Treaty. Owing to the structure and authority of the Commission, individual nationals of the parties as well as the various districts and counties of the two countries may negotiate either through the Commission or with the Commission itself. Accordingly, any establishment of new or extensive alteration of existing works on any parts of the frontier waters referred to in the Treaty must be approved by the Commission. Article 30 of the Treaty provides that, in such cases, there must be public notification, to which the attention of all persons who may clearly suffer damage from the activity must be drawn by registered letter. The requirement of direct notification is thus confined to persons who will *clearly* suffer damage. There must nevertheless be public notification of the activity. The relevant paragraphs of article 30 read:

<sup>58</sup> See footnote 15 above.

<sup>&</sup>lt;sup>59</sup> Agreement of 10 April 1922 for the settlement of questions relating to watercourses and dikes on the German-Danish frontier.

#### Article 30

... the proposed use of the watercourse shall be brought to the notice of the public\* in the manner which is customary in the locality in all communes or manorial districts (Gutsbezirke), the land of which might be affected by the operation of the works in the event of their being authorized.

Further, the attention of all persons who will clearly suffer damage from the authorization of the works shall be drawn to the public notification by means of registered letters.

The notification must describe the authorized activity and name the authorities to which objections may be made or requests for taking preventive measures submitted, etc. It must also establish a time-limit of from two to six weeks for raising objections. Article 31 of the Treaty sets out the detailed content of the notification as well as the appeal procedure by individual claimants:

#### Article 31. Content of notification

Notifications shall state where the drawings and explanations which have been submitted may be inspected, and shall mention the authorities to which objections to the authorization and also applications for the erection and upkeep of installations for the prevention of damage, or applications for compensation shall be addressed in writing or be made orally in official form. A time limit shall also be fixed for lodging objections or making applications. The period allowed shall be not less than two, and not more than six weeks. It shall begin to run from the day following that upon which the gazette containing the final notification is published.

It shall be stated in the notification that all persons who have not lodged any objection or made any application within the time limit fixed shall lose their rights in that connection, but that applications for the erection and upkeep of installations or for compensation may be made at a later date if they are based upon damage which could not be foreseen during the period covered by the time limit.

Even after the expiration of the appointed time a person who has suffered damage shall not be debarred from submitting a claim provided he can show that he was prevented by circumstances over which he had no control from submitting such claim within the time limit.

The right of establishing claims after the expiration of the appointed time is subject to prescription three years after the date on which the person who suffered damage learned of the existence of such damage.

The same time limit shall also be fixed in the notification for other applications for the authorization of a particular use of the water-course by which the use proposed by the first applicant would be restricted. It shall also be made clear that applications of this kind made after the expiration of the appointed time in connection with the same matter will not be taken into consideration.

A suitable additional period may be allowed for the production of evidence.

An elaborate procedure for prior consultation and negotiation is also provided for in article 3 of chapter I of the 1931 Convention between Romania and Yugoslavia concerning their frontier waters. 60 Under this article, if either party intends to make any alterations or changes in its own territory which might affect the hydraulic system in the basin, it shall inform the other State by registered letter with notification of receipt. This is the first step required to reach an agreement with the other State about the proposed changes. Such proposed changes may not be carried out without reaching an agreement with the other State. But if the other State does not acknowledge receipt or make any observation within two and a half months from the date of the communication, the acting State may proceed with its activity without further formalities. If the parties are unable to reach agreement within a reasonable period, the Convention envisages a different procedure, which does not define this period. Article 3 of chapter I provides:

#### Article 3

Should either State propose to make any alterations or take any measures or undertake any works in its own territory such as might change to an appreciable extent the hydraulic system in the basins mentioned in article I above, it shall, by registered letter with notification of receipt\* send to the other State notice of its intentions or measures, with a view to the preliminary establishment of the agreement provided for by article 292 of the Treaty of Trianon.

Such communication shall be confirmed within a period of 15 days. If within two and a half months from the date of the communication the latter State has neither acknowledged receipt nor made any

tion the latter State has neither acknowledged receipt nor made any observations, the proposed alterations, measures or works may be undertaken without further formalities.

In the contrary event, the proposed alterations, measures or works may not be carried out until agreement has been reached between the two States.

If agreement is not reached within a reasonable period, action shall be taken in accordance with article 6 of the regulations of the C.R.E.D.

107. The 1929 Convention between Norway and Sweden<sup>61</sup> provides for a procedure of consultation between the two States concerning activities undertaken by private entities of one of those States that might cause injuries in the other State. Article 14 of the Convention provides that applications for authorizations of certain activities must be addressed to the competent authorities of the State in whose territory the activity is to be undertaken, together with a detailed description of the activity and its plan. The competent authorities must send a copy of this application and plans, etc. to the other State. The article reads:

#### **PROCEDURE**

#### APPLICATIONS

#### Article 14

- 1. Applications for authorizations for an undertaking shall be addressed to the competent authority in the country in which the undertaking is to be carried out. If the waterfall, the immovable property or the transport or floating interest on account of which the undertaking is to be carried out belongs to the other country, the application shall be accompanied by a declaration from that State to the effect that it has no objection to the application being considered.
- 2. Applications shall be accompanied by the plans, specifications and particulars required to enable the effects which the undertaking will produce in both countries to be determined.
- 3. When an application has been received by the authority in the country in which the undertaking is to be carried out, a copy, together with the enclosures, shall be transmitted to the other State.

# (c) Judicial decisions and State practice other than agreements

108. The obligation of prior consultation and negotiation concerning activities with potentially injurious consequences outside the national territory has been developed in a number of judicial decisions. In its advisory opinion of 15 October 1931 in the Railway Traffic between Lithuania and Poland case, the Permanent Court of International Justice stated that the obligation to negotiate was "not only to enter\* into negotiations but also to pursue\* them as far as possible with a view to concluding agreements\*". 62 Repeating the same re-

<sup>60</sup> General Convention of 14 December 1931 between Romania and Yugoslavia concerning the hydraulic system.

<sup>61</sup> See footnote 36 above.

<sup>62</sup> P.C.I.J., Series A/B, No. 42, p. 116.

quirement, the International Court of Justice stated in its judgment of 20 February 1969 relating to the North Sea Continental Shelf cases that, prior to any unilateral delimination of the continental shelf, the parties had an obligation to enter into negotiations going beyond a merely formal process and to behave with the intention of reaching a satisfactory conclusion. The Court ruled as follows:

(a) the parties are under an obligation to enter into negotiations with a view to arriving at an agreement,\* and not merely to go through a formal process of negotiation as a sort of prior condition for the automatic application of a certain method of delimination in the absence of agreement; they are under an obligation so to conduct themselves that the negotiations are meaningful,\* which will not be the case when either of them insists upon its own position without contemplating any modification of it;63

The Court elaborated on the content of the obligation to negotiate: the parties were required to take into account all circumstances and to apply equitable principles. Thus it referred to and enumerated principles and factors which should be evaluated and accommodated in order to reach equitable principles in the case in point:

(b) the parties are under an obligation to act in such a way that, in the particular case, and taking all the *circumstances\** into account, equitable principles are applied,—for this purpose the equidistance method can be used, but other methods exist and may be employed, alone or in combination, according to the areas involved;<sup>44</sup>

This is a clear reference to the need to balance the interests of Denmark and the Netherlands with those of the Federal Republic of Germany. The Court noted that the duty to negotiate was simply an application of the more general principle of international relations that:

the judicial settlement of international disputes "is simply an alternative to the direct and friendly settlement of such disputes between the parties" (Free Zones of Upper Savoy and the District of Gex, P.C.I.J. Series A, No. 22, p. 13)."

The negotiations having been found unsatisfactory, the Court considered that initiation of fresh negotiations was the appropriate remedy:

... In the present case, it needs to be observed that whatever the details of the negotiations carried on in 1965 and 1966, they failed of their purpose because the Kingdoms of Denmark and the Netherlands, convinced that the equidistance principle alone was applicable, in consequence of a rule binding upon the Federal Republic, saw no reason to depart from that rule; and equally, given the geographical considerations stated in the last sentence of paragraph 7 above, the Federal Republic could not accept the situation resulting from the application of that rule. So far therefore the negotiations have not satisfied the conditions indicated in paragraph 85 (a), but fresh negotiations are to take place on the basis of the present judgment.\*\*

109. In its award on 16 November 1957 in the *Lake Lanoux* case, the arbitral tribunal recognized, in fact, the more general requirement of prior negotiation, but not of agreement. The tribunal noted that, in certain situations, the other party, namely, the potentially injured State, might, in violation of the rules of good faith, paralyse genuine negotiation efforts. The tribunal stated that in such circumstances sanctions might be applied:

In effect, in order to appreciate in its essence the necessity for prior agreement, one must envisage the hypothesis in which the interested States cannot reach agreement. In such case, it must be admitted that

the State which is normally competent has lost its right to act alone as a result of the unconditional and arbitrary opposition of another State. This amounts to admitting a "right of assent", a "right of veto", which at the discretion of one State paralyses the exercise of the territorial jurisdiction of another.

That is why international practice prefers to resort to less extreme solutions by confining itself to obliging the States to seek, by preliminary negotiations, terms for an agreement, without subordinating the exercise of their competences to the conclusion of such an agreement. Thus, one speaks, although often inaccurately, of the "obligation of negotiating an agreement". In reality, the engagements thus undertaken by States take very diverse forms and have a scope which varies according to the manner in which they are defined and according to the procedures intended for their execution; but the reality of the obligations thus undertaken is incontestable and sanctions\* can be applied in the event, for example, of an unjustified breaking off of the discussions, abnormal delays, disregard of the agreed procedures, systematic refusals to take into consideration adverse proposals or interests, and, more generally, in cases of violation of the rules of good faith\* (Tacna-Arica arbitration, United Nations, Reports of International Arbitral Awards, vol. II, pp. 921 et seq.; Case of Railway Traffic between Lithuania and Poland, P.C.I.J., Series A/B, No. 42, p. 108).67

The tribunal did not elaborate on what those sanctions might be. The duty of prior negotiation was recognized as essential in balancing interests, since only the affected State could accurately evaluate whether a particular activity might affect its interests:

... The conflicting interests aroused by the industrial use of international rivers must be reconciled by mutual concessions embodied in comprehensive agreements. States have a duty to seek to enter into such agreements. The "interests" safeguarded in the treaties between France and Spain included interests beyond specific legal rights. A State wishing to do that which will affect an international water-course cannot decide whether another State's interests will be affected; the other State is the sole judge of that and has the right to information on the proposals.<sup>68</sup>

110. The obligation to negotiate genuinely and in good faith was thus restated:

... Consultations and negotiations between the two States must be genuine, must comply with the rules of good faith and must not be mere formalities. The rules of reason and good faith\* are applicable to procedural rights and duties relative to the sharing of the use of international rivers; and the subjecting by one State of such rivers to a form of development which causes the withdrawal of some supplies from its basin, are not irreconcilable with the interests of another State.69

111. The principle of prior negotiation applies also to disputes involving the distribution of shared resources in the common domain. For example, in its judgment of 25 July 1974 in the Fisheries Jurisdiction case, the International Court of Justice held that Iceland could not unilaterally extent its fishing rights beyond the area of 12 miles from its baselines because, among other things, Iceland had failed to observe the duty of prior negotiation with the States concerned. The Court held that both the United Kingdom and Iceland were under the mutual obligation to negotiate in good faith for the equitable solution of the question of distribution of their fishery rights. The relevant passage of the Court's judgment reads:

<sup>63</sup> I.C.J. Reports 1969, p. 47, para. 85.

<sup>64</sup> Ibid.

<sup>65</sup> Ibid., para. 87.

<sup>66</sup> Ibid.

<sup>&</sup>lt;sup>67</sup> International Law Reports 1957 (London, 1961), vol. 24, p. 128. See also Yearbook . . . 1974, vol. II (Part Two), pp. 194-199, document A/5409, paras. 1055-1068.

<sup>48</sup> Summary of the tribunal's considerations in *International Law Reports 1957* (see footnote 67 above), p. 119.

<sup>69</sup> Ibid.

79. For these reasons,

THE COURT.

by ten votes to four,

- (1) finds that the regulations concerning the fishery limits off Iceland (Reglugerö um fiskveiöilandhelgi Islands) promulgated by the Government of Iceland on 14 July 1972 and constituting a unilateral extension of the exclusive fishing rights of Iceland to 50 nautical miles from the baselines specified therein are not opposable to the Government of the United Kingdom;
- (2) finds that, in consequence, the Government of Iceland is not entitled unilaterally to exclude United Kingdom fishing vessels from areas between the fishery limits agreed to in the exchange of notes of 11 March 1961 and the limits specified in the Icelandic regulations of 14 July 1972, or unilaterally to impose restrictions on the activities of those vessels in such areas;

by ten votes to four.

(3) holds that the Government of Iceland and the Government of the United Kingdom are under mutual obligations to undertake negotiations in good faith for the equitable solution of their differences concerning their respective fishery rights in the areas specified in subparagraph 2;70

It appears from this decision that activities beyond a State's territorial jurisdiction with potentially direct and substantial injuries to other States may be undertaken only after prior negotiations leading to the consent of the affected States. The Court did not address the issue of what should be done if no agreement was reached after bona fide negotiations.

In State practice, the prior negotiations regarding the site of nuclear power plants in Central Europe confirmed this principle.71 In Dukovany, Czechoslovakia, approximately 35 kilometres from the Austrian border, two Soviet-designed 440 megawatt electrical power reactors were scheduled to be operating by 1980. The closeness of the location of the Austrian border led to a demand by the Austrian Ministry for Foreign Affairs for joint talks with Czechoslovakia about the safety of the facility. This was accepted by the Czech Government.<sup>72</sup> More extensive negotiations took place between Switzerland and Austria regarding Swiss plans to construct a 900 MW nuclear power plant near Rüthi, in the Upper Rhine Valley, close to the Austrian border. As a result of Austrian objections,73 the Swiss Government entered into consultations with the Austrian Federal Government as well as with the Voralberg State Government, the federated State which would have been affected by the Swiss project.74 The talks seem to have focused on the legal principles of good neighbourliness.75 The Swiss Government evidently re-evaluated the entire project but, shortly before the re-evaluation was completed, the Austrian Foreign Minister stated in a press conference that, if the Government of the federated State of Voralberg still believed that the revised project was in conflict with the principle of good neighbourliness, the Austrian Government would be committed to asserting formally the illegality of the project. There is no direct evidence to conclude that the postponement of the construction of the nuclear plant by the Swiss Government was due to the Austrian objection or to acceptance of Austria's legal arguments. A number of other factors could have affected the Swiss decision, such as domestic opposition to the plant, or the Government's energy policy. In evaluating the situation, at least one author has concluded that the Austrian objection was an important element affecting the Swiss decision concerning the nuclear plant.

In 1973, the Belgian Government announced its intention to construct a refinery at Lanaye, near its frontier with the Netherlands. The Netherlands Government voiced its concern because the project threatened not only the nearby Netherlands national park but also other neighbouring countries. It stated that it was an established principle in Europe that, before the initiation of any activities that might cause injuries to neighbouring States, the acting State must negotiate with those States. The Netherlands Government appears to have been referring to an existing or expected regional standard of behaviour. Similar concern was expressed by the Belgian Parliament, which asked the Government how it intended to resolve the problem. The Government stated that the project had been postponed and that the matter was being negotiated with the Netherlands Government. The Belgian Government further assured Parliament that it respected the principles set out in the Benelux accords, to the effect that the parties should inform each other of those of their activities that might have harmful consequences for the other member States.80

#### 1. DEFINITION OF HARM

- 114. The characterization of harm or injury for the purposes of impact assessment may differ from that of harm or injury entailing liability; harm or injury requiring prior consultations may or may not be compensable as a consequence of liability.
- 115. Injury, for purposes of prior negotiation and consultation, may be subdivided into material, non-material and potential. There is no intention here clearly to define injury or harm. For the purposes of this study, material harm means "physical", "quantitative" or "tangible" injury to a State's interests. Non-material harm refers to moral or qualitative harm, for example an affront to the dignity or respect of a State, such as the broadcasting of material to another State that is inconsistent with its internal order and its territorial integrity. With regard to some types of activities, such as nuclear tests in the atmosphere, it is almost certain that

<sup>&</sup>lt;sup>70</sup> I.C.J. Reports 1974, p. 34.

<sup>&</sup>lt;sup>71</sup> Details concerning these negotiations are derived in part from the article by Günther Handl, "Conduct of abnormally dangerous activities in frontier areas: The case of nuclear power plant siting", *Ecology Law Quarterly*, Berkeley, Cal., vol. 7, 1978, p. 1.

<sup>&</sup>lt;sup>72</sup> Österreichische Zeitschrift für Aussenpolitik, vol. 15, 1975, p. 290, cited in Handl, loc. cit., p. 28 and footnote 136.

<sup>&</sup>lt;sup>13</sup> Neue Zürcher Zeitung (foreign edition), No. 118, 1 May 1974, p. 27, cited in Handl, loc. cit., p. 29, footnote 141.

<sup>&</sup>lt;sup>74</sup> Österreichische Zeitschrift . . . , vol. 12, 1972, p. 349, and idem, vol. 14, 1974, p. 224, cited in Handl, loc. cit., p. 29, footnote 143.

<sup>&</sup>lt;sup>75</sup> Österreichische Zeitschrift . . . , vol. 13, 1973, p. 162, cited in Handl, loc. cit., p. 29, footnote 144.

<sup>&</sup>lt;sup>76</sup> The Austrian Foreign Minister stated that this position had been communicated to the Swiss Government. See *Österreichische Zeitschrift*..., vol. 14, 1974, p. 288, cited in Handl, *loc. cit.*, p. 29 and footnote 145.

<sup>&</sup>quot; See footnote 73 above.

<sup>74</sup> Handl, loc. cit., p. 30.

<sup>&</sup>quot; Ibid.

<sup>\*\*</sup> Belgium Parliament, Questions et réponses (Questions and answers) bulletin, 19 July 1973.

at some point in the future some harm will be done to certain interests. However, with respect to other types of activities, harm is not expected to result in every case but may result in some cases only. This latter type of injury is referred to as potential injury and includes prejudice to future interests and harm likely to result from accidental injuries. The characterization of substancial harm, and its point of separation from tolerable harm not requiring prior negotiation and consultation, is a difficult issue that does not appear to have been resolved or treated uniformly in State practice. Some treaties enumerate the kinds of injuries that are not to be tolerated among the parties, so that activities leading to them are prohibited. Other treaties refer in general terms to activities or certain activities leading to some injuries. There are also treaties and judicial decisions which require consultation and prior negotiation for any activity. It would not be totally accurate, however, to assume that the requirement of negotiation and consultation in the latter case is due to the inherent character of certain activities themselves, and not of the injuries they cause. When it is known that certain iniuries will always be caused by certain activities, the activities themselves are regulated so as to prevent or minimize their harmful effects.

#### (a) Multilateral agreements

116. In some multilateral conventions, the concept of harm is described in general terms as a condition affecting human life and changing the quality of a shared resource such as marine fauna and flora. These conventions, then, enumerate hazardous substances whose introduction into shared domains or the territory of another party is considered harmful. Some conventions list substances whose disposal should be progressively either eliminated or restricted. The list of substances not injurious for purposes of liability may nevertheless provide sufficient grounds for negotiation and consultation. For example, the 1974 Convention for the Prevention of Marine Pollution from Land-based Sources provides in article 4 an obligation for parties to eliminate or restrict the pollution of the environment by certain substances:

#### Article 4

- 1. The Contracting Parties undertake:
- (a) to eliminate, if necessary by stages, pollution of the maritime area from land-based sources of substances listed in part I of annex A to the present Convention;
- (b) to limit strictly pollution of the marine area from land-based sources of the substances listed in part II of annex A to the present Convention.
- 2. In order to carry out the undertakings in paragraph 1 of this article, the Contracting Parties, jointly or individually as appropriate, shall implement programmes and measures:
- (a) for the elimination, as a matter of urgency, of pollution of the maritime area from land-based sources by substances listed in part I of annex A to the present Convention;
- (b) for the reduction or, as appropriate, elimination of pollution of the maritime area from land-based sources by substances listed in part II of annex A to the present Convention. These substances shall be discharged only after approval has been granted by the appropriate authorities within each contracting State. Such approval shall be periodically reviewed.
- 3. The programmes and measures adopted under paragraph 2 of this article shall include, as appropriate, specific regulations or standards governing the quality of the environment, discharges into the maritime area, such discharges into watercourses as affect the

maritime area, and the composition and use of substances and products. These programmes and measures shall take into account the latest technical developments.

The programmes shall contain time-limits for their completion.

- 4. The Contracting Parties, furthermore, jointly or individually as appropriate, implement programmes or measures to forestall, reduce or eliminate pollution of the maritime area from land-based sources by a substance not then listed in annex A to the present Convention, if scientific evidence has established that a serious hazard may be created in the maritime area by that substance and if urgent action is necessary.
- 117. Thus the Convention lists substances the deposits of which are to be *prohibited*. Part I of annex A lists these substances:

#### Annex A

The allocation of substances to parts I, II and III below takes account of the following criteria:

- (a) persistence;
- (b) toxicity or other noxious properties;
- (c) tendency to bio-accumulation.

These criteria are not necessarily of equal importance for a particular substance or group of substances, and other factors, such as the location and quantities of the discharge, may need to be considered.

#### PART I

The following substances are included in this part:

- (i) because they are not readily degradable or rendered harmless by natural processes; and
- (ii) because they may either:
  - (a) give rise to dangerous accumulation of harmful material in the food chain, or
  - (b) endanger the welfare of living organisms causing undesirable changes in the marine ecosystems, or
  - (c) interfere seriously with the harvesting of sea foods or with other legitimate uses of the sea; and
- (iii) because it is considered that pollution by these substances necessitates urgent action:
- Organohalogen compounds and substances which may form such compounds in the marine environment, excluding those which are biologically harmless, or which are rapidly converted in the sea into substances which are biologically harmless;
- 2. Mercury and mercury compounds;
- 3. Cadmium and cadmium compounds;
- 4. Persistent synthetic materials which may float, remain in suspension or sink, and which may seriously interfere with any legitimate use of the sea;
- 5. Persistent oils and hydrocarbons of petroleum origin.
- 118. Part II of annex A lists substances the disposal of which is to be *strictly limited*:

#### PART II

The following substances are included in this part because, although exhibiting similar characteristics to the substances in part I and requiring strict control, they seem less noxious or are more readily rendered harmless by natural processes:

- Organic compounds of phosphorus, silicon, and tin and substances which may form such compounds in the marine environment, excluding those which are biologically harmless, or which are rapidly converted in the sea into substances which are biologically harmless;
- 2. Elemental phosphorus;
- 3. Non-persistent oils and hydrocarbons of petroleum origin;
- 4. The following elements and their compounds:

arsenic

chromium

copper lead

nickel

zinc;

- Substances which have been agreed by the Commission as having a
  deleterious effect on the taste and/or smell of products derived
  from the marine environment for human consumption.
- 119. Part III includes a list of substances similar to those listed in part I but, because they are already the subject of research by several international organizations and institutions, they are put into a separate category:

#### PART III

The following substances are included in this part because, although they display characteristics similar to those of substances listed in part I and should be subject to stringent controls with the aim of preventing and, as appropriate, eliminating the pollution which they cause, they are already the subject of research, recommendations and, in some cases, measures under the auspices of several international organizations and institutions; those substances are subject to the provisions of article 5:

Radioactive substances, including wastes.

- 120. After giving an exhaustive list of substances the disposal of which must be prohibited or strictly limited, the Convention offers a general definition of the injury that the disposal of substances not listed in part I may cause to another State. Hence there is no reference to specific substances, but only a general reference to injury.
- 121. Article 9 of the Convention provides for consultations when pollution from land-based sources originating in the territory of a contracting party by substances not listed in part I of annex A is likely to prejudice the interests of one or more other parties to the Convention. The reference to potential injury, requiring consultation, should be noted:

#### Article 9

- 1. When pollution from land-based sources originating from the territory of a Contracting Party by substances not listed in part I of annex A to the present Convention is likely to prejudice the interests of one or more of the other Parties to the present Convention, the Contracting Parties concerned undertake to enter into consultation,\* at the request of any one of them, with a view to negotiating a cooperation agreement.
- 2. At the request of any Contracting Party concerned, the Commission referred to in article 15 of the present Convention shall consider the question and may make recommendations with a view to reaching a satisfactory solution.
- 3. The special agreements specified in paragraph 1 of this article may, among other things, define the areas to which they shall apply, the quality objectives to be achieved, and the methods for achieving these objectives including methods for the application of appropriate standards, and the scientific and technical information to be collected.
- 4. The Contracting Parties signatory to these special agreements shall, through the medium of the Commission, inform the other Contracting Parties of their purport and of the progress made in putting them into effect.
- 122. Similarly, the 1974 Convention on the Protection of the Marine Environment of the Baltic Sea Area provides a list of hazardous substances in annex I and of noxious substances and materials in annex II, deposits of which are either prohibited or strictly limited:

#### Annex I

#### HAZARDOUS SUBSTANCES

The protection of the Baltic Sea area from pollution by the substances listed below can involve the use of appropriate technical means, prohibitions and regulations of the transport, trade, handling, application, and final deposition of products containing such substances.

- DDT (1,1,1-trichloro-2,2-bis-(chlorophenyl)-ethane) and its derivatives DDE and DDD.
- 2. PCB's (polychlorinated biphenyls).

#### Annex II

#### NOXIOUS SUBSTANCES AND MATERIALS

The following substances and materials are listed for the purposes of article 6 of the present Convention.

The list is valid for substances and materials introduced as waterborne into the marine environment. The Contracting Parties shall also endeavour to use best practicable means to prevent harmful substances and materials from being introduced as airborne into the Baltic Sea area.

- A. For urgent consideration
- 1. Mercury, cadmium, and their compounds.

ъ

- Antimony, arsenic, beryllium, chromium, copper, lead, molybdenum, nickel, selenium, tin, vanadium, zinc, and their compounds, as well as elemental phosphorus.
- 3. Phenols and their derivatives.
- 4. Phthalic acid and its derivatives.
- 5. Cyanides.
- 6. Persistent halogenated hydrocarbons.
- 7. Polycyclic aromatic hydrocarbons and their derivatives.
- 8. Persistent toxic organosilicic compounds.
- 9. Persistent pesticides, including organophosphoric and organostannic pesticides, herbicides, slimicides and chemicals used for the preservation of wood, timber, wood pulp, cellulose, paper, hides and textiles, not covered by the provisions of annex I of the present Convention.
- 10. Radioactive materials.
- Acids, alkalis and surface active agents in high concentrations or big quantities.
- Oil and wastes of petrochemical and other industries containing lipid-soluble substances.
- 13. Substances having adverse effects on the taste and/or smell of products for human consumption from the sea, or effects on taste, smell, colour, transparency or other characteristics of the water seriously reducing its amenity values.
- 14. Materials and substances which may float, remain in suspension or sink, and which may seriously interfere with any legitimate use of the sea.
- 15. Lignin substances contained in industrial waste waters.
- The chelators EDTA (ethylenedinitrilotetraacetic acid or ethylenediaminetetraacetic acid) and DTPA (diethylenetriaminopentaacetic acid).
- 123. Annex III of the Convention lists goals, criteria and measures relating to the prevention of land-based pollution, and annex IV lays down the measures to be taken to eliminate or minimize pollution caused by ships. Appendices I to IV of annex IV list the different substances derived from oil. Appendix II contains guidelines for the classification of noxious liquid substances; appendix III lists noxious liquid substances carried in bulk; appendix IV lists other liquid substances carried in bulk. Annex V indicates the exceptions from the general prohibition of dumping of waste and other matter in the Baltic Sea area. These exceptions are as follows:

#### Annex V

EXCEPTIONS FROM THE GENERAL PROHIBITION OF DUMPING OF WASTE AND OTHER MATTER IN THE BALTIC SEA AREA

#### Regulation 1

In accordance with paragraph 2 of article 9 of the present Convention the prohibition of dumping shall not apply to the disposal at sea of dredged spoils provided that:

- 1. they do not contain significant quantities and concentrations of substances to be defined by the Commission and listed in annexes 1 and II of the present Convention; and
- 2. the dumping is carried out under a prior special permit given by the appropriate national authority, either
  - (a) within the area of the territorial sea of the Contracting Party; or
- (b) outside the area of the territorial sea, whenever necessary, after prior consultations in the Commission.

When issuing such permits the Contracting Party shall comply with the provisions in regulation 3 of this annex.

#### Regulation 2

- 1. The appropriate national authority referred to in paragraph 2 of article 9 of the present Convention shall:
  - (a) issue special permits provided for in regulation 1 of this annex;
- (b) keep records of the nature and quantities of matter permitted to be dumped and the location, time and method of dumping;
- (c) collect available information concerning the nature and quantities of matter that has been dumped in the Baltic Sea area recently and up to the coming into force of the present Convention, provided that the dumped matter in question could be liable to contaminate water or organisms in the Baltic Sea area, to be caught by fishing equipment, or otherwise to give rise to harm, and the location, time and method of such dumping.
- 2. The appropriate national authority shall issue special permits in accordance with regulation 1 of this annex in respect of matter intended for dumping in the Baltic Sea area:
  - (a) loaded in its territory;
- (b) loaded by a vessel or aircraft registered in its territory or flying its flag, when the loading occurs in the territory of a State not party to the present Convention.
- 3. When issuing permits under subparagraph 1 (a) above, the appropriate national authority shall comply with regulation 3 of this annex, together with such additional criteria, measures and requirements as they may consider relevant.
- 4. Each Contracting Party shall report to the Commission, and where appropriate to other Contracting Parties, the information specified in subparagraph 1 (c) of regulation 2 of this annex. The procedure to be followed and the nature of such reports shall be determined by the Commission.

#### Regulation 3

When issuing special permits according to regulation 1 of this annex the appropriate national authority shall take into account:

- 1. Quantity of dredged spoils to be dumped.
- 2. The content of the matter referred to in annexes I and II of the present Convention.
- 3. Location (e.g. co-ordinates of the dumping area, depth and distance from coast) and its relation to areas of special interest (e.g. amenity areas, spawning, nursery and fishing areas, etc.).
- 4. Water characteristics, if dumping is carried out outside the territorial sea, consisting of:
- (a) hydrographic properties (e.g. temperature, salinity, density, profile);
  - (b) chemical properties (e.g. pH, dissolved oxygen, nutrients);
- (c) biological properties (e.g. primary production and benthic animals).

The data should include sufficient information on the annual mean levels and the seasonal variation of the properties mentioned in this paragraph.

5. The existence and effects of other dumping which may have been carried out in the dumping area.

#### Regulation 4

Reports made in accordance with paragraph 5 of article 9 of the present Convention shall include the following information:

- 1. Location of dumping, characteristics of dumped material, and counter measures taken:
- (a) location (e.g. co-ordinates of the accidental dumping site, depth and distance from the coast);
  - (b) method of deposit;

- (c) quantity and composition of dumped matter as well as its physical (e.g. solubility and density), chemical and biochemical (e.g. oxygen demand, nutrients), and biological properties (e.g. presence of viruses, bacteria, yeasts, parasites);
  - (d) toxicity
- (e) content of the substances referred to in annexes I and II of the present Convention;
- (f) dispersal characteristics (e.g. effects of currents and wind, and horizontal transport and vertical mixing);
- (g) water characteristics (e.g. temperature, pH, redox conditions, salinity and stratification);
- (h) bottom characteristics (e.g. topography, geological characteristics and redox conditions);
- (i) counter measures taken and follow-up operations carried out or planned.
  - 2. General considerations and conditions:
- (a) possible effects on amenities (e.g. floating or stranded material, turbidity, objectionable odour, discolouration and foaming);
- (b) possible effect on marine life, fish and shellfish culture, fish stocks and fisheries, seaweed harvesting and cultures; and
- (c) possible effects on other uses of the sea (e.g. impairment of water quality for industrial use, underwater corrosion of structures, interference with ship operations from floating materials, interference with fishing or navigation and protection of areas of special importance for scientific or conservation purposes).

The Convention allows exceptions from the general prohibition of the dumping of waste. It permits the dumping of some wastes under certain conditions, including prior negotiations with the Commission (regulation 1, para. 2 (b)), obtaining permission from the appropriate national authorities referred to in the Convention (regulation 2) and observing other detailed regulations concerning the content of the waste or the amount and location of the dumping. These requirements take into account the general interest of all the coastal States as well as the special interests of individual coastal States, and accommodate them by permitting the dumping of substances at locations where they cannot cause immediate tangible harm to another State.

124. The Protocol to the 1976 Convention for the Protection of the Mediterranean Sea against Pollution also lists the hazardous substances the dumping of which requires special care:

#### Annex II

The following wastes and other matter the dumping of which requires special care are listed for the purposes of article 5:

- (i) Arsenic, lead, copper, zinc, beryllium, chromium, nickel, vanadium, selenium, antimony and their compounds;
  - (ii) Cyanides and fluorides;
  - (iii) Pesticides and their by-products not covered in annex I;
  - (iv) Synthetic organic chemicals, other than those referred to in annex I, likely to produce harmful effects on marine organisms or to make edible marine organisms unpalatable.
- (i) Acid and alkaline compounds the composition and quantity
  of which have not yet been determined in accordance with the
  procedure referred to in annex I, paragraph A. 8.
  - (ii) Acid and alkaline compounds not covered by annex I, excluding compounds to be dumped in quantities below thresholds which shall be determined by the Parties in accordance with the procedure laid down in article 14, paragraph 3, of this Protocol.
- Containers, scrap metal and other bulky wastes liable to sink to the sea bottom which may present a serious obstacle to fishing or navigation.
- 4. Substances which, though of a non-toxic nature, may become harmful owing to the quantities in which they are dumped, or which are liable to reduce amenities seriously or to endanger human life or marine organisms or to interfere with navigation.

- 5. Radioactive waste or other radioactive matter which will not be included in annex I. In the issue of permits for the dumping of this matter, the Parties should take full account of the recommendations of the competent international body in this field, at present the International Atomic Energy Agency.
- 125. Annex III of the Protocol lists the *factors* to be considered in establishing criteria governing the issue of permits for the dumping of matter at sea. They include the *quantity* of waste, the *site* of dumping, etc.:

#### Annex III

The factors to be considered in establishing criteria governing the issue of permits for the dumping of matter at sea taking into account article 7 include:

#### A. Characteristics and composition of the matter

- Total amount and average compositions of matter dumped (e.g. per year).
- 2. Form (e.g. solid, sludge, liquid or gaseous).
- 3. Properties: physical (e.g. solubility and density), chemical and biochemical (e.g. oxygen demand, nutrients) and biological (e.g. presence of viruses, bacteria, yeasts, parasites).
- 4. Toxicity.
- 5. Persistence: physical, chemical and biological.
- Accumulation and biotransformation in biological materials or sediments
- Susceptibility to physical, chemical and biochemical changes and interaction in the aquatic environment with other dissolved organic and inorganic materials.
- 8. Probability of production of taints or other changes reducing marketability of resources (fish, shellfish, etc.).

#### B. Characteristics of dumping site and method of deposit

- Location (e.g. co-ordinates of the dumping area, depth and distance from the coast), location in relation to other areas (e.g. amenity areas, spawning, nursery and fishing areas and exploitable resources).
- Rate of disposal per specific period (e.g. quantity per day, per week, per month).
- 3. Methods of packaging and containment, if any.
- Initial dilution achieved by proposed method of release, particularly the speed of the ship.
- Dispersal characteristics (e.g. effects of currents, tides and wind on horizontal transport and vertical mixing).
- 6. Water characteristics (e.g. temperature, pH, salinity, stratification, oxygen indices of pollution—dissolved oxygen (DO), chemical oxygen demand (COD), biochemical oxygen demand (BOD), nitrogen present in organic and mineral form, including ammonia, suspended matter, other nutrients and productivity).
- Bottom characteristics (e.g. topography, geochemical and geological characteristics and biological productivity).
- Existence and effects of other dumpings which have been made in the dumping area (e.g. heavy metal background reading and organic carbon content).
- 9. When issuing a permit for dumping, the Contracting Parties shall endeavour to determine whether an adequate scientific basis exists for assessing the consequences of such dumping in the area concerned, in accordance with the foregoing provisions and taking into account seasonal variations.

#### C. General considerations and conditions

- Possible effects on amenities (e.g. presence of floating or stranded material, turbidity, objectionable odour, discolouration and foaming).
- 2. Possible effects on marine life, fish and shellfish culture, fish stocks and fisheries, seaweed harvesting and culture.
- 3. Possible effects on other uses of the sea (e.g. impairment of water quality for industrial use, underwater corrosion of structures, interference with ship operations from floating materials, interference with fishing or navigation through deposit of waste or solid objects on the sea floor and protection of areas of special importance for scientific or conservation purposes).

 The practical availability of alternative land-based methods of treatment, disposal or elimination or of treatment to render the matter less harmful for sea dumping.

It should be noted that factors 2 and 3 in section C above relate to the *frustration of other uses* of the sea. Factor 4 relates to the possibility of employing alternative methods and bears on the question whether or not some harm should be tolerated in the absence of any other practical method of waste disposal.

126. The aforementioned conventions relate more to the protection of shared domains in which the harmful consequences to coastal States or to a third party may be least direct, immediate or tangible. Nevertheless, the conventions contain detailed instructions concerning the content and quantity of the substances that may be dumped, as well as concerning the balancing of the interests of the parties and the costs and benefits involved. Annex V of the Convention on the Protection of the Marine Environment of the Baltic Sea Area and annex III, section C, of the Protocol to the Convention for the Protection of the Mediterranean Sea against Pollution attempt to introduce factors to be taken into account in order to balance the interests of the parties as well as those of the general community. Similarly, article 194 of the 1982 United Nations Convention on the Law of the Sea calls upon States to take measures to minimize to the fullest possible extent the disposal of certain substances and certain other pollutant materials. Paragraph 3 of that article provides that such measures shall concern, in particular:

- (a) the release of toxic, harmful or noxious substances, especially those which are persistent, from the land-based sources, from or through the atmosphere or by dumping;
- (b) pollution from vessels, in particular measures for preventing accidents and dealing with emergencies, ensuring the safety of operations at sea, preventing intentional and unintentional discharges, and regulating the design, construction, equipment, operation and manning of vessels;
- (c) pollution from installations and devices used in exploration or exploitation of the natural resources of the sea-bed and subsoil, in particular measures for preventing accidents and dealing with emergencies, ensuring the safety of operations at sea, and regulating the design, construction, equipment, operation and manning of such installations or devices;
- (d) pollution from other installations and devices operating in the marine environment, in particular measures for preventing accidents and dealing with emergencies, ensuring the safety of operations at sea, and regulating the design, construction, equipment, operation and manning of such installations or devices.
- 4. In taking measures to prevent, reduce or control pollution of the marine environment, States shall refrain from unjustifiable interference with activities carried out by other States in the exercise of their rights and in pursuance of their duties in conformity with this
- 5. The measures taken in accordance with this part shall include those necessary to protect and preserve rare or fragile ecosystems as well as the habitat of depleted, threatened or endangered species and other forms of marine life.
- 127. The 1976 Convention for the Protection of the Rhine against Chemical Pollution enumerates, in annex I, the substances whose discharge into the Rhine must be *eliminated* and, in annex II, those whose discharge must be *reduced*. The elimination or reduction of discharge of these substances is prescribed on the grounds that they may endanger the uses of the waters of the Rhine, specified as follows:

# Article 1

- 2. . .
- (a) the production of drinking water for human consumption;
- (b) consumption by domestic and wild animals;
- (c) the conservation and development of the natural species of flora and fauna and the preservation of the self-purifying capacity of water;
  - (d) fishing;
- (e) recreational purposes, bearing in mind hygienic and aesthetic requirements;
  - (f) the direct or indirect supply of fresh water to agricultural lands;
- (g) the production of water for industrial purposes; and the need to preserve an acceptable quality of sea water.
- 128. Similar references to injury or harm are made in the 1973 International Convention for the Prevention of Pollution from Ships. The relevant provisions of article 2 are as follows:

# Article 2. Definitions

- (2) "Harmful substance" means any substance which, if introduced into the sea, is liable to create hazards to human health, to harm living resources and marine life, to damage amenities or to interfere with other legitimate uses of the sea, and includes any substance subject to control by the present Convention.
- (3) (a) "Discharge", in relation to harmful substances or effluents containing such substances, means any release howsoever caused from a ship and includes any escape, disposal, spilling, leaking, pumping, emitting or emptying.

This article gives a broad definition of harm: it may include material or non-material injury to human health, injury to the living resources of the sea, as well as interference with the legitimate uses of the sea.

129. Article 1 of the 1979 Convention on Long-range Transboundary Air Pollution defines air pollution in terms of its effects:

#### Article 1

For the purposes of the present Convention:

(a) "air pollution" means the introduction by man, directly or indirectly, of substances or energy into the air resulting in deleterious effects of such a nature as to endanger human health, harm living resources and ecosystems and material property and impair or interfere with amenities and other legitimate uses of the environment, and "air pollutants" shall be construed accordingly;

And in article 1, paragraph 6, of the 1969 Convention on Civil Liability for Oil Pollution Damage, "pollution damage" is defined as "loss or damage caused outside the ship carrying oil by contamination resulting from the escape or discharge of oil from the ship, wherever such escape or discharge may occur, and includes the costs of preventive measures and further loss or damage caused by preventive measures".

130. A general reference to damage by pollution is made in article 198 of the 1982 United Nations Convention on the Law of the Sea. The article requires a State which has become aware of an imminent danger of marine pollution to notify other States which might be affected by it:

# Article 198. Notification of imminent or actual damage

When a State becomes aware of cases in which the marine environment is in imminent danger of being damaged or has been damaged by pollution, it shall immediately notify other States it deems likely to be affected by such damage, as well as the competent international organizations.

Article 206 of the Convention refers to substantial pollution and to significant and harmful changes to the marine environment. This article requires the acting State to communicate reports on the assessment of activities under its jurisdiction when there are reasonable grounds that they may cause the injuries referred to above:

# Article 206. Assessment of potential effects of activities

When States have reasonable grounds for believing that planned activities under their jurisdiction or control may cause substantial pollution of or significant and harmful changes to the marine environment, they shall, as far as practicable, assess the potential effects of such activities on the marine environment and shall communicate reports of the results of such assessments in the manner provided in article 205.

131. Sometimes conventions enumerate or make general references to substances and activities that may frustrate certain other uses of the shared domain. For example, the 1968 European Agreement on the Restriction of the Use of Certain Detergents in Washing and Cleaning Products prohibits the use of certain products containing one or more synthetic detergents which, under conditions of normal use, might adversely affect human or animal health. The relevant articles read:

#### Article 1

The Contracting Parties undertake to adopt measures as effective as possible in the light of the available techniques, including legislation if it is necessary to ensure that:

- (a) in their respective territories, washing or cleaning products containing one or more synthetic detergents are not put on the market unless the detergents in the product considered are, as a whole, at least 80 per cent susceptible to biological degradation;
- (b) the appropriate measurement and control procedures are implemented in their respective territories to guarantee compliance with the provisions of subparagraph (a) of this article.

#### Article 2

Compliance with the provisions of paragraph (a) of article 1 of this Agreement must not result in the usage of detergents which, under conditions of normal use, might affect adversely human or animal health.

132. The 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties grants competence to coastal States to take action to prevent "major harmful consequences" to their coastlines resulting from oil pollution. Paragraph 1 of article I provides:

## Article I

1. Parties to the present Convention may take such measures on the high seas as may be necessary to prevent, mitigate or eliminate grave and imminent danger to their coastline or related interests from pollution or threat of pollution of the sea by oil, following upon a maritime casualty or acts related to such a casualty, which may reasonably be expected to result in major harmful consequences.

The term "major harmful consequences" is not defined in the Convention.

- 133. In at least one convention, a particular interest is accorded legal protection, and any activity frustrating that particular interest requires prior consultation and negotiation. Paragraph 3 of article 1 of the 1960 Convention on the Protection of Lake Constance against Pollution provides:
- 3. Specifically, the riparian States will mutually notify\* each other in advance of projects for water use which, if carried out, could interfere with the interests of another riparian State as regards maintaining the wholesomeness\* of the waters of Lake Constance. Such pro-

jects shall not be carried out\* until they have been jointly discussed by the riparian States, unless either there is danger in delay or the other States have explicitly agreed that they shall be put into effect immediately.

134. A different term for injury is used and a different method of consultation provided for in the 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden, article 1 of which describes environmentally harmful activities as those that entail environmental *nuisance*:

#### Article 1

For the purpose of this Convention, environmentally harmful activities shall mean the discharge from the soil or from buildings or installations of solid or liquid waste, gas or any other substance into watercourses, lakes or the sea and the use of land, the sea-bed, buildings or installations in any other way which entails, or may entail environmental nuisance\* by water pollution or any other effect on water conditions, sand drift, air pollution, noise, vibration, changes in temperature, ionizing radiation, light etc.

The Convention shall not apply in so far as environmentally harmful activities are regulated by a special agreement between two or more of the Contracting States.

The Protocol to the Convention expressly states that the discharges of waste mentioned in article 1 are to be regarded as environmentally harmful only if they entail or may entail a *nuisance* to the surroundings:

In the application of article I, discharge from the soil, or from buildings or installations of solid or liquid waste, gases or other substances into watercourses, lakes or the sea shall be regarded as environmentally harmful activities only if the discharge entails or may entail a nuisance\* to the surroundings.

135. Article 3 of the Convention, without explicitly requiring prior negotiation, provides that any person who is affected or *may be* affected by a *nuisance* caused by environmentally harmful activities in another contracting State may bring before the appropriate authority of that State the question of the *permissibility* of such activities, including measures to prevent damage:

#### Article 3

Any person who is affected or may be affected by a nuisance caused by environmentally harmful activities in another Contracting State shall have the right to bring before the appropriate court or administrative authority of that State the question of the permissibility\* of such activities, including the question of measures to prevent damage,\* and to appeal against the decision of the court or the administrative authority to the same extent and on the same terms as a legal entity of the State in which the activities are being carried out.

The appropriate authority here is the court or the administrative authority of the acting State. Under article 3, the foreign claimant must receive the same treatment from the appropriate court or authorities of the acting State as would a national of that State.

# (b) Bilateral agreements

136. In bilateral agreements, the definition of harm, for the purposes of prior negotiations, varies from a more exact to a more general characterization. For example, in the 1983 Agreement between the United States of America and Mexico, article 2 makes a general reference to "pollution", and article 7, on assessment of projects, refers to significant impacts on the environment. In a number of bilateral agreements the terms "harm", "injury" or "damage" as such are not used.

Instead, reference is made to certain activities, changes in the nature of resources, impact upon material interests, etc. However, bilateral agreements that have a more specific object expressly refer to certain activities that must not be undertaken, or that may be undertaken only in consultation with the other party.

137. Some bilateral agreements provide that, for the commencement of certain activities, and of activities that may have repercussions, prior consultation is required with the other contracting party or with a joint commission. For example, the 1932 Convention between Poland and the Soviet Union relating to their common frontier<sup>82</sup> provides that the contracting parties may not, without the agreement of both parties, remove existing hydraulic installations on frontier waters if the level of the waters were to be changed thereby. Article 14, paragraph 1, of the Convention reads:

#### Article 14

1. The Contracting Parties agree that the presence and further utilization of existing hydraulic installations on frontier waters and on their banks shall not be hindered in any way. If the removal of such installations should involve a change in the *level of the water*,\* the necessary work shall *not*\* be undertaken without the agreement of the frontier authorities of both Contracting Parties.

138. In the 1931 General Convention between Romania and Yugoslavia concerning the hydraulic system, <sup>83</sup> reference is made to any alteration or any measure that might change to an appreciable extent the hydraulic system in the basin. Chapter I, article 3, of the Convention requires the contracting parties to notify each other if they intend to take any measures that might result in such changes. Such notice is a step towards a preliminary agreement between the parties, as provided for in article 292 of the Treaty of Trianon (see para. 106 above).

139. Article III of the 1909 Treaty relating to the boundary waters between Canada and the United States of America provides that the contracting parties shall not undertake activities, other than those prescribed in the Treaty, which may affect the natural level or flow of boundary waters on the other side of the line unless it is agreed upon by at least one party and approved by the International Joint Commission. Thus, under article IV, the contracting parties agree that, except in cases provided for by special agreement between them, they will not permit the construction or maintenance, on their respective territories, of any remedial or protective works or any dams or other obstructions in water flowing from boundary waters or in rivers flowing across the boundary which may raise the natural level of waters on the other side of the boundary, unless it is done with the approval of the International Joint Commission, Articles III and IV provide:

# Article III

It is agreed that, in addition to the uses, obstructions and diversions heretofore permitted or hereafter provided for by special agreement between the Parties hereto, no further or other uses or obstructions or diversions, whether temporary or permanent, of boundary waters on either side of the line, affecting the natural level or flow of boundary

<sup>&</sup>quot; See footnote 16 above.

<sup>&</sup>lt;sup>92</sup> Convention of 10 April 1932 between Poland and the USSR concerning juridical relations on the State frontier.

<sup>3</sup> See footnote 60 above.

waters on the other side of the line, shall be made except by authority of the United States or the Dominion of Canada within their respective jurisdictions and with the approval, as hereinafter provided, of a joint commission, to be known as the International Joint Commission.

#### Article IV

The High Contracting Parties agree that, except in cases provided for by special agreement between them, they will not permit the construction or maintenance on their respective sides of the boundary of any remedial or protective works or any dams or other obstructions in waters flowing from boundary waters or in waters at a lower level than the boundary in rivers flowing across the boundary, the effect of which is to raise the natural level of waters on the other side of the boundary unless the construction or maintenance thereof is approved by the aforesaid International Joint Commission.

It is further agreed that the waters herein defined as boundary waters and waters flowing across the boundary shall not be polluted on either side to the injury of health or property on the other.

- 140. In its 1974 Agreement with Canada, 84 the United States of America, prior to launching two rockets from Canadian territory, assured Canada that, in the event of loss of life, personal injury or damage or loss to property resulting from those rocket launches, the United States Government would take all necessary measures to comply fully with its obligations under the 1967 Treaty on Principles governing the Activities of States in the Exploration and Use of Outer Space, including the Moon and Other Celestial Bodies, in particular article VII thereof, as well as under international law. Article VII of the Treaty does not specify the kinds of injury that may entail liability, whereas the Agreement between Canada and the United States refers to specific injuries. Moreover, article VII refers to the liability of both the launching State and the State from whose territory the rocket has been launched; in the Agreement, the United States alone assumes liability. The injuries referred to in the Agreement appear to be synonymous with "injuries that may entail liability".
- 141. Certain other bilateral agreements contain more general as well as some more specific references to injury. For example, the 1961 Treaty between the USSR and Poland<sup>83</sup> provides in article 29, paragraph 4, for notification of the outbreak of *forest fires* near the frontier. Again, in the 1958 Agreement between Czechoslovakia and Poland, <sup>86</sup> article 3 requires the consent of the other party for *any activity* that might affect its water economy, as well as certain specific uses of the waters shared by the parties. Article 3 reads:

#### Article 3

- 1. Neither Contracting Party may, without the consent of the other Contracting Party, carry out any works in frontier waters which may affect the latter Party's water economy.\*
- 2. The Contracting Parties shall come to agreement on the amount of water to be taken from frontier waters for domestic, industrial, power generation and agricultural requirements and on the discharge of waste water.
- <sup>84</sup> Exchange of notes of 31 December 1974 between the United States of America and Canada constituting an agreement relating to liability for loss or damage from certain rocket launches.
- <sup>43</sup> Treaty of 15 February 1961 between the USSR and Poland concerning the régime of the Soviet-Polish State frontier and co-operation and mutual assistance in frontier matters.
- <sup>16</sup> Agreement of 21 March 1958 between Czechoslovakia and Poland concerning the use of water resources in frontier waters.

- 3. The Contracting Parties shall come to an agreement in each particular case on what runoff ratios are to be preserved in frontier waters.
- 4. The Contracting Parties have agreed to abate the pollution of frontier waters and to keep them clean to such extent as is specifically determined in each particular case in accordance with the economic and technical possibilities and requirements of the Contracting Parties.
- 5. When installations discharging polluted water into frontier waters are constructed or reconstructed, treatment of the waste water shall be required.

Article 3 of the 1922 Convention between Finland and the RSFSR prohibits, unless a special agreement has been concluded between the contracting States, the diversion of water from watercourses, the erection of constructions that may cause damage, the alteration of the existing depth or conditions of parts of watercourses situated in the territory of the other contracting State. or encroachment upon channels used for navigation or timber-floating.87 Article 12 of the 1929 Convention between Norway and Sweden\*\* requires the approval of both contracting parties of activities likely to entail any considerable inconvenience in the territory of one of them in the use of a watercourse for navigation or floating, to hinder the movement of fish or to disturb the conditions governing the water supply over an extensive area:

#### APPROVAL OF THE OTHER COUNTRY

#### Article 12

- 1. One country may not authorize an undertaking unless the other country has given its approval, if the undertaking is likely to involve any considerable inconvenience in the latter country in the use of a watercourse for navigation or floating or to hinder the movement of fish to the detriment of fishing in that country, or if the undertaking is likely to cause considerable disturbance in conditions governing the water supply over an extensive area.
- 2. If there is no reason to believe that the undertaking will produce the effects mentioned in paragraph 1 in the other country, that country cannot oppose the execution of the undertaking.

Article 13 of the same Convention provides that the consent of the potentially injured party is subject only to the planning of the work or the prevention or reduction of public damage or nuisance:

#### Article 13

If the other country's consent is necessary, the question shall be decided in accordance with the principles applicable to similar installations, works or operations under that country's laws, subject, however, to the provisions of articles 4 and 5. The consent may not be subjected to other conditions than those referring to the planning of the work or the prevention or reduction of public damage or nuisances.

142. Activities interfering with the free discharge of waters, changing the quality of waters, or causing flood in the territory of a contracting party may require the approval of a mixed commission. This is stipulated in article 2 of the 1955 Agreement between Yugoslavia and Romania concerning questions of water control at the frontier:\*9

<sup>87</sup> See footnote 27 above.

<sup>&</sup>lt;sup>88</sup> See footnote 36 above.

<sup>&</sup>lt;sup>89</sup> Agreement of 7 April 1955 between Yugoslavia and Romania concerning questions of water control on water control systems and watercourses on or intersected by the State frontier.

#### Article 2

- 1. The two Contracting States undertake, each in its own territory and jointly on the frontier line, to maintain the beds and installations in good condition, and where necessary to improve their condition, and to keep the installations in operation on water control systems and watercourses and in valleys and depressions on or intersected by the State frontier.
- 2. The erection of any new installations and the execution of any new works, in the territory of either Contracting State, which may change the existing régime of the waters, interfere with the free discharge of the waters where it now exists, change the quality of the waters, or cause flooding on water control systems or watercourses or in valleys or depressions on or intersected by the State frontier shall be referred to the Mixed Commission for examination.
- 143. The 1967 Agreement between the Federal Republic of Germany and Austria<sup>90</sup> provides, in article 2, paragraph 2, concerning the hours of operation of the Salzburg Airport, that, before any extension of the hours of operation that might adversely affect the interests of the Federal Republic of Germany in respect of safety, order or noise abatement, the views of the competent German authorities must be ascertained. The paragraph reads:
- 2. If it is contemplated that the hours of operation of the Salzburg airport shall be extended to include periods between 11 p.m. and 6 a.m. local time, permission to change the existing hours of operation shall be granted only if German interests with respect to safety and order or aircraft noise abatement are not thereby affected. Before granting permission, the competent Austrian aeronautical authorities shall ascertain the views of the competent German aeronautical authorities.

Some bilateral agreements, without reference to injury, require the consent of both parties for certain uses of frontier waters. For example, article 6 of the 1934 Agreement between Belgium and Great Britain concerning water rights on the boundary between Tanganyika and Ruanda-Urundi<sup>91</sup> provides that, if either contracting party intends to *utilize the waters of their joint rivers* or to permit any person to utilize such waters for *irrigation purposes*, such contracting party shall give notice to the other party six months in advance, in order to permit the consideration of any objections which the other party may wish to raise:

#### Article 6

In the event of either Contracting Government desiring to utilize the waters of any river or stream on the aforesaid boundary or to permit any person to utilize such water for irrigation purposes, such Contracting Government shall give to the other Contracting Government notice of such desire six months before commencing operations for the utilization of such waters, in order to permit of the consideration of any objections which the other Contracting Government may wish to raise.

144. Similarly, article 15 of the 1932 Convention between Poland and the Soviet Union<sup>92</sup> provides for *prior agreement* between the two Governments for the *erection on waters of new dikes and construction of new mills or other hydraulic installations*:

#### Article 15

The erection on frontier waters of new dykes and the construction of new mills or other hydraulic installations shall in every case be sub-

ject to prior agreement between the competent authorities of the Contracting Parties.

- 145. Similarly, article 5 of the 1971 Act of Santiago concerning Hydrologic Basins, signed by Argentina and Chile, also requires prior notification of activities to be undertaken in their joint waters:
- 5. If a State intends to utilize a common lake or successive river, it shall first transmit to the other State the plans for the works, the plan of operations and other data which may be useful in determining the impact of the works in the territory of the neighbouring State.

Under the 1949 Agreement between Norway and the USSR,<sup>93</sup> the exploitation of mineral deposits near the frontier area requires the consent of the other party. Article 18, paragraph 2, refers to the safeguarding of the *frontier line* and requires a 20 metre-wide space on either side of the borderline within which exploration of mineral resources may not be undertaken, except with the consent of both parties (see para. 44 above). It should be noted that the reference to harm in paragraph 1 of this article is general and undefined.

146. In at least one bilateral agreement, the concept of harm, injury or damage is defined by reference to a standard stipulated in a multilateral convention. Article 12 of the 1970 Treaty between Liberia and the Federal Republic of Germany concerning the use of Liberian ports by the nuclear ship *Otto Hahn*<sup>94</sup> provides that terms such as "nuclear damage" have the same meaning as in the 1962 Convention on the Liability of Operators of Nuclear Ships:

#### Article 12

The terms "nuclear damage", "nuclear incident", "nuclear fuel" and "radioactive products or waste" as used in articles 13-20 of this Treaty shall have the same meaning as in the Convention on the Liability of Operators of Nuclear Ships opened for signature in Brussels on May 25, 1962, hereinafter referred to as "the Convention".

Thus the definition of "nuclear injury", as agreed between the parties, is established in accordance with an international standard.

- 147. By contrast, two other bilateral agreements refer to domestic law standards for injuries. The United States of America, in two agreements, one concluded with the Netherlands, in 1963,95 the other with Ireland, in 1964,96 concerning visits of the N.S. Savannah to the ports of those countries, accepted "liability" for any "nuclear incident", as those terms are defined in section 11 of the United States Atomic Energy Act of 1954, as amended.97 This provision refers to domestic law. It should be noted that, in the above agreements, injury, for the purposes of prior negotiation, means injury entailing liability.
- 148. In a few bilateral agreements, references to harm or injury are general and rather abstract. For example, article 2 of the 1956 Treaty between Hungary and Austria<sup>98</sup> provides that activities that may cause injury are subject to prior notification and consultation, with

<sup>&</sup>quot; See footnote 18 above.

<sup>\*1</sup> Agreement of 22 November 1934 between Belgium and Great Britain regarding water rights on the boundary between Tanganyika and Ruanda-Urundi.

<sup>32</sup> See footnote 82 above.

<sup>33</sup> See footnote 17 above.

<sup>94</sup> See footnote 19 above.

<sup>93</sup> See footnote 22 above.

<sup>%</sup> See footnote 21 above.

<sup>97</sup> United States Code, title 42, section 2014 q (Definitions).

no further specification; the contracting parties are prohibited from taking any unilateral action or from carrying out, without the consent of the other party, any measures or works on the frontier waters that would adversely affect water conditions in the territory of the other contracting State. Article 2, paragraph 1, provides:

#### Article 2. General obligations

- 1. Each Contracting Party undertakes to refrain from unilaterally—without the consent of the other Contracting Party—carrying out any measures or works on frontier waters (article 1, subparagraph 1) which would adversely affect water conditions in the territory of the other Contracting Party. Consent may be refused only if the grounds for such refusal are duly set forth.
- 149. Again, article 4 of the 1954 Convention between Yugoslavia and Austria concerning water economy questions relating to the Drava requires the Austrian Government to negotiate with the Yugoslav Government if it contemplates plans for new installations that would *divert water* from the Drava basin or for construction work that might be *detrimental* to Yugoslavia. The Convention does not define the term "detriment". The article reads in part:

#### Article 4

Should the Austrian authorities seriously contemplate plans for new installations to divert water from the Drava basin or for construction work which might affect the Drava river régime to the detriment of Yugoslavia, the Austrian Federal Government undertakes to discuss such plans with the Federal People's Republic of Yugoslavia prior to legal negotiations concerning rights in the water.

- 150. In the 1928 Convention between Hungary and Czechoslovakia, 9 the parties agree not to allow any works calculated to disturb the flow of the water or the regulation of frontier watercourses. Paragraph 2 (c) of article 26 of the Convention provides:
- (c) The Contracting Parties shall not allow any works calculated to disturb the flow of the water or the regularization of frontier water-courses. If works contemplated are likely to have an undesirable effect on the bed of frontier water-courses, the competent technical department of the other Party must be consulted.

Under the 1960 Treaty between the Netherlands and the Federal Republic of Germany, 100 the contracting party which intends to take measures that may substantially affect the use and management of water resources in the territory of the other contracting party must notify the Permanent Boundary Waters Commission. Article 60 of the Treaty reads:

#### Article 60

1. If it is intended to carry into effect, within the territory of one of the Contracting Parties, any measures which may substantially affect the use and management of water resources in the territory of the other Contracting Party, or to allow such measures to be carried into

<sup>98</sup> Treaty of 9 April 1956 between Hungary and Austria concerning the regulation of water economy questions in the frontier region.

effect, the Permanent Boundary Waters Commission shall be notified thereof as soon as possible.

2. The Contracting Parties shall notify each other of the authorities or corporations within its territory which are competent to make the notification referred to in paragraph 1.

# (c) Judicial decisions and State practice other than agreements

- 151. Most judicial decisions, when referring to harm for purposes of consultation and negotiation, do so in very general terms as of something that affects human life and changes the quality of a shared resource, such as sea water or land-based resources, or the quality of a resource within the exclusive jurisdiction of the injured State, such as land, agriculture or even population. These decisions refer to many hazardous substances which, when introduced into the shared domain or into the territory of another State, would have such effects. In most cases, harm has been considered as interference with or denial of an interest resulting from activities conducted by the acting State in its own territory or within the shared domain.
- There are references to harm in the 1957 arbitral 152 award in the Lake Lanoux case and in the judgment rendered by the International Court of Justice on 19 April 1949 in the Corfu Channel case (merits). In the former case, the tribunal referred to the duty of safeguarding the interests of the parties to a treaty, and stated that those interests went beyond specific legal rights. That statement may be interpreted as meaning that the tribunal also considered the safeguarding of interests not legally protected. In the opinion of the tribunal, then, States could not *unilaterally* determine and evaluate the interests of another State in regard to an international watercourse (see para. 109 above). In the Corfu Channel case, the Court's ruling emphasized a different aspect. The Court stated that Albania had the obligation "not to allow knowingly its territory to be used for acts contrary to the rights of other States".\* 101 The rights here may be interpreted as only "legally protected interests"; it is a narrower concept than that of interests as stated in the Lake Lanoux award. The different emphasis placed by the tribunal and the Court on *interests* to be protected may be explained by the following factors: first, the location of the activities, involving different competing interests and principles of varying degrees of importance. In the Corfu Channel case, the activity occurred in an international channel located within the territorial jurisdiction of the acting State. Certain rights of passage, similar to an international servitude, have been recognized for other States with respect to the use of another State's territory. In this case, it was the right of innocent passage. Hence the coastal State was not, it seems, obliged to consult, negotiate or notify other States exercising innocent passage with regard to activities or conditions that did not affect their right of innocent passage. By contrast, the Lake Lanoux decision dealt with a watercourse in which more than one State had territorial sovereignty. Thus it was not a matter of mere servitude. In addition to the principle of territorial sovereignty, there is a principle of good neighbour-

<sup>&</sup>quot;Convention of 14 November 1928 between Hungary and Czechoslovakia relating to the settlement of questions arising out of the delimitation of the frontier between the two countries (Frontier Statute).

<sup>&</sup>lt;sup>100</sup> Treaty of 8 April 1960 between the Netherlands and the Federal Republic of Germany concerning the course of the common frontier, the boundary waters, real property situated near the frontier, traffic crossing the frontier on land and via inland waters, and other frontier questions (Frontier Treaty).

<sup>101</sup> I.C.J. Reports 1949, p. 22.

liness, which entails different expectations of behaviour between neighbouring States. The second factor is that the Court in the *Corfu Channel* case may have been contemplating activities causing injuries that are not to be tolerated *per se*, such as the frustration of another State's rights, whereas the tribunal in the *Lake Lanoux* case was dealing with injuries which may be tolerated, but only by consent.

- 153. A number of judicial decisions and instances of State practice relate to material injury. Most often this type of injury is of an economic nature or entails injury to human well-being. Material injury may occur to another State's interests either within the shared domain or within the area under the exclusive territorial control of the injured State. In the *Trail Smelter* case, the tribunal formulated the concept of injury in very broad terms (see para. 47 above).
- 154. In equally broad terms, the Italian Court of Cassation stated, in the Société d'énergie électrique du littoral méditerranéen v. Compagnia imprese elettriche liguri case (1939) [hereinafter referred to as the Roya case]:102

... although a State, in the exercise of its rights of sovereignty, may subject public rivers to whatever régime it deems best, it cannot disregard the international duty, derived from that principle, not to impede or to destroy, as a result of this régime, the *opportunity\** of the other States to avail themselves of the flow of water for their own national needs.

According to this decision, the *frustration* or *destruction* of other States' opportunities is *prohibited*. Although many judicial decisions refer to material injury in broad, general and undefined terms as some type

102 Il Foro italiano (Rome), vol. 64, 1939, part I, col. 1036. The following is a summary of the facts of this case:

of injury to interests, in most cases they focus on a particular injury to a particular interest.

- 155. In the Fisheries Jurisdiction cases, the International Court of Justice held that Iceland might not unilaterally extend its exclusive fishing rights beyond its territorial waters where such extension would harm the economic interests of other States. The Court pointed to unemployment in the fishing and related industries as specific injuries resulting from the unilateral determination:
- 64. The Applicant further states that in view of the present situation of fisheries in the North Atlantic, which has demanded the establishment of agreed catch-limitations of cod and haddock in various areas, it would not be possible for the fishing effort of United Kingdom vessels displaced from the Icelandic area to be diverted at economic levels to other fishing grounds in the North Atlantic. Given the lack of alternative fishing opportunity, it is further contended, the exclusion of British fishing vessels from the Icelandic area would have very serious adverse consequences, with immediate results for the affected vessels and with damage extending over a wide range of supporting and related industries. It is pointed out in particular that widespread unemployment would be caused among all sections of the British fishing industry and in ancillary industries and that certain ports—Hull, Grimsby and Fleetwood—specially reliant on fishing in the Icelandic area, would be seriously affected.<sup>103</sup>

The United Kingdom's economic interests receiving protection were historically based:

63. In this case, the Applicant has pointed out that its vessels have been fishing in Icelandic waters for centuries and that they have done so in a manner comparable with their present activities for upwards of 50 years. Published statistics indicate that from 1920 onwards, fishing of demersal species by United Kingdom vessels in the disputed area has taken place on a continuous basis from year to year, and that, except for the period of the Second World War, the total catch of those vessels has been remarkably steady. Similar statistics indicate that the waters in question constitute the most important of the Applicant's distant-water fishing grounds for demersal species.<sup>104</sup>

This passage may be interpreted as meaning that the Court in this context was referring to the legally protected economic interests of the United Kingdom, interests that had acquired legal protection based on history. A different interpretation may also be given. References by the Court to historic use by the United Kingdom may have been intended only to establish a fact: the real and genuine dependence of the United Kingdom on fishing in the area where Iceland was imposing its unilateral prescription. The Court may not have been concerned about whether the United Kingdom's dependence on fishing resources was legally protected or not. The Court at the same time noted that the rights and interests of the States were not static concepts, but changed with changing economic dependence on the resource:

70. This is not to say that the preferential rights of a coastal State in a special situation are a static concept, in the sense that the degree of the coastal State's preference is to be considered as fixed for ever at some given moment. On the contrary, the preferential rights are a function of the exceptional dependence of such a coastal State on the fisheries in adjacent waters and may, therefore, vary as the extent of that dependence changes. Furthermore, as was expressly recognized in the 1961 Exchange of Notes, a coastal State's exceptional dependence on fisheries may relate not only to the livelihood of its people but to its economic development. In each case, it is essentially a matter of appraising the dependence of the coastal State on the fisheries in ques-

<sup>&#</sup>x27;On December 17, 1914, a Convention was concluded in Paris between France and Italy for the regulation in the common interest of the utilization of the waters of the river Roya which flows partly in Italy and partly in France. Article 1 of the Convention provided that the High Contracting Parties will reciprocally refrain from using or from permitting the use of the hydraulic power of the Roya and of its tributaries within the territory subject to their exclusive sovereignty in any manner which might lead to a noticeable modification of the existing régime and of the natural flow of the water in the territory of the lower riparian State. Articles 2 and 3 dealt with the rights of the Contracting Parties in respect of the waters of the Roya where the river formed the common frontier. Article 4 entrusted a permanent international commission consisting of delegates of the two Contracting Parties with the application of the principles laid down in the Convention. Article 5 maintained, as between the two Governments, the agreements reached and obligations incurred by the private users in France and Italy of the water power of the Roya. In substance, article 5 referred to an agreement of February 11, 1914, between the plaintiffs and the defendants which created a modus vivendi to the effect that the defendants promised not to interfere with the waters of the Roya in a manner which might affect the plaintiffs and to remove the effects of interferences in the past. Subsequently, the defendants created new power-stations and plants on Italian territory which, it was alleged, adversely affected the plaintiffs. As a result, the plaintiffs claimed damages for breach of contract in the Court of Nice (France) and obtained judgment in their favour. This decision was affirmed by the Court of Appeal of Aix and by the French Court of Cassation. The plaintiffs now brought an action in the Court of Appeal of Genoa to have the French judgment rendered executory in Italy in accordance with the Franco-Italian Convention of June 3, 1930, for the execution of judgments in commercial matters." (Annual Digest and Reports of Public International Law Cases, 1938-1940 (London, 1942), vol. 9, case No. 47, p. 121.)

<sup>103</sup> I.C.J. Reports 1974, p. 28.

<sup>104</sup> Ibid.

tion in relation to that of the other State concerned and of reconciling them in as equitable a manner as is possible. 105

The Court requested the parties to negotiate.

- 156. Harm, for purposes of impact assessment, must be more than a mere change in the natural situation of resources. Many variables have been taken into account to determine what constitutes harm. The most important, it seems, is that there must be some *value deprivation for human beings*. In the *Lake Lanoux* case, the tribunal, discussing the diversion of international waters from one river basin to the next and in response to Spain's claim that any diversion of international waters was, *per se*, an injury to Spanish interests, noted that the mere withdrawal of water was insufficient to base a claim for injury:
- 8. The prohibition of compensation between the two basins, in spite of equivalence between the water diverted and the water restored, unless the withdrawal of water is agreed to by the other Party, would lead to the prevention in a general way of a withdrawal from a water-course belonging to River Basin A for the benefit of River Basin B, even if this withdrawal is compensated for by a strictly equivalent restitution effected from a watercourse of River Basin B for the benefit of River Basin A. The Tribunal does not overlook the reality, from the point of view of physical geography, of each river basin, which constitutes, as the Spanish Memorial (at p. 53) maintains, "a unit". But this observation does not authorize the absolute consequences that the Spanish argument would draw from it. The unity of a basin is sanctioned at the juridical level only to the extent that it corresponds to human realities. The water which by nature constitutes a fungible item may be the object of a restitution which does not change its qualities in regard to human needs. A diversion with restitution, such as that envisaged by the French project, does not change a state of affairs organized for the working of the requirements of social

The state of modern technology leads to more and more frequent justifications of the fact that waters used for the production of electric energy should not be returned to their natural course. Water is taken higher and higher up and it is carried even farther, and in so doing it is sometimes diverted to another river basin, in the same State or in another country within the same federation, or even in a third State. Within federations, the judicial decisions have recognized the validity of this last practice (Wyoming v. Colorado, United States Reports, vol. 259, p. 419) and the instances cited by F. J. Berber, Die Rechtsquellen des internationalen Wassernutzungsrechts, p. 180, and by G. Sauser Hall, "L'utilisation industrielle des fleuves international de La Haye, 1953-II, vol. 83, p. 544; for Switzerland, Recueil officiel des arrêts du Tribunal fédéral, vol. IV, pp. 14 et seq. 100

Nor is it sufficient, as Spain claimed, that the activity may place into the hands of the acting State an instrument giving it a means of violating its international pledges. The tribunal stated:

... But we must go still further; the growing ascendancy of man over the forces and the secrets of nature has put into his hands instruments which he can use to violate his pledges just as much as for the common good of all; the risk of an evil use has so far not led to subjecting the possession of these means of action to the authorization of the States which may possibly be threatened. Even if we took our stand solely on the ground of neighbourly relations, the political risk alleged by the Spanish Government would not present a more abnormal character than the technical risk which was discussed above. In any case, we do not find either in the Treaty and the Additional Act of May 26, 1866, or in international common law, any rule that forbids one State, acting to safeguard its legitimate interests, to put itself in a situation which would in fact permit it, in violation of its international pledges, seriously to injure a neighbouring State. 107

- 157. The tribunal noted, in addition, that not only might the utilization of international waters by one riparian State not be harmful to the other, but that the utilization might indeed be beneficial to the latter. Not only had France not diverted any of Spain's waters to its own uses without restitution, but French use had also stabilized and equalized the annual water flow:
- ... Thus, if it is admitted that there is a principle which prohibits the upstream State from altering the waters of a river in such a fashion as seriously to prejudice the downstream State, such a principle would have no application to the present case, because it has been admitted by the Tribunal, in connection with the first question examined above, that the French scheme will not alter the waters of the Carol. . . . 108
- 6. In effect, thanks to the restitution effected by the devices described above, none of the guaranteed users will suffer in his enjoyment of the waters (this is not the subject of any claim founded on article 9); at the lowest water level, the volume of the surplus waters of the Carol, at the boundary, will at no time suffer a diminution; it may even, by virtue of the minimum guarantee given by France, benefit by an increase in volume assured by the waters of the Ariège flowing naturally to the Atlantic.<sup>109</sup>
- 158. The tribunal in fact stated that the claims formulated by Spain could not be proved to have caused injuries. It added that pollution, increased temperature, changed chemical composition of the waters or inability to make restitution of water could be considered injuries for the purpose of prior negotiation, and stated that Spain should have argued its position in terms of the actual injuries which might be caused by the French project:

It could have been argued that the works would bring about an ultimate pollution of the waters of the Carol or that the returned waters would have a chemical composition or a temperature or some other characteristic which could injure Spanish interests. Spain could then have claimed that her rights had been impaired in violation of the Additional Act. Neither in the dossier nor in the pleadings in this case is there any trace of such an allegation.

It could also have been claimed that, by their technical character, the works envisaged by the French project could not in effect ensure the restitution of a volume of water corresponding to the natural contribution of the Lanoux to the Carol, either because of defects in measuring instruments or in mechanical devices to be used in making the restitution. The question was lightly touched upon in the Spanish Counter Memorial (p. 86), which underlined the "extraordinary complexity" of procedures for control, their "very onerous" character, and the "risk of damage or of negligence in the handling of the watergates, and of obstruction in the tunnel". But it has never been alleged that the works envisaged present any other character or would entail any other risks than other works of the same kind which today are found all over the world. It has not been clearly affirmed that the proposed works would entail an abnormal risk in neighbourly relations or in the utilization of the waters. As we have seen above, the technical guarantees for the restitution of the waters are as satisfactory as possible. If, despite the precautions that have been taken, the restitution of the waters were to suffer from an accident, such an accident would be only occasional and, according to the two Parties, would not constitute a violation of article 9.110

It is not quite clear whether injuries to "Spanish interests" are meant to apply only to material injuries. But from examples given by the tribunal, i.e. pollution, chemical composition, temperature, etc., it may be deduced that certain material changes with the potential to cause injuries, whether or not material, may be sufficient to constitute harm for purposes of prior negotiations.

<sup>105</sup> Ibid., p. 30.

<sup>106</sup> International Law Reports 1957 (see footnote 67 above), pp. 124-125.

<sup>107</sup> Ibid., p. 126 (para. 9 of the award).

<sup>108</sup> Ibid., p. 129 (para. 13 of the award).

<sup>109</sup> Ibid., p. 123.

<sup>116</sup> Ibid., pp. 123-124 (para. 6 of the award).

159. Similarly, in the Kansas v. Colorado case (1902), the United States Supreme Court noted that utilization of the great rivers by the upper riparian might lead to injury to the lower riparian as defined under international law. The Supreme Court set the principle in a suit brought by Kansas seeking to enjoin Colorado from diverting waters from a shared river. Implicitly, the Supreme Court recognized that the diversion would cause harm to Kansas. Kansas had averred that:

... the State of Colorado, acting directly herself, as well as through private persons thereto licensed, is depriving and threatening to deprive the State of Kansas and its inhabitants of all the water heretofore accustomed to flow in the Arkansas River through its channel on the surface, and through a subterranean course, across the State of Kansas; that this is threatened not only by the impounding, and the use of the water at the river's source, but as it flows after reaching the river . . . The injury is asserted to be threatened, and as being wrought, in respect of lands . . . And it is insisted that Colorado in doing this is violating the fundamental principle that one must use his own so as not to destroy the legal rights of another.

The Supreme Court found the *averments* sufficient to raise the question of injury:

Without subjecting the bill to minute criticism, we think its averments sufficient to present the question as to the power of one State of the Union to wholly deprive another of the benefit of water from a river rising in the former\* and, by nature, flowing into and through the latter, and that, therefore, this court, speaking broadly, has jurisdiction. 112

160. The German Constitutional Court, rendering a provisional decision concerning the flow of the waters of the Danube in the *Donauversinkung* case (1927), 113 stated that "only considerable interference" with the natural flow of international rivers can form the basis for claims under international law". The Court stated further that Württemberg was required "to refrain from such interference" with the natural distribution of water as damages the interests of Baden to any considerable extent".

161. In the Georgia v. Tennessee Copper Co. case (1907), in which the State of Georgia had instituted an action against a private company, seeking to restrain it from discharging noxious gas, the United States Supreme Court stated:

The Supreme Court considered that the defendant had so far shown "due diligence" in making efforts to prevent the discharge of noxious gas, but that preventive measures had proved to be ineffective and that the defendant must take different measures. Citing damage to "forests and vegetable life, if not to health, within the plaintiff State" the Court held that, if after allowing a reasonable time to the defendants to complete the structures, the fumes were not controlled, an injunction would be issued.

162. The pollution of waters through discharge of sewage was also recognized by the Supreme Court in the *Missouri* v. *Illinois* case (1906) as an injury inflicted unilaterally. In that case, Missouri had alleged that:

The Supreme Court found the complaint to pose a question of the "first magnitude", namely, "whether the destiny of the great rivers is to be the sewers of the cities along their banks or to be protected against everything which threatens their purity".\* However, the Court also noted, as would the arbitrators in the Lake Lanoux case, that the activity actually benefited rather than injured the target State:

Thus the pollution of shared resources without detriment to the quality of human life or economic interests does not appear to give rise to the duty to negotiate to seek ways to prevent or to minimize injuries.

163. Potential accidental injury has been the subject of prior negotiation between States in the past. For example, in connection with the construction by Mexico of a highway crossing the *Smugglers* and *Goat Canyons*, close to the United States border, the United States of America had observed that the road constructed on Mexican territory would not withstand torrential rains, and had requested that negotiations commence between the two States to *determine the impact of the construction* and to *develop remedial plans*. Writing to the Mexican Commissioner on the International Boundary and Water Commission on 20 May 1957, the United States Commissioner stated:

... it was observed that the highway construction in Mexico extending west from the city of Tijuana and parallel to the boundary, crosses two canyons draining northward into the United States ... and that the crossing over the first canyon, referred to as "Smugglers Canyon", is being made by an earth fill already up to 60 feet in height without culverts, and that it is understood that the plans for crossing over the second canyon, referred to as "Goat Canyon", provide for similar construction.

This construction which appears in effect to comprise earth dams across the two canyons without outlet works or spillways, and apparently without impervious cores and therefore subject to failure, could result in flows at the mouths of the canyons at rates greatly exceeding those of natural flows. At the mouths of the canyons in the United States there are residences and properties which would be seriously damaged by such flows.

<sup>111</sup> United States Reports, vol. 185, pp. 145-146.

<sup>112</sup> Ibid., p. 145.

<sup>113</sup> See footnote 13 above.

<sup>114</sup> United States Reports, vol. 206, p. 238.

<sup>115</sup> Ibid., vol. 200, p. 517.

<sup>116</sup> Ibid., p. 522.

In view of the aforedescribed situation, I will appreciate an examination of the problem by your Section, and, if the conditions found are as reported to me, that appropriate arrangements be made with the proper authorities in Mexico to take such remedial measures as required to eliminate this threat to interests in my country.<sup>117</sup>

In a note addressed in July 1959 to the Mexican Minister of Foreign Relations, the United States Ambassador to Mexico stated that, in the opinion of the United States Government engineers who were very familiar with the construction, the embankment at Goat Canyon would fail in certain circumstances of flood and that the subsequent modifications by the Mexicans to remedy that problem were not sufficient to ensure its security. The United States therefore urged Mexico to take appropriate steps to prevent the damage to property and the injury to persons that were likely to result from the improper construction of the highway.<sup>118</sup>

164. Similarly, a Mexican-United States Commission for the *Eradication of Foot-and-Mouth Disease* had been established in 1947 at the initiative of the United States in response to the perceived threat of potential injury arising to United States livestock and agriculture from a possible introduction of that disease from Mexico. The Commission's task was to carry out "operations or measures to eradicate, suppress, or control, or to prevent or retard, foot-and-mouth disease, rinderpest, or screw-worm in Mexico" where, it was deemed, such action was necessary to protect the livestock and related industries of the United States.<sup>119</sup> The Commission comprised an equal number of representatives of each country; Mexico and the United States jointly provided the funds for the operation.

165. Lost future interest has also been recognized as an injury which may give rise to impact assessment. Judge Jessup, in his separate opinion in the North Sea Continental Shelf cases, emphasized that harm possibly resulting from delimitation of the continental shelf and mitigated by the majority decision extended beyond direct, immediate and physical harm to indirect, possible and future harm. Judge Jessup referred not only to wasteful or harmful methods of extraction as a basis for negotiation, but also to lost opportunities to exploit resources which might be found in the future, to money wasted in exploratory investigations in areas destined to fall under another State's territorial control and to revenues lost by lack of authority to issue concessionary licences. Those revenues included:

... national revenue to be derived from fees, taxes, royalties or profit-sharing, with increases in national productivity, and also with the impact on the national balance of payments if imports of fuels to meet domestic needs are eliminated or reduced by the production of natural gas in the State's portion of the continental shelf.<sup>120</sup>

166. In the *Nuclear Tests* case, Australia claimed that account be taken of non-material injury. The claims formulated by the Government of Australia were:

(i) The right of Austalia and its people, in common with other States and their peoples, to be free from atmospheric nuclear weapon tests by any country is and will be violated;

- (ii) The deposit of radioactive fallout on the territory of Australia's airspace without Australia's consent:
  - (a) violates Australian sovereignty over its territory;
  - (b) impairs Australia's independent right to determine what acts shall take place within its territory and in particular whether Australia and its people shall be exposed to radiation from artificial sources;
- (iii) The interference with ships and aircraft on the high seas and in the superjacent airspace, and the pollution of the high seas by radioactive fallout, constitute infringements of the freedom of the high seas:<sup>121</sup>

Additionally, the Government of Australia alleged that the French nuclear explosions had caused radioactive fallout on Australian territory and elsewhere in the southern hemisphere and had given rise to considerable concentrations of radioactive substances in foodstuffs and in man. Thus the radioactive substances deposited on Australia were potentially dangerous to that country and its peoples and any injury caused thereby would be irreparable. Australia further claimed that the tests created anxiety and concern among the Australian people and that any effects of the French tests upon the resources of the sea or the conditions of the environment could never be undone and would be irremediable by any payment of damages. 122 In its interim measures order in the Nuclear Tests case, the Court declined to exclude the possibility that damage to Australia might be shown to be caused by the deposit on Australian territory of radioactive fallout resulting from such tests. The issue in that case, however, was whether France was allowed to conduct nuclear testing in the atmosphere. Hence the question of prior consent was not even raised.

167. Harm is referred to in State practice in a variety of forms, such as material injury, frustration or deprivation of some legally protected interests, nuisance or the deposit of certain substances, and in most cases by a general reference to the term "damage" itself. In bilateral agreements and a few judicial decisions, references to harm are more precise than in multilateral agreements. This may be due to the more specific subject matter of bilateral agreements and of disputes leading to judicial decisions. Nevertheless, general reference to the term "harm" in State practice is significant. This trend may be explained by difficulties in, first, determining a fixed content for harm which would be relevant to all circumstances and, secondly, agreeing on a clear and fixed threshold separating tolerable harm from harm that may be tolerated only with the prior consent of the injured party. Pinpointing such a threshold is extremely difficult and appears to be a function of policy decision for particular activities, etc. Consequently, the threshold which separates tolerable injury from that which may be tolerated only with prior consent seems to be fairly flexible. Precedent demonstrates that there are certain criteria, more or less common to various forms of State practice, which could assist in fixing the threshold between the tolerable injury and that which requires consent. First, the harm must be substantial. Although the term "substantial" itself is ambiguous, it suggests a dividing line which may be determined by examining local or regional expectations.

Whiteman, op. cit. (footnote 45 above), vol. 6, p. 260.

<sup>118</sup> Ibid., p. 261.

<sup>119</sup> Ibid., p. 266.

<sup>120</sup> I.C.J. Reports 1969, p. 79.

<sup>&</sup>lt;sup>121</sup> I.C.J. Reports 1973, Nuclear Tests (Australia v. France), Interim measures, Order of 22 June 1973, p. 103, para. 22.

<sup>122</sup> I.C.J. Pleadings, Nuclear tests, vol. 1, pp. 8-14.

Secondly, the harm must affect human beings, that is, there must be direct personal injury, economic loss, damage to property, etc., or entail indirect material and property losses, such as injury to the economic resources of the State through substantial injury to its natural resources (fisheries, coastal waters, drinking or irrigation water, etc.). Thirdly, the injury may be to legally protected interests. Occasionally injury may be to interests not necessarily having explicit legal protection.

#### 2. COMPETENCE TO DECIDE WHAT CONSTITUTES HARM

168. Although, in treaties, the States parties have agreed on the definition of harm, it would not be entirely correct to assume that competence to decide what constitutes harm for the purposes of consultation and negotiation lies with the States parties to a dispute. It seems that primary competence to decide what constitutes harm and to demand negotiation lies with the injured or potentially injured State. Competence to decide whether a particular activity requires the *consent* of the injured State, however, appears to be a competence shared between the acting and the injured State, or devolved on a third party, such as a group of designated consultants, joint commissions, or even an arbitral tribunal.

169. Competence to decide what constitutes harm includes the initial decision on the definition, extent and measure of harm, or the application of those decisions to a particular factual situation. In most treaties such competence appears to be both prescriptive and applicative.

### (a) Multilateral agreements

Some multilateral agreements have already defined harm and sometimes enumerated harmful activities. Decisions regarding these matters, therefore, have already been taken by the parties to the agreements. Occasionally, some multilateral agreements, in addition to defining harm and harmful activities, have provided for review or final decision regarding the permissibility of an activity to be taken by the appropriate authorities. In some agreements there is explicit or implicit language as to who decides what constitutes harm. Under the 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden, for example, each contracting State appoints an organism, known as the "supervisory authority", to decide whether a particular activity which is to take place is environmentally harmful. Once such a decision is made, the supervisory authority can institute proceedings in the courts or with the administrative authority for a decision on the permissibility of the activity. If the court or the administrative authority finds the activity environmentally harmful, the supervisory authority must inform the supervisory authority of the other State of its opinion. The relevant articles of the Convention read:

### Article 4

Each State shall appoint a special authority (supervisory authority) to be entrusted with the task of safeguarding general environmental interests in so far as regards nuisances arising out of environmentally harmful activities in another Contracting State.

For the purpose of sateguarding such interests, the supervisory authority shall have the right to institute proceedings before or be heard by the competent court or administrative authority of another Contracting State regarding the permissibility of the environmentally harmful activities if an authority or other representative of general environmental interests in that State can institute proceedings or be heard in matters of this kind, as well as the right to appeal against the decision of the court or the administrative authority in accordance with the procedures and rules of appeal applicable to such cases in the State concerned.

#### Article 5

If the court or the administrative authority examining the permissibility of environmentally harmful activities (examining authority) finds that the activities entail or may entail nuisance of significance in another Contracting State, the examining authority shall, if proclamation or publication is required in cases of that nature, send as soon as possible a copy of the documents of the case to the supervisory authority of the other State, and afford it the opportunity of giving its opinion. Notification of the date and place of a meeting or inspection shall, where appropriate, be given well in advance to the supervisory authority which, moreover, shall be kept informed of any developments that may be of interest to it.

# With regard to article 5, the Protocol to the Convention provides:

Article 5 shall be regarded as applying also to applications for permits where such applications are referred to certain authorities and organizations for their opinion but not in conjunction with proclamation or publication procedures.

171. Articles 6 and 7 of the Convention further provide:

#### Article 6

Upon the request of the supervisory authority, the examining authority shall, in so far as compatible with the procedural rules of the State in which the activities are being carried out, require the applicant for a permit to carry out environmentally harmful activities to submit such additional particulars, drawings and technical specifications as the examining authority deems necessary for evaluating the effects in the other State.

# Article 7

The supervisory authority, if it finds it necessary on account of public or private interests, shall publish communications from the examining authority in the local newspaper or in some other suitable manner. The supervisory authority shall also institute such investigations of the effects in its own State as it deems necessary.

# Article 11 also provides:

# Article 11

Where the permissibility of environmentally harmful activities which entail or may entail considerable nuisance in another Contracting State is being examined by the Government or by the appropriate Minister or Ministry of the State in which the activities are being carried out, consultations shall take place between the States concerned if the Government of the former State so requests.

Accordingly, the States concerned may resolve differences among themselves concerning the permissibility of environmentally harmful activities which entail or may entail considerable nuisance in other contracting States.

172. An opinion by a commission on the effects of an activity may also be requested. In this connection, article 12 provides:

### Article 12

In cases such as those referred to in article 11, the Government of each State concerned may demand that an opinion be given by a commission which, unless otherwise agreed, shall consist of a chairman from another Contracting State to be appointed jointly by the Parties and three members from each of the States concerned. Where such a

commission has been appointed, the case cannot be decided upon until the commission has given its opinion.

Each State shall remunerate the members it has appointed. Fees or other remuneration of the chairman as well as any other costs incidental to the activities of the commission which are not manifestly the responsibility of one or the other State shall be equally shared by the States concerned.

Where a commission has been appointed, the case according to article 12, cannot be decided upon until the commission has given its opinion.

173. The 1950 Protocol between Belgium, France and Luxembourg for the Establishment of a Tripartite Standing Committee on Polluted Waters provides for a joint technical sub-committee with the following terms of reference:

(a) to define the pollution factors (industrial or communal origin, degree of intensity, etc.), collect any appropriate technical opinions, assess each State's share of responsibility for the pollution;

- 174. Article 9 of the 1974 Convention for the Prevention of Marine Pollution from Land-based Sources provides that decisions shall be made by a joint commission. Under this article, agreement must be reached among the States concerned in regard to certain polluting substances that are likely to prejudice the interests of other parties to the Convention. The Commission referred to in article 15 of the Convention may also make recommendations for the appropriate resolution of the problem, at the request of any contracting party. The final decision appears to be made by the parties involved (see paras. 2, 3 and 4 of art. 9, cited in para. 121 above).
- 175. In some conventions, the decision as to what constitutes harm may be made either by the acting State or by the injured State. However, such unilateral decision may subsequently be subject to review. For example, the 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties appears to leave the decision as to what constitutes "major harmful consequences" to the coastal State (art. I, para. 1). Such decision by the coastal State may eventually be subject to review, should a dispute arise between that State and the flag State. Under article VIII of the Convention, any dispute between the parties, if not settled by negotiation, must be settled by conciliation or abitration. The party (the coastal State) which took the measures may not refuse a request for conciliation or arbitration:

#### Article VIII

- 1. Any controversy between the Parties as to whether measures taken under article I were in contravention of the provisions of the present Convention, to whether compensation is obliged to be paid under article VI, and to the amount of such compensation shall, if settlement by negotiation between the Parties involved or between the Party which took the measures and the physical or corporate claimants has not been possible, and if the Parties do not otherwise agree, be submitted upon request of any of the Parties concerned to conciliation or, if conciliation does not succeed, to arbitration, as set out in the annex to the present Convention.
- 2. The Party which took the measures shall not be entitled to refuse a request for conciliation or arbitration under provisions of the preceding paragraph solely on the grounds that any remedies under municipal law in its own courts have not been exhausted.

# (b) Bilateral agreements

176. In some bilateral agreements, it is for the contracting States to decide what constitutes harm for purposes

of prior negotiation and consultation. The 1931 Convention between Romania and Yugoslavia<sup>123</sup> provides that, if either party intends to take certain measures on the frontier watercourse in its territory which might injuriously affect any interests in the territory of the other State, it must obtain the agreement of that State. Article 19 of chapter III of the Convention provides:

#### Article 19

If either State desires to carry out on a watercourse within its territory new works which might injuriously affect any interests in the territory of the other State, such works may be carried out only by agreement between the two States.

The Convention does not define the interests "injuriously" affected; hence it cannot be assumed that the parties had agreed on the definition of injury. In bilateral agreements concerning all activities with extraterritorial environmental effect, such as the 1983 Agreement between the United States of America and Mexico, 124 there seems to be no indication as to who has the competence to decide what constitutes harm. Since the parties have agreed to co-ordinate their co-operation in activities affecting the environment of border areas, it may be assumed that, in the same spirit of co-operation, decisions as to what constitutes harm will be the result of agreement between the parties. This assumption, however, is not supported by specific provisions of the Agreement.

177. Article 28 of the 1960 Treaty between Belgium and the Netherlands<sup>125</sup> provides that it is for the competent *authorities of the two Governments* to determine the permissible concentration of chemical substances in the waters of the canal in the vicinity of the Belgian-Netherlands frontier:

### Article 28

Pursuant to the provisions of paragraph (I) (d), of annex III to this Treaty, the Belgian and Netherlands Ministers aforesaid shall determine the permissible concentration of chemical substances. The Ministers may by agreement modify the standards of quality set forth in the said annex.

178. Similarly, article 4 of chapter I of the 1964 Agreement between Finland and the Soviet Union<sup>126</sup> provides that the two parties should, to the extent required, *jointly* decide upon the standard of quality of water in each frontier watercourse. The article reads in part:

#### Article 4

... The Contracting Parties shall, to the extent required, jointly decide upon the standards of quality to be set for the water in each frontier watercourse or part thereof and shall, in accordance with the procedure laid down in chapter 11, co-operate in keeping the quality of the water in frontier watercourses under observation and in taking measures to increase the self-cleansing capacity of the said watercourses.

179. Article 3 of the 1958 Treaty between Czechoslovakia and Poland<sup>127</sup> provides that the two

<sup>123</sup> See footnote 60 above.

<sup>124</sup> See footnote 16 above.

<sup>125</sup> Treaty of 20 June 1960 between Belgium and the Netherlands concerning the improvement of the Terneuzen and Ghent Canal, and the settlement of various related matters.

<sup>&</sup>lt;sup>126</sup> Agreement of 24 April 1964 between Finland and the USSR concerning frontier watercourses.

<sup>127</sup> See footnote 86 above.

parties shall *jointly agree* on the amount of water to be taken from frontier waters for domestic, industrial and other uses (see para. 141 above).

- 180. Under some bilateral agreements, a joint commission determines what constitutes tolerable or intolerable harm. Article 2, paragraph 2, of the 1955 Agreement between Yugoslavia and Romania, <sup>128</sup> for example, provides that the erection of any new installations and the execution of any new works in the territory of either contracting party which may change the existing régime of the waters, interfere with their free discharge, change their quality or cause flooding on water control systems, shall be referred to the *Mixed Commission* for examination (see para. 142 above).
- 181. The 1929 Convention between Norway and Sweden<sup>129</sup> establishes a more elaborate system for examining the question of harm. Under article 16, each contracting party may ask the other for the information needed to determine what effects a particular measure, envisaged by the acting State, may have on the other State (see para. 77 above). As specified in article 17, each State may require that the question be examined by a commission consisting of two, four or six members, half of whom to be nominated by each State:

#### Examination of Questions by a Commission130

#### Article 17

Each State may require that, in order to examine the question, a Commission should be appointed consisting of two, four or six members, half of whom shall be nominated by each State.

Under article 18, the Commission may also ask for expert assistance. The Commission is also empowered to examine applications by non-governmental entities. Parties, including individuals, whose rights are affected by an undertaking must have the opportunity to defend their interests before the Commission:

#### Article 18

- 1. The Commission shall examine the questions which concern both countries and may for that purpose call in expert assistance. It shall establish its own rules of procedure.
- 2. Parties whose rights are affected by the undertaking shall receive at reasonable notice an opportunity of defending their interests before the Commission.
- 3. Each State shall fix and pay the remuneration of the members of the commission which it has appointed. The other costs of the Commission shall be paid by the applicant, but shall be advanced by the State which has called for the appointment of the Commission. The applicant may be required to pay an appropriate sum on account or to give security for such costs.

Under article 19, the Commission must give its opinion as to whether the measure should be carried out and on how the work is to be executed with minimum damage and inconvenience, as well as on how to prevent or minimize the damage to or detrimental effect upon public interests. It must also decide on the security that must be given for fulfilling the stipulated conditions governing the work and for any other obligations which may result therefrom:

#### Article 19

- 1. The Commission shall give its opinion as to whether the undertaking should be carried out, and in that case shall decide in so far as circumstances require:
- (a) How the work is to be executed so that the object may be attained without excessive cost and with the minimum damage and inconvenience, and also what measures may be considered necessary to prevent or decrease the damage to or detrimental effect upon public interests;
- (b) What rules should be laid down regarding the conservancy and outflow of the water:
- (c) The amount of the charges to be paid and the funds to be deposited in accordance with the provisions of article 8;
- (d) Whether the arrangements provided for in article 10 regarding participation in the work should be approved;
- (e) What security is to be given for fulfilling the stipulated conditions governing the work and for any other obligations which may result therefrom;
  - (f) Within what period the work is to be begun and completed;
  - (g) For what period the authorization is to be valid;
- (h) Any other questions concerning the two countries in connection with the work.
- 2. When the Commission's inquiry has been concluded, its opinion shall be communicated to both States. Each State may ask the Commission for further information, which shall also be communicated to both States.
- 182. Article 20 of this Convention also provides that the competent authority to decide whether the consent of the other State is required for an undertaking, and the conditions under which such consent may be given, shall be the King. If such consent is required and if it has been made subject to special conditions, the competent authority to decide whether the measures are permissible is also the King of the country where the work is to be carried out:

### COMPETENT AUTHORITY TO GIVE DECISIONS

#### Article 20

The question whether the consent of the other State is required for an undertaking and if so whether such consent should be given and on what conditions, shall be decided by the King. If such consent is required and if it has been subjected to special conditions, the question whether the undertaking is permissible and on what conditions shall also be decided by the King in the country in which the work is to be carried out.

Under article 21, the authorization for an undertaking is not valid in the other country (the potentially injured State) unless the applicant has obtained a certificate from that country:

#### CONTENTS OF THE AUTHORIZATION

#### Article 21

- 1. Authorization for an undertaking shall be granted by the competent authority in the country in which it is to be carried out. The authorization shall contain not only the conditions stipulated by that State but also any conditions which may have been submitted for the other State's approval in accordance with article 13. The authorization shall further stipulate that it is not valid in the other country unless the applicant has obtained the certificate mentioned in article 22 from the competent authority of that country.\*
- 2. When the final decision has been reached by the State in which the undertaking is to be carried out, a copy thereof shall be transmitted to the other State at the same time as the decision is sent to the applicant.

Article 22 provides that, when authorization for an activity has been granted by the acting State, the applicant must within 180 days obtain from the other State (the potentially injured State) a certificate that authorization has been granted in the manner provided in the agree-

<sup>128</sup> See footnote 89 above.

<sup>129</sup> See footnote 36 above.

<sup>130</sup> Articles 18 and 19 below come under the same heading.

ment, otherwise the undertaking may not be carried out without new authorization:

LEGAL EFFECT OF THE AUTHORIZATION IN THE OTHER COUNTRY

#### Article 22

- 1. When authorization for an undertaking has been granted and has acquired legal effect, the applicant must within 180 days obtain from the competent authority in the other country a certificate that authorization has been granted in the manner provided for in this Convention. If the certificate is not applied for within the abovementioned period, the undertaking may not be carried out without fresh authorization.
- 2. If a waterfall, immovable property or transport or floating interest on account of which authorization for an undertaking has been granted belongs to the other country, the certificate may not be issued unless a decision has been taken regarding the regulations to be established under article 3, paragraph 2.
- 3. When such a certificate has been issued, any inhabitant of the country shall be obliged, always subject to compliance with the laws of the country and provided he receives compensation therefor, to give up the such immovable property as may be required and to submit to any servitude upon it and tolerate any damage or nuisance caused by the undertaking.
- 183. The Frontier Water Commission established by the 1922 Treaty between Germany and Denmark<sup>131</sup> is competent to decide on authorizations for certain activities on the joint waters between Germany and Denmark. Article 30 of the Treaty reads:

#### Article 30. Procedure to be followed in making applications

The necessary drawings and explanations shall be attached to all applications for the erection of new works or the alteration of existing works in accordance with article 29. Applications shall be laid before the head district official or head county official concerned whose duty it is to submit them to the Frontier Water Commission, subject, if necessary, to the provision of suitable security for the costs.

If the Frontier Water Commission is definitely of opinion that a proposal should not be adopted, it may at once reject such proposal by means of a decision in which the reasons for the rejection are given.

In other cases, the proposed use of the watercourse shall be brought to the notice of the public in the manner which is customary in the locality in all Communes or manorial districts (Gutsbezirke), the land of which might be affected by the operation of the works in the event of their being authorized.

Further, the attention of all persons who will clearly suffer damage from the authorization of the works shall be drawn to the public notification by means of registered letters.

184. For the disposal of radioactive materials within the potentially injured State from an operation carried out in that territory by the acting State, the prior permission of the potentially injured State is required. The 1964 Agreement between the United States of America and Italy regarding the use of Italian ports by the United States nuclear ship Savannah<sup>132</sup> requires the prior approval of the Italian authorities for the disposal of radioactive materials from the ship in Italian territory. The relevant provisions of article V of the Agreement read:

#### Article V

- (b) Disposal of radioactive liquid or solid substances within Italian territorial waters and ports shall take place from the ship only with the specific prior approval of competent Italian authorities.
- (c) Release of any radioactive gaseous substances from the ship while within Italian territorial waters and ports shall be at or below permissible levels as specified by competent Italian authorities. Disposal or release of any radioactive gaseous substances within

Italian territorial waters and ports which exceed such permissible levels shall be subject to prior approval of competent Italian authorities.

Article 20 of the 1963 Operational Agreement between the Netherlands and the United States of America<sup>133</sup> on the same subject contains a similar provision:

## RADIOACTIVE WASTE

#### Article 20

The Government of the United States shall ensure that gaseous, liquid or solid radioactive waste shall remain on board the Ship in accordance with the *Operating Manual* while the Ship is in Netherlands waters or in the port area of Rotterdam unless the express prior approval of the authorities assigned therefor by the Netherlands Government has been obtained for the disposal of said waste.

# (c) Judicial decisions and State practice other than agreements

185. Judicial decisions and official correspondence recognize that States may unilaterally determine or assess the harm likely to result from activities undertaken by them or within their territories when information about the activity is uniquely within their own knowledge. For example, in the Corfu Channel case, the obligation imposed on Albania to notify shipping authorities about the existence of minefields in its territorial waters (an international strait) may be interpreted as implicitly recognizing Albania's initial duty to decide unilaterally what constitutes harm requiring notification. In its judgment of 9 April 1949, the International Court of Justice noted that Albania had exclusive control of the area.134 Of course, here competence to define injury for purposes of notification should be distinguished from competence to apply that definition to a particular factual situation. It appears that the Court considered Albania as having competence to recognize that the laying of mines in waters under its jurisdiction was bound to cause material injuries to the British ships passing through and that Albania should have taken action, such as informing the British of the existence of the mines. It follows from the judgment that even competence to apply a particular definition of injury to a particular situation is not discretionary. On the contrary, it is obligatory and, in the absence of a proper performance of this competence, the State is liable for the injuries it may have caused.

186. Likewise, the United States Supreme Court, in the *United States* v. *Arjona* case (1887), 135 stated that, with respect to counterfeiting, it was the duty of the State in whose territory an injurious activity was taking place to decide what constituted harm and to take appropriate steps to prevent injury. Here also competence to decide implies a duty to *apply measures to assess injury* to a particular activity:

The law of nations requires every national government to use "due diligence" to prevent a wrong being done within its own dominion to another nation with which it is at peace, or to the people thereof; and because of this the obligation of one nation to punish those who within its own jurisdiction counterfeit the money of another nation has long been recognized. 136

<sup>131</sup> See footnote 59 above.

<sup>132</sup> See footnote 20 above.

<sup>133</sup> See footnote 22 above.

<sup>134</sup> I.C.J. Reports 1949, p. 8.

<sup>135</sup> United States Reports, vol. 120, p. 479.

<sup>136</sup> Ibid., p. 484.

- 187. The Supreme Court further stated that the United States had the *power* and that it was its *duty* to prevent and punish the counterfeiting, within its jurisdiction, of the money of another nation:
- ... it was incumbent on the United States as a nation to use *due diligence to prevent\** any *injury\** to another *nation or its people\** by counterfeiting its money, or its public or *quasi* public securities.<sup>137</sup>
- 188. The decisions requiring States to assess harm unilaterally and to take preventive measures are founded on the notions of the necessity of inter-State relations, of comity and of reciprocity. The Supreme Court gave concrete expression to those notions in the Arjona case, concerning the power and the duty of the United States Government to prevent and punish the counterfeiting within its jurisdiction of the notes, bonds and other securities issued by foreign Governments or under their authority. The Court stated:
- ... Any uncertainty about the genuineness of the security necessarily depreciates its value as a merchantable commodity, and against this international comity requires that national protection shall, as far as possible, be afforded. If there is neglect\* in that, the United States may, with propriety, call on the proper Government to provide for the punishment of such an offence, and thus secure the restraining influences of a fear of the consequences of wrong doing. A refusal may not, perhaps, furnish sufficient cause for war, but would certainly give just ground of complaint, and thus disturb that harmony between the Governments which each is bound to cultivate and promote.

Within this context the Supreme Court also recognized the competence of the injured State to inform the acting State that a particular activity taking place within its jurisdiction and under its control had caused or might cause injury to it. After elaborating on the importance of the genuineness of United States Security Notes to the country's economy, the Court stated that, when there was a counterfeiting of United States Security Notes abroad, the United States Government had the right to call on the proper Government for protection.

189. It also happens that the acting State unilaterally determines what are tolerable or intolerable injuries where the activity or its serious and harmful consequences occur in the shared domain. The United States of America and the United Kingdom, at the time of the Eniwetok Atoll and Christmas Island nuclear tests respectively, unilaterally assessed the possible injuries that the tests might cause to other States and their nationals. Before undertaking nuclear testing in the Pacific Ocean in 1958, the United States Government studied the area which could be affected by nuclear tests, established a "danger area", and then informed the Japanese Government as well as other States and vessels that were planning to pass through that area. Similarly, the British Government, after examining the extent of the area which could be affected by nuclear tests, established a "danger area" on the high seas around Christmas Island for its first hydrogen bomb tests, on 7 January 1957. It may be assumed that the United States and British Governments had made a

- 190. In at least one incident, the acting State assured the injured State that it would comply with the domestic laws of the latter as to standards of tolerable injury. When there was a *serious* possibility of grave pollution originating from a plant to be built in Lorraine, in France, near the border of the Federal Republic of Germany, the local authorities of Lorraine gave assurances to their German counterparts that the plant would comply with German emission standards, and ordered the plant to do so. 140 In this case, competence to define harm appears to have lain with the injured State, while the acting State had the responsibility of applying the prescribed standards.
- 191. The question of observance by the acting State of the standards of the injured State in regard to pollution was touched upon by the United States Supreme Court in an interstate water pollution case. In the *Illinois v. Milwaukee* case (1972), <sup>141</sup> the Supreme Court, stressing the importance of reaching an equitable solution, stated that the high standards of prevention of pollution of the neighbouring State should be taken into account, in addition to the requirements of the federal law:
- ... While federal law governs, consideration of State standards may be relevant ... Thus, a State with high water-quality standards may well ask that its strict standards be honoured and that it not be compelled to lower itself to the more degrading standards of a neighbour....<sup>142</sup>
- 192. In the Lake Lanoux case, the tribunal held that States were required to enter into negotiations with the other States concerned before undertaking the industrial utilization of international rivers. Both States had interests that must be taken into consideration:

France is entitled to exercise her rights; she cannot ignore Spanish interests.

Spain is entitled to demand that her rights be respected and that her interests be taken into consideration.<sup>143</sup>

In addition, the tribunal stated that, while the upstream State had the *right* to give preference to its own scheme, it had the *duty* also to examine schemes proposed by the downstream State:

As a matter of form, the upstream State has, procedurally, a right of initiative; it is not obliged to associate the downstream State in the elaboration of its schemes. If, in the course of discussions, the downstream State submits schemes to it, the upstream State must ex-

unilateral decision regarding the intolerable injuries which might be caused by their activities within the "danger area". Despite claims by the Government of Japan that the tests would also have a devastating impact on Japanese interests outside the "danger area", the acting States assessed the impact on Japanese interests as inferior to the interest of the "free world" in security from nuclear war.<sup>139</sup>

<sup>139</sup> Whiteman, op. cit. (footnote 45 above), vol. 4, pp. 586, 599 and 600.

<sup>&</sup>lt;sup>140</sup> International Environment Reporter, Washington, D.C., vol. 3, No. 9, 10 September 1980, cited by M. Bothe, "International legal problems of industrial siting in border areas and national environment policies", OECD, Transfrontier Pollution and the Role of States, 1981, p. 88, footnote 42.

<sup>141</sup> United States Reports, vol. 406, p. 91.

<sup>142</sup> Ibid., p. 107.

<sup>&</sup>lt;sup>143</sup> International Law Reports 1957 (see footnote 67 above), p. 140, para. 23 of the award.

<sup>&</sup>lt;sup>137</sup> *Ibid.*, p. 488.

<sup>138</sup> Ibid., pp. 486-487.

amine them, but it has the right to give preference to the solution contained in its own scheme provided that it takes into consideration in a reasonable manner the interests of the downstream State.

24. In the case of Lake Lanoux, France has maintained to the end the solution which consists in diverting the waters of the Carol to the Ariège with full restitution. By making this choice France is only making use of a right; the development works of Lake Lanoux are on French territory, the financing of and responsibility for the enterprise fall upon France, and France alone is the judge of works of public utility which are to be executed on her own territory, save for the provisions of articles 9 and 10 of the Additional Act, which, however, the French scheme does not infringe.

On her side, Spain cannot invoke a right to insist on a development of Lake Lanoux based on the needs of Spanish agriculture. In effect, if France were to renounce all of the works envisaged on her territory, Spain could not demand that other works in conformity with her wishes should be carried out. Therefore, she can only urge her interests in order to obtain, within the framework of the scheme decided upon by France, terms which reasonably safeguard them.'44

However, should no agreement occur, the States had the option of seeking the decision of a third party:

- . . . . It is for each State to evaluate in a reasonable manner and in good faith the situations and the rules which will involve it in controversies; its evaluation may be in contradiction with that of another State; in that case, should a dispute arise the Parties normally seek to resolve it by negotiation or, alternatively, by submitting to the authority of a third party; but one of them is never obliged to suspend the exercise of its jurisdiction because of the dispute except when it assumes an obligation to do so; by exercising its jurisdiction it takes the risk of seeing its international responsibility called into question, if it is established that it did not act within the limits of its rights. The commencement of arbitral proceedings in the present case illustrates perfectly these rules in the functioning of the obligations subscribed to by Spain and France in the Arbitration Treaty of July 10, 1929.145
- 193. In cases of distribution of shared resources or delimitation of territorial control over what has been traditionally regarded as the shared domain, judicial decisions recognize negotiation rather than unilateral determination as the most appropriate method. The International Court of Justice required the opening of negotiations in the matter of the distribution of fish resources in the Fisheries Jurisdiction case, in the delimitation of the continental shelf in the North Sea Continental Shelf cases, and in the delimitation of sea areas in the Anglo-Norwegian Fisheries case.
- 194. Recourse to the decision of a third party has also been recognized as appropriate in determining harm in extraterritorial injuries. In the *Trail Smelter* case, the tribunal appointed a panel of technical consultants to assess the injurious impact of smelting activities in British Columbia to the State of Washington. The consultants were only one aspect of a complex temporary régime established to conduct experiments and collect data:

To enable it to establish a permanent régime based on the more adequate and intensive study and knowledge above referred to, the Tribunal establishes the following temporary régime:\*146

(1) For the purpose of administering an experimental period, to continue to a date not later than October 1, 1940, the Tribunal will appoint two Technical Consultants, and in case of vacancy will appoint the successor. Such Technical Consultants to be appointed in the first place shall be Reginald S. Dean and Robert E. Swain, and they shall cease to act as Advisers to the Tribunal under the Convention during such trial period.

- (2) The Tribunal directs that, before May 1, 1938, a consulting meteorologist, adequately trained in the installation and operation of the necessary type of equipment, be employed by the Trail Smelter, the appointment to be subject to the approval of the Technical Consultants. The Tribunal directs that, beginning May 1, 1938, such meteorological observations as may be deemed necessary by the Technical Consultants shall be made, under their direction, by the meteorologist, the scientific staff of the Trail Smelter, or otherwise. The purpose of such observations shall be to determine, by means of captive balloons and other wise, the weather conditions and the height, velocity, temperature, and other characteristics of the gas-carrying and other air currents and of the gas emissions from the stacks.
- 195. The *tribunal determined what types* of experiments were to be conducted, *how* and *when*:
- (3) The Tribunal further directs that beginning May 1, 1938, there shall be installed and put in operation and maintained by the Trail Smelter, for the purpose of providing information which can be used in determining present and prospective wind and other atmospheric conditions, and in making a prompt application of those observations to the control of the Trail Smelter plant operation:
- (a) Such observation stations as the Technical Consultants deem necessary.
- (b) Such equipment at the stacks as the Technical Consultants may find necessary to give adequate information of gas conditions and in connection with the stacks and stack effluents.
- (c) Sulphur dioxide recorders, stationary and portable (the stationary recorders not to exceed three in number).

Within this régime, the consultants were given discretion to modify their *instructions*:

- (d) The Technical Consultants shall have the direction of and authority over the location in both the United States and the Dominion of Canada, and over the installation, maintenance and operation of all apparatus provided for in paragraph 2 and paragraph 3. They may require from the meteorologist and from the Trail Smelter regular reports as to the operation of all such apparatus.
- (e) The Technical Consultants may require regular reports from the Trail Smelter as to the methods of operation of its plant in such form and at such times as they shall direct; and the Trail Smelter shall conduct its smelting operations in conformity with the directions of the Technical Consultants and of the Tribunal, based on the result of the data obtained during the period hereinafter named; and the Technical Consultants and the Tribunal may change or modify at any time its or their instructions as to such operations.
- (f) It is the intent and purpose of the Tribunal that the administration of the observations, experiments, and operations above provided for shall be as flexible as possible, and subject to change or modification by the Technical Consultants and by the Tribunal, to the end that conditions as they at any time may exist, may be changed as circumstances require.
- (4) The Technical Consultants shall make report to the Tribunal at such dates and in such manner as it shall prescribe as to the results obtained and conclusions formed from the observations, experiments, and operations above provided for.
- (5) The observations, experiments, and operations above provided for shall continue on a trial basis through the remainder of the cropgrowing season of 1938, the crop-growing seasons of 1939 and 1940, and the winter seasons of 1938-1939 and 1939-1940 and until October 1, 1940, unless the Tribunal shall find it practicable or necessary to terminate such trial period at an earlier date.
- (6) At the end of the trial period above provided for, or at the end of such shorter trial period as the Tribunal may find to be practicable or necessary, the Tribunal in a final decision will determine upon a permanent régime and upon the indemnity and compensation, if any, to be paid under the Convention. Such final decision, under the agreements for extension, heretofore entered into by the two Governments under article XI of the Convention, shall be reported to the Governments within three months after the date of the end of the trial period.
- (7) The Tribunal shall meet at least once in the year 1939, to consider reports and to take such action as it may deem necessary.
- (8) In case of disagreement between the Technical Consultants, they shall refer the matter to the Tribunal for its decision, and all per-

<sup>144</sup> Ibid.

<sup>145</sup> Ibid., p. 132, para. 16 of the award.

<sup>&</sup>lt;sup>146</sup> United Nations, Reports of International Arbitral Awards, vol. III, pp. 1934-1936.

sons and the Trail Smelter affected hereunder shall act in conformity with such decision.

The régime was to be financed by the acting State:

(10) For the carrying out of the temporary régime herein prescribed by the Tribunal, the Dominion of Canada shall undertake to provide for the payment of the following expenses thereof: (a) the Tribunal will fix the compensation of the Technical Consultants and of such clerical or other assistants as it may find necessary to employ; (b) statements of account shall be rendered by the Technical Consultants to the Tribunal and approved by the Chairman in writing: (c) the Dominion of Canada shall deposit to the credit of the Tribunal from time to time in a financial institution to be designated by the Chairman of the Tribunal, such sums as the Tribunal may find to be necessary for the payment of the compensation, travel, and other expenses of the Technical Consultants and of the clerical or other assistants; (d) written report will be made by the Tribunal to the Dominion of Canada of all the sums received and expended by it, and any sum not expended shall be refunded by the Tribunal to the Dominion of Canada at the conclusion of the trial period.

# C. Balancing of interests

196. An important element in the process of impact assessment is reconciling the interests of the parties concerned with the common interest of the larger community. The balancing of interests appears to be an integral part of treaties and is referred to in judicial decisions and official correspondence concerning activities with potentially harmful impact. The concept of balance of interests in terms of "cost-benefit analysis" in torts law relates to the balancing of economic and financial interests and factors involved in a tortious act. In international relations, treaties and judicial decisions, the concept of balance of interests appears to have a broader meaning; it includes other values, in addition to economic factors, such as the well-being and health of populations, respect for the territorial sovereignty and integrity of other States, the safety and security of neighbouring States, etc.

197. Before reviewing treaties and other forms of State practice dealing with the concept of balance of interests, two points should be made. First, the initial step in balancing interests is to determine what interests each State or the larger community has that should be balanced; secondly, what is the value to be attached to each interest and how they are to be compared with one another. These difficulties arise to some extent in treaties, but they arise more sharply in judicial decisions.

# (a) Multilateral agreements

198. The concept of balance of interests is to some extent developed in the 1974 Convention on the Protection of the Marine Environment of the Baltic Sea Area and in the 1976 Convention for the Protection of Mediterranean Sea against Pollution. The regulations enumerated in annex V of the former Convention are a clear attempt to balance the interests involved and analyse costs and benefits to different parties under various alternatives. These regulations evaluate the substances that may be dumped at sea, the location of dumping, the conditions of dumping, the possible effects of such substances on, for example, marine life, fish stocks and other uses of the sea, etc. The latter Convention makes a similar attempt at balancing interests in annex III of its Protocol,

which enumerates the factors to be considered in establishing criteria governing the issue of permits for the dumping of substances at sea. These two Conventions are primarily concerned with shared domains; consequently the common interests of the larger community of coastal States are predominant. The Conventions attempt to accommodate such common interests with the interests of individual States. The 1963 Vienna Convention on Civil Liability for Nuclear Damage introduces in this respect a concept that may be called risk exclusion. Accordingly, when the risks involved in certain activities are minimal, they are exempt from certain rules, without of course affecting the question of liability in case of injury. Article I, paragraph 2, of the Convention reads:

#### Article I

2. An Installation State may, if the small extent of the risks involved so warrants, exclude any small quantities of nuclear material from the application of this Convention, provided that:

(a) maximum limits for the exclusion of such quantities have been established by the Board of Governors of the International Atomic Energy Agency; and

(b) any exclusion by an Installation State is within such established limits

The maximum limits shall be reviewed periodically by the Board of Governors.

199. In other conventions dealing with the interests of two or more States affected by certain activities, the balancing of interests focuses primarily on the interests of the States directly involved. Article 2 of the 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden clearly illustrates this focus:

#### Article 2

In considering the permissibility of environmentally harmful activities, the nuisance which such activities entail or may entail in another Contracting State shall be equated with a nuisance in the State where the activities are carried out.

Articles 6, 7 and 12 of the Convention further provide for methods to be used in order to balance the interests of the parties more effectively. Under article 6, the supervisory authority may request the examining authority to require, in so far as is compatible with the procedural rules of the State where the activities are being carried out, that the acting entity submit such additional information as the examining authority deems necessary for evaluating the effects in the other State. Article 7 empowers the supervisory authority, if it finds it necessary on account of public or *private interests*, to publish communications from the examining authority in the local newspaper or to publicize them in some other suitable manner. The supervisory authority is also required to institute investigations of the effects in its own State as it deems necessary. The purpose, of course, is to protect the public or private interests within the State in whose territory the activities are taking place. Nevertheless, under article 12, the Government of each State concerned may demand that an opinion be given by a commission concerning the permissibility of environmentally harmful activities which entail or may entail considerable nuisance in another State. The commission, unless otherwise agreed, shall consist of a chairman from a third contracting State to be appointed jointly by the parties and three members from each of the States concerned. The case cannot be decided upon until the commission has given its opinion (see paras. 172-173 above).

- 200. The concept of the balancing of interests is also incorporated in the 1950 Protocol between Belgium, France and Luxembourg to Establish a Tripartite Standing Committee on Polluted Waters. The Protocol provides for the establishment of a joint technical subcommittee with the function of defining the polluting factors, collecting any appropriate technical opinions and assessing each State's share of responsibility for the pollution.
- 201. The physical and technical capacity of States to prevent harm caused by their activities is also regarded as an element affecting the balancing of interests. This does not necessarily mean that these States are authorized to carry out harmful activities, but rather that technical assistance should be accorded to countries that do not have this capacity to enable them to prevent or minimize harm. Article 202 of the 1982 United Nations Convention on the Law of the Sea provides for such technical assistance:

Article 202. Scientific and technical assistance to developing States

States shall, directly or through competent international organizations:

- (a) promote programmes of scientific, educational, technical and other assistance to developing States for the protection and preservation of the marine environment and the prevention, reduction and control of marine pollution. Such assistance shall include, inter alia:
  - (i) training of their scientific and technical personnel;
  - (ii) facilitating their participation in relevant international programmes;
  - (iii) supplying them with necessary equipment and facilities;
  - (iv) enhancing their capacity to manufacture such equipment;
  - (v) advice on and developing facilities for research, monitoring, educational and other programmes;
- (b) provide appropriate assistance, especially to developing States, for the minimization of the effects of major incidents which may cause serious pollution of the marine environment;
- (c) provide appropriate assistance, especially to developing States, concerning the preparation of environmental assessments.

The same Convention, with a view to the balancing of interests, accords preferential treatment to developing countries. That treatment is expressly defined in article 203:

Article 203. Preferential treatment for developing States

Developing States shall, for the purposes of prevention, reduction and control of pollution of the marine environment or minimization of its effects, be granted preference by international organizations in:

- (a) the allocation of appropriate funds and technical assistance; and
  - (b) the utilization of their specialized services.

Such preferential treatment does not reduce the obligation to minimize or to prevent injuries; it simply means that priority should be given to developing countries in terms of allocation of funds and services by international organizations.

202. Again in connection with the balancing of interests, article 193 of the Convention affirms the principle of the sovereign right of States to exploit their

natural resources; it thus attempts to reconcile the principle of State sovereignty with that of protection of the marine environment, which is a matter of international concern:

Article 193. Sovereign right of States to exploit their natural resources

States have the sovereign right to exploit their natural resources pursuant to their environmental policies and in accordance with their duty to protect and preserve the marine environment.

This article, however, does not provide guidelines on how these competing principles and interests should be reconciled and which should prevail in case of conflict.

203. The 1982 United Nations Convention on the Law of the Sea, in defining the exclusive economic zone, grants certain rights to and imposes certain duties on coastal States. Coastal States are required, in most cases, unilaterally to take into account the rights of other States in undertaking activities within their own economic zone. The "rights" referred to relate to legally protected interests which have already been determined either by treaty or under international law. The coastal State is simply required to recognize them in a particular factual situation. Article 56 provides for the sovereign rights and jurisdiction of the coastal State over the establishment and use of artificial islands, installations and structures, marine scientific research and the protection and preservation of the marine environment. It goes on to provide that, in exercising its rights in the exclusive economic zone, the coastal State shall have due regard to the rights and duties of other States:

Article 56. Rights, jurisdiction and duties of the coastal State in the exclusive economic zone

- 1. In the exclusive economic zone, the coastal State has:
- (a) sovereign rights for the purpose of exploring and exploiting, conserving and managing the natural resources, whether living or non-living, of the waters superjacent to the sea-bed and of the sea-bed and its subsoil, and with regard to other activities for the economic exploitation and exploration of the zone, such as the production of energy from the water, currents and winds;
- (b) jurisdiction as provided for in the relevant provisions of this Convention with regard to:
  - (i) the establishment and use of artificial islands, installations and structures;
  - (ii) marine scientific research;
  - (iii) the protection and preservation of the marine environment;
- (c) other rights and duties provided for in this Convention.
- 2. In exercising its rights and performing its duties under this Convention in the exclusive economic zone, the coastal State shall have due regard to the rights and duties of other States and shall act in a manner compatible with the provisions of this Convention.
- 3. The rights set out in this article with respect to the sea-bed and subsoil shall be exercised in accordance with part VI.

In addition, article 58 of the Convention, dealing with the freedoms enjoyed by all States in the exclusive economic zone, namely, freedom of navigation and overflight, freedom of the laying of submarine cables and pipelines and other internationally lawful uses of the sea related to those freedoms, requires States to have due regard to the rights and duties of the coastal State:

Article 58. Rights and duties of other States in the exclusive economic zone

1. In the exclusive economic zone, all States, whether coastal or land-locked, enjoy, subject to the relevant provisions of this Conven-

tion, the freedoms referred to in article 87 of navigation and overflight and of the laying of submarine cables and pipelines, and other internationally lawful uses of the sea related to these freedoms, such as those associated with the operation of ships, aircraft and submarine cables and pipelines, and compatible with the other provisions of this Convention.

- 2. Articles 88 to 115 and other pertinent rules of international law apply to the exclusive economic zone in so far as they are not incompatible with this part.
- 3. In exercising their rights and performing their duties under this Convention in the exclusive economic zone, States shall have due regard to the rights and duties of the coastal State and shall comply with the laws and regulations adopted by the coastal State in accordance with the provisions of this Convention and other rules of international law in so far as they are not incompatible with this part.
- 204. Further, article 59 of the Convention provides that, in circumstances where the Convention does not attribute rights or jurisdiction to the coastal State or to other States within the exclusive economic zone, and a conflict arises, the States involved shall resolve the conflict on the basis of equity, taking into account the importance of the interests of the parties involved as well as the interest of the larger community:

# Article 59. Basis for the resolution of conflicts regarding the attribution of rights and jurisdiction in the exclusive economic zone

In cases where this Convention does not attribute rights or jurisdiction to the coastal State or to other States within the exclusive economic zone, and a conflict arises between the interests of the coastal State and any other State or States, the conflict should be resolved on the basis of equity and in the light of all the relevant circumstances, taking into account the respective importance of the interests involved to the parties as well as to the international community as a whole.

In this article reference is made to "interests" and not "rights". This broad language poses some of the difficulties stated at the beginning of the present section, namely, to determine what those interests are and how they are to be evaluated in relation to the interests of the acting State.

205. Attempts to balance and accommodate interests are also made in articles 60 and 61 of the Convention. Article 60, for example, grants competence to the coastal State to establish *inter alia* artificial islands and other structures, while stating that such installations and the safety zones around them may not be established where they could interfere with the use of recognized sea lanes essential to international navigation. The article reads:

# Article 60. Artificial islands, installations and structures in the exclusive economic zone

- 1. In the exclusive economic zone, the coastal State shall have the exclusive right to construct and to authorize and regulate the construction, operation and use of:
  - (a) artificial islands;
- (b) installations and structures for the purposes provided for in article 56 and other economic purposes;
- (c) installations and structures which may interfere with the exercise of the rights of the coastal State in the zone.
- 2. The coastal State shall have exclusive jurisdiction over such artificial islands, installations and structures, including jurisdiction with regard to customs, fiscal, health, safety and immigration laws and regulations.
- 3. Due notice must be given of the construction of such artificial islands, installations or structures, and permanent means for giving warning of their presence must be maintained. Any installations or structures which are abandoned or disused shall be removed to ensure

safety of navigation, taking into account any generally accepted international standards established in this regard by the competent international organization. Such removal shall also have due regard to fishing, the protection of the marine environment and the rights and duties of other States. Appropriate publicity shall be given to the depth, position and dimensions of any installations or structures not entirely removed.

- 4. The coastal State may, where necessary, establish reasonable safety zones around such artificial islands, installations and structures in which it may take appropriate measures to ensure the safety both of navigation and of the artificial islands, installations and structures.
- 5. The breadth of the safety zones shall be determined by the coastal State, taking into account applicable international standards. Such zones shall be designed to ensure that they are reasonably related to the nature and function of the artificial islands, installations or structures, and shall not exceed a distance of 500 metres around them, measured from each point of their outer edge, except as authorized by generally accepted international standards or as recommended by the competent international organization. Due notice shall be given of the extent of safety zones.
- 6. All ships must respect these safety zones and shall comply with generally accepted international standards regarding navigation in the vicinity of artificial islands, installations, structures and safety zones.
- 7. Artificial islands, installations and structures and the safety zones around them may not be established where interference may be caused to the use of recognized sea lanes essential to international navigation.
- 8. Artificial islands, installations and structures do not possess the status of islands. They have no territorial sea of their own, and their presence does not affect the delimitation of the territorial sea, the exclusive economic zone or the continental shelf.
- 206. Similarly, article 61 grants competence to coastal States to establish policies and programmes for the catch of the living resources in their exclusive economic zone. However, the article indicates that, in establishing such policies, the coastal State shall take into account certain factors, including the economic needs of coastal fishing countries and the special requirements of developing countries. The article reads:

#### Article 61. Conservation of the living resources

- 1. The coastal State shall determine the allowable catch of the living resources in its exclusive economic zone.
- 2. The coastal State, taking into account the best scientific evidence available to it, shall ensure through proper conservation and management measures that the maintenance of the living resources in the exclusive economic zone is not endangered by over-exploitation. As appropriate, the coastal State and competent international organizations, whether subregional, regional or global, shall cooperate to this end.
- 3. Such measures shall also be designed to maintain or restore populations of harvested species at levels which can produce the maximum sustainable yield, as qualified by relevant environmental and economic factors, including the economic needs of coastal fishing communities and the special requirements of developing States, and taking into account fishing patterns, the interdependence of stocks and any generally recommended international minimum standards, whether subregional, regional or global.
- 4. In taking such measures the coastal State shall take into consideration the effects on species associated with or dependent upon harvested species, whith a view to maintaining or restoring populations of such associated or dependent species above levels at which their reproduction may become seriously threatened.
- 5. Available scientific information, catch and fishing effort statistics, and other data relevant to the conservation of fish stocks shall be contributed and exchanged on a regular basis through competent international organizations, whether subregional, regional or global, where appropriate and with participation by all States concerned, including States whose nationals are allowed to fish in the exclusive economic zone.
- 207. With regard to the utilization of the living resources in the exclusive economic zone, article 62 pro-

vides that the coastal State must take into account the requirements of developing States in the region or subregion. It also provides that the coastal State must take certain steps to minimize the negative economic impact of its activities upon States whose nationals have habitually fished in the zone. Article 62 reads:

# Article 62. Utilization of the living resources

- 1. The coastal State shall promote the objective of optimum utilization of the living resources in the exclusive economic zone without prejudice to article 61.
- 2. The coastal State shall determine its capacity to harvest the living resources of the exclusive economic zone. Where the coastal State does not have the capacity to harvest the entire allowable catch, it shall, through agreements or other arrangements and pursuant to the terms, conditions, laws and regulations referred to in paragraph 4, give other States access to the surplus of the allowable catch, having particular regard to the provisions of articles 69 and 70, especially in relation to the developing States mentioned therein.
- 3. In giving access to other States to its exclusive economic zone under this article, the coastal State shall take into account all relevant factors, including, *inter alia*, the significance of the living resources of the area to the economy of the coastal State concerned and its other national interests, the provisions of articles 69 and 70, the requirements of developing States in the subregion or region in harvesting part of the surplus and the need to minimize economic dislocation in States whose nationals have habitually fished in the zone or which have made substantial efforts in research and identification of stocks.
- 4. Nationals of other States fishing in the exclusive economic zone shall comply with the conservation measures and with the other terms and conditions established in the laws and regulations of the coastal State. These laws and regulations shall be consistent with this Convention and may relate, *inter alia*, to the following:
- (a) licensing of fishermen, fishing vessels and equipment, including payment of fees and other forms of remuneration, which in the case of developing coastal States, may consist of adequate compensation in the field of financing, equipment and technology relating to the fishing industry;
- (b) determining the species which may be caught, and fixing quotas of catch, whether in relation to particular stocks or groups of stocks or catch per vessel over a period of time or to the catch by nationals of any State during a specified period;
- (c) regulating seasons and areas of fishing, the types, sizes and amount of gear and the types, sizes and number of fishing vessels that may be used;
- (d) fixing the age and size of fish and other species that may be caught;
- (e) specifying information required of fishing vessels, including catch and effort statistics and vessel position reports;
- (f) requiring, under the authorization and control of the coastal State, the conduct of specified fisheries research programmes and regulating the conduct of such research, including the sampling of catches, disposition of samples and reporting of associated scientific data:
- (g) the placing of observers or trainees on board such vessels by the coastal State:
- (h) the landing of all or any part of the catch by such vessels in the ports of the coastal State;
- (i) terms and conditions relating to joint ventures or other cooperative arrangements;
- (j) requirements for the training of personnel and the transfer of fisheries technology, including enhancement of the coastal State's capability of undertaking fisheries research;
  - (k) enforcement procedures
- 5. Coastal States shall give due notice of conservation and management laws and regulations.

# (b) Bilateral agreements

208. The concept of balancing interests by deciding whether and under what conditions certain activities with potentially injurious effect may be undertaken is

also incorporated in certain bilateral agreements. Here the interests of the acting entities, including private entities, as well as the common interest of the States parties to an agreement, are taken into account. For example, article 4 of chapter 1 of the 1971 Agreement between Finland and Sweden concerning frontier rivers provides that, where a number of projects are involved affecting the same waters, preference shall be given to the project which may be assumed to be of the *greatest public* and *private* benefit. Thus, conflicting interests are to be accommodated in such a way that each may be satisfied without substantial injury to the others. Article 4 reads:

# Article 4

In cases involving a number of different projects which affect the same waters or for some other reason cannot be carried out concurrently, preference shall be given to the project which may be assumed to be of the greatest public and private benefit. Conflicting interests shall, in so far as possible, be adjusted in such a way that each may be satisfied without substantial injury to the others.

209. Article 3 of chapter 3 of this Agreement further provides that, where any person would suffer damage or inconvenience as a result of hydraulic construction works, the works shall be carried out only if they can be shown to bring public or private benefit, that substantially outweighs the inconvenience. The same article provides that, where the injury from an activity is a substantial deterioration in the living conditions of the population or causes a permanent change in natural conditions which might entail substantially diminished comfort for people living in the vicinity or a significant nature conservancy loss, or where significant public interests would be otherwise prejudiced, the construction may be permitted only if it is of particular importance to public interests. This article reads:

#### Article 3

Where any person would suffer damage or inconvenience as a result of hydraulic construction works, the works shall be carried out only if they can be shown to bring public or private benefit that substantially outweighs the inconvenience.

Where the construction would result in a substantial deterioration in the living conditions of the population or cause a permanent change in natural conditions such as might entail substantially diminished comfort for people living in the vicinity or a significant nature conservancy loss or where significant public interests would be otherwise prejudiced, the construction shall be permitted only if it is of particular importance for the economy or for the locality or from some other public standpoint.

Compensation pursuant to chapter 7 shall be paid in respect of any damage or inconvenience.

210. Finally, article 5 of chapter 6 of this Agreement provides that, in deciding whether permission should be granted to undertake the activities, equal consideration shall be given to conditions in both countries. Thus, a site shall be selected for the operations so that the purpose can be achieved in such a way as to cause minimum inconvenience:

#### Article 5

Compensation pursuant to chapter 7 shall be paid in respect of any damage or inconvenience caused by the operations referred to in article 3.

In deciding whether permission should be granted for the operations, equal consideration shall be given to conditions in the two States. A site shall be selected for the operations such that their purpose can be achieved in such a manner as to cause minimum inconvenience and without unreasonable costs.

This Agreement attempts to reconcile the public with the private interest as well as with any other interests that both countries may have.

211. The preamble to the 1983 Agreement between the United States of America and Mexico<sup>147</sup> refers to the long-term social well-being and the economic interests of the contracting parties as well as of the global community:

Recognizing the importance of a healthful environment to the longterm economic and social well-being of present and future generations of each country as well as the global community;

212. The 1929 Convention between Norway and Sweden relating to the law on watercourses<sup>148</sup> incorporates the concept of balance of interests in article 5, which provides that, in deciding whether a particular activity may be carried out, its effects on both countries shall be taken into consideration. The utility of an activity shall be considered solely in relation to the maintenance of the waterfall, or to the transport or floating interest on account of which the activity is to be carried out. Therefore, as a matter of general principle, the evaluation of any undertaking should be based on its usefulness to the joint waters, while taking into account its effect on both countries. The article reads:

#### Article 5

In deciding whether an undertaking may be carried out, its effects in both countries shall be taken into consideration. As a rule, however, the utility of the undertaking shall be considered to be solely its utility for the waterfall, the immovable property, or the transport or floating interest on account of which the undertaking is to be carried out.

213. The 1909 Treaty concerning the boundary waters between the United States of America and Canada<sup>149</sup> lays down a set of preferences for use of the joint waters by each State even within its own territory. Furthermore, it states that the International Joint Commission established by the two States may, at its discretion, approve any undertaking conditional upon the construction of remedial or protective works to compensate so far as possible for the particular use or diversion proposed. In such cases, the Commission may require that suitable and adequate provision be made for the protection and indemnity against injury of any interests on either side of the boundary. Article VIII of the Treaty reads:

#### Article VIII

This International Joint Commission shall have jurisdiction over and shall pass upon all cases involving the use or obstruction or diversion of the waters with respect to which, under articles III and IV of this Treaty, the approval of this Commission is required, and in passing upon such cases the Commission shall be governed by the following rules and principles which are adopted by the High Contracting Parties for this purpose:

The High Contracting Parties shall have, each on its own side of the boundary, equal and similar rights in the use of the waters hereinbefore defined as boundary waters.

The following order of precedence shall be observed among the various uses enumerated hereinafter for these waters, and no use shall be permitted which tends materially to conflict with or restrain any other use which is given preference over it in this order of precedence:

- 1. Uses for domestic and sanitary purposes;
- 2. Uses for navigation, including the service of canals for the purposes of navigation;
  - 3. Uses for power and for irrigation purposes.

The foregoing provisions shall not apply to or disturb any existing uses of boundary waters on either side of the boundary.

The requirement for an equal division may, in the discretion of the Commission, be suspended in the cases of temporary diversions along boundary waters at points where such equal division cannot be made advantageously on account of local conditions, and where such diversion does not diminish elsewhere the amount available for use on the other side.

The Commission, in its discretion, may make its approval in any case conditional upon the construction of remedial or protective works to compensate, so far as possible, for the particular use or diversion proposed, and in such cases may require that suitable and adequate provision, approved by the Commission, be made for the protection and indemnity against injury of any interests on either side of the boundary.

In cases involving the elevation of the natural level of waters on either side of the line as a result of the construction or maintenance on the other side of remedial or protective works or dams or other obstructions in boundary waters or in waters flowing therefrom or in waters below the boundary in rivers flowing across the boundary, the Commission shall require, as a condition of its approval thereof, that suitable and adequate provision, approved by it, be made for the protection and indemnity of all interests on the other side of the line which may be injured thereby.

The majority of the Commissioners shall have power to render a decision. In case the Commission is evenly divided upon any question or matter presented to it for decision, separate reports shall be made by the Commissioners on each side to their own Government. The High Contracting Parties shall thereupon endeavour to agree upon an adjustment of the question or matter of difference, and if an agreement is reached between them, it shall be reduced to writing in the form of a protocol, and shall be communicated to the Commissioners, who shall take such further proceedings as may be necessary to carry out such agreement.

214. Under article 29 of the 1922 Treaty between Germany and Denmark concerning their frontier waters, <sup>150</sup> the Frontier Water Commission must balance the interests of the parties in the case of works on a large scale. The Commission may take certain decisions regarding the direction of the flow of the river regardless of the opposition of the parties, but in such cases compensation must be paid to injured individuals. The relevant provision of article 29 reads:

In the case of works on a large scale, the Frontier Water Commission may, however, direct that the water should be caused to flow round one or more properties adjacent to the watercourse, or that the water shall be discharged into another watercourse without regard to the opposition of the parties concerned. In such cases, compensation shall be granted to persons suffering prejudice for any loss and damage caused.

The second paragraph of article 26 is also aimed at accommodating the interests of the parties; it provides that the riparian proprietors must, subject to compensation, permit certain changes in the watercourse. Hence, when certain activities are important to the acting State and the injuries are not devastating to the injured State and can easily be compensated, the activities may be undertaken, subject to compensation. The paragraph reads:

<sup>147</sup> See footnote 16 above.

<sup>148</sup> See footnote 36 above.

<sup>149</sup> See footnote 35 above.

<sup>150</sup> See footnote 59 above.

The riparian proprietors must permit, subject to compensation, the erection at or in the watercourse of subsidiary works necessary to carry out the regularization of a river bed, the deposit of earth, stones, gravel, sand, wood, etc., on the land on the banks, the transport to and fro of such materials and the storing and transport to and fro of building materials, and must also grant regular right of access to the workmen and inspectors.

In some bilateral agreements, interests are balanced by the division of responsibilities between the parties. In two agreements concluded by the United States of America, one in 1963 with the Netherlands, 151 the other in 1964 with Italy,152 concerning the use of the ports of those countries by the nuclear ship Savannah, the responsibilities for port security and inspection of the ship are divided between the master of the ship and the port authorities. The host Governments are responsible for taking all necessary measures for fire safety, police protection, crowd control and general provision of facilities relating to the ship's entry into port. The designated authorities of the host countries must have reasonable access to the ship for purposes of inspection to determine whether the appropriate regulations have been applied. The relevant articles of the Operational Agreement with the Netherlands are the following:

#### Article 14

Local authorities shall provide for normal fire and police protection, and crowd control and shall make general preparations in the port area for the visit of the ship.

# Article 15

Control of public access to the ship shall be the responsibility of the master of the ship. Special arrangements relating to such control shall be made by the master with the concurrence of the authorities assigned therefor by the Netherlands Government.

#### INSPECTION

#### Article 16

While the ship is in Netherlands waters or in the port area of Rotter-dam the authorities assigned therefor by the Netherlands Government shall have reasonable access to the ship to enable them to carry out the inspections as described in recommendation 11 of annex C to the Final Act of the International Conference on Safety of Life at Sea, 1960, and to determine whether the ship has been and is being operated in accordance with its Safety Assessment and its Operating Manual.

# SECURITY

# Article 22

As regards the security of the ship while in Netherlands waters the Netherlands Government only accepts the responsibilities it usually accepts with regard to conventional ships.

The corresponding articles of the Agreement concluded with Italy are the following:

# Article III. Port arrangements

- (a) The Italian Government shall give the competent authorities the instructions necessary for the entry of the ship into Italian ports and for the use thereof.
- (b) The competent Italian authorities shall take all necessary measures for fire safety and police protection, crowd control, and the general preparation of facilities relating to the entry of the ship.
- (c) Control of public access to the ship shall be the responsibility of the master of the ship. Special arrangements for such control shall be agreed upon by the master and the authorities designated by the Italian Government.
- (d) The master shall comply with local regulations. If the operator or the master himself considers that the application of those regulations does not fulfil the safety requirements of operation of the

nuclear plant, the necessary measures shall be agreed upon in this connection.

(e) The Italian Government shall see to the surveillance of the areas in the vicinity of the ship, with the assistance of the Government of the United States, as mutually agreed.

#### Article IV. Inspection

While the ship is in Italian territorial waters, the designated authorities shall have reasonable access to it for purposes of inspecting the ship and its operating records and programme data, to determine whether it has been operated in accordance with the Manual of Operations

It thus appears that, once an activity takes place within the territory of a potentially injured State, that State is responsible for local security.

# (c) Judicial decisions and State practice other than agreements

216. In judicial decisions and official correspondence, the characterization of harm has been much influenced by the need to balance interests. These appears to be no fixed and definite substantive rule as to what constitutes injury; rather, there are a set of factors that are balanced against one another. In some judicial decisions regarding competing uses of shared resources, certain uses have been given priority over others. The priority of one use over others occasionally appears to have been affected by crisis conditions, such as the degree of tension and instability in international relations. For example, the United States of America and the United Kingdom, in preparing for their nuclear tests in the Eniwetok Atoll and Christmas Islands in the mid-1950s, took the position that military exercises were a traditional use of the high seas. They maintained that the danger areas which they had established, on the basis of their investigations, would prevent the occurrence of substantial losses. Finally, they claimed that the inconvenience that the tests might cause for other traditional uses of the high seas could not validly be argued against military uses, the purpose of which was not only the protection of important security interests of their respective countries but also the strengthening of the security of the "free world". Thus they unilaterally balanced their security needs against the interests of certain other States in remaining free of the health hazards caused by radioactive fallout. Security won in the

217. In a note addressed on 19 March 1956 to the ambassador of Japan in Washington, the Department of State set forth the position of the United States as follows:

The United States recognizes and strongly sympathizes with the humane motivations which inspired the resolutions of the Japanese Diet, but is constrained to point out that the problem of suspending nuclear weapons tests cannot be treated separately from the establishment of a safeguarded and controlled disarmament programme.

The United States Government is convinced that the proposed nuclear tests are vital to its own defence and the defence of the free world because the possession and competence in the use of nuclear weapons by leading nations of the free world are the chief deterrent to aggression and to war. International agreement to abandon tests without effective safeguards against the clandestine development of new weapons would involve a reliance by the United States upon the good intentions of certain nations not justified by the record of their actions in the past.

The United States Government is convinced that no world-wide health hazard exists from the past or planned tests. In this connection

<sup>151</sup> See footnote 22 above.

<sup>152</sup> See footnote 20 above.

the United States proposed a resolution unanimously adopted by the United Nations tenth General Assembly establishing a scientific committee on radiation, of which Japan is a member, to facilitate pooling and distribution of all available scientific data on the effects of radiation upon man and his environment. During the forthcoming tests the United States will make every effort to eliminate any danger and to minimize any inconvenience to maritime commerce and fishing.

It cannot be regarded as established on the basis of present information that substantial economic losses will result from the establishment of the danger area. Military exercises are a traditional use of the high seas, and the Government of the United States considers that inconvenience for other traditional uses which may result therefrom is not compensable as a matter of right.

In conclusion the Acting Secretary wishes to give the assurance that the United States continues only such tests as are essential to the strength of the free world defence and security. It has sought and will continue to seek with renewed efforts a system for a safeguarded and controlled disarmament programme which ultimately may lead to the type of action envisaged by the resolutions of the Japanese Diet.<sup>153</sup>

218. The United Kingdom, in defending its nuclear tests, maintained that the temporary use of areas outside territorial waters for nuclear testing corresponded to the use of those waters for gunnery or bombing practice; such use had never been considered to be a violation of the principle of freedom of navigation on the high seas, and hence no special agreement was called for. The United Kingdom considered that, by announcing the establishment of a temporary danger area on the high seas, it was acting in the safety interest of other countries, and stated that it had tried not to interfere with regular shipping routes.<sup>154</sup>

219. Some decisions imposing the duty to balance the interests of the States parties refer to interests in general terms and leave it up to each State individually to determine which factors are to be weighed. The tribunal in the *Lake Lanoux* case took this approach:

... The Tribunal is of the opinion that, according to the rules of good faith, the upstream State is under the obligation to take into consideration the various interests involved, to seek to give them every satisfaction compatible with the pursuit of its own interests, and to show that in this regard it is genuinely concerned to reconcile the interests of the other riparian State with its own.<sup>155</sup>

The co-operation established in 1957 between the United States of America and Mexico in order to modify, in their respective interests, the plans for a highway constructed by Mexico parallel to the United States border, but without halting the progress of the work undertaken, is a clear example of the balancing of interests. As mentioned above (para. 163), the construction, as initially planned, had threatened to provoke flooding and thus of causing injury to residents and their property on the other side of the border. Realizing the importance of the construction of the highway to Mexico, the United States section of the International Boundary and Water Commission acted for two years in an engineering advisory capacity to the United States Department of State and the United States Consulate in Mexico in their discussions of the projects and safety precautions in order to ensure that the highway, once completed, did not cause injuries.156 As a result of the technical discussions, several modifications of the original plan were agreed upon.

221. The issue of tolerance of a "natural risk" was dealt with in a judicial decision in a dispute between two Swiss cantons. At the end of the first phase of the Solothurn v. Aargau case (1900),157 the Swiss Federal Court ruled, on the basis of the applicable principles of international law, in favour of the complete protection of Solothurn from the risks associated with target practice at its border by the neighbouring canton of Aargau. However, at the conclusion of the second phase (1915), the Court reversed its decision and authorized the continuation of the target practice.158 The Court had concluded that, despite additional safety measures, the probability of stray bullets could not be eliminated, but that the use of the range created a "practically inevitable natural risk" which must be tolerated between neighbours. The reversal of the judgment was apparently due to federal legislation passed after the first decision, which required local communities to provide military target practice facilities. Since absolutely safe practice facilities in the communities concerned were unavailable, the Court found that the neighbouring canton's demand for absolute protection against transboundary crossing of bullets was in conflict with the implementation of federal legislation.

222. In judicial decisions between federated States, the United State Supreme Court has referred to the *prior existence of a beneficial use* as a factor that deserves relative but not absolute protection. In the Washington v. Oregon case (1936), 159 the Supreme Court ruled:

A priority once acquired or put in course of acquisition by the posting of a notice may be lost to the claimant by abandonment or laches . . . The essence of the doctrine of prior appropriation is beneficial use, not a stale or barren claim. Only diligence and good faith will keep the privilege alive.\* 160

223. In the Nebraska v. Wyoming case (1945), 161 while the Supreme Court acknowledged the importance of prior appropriation of waters, it held that it must be balanced against other factors and thus that that factor was not per se the determining element:

... Priority of appropriation is the guiding principle. But physical and climatic conditions, the consumptive use of water in the several sections of the river, the character and rate of return flows, the extent of established uses, the availability of storage water, the practical effect of wasteful uses on downstream areas, the damage to upstream areas as compared to the benefits to downstream areas if a limitation is imposed on the former—these are all relevant factors....<sup>162</sup>

224. The Supreme Court has also referred to reciprocal restraints in demands by parties about using a shared resource. In the New Jersey v. New York case (1931), 163 the Court emphasized that, just as the upper riparian could not cut off the flow of water towards the lower riparian, the latter could not require the former to

<sup>&</sup>lt;sup>153</sup> Whiteman, op. cit. (footnote 45 above), vol. 4, pp. 576-577.

<sup>154</sup> Ibid., p. 600.

<sup>155</sup> International Law Reports 1957 (see footnote 67 above), p. 139, para. 22 of the award.

<sup>156</sup> Whiteman, op. cit., vol. 6, p. 260.

<sup>137</sup> Judgment of 1 November 1900, Entscheidungen des schweizerischen Bundesgerichtes, 1900, vol. 26, part 1, p. 444.

<sup>138</sup> Aargau v. Solothurn, judgment of 4 February 1915, Entscheidungen . . ., 1915, vol. 41, part 1, p. 126.

<sup>159</sup> United States Reports, vol. 297, p. 517.

<sup>160</sup> Ibid., p. 527.

<sup>&</sup>lt;sup>161</sup> *Ibid.*, vol. 325, p. 589.

<sup>162</sup> Ibid., p. 618.

<sup>163</sup> Ibid., vol. 283, p. 336.

give up its interests in the river so that the river might come down to the latter undiminished.<sup>164</sup>

225. The United States Supreme Court has rejected the concept of *future use* as an element which should frustrate present equitable use. In the *Connecticut* v. *Massachusetts* case (1931), 165 the Supreme Court, after reviewing arguments supporting the granting of an injunction against a diversion by Massachusetts of waters of an interstate river, concluded:

... At most, there is a mere possibility that at some undisclosed time the owner [of an allegedly affected power station in Connecticut], were it not for the diversion, might construct additional works capable of using all of the flow of the river including the waters proposed to be taken by Massachusetts. Injunction will not issue in the absence of actual or presently threatened interference. 166

226. In the same case, the United States Supreme Court also touched upon the principles of *efficient* use and *alternative* modes of resource use to avoid transboundary injuries. The Court examined in detail the possible consequences for both Connecticut and Massachusetts of feasible alternative arrangements whereby Massachusetts could have avoided a diversion of waters from an interstate stream and any possibility of conflict with the lower riparian, Connecticut.<sup>167</sup> Similarly, in the *Kansas* v. *Colorado* case (1907), <sup>168</sup> the Supreme Court examined the changes brought about in Kansas, the lower riparian, as a result of the appropriation of a certain portion of the flow of their shared river by Colorado. The Court stated:

In the same case the Court referred to the principle of *non-discrimination* as relevant in balancing the interests of the two federated States. The Court noted:

227. The superiority of the principle of perfect equality of States over any preferential treatment was reflected in the decision of 10 September 1929 of the Permanent Court of International Justice in the *River Oder* case:

... [A] community of interest in a navigable river becomes the basis of a common legal right, the essential features of which are the perfect equality of all riparian States in the use of the whole course of the river

and the exclusion of any preferential privilege of any one riparian State in relation to the others.  $^{\rm trt}$ 

If the concept of "perfect equality" refers only to the physical and geographical relationship among States and between them, on the one hand, and a shared resource on the other hand, it is then a simple and rather mechanical formula. But if it purports to consider factors of an economic, social, historical, humanitarian, etc. nature, then this concept is much too complicated.

The position of the International Court of Justice in two cases involving the determination of the distribution of a shared resource or its use between two or more States was not uniform. In the Anglo-Norwegian Fisheries case, 172 the Court, in deciding whether the 1935 Norwegian decree concerning the delimitation of the Norwegian fisheries zone was compatible with the principles of international law, emphasized primarily the geographical factors of the coastal configuration. After referring to some rather vague criteria to provide an adequate basis for a decision, the Court basically examined the geographical configuration of the coastal States. Although the Court referred to "practical needs and local requirements", it concluded that the drawing of baselines must not depart to any appreciable extent from the general direction of the coast. Finally, the Court mentioned certain economic interests peculiar to a region as relevant factors:

In this connection, certain basic considerations inherent in the nature of the territorial sea, bring to light certain criteria which, though not entirely precise, can provide courts with an adequate basis for their decisions,\* which can be adapted to the diverse facts in question.

Among these considerations, some reference must be made to the close dependence of the territorial sea upon the land domain. It is the land which confers upon the coastal State a right to the waters off its coasts. It follows that while such a State must be allowed the latitude necessary in order to be able to adapt its delimitation to practical needs and local requirements,\* the drawing of baselines must not depart to any appreciable extent from the general direction of the coast.

Another fundamental consideration, of particular importance in this case, is the more or less close relationship existing between certain sea areas and the land formations which divide or surround them. The real question raised in the choice of baselines is in effect whether certain sea areas lying within these lines are sufficiently closely linked to the land domain to be subject to the regime of internal waters. This idea, which is at the basis of the determination of the rules relating to bays, should be liberally applied in the case of a coast, the geographical configuration of which is as unusual as that of Norway.

Finally, there is one consideration not to be overlooked, the scope of which extends beyond purely geographical factors: that of *certain economic interests\** peculiar to a region, the reality and importance of which are clearly evidenced by a *long usage.\** 173

229. Eighteen years later, in the North Sea Continental Shelf cases,<sup>174</sup> the Court seems to have placed less emphasis on geographical factors. In those cases, Denmark and the Netherlands claimed that the continental shelf which they shared with the Federal Republic of Germany should be delimited in accordance with article 6 of the 1958 Geneva Convention on the Continental Shelf. That article refers to "equidistance" as the ap-

<sup>164</sup> Ibid., p. 342.

<sup>165</sup> Ibid., vol. 282, p. 660.

<sup>166</sup> Ibid., p. 673.

<sup>167</sup> Ibid., pp. 668-674.

<sup>&</sup>lt;sup>168</sup> *Ibid.*, vol. 206, p. 46.

<sup>169</sup> Ibid., pp. 100-101.

<sup>170</sup> Ibid., pp. 104-105.

<sup>&</sup>lt;sup>171</sup> Case relating to the Territorial Jurisdiction of the International Commission of the River Oder, *P.C.I.J.*, Series A, No. 23, p. 27.

<sup>1721.</sup>C.J. Reports 1951, p. 116.

<sup>173</sup> Ibid., p. 133.

<sup>174</sup> I.C.J. Reports 1969, p. 3.

plicable principle for the division of the continental shelf in the absence of agreement among the States, unless, or to the extent to which circumstances" are recognized to exist. The Court, after considering that article 6 of the Geneva Convention did not impose a mandatory obligation upon the parties, introduced the concept of "equitable principles" founded upon the principles of justice and good faith. The Court stated that geographical factors were not the only considerations; that there was in fact no legal limitation to the factors to be considered in order to make it possible to apply equitable procedures:

- 93. In fact, there is no legal limit to the considerations\* which States may take account of for the purpose of making sure that they apply equitable procedures, and more often than not it is the balancing-up of all such considerations that will produce this result rather than reliance on one to the exclusion of all others. The problem of the relative weight to be accorded to different considerations naturally varies with the circumstances of the case.
- 94. In balancing the factors in question it would appear that various aspects must be taken into account. Some are related to the geological, others to the geographical aspect of the situation, others again to the idea of the unity of any deposits. These criteria, though not entirely precise, can provide adequate bases for decision adapted to the factual situation.
- 95. The institution of the continental shelf has arisen out of the recognition of a physical fact; and the link between this fact and the law, without which that institution would never have existed, remains an important element for the application of its legal régime. . . . !
- That approach was followed more explicitly by the Court in the Fisheries Jurisdiction case, 176 in which the United Kingdom disputed Iceland's unilateral expansion of its fisheries zone. The Court ordered the parties to take into account factors such as the public interest of the populations of the States involved, the dependence of coastal populations for their livelihood on the fishing resources of the disputed area, and consequently the interest of the parties in the conservation and equitable exploitation of those resources. Thus the rights and interests of the respective States varied according to their economic dependence on the resource. The Court stated that the preferential rights of a coastal State in a particular circumstance was not a static concept. A State's preferential rights as a function of an exceptional dependence varied as the extent of that dependence changed. The Court further referred to two distinct interests of each of the States involved: the livelihood of people and economic development. It was essential, the Court stated, that in each case the dependence of the coastal State on the fisheries be appraised in relation to that of the other State concerned in order to design an equitable régime of exploitation of fishing resources (see paragraph 70 of the judgment, at para. 155 above). Finally, the Court held that Iceland and the United Kingdom were under the obligation to negotiate, taking into account the following factors:
- (a) that in the distribution of the fishing resources in the areas specified in subparagraph 2 Iceland is entitled to a preferential share to the extent of the special dependence of its people upon the fisheries in the seas around its coasts for their livelihood and economic development;
- (b) that by reason of its fishing activities in the areas specified in sub-

- fishery resources of the said areas on which elements of its people depend for their livelihood and economic well-being;
- (c) the obligation to pay due regard to the interests of other States in the conservation and equitable exploitation of these resources;
- (d) that the above-mentioned rights of Iceland and of the United Kingdom should each be given effect to the extent compatible with the conservation and development of the fishery resources in the areas specified in subparagraph 2 and with the interests of other States in their conservation and equitable exploitation;
- (e) their obligation to keep under review those resources and to examine together, in the light of scientific and other available information, such measures as may be required for the conservation and development, and equitable exploitation, of those resources, making use of the machinery established by the North-East Atlantic Fisheries Convention or such other means as may be agreed upon as a result of international negotiations.17
- An important element that may have affected the decision of the Court in this case was the Third United Nations Conference on the Law of the Sea. The Court may not have wished to make any ruling that would have been incompatible with the Conference negotiations on the delimitation of the fishery zone. In fact, the Court in its findings asked the parties to make use in their negotiations of the machinery established by the 1959 North-East Atlantic Fisheries Convention or "such other means as may be agreed upon as a result of international negotiations".
- More recently, in its judgment of 24 February 1982 in the Continental Shelf case between Tunisia and the Libyan Arab Jamahiriya, 178 the International Court of Justice decided that it was bound to take a decision on the basis of equitable principles, "divorced from the concept of natural\* prolongation". The Court further stated:
- . The result of the application of equitable principles must be equitable. This terminology, which is generally used, is not entirely satisfactory because it employs the term equitable to characterize both the result to be achieved and the means to be applied to reach this result. It is, however, the result which is predominant, the principles are subordinate to the goal. . . . 180

The Court did not identify any equitable principle, but it deemed it to be its task "to balance up the various considerations which it regards as relevant in order to produce an equitable result". 181 It nevertheless sought to propose a procedure for the consideration of various factors by stating:

. . . While it is clear that no rigid rules exist as to the exact weight to be attached to each element in the case, this is very far from being an exercise of discretion or conciliation; nor is it an operation of distributive justice. . . . 182

The Court stated that, in addition to giving consideration to the maritime limits claimed by both parties, it must take into account the claims to historic rights made by Tunisia as well as a number of economic considerations which one or the other party had urged as relevant.183 Tunisia had emphasized its relative poverty vis-à-vis the Libyan Arab Jamahiriya in terms of absence of natural resources such as oil and gas, as well

paragraph 2, the United Kingdom also has established rights in the

<sup>125</sup> Ibid., pp. 50-51.

<sup>176</sup> I.C.J. Reports 1974, p. 3.

<sup>177</sup> Ibid., pp. 34-35, para. 79, subpara. 4, of the judgment.

<sup>178</sup> I.C.J. Reports 1982, p. 18.

<sup>179</sup> Ihid., p. 59, para. 70 of the judgment.

<sup>180</sup> Ibid., p. 60.

<sup>&</sup>lt;sup>181</sup> Ibid., p. 60, para. 71.

<sup>182</sup> Ibid.

<sup>183</sup> Ibid., pp. 64-65, para. 82.

as its economic dependence on fishing resources derived from its "historic waters", which supplemented its national economy and enabled it to survive as a country. The Court, however, found that those economic considerations could not be taken into account in that case, on the grounds that:

- ... They [these economic factors] are virtually extraneous factors since they are variables which unpredictable national fortune or calamity, as the case may be, might at any time cause to tilt the scale one way or the other. A country might be poor today and become rich tomorrow as a result of an event such as the discovery of a valuable economic resource. . . . 125
- 233. Finally, the Court decided that the delimitation should be effected "in accordance with equitable principles, and taking into account of all relevant circumstances", those circumstances including the following:
- (1) the fact that the area relevant to the delimitation in the present case is bounded by the Tunisian coast from Ras Ajdir to Ras Kaboudia and the Libyan coast from Ras Ajdir to Ras Tajoura and by the parallel of latitude passing through Ras Kaboudia and the meridian passing through Ras Tajoura, the rights of third States being reserved;
- (2) the general configuration of the coast of the Parties, and in particular the marked change in direction of the Tunisian coastline between Ras Ajdir and Ras Kaboudia;
- (3) the existence and position of the Kerkennah Islands;
- (4) the land frontier between the Parties, and their conduct prior to 1974 in the grant of petroleum concessions, resulting in the employment of a line seawards from Ras Ajdir at an angle of approximately 26° east of the meridian, which line corresponds to the line perpendicular to the coast at the frontier point which had in the past been observed as a de facto maritime limit;
- (5) the element of a reasonable degree of proportionality, which a delimitation carried out in accordance with equitable principles ought to bring about between the extent of the continental shelf areas appertaining to the coastal State and the length of the relevant part of its coast, measured in the general direction of the coastlines, account being taken for this purpose of the effects, actual or prospective, of any other continental shelf delimitation between States in the same region. 186

With one exception, the Court cited basically physical and geographical considerations. The exception is indicated in factor 4, which refers to "conduct prior to 1974 in the grant of petroleum concessions". The Court appears to have rejected some of the economic factors, which it had considered appropriate in its previous decisions (see paras. 228-230 above).

234. The difficulties associated with the concept of equitable principles, which do not appear to have been resolved, concern first the kind of and limit to the factors to be considered and, secondly, the value to be attached to the different factors. The decision of the Court in the *Continental Shelf* case does not appear to have *resolved* these questions.<sup>187</sup>

- 236. Precedent demonstrates that, where the interests of the parties concerned have had to be balanced in connection with a particular act or course of conduct, such balancing has been effected either by the parties jointly or by a third party. In two cases examined in this study (the nuclear tests in the *Eniwetok Atoll* and *Christmas Islands*), the acting States *unilaterally* balanced their own security interests against the interests of other States (see paras. 216 and 218 above).
- 237. In the Lake Lanoux case, the tribunal considered that the conclusion of a comprehensive agreement between the parties was the only way to achieve a balancing of the parties' interests:
- ... The only way to arrive at such compromises of interests is to conclude agreements on an increasingly comprehensive basis. International practice reflects the conviction that States ought to strive to conclude such agreements; there would thus appear to be an obligation to accept in good faith all communications and contacts which could, by broad comparison of interests and reciprocal good will, provide States with the best conditions for concluding agreements....<sup>190</sup>

The parties were thus considered the best placed to balance their interests. Similarly, in the North Sea Continental Shelf cases, the International Court of Justice directed the States parties to take into consideration certain factors to balance their interests in delimiting the continental shelf. The States parties voluntarily recognized the competence of the Court to prescribe how the continental shelf was to be divided, while the States reserved the right to apply those prescriptions themselves. In the Fisheries Jurisdiction case, the Court ordered the parties to take into account certain factors which would ensure the balance of the parties' interests in dividing the exploitation of certain fishery resources. The States parties were to apply those prescriptions themselves. In the Anglo-Norwegian Fisheries case, however, the Court, with the consent of both States parties, prescribed the criteria relevant to the balancing of interests and then itself applied those criteria to determine whether or not Norway's unilateral action in delimiting its fisheries zone had taken into account the interests of the neighbouring State.

<sup>235.</sup> In its arbitral award to 30 June 1977 in the Delimitation of the continental shelf (Channel Islands) case, 188 the tribunal referred to the "equitable principle" as relevant in balancing the rights and interests of the parties in delimiting their continental shelf. In that respect it found that the claim made by the United Kingdom that, in dividing the continental shelf, account should be taken, among other considerations, of that country's responsibility for the defence and security of its islands in the Channel, carried a "certain weight". 189

<sup>&</sup>lt;sup>184</sup> *Ibid.*, p. 77, para. 106.

<sup>185</sup> Ibid., para. 107.

<sup>186</sup> Ibid., p. 93, para. 133.

<sup>187</sup> One of the first references to equitable principles is to be found in the proclamation made on 28 September 1945 by President Truman in connection with the policy of the United States of America in regard to the sea-bed and subsoil of the continental shelf. The proclamation refers to the "equitable principles" on the basis of which the continental shelves of adjacent States should be divided. Without defining the principle, the proclamation provides that: "where the continental shelf [of the United States] extends to the shores of another State, or is shared with an adjacent State, the boundary shall

be determined by the United States and the State concerned in accordance with equitable principles". (United States of America, *The Department of State Bulletin*, vol. XIII, No. 327, 30 September 1945, p. 485.)

<sup>\*\*\*</sup> Delimitation of the continental shelf between the United Kingdom of Great Britain and Northern Ireland and the French Republic (United Nations, Reports of International Arbitral Awards, vol. XVIII, pp. 3 et seq.).

<sup>189</sup> Ibid., pp. 93-94, paras. 197-198.

<sup>190</sup> International Law Reports 1957 (see footnote 67 above), pp. 129-130, para. 13 of the award.

238. On occasion, the acting State, at the request of the injured State, has unilaterally decided to halt or modify the conduct of a particular hazardous activity. In 1892, French troops staged target practice exercises near the Swiss border. After Switzerland had protested that the exercises endangered a Swiss locality near the border, the French military authorities halted the exercises until steps had been taken to avoid *accidental* transboundary injuries.<sup>191</sup>

239. In the case of the construction of a highway in Mexico, the United States of America and Mexico, through negotiation, balanced their interests and agreed upon a solution, while in the case of the nuclear testing in the atmosphere by the United States and the United Kingdom in the Eniwetok Atoll and Christmas Islands, the acting States themselves balanced their own security interests against the interests of the affected entities. They claimed, of course, that in thus balancing those interests they had taken into account the principles of international law.

240. In connection with another series of nuclear tests, the United States unilaterally balanced its own security interests with those of other States. In 1971, the United States planned to conduct a third series of underground nuclear tests, named Cannikin, 192 in the Aleutian island of Amchitka. Upon the announcement of the plan, the Canadian Government protested and expressed its concern to the Government of the United States. The reason why the protest was addressed to the United States and not to other Governments engaged in underground nuclear testing at the time was, as explained by the Canadian Secretary of State for External Affairs, that these tests would have a special effect on Canada:

... such a test as was proposed could have a direct effect on people living on the Pacific Coast in both Canada and the United States. Indeed, such a nuclear explosion [was] to be condemned on two counts: first, it [was] a continuation of the testing and, second, because it happen[ed] to be in an area of difficult terrain where there might be untoward effects.<sup>193</sup>

Canada feared that the tests might produce a major earthquake, a tidal wave, or leakage of radioactive materials into the environment. In response to the Canadian concern, the United States assured that country that it would take full account of Canada's interests.<sup>194</sup>

# D. Exoneration from the duty of prior negotiation

241. Under certain conditions, States may undertake activities which they know will cause extraterritorial injuries without prior consultation. Such situations may be rare, but nevertheless can occur, as for example in cases of "self-help", "self-defence" or *force majeure*. Exemption from prior negotiations or consultations does not necessarily secure exemption from all levels of

impact assessment. Depending upon the conditions under which particular activities are undertaken, the impact assessment may be carried out by different procedures appropriate to the "crisis" situation. Even during the "crisis" situation, it may be expected that some consideration will be given to minimizing injuries to others. Nor does it appear that exemption from prior consultation necessarily entails the exoneration of the acting State from liability for damage.

# (a) Multilateral agreements

242. The 1973 Protocol relating to Intervention on the High Seas in Cases of Marine Pollution by Substances other than Oil contains no provision requiring prior negotiation in cases of self-help. Article I reads:

#### Article I

- 1. Parties to the present Protocol may take such measures on the high seas as may be necessary to prevent, mitigate or climinate grave and imminent danger to their coastline or related interests from pollution or threat of pollution by substances other than oil following upon a maritime casualty or acts related to such a casualty, which may reasonably be expected to result in major harmful consequences.
- 2. "Substances other than oil" as referred to in paragraph I shall be:
- (a) those substances enumerated in a list which shall be established by an appropriate body designated by the Organization and which shall be annexed to the present Protocol, and
- (b) those other substances which are liable to create hazards to human health, to harm living resources and marine life, to damage amenities or to interfere with other legitimate uses of the sea.
- 3. Whenever an intervening Party takes action with regard to a substance referred to in paragraph 2 (b) above that Party shall have the burden of establishing that the substance, under the circumstances present at the time of the intervention, could reasonably pose a grave and imminent danger analogous to that posed by any of the substances enumerated in the list referred to in paragraph 2 (a) above.

The potentially injured State, of course, has the burden of establishing that the substances, under the circumstances present at the time of intervention, could reasonably *pose a grave and imminent danger* analogous to that posed by any of the substances enumerated in the Protocol.

243. The 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties sets aside the requirement of prior negotiation in cases of extreme urgency requiring measures to be taken immediately. Paragraph (d) of article III reads:

(d) in cases of extreme urgency requiring measures to be taken immediately, the coastal State may take measures rendered necessary by the urgency of the situation, without prior notification or consultation or without continuing consultations already begun;

However, under articles V and VI of the Convention, the coastal State will be responsible for injuries if its measures go beyond what is proportionate to the danger and reasonable to prevent injuries.

244. Section 5 of part XII (arts. 207-212) of the 1982 United Nations Convention on the Law of the Sea provides that States, in addition to respecting prescribed international rules, have the competence to prescribe domestic laws to prevent, reduce and control pollution of the marine environment. Section 6 (arts. 213-222) enumerates measures for enforcing the rules set out in section 5, including the requirement that coastal States adopt national legislation and take *unilateral* measures

<sup>&</sup>lt;sup>191</sup> P. Guggenheim, "La pratique suisse (1956)", Annuaire suisse de droit international, 1957 (Zurich), vol. XIV, p. 168.

<sup>192</sup> See International Canada, Toronto, vol. 2, 1971, p. 97.

<sup>&</sup>lt;sup>193</sup> *Ibid.*, p. 185.

<sup>194</sup> Ibid., p. 199.

to enforce the principle of protection of the environment embodied in the Convention and to protect their coastal interests. In order to balance the various interests involved, article 232, while recognizing the unilateral competence of coastal States, warns against the liability they may incur if they take unlawful or unreasonable enforcement measures and cause injuries:

# Article 232. Liability of States arising from enforcement measures

States shall be liable for damage or loss attributable to them arising from measures taken pursuant to section 6 when such measures are unlawful or exceed those reasonably required in the light of available information. States shall provide for recourse in their courts for actions in respect of such damage or loss.

Paragraph 3 of article 142 of the Convention also provides for exoneration from prior consultations, required by paragraph 2 of the same article, when coastal States have to take measures of self-help to prevent imminent danger to their coastlines:

3. Neither this part nor any rights granted or exercised pursuant thereto shall affect the rights of coastal States to take such measures consistent with the relevant provisions of part XII as may be necessary to prevent, mitigate or eliminate grave and imminent danger to their coastline, or related interests from pollution or threat thereof or from other hazardous occurrences resulting from or caused by any activities in the area.

# (b) Bilateral agreements

245. Exoneration from prior negotiation concerning measures of self-help is provided for in at least two bilateral agreements examined in this study. These agreements do not remove the requirement of prior consultation and negotiation altogether, but set it aside in certain crisis situations. For example, the 1975 Agreement between Canada and the United States of America relating to weather modification activities195 provides in article VI that extreme emergencies, such as forest fires, may require immediate commencement by one party of weather modification activities of mutual interest, notwithstanding the lack of sufficient time for prior notification or consultation as required by the Agreement. In such cases, however, the party undertaking such activities shall notify and fully inform the other party as soon as practicable and shall promptly enter into consultations at the request of the other party. Article VI reads:

# Article VI

The parties recognize that extreme emergencies, such as forest fires, may require immediate commencement by one of them of weather modification activities of mutual interest notwithstanding the lack of sufficient time for prior notification pursuant to article IV, or for consultation pursuant to article V. In such cases, the party commencing such activities shall notify and fully inform the other party as soon as practicable, and shall promptly enter into consultations at the request of the other party.

246. The 1922 Treaty between Germany and Denmark concerning frontier waters<sup>196</sup> also provides that *protective* measures may be taken without prior authorization in *case of danger*. If these measures are to become permanent, then the party that has taken them must obtain authorization immediately after the danger has been averted. The last paragraph of article 29 reads:

Protective measures taken in cases of necessity when danger is threatening require no authorization. If, however, they become permanent, authorization shall be obtained when the immediate danger has been averted.

# (c) Judicial decisions and State practice other than agreements

247. In the judicial decisions and official correspondence examined in this study, there is no instance in which exoneration from prior negotiation has been recognized. Of course, by negotiation, the present study refers in this context to any form of submission of notice regarding the commencement of an activity. For example, there might be serious question whether the general notices given by the United States of America and the United Kingdom about their nuclear tests constituted prior negotiation. Similarly, it is open to question whether the exchange of official correspondence between those States and Japan and between the United States and Canada regarding the Cannikin tests can be considered prior negotiations. Most probably, the correspondence between the acting and the affected States forced the acting States to re-examine their projects in the light of the objections and concerns raised by the affected States. They were not negotiations in the sense of a joint re-evaluation of the activities. At the same time, the overall reaction of Japan and Canada cannot be interpreted as their expression of consent to the procedure, nor as complete opposition to it.

248. Another ambiguity in the characterization of exoneration from prior negotiation may be observed in a communication addressed by Mexico in 1955 to the United States of America informing the Government of that country that the facilities for protection against flooding which Mexico was planning to construct might in turn cause flooding in the United States. The issue had first been raised in 1951, in a series of exchanges of official correspondence between Mexico and the United States concerning the construction of a drainage canal for the purpose of preventing the flooding caused by the collection of rain water in the Rose Street Canal in Douglas, in the United States. The damage annually caused by flooding to the Mexican city of Agua Prieta was substantial. The damage to the United States was insignificant. However, the facilities for protection constructed in Mexico would have reversed the flooding to the United States. Two years of negotiations concerning the construction of a dike failed to produce agreement on the plan. Finally, in a letter addressed to the United States Secretary of State on 24 March 1955, the Mexican Ambassador, after referring to the fruitless efforts Mexico had made to reach agreement with the United States, and the substantial injuries Mexico was suffering annually because of the flood, informed the United States Government that, as of 1 May of that year, the Mexican Government would begin to construct certain protective works to prevent the entry into the Mexican city of rain water collected by the Rose Street Canal, in Douglas, in the United States. Thus, Mexico informed the United States Government of the situation so that the latter might take such measures as it considered advisable to prevent consequences which the return of such water might have in the city of Douglas.197 In a letter dated 12

<sup>195</sup> See footnote 16 above.

<sup>196</sup> See footnote 59 above.

<sup>197</sup> Whiteman, op. cit. (footnote 45 above), vol. 6, p. 264.

May 1955 to the mayor of Douglas, the Assistant Secretary of State explicitly recognized the right of Mexico to take protective measures at any time to prevent injuries to its territory:

... there would seem to be no doubt that Mexico has the right\* to prevent water coming into Mexico through the Rose Street Canal by the construction at any time of a dike on the Mexican side of the international boundary....<sup>198</sup>

The Assistant Secretary of State, however, referred to principles of international law obliging every State to respect the full sovereignty of other States and to refrain from *taking* or *authorizing* actions on its territory which caused injuries to another State:

... On the other hand, the principle of international law which obligates every State to respect the full sovereignty of other States and to refrain from creating or authorizing or countenancing the creation on its territory of any agency, such as the Rose Street Canal, which causes injury to another State or its inhabitants, is one of long standing and universal recognition.<sup>199</sup>

# CHAPTER III

# Preventive measures

249: Preventive measures entail the taking of decisions in connection with harmful activities in order to prevent or minimize injuries. In the context of the concept of negligence, preventive measures are those taken in order to minimize or prevent the injurious effects of conduct involving excessive risk of harm to others. Some of these measures are of a general nature, such as would be expected of any reasonable person in the exercise of his own best judgment. Others-external measures-are reguired by law and entail liability for the person who is ignorant of them, regardless of his intent. In the context of inter-State relations, the acting State is obliged to carry out the measures upon which it has agreed with the injured State. Sometimes the acting State may not be required to take specific preventive measures, and to adopt them only if it deems them necessary. However, in situations where preventive measures, whether of a general nature or externally imposed, have to be taken, the acting State is not necessarily exempted from the obligation to use its best judgment. Such judgment must be that of an expert, not of a layman. The originators of acts that may produce extraterritorial consequences are specialists. Government personnel, who are normally responsible for the application of government regulations concerning certain activities of private entities, are expected to be skilled and to have special competence. It is thus in this sense that the concept of the best judgment of a reasonable person in the context of activities with extraterritorial consequences is to be interpreted.

250. Preventive measures have been developed in treaties and other forms of State practice as regards both procedure and content. At the procedural level, such measures include the process of management and monitoring, the setting up of an institution for such a function, securing it operational activities and thus ensuring the continuity and effectiveness of assessment of activities, etc. This process may be equipped to take into account new elements developed during the preparations for or the operation of potentially harmful activities. State practice demonstrates that the management and monitoring process may be undertaken by the States concerned, by a third party (including a commission) or occasionally by other entities which may be injured by the activities.

251. The content of preventive measures includes recommendations concerning changes to be made to prevent or minimize injuries or remedy damage. Recommendations or monitoring decisions may require a specific or a more general change in the process of the performance of activities. Thus they may relate to prevention or minimization of harm, or to methods to ensure payment of compensation in case of liability.

# A. Management and monitoring

# (a) Multilateral agreements

Preventive measures may involve the adoption of national legislation and other regulatory provisions. The 1982 United Nations Convention on the Law of the Sea, for example, provides, in section 5 of part XII, for international rules and national legislation to prevent, reduce and control pollution of the marine environment. Section 6 of the same part empowers coastal States themselves to take unilateral enforcement measures to enforce the principles set out in section 5 and to prevent or minimize injuries to themselves which may be caused by activities undertaken by other States. The most directly relevant article of this section is article 221 regarding measures to avoid pollution arising from maritime casualties. This article in fact grants unilateral management and monitoring competence to coastal States to take and enforce, pursuant to international law, measures beyond their territorial sea proportionate to the actual or threatened damage to protect their coastline or related interests against incidents which may reasonably be expected to result in major harmful consequences:

# Article 221. Measures to avoid pollution arising from maritime casualties

- 1. Nothing in this part shall prejudice the right of States, pursuant to international law, both customary and conventional, to take and enforce measures beyond the territorial sea proportionate to the actual or threatened damage to protect their coastline or related interests, including fishing, from pollution or threat of pollution following upon a maritime casualty or acts relating to such a casualty, which may reasonably be expected to result in major harmful consequences.
- 2. For the purposes of this article, "maritime casualty" means a collision of vessels, stranding or other incident of navigation, or other

<sup>198</sup> Ibid., p. 265.

<sup>199</sup> Ibid.

occurrence on board a vessel or external to it resulting in material damage or imminent threat of material damage to a vessel or cargo.

253. Other relevant articles of section 6 are the following:

# Article 218. Enforcement by port States

- 1. When a vessel is voluntarily within a port or at an off-shore terminal of a State, that State may undertake investigations and, where the evidence so warrants, institute proceedings in respect of any discharge from that vessel outside the internal waters, territorial sea or exclusive economic zone of that State in violation of applicable international rules and standards established through the competent international organization or general diplomatic conference.
- 2. No proceedings pursuant to paragraph I shall be instituted in respect of a discharge violation in the internal waters, territorial sea or exclusive economic zone of another State unless requested by that State, the flag State, or a State damaged or threatened by the discharge violation, or unless the violation has caused or is likely to cause pollution in the internal waters, territorial sea or exclusive economic zone of the State instituting the proceedings.
- 3. When a vessel is voluntarily within a port or at an offshore terminal of a State, that State shall, as far as practicable comply with requests from any State for investigation of a discharge violation referred to in paragraph 1, believed to have occurred in, caused, or threatened damage to the internal waters, territorial sea or exclusive economic zone of the requesting State. It shall likewise, as far as practicable, comply with requests from the flag State for investigation of such a violation, irrespective of where the violation occurred.
- 4. The records of the investigation carried out by a port State pursuant to this article shall be transmitted upon request to the flag State or to the coastal State. Any proceedings instituted by the port State on the basis of such an investigation may, subject to section 7, be suspended at the request of the coastal State when the violation has occurred within its internal waters, territorial sea or exclusive economic zone. The evidence and records of the case, together with any bond or other financial security posted with the authorities of the port State, shall in that event be transmitted to the coastal State. Such transmittal shall preclude the continuation of proceedings in the port State.

# Article 219. Measures relating to seaworthiness of vessels to avoid pollution

Subject to section 7, States which, upon request or on their own initiative, have ascertained that a vessel within one of their ports or at one of their off-shore terminals is in violation of applicable international rules and standards relating to seaworthiness of vessels and thereby threatens damage to the marine environment shall, as far as practicable, take administrative measures to prevent the vessel from sailing. Such States may permit the vessel to proceed only to the nearest appropriate repair yard and, upon removal of the causes of the violation, shall permit the vessel to continue immediately.

# Article 220. Enforcement by coastal States

- 1. When a vessel is voluntarily within a port or at an off-shore terminal of a State, that State may, subject to section 7, institute proceedings in respect of any violation of its laws and regulations adopted in accordance with this Convention or applicable international rules and standards for the prevention, reduction and control of pollution from vessels when the violation has occurred within the territorial sea or the exclusive economic zone of that State.
- 2. Where there are clear grounds for believing that a vessel navigating in the territorial sea of a State has, during its passage therein, violated laws and regulations of that State adopted in accordance with this Convention or applicable international rules and standards for the prevention, reduction and control of pollution from vessels, that State, without prejudice to the application of the relevant provisions of part 11, section 3, may undertake physical inspection of the vessel relating to the violation and may, where the evidence so warrants, institute proceedings, including detention of the vessel. in accordance with its laws, subject to the provisions of section 7.
- 3. Where there are clear grounds for believing that a vessel navigating in the exclusive economic zone or the territorial sea of a State has, in the exclusive economic zone, committed a violation of applicable international rules and standards for the prevention, reduction and control of pollution from vessels or laws and regulations of that State conforming and giving effect to such rules and standards, that State may require the vessel to give information regarding its identity and port of registry, its last and its next port of call and other

relevant information required to establish whether a violation has occurred.

- 4. States shall adopt laws and regulations and take other measures so that vessels flying their flag comply with requests for information pursuant to paragraph 3.
- 5. Where there are clear grounds for believing that a vessel navigating in the exclusive economic zone or the territorial sea of a State has, in the exclusive economic zone, committed a violation referred to in paragraph 3 resulting in a substantial discharge causing or threatening significant pollution of the marine environment, that State may undertake physical inspection of the vessel for matters relating to the violation if the vessel has refused to give information or if the information supplied by the vessel is manifestly at variance with the evident factual situation and if the circumstances of the case justify such inspection.
- 6. Where there is clear objective evidence that a vessel navigating in the exclusive economic zone or the territorial sea of a State has, in the exclusive economic zone, committed a violation referred to in paragraph 3 resulting in a discharge causing major damage or threat of major damage to the coastline or related interests of the coastal State, or to any resources of its territorial sea or exclusive economic zone, that State may, subject to section 7, provided that the evidence so warrants, institute proceedings, including detention of the vessel, in accordance with its laws.
- 7. Notwithstanding the provisions of paragraph 6, whenever appropriate procedures have been established, either through the competent international organization or as otherwise agreed, whereby compliance with requirements for bonding or other appropriate financial security has been assured, the coastal State if bound by such procedures shall allow the vessel to proceed.
- 8. The provisions of paragraphs 3, 4, 5, 6 and 7 also apply in respect of national laws and regulations adopted pursuant to article 211, paragraph 6.
- 254. Article 11 of the 1974 Convention for the Prevention of Marine Pollution from Land-based Sources provides for a permanent monitoring system operating through the individual or joint efforts of member States:

#### Article 11

The Contracting Parties agree to set up progressively and to operate within the area covered by the present Convention a permanent monitoring system allowing:

the earliest possible assessment of the existing level of marine pollution;

the assessment of the effectiveness of measures for the reduction of marine pollution from land-based sources taken under the terms of the present Convention.

For this purpose the Contracting Parties shall lay down the ways and means of pursuing individually or jointly systematic and *ad hoc* monitoring programmes. These programmes shall take into account the deployment of research vessels and other facilities in the monitoring area.

The programmes shall take into account similar programmes pursued in accordance with Conventions already in force and by the appropriate international organizations and agencies.

Article 11 provides for a flexible monitoring system. Parties to the Convention can set up individual or joint monitoring systems, on a continuous or on an ad hoc basis. These systems also require the co-operation of appropriate international organizations and agencies. In addition, the Convention establishes a Commission with overall supervision of the implementation of the Convention as well as of monitoring. Articles 15, 16 and 17 of the Convention read:

#### Article 15

A Commission composed of representatives of each of the Contracting Parties is hereby established. The Commission shall meet at regular intervals and at any time when due to special circumstances it is so decided in accordance with its rules of procedure.

#### Article 16

It shall be the duty of the Commission:

- (a) to exercise overall supervision over the implementation of the present Convention;
- (b) to review generally the condition of the seas within the area to which the present Convention applies, the effectiveness of the control measures being adopted and the need for any additional or different measures:
- (c) to fix, if necessary, on the proposal of the Contracting Party or Parties bordering on the same watercourse and following a standard procedure, the limit to which the maritime area shall extend in that watercourse:
- (d) to draw up, in accordance with article 4 of the present Convention, programmes and measures for the elimination or reduction of pollution from land-based sources;
- (e) to make recommendations in accordance with the provisions of article 9:
- (f) to receive and review information and distribute it to the Contracting Parties in accordance with the provisions of articles 11, 12 and 17 of the present Convention:
- (g) to make, in accordance with article 18, recommendations regarding any amendment to the lists of substances included in annex A to the present Convention;
- (h) to discharge such other functions, as may be appropriate, under the terms of the present Convention.

#### Article 17

The Contracting Parties, in accordance with a standard procedure, shall transmit to the Commission:

- (a) the results of monitoring pursuant to article 11;
- (b) the most detailed information available on the substances listed in the annexes to the present Convention and liable to find their way into the maritime area.

The Contracting Parties shall endeavour to improve progressively techniques for gathering such information which can contribute to the revision of the pollution reduction programmes adopted in accordance with article 4 of the present Convention.

255. Article 12 of the Convention requires the contracting States to ensure compliance with the Convention and to take appropriate domestic measures to prevent and punish conduct in contravention of its provisions; these measures may be of a legislative or administrative nature:

# Article 12

- 1. Each Contracting Party undertakes to ensure compliance with the provisions of this Convention and to take in its territory appropriate measures to prevent and punish conduct in contravention of the provisions of the present Convention.
- 2. The Contracting Parties shall inform the Commission of the legislative and administrative measures they have taken to implement the provisions of the preceding paragraph.
- 256. Article 13 of the Convention may also be interpreted as providing for a monitoring system. Under this article, the contracting parties must assist one another in preventing incidents that may result in pollution from land-based sources, reducing and eliminating the impact of such incidents and exchanging information to that end:

#### Article 13

The Contracting Parties undertake to assist one another as appropriate to prevent incidents which may result in pollution from land-based sources, to minimize and eliminate the consequences of such incidents, and to exchange information to that end.

257. The 1972 Convention on the Prevention of Marine Pollution by Dumping of Wastes and other matter requires the contracting parties to designate authorities with the competence to monitor the applica-

tion of the Convention. Article VI of the Convention reads:

#### Article VI

- 1. Each Contracting Party shall designate an appropriate authority or authorities to:
- (d) monitor individually, or in collaboration with other Parties and competent international organizations, the condition of the seas for the purposes of this Convention.
- 258. Article 3 of the 1960 Convention for the Protection of Lake Constance against Pollution also establishes a supervisory commission, whose functions are set out in article 4 (see para. 74 above).
- 259. A Tripartite Standing Committee was established by Belgium, France and Luxembourg under the 1950 Protocol to study and monitor problems raised by the installation in the vicinity of the frontier of explosive materials for civil use as well as problems of water pollution. One of the functions of the joint technical sub-committee established under the Protocol is to define the pollution factors, collect any appropriate technical opinions and assess each State's share of responsibility for the pollution (see para. 173 above).
- 260. Similarly, by an exchange of notes of 22 October 1975,<sup>200</sup> the Governments of France, the Federal Republic of Germany and Switzerland decided to create an intergovernmental commission to deal with frontier problems, and to examine environmental problems.
- 261. Intergovernmental commissions have also been established to study the sources of and preventive measures against pollution in the Moselle and the Rhine. The 1961 Protocol between France, the Federal Republic of Germany and Luxembourg concerning the establishment of an International Commission to protect the Moselle against pollution provides *inter alia* as follows:

# Article 2

The Commission established by virtue of article 1 is intended to ensure co-operation between the competent agencies of the three signatory Governments with a view to protecting the waters of the Moselle against pollution.

To this end, the Commission may:

- (a) prepare, commission and avail itself of the results of all inquiries necessary for determining the nature, extent and origin of the pollution;
- (b) propose to the signatory Governments suitable measures for protecting the Moselle against pollution.

The Commission shall also concern itself with all other matters which the signatory Governments refer to it by common agreement.

262. Similar language is used in article 2 of the 1963 Agreement concluded between Switzerland, France, the Federal Republic of Germany, Luxembourg and the Netherlands on the International Commission for the Protection of the Rhine against Pollution:

# Article 2

- 1. The Commission shall:
- (a) prepare, commission and avail itself of the results of all investigations necessary to determine the nature, extent and origin of the pollution of the Rhine;

<sup>&</sup>lt;sup>200</sup> Exchange of letters of 22 October 1975 between France, the Federal Republic of Germany and Switzerland constituting an agreement concerning the establishment of an Intergovernmental Commission on contiguity problems in frontier regions.

- (b) propose to the signatory Governments appropriate measures to protect the Rhine against pollution;
- (c) prepare the bases for possible future arrangements between the signatory Governments concerning the protection of the waters of the Rhine.
- 2. The Commission's competence shall also extend to all other matters which the signatory Governments refer to it by common consent.

In 1976, the European Economic Community became a party to this Agreement.<sup>201</sup>

263. Under the 1976 Convention for the Protection of the Rhine against Chemical Pollution, the aforementioned International Commission is required to monitor the chemical substances discharged into the Rhine. The Commission has to modify the list of prohibited substances and make recommendations in the light of new scientific and other developments. The following are the relevant provisions of the Convention:

#### Article 2

2. Pursuant to annex III, paragraph 2, the Governments shall communicate to the International Commission for the Protection of the Rhine against Pollution (hereinafter referred to as the "International Commission") the contents of their inventories, which shall be regularly updated at intervals not exceeding three years.

#### Article 5

- 1. The International Commission shall propose the limit values provided for in article 3, paragraph 2, and if necessary their application to discharges into sewers. These limit values shall be laid down in conformity with the procedure provided for in article 14. Upon adoption, they shall be included in annex IV.
  - 2. These limit values are fixed in terms of:
- (a) the maximum permissible concentration of a substance in discharges and,
- (b) where appropriate, the maximum permissible quantity of such a substance expressed as a unit of weight of the pollutant per unit of the characteristic element of the polluting activity (e.g. unit of weight per unit of raw material or product unit).

Where appropriate, the limit values applicable to industrial effluent shall be laid down individually by sector and by type of product.

The limit values applicable to annex I substances shall be laid down mainly on the basis of:

- toxicity,
- persistence,
- bioaccumulation,

taking into account the best available technical means.

- 3. The International Commission shall propose to the Contracting Parties the time limits referred to in article 3, paragraph 3, making due allowance for the specific characteristics of the industrial sectors involved and, as appropriate, the types of product. These time limits shall be determined in accordance with the procedure laid down in article 14.
- 4. The International Commission shall use the results obtained at international measuring points to determine the extent to which the level of annex I substances in the Rhine varies following the application of the above provisions.
- 5. As regards the quality of Rhine water, the International Commission may if necessary propose other measures for reducing the pollution of the Rhine taking into account, *inter alia*, the toxicity, persistence and bioaccumulation of the substance under consideration. These proposals shall be adopted in accordance with the procedure laid down in article 14.

#### Article 12

- 1. The Contracting Parties shall regularly inform the International Commission of the experience gained in the course of implementing this Convention.
- 2. The International Commission shall also make recommendations, as appropriate, designed gradually to improve the implementation of this Convention.

#### Article 13

The International Commission shall work out draft recommendations for achieving comparable results by the use of appropriate measuring and analysis methods.

#### Article 14

- 1. Annexes I to IV inclusive, which shall form an integral part of this Convention, may be amended or added to for the purposes of adapting them to technical or scientific advances or of more effectively combating the chemical pollution of the Rhine.
- 2. To this end, the International Commission shall recommend the admendments or additions which it considers appropriate.
- 3. The amended or supplemented texts shall enter into force following unanimous acceptance by the Contracting Parties.
- 264. Some multilateral agreements aim only at monitoring certain activities and at co-operation in minimizing injuries without reference to issues of liability. For example, the 1969 Agreement for Co-operation in Dealing with Pollution of the North Sea by Oil is aimed at organizing immediate consultation and rapid co-operation among the signatories to combat oil pollution in the North Sea area. The relevant provisions of the Agreement are the following:

#### Article 4

Contracting Parties undertake to inform the other Contracting Parties about

- (a) their national organization for dealing with oil pollution;
- () the competent authority responsible for receiving reports of oil pollution and for dealing with questions concerning measures of mutual assistance between Contracting Parties;
- (c) new ways in which oil pollution may be avoided and about new effective measures to deal with oil pollution.

#### Article 6

- 1. For the sole purposes of this Agreement the North Sea area is divided into the zones described in the annex to this Agreement.
- 2. The Contracting Party within whose zone a situation of the kind is described in article 1 occurs, shall make the necessary assessments of the nature and extent of any casualty or, as the case may be, of the type and approximate quantity of oil floating on the sea, and the direction and speed of movement of the oil.
- 3. The Contracting Party concerned shall immediately inform all the other Contracting Parties through their competent authorities of its assessments and of any action which it has taken to deal with the floating oil and shall keep the oil under observation as long as it is drifting in its zone.
- 4. The obligations of the Contracting Parties under the provisions of this article with respect to the zones of joint responsibility shall be the subject of special technical arrangements to be concluded between the Parties concerned. These arrangements shall be communicated to the other Contracting Parties.

This Agreement, whose main purpose is the institution of a co-operative monitoring system among signatory States, is not concerned with liability or compensation.

265. Similarly, article III of the 1966 International Convention for the Conservation of Atlantic Tunas establishes a commission whose functions include monitoring the conditions of tuna fishery resources and study and appraisal of information concerning methods to ensure the maintenance of the population of tuna, etc.

<sup>&</sup>lt;sup>201</sup> See the Additional Agreement of 3 December 1976 to the Agreement of 29 April 1963 concerning the International Commission for the Protection of the Rhine against Pollution.

Article IV enumerates the Commission's functions (see para. 72 above).

266. The 1985 Vienna Convention for the Protection of the Ozone Layer also provides for a continuous and systematic research and monitoring mechanism, either through co-operation among States or through competent international organizations. The functions of this mechanism include the detection and study of phenomena that may affect the ozone layer as well as the study of the effects of the modification of that layer on human health. Article 3 of the Convention reads:

#### Article 3. Research and systematic observations

- 1. The Contracting Parties undertake, as appropriate, to initiate and co-operate in, directly or through competent international bodies, the conduct of research and scientific assessments on:
- (a) The physical and chemical processes that may affect the ozone layer:
- (b) The human health and other biological effects deriving from any modifications of the ozone layer, particularly those resulting from changes in ultra-violet solar radiation having biological effects;
- (c) Climatic effects deriving from any modifications of the ozone layer;
- (d) Effects deriving from any modifications of the ozone layer and any consequent change in UV-B radiation on natural and synthetic materials useful to mankind;
- (e) Substances, practices, processes and activities that may affect the ozone layer, and their cumulative effects;
- (f) Related socio-economic matters;
- as further elaborated in annex I.
- 2. The Parties undertake to promote or establish, as appropriate, directly or through competent international bodies and taking fully into account national legislation and relevant ongoing activities at both the national and international levels, joint or complementary programmes for systematic observation of the state of the ozone layer and other relevant parameters, as elaborated in annex I.
- 3. The Parties undertake to co-operate, directly or through competent international bodies, in ensuring the collection, validation and transmission of research and observational data through appropriate world data centres in a regular and timely fashion.

The Convention is concerned only with research and monitoring, not with liability and compensation.

267. Similary, the 1971 Treaty on the Prohibition of the Emplacement of Nuclear Weapons and Other Weapons of Mass Destruction on the Sea-Bed and the Ocean Floor and in the Subsoil thereof provides for a monitoring system or "verification" for compliance with treaty obligations. Such a monitoring function may be undertaken by the signatories individually or in co-operation. Nevertheless, when there are "reasonable" doubts about the fulfilment of treaty obligations, the State party which has doubts and the State party which has given rise to such doubts by its activities are under obligation to co-operate and consult with the aim of allaying them. If a State party continues to have doubts then it may, through the procedures provided under article III of the Treaty and in accordance with the provisions of the United Nations Charter, bring the matter to the attention of the Security Council, as follows:

#### Article III

1. In order to promote the objectives of and ensure compliance with the provisions of this Treaty, each State Party to the Treaty shall have the right to verify through observation the activities of other States Parties to the Treaty on the sea-bed and the ocean floor and in the subsoil thereof beyond the zone referred to in article I, provided that observation does not interfere with such activities.

- 2. If after such observation reasonable doubts remain concerning the fulfilment of the obligations assumed under the Treaty, the State Party having such doubts and the State Party that is responsible for the activities giving rise to the doubts shall consult with a view to removing the doubts. If the doubts persist, the State Party having such doubts shall notify the other States Parties, and the Parties concerned shall co-operate on such further procedures for verification as may be agreed, including appropriate inspection of objects, structures, installations or other facilities that reasonably may be expected to be of a kind described in article I. The Parties in the region of the activities, including any coastal State, and any other Party so requesting, shall be entitled to participate in such consultation and co-operation. After completion of the further procedures for verification, an appropriate report shall be circulated to other Parties by the Party that initiated such procedures.
- 3. If the State responsible for the activities giving rise to the reasonable doubts is not identifiable by observation of the object, structure, installation or other facility, the State Party having such doubts shall notify and make appropriate inquiries of States Parties in the region of the activities and of any other State Party. If it is ascertained through these inquiries that a particular State Party is responsible for the activities, that State Party shall consult and co-operate with other Parties as provided in paragraph 2 of this article. If the identity of the State responsible for the activities cannot be ascertained through these inquiries, then further verification procedures, including inspection, may be undertaken by the inquiring State Party, which shall invite the participation of the Parties in the region of the activities, including any coastal State, and of any other Party desiring to co-operate.
- 4. If consultation and co-operation pursuant to paragraphs 2 and 3 of this article have not removed the doubts concerning the activities and there remains a serious question concerning fulfilment of the obligations assumed under this Treaty, a State Party may, in accordance with the provisions of the Charter of the United Nations, refer the matter to the Security Council, which may take action in accordance with the Charter.
- 5. Verification pursuant to this Article may be undertaken by any State Party using its own means, or with the full or partial assistance of any other State Party, or through appropriate international procedures within the framework of the United Nations and in accordance with its Charter.
- 6. Verification activities pursuant to this Treaty shall not interfere with activities of other States Parties and shall be conducted with due regard for rights recognized under international law, including the freedoms of the high seas and the rights of coastal States with respect to the exploration and exploitation of their continental shelves.
- 268. Similarly, the 1976 Convention on the Prohibition of Military or any other Hostile Use of Environmental Modification Techniques provides for a monitoring system. Under article V, consultation and co-operation among States may also take place through appropriate international organizations. Furthermore, the depositary, the United Nations, must within one month of the receipt of a request to that effect by any signatory State convene a consultative committee of experts. Any signatory may appoint an expert to the committee. The committee is a fact-finding body, which submits its findings to the depositary, which in turn has them circulated to the signatories. Article V of the Convention reads:

#### Article V

- 1. The States Parties to this Convention undertake to consult one another and to co-operate in solving any problems which may arise in relation to the objectives of, or in the application of the provisions of, the Convention. Consultation and co-operation pursuant to this article may also be undertaken through appropriate international procedures within the framework of the United Nations and in accordance with its Charter. These international procedures may include the services of appropriate international organizations, as well as of a consultative committee of experts as provided for in paragraph 2 of this article.
- 2. For the purposes set forth in paragraph 1 of this article, the Depositary shall, within one month of the receipt of a request from

any State Party, convene a consultative committee of experts. Any State Party may appoint an expert to this Committee, whose functions and rules of procedures are set out in the annex, which constitutes an integral part of this Convention. The committee shall transmit to the Depositary a summary of its findings of fact, incorporating all views and information presented to the committee during its proceedings. The Depositary shall distribute the summary to all States Parties.

- 3. Any State Party to this Convention which has reasons to believe that any other State Party is acting in breach of obligations deriving from the provisions of the Convention may lodge a complaint with the Security Council of the United Nations. Such a complaint should include all relevant information as well as all possible evidence supporting its validity.
- 4. Each State Party to this Convention undertakes to co-operate in carrying out any investigation which the Security Council may initiate, in accordance with the provisions of the Charter of the United Nations, on the basis of the complaint received by the Council. The Security Council shall inform the States Parties of the results of the investigation.
- 5. Each State Party to this Convention undertakes to provide or support assistance, in accordance with the provisions of the Charter of the United Nations, to any Party to the Convention which so requests, if the Security Council decides that such Party has been harmed or is likely to be harmed as a result of violation of the Convention.
- 269. The function of the Consultative Committee of Experts and its rules of procedure are defined in the annex to the Convention, which reads:

# Annex to the Convention

#### Consultative Committee of Experts

- 1. The Consultative Committee of Experts shall undertake to make appropriate findings of fact and provide expert views relevant to any problem raised pursuant to article V, paragraph 1, of this Convention by the State Party requesting the convening of the Committee.
- 2. The work of the Consultative Committee of Experts shall be organized in such a way as to permit it to perform the functions set forth in paragraph 1 of this annex. The Committee shall decide procedural questions relative to the organization of its work, where possible by consensus, but otherwise by a majority of those present and voting. There shall be no voting on matters of substance.
- 3. The Depositary or his representative shall serve as the Chairman of the Committee.
- 4. Each expert may be assisted at meetings by one or more advisers.
- 5. Each expert shall have the right, through the Chairman, to request from States, and from international organizations, such information and assistance as the expert considers desirable for the accomplishment of the Committee's work.

Furthermore, each State party to the Convention undertakes to take measures within its own constitutional framework to prevent any activity within its jurisdiction or under its control in violation of the Convention. Article IV of the Convention reads:

#### Article IV

Each State Party to this Convention undertakes to take any measures it considers necessary in accordance with its constitutional processes to prohibit and prevent any activity in violation of the provisions of the Convention anywhere under its jurisdiction or control.

270. Other conventions, on the other hand, whose basic aim is prevention or reduction of injuries through monitoring and co-operation, include provisions on liability and compensation. For examples, articles 11 and 13 of the 1983 Convention for the Protection and Development of the Marine Environment of the Wider Caribbean Region provide for co-operation among contracting States in monitoring activities in order to prevent injuries to the area covered by the Convention. They also provide that States shall co-operate in notification of emergencies and in provision of assistance during such situations:

#### Article 11. Co-operation in cases of emergency

- 1. The Contracting Parties shall co-operate in taking all necessary measures to respond to pollution emergencies in the Convention area, whatever the cause of such emergencies, and to control, reduce or eliminate pollution or the threat of pollution resulting therefrom. To this end, the Contracting Parties shall, individually and jointly, develop and promote contingency plans for responding to incidents involving pollution or the threat thereof in the Convention area.
- 2. When a Contracting Party becomes aware of cases in which the Convention area is in imminent danger of being polluted or has been polluted, it shall immediately notify other States likely to be affected by such pollution, as well as the competent international organizations. Furthermore, it shall inform, as soon as feasible, such other States and competent international organizations of measures it has taken to minimize or reduce pollution or the threat thereof.

#### Article 13. Scientific and technical co-operation

- 1. The Contracting Parties undertake to co-operate, directly and, when appropriate, through the competent international and regional organizations, in scientific research, monitoring and the exchange of data and other scientific information relating to the purposes of this Convention.
- 2. To this end, the Contracting Parties undertake to develop and co-ordinate their research and monitoring programmes relating to the Convention area and to ensure, in co-operation with the competent international and regional organizations, the necessary links between their research centres and institutes with a view to producing compatible results. With the aim of further protecting the Convention area, the Contracting Parties shall endeavour to participate in international arrangements for pollution research and monitoring.
- 3. The Contracting Parties undertake to co-operate, directly and, when appropriate, through the competent international and regional organizations, in the provision to other Contracting Parties of technical and other assistance in fields relating to pollution and sound environmental management of the Convention area, taking into account the special needs of the smaller island developing countries and territories.

It should be noted that the Convention requires, in this final paragraph, that special attention be paid to the needs of smaller island developing countries and territories in the region.

# (b) Bilateral agreements

- 271. Bilateral agreements relating to the use of frontier resources usually provide for the monitoring of activities, particularly if such activities are conducted in the vicinity of the frontier and concern resources shared by the States. Some bilateral agreements also provide for the monitoring of activities unrelated to frontier resources, but in which both States parties have an interest. Such monitoring may be affected either jointly by the contracting States or through a commission.
- 272. Under the 1931 General Convention between Romania and Yugoslavia, <sup>202</sup> the monitoring of the operation of the hydraulic system of the frontier waters may apparently be performed by *either State*. Articles 9 and 10 of chapter I of the Convention provide that the parties agree to communicate to one another any laws or regulations in force or that may be adopted regarding the hydraulic system, the forestry system and fisheries, and that each party, at the request of the other, must supply certain data or information concerning the hydraulic system:

### Article 9

The two States shall communicate to one another any laws or other regulations which are now in force or which may be promulgated in

<sup>202</sup> See footnote 60 above.

the future, regarding the hydraulic system, the forestry system and fisheries; they shall also communicate to one another their official periodicals concerning hydrometrical, hydrological, meteorological and geological data collected in their respective territories which may be of assistance in the study of the hydraulic system.

#### Article 10

When either State requests, for purposes of study, certain data or information regarding the hydraulic system of the other State, the latter undertakes, in the absence of any legitimate objection, to supply them.

- 273. By virtue of an Agreement concluded in 1983, the United States of America and Mexico<sup>203</sup> undertook to ensure the monitoring of the environment in a coordinated manner (see art. 6, at para. 73 above). In order to monitor polluting activities, the parties are required to consult one another on the measurement and analysis of polluting elements in the border area. Article 15, paragraph 2, of the Agreement provides:
- 2. In order to undertake the monitoring of polluting activities in the border area, the parties shall undertake consultations relating to the measurement and analysis of polluting elements in the border area.
- 274. Under the 1954 Agreement between Yugoslavia and Austria, <sup>204</sup> the *local authorities* in those countries are required to advise each other, as quickly as possible, of any danger from high water or ice or any other impending danger which comes into their notice in connection with the river Mura. Article 7 of the Agreement reads:

#### Article 7

The local authorities of the Contracting States shall advise each other, by the quickest possible means, of any danger from high water or ice and of any other impending danger which comes to their notice in connection with the Mura. The same shall apply to the frontier waters of the Mura where such dangers come to the notice of the local authorities.

275. In the 1974 Agreement between the Federal Republic of Germany and Norway relating to the transmission of petroleum by pipeline, <sup>205</sup> the monitoring of the operation is *divided between the States concerned*. The search for and removal of any mines, or other explosive devices, lying on the sea-bed route of the pipeline, or on the continental shelf or in the territorial sea of the Federal Republic of Germany, is the responsibility of the Government of that country. Paragraph 2 of article 7 of the Agreement reads:

# Article 7

- 2. The Government of the Federal Republic of Germany is prepared, to the extent that available technical facilities are adequate and other conditions so permit, to search for and remove any mines, or other explosive devices, lying on or projecting upwards from the seabed on the pipeline route in the continental shelf or territorial sea of the Federal Republic of Germany.
- 276. The final safety clearance of the pipeline is the responsibility of the Norwegian Government after consultation with the Government of the Federal Republic

<sup>203</sup> See footnote 16 above.

of Germany. Paragraph 2 of article 8 of the Agreement reads:

#### Article 8

- 2. The final safety clearance of the pipeline shall be given by the Norwegian Government after consultation with the Government of the Federal Republic of Germany on the basis of existing German and Norwegian law and this Agreement.
- 277. Under article 9 of the Agreement, the competent supervisory authorities of each contracting party shall have the right to inspect the pipeline facilities, including those situated within the territorial jurisdiction of the other party:

#### Article 9

- 1. To the extent required for the monitoring of safety regulations relating to the construction, laying and operation of the pipeline, the competent supervisory authorities of each Contracting Party shall have the right to inspect the pipeline facilities, including those situated in the continental shelf or national territory of the other State, and to obtain information for that purpose.
- 2. The details of the procedure shall be agreed upon by the competent supervisory authorities of the two Contracting Parties.
- 278. Finally, the issuance of new licences, as well as the modification of existing licences relating to the operation of the pipeline, are to be under the joint control of the two States. This is a case of joint control and monitoring of the activities of private entities affecting States and expressed by the extension or cancellation of the companies' licences. Under this article, both Norway and the Federal Republic of Germany have to agree upon the substantive content of licences as well as upon any alteration in or issuance of new licences. In case of serious or repeated violations of the terms of a licence, the Government concerned may revoke it, but only after prior consultation with the other Government. Control over the substantive content of licences is of course an effective means of controlling the activities of the operating entity. Article 10 of the Agreement reads:

#### Article 10

- 1. The substantive content of licences, including their period of validity, shall be agreed upon by the two Governments on the basis of the law in force and this Agreement.
- 2. A copy of the licence or licences issued by one Government shall be made available to the other Government.
- 3. No licences shall be altered or assigned to a new licensee by the Government concerned without prior consultation with the other Government.
- 4. In the event of serious or repeated violations of the terms of a licence, the Government concerned may revoke such licence but not without prior consultation with the other Government.

It should be noted that this Agreement does not relate to the utilization of a shared frontier resource but rather to an operation entailing economic benefits for both States.

279. The co-operation in management and monitoring provided for in some bilateral treaties is also aimed at averting danger. Article 19 of the 1950 Treaty between Hungary and the USSR,<sup>206</sup> in addition to requiring the gathering and exchange of information concerning the condition and the level of their frontier rivers in order to avert danger from floods, provides that the parties shall agree upon a regular system of signals to be used during

<sup>&</sup>lt;sup>204</sup> Agreement of 16 December 1954 between Yugoslavia and Austria concerning water economy questions in respect of the frontier sector of the Mura and the frontier waters of the Mura (Mura Agreement).

<sup>205</sup> See foonote 14 above.

<sup>206</sup> See foonote 40 above.

periods of high water or drifting ice. This Agreement does not itself establish a monitoring system, but urges the parties to agree upon such a system; information must be exchanged in a spirit of co-operation, and delay or failure in communication does not constitute grounds for claims to compensation for damage caused by flooding or drifting ice (see para. 81 above).

280. A number of other agreements provide for a system of monitoring involving advance notification of danger and the establishment of a warning system. Thus paragraph 3 of article 17 of the 1949 Agreement between Norway and the USSR<sup>207</sup> provides that, in case of forest fire, the contracting party in whose territory the fire breaks out shall immediately notify the other party so that the necessary measures may be taken to halt the fire at the frontier:

#### Article 17

- 3. If a forest fire threatens to spread across the frontier, the Contracting Party in whose territory the danger arises shall forthwith notify the other Contracting Party so that the necessary measures may be taken to stop the fire at the frontier.
- 281. An identical provision is included in the 1948 Agreement between the Soviet Union and Finland,<sup>208</sup> paragraph 3 of article 17 of which reads:

#### Article 17

- 3. If a forest fire threatens to spread across the frontier, the Contracting Party in whose territory the threat arises shall forthwith notify the other Contracting Party so that necessary measures may be taken to localize the fire.
- 282. Under article 11 of their 1956 Treaty, <sup>209</sup> Hungary and Austria also agree to notify each other as quickly as possible of any danger of flood, ice or other danger arising in connection with frontier waters which comes to their attention:

# Article 11. Warning service

The authorities of the Contracting Parties, particularly the hydrographic service and local authorities, shall notify each other as quickly as possible of any danger of flood or ice or other danger arising in connection with frontier waters which comes to their attention.

- 283. By an exchange of letters, France and the Soviet Union<sup>210</sup> agreed to notify each other immediately of any accidental occurrence or any other unexplained incident that could lead to the explosion of one of their nuclear weapons that might have harmful effects on the other party. Provisions 1 and 2 of the letter addressed by France to the Soviet Union read:
- 1. Each Party undertakes to maintain and to make such improvements as it deems necessary to the organizational and technical measures taken by it to prevent the accidental or unauthorized use of any nuclear weapon under its control.
- 2. The two Parties undertake to notify each other immediately of any accident or any other unexplained incident which may appear to involve the possibility of the explosion of one of their nuclear weapons

and may be interpreted as liable to have harmful consequences for the other Party.

284. The 1952 Agreement between Poland and the German Democratic Republic concerning navigation in frontier waters<sup>211</sup> also provides, in article 20, for mutual assistance in case of accident during blasting operations. The parties undertake to come to each other's assistance, subject to reimbursement of the expenses entailed in the provision of such assistance:

#### Article 20

In the event of damage or accident during blasting operations, each Party undertakes to come to the other's assistance, subject to reimbursement of the expenses entailed in the provision of such assistance.

This mutual assistance is only a part of an entire system of co-operation established between the two States in the utilization of their joint waters. Under this system, Poland and the German Democratic Republic divide the monitoring and management of their joint waters between them. Accordingly, as provided in article 21, each contracting party is required to take precautions against flooding in its own territory and, where necessary, to inform the other party of any burst in a dike; if such a burst occurs, the two parties must immediately concert their efforts to repair the damage.

285. In at least one bilateral agreement, measures for the prevention of danger may be taken by one State alone, while the other State reimburses it for the cost of such preventive measures. Thus the 1961 Treaty between Canada and the United States of America concerning the Columbia River Basin<sup>212</sup> sets out, in paragraphs 2 and 3 of article IV, the measures that Canada must take to combat flooding. Article VI of the Treaty specifies the amount of the reimbursement the United States agrees to make to Canada for its flood control measures:

# Article VI. Payment for flood control

- 1. For the flood control provided by Canada under article IV (2) (a) the United States of America shall pay Canada in United States funds:
- (a) 1,200,000 dollars upon the commencement of operation of the storage referred to in subparagraph (a) (i) thereof,
- (b) 52,100,000 dollars upon the commencement of operation of the storage referred to in subparagraph (a) (ii) thereof, and
- (c) 11,100,000 dollars upon the commencement of operation of the storage referred to in subparagraph (a) (iii) thereof.
- 2. If full operation of any storage is not commenced within the time specified in article IV, the amount set forth in paragraph 1 of this article with respect to that storage shall be reduced as follows:
- (a) under paragraph 1 (a), 4,500 dollars for each month beyond the required time,
- (b) under paragraph 1 (b), 192,100 dollars for each month beyond the required time, and
- (c) under paragraph 1 (c), 40,800 dollars for each month beyond the required time.
- 3. For the flood control provided by Canada under article IV (2) (b) the United States of America shall pay Canada in United States funds in respect only of each of the first four flood periods for which a call is made 1,875,000 dollars and shall deliver to Canada in respect of

<sup>267</sup> See foonote 17 above.

<sup>&</sup>lt;sup>208</sup> Agreement of 9 December 1948 between the USSR and Finland concerning the régime of the Soviet-Finnish frontier.

<sup>209</sup> See footnote 98 above.

<sup>&</sup>lt;sup>210</sup> Exchange of letters of 16 July 1976 between France and the USSR constituting an agreement concerning the prevention of the accidental or unauthorized use of nuclear weapons.

<sup>&</sup>lt;sup>211</sup> Agreement of 6 February 1952 between Poland and the German Democratic Republic concerning navigation in frontier waters and the use and maintenance of frontier waters.

<sup>&</sup>lt;sup>212</sup> Treaty of 17 January 1961 between Canada and the United States of America relating to co-operative development of the water resources of the Columbia River Basin.

each and every call made, electric power equal to the hydroelectric power lost by Canada as a result of operating the storage to meet the flood control need for which the call was made, delivery to be made when the loss of hydroelectric power occurs.

- 4. For each flood period for which flodd control is provided by Canada under article IV (3) the United States of America shall pay Canada in United States funds:
- (a) the operating cost incurred by Canada in providing the flood control, and
- (b) compensation for the economic loss to Canada arising directly from Canada foregoing alternative uses of the storage used to provide the flood control.
- 5. Canada may elect to receive in electric power, the whole or any portion of the compensation under paragraph 4 (b) representing loss of hydroelectric power to Canada.

In case the causes of transboundary injuries should be disputed, neighbouring States have concluded agreements for co-operation, mutual consultation and monitoring to determine the source of the injuries. The most recent agreement of this type is the 1983 Agreement between Canada and the United States of America,213 the object of which is to determine the cause of the acid rain which has produced injuries in both Canada and the United States. Under the Agreement, the two States are to monitor the flow of pollution from industrial plants in Ohio and Ontario, regarded as the prime sources of the pollution that has damaged forests on both sides of the border and destroyed fish and plant life in hundreds of lakes in the Adirondacks, in New York State, as well as in New England and eastern Canada. A scientific experiment known as Captex<sup>214</sup> was conducted under the Agreement and was expected to show whether the atmospheric pollutants were carried over long distances by wind currents and, if so, how they were carried.

# (c) Judicial decisions and State practice other than agreements

287. Where States engage in potentially injurious activities, they may take preventive measures by adopting appropriate legislative or other regulatory measures.

288. Prior to and during the Eniwetok Atoll nuclear tests series, the United States unilaterally took preventive measures to minimize injuries and to monitor the radioactive fallout from the activity. In addition to the vast programme of scientific, technical and public health measures established in the testing area of the Eniwetok Atoll and adjacent areas (forecasts of radioactive fallout, establishment of a danger area, checking for radiation, etc.) and analysis of radioactivity in the ocean and in marine organisms as described in detail in paragraphs 84 to 89 of the present study,

attention must be drawn to the study of fallout recorded in the United States and other parts of the world.

289. In the United States, the Public Health Service monitoring stations were instructed to take daily radiation readings and to filter samples for analysis. They were also required to report data to the health officers of the States or territories in which the stations were located. These stations were manned by trained technicians from State health departments, local universities and scientific institutions. Another network in the United States was required to gather data to be used in a long-range scientific study of the behaviour of radioactive materials in the environment and their effect on man.<sup>215</sup>

290. Samples of dust, soil, milk, cheese and animal bones were collected from around the world and analysed. That programme was part of a study of the world-wide distribution and uptake of radioactive fission products. Monitoring was carried out throughout the world in order to make an early assessment of any potential injury to human life.<sup>216</sup>

291. The United Kingdom, for its part, in connection with its hydrogen bomb tests in the *Christmas Islands* in 1957, established a monitoring system similar to that established by the United States at Eniwetok Atoll. Details of the system are given in paragraphs 90 and 91 of this report.

292. The United States also undertook unilateral monitoring of Mexican construction of a highway across the Smugglers and Goat Canyons. Although the United States had discussed the plans with Mexico prior to the construction of the highway, it nevertheless considered it appropriate to monitor the result of the work itself and to assess its impact upon the United States. The United States Government asked a group of engineers to analyse the blueprints in order to ascertain whether sufficient safety measures had been taken to prevent any injuries to the United States. In a letter to the Mexican Foreign Minister, the United States Ambassador to Mexico wrote:

As a result of the technical discussions, several modifications of the original plans were understood to have been agreed upon. . . . [These modifications were listed.]

It was believed by the United States engineers that these modifications would barely meet the minimum standards for such embankments.

When construction was resumed, culverts were installed at the base of the embankment at the Arroyo de las Cabras [Smugglers Canyon] but were encased in concrete only up to about two thirds of their height [instead of to a height of about four inches above the top of the culverts]. Since the culverts are now being covered with fill, it seems improbable that the State of Baja California intends to complete the encasement and erect the cutoff collars at the Arroyo de las Cabras. The remedial measures have not been started at the Arroyo de San Antonio.

In the opinion of engineers of the United States Government who are closely familiar with the recent construction, the embankment at Arroyo de San Antonio [Goat Canyon] will fail in certain circumstances of flood, and the modifications made at the Arroyo de las Cabras are not adequate to ensure its security. It too must be expected to fail in certain circumstances. Since the rainy season in that area begins as a rule in November, when considerable runoff in the arroyos must be anticipated, the matter is not only grave but urgent.

<sup>&</sup>lt;sup>213</sup> Agreement of 23 August 1983 between Canada and the United States of America to Track Air Pollution across Eastern North America (Acid Rain Research).

Canadian concern was reflected as follows:

<sup>&</sup>quot;The Canadian Government had argued in recent years that the breadth of acid rain pollution demanded urgent action, but the Reagan Administration has maintained that there is insufficient evidence to tie the death of lakes to the flow of pollutants from industrial plants." (The New York Times, 24 August 1983, p. A.3, col. 4.)

<sup>&</sup>lt;sup>214</sup> The *Captex* experiment—an acronym for Cross-Appalachian Trace Experiment—began in September 1983 and was conducted for a period of six weeks.

Whiteman, op. cit. (footnote 45 above), vol. 4, p. 591.

<sup>216</sup> Ibid., pp. 591-592.

My Government has accordingly instructed me to urge the Government of Mexico to take appropriate steps to prevent the damage to property and the injury to persons that are likely to result from the improper construction of the highway. I urge particularly that further construction at the Arroyo de las Cabras be suspended until arrangements can be made by the Government of Mexico for adoption of features essential for the security of the embankment in that canyon, and that the embankment at the Arroyo de San Antonio be opened to prevent the accumulation of flood water pending installation of similar modifications at that canyon.<sup>217</sup>

293. Besides acting unilaterally to monitor and assess the extraterritorial impact of activities, States have in the past acted jointly in the impact monitoring process. For example, the United States acted in conjunction with Mexico to impede the spread of foot-and-mouth disease. In 1947, the two States established a joint commission for the eradication of foot-and-mouth disease (see para. 164 above) and jointly financed a campaign to combat the disease, organized by a joint office established at the same date.

294. In the *Trail Smelter* case, the tribunal prescribed the establishment of a far more detailed régime for the monitoring and management of activities with harmful extraterritorial impact. Among its various detailed provisions, the régime called for placement of permanent instruments to monitor fumes from the smelter:

#### I. Instruments

A. The instruments for recording meteorological conditions shall be as follows:

(a) Wind Direction and Wind Velocity shall be indicated by any of the standard instruments used for such purposes to provide a continuous record and shall be observed and transcribed for use of the Smoke Control Office at least once every hour.

(b) Wind Turbulence shall be measured by the Bridled Cup Turbulence Indicator. This instrument consists of a light horizontal wheel around whose periphery are twenty-two equally spaced curved surfaces cut from one-eighth inch aluminium sheet and shaped to the same-sized blades or cups. This wind-sensitive wheel is attached to an aluminium sleeve rigidly screwed to one end of a three-eighth inch vertical steel shaft supported by almost frictionless bearings at the top and bottom of the instrument frame. The shaft of the wheel is bridled to prevent continuous rotation and is so constrained that its angle of rotation is directly proportional to the square of the wind velocity. One complete revolution of the anemometer shaft corresponds to a wind velocity of 36 miles per hour and, with eighteen equally spaced contact points on the commutator, one make and one break in the circuit is equivalent to a change in wind velocity of two miles per hour, recorded on a standard anemograph. . . .

The instruments noted in (a) and (b) above, shall be located at the present site near the zinc stack of the Smelter or at some other location not less favourable for such observations.

(c) Atmospheric temperature and barometric pressure shall be determined by the standard instruments in use for such meteorological observations.

B. Sulphur dioxide concentrations shall be determined by the standard recorders, which provide automatically an accurate and continuous record of such concentrations.

One recorder shall be located at Columbia Gardens, as at present installed with arrangements for the automatic transcription of its record to the Smoke Control Office at the Smelter. A second recorder shall be maintained at the present site near Northport. A third recorder shall be maintained at the present site near Waneta, which recorder may be discontinued after December 31, 1942.<sup>218</sup>

295. Records of the data thus collected were to be kept. A summary of the smelter operation covering the daily sulphur balances was to be compiled monthly and

296. The permanent régime also provided for amendment or modification of the monitoring system:

#### VI. Amendment or suspension of the régime

If at any time after December 31, 1942, either Government shall request an amendment or suspension of the régime herein prescribed and the other Government shall decline to agree to such request, there shall be appointed by each Government, within one month after the making or receipt respectively of such request, a scientist of repute; and the two scientists so appointed shall constitute a Commission for the purpose of considering and acting upon such request. If the Commission within three months after appointment fail to agree upon a decision, they shall appoint jointly a third scientist who shall be Chairman of the Commission; and thereupon the opinion of the majority, or in the absence of any majority opinion, the opinion of the Chairman shall be decisive; the opinion shall be rendered within one month after the choice of the Chairman. If the two scientists shall fail to agree upon a third scientist within the prescribed time, upon the request of either, he shall be appointed within one month from such failure by the President of the American Chemical Society, a scientific body having a membership both in the United States, Canada, Great Britain and other countries.

Any of the periods of time herein prescribed may be extended by agreement between the two Governments.

The Commission of two, or three scientists as the case may be, may take such action in compliance with or in denial of the request above referred to, either in whole or in part, as it deems appropriate for the avoidance or prevention of damage occurring in the State of Washington. The decision of the Commission shall be final, and the Governments shall take such action as may be necessary to ensure due conformity with the decision, in accordance with the provisions of article XII of the Convention.

The compensation of the scientists appointed and their reasonable expenditures shall be paid by the Government which shall have requested a decision; if both Governments shall have made a request for decision, such expenses shall be shared equally by both Governments; provided, however, that if the Commission in response to the request of the United States shall find that notwithstanding compliance with the régime in force damage has occurred through fumes in the State of Washington, then the above expenses shall be paid by the Dominion of Canada.<sup>274</sup>

The tribunal also recommended that the acting State, Canada, maintain a scientific staff for the purpose of unilateral monitoring. While the tribunal refrained from making it a part of the prescribed monitoring régime, it considered that it would be to the clear advantage of Canada if, during the interval between the date of the filing of the final report and 31 December 1942 (when the prescribed monitoring régime was to be put into effect), Canada were to continue, at its own expense, the maintenance of experimental and observational work by two scientists similar to that previously prescribed by the tribunal and which had been in operation during the trial period since 1938. The tribunal thought such continuance of investigation would pro-

copies were to be sent to the Governments of the United States and the Dominion of Canada. In addition, the structure of the stacks was regulated. Sulphur dioxide was allowed to be discharged into the atmosphere from smelting operations of zinc and lead plants at a height no lower than that of the stacks at that time. In the case of cooling of the stacks by a lengthy shutdown, gases containing sulphur dioxide were not to be emitted until the stacks had been heated to normal operating temperatures by hot gases free of sulphur dioxide, and standards of maximum hourly emissions were to be carefully adhered to.<sup>219</sup>

<sup>&</sup>lt;sup>217</sup> *Ibid.*, vol. 6, pp. 261-262.

<sup>&</sup>lt;sup>210</sup> United Nations, Reports of International Arbitral Awards, vol. III, pp. 1974-1975.

<sup>219</sup> Ibid., pp. 1975-1976.

<sup>220</sup> Ibid., p. 1978.

vide additional valuable data not only for testing purposes and for the effective operation of the régime prescribed, but also in order to make it possible, if need be to improve that régime.<sup>221</sup>

297. Unilateral monitoring was also conducted by Belgium when a resort area at the mouth of the Yser, near the French frontier, was affected by air pollution, the cause of which was undetermined. The pollution had affected tourism in the area. The Belgian Parliament asked the Government to negotiate with France to resolve the problem. The Government replied that, although it was suspected that the pollution was caused by industries in Dunkirk, it did not yet have any facts to verify that. It stated that it was monitoring the pollution to determine its origin. Once it was sure that the pollution came from the Dunkirk region, it would take the matter up with the French Government.<sup>222</sup>

298. In a debate in the Belgian Parliament, it was stated that the Thure, an affluent of the Sambre, at the frontier between France and Belgium, was being seriously polluted by the activities of two quarries, one situated on French territory, the other on Belgian territory, whose operators were washing the rock extracted in the waters of the river. The Belgians had apparently taken some measures to reduce, but not to eliminate, the pollution caused by the operation. The French, however, had done nothing. When the issue was raised in the Belgian Parliament, the Government replied that it had taken up the matter with the French delegation to the Tripartite Standing Committee on Polluted Waters, composed of France, Belgium and Luxembourg. The Parliament, however, seemed to consider that action insufficient and expressed the wish that the Belgian Government should raise the issue directly with the French Government.223

299. States have also provided for the monitoring of an activity by a joint body. For example, in the dispute between Mexico and the United States of America concerning the Rose Street Canal, the United States requested the International Boundary and Water Commission to study the problems caused by that canal, on the border between the two States, and to submit a joint report as early as possible. The report was to include recommendations not only concerning remedial measures but also with respect to an equitable division of costs between the two Governments.<sup>224</sup>

## B. Provisions for prevention of harm

300. Recommendations for the prevention of injuries may require a specific or a more general change in the modes of performance of activities. In order to reduce the risk of harm arising from an activity, changes may be made in the manner in which the activity is conducted. There may be an alteration in the actual structure or

operation of the ongoing activity, a partial or occasional halt, a complete halt or an emergency back-up plan in case of unexpected occurrences. There appears to be a relation between the content of the change and the magnitude of the perceived harm from the activity.

## (a) Multilateral agreements

- 301. Article 4 (paras. (c) to (f)) of the 1960 Convention on the Protection of Lake Constance against Pollution enumerates the types of recommendations that the Commission may make to prevent pollution of the lake and to protect the interests of each of the riparian States (see para. 74 above).
- 302. Recommended changes may relate to the structure and operation of activities. Paragraph (c) of article XI of the 1978 Kuwait Regional Convention for Cooperation on the Protection of the Marine Environment from Pollution provides for this type of recommendation (see para. 70 above).
- 303. The 1949 International Convention for the North-West Atlantic Fisheries also empowers the Commission to recommend, if necessary, a catch limit and a size limit for any species or to prohibit the use of certain appliances. Paragraph 1, subparagraphs (c), (d) and (e), of article VIII read:
  - (c) establishing size limits for any species;
- (d) prescribing the fishing gear and appliances the use of which is prohibited;
  - (e) prescribing an over-all catch limit for any species of fish.
- 304. In addition to changes in the structure and operation of activities, partial or occasional cessation of activities may also be recommended. For example, paragraph 1, subparagraphs (a) and (b), of the same article VIII of the Convention provides that the Commission may establish open or close seasons for catching fish or close to fishing such portions of the area as are populated by small or immature fish:
  - (a) establishing open and closed seasons;
- (b) closing to fishing such portions of a sub-area as the Panel concerned finds to be a spawning area or to be populated by small or immature fish;
- 305. When injury is caused, the injured State may request the assistance of the acting State in taking certain measures in its territory to reduce the injury. For example, article XXI of the 1972 Convention on International Liability for Damage Caused by Space Objects requires the acting State—the launching State—to provide rapid assistance to the injured State when the latter so requests:

#### Article XXI

If the damage caused by a space object presents a large-scale danger to human life or seriously interferes with the living conditions of the population or the functioning of vital centres, the States Parties, and in particular the launching State, shall examine the possibility of rendering appropriate and rapid assistance to the State which has suffered the damage, when it so requests. However, nothing in this article shall affect the rights or obligations of the States Parties under this Convention.

306. Article 199 of the 1982 United Nations Convention on the Law of the Sea provides for contingency plans to be designed by States as well as international organizations to combat pollution:

<sup>221</sup> Ibid.

<sup>&</sup>lt;sup>222</sup> Belgian Parliament, *Questions et réponses* (Questions and answers) bulletin, 29 May 1973.

<sup>&</sup>lt;sup>123</sup> Ibid., 4 July 1973.

<sup>&</sup>lt;sup>224</sup> Whiteman, op. cit. (footnote 45 above), vol. 6, p. 264.

#### Article 199. Contingency plans against pollution

In the cases referred to in article 198, States in the area affected, in accordance with their capabilities, and the competent international organizations shall co-operate, to the extent possible, in eliminating the effects of pollution and preventing or minimizing the damage. To this end, States shall jointly develop and promote contingency plans for responding to pollution incidents in the marine environment.

307. The recommended changes may be compulsory. Article 6 of the 1960 Convention on the Protection of Lake Constance against Pollution requires the riparian States to do their utmost to ensure that the recommendations of the Commission are put into effect within the limits of their domestic legislation. However, the riparian State, in implementing a recommendation of the Commission, may recognize it as binding. Hence it appears that the compulsory nature of the recommendation depends entirely on the willingness of the acting State. Article 6 of the Convention reads:

#### Article 6

- 1. The riparian States undertake to give careful consideration to water protection measures recommended by the Commission which affect their territory and shall do their utmost to ensure that such measures are put into effect within the limits of their domestic legislation
- 2. Riparian States in whose territory water protection measures on which the Commission makes recommendations have to be put into effect may recognize a recommendation of the Commission as being binding on them and instruct their delegation to make a statement to that effect.
- 308. Article VIII of the 1966 International Convention for the Conservation of Atlantic Tunas provides for an elaborate procedure for acceptance of the decisions of the Commission. If a recommendation of the Commission faces opposition by a minority of the contracting parties, that recommendation becomes *effective* for *only* those contracting parties that have not presented an objection thereto. Article VIII reads:

### Article VIII

- 1. (a) The Commission may, on the basis of scientific evidence, make recommendations designed to maintain the populations of tuna and tuna-like fishes that may be taken in the Convention area at levels which will permit the maximum sustainable catch. These recommendations shall be applicable to the Contracting Parties under the conditions laid down in paragraphs 2 and 3 of this article.
  - (b) The recommendations referred to above shall be made:
  - (i) at the initiative of the Commission if an appropriate Panel has not been established or with the approval of at least two thirds of all the Contracting Parties if an appropriate Panel has been established:
  - (ii) on the proposal of an appropriate Panel if such a Panel has been established;
  - (iii) on the proposal of the appropriate Panels if the recommendation in question relates to more than one geographic area, species or groups of species.
- 2. Each recommendation made under paragraph 1 of this article shall become effective for all Contracting Parties six months after the date of the notification from the Commission transmitting the recommendation to the Contracting Parties, except as provided in paragraph 3 of this article.
- 3. (a) If any Contracting Party in the case of a recommendation made under paragraph 1 (b) (i) above, or any Contracting Party member of a Panel concerned in the case of a recommendation made under paragraph 1 (b) (ii) or (iii) above, presents to the Commission an objection to such recommendation within the six months period provided for in paragraph 2 above the recommendation shall not become effective for an additional sixty days.
- (b) Thereupon any other Contracting Party may present an objection prior to the expiration of the additional sixty days period, or

- within forty-five days of the date of the notification of an objection made by another Contracting Party within such additional sixty days, whichever date shall be the later.
- (c) The recommendation shall become effective at the end of the extended period or periods for objection, except for those Contracting Parties that have presented an objection.
- (d) However, if a recommendation has met with an objection presented by only one or less than one fourth of the Contracting Parties, in accordance with subparagraphs (a) and (b) above, the Commission shall immediately notify the Contracting Party or Parties having presented such objection that it is to be considered as having no effect.
- (e) In the case referred to in subparagraph (d) above the Contracting Party or Parties concerned shall have an additional period of sixty days from the date of said notification in which to reaffirm their objection. On the expiry of this period the recommendation shall become effective, except with respect to any Contracting Party having presented an objection and reaffirmed it within the delay provided for.
- (f) If a recommendation has met with objection from more than one fourth but less than the majority of the Contracting Parties, in accordance with subpagraphs (a) and (b) above, the recommendation shall become effective for the Contracting Parties that have not presented an objection thereto.
- (g) If objections have been presented by a majority of the Contracting Parties the recommendation shall not become effective.
- 4. Any Contracting party objecting to a recommendation may at any time withdraw that objection, and the recommendation shall become effective with respect to such Contracting Party immediately if the recommendation is already in effect, or at such time as it may become effective under the terms of this article.
- 5. The Commission shall notify each Contracting Party immediately upon receipt of each objection and of each withdrawal of an objection, and of the entry into force of any recommendation.
- 309. The 1969 International Convention relating to intervention on the High Seas in Cases of Oil Pollution Casualties provides, in paragraph (e) of article III, that the coastal State, before taking measures to minimize damage which may result from its action, shall "use its best endeavours... to afford persons in distress any assistance of which they may stand in need, and in appropriate cases to facilitate the repatriation of ships' crews, and to raise no obstacle thereto".

## (b) Bilateral agreements

- 310. Some bilateral agreements provide that the acting State must notify the potentially injured State of any danger caused as a result of activities in its territory. Thus, for example, the 1971 Act of Santiago concerning hydrologic basins, signed by Argentina and Chile, provides in article 6:
- 6. Within a reasonable period of time, which in any case shall not exceed five months, the requested Party must indicate whether there are any aspects of the plans or plan of operations which might cause it appreciable damage. if so, it shall indicate the technical reasons and calculations substantiating that claim and shall suggest changes in the plans or plan of operations in question which would avoid such damage.
- 311. Again, in the 1956 Treaty between Czechoslovakia and Hungary concerning the régime of State frontiers, paragraphs 2 and 3 of article 24 read:

## Article 24

- 2. If a forest fire breaks out near the frontier line, the competent authorities of the Party on whose territory the fire breaks out shall, as far as possible, do everything in their power to extinguish it and to prevent it from spreading across the State frontier.
- 3. If there is a danger of a forest fire spreading across the State frontier, the Party on whose territory the danger originated shall immediately warn the other Party, so that action may be taken to prevent the fire from spreading across the State frontier.

312. The 1948 Agreement between Poland and the USSR<sup>225</sup> contains similar provisions in article 19 (see para. 80 above) and in article 27, which reads:

#### Article 27

- 1. In sectors adjacent to the frontier line the Contracting Parties will exploit their forests in such a way as not to damage the forests of the other Party.
- 2. If a forest fire breaks out near the frontier, the Contracting Party on whose territory the fire breaks out must, as far as possible, do everything in its power to localize and extinguish the fire and to prevent it from spreading across the frontier.
- 3. Should a forest fire threaten to spread across the frontier, the Contracting Party on whose territory the danger originated shall immediately warn the other Contracting Party so that appropriate action may be taken to localize the fire on the frontier.
- 4. If trees fall beyond the frontier line as the result of elemental causes or logging operations, the competent authorities of the Contracting Parties shall take all steps to enable the persons concerned of the neigbouring Party to cut up and remove the trees to their own territory. The competent authorities of the Contracting Party to which the trees belong must inform the competent authorities of the other Party of such occurrences.
- 5. In such cases the transportation of the trees across the frontier shall be exempt from all duties or taxes.
- 313. Similarly, article 4 of the 1966 Convention between Belgium and France<sup>226</sup> provides for *mutual assistance* in case of *accident*, the means of assistance to be placed under a single authority:

#### Article 4

In case of accident, the Contracting Parties, being desirous of assisting each other to the greatest possible extent, shall place the means of assistance that they furnish under a single authority, which shall be responsible for the general administration of the aid and emergency action.

The provisions relating to this mutual assistance are set out in annex III.

314. Finland and Norway, in their 1951 Agreement on the transfer of water from their frontier rivers, 227 agreed on certain *measures* that each country must take individually in order to *offset any inconvenience* that such transfer might create for the inhabitants of the river banks. Those measures are stipulated in article 2:

#### Article 2

To offset any inconvenience which the transfer of water referred to in article 1 could cause the inhabitants along the banks of the Näätämo river, the Governments shall take the following measures:

(a) The Government of Norway shall take steps to facilitate the movement of salmon upstream past Koltaakoski on the Näätämo river so that the fish can gain access to the upper reaches of the river.

The plans for the installation shall be laid before fishery experts designated by the Government of Finland, for their opinion.

The work shall be carried out at the expense of the Government of Norway as soon as possible after the entry into force of this Agreement.

- (b) The Government of Finland shall arrange for the removal of a number of large boulders and holms lying in the stretch of about four kilometres along the Näätämo river between the confluence of the Kallo and Näätämo rivers and the frontier between Finland and Norway and obstructing timber-floating.
- 315. Similarly, article 20 (see para. 284 above) and article 21 of the 1952 Agreement between Poland and the

German Democratic Republic<sup>228</sup> contain a recommendation for mutual assistance in case of crisis. Article 21 reads:

#### Article 21

If a dike bursts, the two Parties shall immediately combine their efforts to repair the damage, furnishing technical facilities and the necessary labour.

The Party which asks for assistance shall bear the cost involved.

316. In an exchange of notes in 1974, the United States of America and Canada agreed to establish joint pollution contingency plans for waters of mutual interest.<sup>229</sup> Under the 1961 Treaty between Canada and the United States of America relating to the development of the Columbia River Basin,<sup>230</sup> each party is required to exercise due diligence to remove the cause and mitigate the effect of any injury to the other's territory. Paragraph 3 of article XVIII of that Treaty reads:

## Article XVIII. Liability for damage

- 3. Canada and the United States of America, each to the extent possible within its territory, shall exercise due diligence to remove the cause of and to mitigate the effect of any injury, damage or loss occurring in the territory of the other as a result of any act, failure to act, omission or delay under the Treaty.
- 317. The 1931 General Convention between Romania and Yugoslavia<sup>231</sup> recommends, in chapter I, article 7, that the two parties should effect *joint inspections* of the hydraulic system on their common waters:

#### Article 7

On the proposal of either State, and subject to previous consent, joint inspections of the places affected may be made from time to time for the purpose of studying the hydraulic system of the hydrotechnical areas and watercourses and their basins, in order to consider what measures are advisable or what works should be carried out for the maintenance or improvement of the hydraulic system affecting either or both of the States.

Accordingly, subject to the previous consent of both States, and on the proposal of either State, a joint inspection of affected areas may take place in order to recommend measures to improve the hydraulic system affecting either or both States.

318. In some cases, recommendations for the prevention or minimization of harm have taken the form of the banning of certain activities at certain places. For example, in the 1948 Agreement between the Soviet Union and Finland, 232 the parties agreed, in order to safeguard their joint frontier line, to establish a belt 20 metres wide within and around their border, where exploitation of mineral deposits was ordinarily to be prohibited and permitted only in exceptional cases and by agreement between the two States. Article 18 of the Agreement reads in part:

<sup>225</sup> See footnote 39 above.

<sup>&</sup>lt;sup>226</sup> See footnote 42 above.

<sup>&</sup>lt;sup>227</sup> Agreement of 25 April 1951 between Finland and Norway on the transfer from the course of the Näätämo (Neiden) River to the course of the Gandvik River of water from the Garsjöen, Kjerringvatn and Förstevannene Lakes.

<sup>228</sup> See foonote 211 above.

<sup>&</sup>lt;sup>229</sup> Exchange of notes of 19 June 1974 between the United States of America and Canada constituting an agreement relating to joint pollution contingency plans for spills of oil and other noxious substances in waters of mutual interest.

<sup>230</sup> See foonote 212 above.

<sup>&</sup>lt;sup>231</sup> See footnote 60 above.

<sup>&</sup>lt;sup>232</sup> See footnote 208 above.

#### Article 18

- 1. Mining and the prospecting of mineral deposits in the immediate vicinity of the frontier shall be governed by the regulations of the Party in whose territory the workings are situated.
- 2. In order to safeguard the frontier line there shall on each side thereof be a belt twenty metres wide within which the work referred to in paragraph 1 of this article shall ordinarily be prohibited and shall be permitted only in exceptional cases by agreement between the competent authorities of the Contracting Parties.

## (c) Judicial decisions and State practice other than agreements

- 319. With regard to prevention of injury, judicial decisions and official correspondence show that, through various methods of deterrence, States and non-State entities have sought to postpone or to re-evaluate an activity. For example, in 1961, the United States decided to release 20 kilograms of tiny copper "hairs" or "needles" in outer space to form a belt around the earth about 15 kilometres wide and 30 kilometres deep. The purpose was to test the capability of the belt to intercept communications signals. The prospect of such use of the shared resource caused international as well as national concern among scientific groups about its possible adverse effects upon radiocommunications and optical astronomy. The Soviet Union also complained about possible interference with the movement of spacecraft. As a result of many protests, a special meeting of the United States President's Scientific Advisory Council (PSAC) was called to review the project and advise whether the launching should be stopped, but PSAC finally decided that the operation should be a safe undertaking. The West Ford Test consequently went ahead a month later.233
- 320. Courts have also used the procedure of *injunctions* to postpone an activity with potentially harmful consequences pending a final decision on merits. This is a fairly routine procedure in the United States with regard to environmental problems. For example, in connection with the construction of the trans-Alaska pipeline, an action by three United States conservation groups, with intervention by their Canadian counterparts, was brought before the United States District Court, District of Columbia.<sup>234</sup> The Court issued a preliminary injunction, which it subsequently dissolved; it denied a permanent injunction and dismissed the complaint.<sup>235</sup>
- 321. In certain cases regarding acts with extraterritorial harmful consequences, international tribunals have also granted the equivalent of injunctions. Thus at an early stage of the *Fisheries Jurisdiction* cases, the International Court of Justice, by its order of 17 August 1972, prescribed interim measures of protection which,

<sup>233</sup> See *The New York Times*, issues of 30 July 1961, p. 48, col. 1; 3 February 1962, p. 5, col. 1; 10 May 1962, p. 16, col. 4; 13 May 1963, p. 1, col. 5; 21 May 1963, p. 3, col. 1; 23 September 1963, p. 28,

col. 2.

among other things, provided that the parties should "ensure that no action of any kind is taken which might aggravate or extend the dispute submitted to the Court" and also "ensure that no action is taken which might prejudice the rights of the other party in respect of the carrying out of whatever decision on the merits the Court may render".236 The Court also held that Iceland should refrain from taking any measures to enforce its purported new fisheries regulations against ships registered in the United Kingdom or the Federal Republic of Germany outside the agreed 12-mile fisheries zone and that it should refrain from applying any administrative, judicial or other measures against such ships, their crews, or other related persons because they had engaged in fishing activities between 12 and 50 miles offshore. For their part, the United Kingdom and the Federal Republic of Germany were directed not to take more than 170,000 and 119,000 metric tons of fish respectively from the "sea area of Iceland". Iceland, however, ignored the order.

- 322. A preliminary injunction was also granted by the International Court of Justice in the *Nuclear Tests* cases. By its order of 22 June 1973, the Court requested France to "avoid nuclear tests causing the deposit of radioactive fallout" over Australia and New Zealand, pending final decisions in its proceedings.<sup>237</sup>
- 323. State practice shows that acting States are reluctant to halt completely activities that are in their legitimate interests. However, the results of management and monitoring indicate in certain instances that a change has been made in the operation of the activity to take into account the interests of other States.
- 324. Following an exchange of notes, in 1961, between the United States of America and Mexico concerning two United States companies, *Peyton Packing* and *Casuco*, located near the Mexican border, whose activities were prejudicial to Mexico, the two companies took substantial measures to ensure that their operations ceased to inconvenience the Mexican border cities. Those measures included *phasing out* certain activities, *changing working hours* and establishing systems of disinfection:

The Peyton Packing Plant:

- 3. Reduced the number of cattle so that there are no more than 6,500 head in the pens at any time.
- 4. Constructed a system of spray heads on the fences of all four sides of the property for very high pressure dispensing of a masking agent to alleviate any remaining odour emanating from the premises.
- 5. Began plans which in approximately twelve months will remove all cattle feeding operations from the present area.

The Casuco Company:

- 1. Eliminated the rendering of partially decomposed carcasses.
- 2. Changed operations from night hours to day hours to take the most advantage of meteorological conditions.
- 3. Built an oxidizing furnace fired by natural gas to oxidize any odours from the plant operation.
- 4. Constructed a condenser to condense all possible vapours, which are disposed of with the liquid waste.

<sup>&</sup>lt;sup>234</sup> Wilderness Society v. Hickel (1970) (Federal Supplement, 1971, vol. 325, pp. 422 and 424). See also Natural Resources Defense Council v. Morton (1972) (Federal Reporter, 2nd Series, 1972, vol. 458, p. 827).

<sup>&</sup>lt;sup>235</sup> Wilderness Society v. Morton (1973) (Federal Reporter, 2nd Series, 1973, vol. 479, p. 842).

<sup>&</sup>lt;sup>236</sup> Fisheries Jurisdiction (United Kingdom v. Iceland; Federal Republic of Germany v. Iceland), Interim measures of protection, Order of 17 August 1972, I.C.J. Reports 1972, pp. 17 and 35.

5. Installed a system of sprays to counteract any remaining odour that might otherwise escape into the atmosphere.

With respect to industrial waste, the Peyton Packing Company constructed a primary treatment plant. It removes from the waste going into the Rio Grande all the blood and much of the solid organic matter. While this is not complete treatment, when the public sewers become available to the company, the effluent will be disposed of by that method.<sup>236</sup>

325. Similarly, the company operating the Trail Smelter undertook to contain and reduce the fumes emanating from the plant and causing damage to the State of Washington by treating the sulphur dioxide emitted:

The Consolidated Mining and Smelting Company of Canada, Limited, proceeded after 1930 to make certain changes and additions in its plant, with the intention and purpose of lessening the sulphur contents of the fumes, and in an attempt to lessen injurious fumigations, a new system of control over the emission of fumes during the crop growing season came into operation about 1934. To the three sulphuric acid plants in operation since 1932, two others have recently been added. The total capacity is now of 600 tons of sulphuric acid per day, permitting, if these units could run continually at capacity, the fixing of approximately 200 tons of sulphur per day. In addition, from 1936, units for the production of elemental sulphur have been put into operation. There are at present three such units with a total capacity of 140 tons of sulphur per day. The capacity of absorption of sulphur dioxide is now 600 tons of sulphur dioxide per day (300 tons from the zinc plant gases and 300 tons from the lead plant gases). As a result, the maximum possible recovery of sulphur dioxide, with all units in full operation has been brought to a figure which is about equal to the amount of that gas produced by smelting operations at the plant in 1939. However, the normal shutdown of operating units for repairs, the power supply, ammonia available, and the general market situation are factors which influence the amount of sulphur dioxide treated.239

326. These measures greatly lessened the amount of sulphur dioxide dispersed into the air:

In 1939, 360 tons, and in 1940, 416 tons, of sulphur per day were oxidized to sulphur dioxide in the metallurgical processes at the plant. Of the above, for 1939, 253 tons, and for 1940, 289 tons per day, of the sulphur which was oxidized to sulphur dioxide was utilized. One hundred and seven tons and 127 tons of sulphur per day for those two years, respectively, were emitted as sulphur dioxide to the atmosphere.

The tons of sulphur emitted into the air from the Trail Smelter fell from about 10,000 tons per month in 1930 to about 7,200 tons in 1931 and 3,400 tons in 1932 as a result both of sulphur dioxide beginning to be absorbed and of depressed business conditions. As depression receded, this monthly average rose in 1933 to 4,000 tons, in 1934 to nearly 6,300 tons and in 1935 to 6,800 tons. In 1936, however, it had fallen to 5,600 tons; in 1937, it further fell to 4,850 tons; in 1938, still further to 4,230 tons to reach 3,250 tons in 1939. It rose again, however, to 3,875 tons in 1940.240

- 327. In cases of dispute over the distribution and delimitation of resources, a régime is sometimes established to ensure the more equitable distribution of the resource between interested States. For example, in the Fisheries Jurisdiction cases, the parties were required to negotiate, in good faith, an equitable distribution of their fishing rights off the coast of Iceland.
- 328. Changes may also consist in expanding a designated danger area and giving notice of imminent danger to States or other international actors. Thus, in the Eni-

wetok Atoll nuclear tests series, after the monitoring of the activity had shown that harm might occur to others outside the initially calculated danger zone, that zone was expanded in 1953. The expansion was announced in a Notice to Mariners issued in 1954.<sup>241</sup>

- Once it becomes evident that there is no further possibility of injuries from a particular activity, preventive measures are no longer required. Thus the maintenance of a danger zone becomes unnecessary once tests are completed and it has been established that no harm is likely to ensue from those tests in the future. That was the position taken by the United States following its nuclear tests: it stated that the personnel of Joint Task Force Seven had suffered no injuries attributable to the effects of the test; consequently, following a post-operation radiological survey of the Eniwetok-Bikini danger area, it was determined that the danger area could be disestablished without hazard. However, the land area of the Bikini and Eniwetok atolls, the water area of their lagoons and the adjacent areas within three miles to seaward of the atolls and the overlying airspace remained closed to vessels and aircraft which did not have specific clearance.242
- 330. In addition to the restructuring of an activity or the establishment of additional safeguards or curtailment of operations, the *partial* or *occasional* halting of the activity may be required. In the *Trail Smelter* arbitration, such a preventive measure was instituted in the event that fumes emissions exceeded well-defined limits. Guidelines were established taking into consideration agricultural activities that might be harmed by fumes emissions, whether during the growing or the nongrowing season:

#### General restrictions and provisions

(a) If the Columbia Gardens recorder indicates 0.3 part per million or more of sulphur dioxide for two consecutive twenty-minute periods during the growing season, and the wind direction is not favourable, emission shall be reduced by four tons of sulphur per hour or shut down completely when the turbulence is bad, until the recorder shows 0.2 part per million or less of sulphur dioxide for three consecutive twenty-minute periods.

If the Columbia Gardens recorder indicates 0.5 part per million or more of sulphur dioxide for three consecutive twenty-minute periods during the non-growing season and the wind direction is not favourable, emission shall be reduced by four tons of sulphur per hour or shut down completely when the turbulence is bad, until the recorder shows 0.2 part per million or less of sulphur dioxide for three consecutive twenty-minute periods.

- (b) In case of rain or snow, the emission of sulphur shall be reduced by two (2) tons per hour. This regulation shall be put into effect immediately when precipitation can be observed from the Smelter and shall be continued in effect for twenty (20) minutes after such precipitation has ceased.
- (c) If the slag retreatment furnace is not in operation the emission of sulphur shall be reduced by two (2) tons per hour.
- (d) If the instrumental reading shows turbulence excellent, good or fair, but visual observations made by trained observers clearly indicate that there is poor diffusion, the emission of sulphur shall be reduced to the figures given in column (1) if wind is not favourable, or column (2) if wind is favourable.
- (e) When more than one of the restricting conditions provided for in (a), (b), (c) and (d) occur simultaneously, the highest reduction shall apply.

<sup>&</sup>lt;sup>237</sup> Nuclear Tests (Australia v. France; New Zealand v. France), Interim measures of protection, Order of 22 June 1973, I.C.J. Reports 1973, pp. 106 and 142. Less than a month after the order, France exploded another device over its Pacific atoll of Mururoa.

<sup>&</sup>lt;sup>236</sup> Whiteman op. cit. (see footnote 45 above), vol. 6, pp. 258-259.

<sup>&</sup>lt;sup>239</sup> United Nations, Reports of International Arbitral Awards, vol. III, p. 1946.

<sup>240</sup> Ibid., pp. 1946-1948.

<sup>&</sup>lt;sup>241</sup> Whiteman, op. cit., vol. 4, pp. 560-561.

<sup>242</sup> Ibid., pp. 594-595.

(f) If, during the non-growing season, the instrumental reading shows turbulence fair and wind not favourable but visual observations by trained observers clearly indicate that there is excellent diffusion, the maximum permissible emission of sulphur may be increased to the figures in column (5). The general restrictions under (a), (b), (c) and (e), however, shall be applicable.<sup>243</sup>

Notice was to be issued when the emission limits were exceeded.

331. Some activities may cause such extensive injuries that their complete prohibition is demanded, particularly if it is considered that no precautions can adequately safeguard against the perceived harm. Although this survey of judicial decisions and official correspondence has not uncovered an instance where a legitimate activity was permanently banned, requests for a total ban have been made in the area of nuclear activities. During the *Eniwetok Atoll* nuclear tests, Japan protested against atmospheric nuclear tests and called for an immediate suspension of all tests:

In view of this menace posed by nuclear tests to mankind and from a humanitarian standpoint, the Japanese Government and people have consistently had an earnest desire that all nuclear bomb tests be suspended immediately. This desire was stated in a note verbale sent by the Foreign Ministry to the United States Embassy in Japan on September 15, 1957, asking for the suspension of tests on Eniwetok, and also in Prime Minister Kishi's letter to President Eisenhower, dated September 24, 1957.

The Japanese Government regrets that the United States Government, in spite of the desire of the Japanese Government and people, has announced the establishment of a danger zone to conduct nuclear bomb tests. The Japanese Government takes this opportunity to request again that the United States Government consider seriously the suspension of the aforementioned tests.

The United States Government states that every possible precaution will be taken to prevent damage and injury to human lives and property in the danger zone and that there is no probability of any accidents outside the danger zone. Whatever precaution is taken, however, the Japanese Government is greatly concerned over conducting of nuclear tests and establishment of a danger zone for that purpose in view of the fact that said zone is near to routes of the Japanese merchant marine and to fishing grounds of Japanese fishing boats.<sup>244</sup>

- 332. In a border incident between France and Switzerland in 1892, the French Government decided to halt the military target practice exercises near the Swiss border until steps had been taken to avoid accidental transboundary injuries.<sup>245</sup>
- 333. Emergency back-up plans may also be requested to minimize injury in case it occurs. Such plans were in effect during the *Eniwetok Atoll* nuclear tests in the event of error in wind predictions. Those plans included immediate *evacuation* of persons and immediate *medical care*. The plans were executed when the magnitude of the 1 March 1954 test explosion had been underestimated by half and the error had been compounded by erroneous wind predictions:

The United States meanwhile took swift action to mitigate the effects of the test mishaps. Injured Marshallese were given immediate medical care at naval facilities on nearby Kwajalein Atoll; expert medical personnel were rushed to their assistance, and to that of the injured Japanese fishermen as soon as their plight became known; and prompt assurances were given that all financial loss would be made good. . . . Two million dollars have been paid to Japan for damages resulting from the tests, including both personal injuries suffered by

- 334. In the Lake Lanoux case, the tribunal noted that the potentially injured or affected State had the right to assert its interests and to demand modification of the acting State's activities. The acting State must take the affected State's proposals into consideration. Procedurally, the upstream State had a right of initiative, but it must nevertheless examine the schemes proposed by the downstream State. Of course, the upstream State had the right to give preference to the solution contained in its own plan, provided that it took into consideration in a reasonable manner the interests of the downstream State.<sup>247</sup>
- 335. At least one international judicial decision indicates that, upon failure by the acting State to fulfil its duty of care unilaterally or to reach agreement through negotiations with the affected State, the decision process will be subject to review by an international tribunal. In its decision in the *Trail Smelter* case, the tribunal expressly provided for recourse by the parties, in the event of failure to agree on amendment or suspension of the permanent régime, to decision-making by a joint body consisting of reputable scientists (see para. 296 above).
- 336. In the Fisheries Jurisdiction cases, the International Court of Justice stated that competence in the matter of the equitable distribution of fishery resources lay with the parties to the dispute themselves. After enumerating the factors to be considered in designing an equitable régime, the Court concluded that in that particular case "negotiation" was the most appropriate method for resolving the dispute; the decision must be based on scientific data, which were mainly in the possession of the parties:
- . . . This necessitates detailed scientific knowledge of the fishing grounds. It is obvious that the relevant information and expertise would be mainly in the possession of the Parties. The Court would, for this reason, meet with difficulties if it were itself to attempt to lay down a precise scheme for an equitable adjustment of the rights involved. It is thus obvious that both in regard to merits and to jurisdiction the Court only pronounces on the case which is before it and not on any hypothetical situation which might arise in the future.<sup>248</sup>
- 337. Similarly, in the *Corfu Channel* case, the Court noted that the decision whether notice of a condition existing in the territorial waters of the State concerned must be given to other States depended on information uniquely within the possession of that State. The Court took note of the lack of notice and held Albania liable for failure to notify, and for the injuries resulting therefrom.
- 338. Again, in the North Sea Continental Shelf cases, the Court, quoting from the order of 19 August 1929 of the Permanent Court of International Justice in the case of the Free Zones of Upper Savoy and the District of Gex, stated that the judicial settlement of international disputes "is simply an alternative to the direct and friendly settlement of such disputes between the

the crew of the Fukuryu Maru and damage to the Japanese fishing industry. . . . 246

<sup>&</sup>lt;sup>243</sup> United Nations, Reports of International Arbitral Awards, vol. III. pp. 1976-1977.

<sup>&</sup>lt;sup>244</sup> Whiteman, op. cit., vol. 4, p. 585.

<sup>245</sup> See footnote 191 above.

<sup>&</sup>lt;sup>246</sup> Whiteman, op. cit., vol. 4, p. 571.

<sup>&</sup>lt;sup>247</sup> International Law Reports 1957 (see footnote 67 above), p. 140, para. 23 of the award.

<sup>&</sup>lt;sup>248</sup> I.C.J. Reports 1974, pp. 31-32, para. 73.

parties".249 The Court added that the obligation to negotiate was emphasized by the observable fact that judicial or arbitral settlement was not universally accepted.

339. The operators of certain activities have sometimes determined for themselves what changes should be made to prevent or minimize injuries to others. In the case of the *Peyton Packing Company* and *Casuco Company* plants, the companies made changes in their activities (slaughtering and meat processing) in order to

reduce the injurious effects on Mexico and its nationals; specifically, they modified the technical operations of the plants as well as the schedule of operations (see para. 324 above).

340. Similarly, the Canadian operator of the Trail Smelter took steps to diminish the emissions of fumes which were polluting the State of Washington, in particular by building additional sulphur dioxide treatment plants (see para. 325 above). However, the measures taken by the company were reviewed and added to by the arbitral tribunal in the light of the widespread damage caused in that State.

## CHAPTER IV

## Guarantees of compensation for injuries

341. When it is decided to permit the performance of certain activities, in the knowledge that they may cause injuries, it is necessary to provide, in advance, for guarantees of payment of damages. This means that the operator of certain activities must either take out an insurance policy or provide financial security. Such requirements are similar to those stipulated in the domestic laws of many States in connection with the operation of complex industries, as well as with more routine activities such as driving a car.

## (a) Multilateral agreements

342. Some multilateral agreements include provisions to ensure the payment of compensation in case of harm and liability. Most multilateral agreements concerning nuclear activities are in this category. Thus, they require the maintenance of insurance or other financial security for the payment of damages in case of liability. The 1962 Convention on the Liability of Operators of Nuclear Ships requires the maintenance of such security. The terms and the amount of the insurance carried by the operator of nuclear ships are determined by the licensing State. Although the licensing State is not required to carry insurance or to provide other financial security, it must "ensure" the payment of claims for compensation for nuclear damage if the operator's insurance or security proves to be inadequate. The relevant paragraphs of article III of the Convention read:

## Article III

- 1. The liability of the operator as regards one nuclear ship shall be limited to 1,500 million francs in respect of any one nuclear incident, notwithstanding that the nuclear incident may have resulted from any fault or privity of that operator; such limit shall include neither any interest nor costs awarded by a court in actions for compensation under this Convention.
- 2. The operator shall be required to maintain insurance, or other financial security covering his liability for nuclear damage, in such amount, of such type and in such terms as the licensing State shall specify. The licensing state shall ensure the payment of claims for compensation for nuclear damage established against the operator by providing the necessary funds up to the limit laid down in paragraph 1 of this article to the extent that the yield of the insurance or the financial security is inadequate to satisfy such claims.

- 3. However, nothing in paragraph 2 of this article shall require any Contracting State or any of its constituent subdivisions, such as States, Republics or Cantons, to maintain insurance or other financial security to cover their liability as operators of nuclear ships.
- 343. Similar requirements are stipulated in article VII of the 1963 Vienna Convention on Civil Liability for Nuclear Damage. The operator is required to maintain an insurance or other financial security required by the installation State. While the installation State is not required to carry insurance or to provide other financial security to cover the injuries that may be caused by the operation of the nuclear plant, it must ensure the payment of claims for compensation established against the operator by providing the necessary funds if the insurance is inadequate. Article VII reads:

## Article VII

- 1. The operator shall be required to maintain insurance or other financial security covering his liability for nuclear damage in such amount, of such type and in such terms as the Installation State shall specify. The Installation State shall ensure the payment of claims for compensation for nuclear damage which have been established against the operator by providing the necessary funds to the extent that the yield of insurance or other financial security is inadequate to satisfy such claims, but not in excess of the limit, if any, established pursuant to article V.
- 2. Nothing in paragraph 1 of this article shall require a Contracting Party or any of its consituent subdivisions, such as States or Republics, to maintain insurance or other financial security to cover their liability as operators.
- 3. The funds provided by insurance, by other financial security or by the Installation State pursuant to paragraph 1 of this article shall be exclusively available for compensation due under this Convention.
- 4. No insurer or other financial guarantor shall suspend or cancel the insurance or other financial security provided pursuant to paragraph 1 of this article without giving notice in writing of at least two months to the competent public authority or, in so far as such insurance or other financial security relates to the carriage of nuclear material, during the period of the carriage in question.
- 344. Likewise, article 10 of the 1960 Convention on Third Party Liability in the Field of Nuclear Energy requires the operator of nuclear plants to maintain insurance or provide other financial security in accordance with the Convention:

<sup>&</sup>lt;sup>249</sup> I.C.J. Reports 1969, p. 47, para. 87.

#### Article 10

- (a) To cover the liability under this Convention, the operator shall be required to have and maintain insurance or other financial security of the amount established pursuant to article 7 and of such type and terms as the competent public authority shall specify.
- (b) No insurer or other financial guarantor shall suspend or cancel the insurance or other financial security provided for in paragraph (a) of this article without giving notice in writing of at least two months to the competent public authority or, in so far as such insurance or other financial security relates to the carriage of nuclear substances, during the period of the carriage in question.
- (c) The sums provided as insurance, reinsurance, or other financial security may be drawn upon only for compensation for damage caused by a nuclear incident.
- 345. In addition to conventions dealing with nuclear materials, conventions regulating other activities also require guarantees for payment of compensation in case of injury. Under article 15 of the 1952 Convention on Damage Caused by Foreign Aircraft to Third Parties on the Surface, the operators of aircraft registered in another contracting State are required to maintain insurance or provide other security for possible damage they may cause on the surface. Paragraph 4 (c) of that article provides that a contracting State may accept, instead of insurance, the guarantee of the contracting State in which the aircraft is registered, provided that that State undertakes to waive immunity from suit in respect of that guarantee. Article 15 reads:

#### SECURITY FOR OPERATOR'S LIABILITY

#### Article 15

- 1. Any Contracting State may require that the operator of an aircraft registered in another Contracting State shall be insured in respect of his liability for damage sustained in its territory for which a right to compensation exists under article 1 by means of insurance up to the limits applicable according to the provisions of article 11.
- 2. (a) The insurance shall be accepted as satisfactory if it conforms to the provisions of this Convention and has been effected by an insurer authorized to effect such insurance under the laws of the State where the aircraft is registered or of the State where the insurer has his residence or principal place of business and whose financial responsibility has been verified by either of those States.
- (b) If insurance has been required by any State under paragraph 1 of this article, and a final judgment in that State is not satisfied by payment in the currency of that State, any Contracting State may refuse to accept the insurer as financially responsible until such payment, if demanded, has been made.
- 3. Notwithstanding the last preceding paragraph, the State overflown may refuse to accept as satisfactory insurance effected by an insurer who is not authorized for that purpose in a Contracting State.
- 4. Instead of insurance, any of the following securities shall be deemed satisfactory if the security conforms to article 17:
- (a) a cash deposit in a depository maintained by the Contracting State where the aircraft is registered or with a bank authorized to act as a depository by that State;
- (b) a guarantee given by a bank authorized to do so by the Contracting State where the aircraft is registered, and whose financial responsibility has been verified by that State;
- (c) a guarantee given by the Contracting State where the aircraft is registered, if that State undertakes that it will not claim immunity from suit in respect of that guarantee.
- 5. Subject to paragraph 6 of this article, the State overflown may also require that the aircraft shall carry a certificate issued by the insurer certifying that insurance has been effected in accordance with the provisions of this Convention, and specifying the person or persons whose liability is secured thereby, together with a certificate or endorsement issued by the appropriate authority in the State where the aircraft is registered or in the State where the insurer has his residence or principal place of business certifying the financial responsibility of the insurer. If other security is furnished in accordance with the provisions of paragraph 4 of this article, a certificate to that effect shall

be issued by the appropriate authority in the State where the aircraft is registered.

- 6. The certificate referred to in paragraph 5 of this article need not be carried in the aircraft if a certified copy has been filed with the appropriate authority designated by the State overflown or, if the International Civil Aviation Organization agrees, with that Organization, which shall furnish a copy of the certificate to each Contracting State.
- 7. (a) Where the State overflown has reasonable grounds for doubting the financial responsibility of the insurer, or of the bank which issues a guarantee under paragraph 4 of this article, that State may request additional evidence of financial responsibility, and if any question arises as to the adequacy of that evidence the dispute affecting the States concerned shall, at the request of one of those States, be submitted to an arbitral tribunal which shall be either the Council of the International Civil Aviation Organization or a person or body mutually agreed by the parties.
- (b) Until this tribunal has given its decision the insurance or guarantee shall be considered provisionally valid by the State overflown.
- 8. Any requirements imposed in accordance with this article shall be notified to the Secretary General of the International Civil Aviation Organization who shall inform each Contracting State thereof.
- 9. For the purpose of this article, the term "insurer" includes a group of insurers, and for the purpose of paragraph 5 of this article, the phrase "appropriate authority in a State" includes the appropriate authority in the highest political subdivision thereof which regulates the conduct of business by the insurer.
- 346. Similarly, articles 11 and 11A of the draft convention prepared by IMO in 1984 on liability and compensation in connection with the carriage of noxious and hazardous substances by sea provide for compulsory insurance of the shipowner and shipper:

## Article 11. Compulsory insurance of the shipowner

- 1. The owner of a ship registered in a Contracting State shall be required to maintain insurance or other financial security, such as the guarantee of a bank or a certificate delivered by an international compensation fund, in the sums fixed by applying the limits of liability prescribed in article 6 to cover his liability for damage under the present Convention. The same shall apply to a ship not registered in a Contracting State entering or leaving a port or other place for the loading or discharge of cargo [within the territory] [in an area under the jurisdiction] of a Contracting State.
- 2. A certificate attesting that insurance or other financial security is in force in accordance with the provisions of the present Convention shall be issued by the appropriate authority to each ship after determining that the requirements of paragraph 1 have been complied with. With respect to ships registered in a Contracting State, the certificate shall be issued or certified by the appropriate authority of the State of the ship's registration and, with respect to ships not registered in a Contracting State, the certificate shall be issued or certified by the appropriate authority of [any Contracting State] [the Contracting State referred to in the second sentence of paragraph 1] [such other Contracting State as a Contracting State may authorize]. This certificate shall be in the form of the annexed model and shall contain the following particulars:
  - (a) name of the ship and port of registration;
  - (b) name and principal place of business of the owner;
  - (c) type of security;
- (d) name and principal place of business of insurer or other person giving security and, where appropriate, place of business where the insurance or security is established; and
- (e) period of validity of certificate which shall not be longer than the period of validity of the insurance or other security.
- 3. The certificate shall be carried on board the ship and a copy shall be deposited with the appropriate authorities of the State of the ship's registry.
- 4. Insurance or other financial security shall not satisfy the requirements of this article if it can cease, for reasons other than the expiry of the period of validity of the insurance or security specified in the certificate under paragraph 2, before three months have elapsed from the date on which notice of its termination is given to the

authorities referred to in paragraph 3, unless the certificate has been surrendered to these authorities or a new certificate has been issued within the said period. The foregoing provisions shall similarly apply to any modification which results in the insurance or security no longer satisfying the requirements of this article.

- 5. The State where the certificate is issued or certified shall, subject to the provisions of this article and of article 11B, determine the conditions of issue and validity of the certificate.
- [6. Any sums provided by insurance or by other financial security maintained in accordance with paragraph 1 shall be available only for the satisfaction of claims under the present Convention.]

#### Article 11A. Compulsory insurance of the shipper

- 1. The shipper of a consignment of hazardous substances shall be required to maintain insurance or other financial security, such as a bank guarantee in the sum laid down in article 8, paragraph 1, to cover his liability for damage under the present Convention.
- 2. A certificate attesting that insurance or other financial security is in force in accordance with the provisions of the present Convention shall be issued by the insurer or other person providing financial security for the shipper's liability with respect to each consignment. This certificate shall be delivered by the shipper to the owner when the consignment is handed over for carriage by sea.
- 3. This certificate shall be in the form of the annexed model and shall contain the following particulars:
- (a) the name of the ship or the ships on board of which the consignment is expected to be carried and their port of registration;
  - (b) the name and principal place of business of the insured person;
- (c) any particulars necessary for identification of the consignment; these particulars shall also contain a description of the substances which is in accordance with the requirements of any international generally accepted standards relating to sea carriage of dangerous substances;
  - (d) the type of security referred to in paragraph 1;
- (e) the name and principal place of business of the insurer or other person giving security; and
  - (f) the period of validity of the insurance or other security.
- 4. The insurance or other financial security shall be effected with an insurer or other person providing security approved for this purpose by any Contracting State.
- 5. The insurance or security shall cover the entire period of the shipper's liability and shall cover the liability under the present Convention of the person named in the certificate as shipper or, if that person should not be the shipper as defined, of such person as does incur liability as shipper under the present Convention.
- 6. Any sums provided by insurance or by other financial security maintained in accordance with paragraph 1 shall be available only for the satisfaction of claims under the present Convention.
- 347. The 1969 International Convention on Civil Liability for Oil Pollution Damage requires that the owner of a ship registered in a contracting State which carries more than 2,000 tons of oil as cargo maintain insurance or other financial security. Paragraph 1 of article VII reads:

#### Article VII

- 1. The owner of a ship registered in a Contracting State and carrying more than 2,000 tons of oil in bulk as cargo shall be required to maintain insurance or other financial security, such as the guarantee of a bank or a certificate delivered by an international compensation fund, in the sums fixed by applying the limits of liability prescribed in article V, paragraph 1, to cover his liability for pollution damage under this Convention.
- 348. Again, under paragraph 1 of article 8 of the 1976 Convention on Civil Liability for Oil Pollution Damage Resulting from Exploration for and Exploitation of Seabed Mineral Resources, the operator of an installation is required to have and maintain insurance or other financial security to the amount and on the terms required by the controlling State.

349. Article 235 of the 1982 United Nations Convention on the Law of the Sea also provides, in paragraph 3, that States shall co-operate in developing procedures for payment of adequate compensation, such as "compulsory insurance or compensation funds".

## (b) Bilateral agreements

350. Some bilateral agreements also take account of the need to provide for guarantees of compensation in case of injury. At least two bilateral agreements examined in the present study require such assurances. In the 1973 Agreement between Norway and the United Kingdom regarding the transmission of petroleum by pipeline from the Ekofisk field to the United Kingdom,<sup>250</sup> the licensees are required to take out insurance or to furnish security or guarantees in respect of possible damage, as provided in article 11:

#### Article 11

Liability for pollution damage including the costs of preventive and remedial action, shall be governed in accordance with the provisions of article 4. The licence or licences may\* contain conditions concerning the liability of the licensees and their obligations to insure against or to furnish security or guarantees in respect of possible pollution damage.

351. A similar provision is contained in article 12 of the 1974 Agreement between the Federal Republic of Germany and Norway for the transmission of petroleum by pipeline from the Ekofisk field to the Federal Republic of Germany,<sup>251</sup> but the terms appear to be more mandatory:

#### Article 12

Liability for pollution damage, including the costs of preventive and remedial action, shall be governed in accordance with the provisions of article 4. Licences shall\* contain provisions concerning the liability of the licensees and their obligations to insure against or to furnish security or guarantees\* in respect of possible pollution damage.

352. It should be noted that in these two agreements the operators responsible for the construction and maintenance of the pipelines are private entities.

# (c) Judicial decisions and State practice other than agreements

353. In a few cases, a State engaged in activities entailing risks of damage to other States has unilaterally guaranteed reparation of possible damage. The United States of America has adopted legislation guaranteeing reparation for damage caused by certain nuclear incidents. On 6 December 1974, by Public Law 93-513, adopted in the form of a joint resolution of Congress, the United States assured compensation for damage that might be caused by nuclear incidents involving the nuclear reactor of a United States warship:

Whereas it is vital to the national security to facilitate the ready acceptability of United States nuclear powered warships into friendly foreign ports and harbours; and

Whereas the advent of nuclear reactors has led to various efforts throughout the world to develop an appropriate legal régime for compensating those who sustain damages in the event there should be an incident involving the operation of nuclear reactors; and

<sup>&</sup>lt;sup>250</sup> Agreement of 22 May 1973 between Norway and the United Kingdom relating to the transmission of petroleum by pipeline from the Ekofisk field and neighbouring areas to the United Kingdom.

<sup>251</sup> See footnote 14 above.

Whereas the United States has been exercising leadership in developing legislative measures designed to assure prompt and equitable compensation in the event a nuclear incident should arise out of the operation of a nuclear reactor by the United States as is evidenced in particular by section 170 of the Atomic Energy Act of 1954, as amended; and

Whereas some form of assurance as to the prompt availability of compensation for damage in the unlikely event of a nuclear incident involving the nuclear reactor of a United States warship would, in conjunction with the unparalleled safety record that has been achieved by United States nuclear powered warships in their operation throughout the world, further the effectiveness of such warships: Now, therefore, be it

Resolved by the Senate and House of Representatives of the United States of America in Congress assembled, That it is the policy of the United States that it will pay claims or judgments for bodily injury, death, or damage to or loss of real or personal property proven to have resulted from a nuclear incident involving the nuclear reactor of a United States warship: Provided, That the injury, death, damage, or loss was not caused by the act of an armed force engaged in combat or as a result of civil insurrection. The President may authorize, under such terms and conditions as he may direct, the payment of such claims or judgments from any contingency funds available to the Government or may certify such claims or judgments to the Congress for appropriation of the necessary funds.<sup>252</sup>

354. Public Law 93-513 subsequently supplemented by Executive Order 11918, of 1 June 1976, which provided for *prompt*, adequate and effective compensation in the case of certain nuclear incidents:

By virtue of the authority vested in me by the joint resolution approved December 6, 1974 [Public Law 93-513.88 Stat. 1610.42 U.S.C.2211), and by section 301 of Title 3 of the *United States Code*, and as President of the United States of America, in order that prompt, adequate and effective compensation will be provided in the unlikely event of injury or damage resulting from a nuclear incident involving the nuclear reactor of a United States warship, it is hereby ordered as follows:

Section 1. (a) With respect to the administrative settlement of claims or judgments for bodily injury, death, or damage to or loss of real or personal property proven to have resulted from a nuclear incident involving the nuclear reactor of a United States warship, the Secretary of Defense is designated and empowered to authorize, in accord with Public Law 93-513, the payment, under such terms and conditions as he may direct, of such claims and judgments from contingency funds available to the Department of Defense.

(b) The Secretary of Defense shall, when he considers such action appropriate, certify claims or judgments described in subsection (a) and transmit to the Director of the Office of Management and Budget his recommendation with respect to appropriation by the Congress of such additional sums as may be necessary.

Sec. 2. The provisions of section 1 shall not be deemed to replace, alter, or diminish, the statutory and other functions vested in the Attorney General, or the head of any other agency, with respect to litigation against the United States and judgments and compromise settlements arising therefrom.

Sec. 3. The functions herein delegated shall be exercised in consultation with the Secretary of State in the case of any incident giving rise to a claim of a foreign country or national thereof, and, international

negotiations relating to Public Law 93-513 shall be performed by or under the authority of the Secretary of State.<sup>273</sup>

355. In an exchange of notes between the United States and Spain in connection with the Treaty of friendship and co-operation concluded between them in 1976, the United States gave the assurance:

... that it will endeavour, should the need arise, to seek legislative authority to settle in a similar manner claims for bodily injury, death or damage to or loss of real or personal property proven to have resulted from a nuclear incident involving any other United States nuclear component giving rise to such claims within Spanish territory.<sup>254</sup>

In other words, the United States unilaterally expanded its liability and volunteered, if necessary, to enact legislation expressing such obligation towards Spain.

356. Similarly, a statement made by the United States Department of State in connection with weather modification activities also speaks of advance agreements with potential victim States. In connection with the 1966 hearings before the United States Senate on pending legislation concerning a programme to increase usable precipitation in the United States, the State Department made the following statement:

The Department of State's only concern would be in case the experimental areas selected would be close to national boundaries which might create problems with the adjoining countries of Canada and Mexico. In the event of such possibilities the Department would like to ensure that provision is made for advance agreements with any affected countries before such experimentation took place.<sup>255</sup>

357. In at least one case, a State undertook to guarantee compensation for injuries that might be caused in a neighbouring State by a private company operating in its territory. Thus Canada and the United States conducted negotiations concerning a project for petroleum prospection that a private Canadian company planned to undertake in the Beaufort Sea, off the Mackenzie delta. The project aroused grave concern in the neighbouring territory of Alaska, in particular in respect of the safety measures envisaged and the funds available for compensating potential victims in the United States. As a result of the negotiations, the Canadian company was required to constitute a fund that would ensure payment of the required compensation. The Canadian Government, in turn, undertook to guarantee the payment of compensation.256

## CHAPTER V

## Liability

358. Regardless of any preventive measures that States may take in undertaking activities, they may nevertheless be unable to prevent the occurrence of injuries in the territory of another State. The concept of liability

for injuries to others, in the absence of fault, is not new in domestic law. In the case of certain activities, a causal relationship between the activity and the injury is sufficient to entail liability. This concept in domestic law has

<sup>&</sup>lt;sup>252</sup> Public Law 95-513, *United States Statutes at Large, 1974*, vol. 88, part 2, pp. 1610-1611.

<sup>&</sup>lt;sup>253</sup> Federal Register, vol. 41, No. 108, 3 June 1976, p. 22329.

<sup>&</sup>lt;sup>254</sup> Digest of United States Practice in International Law 1976 (Washington, D.C.), p. 441.

<sup>&</sup>lt;sup>235</sup> Letter addressed by the Department of State to Senator Magnuson, Chairman of the Senate Committee on Commerce, "Weather Modification", *Hearings before the Committee on Commerce*, United States Senate, 89th Congress, 2nd session, part 2, 1966, p. 321.

<sup>&</sup>lt;sup>256</sup> International Canada, Toronto, vol. 7, 1976, pp. 84-85.

been continuously promoted for reasons of morality, social policy and maintenance of public order. In countries with more complex and developed torts law, legislators and courts have begun to recognize that, while some activities are tolerated by law, they "must pay their way". <sup>257</sup> Furthermore, there is the question of who must bear the responsibility to compensate for damage when neither party, under the law, can be blamed. In some instances, strict liability has been imposed upon the party that has initiated the activity as the party that can best bear the loss, or for other reasons connected with social policy. <sup>258</sup>

359. In domestic law, liability for injuries caused by certain permissible activities is termed "strict" or "no-fault" liability. Strict liability has been imposed for a number of activities; some have a longer history than others. Before reviewing the application of a similar principle of liability in relations between States, it may be useful to examine briefly the application of the law of liability for permissible activities. One early application of what is called strict or no-fault liability in domestic

<sup>257</sup> A. A. Ehrenzweig, *Negligence Without Fault* (Berkeley, University of California Press, 1951).

234 William Prosser, an authority on United States torts law, enumerates in his Handbook of the Law of Torts, 4th ed. (St. Paul, Minn., West Publishing Co., 1971), pp. 494-496, instances in which strict liability has been recognized to be relevant, as follows:

"This new policy frequently has found expression where the defendant's activity is unusual and abnormal in the community, and the danger which it threatens to others is unduly great—and particularly where the danger will be great even though the enterprise is conducted with every possible precaution. The basis of the liability is the defendant's intentional behaviour in exposing those in his vicinity to such a risk. The conduct which is dealt with here occupies something of a middle ground. It is conduct which does not so far depart from social standards as to fall within the traditional boundaries of negligence—usually because the advantages which it offers to the defendant and to the community outweigh even the abnormal risk; but which is still so far socially unreasonable that the defendant is not allowed to carry it on without making good any actual harm which it does to his neighbours.

"The courts have tended to lay stress upon the fact that the defendant is acting for his own purposes, and is seeking a benefit or a profit of his own from such activities, and that he is in a better position to administer the unusual risk by passing it on to the public than is the innocent victim. The problem is dealt with as one of allocating a more or less inevitable loss to be charged against a complex and dangerous civilization, and liability is imposed upon the party best able to shoulder it. The defendant is held liable merely because, as a matter of social adjustment, the conclusion is that the responsibility should be his. This modern attitude, which is largely a thing of the last four decades, is of course a far cry from the individualistic viewpoint of the common law courts.

"While such strict liability often is said to be imposed 'without fault', it can scarcely be said that there is less of a moral point of view involved in the rule that one who innocently causes harm, should make it good. . . .

"... The basis of his liability in either case is the creation of an undue risk of harm to other members of the community. It has been said that there is 'conditional fault', meaning that the defendant is not to be regarded as at fault unless or until his conduct causes some harm to others, but he is then at fault, and to be held responsible. If this analysis helps anyone, it is certainly as permissible as another.

"Once the legal concept of 'fault' is divorced, as it has been, from the personal standard of moral wrongdoing, there is a sense in which liability with or without 'fault' must beg its own conclusion. The term requires such extensive definition, that it seems better not to make use of it at all, and to refer instead to strict liability, apart from either wrongful intent or negligence."

law concerns owners of dangerous animals, 259 who were required to protect the community against the attendant risks.

360. The concept of "strict liability" for damage caused by animals was recognized in Roman law. Under the actio de pauperis derived from the XII Tables, an owner was required either to compensate the victim for his loss or to make surrender of the offending animal. 60 The civil codes (CC) of many States, including those of France, Belgium and Italy, also impose strict liability upon the owner or keeper of an animal for the damage it causes, whether the animal was in his keeping or had strayed or escaped. 61 The German Civil Code of 1900, as amended in 1908, provided for exceptions to strict liability only in the case of domestic animals used by the owner in his profession or in his business, or under his care. 62

361. Strict liability is also recognized in respect of owners or keepers of animals in Argentina (CC, art. 1126), Brazil (CC, art. 1527), Colombia (CC, art. 2353), Greece (CC, art. 924), Hungary (CC, art. 353), Mexico (CC, art. 1930), the Netherlands (CC, art. 1404), Poland (CC, art. 431), Switzerland (CC, art. 56) and Yugoslavia.263 Strict liability for damage caused by fire is not widely recognized in domestic law and the elements of fault or negligence are still essential for liability. For example, the French Civil Code, in article 1384, holds a person who possesses by whatever right all or part of a building or personal property in which a fire occurs liable vis-à-vis third persons for damage caused by such fire only if it is proved that it was attributable to his fault or to the fault of a person for whom he is responsible.

362. The theory of strict liability has been incorporated in the Workmen's Compensation Acts in the United States; the employer is strictly liable for injuries to his employees. The policy behind liability for employers is one of "social insurance", and of determining who can best carry the loss when fault is involved.264 These laws, however, do not cover all activities, but in the last few years there has been strong advocacy in the United States for "strict liability" on a broader scale. The strict liability of employers is also recognized in France. Under article 1 of the 1898 law concerning liability for industrial accidents to workers (loi concernant la responsabilité des accidents dont les ouvriers sont victimes dans leur travail), the victim or his representatives are entitled to demand compensation from the employer if, in consequence of the accident, the person concerned is obliged to stop work for more than four days.

<sup>249</sup> Prosser, op. cit., pp. 496 et seq.

<sup>&</sup>lt;sup>260</sup> F. F. Stone, "Liability for damage caused by things", in A. Tunc, ed., *International Encyclopedia of Comparative Law*, vol. XI, *Torts*, part 1 (Nijhoff, The Hague, 1983), chap. 5, p. 11, para. 39.

<sup>&</sup>lt;sup>261</sup> *Ibid.*, p. 12, para. 42.

<sup>&</sup>lt;sup>262</sup> Article 833 of the German Civil Code, *ibid.*, p. 13, para. 47.

<sup>&</sup>lt;sup>263</sup> *Ibid.*, p. 14, paras. 51-52.

<sup>&</sup>lt;sup>264</sup> The concept of workmen's compensation derives from the duties under common law formerly incumbent upon the master for the protection of his servants. See Prosser, *op. cit.*, pp. 525 *et seq*, see also p. 531, footnote 43.

363. Strict liability in the case of abnormally dangerous activities and objects is a comparatively new concept. The leading decision which has influenced domestic law in the United Kingdom and the United States, and which is thought to have given rise to the doctrine of strict liability, is that rendered in the United Kingdom in 1868 in the *Rylands v. Fletcher* case.<sup>265</sup> Justice Blackburn, in the Exchequer Chamber, had stated:

This broad language was later limited by the House of Lords, which stated that the principle applied only to a "non-natural" use of the defendant's lands, as distinguished from "any purpose for which it might in the ordinary course of the enjoyment of land be used".267 More than one hundred subsequent decisions in the United Kingdom have followed the ruling in this case, and strict liability has been confined to things or activities that are "extraordinary", "exceptional" or "abnormal", to the exclusion of those that are "usual and normal". This doctrine does not appear to be applicable to ordinary use of land or to such use as is proper for the benefit of the general community.269 In determining what is a "non-natural use" the English courts appear to have looked not only to the *character* of the thing or activity in question, but also to the place and manner in which it is maintained and its relation to its environment.270

364. In the United States, the *Rylands* v. *Fletcher* precedent was followed by a large number of courts, but rejected by others, among them the courts of New York, New Hampshire and New Jersey. Since the cases before the latter courts bore on customary, natural uses "to which the English courts would certainly never have applied the rule", it was held that the *Rylands* v. *Fletcher* rule had been "misstated" and, as such, must be "rejected in cases in which it had no proper application in the first place"." The *American Restatement of the* 

Law of Torts, established by the American Law Institute, 272 adopted the principle of the Rylands v. Fletcher decision, but confined its application to ultrahazardous activities of the defendant. Section 520 enumerates factors to be considered in determining whether an activity is abnormally dangerous:

- (a) Existence of a high degree of risk of some harm to the person, land or chattels of others;
- (b) Likelihood that the harm that results from it will be great;
- (c) Impossibility of eliminating the risk by the exercise of reasonable care;
- (d) Extent to which the activity is not one of common usage:
- (e) Inappropriateness of the activity to the place where it is carried on;
- (f) Extent to which its value to the community is outweighed by its dangerous attributes.

Ultrahazardous activities have been defined as those that necessarily involve a risk of serious harm to the person, land or chattels of others which cannot be eliminated by the exercise of the utmost care and is not a matter of common usage. This definition has been criticized on the grounds that it is narrower than the ruling in the *Rylands* v. *Fletcher* case and for its emphasis on the *nature* of the activity—"extreme danger and impossibility of eliminating it with all possible care"—rather than on its *relation to its surroundings*.<sup>271</sup> At the same time, the *Restatement* is broader than the ruling in the case, for it does not limit the concept to cases where the material "escapes" from the defendant's land.

365. The rule of strict liability for ultrahazardous activities appears to be provided for in article 1384, paragraph 1, of the French Civil Code,<sup>274</sup> which stipulates:

A person is liable not only for the damage he causes by his own act, but also for that caused by the acts of persons for whom he is responsible or by things that he has under his charge.

Under the rules laid down by this article and first confirmed by the Cour de Cassation in June 1896, it suffices that the plaintiff show that he has suffered damage from an inanimate object in the defendant's keeping for liability to be established:<sup>775</sup>

The Law Reports, Court of Exchequer, vol. 1, 1866, p. 265. In regard to this case and its implications in United States law, see Prosser, op. cii., and Anderson, "The Rylands v. Fletcher doctrine in America: abnormally dangerous, ultrahazardous, or absolute nuisance?", Arizona State Law Journal, Tempe, 1978, p. 99.

<sup>266</sup> Ibid., p. 279.

<sup>&</sup>quot;' The Law Reports, English and Irish Appeal Cases before the House of Lords, vol. 111, 1868, pp. 330, at 338.

<sup>&</sup>lt;sup>268</sup> Prosser, op. cit., p. 506, and footnotes 48, 50 and 51.

<sup>&</sup>lt;sup>260</sup> The House of Lords halted the expansion of that doctrine in a case in which the plaintiff, a government inspector, had been injured by an explosion in the defendant's munitions plant. The judges in this case limited the principle of strict liability to cases in which there had been an *escape* of a dangerous substance from land under the control of the defendant, and two other judges held that the principle was not applicable to *personal injury*. This decision was a sudden departure from the holdings of the leading case; however, it is uncertain whether it has changed the trend towards the application of the principle of strict liability established by the decision in the *Rylands* v. *Fletcher* case (see Prosser, *op. cit.*, p. 506, footnote 52).

of land", The Cambridge Law Journal, London, vol. 111, 1929, p. 387. See also The Law Commission, Civil Liability for Dangerous Things and Activities (London, 1970).

Prosser, Selected Topics on the Law of Torts (Ann Arbor, University of Michigan Law School, 1954), pp. 149-152.

<sup>&</sup>lt;sup>272</sup> See American Law Institute, American Restatement of the Law of Torts (Washington, D.C., 1938), vol. III, chap. 21, sects. 519-524.

<sup>&</sup>lt;sup>271</sup> See Prosser, Selected Topics. . ., p. 158.

<sup>&</sup>lt;sup>274</sup> See H. and L. Mazeaud, Traité théorique et pratique de la responsabilité civile délictuelle et contractuelle, vol. II, 5th ed. established by A. Tunc (Paris, Montchrestien, 1958), p. 342; A. von Mehren and J. R. Gordley, The Civil Law System, 2nd ed. (Boston, Little, 1977), p. 555; F. H. Lawson, Negligence in the Civil Law (Oxford, Clarendon Press, 1950), pp. 46-50; R. Rodière, "Responsabilité civile et risque atomique", Revue internationale de droit comparé paris, 11th year, 1959, p. 505; B. Starck, "The foundation of delictual liability in contemporary French law: An evaluation and a proposal", Tulane Law Review, New Orleans, La., vol. 48, 1973-1974, pp. 1044-1049.

See also Jand'heur v. Galeries belfortaises (1930) (Dalloz, Recueil périodique et critique, 1930 (Paris), part 1, p. 57). The decision in this case also established a presumption of fault on the part of the person having in his charge the inanimate object that has caused the injury.

... A literal interpretation of the article [1384] undoubtedly gives a result comparable to—or rather more far-reaching than—that in *Rylands* v. *Fletcher*, for there is nothing in the words of the article to restrict liability to cases where defendant can be proved to have been negligent in the custody of the things, or even to things which are inherently dangerous....<sup>276</sup>

366. The concept of strict liability also appears in the legal system of the Soviet Union. The 1922 Civil Code of the Russian Socialist Federal Soviet Republic (RSFSR) contained a chapter entitled "Obligations arising from injury caused to another" (arts. 403-415), which correspond to the concepts of torts and delictual liability in the Roman law and common law systems.<sup>277</sup> According to those provisions, "causation" alone suffices to establish liability, the requirement of "fault" being often a cause of injustice. Article 404 provided as follows:

Individuals and enterprises whose activities involve increased hazard for persons coming into contact with them, such as railways, tramways, industrial establishments, dealers in inflammable materials, keepers of wild animals, persons erecting buildings and other structures, and the like, shall be liable for the injury caused by the source of increased hazard, if they do not prove that the injury was the result of *force majeure* or occurred through the intent or gross negligence of the person injured.

367. Once the Roman law maxim cuius commodum eius periculum had been rejected, this article was justified largely in terms of social policy, including the need to promote safety measures and to remedy the hardship that a plaintiff would suffer if the enterprise could escape responsibility merely by showing that all reasonable care had been taken. <sup>278</sup> In 1964, this article was replaced by article 454, based on article 90 of the Fundamental Principles of 1961. Article 454 reads:

Liability for harm caused by a source of increased danger. Organizations and citizens whose activity involves increased danger for those in the vicinity (transport organizations, industrial enterprises, building projects, possessors of motor cars, etc.) must make good the harm caused by the source of increased danger unless they prove that the harm arose in consequence of irresistible force or as a result of the intention of the victim.

368. Recognition of the principle of strict liability is also embodied in the 1964 Polish Civil Code, articles 435 to 437 of which recognize strict liability for damage caused by ultrahazardous activities. The Civil Code of the German Democratic Republic, adopted in 1975, incorporates strict liability in article 344, under which enterprises whose activities lead to increased danger to others are strictly liable for damage resulting from those activities. The same applies to damage resulting from the operation of enterprises as well as the location of things and substances with regard to which increased danger for the life, health and property of others cannot altogether be excluded.<sup>279</sup>

369. Article 178 of the Egyptian Civil Code, article 231 of the Iraqi Civil Code, article 291 of the Jordanian Civil Code and article 161 of the Sudanese Civil Code all establish the strict liability of persons in charge of machines or other objects requiring special care. Article 133 of the Algerian Civil Code goes even farther and recognizes the strict liability of a person in charge of any object when that object causes damage. The Austrian Civil Code (article 1318) and the 1928 Mexician Civil Code (articles 1913 and 1932) also recognize strict liability in respect of dangerous activities or things.

370. The principle of strict liability has been applied in regard to defective products. The policies underlying this practice were stated in the *Escola v. Coca Cola Bottling Co.* case (1944):

... Those who suffer injury from defective products are unprepared to meet its consequences. The cost of an injury and the loss of time or health may be an overwhelming misfortune to the person injured, and a needless one, for the risk of injury can be insured by the manufacturer and distributed among the public as a cost of doing business. It is to the public interest to discourage the marketing of products having defects that are a menace to the public. If such products nevertheless find their way into the market it is to the public interest to place the responsibility for whatever injury they may cause upon the manufacturer, who, even if he is not negligent in the manufacture of the product, is responsible for its reaching the market. However intermitently such injuries may occur and however haphazardly they may strike, the risk of their occurrence is a constant risk and a general one. Against such a risk there should be general and constant protection and the manufacturer is best situated to afford such protection.<sup>230</sup>

This has become the official doctrine in some of the States of the United States. In others, for example New York, it has been supported by additional reasons that were not applicable in the aforementioned case. In its modified form, strict liability in respect of defective products is based on the theory that the manufacturer was in breach of an *implied warranty* to the plaintiff that the article had been properly made.<sup>281</sup> However, a leading United States specialist on the law of torts has strongly objected to this concept of "warranty" as a device that "carries far too much luggage in the way of undesirable complications, and is more trouble than it is worth".<sup>282</sup>

371. In a number of European countries, "strict liability" has not been fully applicable in cases of injury caused by the consumption or use of manufactured products. In the Federal Republic of Germany, for example, the liability of the manufacturer generally requires proof of fault, the concept of strict liability being ruled out. In cases where the injured consumer and the manufacturer are related by contract, the basis for a legal action is the bad performance of the contract and, in this case, the fault of the manufacturer is usually presumed (CC art. 282). Where the parties are not related by contract, action can be based only on the notion of violation of a duty of security to the public

<sup>276</sup> Lawson, op. cit., p. 44.

<sup>&</sup>lt;sup>277</sup> See V. Gsovski, Soviet Civil Law (Ann Arbor, University of Michigan Law School, 1948), vol. 1, p. 489; V. Gsovski and K. Grzybowski, eds., Government, Law and Courts in the Soviet Union and Eastern Europe (London, Stevens, 1959), vol. 2, p. 1175.

See M. J. L. Hardy, "Nuclear liability: The general principles of law and further proposals", *The British Year Book of International Law*, 1960 (London), vol. 36, p. 235; and A. Erh-Soon Tay, "Principles of liability and the 'source of increased danger' in the Soviet law of tort", *The International and Comparative Law Quarterly* (London), vol. 18, 1969, pp. 424-425.

 $<sup>^{\</sup>rm the}$  See also articles 345 and 347 of the 1975 Civil Code of the German Democratic Republic.

<sup>&</sup>lt;sup>280</sup> California Reports 2d Series, vol. 24, p. 453, at p. 462.

<sup>&</sup>lt;sup>281</sup> Goldberg v. Kollsman Instrument Corp. (1963), New York Supplement 2d Series, vol. 240, p. 592.

<sup>&</sup>lt;sup>282</sup> Prosser, *The Law of Torts*, (op. cit., footnote 258 above), p. 656. See also R. M. Sachs, "Negligence or strict product liability: is there really a difference in law or economics?", *Georgia Journal of International and Comparative Law*, Athens, vol. 8, 1978, p. 259; and D. J. Gingerich, "The interagency task force 'blueprint' for reforming product liability tort law in the United States", *ibid.*, p. 279.

(CC art. 831), or on that of the liability of the master for damage done by his servant. Some exceptions to this rule are provided for (*ibid.*), but in general the consumer sues only his fellow contracting party, who is usually not the manufacturer.<sup>283</sup>

372. Similarly, the draft civil code of the Netherlands provides that a person who unknowingly manufactures defective products which constitute a danger to persons or things is liable, if that danger materializes, in the same way as if the defect had been known to him, unless he can prove that the injury was due neither to his own fault or to that of another who, upon his orders, had engaged in the manufacture of the product, nor to the failure of the appliances used by him. However, some Netherlands jurists wish to render the manufacturer responsible for unexplained defects or failures in a product, that is, to hold the manufacturer strictly liable for damage caused by his products.284 In the Soviet Union, the domestic law does not contain provisions relating to the strict liability of the manufacturer of defective products. However, article 454 of the Civil Code of the RSFSR expressed the principle that a person was strictly liable if he caused injury by exceptionally hazardous means. Whether this provision was applicable to manufacturers as opposed to "owners" is unclear.285

373. In the United States of America, the principle of strict liability was also apparent in the Aeronautics Act of 1922. That legislation, adopted in whole or in part by twenty-four States of the Union, provides for the "absolute liability" of the owners of aircraft for injuries to persons or property on land or water caused by the ascent, descent or flight of the aircraft or the dropping or falling of any object therefrom, unless the injury was caused in whole or in part by the negligence of the person injured, or of the owner or bailee of the property damaged. The object of the Act was to place the liability for damage caused by accidents of aircraft upon operators, and to protect innocent victims, even though the accident might not be attributable to the fault of the operator. 287

374. A number of Latin American and European countries have also adopted the principle of strict liability, often similar to the 1933 and 1952 Rome Conventions for accidents involving aircraft. Argentina, Guatemala, Honduras and Mexico are among the Latin American countries which have imposed strict liability based on the concept of risk, and among European

countries having done the same are Italy, Spain, Denmark, Sweden, Norway, Finland, Switzerland, France and the Federal Republic of Germany.<sup>288</sup>

375. The rule of strict liability has also been applied in respect of owners and operators of sources of power for damage caused by the production or storage of electricity. In this area, the concept of strict liability corresponds to the notion that "electricity is a thing in one's keeping" (France, CC art. 1384), or to the notion that "the owner is presumed to be at fault" (Argentina, CC art. 1135), or to the notions of "dangerous things" (United States and United Kingdom), or of "dangerous activities" (Italy, CC art. 2050).<sup>289</sup>

376. Originally, nuisance meant nothing more than harm or annoyance.<sup>290</sup> The principle of strict liability has been applied to cases of absolute nuisance, without regard to the defendant's intent or precautions. There has been little discussion of nuisance in the context of liability. The reasons for this have been described as follows:

. . . One reason is that nuisance suits frequently have been in equity, seeking an injunction, so that the question is not so much one of the nature of the defendant's conduct as of whether he shall be permitted to continue it. Even where the action is one for damages, it usually has been brought after long continuance of the conduct and repeated requests to stop it; and whatever may have been his state of mind in the first instance, the defendant's persistence after notice of the harm he is doing takes on the aspects of an intentional tort. Another reason is that in nuisance cases the threat of future harm may in itself amount to a present interference with the public right or the use and enjoyment of land, so that the possible bases of liability tend to merge and become more or less indistinguishable. Nevertheless it is quite clear that a substantial part of the law of nuisance rests upon neither wrongful intent nor negligence.<sup>291</sup>

377. It has been claimed that the concept of absolute nuisance is closely related to the rule in Rylands v. Flet*cher.* To distinguish that rule, some have claimed that it applies to conduct which is not wrongful in itself, and so will not be prohibited or enjoined in advance, but will make the defendant strictly liable if it causes actual damage; whereas a nuisance is in itself wrongful, and may always be enjoined. Others have rejected this distinction on the grounds that there are no cases or decisions to sustain it.292 It has also been stated that the concept of absolute nuisance and the Rylands rule are related to one another as intersecting circles; they have a large area in common, but nuisance is the *older tort* and its historical development has limited it to two kinds of interference; with the public interest and with the enjoyment of land, excluding such other damage as personal injuries not connected with either. Thus the underlying principles appear to be the same in each case and they are indistinguishable except by the accident of their history.293

378. Many legal systems have shown a persistent tendency to recognize the concept of strict liability while

<sup>&</sup>lt;sup>283</sup> See Stone, *loc. cit.* (footnote 260 above), p. 74, paras. 265-266.

<sup>&</sup>lt;sup>284</sup> *Ibid.*, pp. 73-74, para. 264. See also Italian practice, *ibid.*, p. 74, para. 268.

<sup>&</sup>lt;sup>285</sup> See F. A. Orban, "Product liability: a comparative legal restatement—foreign national law and EEC directive", *Georgia Journal of International and Comparative Law*, vol. 8, 1978, pp. 371-373.

<sup>&</sup>lt;sup>286</sup> United States of America, *Uniform Laws Annotated*, vol. II, pp. 159-171. This act was withdrawn in 1938 by the National Conference of Commissioners on Uniform State Laws and replaced by other texts drafted by that body, imposing substantially the same limited absolute liability. See *Handbook of National Conference of Commissioners on Uniform State Laws*, 1938, p. 318, and Uniform Aviation Liability Act, art. II, paras. 201-202.

<sup>&</sup>lt;sup>287</sup> See E. C. Sweeney, "Is special aviation liability legislation essential", The Journal of Air Law and Commerce, Chicago, Ill., vol. 19, p. 166; Prentiss et al. v. National Airlines, Inc., 112 Federal Supplement, pp. 306 and 312.

<sup>&</sup>lt;sup>288</sup> See Stone, *loc. cit.* (footnote 360 above), pp. 45-46, paras. 178-181.

<sup>&</sup>lt;sup>289</sup> *Ibid.*, pp. 48-49, paras. 193-197.

<sup>&</sup>lt;sup>290</sup> Prosser, Selected Topics . . . (see footnote 271 above), p. 164.

<sup>&</sup>lt;sup>291</sup> *Ibid.*, p. 166. See also P. H. Winfield, "The myth of absolute liability", *The Law Quarterly Review*, London, vol. 42, 1926, p. 37.

<sup>&</sup>lt;sup>292</sup> Prosser, Selected Topics . . ., p. 172.

<sup>&</sup>lt;sup>293</sup> Ibid., p. 177. See also Winfield, "Nuisance as a tort", The Cambridge Law Journal, London, vol. 1V, 1932, p. 195.

maintaining liability dependent on "fault" as the general principle. Originally, strict liability was recognized in legal systems in connection with keeping animals or causing fire and was based on reasons of morality.

- 379. As legal systems developed, the concept of strict liability appears to have been introduced as a means to accommodate diverse social interests. The reason for the application of this concept to relations between employers and employees under certain conditions (ultrahazardous activities, product liability, air transport, etc.) has often been social necessity rather than morality.
- 380. Strict liability as a legal concept now appears to have been accepted by most legal systems, especially those of technologically developed countries with more complex torts laws. The extent of activities subject to strict liability may differ; in some countries it is more limited than in others. The legal basis for strict liability also varies from "presumed fault" to the notion of "risk", "dangerous activity involved", etc. But it is evident that strict liability is a principle common to a sizable number of countries with different legal systems, which have had to regulate activities to which this principle is relevant. While States may differ as to the particular application of this principle, their understanding and formulation of it are substantially similar.
- 381. From an examination of municipal laws, the following characteristics of strict liability emerge:
- (1) The concept of strict liability has been stipulated in the civil code or in general terms in the decisions of domestic courts, as opposed to special legislation. This demonstrates the importance of strict liability as a general legal principle and its acceptance by many States with diverse legal systems.
- (2) In municipal law, the definition of strict liability is characterized by its lack of reference to fault; "fault" is not a criterion of liability.
- (3) It has been recognized by municipal law that it would be inequitable and unduly harsh to the public if operators of hazardous activities were permitted to avoid liability for damage caused by their industry under the rules of "fault liability".
- (4) Municipal law limits strict liability to "the kind of harm, the risk of which makes the activity abnormally dangerous". The RSFSR Supreme Court held that a railroad was not liable, under section 404 of the Civil Code, for the injury sustained by a passenger "who was the victim of a hold-up on the train, because such injury was not caused by the increased hazard incidental to railways as a special kind of transportation". There must be a causal nexus between the activity and the harm from which relief is sought and, even if causation is admitted, liability may be avoided under certain conditions: 296

- (a) When the victim assumes the risk of harm; <sup>347</sup>
- (b) When the victim intentionally suffers the damage; however, the contributory negligence of the plaintiff does not exempt the person who carries on an abnormally dangerous activity from strict liability; how
- (c) When the damage is caused by an irresistible force or in a case of *force majeure*. 100
- The introduction and application of the concept of liability in inter-State relations, on the other hand, is relatively newer and less developed than in domestic law. One reason for this late start may have been the fact that States were not so involved in activities which could injure other States and their nationals. The difficulties in accommodating the concept of liability with other well-established concepts of international law, such as domestic jurisdiction and territorial sovereignty, should not be ignored. Of course, the development of no-fault liability in domestic law has faced similar difficulties. But social necessity, in many States, has led to accommodating this new legal concept with others in a way which serves social policies and public order. In inter-State relations, activities which may cause injuries to others, beyond the territorial jurisdiction or control of the acting State, have in most cases been singled out, and the liability issue has been subject to agreements between States. This may be more akin to laws on liability, such as those relating to the liability of keepers or owners of animals, product liability, employers' liability, etc. But in State practice there are nevertheless a few references to a broader concept of liability, such as the principle of due care, good neighbourliness, etc.
- 383. It is not suggested here that the development of the liability concept in State practice has the same content and procedure as in domestic law. The domestic law references are mentioned only to provide guidelines when appropriate for understanding the concept of liability and its development.

## (a) Multilateral agreements

384. Sometimes the main purpose of a multilateral agreement is to resolve the question of liability and compensation which may be involved in certain activities without limiting or hindering the activities themselves. It seems to have been accepted in principle that such activities should be authorized regardless of the injuries they may cause. Agreements have been concluded only to deal with liability, compensation and jurisdictional questions which may arise from an accident, as in the 1976 Convention on Civil Liability for Oil Pollution Damage Resulting from Exploration for and Exploitation of Seabed Mineral Resources. The preamble to the Convention reads:

<sup>&</sup>lt;sup>294</sup> American Law Institute, Restatement (Second) of Torts, art. 519, 2 (Tentative Draft No. 10, 1964).

<sup>&</sup>lt;sup>295</sup> Gsovski, op. cit. (footnote 277 above), p. 506.

<sup>&</sup>lt;sup>296</sup> See J. M. Kelsen, "State responsibility and the abnormally dangerous activity", *Harvard International Law Journal*, Cambridge, Mass., vol. 13, 1972, pp. 230-231.

<sup>&</sup>lt;sup>29</sup> American Law Institute, Restatement (Second) of Torts, sect. 523 (Tentative Draft No. 10, 1964).

<sup>&</sup>lt;sup>298</sup> Ibid., sect. 524, 2; RSFSR, CC art. 454 (see Tay, loc. cit. (footnote 278 above), p. 441.

<sup>&</sup>lt;sup>299</sup> American Law Institute, Restatement (Second) of Torts, sect. 545, 1 (Tentative Draft No. 10, 1964).

<sup>&</sup>lt;sup>300</sup> Idem; RSFSR, CC art. 454 (see Tay, loc. cit., pp. 441-442); France, CC art. 1384, and Jand'heur v. Galeries belfortaises case (1930) (see footnote 275 above).

The States Parties to this Convention.

Conscious of the dangers of oil pollution posed by the exploration for, and exploitation of, certain seabed mineral resources,

Convinced of the need to ensure that adequate compensation is available to persons who suffer damage caused by such pollution,

Desiring to adopt uniform rules and procedures for determining questions of liability and providing adequate compensation in such cases.

. .

385. Some other conventions refer to the concept of liability but do not resolve the questions of compensation and jurisdiction. For example, the 1978 Kuwait Regional Convention for Co-operation on the Protection of the Marine Environment from Pollution provides that the contracting States shall co-operate in formulating rules and procedures for civil liability and compensation for damage resulting from pollution of the marine environment, but it does not stipulate those rules and procedures. Article XIII of the Convention reads:

#### Article XIII. Liability and compensation

The Contracting States undertake to co-operate in the formulation and adoption of appropriate rules and procedures for the determination of:

- (a) civil liability and compensation for damage resulting from pollution of the marine environment, bearing in mind applicable international rules and procedures relating to those matters; and
- (b) liability and compensation for damage resulting from violation of obligations under the present Convention and its protocols.
- 386. Similar requirements are stipulated in the 1976 Convention for the Protection of the Mediterranean Sea against Pollution and in the 1974 Convention on the Protection of the Marine Environment of the Baltic Sea Area. Article 12 of the Convention for the Protection of the Mediterranean Sea reads:

## Article 12. Liability and compensation

The Contracting Parties undertake to co-operate as soon as possible in the formulation and adoption of appropriate procedures for the determination of liability and compensation for damage resulting from the pollution of the marine environment deriving from violations of the provisions of this Convention and applicable protocols.

Article 17 of the Convention on the Protection of the Marine Environment of the Baltic Sea Area reads:

## Article 17. Responsibility for damage

The Contracting Parties undertake, as soon as possible, jointly to develop and accept rules concerning responsibility for damage resulting from acts or omissions in contravention of the present Convention, including, *inter alia*, limits of responsibility, criteria and procedures for the determination of liability and available remedies.

387. Article X of the 1972 Convention on the Prevention of Marine Pollution by Dumping of Wastes and Other Matter recommends that the contracting States develop rules and procedures on liability, but is based on a different assumption, namely, the existence of principles of international law on State responsibility for damage to the environment of other States or to any other area of the environment. Thus article X stipulates that procedures for the assessment of liability and settlement of disputes should be formulated in accordance with those principles of international law:

### Article X

In accordance with the principles of international law regarding State responsibility for damage to the environment of other States or to any other area of the environment, caused by dumping of wastes and other matter of all kinds, the Contracting Parties undertake to develop procedures for the assessment of liability and the settlement of disputes regarding dumping.

388. Article 235 of the 1982 United Nations Convention on the Law of the Sea uses a different and a more complex language. It provides that States shall cooperate to "implement" the existing international law and its future development relating to responsibility and liability for assessment of compensation for damage and the settlement of disputes:

### Article 235. Responsibility and liability

- 1. States are responsible for the fulfilment of their international obligations concerning the protection and preservation of the marine environment. They shall be liable in accordance with international law.
- 2. States shall ensure that recourse is available in accordance with their legal systems for prompt and adequate compensation or other relief in respect of damage caused by pollution of the marine environment by natural or juridical persons under their jurisdiction.
- 3. With the objective of assuring prompt and adequate compensation in respect of all damage caused by pollution of the marine environment, States shall co-operate in the implementation of existing international law and the further development of international law relating to responsibility and liability for the assessment of and compensation for damage and the settlement of related disputes, as well as, where appropriate, development of criteria and procedures for payment of adequate compensation, such as compulsory insurance or compensation funds.

This article assumes the existence of principles of international law governing issues of liability.

389. In other conventions, finally, the contracting parties are merely requested to co-operate and to develop rules on liability and compensation in conformity with international law. For example, article 14 of the 1983 Convention for the Protection and Development of the Marine Environment of the Wider Caribbean Region reads:

## Article 14. Liability and compensation

The Contracting Parties shall co-operate with a view to adopting appropriate rules and procedures, which are in conformity with international law, in the field of liability and compensation for damage resulting from pollution of the Convention area.

## (b) Bilateral agreements

390. Few bilateral agreements are designed to resolve the question of liability for extraterritorial injuries. Some make no reference to anything bearing on liability; others make general references that may be interpreted to mean that other questions, including that of liability, will be dealt with under a different formula. The 1983 Agreement between the United States of America and Mexico, <sup>301</sup> for example, states in article 17 that the Agreement does not affect the rights and obligations of the parties under international agreements to which they are party or existing or future agreements between the parties themselves:

#### Article 17

Nothing in this Agreement shall be construed to prejudice other existing or future agreements concluded between the two Parties, or affect the rights and obligations of the Parties under international agreements to which they are a party.

391. In some agreements it is expressly stated that the question of liability is not examined. For example, the

<sup>&</sup>lt;sup>301</sup> See footnote 16 above.

1975 Agreement between Canada and the United States of America relating to weather modification activities<sup>302</sup> excludes the resolution of liability questions. After the formulation of certain procedures regarding weather modification activities affecting the contracting States, it is provided, in article VII, that the Agreement should not be construed as affecting the liability and responsibility issues arising between two countries, nor as implying the existence of any generally applicable rule of international law:

#### Article VII

Nothing herein relates to or shall be construed to affect the question of responsibility or liability for weather modification activities, or to imply the existence of any generally applicable rule of international law.

This article neither confirms nor denies the existence of any principles of liability accepted by the two States. Nevertheless, the Agreement recognizes that such a question may be relevant and may be raised in connection with weather modification activities.

392. In the 1974 Agreement between the United States and Canada regarding certain rocket launches, the two Governments agree, in the event of loss or damage resulting from those launches, to consult each other promptly, prior to the settlement of any claim, in order to expedite such claims in accordance with international law and the domestic law of each State. The relevant paragraph of the Agreement (note I, third para.) reads:

The Embassy [of the United States] has the honour to propose that, in the event of such loss or damage, the Government of the United States and the Government of Canada shall consult promptly, and in any case prior to the settlement of any claim arising out of these launches, with a view to arriving at an expeditious and mutually acceptable disposition of such claim, in accordance with international law and the domestic law of each state. . . .

393. The 1909 Treaty relating to the boundary waters between Canada and the United States of America<sup>104</sup> contains a different language on liability. Article II refers to *legal remedies* available to parties suffering injuries caused by activities occurring in the boundary waters under the territorial control of the other State. In such cases, under article II, the *injured parties are entitled to the local legal remedies available in the country where the activities occurred*:

#### Article II

Each of the High Contracting Parties reserves to itself or to the several State Governments on the one side and the Dominion or Provincial Governments on the other, as the case may be, subject to any treaty provisions now existing with respect thereto, the exclusive jurisdiction and control over the use and diversion, whether temporary or permanent, of all waters on its own side of the line which in their natural channels would flow across the boundary or into boundary waters; but it is agreed that any interference with or diversion from their natural channel of such waters on either side of the boundary, resulting in any injury on the other side of the boundary, shall give rise to the same rights and entitle the injured parties to the same legal remedies as if such injury took place in the country where such diversion or interference occurs;\* but this provision shall not apply to cases already existing or to cases expressly covered by special agreement between the parties hereto.

It is understood, however, that neither of the High Contracting Parties intends by the foregoing provision to surrender any right which it may have to object to any interference with or diversions of waters on the other side of the boundary the effect of which would be productive of material injury to the navigation interests on its own side of the boundary.

Hence the applicable rules on liability are either the domestic law of Canada or that of the United States.

394. The 1922 Treaty between Germany and Denmark<sup>30</sup> provides remedies for individuals who have suffered injuries. Article 26 states that any person who suffers damage as a result of a new water regulation or alteration has the *right to claim full compensation* from the *person* who *benefited* from those regulations. The article does not refer to any particular principles of domestic or international law on liability; it states only that the matter shall be decided by the Frontier Water Commission. The relevant paragraph of article 26 reads:

#### Article 26. Compensation for damage caused by regularization

Any person who suffers loss or damage in consequence of the regularization or of the alteration in the condition of the watercourse occasioned by such regularization has the right to claim full compensation from the person who benefits by the work in question. The matter shall be decided by the Frontier Water Commission.

# (c) Judicial decisions and State practice other than agreements

395. The concept of liability for damage caused by an activity beyond the territorial jurisdiction or control of the acting State appears to have been developed through State practice to a limited extent for some potentially harmful activities. Some sources refer to the concept in general terms, leaving its content and procedure for implementation to future developments. Other sources deal with the concept of liability only in a specific case.

In the past, liability has been considered as an outgrowth of failure to exercise "due care" or "due diligence". In determining whether there has been a failure to exercise due diligence, the test has been the balancing of interests. This criterion is similar to that used in determining harm and the permissibility of harmful activities, given the assessment of their impact. Liability for failure to exercise due care was established as early as 1872, in the Alabama case. In that dispute between the United States and the United Kingdom over the alleged failure of the United Kingdom to fulfil its duty of neutrality during the American Civil War, both sides attempted to articulate what "due diligence" entailed. The United States argued that due diligence was proportioned to the *magnitude* of the subject and to the dignity and strength of the power which was to exercise

#### Due diligence

The rules of the treaty,[\*] said the Case of the United States, imposed upon neutrals the obligation to use *due diligence* to prevent certain acts. These words were not regarded by the United States as changing in any respect the obligations imposed by international law. "The United States", said the Case, "understands that the diligence which is called for by the rules of the treaty of Washington is a *due* diligence—that is, a diligence proportioned to the magnitude of the subject and to the dignity and strength of the power which is to exer-

see footnote 15 above.

<sup>103</sup> See footnote 84 above.

<sup>304</sup> See footnote 35 above.

<sup>101</sup> See footnote 59 above.

cise it; a diligence which shall, by the use of active vigilance, and of all the other means in the power of the neutral, through all stages of the transaction, prevent its soil from being violated; a diligence that shall in like manner deter designing men from committing acts of war upon the soil of the neutral against its will, and thus possibly dragging it into a war which it would avoid; a diligence which prompts the neutral to the most energetic measures to discover any purpose of doing the acts forbidden by its good faith as a neutral, and imposes upon it the obligation, when it receives the knowledge of an intention to commit such acts, to use all the means in its power to prevent it. No diligence short of this would be "due", that is, commensurate with the emergency or with the magnitude of the results of negligence\*....<sup>306</sup>

[\* Treaty of Washington of 8 May 1871 whereby the United Kingdom and the United States of America agreed to submit their dispute to arbitration.]

397. By contrast, the British Government argued that, in order to show lack of due diligence and invoke the liability of a State, it must be proved that there had been a failure to use, for the prevention of a harmful act, such care as Governments *ordinarily* employed in their domestic concerns:

... it was necessary to show that there had been "a failure to use, for the prevention of an act which the government was bound to endeavour to prevent, such care as governments ordinarily employ in their domestic concerns, and may reasonably be expected to exert in matters of international interest and obligation"."

The tribunal referred to "due diligence" as a duty arising "in exact proportion to the risks to which either of the belligerents may be exposed, from a failure to fulfil the obligations of neutrality on their part". <sup>308</sup> Thus, due diligence is a function of the circumstances of the activity.

398. Later State practice *appears* not to have dealt so much with State liability arising out of failure to exercise due care, except in the area of the protection of aliens. These categories of claims include nationalization and confiscation of foreign properties, police protection and safety of foreigners, etc., which have been excluded from this study.

399. In the claim against the USSR for damage caused by the crash of the Soviet satellite *Cosmos 954* on Canadian territory in January 1978, Canada referred to the general principle of the law of "absolute liability" for injury resulting from activities with a high degree of risk:

The standard of absolute liability for space activities,\* in particular activities involving the use of nuclear energy\* is considered to have become a general principle of international law.\* A large number of States, including Canada and the Union of Soviet Socialist Republics, have adhered to this principle as contained in the 1972 Convention on International Liability for Damage caused by Space Objects.\* The principle of absolute liability applies to fields of activities having in common a high degree of risk. It is repeated in numerous international agreements and is one of "the general principles of law recognized by civilized nations" (Article 38 of the Statute of the International Court of Justice). Accordingly, this principle has been accepted as a general principle of international law.\*

400. Similarly, in the *Trail Smelter* case, the permanent régime called for compensation for injury to United States interests arising from fume emissions *even* 

if the smelting activities conformed fully to the permanent régime as defined in the decision:

The Tribunal is of opinion that the prescribed régime will probably remove the causes of the present controversy and, as said before, will probably result in preventing any damage of a material nature occurring in the State of Washington in the future.

But since the desirable and expected result of the régime or measure of control hereby required to be adopted and maintained by the Smelter may not occur, and since in its answer to Question No. 2, the Tribunal has required the Smelter to refrain from causing damage in the State of Washington in the future, as set forth therein, the Tribunal answers Question No. 4 and decides that on account of decisions rendered by the Tribunal in its answers to Question No. 2 and Question No. 3 there shall be paid as follows: (a) if any damage as defined under Question No. 2 shall have occurred since October 1, 1940, or shall occur in the future, whether through failure on the part of the Smelter to comply with the regulations herein prescribed or notwithstanding the maintenance of the régime, an indemnity shall be paid for such damage but only when and if the two Governments shall make arrangements for the disposition of claims for indemnity under the provisions of article XI of the Convention;[\*] (b) if as a consequence of the decision of the Tribunal in its answers to Question No. 2 and Question No. 3, the United States shall find it necessary to maintain in the future an agent or agents in the area in order to ascertain whether damage shall have occurred in spite of the régime prescribed herein, the reasonable cost of such investigations not in excess of \$7,500 in any one year shall be paid to the United States as a compensation, but only if and when the two Governments determine under article XI of the Convention that damage has occurred in the year in question, due to the operation of the Smelter, and "disposition of claims for indemnity for damage" has been made by the two Governments; but in no case shall the aforesaid compensation be payable in excess of the indemnity for damage; and further it is understood that such payment is hereby directed by the Tribunal only as a compensation to be paid on account of the answers of the Tribunal to Question No. 2 and Question No. 3 (as provided for in Question No. 4) and not as any part of indemnity for the damage to be ascertained and to be determined upon by the two Governments under article XI of the Convention.310

401. The standard for imposing liability on the State under whose control an injurious condition exists is even more obfuscated in the decision of 9 April 1949 in the *Corfu Channel* case (*merits*). There the International Court of Justice found that Albania had known or should have known of the mines lying within its territorial waters in sufficient time to give warning to other States and their nationals. The Court found that:

In fact, nothing was attempted by the Albanian authorities to prevent the disaster. These grave omissions involve the international responsibility\* of Albania.

The Court therefore reaches the conclusion that Albania is responsible under international law for the explosions which occurred on October 22nd, 1946, in Albanian waters, and for the damage and loss of human life which resulted from them, and that there is a duty upon Albania to pay compensation to the United Kingdom.<sup>311</sup>

Owing to the difficult and circumstantial nature of the proof of Albania's knowledge of the injurious condition, it is unclear whether liability was based on a breach of the duty of due care in warning other international actors or on a standard of "strict liability" without regard to the concept of due care.

<sup>&</sup>lt;sup>106</sup> J. B. Moore, International Arbitrations to which the United States has been a Party (Washington, D.C., 1898), vol. I, pp. 572-573.

<sup>&#</sup>x27;07 Ibid., p. 610.

<sup>10</sup>x Ibid., p. 654.

<sup>&</sup>lt;sup>104</sup> See *International Legal Materials* (Washington, D.C.), vol. 18, p. 907, para. 22.

<sup>1\*</sup> Convention of 15 April 1935 between the United States of America and Canada for the Final Settlement of the Difficulties arising through Complaints of Damage done in the State of Washington by Furnes discharged from the Smelter of the Consolidated Mining and Smelting Company, Trail, British Columbia.1

<sup>&</sup>lt;sup>110</sup> United Nations, Reports of International Arbitral Awards, vol. 111, pp. 1980-1981.

<sup>311</sup> I.C.J. Reports 1949, p. 23.

402. In the same judgment, the International Court of Justice made some general statements regarding State liability which are of considerable importance. In one passage, the Court stated that it was "every State's obligation\* not to allow knowingly its territory to be used for acts contrary to the rights of other States". 112 It should be noted that in this passage the Court was making a general statement of law and policy, not limited or narrowed to any specific case. When the Court renders a decision in a case in accordance with Article 38 of its Statute, it may also declare general statements of law. The aforementioned passages are among such statements. It may therefore be concluded that, while the Court's decision addressed the point debated by the parties in connection with the Corfu Channel, it also stressed a more general issue. It was a declaratory general statement regarding the conduct of any State which might cause extraterritorial injuries.

403. In the Lake Lanoux case, on the other hand, the tribunal, responding to the allegation of Spain that the French projects would entail an abnormal risk to Spanish interests, stated that only failure to take all necessary safety precautions would have entailed France's responsibility if Spanish rights had in fact been infringed. The tribunal stated:

... The question was lightly touched upon in the Spanish counter memorial, which underlined the "extraordinary complexity" of procedures for control, their "very onerous" character, and the "risk of damage or of negligence in the handling of the watergates, and of obstruction in the tunnel". But it has never been alleged that the works envisaged present any other character or would entail any other risks than other works of the same kind which today are found all over the world. It has not been clearly affirmed that the proposed works would entail an abnormal risk in neighbourly relations or in the utilization of the waters. As we have seen above, the technical guarantees for the restitution of the waters are as satisfactory as possible. If, despite the precautions that have been taken, the restitution of the waters were to suffer from an accident, such an accident would be only occasional and, according to the two Parties, would not constitute a violation of article 9.313

In other words, responsibility would not arise as long as all possible precautions against the occurrence of the injurious event had been taken. Although the authority of the tribunal was limited by the parties to the examination of compatibility of French activities on the Carol River with a treaty, the tribunal also touched upon the question of dangerous activities. In the passage quoted above, the tribunal stated: "It has not been clearly affirmed that the proposed works [by Francel would entail an abnormal risk in neighbourly relations or in the utilization of the waters." This passage may be interpreted as meaning that the tribunal was of the opinion that abnormally dangerous activities constituted a special problem, and that, if Spain had established that the proposed French project would entail an abnormal risk of injury to Spain, the decision of the tribunal might have been different.

## A. Balancing of interests

405. State practice shows that the concept of balance of interests is also involved in questions of liability and

compensation. The role of balance of interests in the determination of liability and compensation is sometimes the same as in the assessment of activities. However, in some agreements a clear distinction has been made between the two roles. Terms such as "equitable compensation", "fair compensation", "limited liability", etc., as used in State practice, clearly refer to the balancing of interests in determining liability and compensation.

406. It is obvious that the concept of balance of interests may vary according to circumstances. Negotiations concerning liability and compensation must not obstruct the undertaking and future development of commercial, industrial and technological activities which have become indispensable to and inseparable from human society. The concept of "limited liability", for example, has been developed to balance interests in connection with such activities. In the case of certain other activities, considerations of balance of interests and priorities may be different; the interests of injured parties, for example, may prevail over the continuation of certain potentially harmful activities.

### (a) Multilateral agreements

407. In at least two multilateral conventions, the concept of balance of interests is expressly stated in the preamble with reference to liability and compensation. The preamble to the 1960 Convention on Third Party Liability in the Field of Nuclear Energy expresses the desire of the contracting parties to ensure "adequate" and "equitable" compensation for individuals who suffer injury caused by nuclear incidents, without hindering the development of peaceful uses of nuclear energy. The relevant paragraphs of the preamble provide:

[The Contracting Governments,]

Desirous of ensuring adequate and equitable compensation for persons who suffer damage caused by nuclear incidents whilst taking the necessary steps to ensure that the development of the production and uses of nuclear energy for peaceful purposes is not thereby hindered;

Convinced of the need for unifying the basic rules applying in the various countries to the liability incurred for such damage, whilst leaving these countries free to take, on a national basis, any additional measures which they deem appropriate, including the application of the provisions of this Convention to damage caused by incidents due to ionizing radiations not covered therein;

408. The preamble to the 1952 Convention on Damage Caused by Foreign Aircraft to Third Parties on the Surface is even more explicit in stating its aim, namely, the balance of interests in determining liability and compensation. It states the desire of the contracting parties to ensure "adequate" compensation for injured persons, while limiting in a reasonable manner the extent of the liabilities incurred for such damage. The relevant paragraph of the preamble reads:

The States signatory to this Convention,

Moved by a desire to ensure adequate compensation for persons who suffer damage caused on the surface by foreign aircraft, while limiting in a reasonable manner the extent of the liabilities incurred for such damage in order not to hinder the development of international civil air transport, . . .

409. The 1976 Convention on Limitation of Liability for Maritime Claims is implicitly based on the accommodation and balancing of the interests of injured par-

<sup>312</sup> Ibid., p. 22.

<sup>&</sup>quot;International Law Reports (1957) (see footnote 67 above), pp. 123-124, para. 6 of the award.

ties with the interest of the larger community in protecting and promoting maritime transportation essential to the present-day world economy. Paragraph 1 of article 2 itemizes claims subject to limitation, as follows:

Article 2. Claims subject to limitation

- 1. ...
- (a) claims in respect of loss of life or personal injury or loss of or damage to property (including damage to harbour works, basins and waterways and aids to navigation), occurring on board or in direct connection with the operation of the ship or with salvage operations and consequential loss resulting therefrom;
- (b) claims in respect of loss resulting from delay in the carriage by sea of cargo, passengers or their luggage;
- (c) claims in respect of other loss resulting from infringement of rights other than contractual rights, occurring in direct connection with the operation of the ship or salvage operations;
- (d) claims in respect of the raising, removal, destruction or the rendering harmless of a ship which is sunk, wrecked, stranded or abandoned, including anything that is or has been on board such ship;
- (e) claims in respect of the removal, destruction or the rendering harmless of the cargo of the ship;
- (f) claims of a person other than the person liable in respect of measures taken in order to avert or minimize loss for which the person liable may limit his liability in accordance with this Convention, and further loss caused by such measures.

Paragraph 2 of the same article provides that limitation on liability applies to these claims even if brought by way of recourse or for indemnity under a contract or otherwise. Only claims under subparagraphs (d), (e) and (f) shall not be subject to limitation of liability to the extent that they relate to remuneration under a contract with the person liable. Paragraph 2 provides:

- 2. Claims set out in paragraph 1 shall be subject to limitation of liability even if brought by way of recourse or for indemnity under a contract or otherwise. However, claims set out under paragraph 1 (d), (e) and (f) shall not be subject to limitation of liability to the extent that they relate to remuneration under a contract with the person liable.
- 410. This policy of accommodation and balancing of interests appears again in the 1969 International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties, in which coastal States are granted competence to take unilateral action on the high seas to protect their own interests. This Convention accommodates the interests of the coastal State, by granting it the right to take unilateral action, with those of the flag State, the latter being entitled to compensation if the measures taken by the coastal State go beyond what is reasonable. The security interests of the flag State are also taken into account, in article I, paragraph 2, since the coastal State may take no action, under the Convention, if the ship involved in the casualty is a warship or other ship owned or operated by a State and used for the time being on government noncommercial service:

#### Article I

2. However, no measures shall be taken under the present Convention against any warship or other ship owned or operated by a State and used, for the time being, only on government noncommercial service.

## (b) Bilateral agreements

411. In bilateral agreements, the concept of balancing interests to determine liability and compensation is the same as that used to determine harm for purposes of

prior negotiation and consultation. Howevever, the language used in connection with the balancing of interests in determining liability is less explicit in bilateral than in multilateral agreements. The reason may lie in the very nature of multilateral and bilateral agreements. Multilateral agreements are of a more general nature, dealing as they do with a more general set of activities, referring to more than two parties and accommodating various interests. To take account of all these factors it is necessary to use explicit language, for example, in balancing interests. By contrast, bilateral agreements are less complex, since they deal with a more precise subject; sometimes, indeed, a large part of the agreement sets out a detailed procedure for accommodating the two parties' interests when determining liability and compensation.

412. For example, article 27 of the 1922 Treaty between Germany and Denmark<sup>314</sup> is in fact based on the concept of balance of interests. Without mentioning the concept as such, the Treaty provides that the cost of upkeep, if increased by a new regulation of water-courses, shall be paid by those who benefit from the regulation regardless whether they previously shared the cost of upkeep or not:

Article 27. Liability for upkeep after regularization

If the cost of upkeep is increased by the regularization of a watercourse, the increase shall be apportioned among all the proprietors to whom the regularization is of use or advantage, regardless of the fact whether they previously shared in the cost of upkeep or not.

The first paragraph of article 26 of the treaty also bears on the concept of balance of interests (see para. 394 above).

## B. Operator's liability

- 413. In activities conducted primarily by private entities, liability rests with the operator. Some activities are conducted by both private entities and government agencies, but in the latter case the régime of liability is directly based on that applying to the private operator, in other words, the relevant government agency (the operator) has the same liability as a private operator. This is true in particular in connection with activities involving the transport of goods and services by air, land and sea. This area of activities is predominantly controlled by the private sector, although government agencies are active in it too, but the principles of liability and compensation applying to government operators are the same as those applying to private operators. This similarity is apparent even in the requirement of carrying insurance before undertaking an activity.
- 414. The need to protect the interests of injured parties is one of the reasons for the uniform application of the rules on liability to both private and government operators. If different rules were established, based on the capacity of operators, then Governments might attempt to limit or avoid liability by subsidizing and sponsoring commercial activities normally conducted by private operators. Furthermore, given the commercial character of these activities, there is no justification for

<sup>314</sup> See footnote 59 above.

limiting or eliminating the liability of government operators.

## (a) Multilateral agreements

415. The operator of activities causing extraterritorial damage or the insurer of the operator may be liable for damage. This is standard practice in conventions primarily concerned with commercial activities, such as the 1966 Additional Convention to the 1961 International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 25 February 1961 relating to the liability of the railway for death of and personal injury to passengers. Article 2 of the Convention reads in part:

## Article 2. Extent of liability

1. The railway shall be liable for damage resulting from the death of, or personal injury or any other bodily or mental harm to, a passenger, caused by an accident arising out of the operation of the railway and happening while the passenger is in, entering or alighting

The railway shall also be liable for damage to, or total or partial loss of any articles which the passenger who has sustained such an accident had either on him or with him as hand luggage, including any animals which he had with him.

4. ...

If the railway is not relieved of liability in accordance with the preceding sub-paragraph, the railway shall be wholly liable up to the limits laid down in this Convention, but without prejudice to any right of action which the railway may have against the third party.

- 5. This Convention shall not affect any liability which may be incurred by the railway in cases not provided for under paragraph 1.
- 6. For the purposes of this Convention, the "responsible railway" is that which, according to the list of lines provided for in article 59 of CIV, operates the line on which the accident occurs. If, in accordance with the aforementioned list, there is joint operation of the line by two railways, each of them shall be liable.

The operators of railways may be private entities or government agencies. The Convention, however, makes no distinction between them as far as liability and compensation are concerned.

416. Similarly, the 1952 Convention on Damage Caused by Foreign Aircraft to Third Parties on the Surface provides for the liability of the operator of an aircraft causing injury to a person on the surface. The relevant articles of the Convention read:

## PRINCIPLLS OF LIABILITY

#### Article 1

- Any person who suffers damage on the surface shall, upon proof only that the damage was caused by an aircraft in flight or by any person or thing falling therefrom, be entitled to compensation as provided by this Convention. Nevertheless there shall be no right to compensation if the damage is not a direct consequence of the incident giving rise thereto, or if the damage results from the mere fact of passage of the aircraft through the airspace in conformity with existing air traffic regulations.
- 2. For the purpose of this Convention, an aircraft is considered to be in flight from the moment when power is applied for the purpose of actual take-off until the moment when the landing run ends. In the case of an aircraft lighter than air, the expression "in flight" relates to the period from the moment when it becomes detached from the surface until it becomes again attached thereto.

- 1. The liability for compensation contemplated by article 1 of this Convention shall attach to the operator of the aircraft.
- 2. (a) For the purposes of this Convention the term "operator" shall mean the person who was making use of the aircraft at the time

the damage was caused, provided that if control of the navigation of the aircraft was retained by the person from whom the right to make use of the aicraft was derived, whether directly or indirectly, that person shall be considered the operator.

- (b) A person shall be considered to be making use of an aircraft when he is using it personally or when his servants or agents are using the aircraft in the course of their employment, whether or not within the scope of their authority.
- The registered owner of the aircraft shall be presumed to be the operator and shall be liable as such unless, in the proceedings for the determination of his liability, he proves that some other person was the operator and, in so far as legal procedures permit, takes appropriate measures to make that other person a party in the proceedings.

#### Article 3

If the person who was the operator at the time the damage was caused had not the exclusive right to use the aircraft for a period of more than fourteen days, dating from the moment when the right to use commenced, the person from whom such right was derived shall be liable jointly and severally with the operator, each of them being bound under the provisions and within the limits of liability of this Convention.

#### Article 4

If a person makes use of an aircraft without the consent of the person entitled to its navigational control, the latter, unless he proves that he has exercised due care to prevent such use, shall be jointly and severally liable with the unlawful user for damage giving a right to compensation under article 1, each of them being bound under the provisions and within the limits of liability of this Convention.

#### Article 5

Any person who would otherwise be liable under the provisions of this Convention shall not be liable if the damage is the direct consequence of armed conflict or civil disturbance, or if such person has been deprived of the use of the aircraft by act of public authority.

#### Article 6

- 1. Any person who would otherwise be liable under the provisions of this Convention shall not be liable for damage if he proves that the damage was caused solely through the negligence or other wrongful act or omission of the person who suffers the damage or of the latter's servants or agents. If the person liable proves that the damage was contributed to by the negligence or other wrongful act or omission of the person who suffers the damage, or of his servants or agents, the compensation shall be reduced to the extent to which such negligence or wrongful act or omission contributed to the damage. Nevertheless there shall be no such exoneration or reduction if, in the case of the negligence or other wrongful act or omission of a servant or agent, the person who suffers the damage proves that his servant or agent was acting outside the scope of his authority.
- 2. When an action is brought by one person to recover damages arising from the death or injury of another person, the negligence or other wrongful act or omission of such other person, or of his servants or agents, shall also have the effect provided in the preceding paragraph.

#### Article 7

When two or more aircraft have collided or interfered with each other in flight and damage for which a right to compensation as contemplated in article 1 results, or when two or more aircraft have jointly caused such damage, each of the aircraft concerned shall be considered to have caused the damage and the operator of each aircraft shall be liable, each of them being bound under the provisions and within the limits of liability of this Convention.

## Article 8

The persons referred to in paragraph 3 of article 2 and in articles 3 and 4 shall be entitled to all defences which are available to an operator under the provisions of this Convention.

## Article 9

Neither the operator, the owner, any person liable under article 3 or article 4, nor their respective servants or agents, shall be liable for damage on the surface caused by an aircraft in flight or any person or thing falling therefrom otherwise than as expressly provided in this Convention. This rule shall not apply to any such person who is guilty of a deliberate act or omission done with intent to cause damage. 417. The operators of aircraft may also be private or government entities. The operators enjoy limitation on liability. Article 11 of the Convention reads:

#### EXTENT OF LIABILITY

#### Article L

- 1. Subject to the provisions of article 12, the liability for damage giving a right to compensation under article 1, for each aircraft and incident, in respect of all persons liable under this Convention, shall not exceed:
  - (a) 500,000 francs for aircraft weighing 1,000 kilogrammes or less;
- (b) 500,000 francs plus 400 francs per kilogramme over 1,000 kilogrammes for aircraft weighing more than 1,000 but not exceeding 6,000 kilogrammes;
- (c) 2,500,000 francs plus 250 francs per kilogramme over 6,000 kilogrammes for aircraft weighing more than 6,000 but not exceeding 20,000 kilogrammes;
- (d) 6,000,000 francs plus 150 francs per kilogramme over 20,000 kilogrammes for aircraft weighing more than 20,000 but not exceeding 50,000 kilogrammes;
- (e) 10,500,000 francs plus 100 francs per kilogramme over 50,000 kilogrammes for aircraft weighing more than 50,000 kilogrammes.
- 2. The liability in respect of loss of life or personal injury shall not exceed 500,000 francs per person killed or injured.
- 3. "Weight" means the maximum weight of the aircraft authorized by the certificate of airworthiness for take-off, excluding the effect of lifting gas when used.
- 4. The sums mentioned in francs in this article refer to a currency unit consisting of 65.5 milligrammes of gold of millesimal fineness 900. These sums may be converted into national currencies in round figures. Conversion of the sums into national currencies other than gold shall, in case of judicial proceedings, be made according to the gold value of such currencies at the date of the judgment, or, in cases covered by article 14, at the date of the allocation.
- 418. The operators do not enjoy limitation on liability if the injury was due to their negligence. Article 12 reads:

## Article 12

- 1. If the person who suffers damage proves that it was caused by a deliberate act or omission of the operator, his servants or agents, done with intent to cause damage, the liability of the operator shall be unlimited; provided that in the case of such act or omission of such servant or agent, it is also proved that he was acting in the course of his employment and within the scope of his authority.
- 2. If a person wrongfully takes and makes use of an aircraft without the consent of the person entitled to use it, his liability shall be unlimited.
- 419. In some circumstances, liability can be imputed to the insurer of the aircraft. The relevant paragraphs of article 16 read:

### Article 16

- 5. Without prejudice to any right of direct action which he may have under the law governing the contract of insurance or guarantee, the person suffering damage may bring a direct action against the insurer or guarantor only in the following cases:
- (a) where the security is continued in force under the provisions of paragraph 1 (a) and (b) of this article;
  - (b) the bankruptcy of the operator.
- 6. Lxcepting the defences specified in paragraph 1 of this article, the insurer or other person providing security may not, with respect to direct actions brought by the person suffering damage based upon application of this Convention, avail himself of any grounds of nullity or any right of retroactive cancellation.
- 7. The provisions of this article shall not prejudice the question whether the insurer or guarantor has a right of recourse against any other person.

- 420. The two aforementioned conventions provide for limited liability. Both deal with the transport of goods and services across boundaries, an operation essential to modern civilization. It is interesting to note that article 12 of the 1952 Convention relating to damage caused by foreign aircraft allows of no limitation of liability if the operator is negligent (para. 1), and if a person wrongfully seizes and uses an aircraft without the consent of the person entitled to use it (para. 2).
- 421. The 1962 Convention on the Liability of Operators of Nuclear Ships also provides for the liability of the operator of nuclear ships, who may be either a private or a public entity. The relevant articles of the Convention read:

#### Article II

- 1. The operator of a nuclear ship shall be absolutely liable for any nuclear damage upon proof that such damage has been caused by a nuclear incident involving the nuclear fuel of, or radioactive products or waste produced in, such ship.
- 2. Except as otherwise provided in this Convention no person other than the operator shall be liable for such nuclear damage.
- 3. Nuclear damage suffered by the nuclear ship itself, its equipment, fuel or stores shall not be covered by the operator's liability as defined in this Convention.
- 4. The operator shall not be liable with respect to nuclear incidents occurring before the nuclear fuel has been taken in charge by him or after the nuclear fuel or radioactive products or waste have been taken in charge by another person duly authorized by law and liable for any nuclear damage that may be caused by them.
- 5. If the operator proves that the nuclear damage resulted wholly or partially from an act or omission done with intent to cause damage by the individual who suffered the damage, the competent courts may exonerate the operator wholly or partially from his liability to such individual.
- 6. Notwithstanding the provisions of paragraph 1 of this Article, the operator shall have a right of recourse:
- (a) If the nuclear incident results from a personal act or omission done with intent to cause damage, in which event recourse shall lie against the individual who has acted, or omitted to act, with such intent;
- (b) It the nuclear incident occurred as a consequence of any wreckraising operation, against the person or persons who carried out such operation without the authority of the operator or of the State having licensed the sunken ship or of the State in whose waters the wreck is situated;
  - (c) If recourse is expressly provided for by contract.

## Article III

1. The liability of the operator as regards one nuclear ship shall be limited to 1500 million francs in respect of any one nuclear incident, notwithstanding that the nuclear incident may have resulted from any fault or privity of that operator; such limit shall include neither any interest nor costs awarded by a court in actions for compensation under this Convention.

### Article VII

- 1. Where nuclear damage engages the liability of more than one operator and the damage attributable to each operator is not reasonably separable, the operators involved shall be jointly and severally liable for such damage. However, the liability of any one operator shall not exceed the limit laid down in article III.
- 2. In the case of a nuclear incident where the nuclear damage arises out of or results from nuclear fuel or radioactive products or waste of more than one nuclear ship of the same operator, that operator shall be liable in respect of each ship up to the limit laid down in article 111.
- 3. In case of joint and several liability, and subject to the provisions of paragraph 1 of this article:
- (a) Lach operator shall have a right of contribution against the others in proportion to the fault attaching to each of them;

- (b) Where circumstances are such that the degree of fault cannot be apportioned, the total liability shall be borne in equal parts.
- 422. Similarly, the 1969 International Convention on Civil Liability for Oil Pollution Damage provides for liability of the owner of the ship at the time of an accident. There too the operator of a ship may be a private or a government entity. The relevant articles of the Convention read:

#### Article III

1. Except as provided in paragraphs 2 and 3 of this article, the owner of a ship at the time of an incident, or where the incident consists of a series of occurrences at the time of the first such occurrence, shall be liable for any pollution damage caused by oil which has escaped or been discharged from the ship as a result of the incident.

#### Article IV

When oil has escaped or has been discharged from two or more ships, and pollution damage results therefrom, the owners of all the ships concerned, unless exonerated under article III, shall be jointly and severally liable for all such damage which is not reasonably separable.

#### Article V

- 1. The owner of a ship shall be entitled to limit his liability under this Convention in respect of any one incident to an aggregate amount of 2,000 francs for each ton of the ship's tonnage. However, this aggregate amount shall not in any event exceed 210 million francs.
- 2. If the incident occurred as a result of the actual fault or privity of the owner, he shall not be entitled to avail himself of the limitation provided in paragraph 1 of this article.
- 423. Claims for compensation may also be brought directly against the insurer of the owner. Paragraph 8 of article VII of the Convention reads:

#### Article VII

8. Any claim for compensation for pollution damage may be brought directly against the insurer or other person providing financial security for the owner's liability for pollution damage. In such case the defendant may, irrespective of the actual fault or privity of the owner, avail himself of the limits of liability prescribed in article V, paragraph 1. He may further avail himself of the defences (other than the bankruptcy or winding up of the owner) which the owner himself would have been entitled to invoke. Furthermore, the defendant may avail himself of the defence that the pollution damage resulted from the wilful misconduct of the owner himself, but the defendant shall not avail himself of any other defence which he might have been entitled to invoke in proceedings brought by the owner against him. The defendant shall in any event have the right to require the owner to be joined in the proceedings.

424. The preamble to the 1971 Convention relating to Civil Liability in the Field of Maritime Carriage of Nuclear Material explicitly states the desire of the contracting parties to hold the operator of a nuclear installation *exclusively* liable for damage caused as a result of any incident occurring during the maritime carriage of nuclear material. The relevant paragraph of the preamble states:

Desirous of ensuring that the operator of a nuclear installation will be exclusively liable for damage caused by a nuclear incident occurring in the course of maritime carriage of nuclear material,

425. The liability of the operator of a nuclear installation for the injuries it may cause is also stipulated in article II of the 1963 Vienna Convention on Civil Liability for Nuclear Damage:

#### Article II

- 1. The operator of a nuclear installation shall be liable for nuclear damage upon proof that such damage has been caused by a nuclear incident:
  - (a) in his nuclear installation; or
- (b) involving nuclear material coming from or originating in his nuclear installation, and occurring:
  - (i) before liability with regard to nuclear incidents involving the nuclear material has been assumed, pursuant to the express terms of a contract in writing, by the operator of another nuclear installation;
  - (ii) in the absence of such express terms, before the operator of another nuclear installation has taken charge of the nuclear material; or
  - (iii) where the nuclear material is intended to be used in a nuclear reactor with which a means of transport is equipped for use as a source of power, whether for propulsion thereof or for any other purpose, before the person duly authorized to operate such reactor has taken charge of the nuclear material; but
  - (iv) where the nuclear material has been sent to a person within the territory of a non-Contracting State, before it has been unloaded from the means of transport by which it has arrived in the territory of that non-Contracting State;
- (c) involving nuclear material sent to his nuclear installation, and occurring:
  - (i) after liability with regard to nuclear incidents involving the nuclear material has been assumed by him, pursuant to the express terms of a contract in writing, from the operator of another nuclear installation;
  - (ii) in the absence of such express terms, after he has taken charge of the nuclear material; or
  - (iii) after he has taken charge of the nuclear material from a person operating a nuclear reactor with which a means of transport is equipped for use as a source of power, whether for propulsion thereof or for any other purpose; but
  - (iv) where the nuclear material has, with the written consent of the operator, been sent from a person within the territory of a non-Contracting State, only after it has been loaded on the means of transport by which it is to be carried from the territory of that State:

provided that, if nuclear damage is caused by a nuclear incident occurring in a nuclear installation and involving nuclear material stored therein incidentally to the carriage of such material, the provisions of sub-paragraph (a) of this paragraph shall not apply where another operator or person is solely liable pursuant to the provisions of sub-paragraph (b) or (c) of this paragraph.

- 2. The Installation State may provide by legislation that, in accordance with such terms as may be specified therein, a carrier of nuclear material or a person handling radioactive waste may, at his request and with the consent of the operator concerned, be designated or recognized as operator in the place of that operator in respect of such nuclear material or radioactive waste respectively. In this case such carrier or such person shall be considered, for all the purposes of this Convention, as an operator of a nuclear installation situated within the territory of that State.
- 3. (a) Where nuclear damage engages the liability of more than one operator, the operators involved shall, in so far as the damage attributable to each operator is not reasonably separable, be jointly and severally liable.
- (b) Where a nuclear incident occurs in the course of carriage of nuclear material, either in one and the same means of transport, or, in the case of storage incidental to the carriage, in one and the same nuclear installation, and causes nuclear damage which engages the liability of more than one operator, the total liability shall not exceed the highest amount applicable with respect to any one of them pursuant to article V.
- (c) In neither of the cases referred to in sub-paragraphs (a) and (b) of this paragraph shall the liability of any one operator exceed the amount applicable with respect to him pursuant to article V.
- 4. Subject to the provisions of paragraph 3 of this article, where several nuclear installations of one and the same operator are involved in one nuclear incident, such operator shall be liable in respect of each nuclear installation involved up to the amount applicable with respect to him pursuant to article V.

- 5. Except as otherwise provided in this Convention, no person other than the operator shall be liable for nuclear damage. This, however, shall not affect the application of any international convention in the field of transport in force or open for signature, ratification or accession at the date on which this Convention is opened for signature.
- 6. No person shall be liable for any loss or damage which is not nuclear damage pursuant to sub-paragraph (k) of paragraph 1 of article 1 but which could have been included as such pursuant to sub-paragraph (k) (ii) of that paragraph.
- 7. Direct action shall lie against the person furnishing financial security pursuant to article VII, if the law of the competent court so provides.
- 426. Under article IV of the Convention, the operator's liability is absolute. The article reads in part:

## Article IV

- 1. The liability of the operator for nuclear damage under this Convention shall be absolute.
  - 7. Nothing in this Convention shall affect:
- (a) the liability of any individual for nuclear damage for which the operator, by virtue of paragraph 3 or 5 of this article, is not liable under this Convention and which that individual caused by an act or omission done with intent to cause damage; or
- (b) the liability outside this Convention of the operator for nuclear damage for which, by virtue of sub-paragraph (b) of paragraph 5 of this article, he is not liable under this Convention.
- 427. Under article V, the installation State may limit the operator's liability, but to not less than \$5 million for any one nuclear incident:

#### Article V

- 1. The liability of the operator may be limited by the Installation State to not less than US \$5 million for any one nuclear incident.
- 2. Any limits of liability which may be established pursuant to this article shall not include any interest or costs awarded by a court in actions for compensation of nuclear damage.
- 3. The United States dollar referred to in this Convention is a unit of account equivalent to the value of the United States dollar in terms of gold on 29 April 1963, that is to say US \$35 per one troy ounce of fine gold.
- 4. The sum mentioned in paragraph 6 of article IV and in paragraph 1 of this article may be converted into national currency in round figures.
- 428. The provisions of the 1960 Convention on Third Party Liability in the Field of Nuclear Energy concerning the liability of the nuclear operator are as follows:

#### Article 3

- (a) The operator of a nuclear installation shall be liable, in accordance with this Convention, for:
  - (i) damage to or loss of life of any person; and
  - (ii) damage to or loss of any property other than
    - the nuclear installation itself and any property on the site of that installation which is used or to be used in connection with that installation;
    - in the cases within article 4, the means of transport upon which the nuclear substances involved were at the time of the nuclear incident.

upon proof that such damage or loss (hereinafter referred to as "damage") was caused by a nuclear incident involving either nuclear fuel or radioactive products or waste in, or nuclear substances coming from such installation, except as otherwise provided for in article 4.

(b) Where the damage or loss is caused jointly by a nuclear incident and by an incident other than a nuclear incident, that part of the damage or loss which is caused by such other incident shall, to the extent that it is not reasonably separable from the damage or loss caused by the nuclear incident, be considered to be damage caused by the nuclear incident. Where the damage or loss is caused jointly by a

nuclear incident and by an emission of ionizing radiation not covered by this Convention, nothing in this Convention shall limit or otherwise affect the liability of any person in connection with that emission of ionizing radiation.

(c) Any Contracting Party may by legislation provide that the liability of the operator of a nuclear installation situated in its territory shall include liability for damage which arises out of or results from ionizing radiations emitted by any source of radiation inside that installation, other than those referred to in paragraph (a) of this article.

#### Article 4

In the case of carriage of nuclear substances, including storage incidental thereto, without prejudice to article 2:

- (a) The operator of a nuclear installation shall be liable, in accordance with this Convention, for damage upon proof that it was caused by a nuclear incident outside that installation and involving nuclear substances in the course of carriage therefrom, only if the incident occurs:
  - (i) before liability with regard to nuclear incidents involving the nuclear substances has been assumed, pursuant to the express terms of a contract in writing, by the operator of another nuclear installation;
  - (ii) in the absence of such express terms, before the operator of another nuclear installation has taken charge of the nuclear substances; or
  - (iii) where the nuclear substances are intended to be used in a reactor comprised in a means of transport, before the person duly authorized to operate that reactor has taken charge of the nuclear substances; but
  - (iv) where the nuclear substances have been sent to a person within the territory of a non-Contracting State, before they have been unloaded from the means of transport by which they have arrived in the territory of that non-Contracting State.
- (b) The operator of a nuclear installation shall be liable, in accordance with this Convention, for damage upon proof that it was caused by a nuclear incident outside that installation and involving nuclear substances in the course of carriage thereto, only if the incident occurs:
  - (i) after liability with regard to nuclear incidents involving the nuclear substances has been assumed by him, pursuant to the express terms of a contract in writing, from the operator of another nuclear installation;
  - (ii) in the absence of such express terms, after he has taken charge of the nuclear substances; or
  - (iii) after he has taken charge of the nuclear substances from a person operating a reactor comprised in a means of transport; but
  - (iv) where the nuclear substances have, with the written consent of the operator, been sent from a person within the territory of a non-Contracting State, after they have been loaded on the means of transport by which they are to be carried from the territory of that State.
- (c) The operator liable in accordance with this Convention shall provide the carrier with a certificate issued by or on behalf of the insurer or other financial guarantor furnishing the security required pursuant to article 10. The certificate shall state the name and address of that operator and the amount, type and duration of the security, and these statements may not be disputed by the person by whom or on whose behalf the certificate was issued. The certificate shall also indicate the nuclear substances and the carriage in respect of which the security applies and shall include a statement by the competent public authority that the person named is an operator within the meaning of this Convention.
- (d) A Contracting Party may provide by legislation that, under such terms as may be contained therein and upon fulfilment of the requirements of article 10 (a), a carrier may, at his request and with the consent of an operator of a nuclear installation situated in its territory, by decision of the competent public authority, be liable in accordance with this Convention in place of that operator. In such case for all the purposes of this Convention the carrier shall be considered, in respect of nuclear incidents occurring in the course of carriage of nuclear substances, as an operator of a nuclear installation on the territory of the Contracting Party whose legislation so provides.

#### Article 5

(a) If the nuclear fuel or radioactive products or waste involved in a nuclear incident have been in more than one nuclear installation and

are in a nuclear installation at the time damage is caused, no operator of any nuclear installation in which they have previously been shall be liable for the damage.

- (b) Where, however, damage is caused by a nuclear incident occurring in a nuclear installation and involving only nuclear substances stored therein incidentally to their carriage, the operator of the nuclear installation shall not be liable where another operator or person is liable pursuant to article 4.
- (c) If the nuclear fuel or radioactive products or waste involved in a nuclear incident have been in more than one nuclear installation and are not in a nuclear installation at the time damage is caused, no operator other than the operator of the last nuclear installation in which they were before the damage was caused or an operator who has subsequently taken them in charge shall be liable for the damage.
- (d) If damage gives rise to liability of more than one operator in accordance with this Convention, the liability of these operators shall be joint and several; provided that where such liability arises as a result of damage caused by a nuclear incident involving nuclear substances in the course of carriage in one and the same means of transport, or, in the case of storage incidental to the carriage, in one and the same nuclear installation, the maximum total amount for which such operators shall be liable shall be the highest amount established with respect to any of them pursuant to article 7 and provided that in no case shall any one operator be required, in respect of a nuclear incident, to pay more than the amount established with respect to him pursuant to article 7.

#### Article 6

- (a) The right to compensation for damage caused by a nuclear incident may be exercised only against an operator liable for the damage in accordance with this Convention, or, if a direct right of action against the insurer or other financial guarantor furnishing the security required pursuant to article 10 is given by national law, against the insurer or other financial guarantor.
- (b) Except as otherwise provided in this article, no other person shall be liable for damage caused by a nuclear incident, but this provision shall not affect the application of any international agreement in the field of transport in force or open for signature, ratification or accession at the date of this Convention.
  - (c) (i) Nothing in this Convention shall affect the liability:
    - of any individual for damage caused by a nuclear incident for which the operator, by virtue of article 3 (a) (ii) (1) and (2) or article 9, is not liable under this Convention and which results from an act or omission of that individual done with intent to cause damage;
    - 2. of a person duly authorized to operate a reactor comprised in a means of transport for damage caused by a nuclear incident when an operator is not liable for such damage pursuant to article 4 (a) (iii) or (b) (iii).
  - (ii) The operator shall incur no liability outside this Convention for damage caused by a nuclear incident except where use has not been made of the right provided for in article 7 (c), and then only to the extent that national legislation or the legislation of the Contracting Party in whose territory the nuclear installation of the operator liable is situated has made specific provisions concerning damage to the means of transport.
- (d) Any person who has paid compensation in respect of damage caused by a nuclear incident under any international agreement referred to in paragraph (b) of this article or under any legislation of a non-Contracting State shall, up to the amount which he has paid, acquire by subrogation the rights under this Convention of the person suffering damage whom he has so compensated.
- (e) Any person who has his principal place of business in the territory of a Contracting Party or who is the servant of such a person and who has paid compensation in respect of damage caused by a nuclear incident occurring in the territory of a non-Contracting State or in respect of damage suffered in such territory shall, up to the amount which he has paid, acquire the rights which the person so compensated would have had against the operator but for the provisions of article 2.
  - (f) The operator shall have a right of recourse only:
  - (i) if the damage caused by a nuclear incident results from an act or omission done with intent to cause damage, against the individual acting or omitting to act with such intent;
  - (ii) if and to the extent that it is so provided expressly by contract.

- (g) If the operator has a right of recourse to any extent pursuant to paragraph (f) of this article against any person, that person shall not, to that extent, have a right against the operator under paragraphs (d) or (e) of this article.
- (h) Where provisions of national or public health insurance, social security, workmen's compensation or occupational disease compensation systems include compensation for damage caused by a nuclear incident, rights of beneficiaries of such systems and rights of recourse by virtue of such systems shall be determined by the law of the Contracting Party or by the regulations of the inter-Governmental organization which has established such systems.
- 429. The operator's liability is formulated in more general terms in the 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden, article 3 of which provides that the injured party shall have access to local courts or the administrative authority of the State in whose territory the act causing damage has occurred:

#### Article 3

Any person who is affected or may be affected by a nuisance caused by environmentally harmful activities in another Contracting State shall have the right to bring before the appropriate Court or Administrative Authority of that State the question of the permissibility of such activities, including the question of measures to prevent damage, and to appeal against the decision of the Court or the Administrative Authority to the same extent and on the same terms as a legal entity of the State in which the activities are being carried out.

The provisions of the first paragraph of this article shall be equally applicable in the case of proceedings concerning compensation for damage caused by environmentally harmful activities. The question of compensation shall not be judged by rules which are less favourable to the injured party than the rules of compensation of the State in which the activities are being carried out.

### The Protocol to the Convention further provides:

The right established in article 3 for anyone who suffers injury as a result of environmentally harmful activities in a neighbouring State to institute proceedings for compensation before a court or administrative authority of that State shall, in principle, be regarded as including the right to demand the purchase of his real property.

430. The draft convention on liability and compensation in connection with the carriage of noxious and hazardous substances by sea, prepared by IMO in 1984, provides in article 3 for the liability of the owner of the ship carrying hazardous substances, and in article 7 for that of the shipper if the injured person has been unable to obtain from the owner full compensation for the damage, in accordance with the Convention. The relevant provisions of articles 3 and 7 are as follows:

### Article 3

1. Except as provided in paragraphs 2 and 3, the owner at the time of an incident of a ship carrying hazardous substances as cargo shall be liable for damage caused by any such substance during its carriage by sea, provided that if an incident consists of a series of occurrences having the same origin the liability shall attach to the owner at the time of the first of such occurrences.

## Article 7

. . .

- 1. The shipper of a hazardous substance shall be liable to pay compensation to any person suffering damage caused by that substance during its carriage by sea if such person has been unable to obtain from the owner full compensation for the damage under this Convention:
- (a) because the damage exceeds the owner's liability under this Convention as limited in accordance with article 6;
- (b) because the owner liable for the damage under article 3 is financially incapable of meeting his obligations in full; an owner being

treated as financially incapable of meeting his obligations if the person suffering the damage has been unable to obtain full satisfaction of the amount of compensation due under the present Convention after having taken all reasonable steps to pursue the legal remedies available to him.

(b) Bilateral agreements

431. Some bilateral agreements provide for the liability of the operator of an activity causing extraterritorial injury. Thus the 1973 Agreement between Norway and the United Kingdom<sup>315</sup> and the 1974 Agreement between the Federal Republic of Germany and Norway, <sup>316</sup> relating to the transmission of petroleum from the Ekofisk field, expressly state—the former in article 11, the latter in article 12—that the licensees are liable for pollution damage, including the costs of preventive and remedial action (see paras. 350 and 351 above, respectively).

432. The 1973 Agreement between Norway and the United Kingdom provides further, in article 4, that the pipeline company shall be subject to Norwegian law for the purposes of civil and criminal jurisdiction as well as of enforcement:

#### Article 4

The pipeline company shall be subject to Norwegian law and jurisdiction as regards civil and criminal proceedings, forum and enforcement. This shall also apply in relation to the pipeline and incidents pertaining thereto; it being understood, however, that this shall not exclude the concurrent jurisdiction of the United Kingdom courts and the application of United Kingdom law subject to the rules of United Kingdom law governing the conflict of laws.

433. Operator liability is provided for in the 1971 Agreement between Finland and Sweden concerning frontier rivers. Chapter 7 of the Agreement, dealing with compensation, provides, in article 1, that any person who is granted the right to use property belonging to a third party is liable for injuries resulting from such use:

#### Article 1

Any person who is granted the right under this Agreement to use property belonging to a third party, to use water power belonging to a third party or to take measures which otherwise cause damage or inconvenience to property belonging to a third party shall be liable to pay compensation for the property used or for the loss, damage or inconvenience caused.

Save as otherwise provided, compensation shall be fixed at the same time that permission is granted for the measure in question.

434. The 1922 Convention between Germany and Denmark<sup>317</sup> provides, in article 26, for the liability not of the operator of the activity, but of the persons who benefit from the activity, and authorizes a joint commission to decide on certain measures regarding the joint waters. Those who have suffered injuries from the new measures have the right to full compensation from those who benefit from those measures:

Article 26. Compensation for damage caused by regularization

Any person who suffers loss or damage in consequence of the regularization or of the alteration in the condition of the watercourse occasioned by such regularization has the right to claim full compensation from the person who benefits by the work in question. The matter shall be decided by the Frontier Water Commission.

The riparian proprietors must permit, subject to compensation, the erection at or in the watercourse of subsidiary works necessary to carry out the regularization of a river bed, the deposit of earth, stones, gravel, sand, wood, etc., on the land on the banks, the transport to and fro of such materials and the storing and transport to and fro of building materials, and must also grant regular right of access to the workmen and inspectors.

These provisions are also applicable to land situated behind the riparian land and to the proprietors thereof.

In the absence of agreement, the Frontier Water Commission shall determine the amount of compensation.

The Convention stipulates further, in article 27, that, if the cost of upkeep is increased by the new measures, such increase shall be borne by those who benefit from them (see para. 412 above).

# (c) Judicial decisions and State practice other than agreements

435. No clear picture of the liability of the operator can be derived from judicial decisions or official correspondence. These sources indicate no instances where the operator has been held to be solely liable for payment of compensation for injuries resulting from his activities. In the case of some incidents, private operators have voluntarily paid compensation and taken unilateral action to minimize or prevent injuries, but without admitting liability. It is obviously difficult to determine the real reason for the unilateral and voluntary action. But it would not be entirely correct to assume that this action was taken solely on "moral" grounds. The factors of pressure from the home Government, public opinion, or the necessity of a relaxed atmosphere for doing business, should not be underestimated. All these pressures lead to the creation of an expectation which is stronger than a mere moral obligation.

436. In 1972, the World Bond, a tanker registered in Liberia, leaked 12,000 gallons of crude oil into the sea while unloading at the refinery of the Atlantic Richfield Corporation, at Cherry Point, in the State of Washington. The oil spread to Canadian waters and befouled five miles of beaches in British Columbia. The spill was relatively small, but it had major political repercussions. Prompt action was taken both by the refinery and by the authorities on either side of the frontier to contain and limit the damage, so that the injury to Canadian waters and shorelines could be minimized. The cost of the clean-up operations was borne by the private operator, the Atlantic Petroleum Corporation.<sup>318</sup>

437. In the case of the transfrontier pollution caused by the activities of the *Peyton Packing Company* and the *Casuco Company*, action was taken unilaterally by those two United States companies to remedy the injury (see para. 324 above). Similarly, in the *Trail Smelter* case, the Canadian operator, the Consolidated Mining and Smelting Company, acted unilaterally to repair the damage caused by the plant's activities in the State of Washington. On the other hand, in the case of an oil prospection project contemplated by a private Canadian corporation in the *Beaufort Sea*, near the Alaskan border, the Canadian Government undertook to ensure

<sup>315</sup> See footnote 250 above.

<sup>316</sup> See footnote 14 above.

<sup>317</sup> See footnote 59 above.

<sup>&</sup>lt;sup>318</sup> See *The Canadian Yearbook of International Law 1973* (Vancouver), vol. XI, pp. 333-334.

compensation for any damage that might be caused in the United States in the event that the guarantees furnished by the corporation proved insufficient (see para. 357 above).

## C. State liability

438. Past trends demonstrate that States have been held liable for injuries caused to other States and their nationals as a result of activities occurring within their territorial jurisdiction or under their control. Even treaties imposing liability on the operators of activities have not in all cases exempted States from liability.

## (a) Multilateral agreements

439. In some multilateral treaties, States have agreed to be held liable for injuries caused by activities occurring within their territorial jurisdiction or under their control. Some conventions regulating activities undertaken mostly by private operators impose certain responsibilities upon the State to ensure that its operators abide by those regulations. If the State fails to do so, it is held liable for the injuries the operator causes. For example, under paragraph 2 of article III of the 1962 Convention on the Liability of Operators of Nuclear Ships, the operator is required to maintain insurance or other financial security covering his liability for nuclear damage in such forms as the licensing State specifies. Furthermore, the licensing State has to ensure the payment of claims for compensation for nuclear damage established against the operator by providing the necessary funds up to the limit laid down in paragraph 1 of article III, to the extent that the yield of the insurance or the financial security is inadequate to satisfy such claims (see para. 342 above). Hence the licensing State is obliged to ensure that the insurance of the operator or the owner of the nuclear ship satisfies the requirements of the Convention. Otherwise the State itself is liable and has to pay compensation. In addition, under article XV of the Convention, the State is required to take all necessary measures to prevent a nuclear ship flying its flag from operating without a licence. If a State fails to do so, and a nuclear ship flying its flag causes injury to others, the flag State is considered to be the licensing State, and it will be held liable for compensation to victims in accordance with the obligations laid down in article III. Article XV of the Convention reads:

## Article XV

- 1. Each Contracting State undertakes to take all measures necessary to prevent a nuclear ship flying its flag from being operated without a licence or authority granted by it.
- 2. In the event of nuclear damage involving the nuclear fuel of, or radioactive products or waste produced in, a nuclear ship flying the flag of a Contracting State, the operation of which was not at the time of the nuclear incident licensed or authorized by such Contracting State, the owner of the nuclear ship at the time of the nuclear incident shall be deemed to be the operator of the nuclear ship for all the purposes of this Convention, except that his liability shall not be limited in amount.
- 3. In such an event, the Contracting State whose flag the nuclear ship flies shall be deemed to be the licensing State for all the purposes of this Convention and shall, in particular, be liable for compensation for victims in accordance with the obligations imposed on a licensing State by article III and up to the limit laid down therein.

- 4. Each Contracting State undertakes not to grant a licence or other authority to operate a nuclear ship flying the flag of another State. However, nothing in this paragraph shall prevent a Contracting State from implementing the requirements of its national law concerning the operation of a nuclear ship within its internal waters and territorial sea.
- 440. For activities involving primarily States, the States themselves have accepted liability. Such is the case under the 1972 Convention on International Liability for Damage Caused by Space Objects. Article II of the Convention provides for the absolute liability of the launching State for damage caused by its space object:

#### Article II

A launching State shall be absolutely liable to pay compensation for damage caused by its space object on the surface of the earth or to aircraft in flight.

441. In the event of an accident involving two space objects and causing injury to a third State or its nationals, both launching States are liable to the third State, as provided in article IV:

#### Article IV

- 1. In the event of damage being caused elsewhere than on the surface of the earth to a space object of one launching State or to persons or property on board such a space object by a space object of another launching State, and of damage thereby being caused to a third State or to its natural or juridical persons, the first two States shall be jointly and severally liable to the third State, to the extent indicated by the following:
- (a) If the damage has been caused to the third State on the surface of the earth or to aircraft in flight, their liability to the third State shall be absolute;
- (b) If the damage has been caused to a space object of the third State or to persons or property on board, that space object elsewhere than on the surface of the earth, their liability to the third State shall be based on the fault of either of the first two States or on the fault of persons for whom either is responsible.
- 2. In all cases of joint and several liability referred to in paragraph I of this article, the burden of compensation for the damage shall be apportioned between the first two States in accordance with the extent to which they were at fault: if the extent of the fault of each of these States cannot be established, the burden of compensation shall be apportioned equally between them. Such apportionment shall be without prejudice to the right of the third State to seek the entire compensation due under this Convention from any or all of the launching States which are jointly and severally liable.
- 442. Furthermore, article V provides that, when two or more States jointly launch a space object, they are both jointly and severally liable for any damage the space object may cause:

## Article V

- 1. Whenever two or more States jointly launch a space object, they shall be jointly and severally liable for any damage caused.
- 2. A launching State which has paid compensation for damage shall have the right to present a claim for indemnification to other participants in the joint launching. The participants in a joint launching may conclude agreement regarding the apportioning among themselves of the financial obligation in respect of which they are jointly and severally liable. Such agreements shall be without prejudice to the right of a State sustaining damage to seek the entire compensation due under this Convention from any or all of the launching States which are jointly and severally liable.
- 3. A State from whose territory or facility a space object is launched shall be regarded as a participant in a joint launching.
- 443. Paragraphs 1 and 2 of article XXII provide that, if the launching entity is an international organization, it has the same liability as a launching State:

#### Article XXII

- 1. In this Convention, with the exception of articles XXIV to XXVII, reference to States shall be deemed to apply to any international intergovernmental organization which conducts space activities if the organization declares its acceptance of the rights and obligations provided for in this Convention and if a majority of the States members of the organization are States Parties to this Convention and to the Treaty on Principles Governing the Activities of States in the Exploration and Use of Outer Space, including the Moon and Other Celestial Bodies.
- 2. States members of any such organization which are States Parties to this Convention shall take all appropriate steps to ensure that the organization makes a declaration in accordance with the preceding paragraph.

Article XXII further provides, in paragraphs 3 and 4, that, independently of the launching international organization, those of its members which are parties to the Convention are also jointly and severally liable:

- 3. If an international intergovernmental organization is liable for damage by virtue of the provisions of this Convention, that organization and those of its members which are States Parties to this Convention shall be jointly and severally liable; provided, however, that:
- (a) any claim for compensation in respect of such damage shall be first presented to the organization:
- (b) only where the organization has not paid, within a period of six months, any sum agreed or determined to be due as compensation for such damage, may the claimant State invoke the liability of the members which are States Parties to this Convention for the payment of that sum
- 4. Any claim, pursuant to the provision of this Convention, for compensation is respect of damage caused to an organization which has made a declaration in accordance with paragraph 1 of this article shall be presented by a State member of the organization which is a State Party to this Convention.
- 444. Finally, the 1982 United Nations Convention on the Law of the Sea provides in article 139 that States parties to the Convention shall ensure that activities in the Area, whether carried out by the State or its nationals, are in conformity with the Convention. When a State party fails to carry out its obligation it will be liable for damage. The same liability is imposed upon an international organization for activities in the Area. In this case, States members of international organizations acting together bear joint and several liability. States members of international organizations involved in activities in the Area must ensure the implementation of the requirements of the Convention with respect to those international organizations. Article 139 of the Convention reads:

## Article 139. Responsibility to ensure compliance and liability for damage

- 1. States Parties shall have the responsibility to ensure that activities in the Area, whether carried out by States Parties, or State enterprises or natural or juridical persons which possess the nationality of States Parties or are effectively controlled by them or their nationals, shall be carried out in conformity with this part. The same responsibility applies to international organizations for activities in the Area carried out by such organizations.
- 2. Without prejudice to the rules of international law and annex III, article 22, damage caused by the failure of a State Party or international organization to carry out its responsibilities under this part shall entail liability; States Parties or international organizations acting together shall bear joint and several liability. A State Party shall not however be liable for damage caused by any failure to comply with this part by a person whom it has sponsored under article 153, paragraph 2 (b), if the State Party has taken all necessary and appropriate measures to secure effective compliance under article 153, paragraph 4, and annex III, article 4, paragraph 4.
- 3. States Parties that are members of international organizations shall take appropriate measures to ensure the implementation of this article with respect to such organizations.

445. Similarly, article 263 of the Convention provides that States and international organizations shall be liable for damage caused by pollution of the marine environment arising out of marine scientific research undertaken by them or on their behalf. States and international organizations are also liable for the measures they take which violate the Convention in respect of marine scientific research undertaken by other States and their nationals, and by international organizations. If those measures cause injury, they must pay compensation. Article 263 of the Convention reads:

#### Article 263. Responsibility and liability

- 1. States and competent international organizations shall be responsible for ensuring that marine scientific research, whether undertaken by them or on their behalf, is conducted in accordance with this Convention.
- 2. States and competent international organizations shall be responsible and liable for the measures they take in contravention of this Convention in respect of marine scientific research conducted by other States, their natural or juridical persons or by competent international organizations, and shall provide compensation for damage resulting from such measures.
- 3. States and competent international organizations shall be responsible and liable pursuant to article 235 for damage caused by pollution of the marine environment arising out of marine scientific research undertaken by them or on their behalf.

## (b) Bilateral agreements

446. One bilateral agreement refers to State liability for *injuries caused* by *fault or by deliberate* destructive activities. The 1948 Agreement between Poland and the USSR concerning their frontier régime<sup>319</sup> provides, in article 14, for the liability of a contracting party causing material damage as a result of failure to keep the frontier river in proper order and to prevent deliberate destruction of the banks of the frontier rivers and lakes. Article 14 of chapter I of the Agreement reads in part:

### Article 14

- 1. The Contracting Parties shall see that frontier waters are kept in proper order. They shall also take appropriate steps to prevent deliberate destruction of the banks of frontier rivers and lakes.
- 2. If, through the fault of one Contracting Party material damage is caused to the other Contracting Party as a result of failure to carry out the provisions of paragraph 1 of this article, compensation for such damage shall be paid by the Party responsible therefor.
- 447. Yugoslavia and Hungary also recognize the liability of both parties for any damage resulting from failure to respect the requirements of article 5 of their 1957 Agreement: 320

#### Article 5

It shall be prohibited to ret flax and hemp in the frontier waters and to discharge untreated waste waters and other substances harmful to aquatic wildlife, irrespective of the manner in which and the distance from which such substances reach the frontier waters. A Contracting Party failing to respect this provision shall make compensation for any damage caused.

448. The 1950 Treaty between Hungary and the USSR<sup>321</sup> provides for the liability of the contracting party which has caused extraterritorial injuries because

<sup>319</sup> See footnote 39 above.

<sup>&</sup>lt;sup>320</sup> Agreement of 25 May 1957 between Yugoslavia and Hungary concerning fishing in frontier waters.

<sup>321</sup> See footnote 40 above.

of its failure to keep the frontier in proper order, as required in article 14 of the Treaty:

#### Article 14

- 1. The Contracting Parties shall ensure that the frontier waters are kept in proper order. They shall also take steps to prevent deliberate damage to the banks of frontier rivers.
- 2. Where one Contracting Party occasions material damage to the other Contracting Party by failing to comply with the provisions of paragraph 1 of this article, compensation for such damage shall be paid by the Party responsible therefor.
- 449. The Netherlands and the Federal Republic of Germany recognize State liability for injuries caused in violation of certain provisions of their 1960 frontier Treaty.322 These provisions primarily refer to procedural steps which should be taken when an objection is raised regarding certain activities by the other contracting party. The importance attached to the procedural aspects of evaluating an activity involving extraterritorial injuries is quite explicit in this Treaty. It appears that a unilateral decision to carry out an activity likely to cause extraterritorial injuries may lead to a greater degree of liability for damage than in the case of activities preceded by some recommended procedural steps such as consultation, etc. Of course, consultation and other procedural steps should be taken only when there is an objection to the undertaking of a particular activity. Article 63 of the Treaty reads:

#### Article 63

- 1. If one of the Contracting Parties, notwithstanding the objections raised by the other Party under the terms of article 61, acts in violation of its obligations under this chapter or arising under any of the special agreements to be concluded as provided in article 59, thereby causing damage within the territory of the other Contracting Party, it shall be liable for damages.
- 2. Liability for damages shall arise in respect only of such damage as was sustained after the objections were raised.
- 450. State liability is also provided for in bilateral agreements concerning extraterritorial injuries to one contracting party resulting from any kind of activities. For example, the 1964 Agreement between Finland and the USSR<sup>323</sup> provides, in article 5, that the contracting party which causes injury in the territory of the other party through activities in its own territory shall be liable and shall pay compensation:

#### Article 5

Where the execution of certain measures by one Contracting Party causes loss or damage in the territory of the other Contracting Party, the Contracting Party permitting such measures in its territory shall be liable to make reparation to the Party suffering the loss or damage. Each Contracting Party shall ensure that reparation for the loss or damage is made to nationals, organizations and institutions of its own country.

The Contracting Parties may agree separately to make reparation for any loss or damage caused by the measures referred to in this article by granting the Party suffering the loss or damage certain privileges in the watercourses of the other Party.

451. The 1973 Treaty between Argentina and Uruguay concerning the La Plata River<sup>324</sup> contains an express provision concerning the general liability of the State

for extraterritorial damage. Each contracting party is liable for damage resulting from polluting activities conducted in its territory and causing detriment to the other contracting party. State liability is explicitly provided for in article 51 of the Treaty, regardless whether the polluting activity is carried out by the State or by private entities:

#### Article 51

Each Party shall be liable to the other for detriment suffered as a consequence of pollution caused by their operations, or by those of physical or corporate persons domiciled on their soil.

452. The 1951 Agreement between Finland and Norway concerning the transfer of the waters of a joint river (the Näätämo)<sup>325</sup> also contains provisions on State liability. However, a difference may be noted between this Agreement and those mentioned above. Whereas the latter deal with activities which either contracting party may undertake with or without prior consultation, and which may be harmful to the other party, this Agreement deals with certain agreed changes to be made in the course of a joint river which may be injurious to one of the parties. Consequently the party which benefits more from such changes agrees to compensate the other party for the injuries it may sustain. Article 2, paragraph (c), of the Agreement provides:

#### Article 2

- (c) The Government of Norway shall compensate the Government of Finland for any loss of water power which may be caused as a result of this Agreement and for the cost of the clearing operations referred to under (b) above, by an overall payment which has been fixed at 15,000 Norwegian kronor.
- 453. The concept of State liability as expressed in the preceding agreement is also to be found in an agreement concluded in 1959 between the USSR and Finland concerning the regulation of Lake Inari. Article 1 provides for the payment of a lump sum of 75 million Finnish marks for any loss or damage that might be caused to land, waters, structures or other property belonging to the State, to communes or to private persons of Finland as a result of the implementation of certain agreed activities. Consequently, and in consideration of the payment of the lump sum, the Soviet Union is exonerated from all responsibility towards Finland and its nationals arising out of these activities:

#### Article I

In consideration of such loss and damage as have been or may be caused to the lands, waters, structures or other property of any kind belonging to the State, communes and private persons and bodies of Finland as a result of the regulation of Lake Inari under the Agreement of 24 April 1947 and the Agreement concluded this day, and as payment for the works which have been and are to be carried out by the Finnish Ministry under the Regulations referred to in article 2 of the said Agreements, the Government of the Soviet Union has paid to the Government of Finland a lump sum of seventy-five million (75,000,000) Finnish markkaa.

<sup>322</sup> See footnote 100 above.

<sup>323</sup> See footnote 126 above.

<sup>&</sup>lt;sup>324</sup> Treaty of 19 November 1973 between Argentina and Uruguay concerning the La Plata River and its maritime limits.

<sup>325</sup> See footnote 227 above.

<sup>&</sup>lt;sup>326</sup> Additional Protocol of 29 April 1959 between the USSR and Finland concerning compensation for loss and damage and for the works to be carried out by Finland in connection with the implementation of the Agreement of 29 April 1959 between the Government of the Union of Soviet Socialist Republics, the Government of Finland and the Government of Norway concerning the regulation of Lake Inari by means of the Kaitakoski hydroelectric power station and dam.

The Government of the Soviet Union is consequently exonerated of all responsibility of the State, communes, individuals and corporate bodies of Finland for the loss and damage referred to in the first paragraph of this article and for the works which have been and are to be carried out by the Finnish Ministry under the Regulations referred to in article 2 of the said Agreements. The Finnish Ministry assumes all such responsibility to the said authorities, persons and bodies.

A different form of State liability may be observed in the 1959 Convention between France and Spain,<sup>327</sup> which provides for mutual assistance between the two contracting parties in case of fire. The Convention exonerates the party that was called upon for assistance from liability for any damage that may be caused to third parties; liability for such damage lies with the party requesting the assistance. As for the damage that may be caused to third parties by the fire services while on their way to or from the place where they are employed, the authorities in whose territory the injury has occurred will be liable for damage. Moreover, if the emergency assistance causes injury or death to the service personnel, the party to which the personnel belong shall waive any claim against the other party. Article VI of the Convention reads:

## Article VI. Payment of damages and compensation consequent upon accidents

- 1. In the event of the death of or injury to emergency personnel, the Party to which the personnel in question belong shall waive any claim against the other Party.
- 2. If the emergency services called in to assist cause damage to third parties at the place where they are employed, such damage being attributable to the emergency operations, the damage shall be the responsibility of the Party which requested the assistance, even if it results from a faulty action or technical error.
- 3. If the emergency services called in to assist cause damage to third parties while on the way to or from the place where they are employed, such damage shall be the responsibility of the authorities in whose territory it was caused.
- 455. Similarly, article 5 of the 1967 Agreement between the Federal Republic of Germany and Austria regarding the Salzburg airport<sup>328</sup> provides for the liability of the Federal Republic of Germany if injury to third parties occurs in its territory as a result of unlawful conduct by Austrian airport officials. Austria is of course obliged to compensate the Federal Republic of Germany for its discharge of liability arising from the claim. This article does not apply to injuries that may be sustained by Austrian nationals. It reads:

#### Article 5

- 1. In the event of damage to persons, property or interests resulting in the territory of the Federal Republic of Germany from the effects of airport traffic or of operation of the Salzburg airport and culpably caused, through unlawful conduct, by agents of the Republic of Austria in connection with their official activities, the Federal Republic of Germany shall be liable in accordance with the laws and regulations governing its liability in respect of its own agents.
- 2. The Federal Republic of Germany shall, if a claim is made against it pursuant to paragraph !, notify the Republic of Austria accordingly without delay, and shall also inform it if the claim is brought before a court.
- 3. The Republic of Austria shall be obligated, to the extent that its laws and regulations permit, to make available to the Federal Republic

- of Germany such information and evidence obtainable by it as may be helpful in dealing with the damage claim.
- 4. The Federal Republic of Germany shall notify the Republic of Austria of the settlement of the claim; copies of the decision, agreement or other disposition resulting in a settlement shall be attached.
- 5. The Republic of Austria shall compensate the Federal Republic of Germany for its discharge of the liabilities arising from paragraph (1).
- 6. This article shall not apply where the damage is sustained by an Austrian national.
- 456. The Agreement also provides that, if the Federal Republic of Germany takes measures in connection with the airport that give rise to liability on the part of the airport operator under German law, the Federal Republic shall accept the liability. However, Austria must reimburse the Government of the Federal Republic of Germany for all necessary costs and damages resulting from those measures. Article 4 reads in part:

#### Article 4

- 1. Where measures taken by German authorities in connection with contruction and operation of the Salzburg airport give rise under German law to liability for compensation on the part of the airport operator, such liability shall be assumed by the Federal Republic of Germany.
- 2. The Republic of Austria shall reimburse the Federal Republic of Germany, the State of Bavaria and its municipal corporations for all necessary costs and all damage incurred in connection with construction and operation of the airport, especially costs arising under paragraph 1 and other costs incurred in meeting third-party claims.
- 457. In a number of agreements with Ireland, Italy and the Netherlands, the United States of America accepts liability for certain injuries which may arise out of the use of the ports of those States by the United States nuclear ship Savannah. The United States further accepts liability for injuries arising from the operation of the Savannah by a private company. In the 1964 Agreement between the United States and Ireland, <sup>329</sup> paragraphs 1 and 4 of note I provide:
- 1. The United States Government shall provide compensation for all loss, damage, death or injury in Ireland (including Irish territorial seas) arising out of or resulting from the operation of N.S. Savannah to the extent that the United States Government, the United States Maritime Administration or a person indemnified under the Indemnification Agreement is liable for public liability in respect of such loss, damage, death or injury.
- 4. The United States Government being liable in the conditions specified in paragraph (I) of this Agreement, shall not pursue any right of recourse against any person who might otherwise be liable for such loss, damage, death or injury.
- 458. A similar agreement concluded between the United States of America and Italy in 1964<sup>330</sup> provides:

## Article VIII. Liability for damage

Within the limitations of liability set by United States Public Law 85-256 (annex "A"), as amended by 85-602 (annex "B"), in any legal action or proceeding brought in personam against the United States in an Italian court, the United States Government will pay compensation for any responsibility which an Italian court may find, according to Italian law, for any damage to people or goods deriving from a nuclear incident in connection with, arising out of or resulting from the operation, repair, maintenance or use of the ship, in which the N.S. Savannah may be involved within Italian territorial waters, or outside of them on a voyage to or from Italian ports if damage is caused in Italy or on ships of Italian registry.

<sup>&</sup>lt;sup>327</sup> Convention of 14 July 1959 between France and Spain on mutual assistance between French and Spanish fire and emergency services, and Additional Agreement of 8 February 1973 to that Convention.

<sup>328</sup> See footnote 18 above.

<sup>329</sup> See footnote 21 above.

<sup>330</sup> See footnote 20 above.

Subject to the \$500 million limitation in such public laws, the United States Government agrees not to interpose the defence of sovereign immunity and to submit to the jurisdiction of the Italian courts and not to invoke the provisions of Italian laws or any other law relating to the limitation of shipowners' liability.

- 459. In an exchange of notes of 1965 contituting an agreement between the United States of America and Italy concerning liability during the private operation of the N.S. Savannah,<sup>331</sup> the United States also assumes liability. The relevant paragraphs of the agreement read:
- ... concerning visits of the N.S. Savannah to Italy and ... recent conversations with respect to the situation arising from the operation of the N.S. Savannah by a private company,
- . . . [the United States proposes:]

Within the limitation of liability set by United States Public Law 85-256 (Annex A), as amended by 85-602 (Annex B) in any legal action or proceeding brought in personam against the operator to the N.S. Savannah in an Italian court, the United States Government will provide compensation by way of indemnity for any legal liability which an Italian court may find for any damage to people or goods deriving from a nuclear incident in connection with, arising out of or resulting from the operation, repair, maintenance or use of the N.S. Savannah, in which the N.S. Savannah may be involved within Italian territorial waters, or outside of them on a voyage to or from Italian ports if damage is caused in Italy or on ships of Italian registry. . . .

460. Similarly, articles 1 and 3 of the 1963 Agreement between the Netherlands and the United States of America on liability for damage caused by the N.S. Savannah<sup>132</sup> provide:

#### Article 1

The United States shall provide compensation for damage which arises out of or results from a nuclear incident in connection with the design, development, construction, operation, repair, maintenance or use of the N.S. Savannah provided, and to the extent, that any competent court of the Netherlands or a Commission to be established under Netherlands law, determines the United States to be liable for public liability. The principles of law which shall govern the liability of the United States for any such damage shall be those in existence at the time of the occurrence of the said nuclear incident.

### Article 3

The United States shall pursue no rights of recourse against any person who on account of any act or omission committed on Netherlands territory would be liable for damage as described in article 1.

461. The 1963 Operational Agreement on arrangements for a visit of the N.S. Savannah to the Netherlands<sup>333</sup> provides in article 26 that, in the event of the ship running ashore, running aground or sinking in Netherlands waters, the Netherlands may take the necessary measures at the owner's expense:

## Article 26

In the event of the ship running ashore, running aground or sinking in Netherlands waters the competent authorities under Netherlands law may take the necessary action at the owners' expense. The United States Government shall offer all possible assistance and in particular shall make available any equipment which might prove necessary to expedite required operations.

462. In 1970, the Federal Republic of Germany concluded a similar treaty with Liberia concerning its nuclear ship *Otto Hahn*. <sup>334</sup> Article 16 of the Treaty provides that the Federal Republic of Germany shall ensure

the payment of damages caused by the operator of the ship:

#### Article 16

- 1. The Federal Republic of Germany shall ensure the payment of claims for compensation for nuclear damage established under this Treaty against the operator of the ship by providing the necessary funds up to a maximum amount of DM 400 million (four hundred million). Funds shall be provided only to the extent that the yield of the insurance or other financial security is inadequate to satisfy such claims.
- 2. The Government of the Federal Republic of Germany shall, upon request of the Liberian Government, make the amount available three months after the judgment against the operator has become final.

# (c) Judicial decisions and State practice other than agreements

- 463. Judicial decisions, official correspondence and inter-State relations show that States are responsible both for the private activities conducted within their territorial jurisdiction and for the activities they themselves conduct within or beyond the limits of their territorial control. Even when States have apparently refused to accept liability as a legal principle characterizing the consequences of their actions, they have nevertheless acted as though they accepted such liability, whatever the terms used to describe their position. Most of the cases and incidents examined in this section relate to activities normally conducted by States.
- 464. In its claim against the USSR in 1979 following the accidental crash of the nuclear-powered Soviet satellite, Cosmos 954, on Canadian territory, Canada sought to impose "absolute liability" on the Soviet Union by reason of the damage caused by the accident. In arguing the liability of the Soviet Union, Canada invoked not only "relevant international agreements", including the 1972 Convention on International Liability for Damage caused by Space Objects, but also "general principles of international law" (see para. 399 above).
- 465. In its judgment of 9 April 1949 in the Corfu Channel case (merits), the International Court of Justice placed liability upon Albania for failure to notify British shipping of a dangerous situation in its territorial waters, whether or not that situation had been caused by the Government of Albania. The Court found that it was the obligation of Albania to notify, for the benefit of shipping in general, the existence of mines in its territorial waters, not only by virtue of The Hague Convention No. VIII of 1907, but also of "certain general and well recognized principles, namely: elementary considerations of humanity, even more enacting in peace than in war, . . ., and every State's obligation not to allow knowingly its territory to be used for acts contrary to the rights of other States." \*\*335 The Court found that no attempt had been made by Albania to prevent the disaster and it therefore held Albania "responsible under international law for the explosions . . . and for the damage and loss of human life . . . ". 336
- 466. In connection with the construction of a highway in Mexico, in proximity to the United States border, the

<sup>331</sup> Ibid.

<sup>332</sup> See footnote 22 above.

<sup>333</sup> Ibid.

<sup>334</sup> See footnote 19 above.

<sup>335</sup> I.C.J. Reports 1949, p. 22.

<sup>&</sup>lt;sup>336</sup> *Ibid.*, p. 36.

United States Government, considering that, not-withstanding the technical changes that had been made in the project at its request, the highway did not offer sufficient guarantees for the security of property situated in United States territory (see paras. 163 and 292 above), reserved its rights in the event of damage resulting from the construction of the highway. In a note addressed on 29 July 1959 to the Mexican Minister of Foreign Relations, the United States Ambassador to Mexico concluded:

In view of the foregoing, I am instructed to reserve all the rights that the United States may have under international law in the event that damage in the United States results from the construction of the highway.<sup>337</sup>

467. In the case of the Rose Street Canal (see para. 248 above), both the United States and Mexico reserved the right to invoke the responsibility of the State whose construction activities might cause damage in the territory of the other State.

468. In the correspondence between Canada and the United States regarding the United States Cannikin underground nuclear tests on Amchitka (see para. 240 above), Canada reserved its rights to compensation in the event of damage.

469. The series of United States nuclear tests on Eniwetok Atoll on 1 March 1954 caused injuries extending far beyond the danger area: they injured Japanese fishermen on the high seas and contaminated a great part of the atmosphere and a considerable quantity of fish, thus seriously disrupting the Japanese fish market. Japan demanded compensation. In a note dated 4 January 1955, the United States Government, completely avoiding any reference to legal liability, agreed to pay compensation for injury caused by the tests:

... The Government of the United States of America has made clear that it is prepared to make monetary compensation as an additional expression of its concern and regret over the injuries sustained ...

... the United States of America hereby tenders, ex gratia, to the Government of Japan, without reference to the question of legal liability, the sum of two million dollars for purposes of compensation for the injuries or damages sustained as a result of nuclear tests in the Marshall Islands in 1954.

It is the understanding of the Government of the United States of America that the Government of Japan, in accepting the tendered sum of two million dollars, does so in full settlement of any and all claims\* against the United States of America or its agents, nationals or juridical entities for any and all injuries, losses or damages arising out of the said nuclear tests.<sup>138</sup>

470. In the case of the injuries sustained in 1954 by the inhabitants of the *Marshall Islands*, then a Trust Territory administered by the United States, the latter agreed to pay compensation. A report of the Committee on Interior and Insular Affairs of the United States Senate stated that, owing to an unexpected wind shift immediately following the nuclear explosion, the 82 inhabitants of the Rongelap Atoll had been exposed to heavy radioactive fallout. After describing the injuries to persons and property suffered by the inhabitants and the immediate and extensive medical assistance provided by the United States, the report concluded: "It

cannot be said, however, that the compensatory measures heretofore taken are fully adequate . . . ". The report disclosed that in February 1960 a complaint against the United States had been lodged with the high court of the Trust Territory with a view to obtaining \$8,500,000 as compensation for property damage, radiation sickness, burns, physical and mental agony, loss of consortium and medical expenses. The suit had been dismissed for lack of jurisdiction. The report indicated, however, that bill No. 1988 (on payment of compensation) presented in the House of Representatives was "needed to permit the United States to do justice\* to these people". On 22 August 1964, President Johnson signed into law an act under which the United States assumed "compassionate responsibility" to compensate inhabitants of the Rongelap Atoll, in the Trust Territory of the Pacific Islands, for radiation exposures sustained by them as a result of a thermonuclear detonation at Bikini Atoll in the Marshall Islands on March 1, 1954" and authorized \$950,000 to be paid in equal amounts to the affected inhabitants of Rongelap.339 According to another report, in June 1982 the Reagan Administration was prepared to pay \$100 million to the Government of the Marshall Islands in settlement of all claims against the United States by islanders whose health and property had been affected by United States nuclear weapons tests in the Pacific between 1946 and 1963.340 Reportedly, the islanders have so far filed suits in the United States in excess of \$4 billion.

The arbitral award rendered on 27 September 1968 in the Gut Dam case also bears on State liability. In 1874, a Canadian engineer had proposed to his Government the construction of a dam between Adams Island, in Canadian territory, and Les Galops Island, in the United States, in order to improve navigation on the St. Lawrence River. Following investigations and the exchange of many reports, as well as the adoption of legislation by the United States Congress approving the project, the Canadian Government undertook the construction of the dam in 1903. However, it soon became clear that the dam was too low to serve the desired purposes and, with United States permission, Canada increased its height. Between 1904 and 1951, several manmade changes affected the flow of water in the Great Lakes-St. Lawrence River basin. While the dam itself was not altered in any way, the level of the waters in the river and in nearby Lake Ontario increased. In 1951-1952, the waters reached unprecedented levels which, in combination with storms and other natural phenomena, resulted in extensive flooding and erosion, causing injuries on both the north and south shores of the lakes. In 1953, Canada removed the dam as part of the construction of the St. Lawrence Seaway, but the United States claims for damages allegedly resulting from the presence of the Gut Dam continued to fester for some years.341

472. The Lake Ontario Claims Tribunal, established in 1965 to resolve the matter, recognized the liability of

<sup>337</sup> Whiteman, op. cit. (footnote 45 above), vol. 6, p. 262.

<sup>&</sup>quot;The Department of State Bulletin, Washington, D.C., vol. 32, No. 812, 17 January 1955, pp. 90-91.

<sup>339</sup> Whiteman, op. cit. (footnote 45 above), vol. 4, p. 567.

<sup>340</sup> International Herald Tribune, 15 June 1982, p. 5, col. 2.

<sup>&</sup>lt;sup>341</sup> See the report of the United States agent before the Lake Ontario Claims Tribunals, "Canada-United States Settlement of Gut Dam Claims (September 27, 1968)", *International Legal Materials* (Washington, D.C.), vol. VIII, 1969, pp. 128-138.

Canada, without finding any fault or negligence on the part of Canada. The Tribunal, of course, relied a great deal on the terms of the second condition stipulated in the instrument signed on 18 August 1903 and 10 October 1904 whereby the United States Secretary of War had approved construction of the dam, as well as on Canada's unilateral acceptance of liability. Furthermore, the Tribunal found Canada liable not only towards the inhabitants of Les Galops in connection with the injuries caused by the dam but also towards all United States citizens. Such responsibility was moreover found not to be limited in time to some initial testing period. The Tribunal concluded that the only questions remaining to be settled were whether the Gut Dam had caused the damage for which claims had been filed and the amount of compensation.

473. Other transboundary incidents have occurred owing to activities carried out by Governments within their territories, with effects on a neighbouring State, but they have not given rise to official demands for compensation. These incidents have of course been minor and of an accidental nature.

474. In 1949, Austria made a formal protest to the Hungarian Government for installing mines in its territory close to the Austrian border, and demanded their removal, but it did not claim compensation for injuries caused by the explosion of some of the mines on its territory. Hungary had apparently laid the mines to prevent illegal passage across the border. Austria was concerned that during a flood the mines might be washed into Austrian territory and endanger the lives of its nationals resident near the border. These protests, however, did not prevent Hungary from maintaining its minefields. In 1966, a Hungarian mine exploded in Austrian territory, causing extensive damage. The Austrian Ambassador lodged a strong protest with the Hungarian Foreign Ministry, accusing Hungary of violating the uncontested international legal principle according to which measures taken in the territory of one State must not endanger the lives, health and property of citizens of another State. Following a second accident, occurring shortly after, Austria again protested to Hungary, stating that the absence of a *public* commitment by Hungary to take all measures to prevent such accidents in the future was totally inconsistent with the principle of "good neighbourliness". Hungary subsequently removed or relocated all minefields away from the Austrian border.342

475. In October 1968, during a shooting exercise, a Swiss artillery unit erroneously fired four shells into the territory of Liechtenstein. The facts concerning this incident are difficult to ascertain. However, the Swiss Government, in a note to the Government of Liechtenstein, expressed regret for the involuntary violation of the frontier. The Swiss Government stated that it was prepared to compensate all damage caused and that it would take all necessary measures to prevent a recurrence of such incidents.<sup>343</sup>

476. Judicial decisions and official correspondence demonstrate that States have agreed to assume liability

for the injurious impact of activities by private entities operating within their territory. The legal basis for such State liability appears to derive from the principle of territorial sovereignty, a concept investing States with exclusive rights within certain portions of the globe. This concept of the function of territorial sovereignty was emphasized in the *Island of Palmas* case.<sup>344</sup> The arbitrator stated that territorial sovereignty

... cannot limit itself to its negative side, i.e. to excluding the activities of other States; for it serves to divide between nations the space upon which human activities are employed, in order to assure them at all points the minimum of protection of which international law is the guardian.<sup>345</sup>

This concept was later formulated in a more realistic way, namely, that actual physical control is the sound basis for State liability and responsibility. The International Court of Justice, in its advisory opinion of 21 June 1971 concerning Namibia, stated:

... Physical control of a territory, and not sovereignty or legitimacy of title, is the basis of State liability for acts affecting other States. The from this perspective, the liability of States for extraterritorial damage caused by private persons under their control is an important issue to be examined in the context of this study. The following are examples of State practice touching upon this source of State liability.

477. In 1948, a munitions factory in Arcisate, in Italy, near the Swiss frontier, exploded and caused varying degrees of damage in several Swiss communes. The Swiss Government demanded reparation from the Italian Government for the damage sustained; it invoked the principle of good neighbourliness and argued that Italy was liable since it tolerated the existence of an explosives factory, with all its attendant hazards, in the immediate vicinity of an international border.<sup>347</sup>

478. In 1956, the river Mura, forming the international boundary between Yugoslavia and Austria, was extensively polluted by the sediments and mud which several Austrian hydroelectric facilities had released by partially draining their reservoirs in order to forestall major flooding. Yugoslavia claimed compensation for the economic loss incurred by two paper mills and for damage to fisheries. In 1959, the two States agreed on a settlement, pursuant to which Austria paid monetary compensation and delivered a certain quantity of paper to Yugoslavia.<sup>348</sup> Although the settlement was reached in the framework of the Permanent Austro-Yugoslavian Commission for the River Mura, this is a case in which the injured State invoked the direct liability of the controlling State and the controlling State accepted the claim to pay compensation.

479. In 1971, the Liberian tanker *Juliana* ran aground and split apart off Niigata, on the west coast of the

<sup>342</sup> See Handl, loc. cit. (footnote 71 above), pp. 23-24.

<sup>&</sup>lt;sup>343</sup> Annuaire suisse de droit international, 1969-1970, Zurich, vol. 26, p. 158.

<sup>&</sup>lt;sup>344</sup> Netherlands v. United States of America, United Nations, Reports of International Arbitral Awards, vol. II, p. 829.

<sup>345</sup> Ibid., p. 839.

<sup>&</sup>lt;sup>346</sup> Legal Consequences for States of the continued presence of South Africa in Namibia (South West Africa) notwithstanding Security Council resolution 276 (1970), advisory opinion, I.C.J. Reports 1971, p. 54, para. 118.

<sup>347</sup> Guggenheim, loc. cit. (footnote 191 above), p. 169

<sup>&</sup>lt;sup>348</sup> See Handl, "State liability for accidental transnational environmental damage by private persons", American Journal of International Law, vol. 74, 1980, pp. 545-546; The Times, London, 2 December 1971, p. 8, col. 1.

Japanese island of Honshu. The oil of the tanker washed ashore and extensively damaged local fisheries. The Liberian Government (the flag State) offered 200 million yen to the fishermen for damage, which they accepted.<sup>349</sup> In this affair, the Liberian Government accepted the claims for damage caused by the act of a private person. It seems that no allegations of wrongdoing on the part of Liberia were made at an official diplomatic level.

480. Following the accidental spill of 12,000 gallons of crude oil into the sea at *Cherry Point*, in the State of Washington, and the resultant pollution of Canadian beaches (see para. 436 above), the Canadian Government addressed a note to the United States Department of State in which it expressed its grave concern about this "ominous incident" and noted that "the Government wishes to obtain firm assurances that full compensation for all damages, as well as the cost of clean-up operations, will be paid by those legally responsible". 350 Reviewing the legal implications of the incident before the Canadian Parliament, the Canadian Secretary of State for External Affairs stated:

We are especially concerned to ensure observance of the principle established in the 1938 Trail Smelter arbitration between Canada and the United States. This has established that one country may not permit the use of its territory in such a manner as to cause injury to the territory of another and shall be responsible to pay compensation for any injury so suffered. Canada accepted this responsibility in the Trail Smelter case and we would expect that the same principle would be implemented in the present situation. Indeed, this principle has already received acceptance by a considerable number of States and hopefully it will be adopted at the Stockholm Conference as a fundamental rule of international environmental law.<sup>331</sup>

- 481. Canada, referring to the precedent of the *Trail Smelter* case, claimed that the United States was responsible for the extraterritorial damage caused by acts occurring under its territorial control, regardless whether the United States was at fault. The final resolution of the dispute did not involve the legal principle invoked by Canada; the private company responsible for the pollution offered to pay the costs of the clean-up operations; the official United States response to the Canadian claim remains unclear.
- 482. In 1973, a major contamination occurred in the Swiss canton of Bale-Ville owing to the production of insecticides by a French chemical factory across the border. The contamination caused damage to the agriculture and environment of that canton and destroyed some 10,000 litres of milk production per month. The facts about the case and the diplomatic negotiations that followed are difficult to ascertain. The Swiss Government apparently intervened and negotiated with the French authorities in order to halt the pollution and obtain compensation for the damage. The reaction of the French authorities is unclear; however, it appears that persons injured brought charges in French courts.
- 483. During negotiations between the United States and Canada regarding a plan for oil prospection in the

Beaufort Sea, near the Alaskan border, the Canadian Government undertook to guarantee payment of any damage that might be caused in the United States by the activities of the private corporation that was to undertake the prospection (see para. 357 above). It should be noted that, although the private corporation was to furnish a bond covering compensation for potential victims in the United States, the Canadian Government accepted liability on a subsidiary basis for payment of the cost of transfrontier damage should the bonding arrangement prove to be inadequate.

## D. Exoneration from liability

484. In interstate relations as under domestic law, there are certain circumstances in which liability may be ruled out. The principles governing exoneration from liability in interstate relations are similar to those applying in domestic law, such as prescription, contributory negligence, war, civil insurrection, natural disasters of an exceptional character, etc.

### (a) Multilateral agreements

485. In certain circumstances, the liability of the operator or of the State may be precluded. Some multilateral conventions provide for such exoneration. The typical exoneration is that which results from prescription. Article 21 of the 1952 Convention on Damage Caused by Foreign Aircraft to Third Parties on the Surface provides that actions under the Convention are limited to two years from the date of the incident. Any suspension or interruption of these two years is determined by the law of the court where the action is brought. Nevertheless, the maximum time for bringing an action may not extend beyond three years from the date of the accident. The article reads:

#### Article 21

- 1. Actions under this Convention shall be subject to a period of limitation of two years from the date of the incident which caused the damage.
- 2. The grounds for suspension or interruption of the period referred to in paragraph 1 of this article shall be determined by the law of the court trying the action; but in any case the right to institute an action shall be extinguished on the expiration of three years from the date of the incident which caused the damage.
- 486. Articles 16 and 17 of the 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 25 February 1961, relating to the liability of the railway for death of and personal injury to passengers, provide for a period of time after which a right of action will be extinguished. These articles read:

## Article 16. Extinction of rights of action

1. A claimant shall lose his right of action if he does not give notice of the accident to a passenger to one of the railways to which a claim may be presented in accordance with article 13 within three months of his becoming aware of the damage.

When notice of the accident is given orally by the claimant, confirmation of this oral notice must be delivered to the claimant by the railway to which the accident has been notified.

- 2. Nevertheless the right of action shall not be extinguished:
- (a) if, within the period of time provided for in paragraph 1, the claimant has made a claim to one of the railways designated in article 13 (1);

<sup>349</sup> The Times, London, 1 October 1974; Revue générale de droit international public, Paris, vol. 80, 1975, p. 842.

<sup>350</sup> Loc. cit. (footnote 318 above).

<sup>&</sup>lt;sup>351</sup> *Ibid.*, p. 334.

<sup>&</sup>lt;sup>352</sup> See Annuaire suisse de droit international, 1974, Zurich, vol. 30, p. 147.

- (b) if the claimant proves that the accident was caused by the wrongful act or neglect of the railway;
- (c) if notice of the accident has not been given, or has been given late, as a result of circumstances for which the claimant is not responsible:
- (d) if during the period of time specified in paragraph (1), the railway responsible—or one of the two railways if in accordance with article 2 (6) two railways are responsible—knows of the accident to the passenger through other means.

#### Article 17. Limitation of actions

- 1. The periods of limitation for actions for damages brought under this Convention shall be:
- (a) in the case of the passenger who has sustained an accident, three years from the day after the accident;
- (b) in the case of other claimants, three years from the day after the death of the passenger, or five years from the day after the accident, whichever is the earlier.
- 2. When a claim is made to the railway in accordance with article 13, the three periods of limitation provided for in paragraph 1 shall be suspended until such date as the railway rejects the claim by notification in writing, and returns the documents attached thereto. If part of the claim is admitted, the period of limitation shall start to run again only in respect of that part of the claim still in dispute. The burden of proof of the receipt of the claim or of the reply and of the return of the documents, shall rest with the party relying upon these facts.

The running of the period of limitation shall not be suspended by further claims having the same object.

- 3. A right of action which has become barred by lapse of time may not be exercised even by way of counterclaim or set-off.
- 4. Subject to the foregoing provisions, the limitation of actions shall be governed by national law.
- 487. The Convention further exonerates the railway from liability if the accident has been caused by circumstances not connected with the operation of the railway and if the railway, in spite of having taken the care required, could not have avoided it. Paragraph 2 of article 2 of the Convention reads:

## Article 2. Extent of liability

- 2. The railway shall be relieved of liability if the accident has been caused by circumstances not connected with the operation of the railway and which the railway, in spite of having taken the care required in the particular circumstances of the case, could not avoid and the consequences of which it was unable to prevent.
- 488. The 1962 Convention on the Liability of Operators of Nuclear Ships provides for a ten-year period of prescription from the date of the nuclear incident. The domestic law of the licensing State may provide for a longer period. Article V of the Convention reads:

#### Article V

- 1. Rights of compensation under this Convention shall be extinguished if an action is not brought within ten years from the date of the nuclear incident. If, however, under the law of the licensing State the liability of the operator is covered by insurance or other financial security or State indemnification for a period longer than ten years, the applicable national law may provide that rights of compensation against the operator shall only be extinguished after a period which may be longer than ten years but shall not be longer than the period for which his liability is so covered under the law of the licensing State. However, such extension of the extinction period shall in no case affect the right of compensation under this Convention of any person who has brought an action for loss of life or personal injury against the operator before the expiry of the aforesaid period of ten years.
- 2. Where nuclear damage is caused by nuclear fuel, radioactive products or waste which were stolen, lost, jettisoned, or abandoned,

- the period established under paragraph 1 of this article shall be computed from the date of the nuclear incident causing the nuclear damage, but the period shall in no case exceed a period of twenty years from the date of the theft, loss, jettison or abandonment.
- 3. The applicable national law may establish a period of extinction or prescription of not less than three years from the date on which the person who claims to have suffered nuclear damage had knowledge or ought reasonably to have had knowledge of the damage and of the person responsible for the damage, provided that the period established under paragraphs 1 and 2 of this article shall not be exceeded.
- 4. Any person who claims to have suffered nuclear damage and who has brought an action for compensation within the period applicable under this article may amend his claim to take into account any aggravation of the damage, even after the expiry of that period, provided that final judgment has not been entered.
- 489. The Convention also provides for exoneration from liability of the operators of a nuclear ship if the damage was due to an act of war, hostilities, civil war or insurrection. Article VIII of the Convention reads:

#### Article VIII

No liability under this Convention shall attach to an operator in respect to nuclear damage caused by a nuclear incident directly due to an act of war, hostilities, civil war or insurrection.

490. A ten-year period of prescription is also provided for in the 1963 Vienna Convention on Civil Liability for Nuclear Damage, article VI of which reads:

#### Article VI

- 1. Rights of compensation under this Convention shall be extinguished if an action is not brought within ten years from the date of the nuclear incident. If, however, under the law of the Installation State the liability of the operator is covered by insurance or other financial security or by State funds for a period longer than ten years, the law of the competent court may provide that rights of compensation against the operator shall only be extinguished after a period which may be longer than ten years, but shall not be longer than the period for which his liability is so covered under the law of the Installation State. Such extension of the extinction period shall in no case affect rights of compensation under this Convention of any person who has brought an action for loss of life or personal injury against the operator before the expiry of the aforesaid period of ten years.
- 2. Where nuclear damage is caused by a nuclear incident involving nuclear material which at the time of the nuclear incident was stolen, lost, jettisoned or abandoned, the period established pursuant to paragraph 1 of this article shall be computed from the date of that nuclear incident, but the period shall in no case exceed a period of twenty years from the date of the theft, loss, jettison or abandonment.
- 3. The law of the competent court may establish a period of extinction or prescription of not less than three years from the date on which the person suffering nuclear damage had knowledge or should have had knowledge of the damage and of the operator liable for the damage, provided that the period established pursuant to paragraphs 1 and 2 of this article shall not be exceeded.
- 491. The same period of prescription is provided for in the 1960 Convention on Third Party Liability in the Field of Nuclear Energy, articles 8 and 9 of which read:

#### Article 8

(a) The right of compensation under this Convention shall be extinguished if an action is not brought within ten years from the date of the nuclear incident. National legislation may, however, establish a period longer than ten years if measures have been taken by the Contracting Party in whose territory the nuclear installation of the operator liable is situated to cover the liability of that operator in respect of any actions for compensation begun after the expiry of the period of ten years and during such longer period: provided that such extension of the extinction period shall in no case affect the right of compensation under this Convention of any person who has brought an action in respect of loss of life or personal injury against the operator before the expiry of the period of ten years.

- (b) In the case of damage caused by a nuclear incident involving nuclear fuel or radioactive products or waste which, at the time of the incident have been stolen, lost, jettisoned or abandoned and have not yet been recovered, the period established pursuant to paragraph (a) of this article shall be computed from the date of that nuclear incident, but the period shall in no case exceed twenty years from the date of the theft, loss, jettison or abandonment.
- (c) National legislation may establish a period of not less than two years for the extinction of the right or as a period of limitation either from the date at which the person suffering damage has knowledge or from the date at which he ought reasonably to have known of both the damage and the operator liable: provided that the period established pursuant to paragraphs (a) and (b) of this article shall not be exceeded.
- (d) Where the provisions of article 13 (c) (ii) are applicable, the right of compensation shall not, however, be extinguished if, within the time provided for in paragraph (a) of this article,
  - (i) prior to the determination by the Tribunal referred to in article 17, an action has been brought before any of the courts from which the Tribunal can choose; if the Tribunal determines that the competent court is a court other than that before which such action has already been brought, it may fix a date by which such action has to be brought before the competent court so determined; or
  - (ii) a request has been made to a Contracting Party concerned to initiate a determination by the Tribunal of the competent court pursuant to article 13 (c) (ii) and an action is brought subsequent to such determination within such time as may be fixed by the Tribunal.
- (e) Unless national law provides to the contrary, any person suffering damage caused by a nuclear incident who has brought an action for compensation within the period provided for in this article may amend his claim in respect of any aggravation of the damage after the expiry of such period provided that final judgment has not been entered by the competent court.

#### Article 9

The operator shall not be liable for damage caused by a nuclear incident directly due to an act of armed conflict, hostilities, civil war, insurrection or, except in so far as the legislation of the Contracting Party in whose territory his nuclear installation is situated may provide to the contrary, a grave natural disaster of an exceptional character.

492. The 1972 Convention on International Liability for Damage Caused by Space Objects provides for a one-year limit for bringing actions for damages. The one year runs from the occurrence of the damage or from the identification of the launching State which is liable. This period, however, shall not exceed one year following the date by which the State could *reasonably* be expected to have learned of the facts. Article X of the Convention reads:

## Article X

- 1. A claim for compensation for damage may be presented to a launching State not later than one year following the date of the occurrence of the damage or the identification of the launching State which is liable.
- 2. If, however, a State does not know of the occurrence of the damage or has not been able to identify the launching State which is liable, it may present a claim within one year following the date on which it learned of the aforementioned facts; however, this period shall in no event exceed one year following the date on which the State could reasonably be expected to have learned of the facts through the exercise of due diligence.
- 3. The time-limits specified in paragraphs 1 and 2 of this article shall apply even if the full extent of the damage may not be known. In this event, however, the claimant State shall be entitled to revise the claim and submit additional documentation after the expiration of such time-limits until one year after the full extent of the damage is known.
- 493. An action for damages may be brought within three years from the date of the occurrence of the damage under the 1969 International Convention on

Civil Liability for Oil Pollution Damage. No action may be brought after six years from the date of the incident which caused damage. Article VIII of the Convention reads:

#### Article VIII

Rights of compensation under this Convention shall be extinguished unless an action is brought thereunder within three years from the date when the damage occurred. However, in no case shall an action be brought after six years from the date of the incident which caused the damage. Where this incident consists of a series of occurrences, the six years' period shall run from the date of the first such occurrence.

494. The provisions of this Convention do not apply to warships or other ships owned or operated only for governmental and non-commercial service. Paragraph 1 of article XI reads:

#### Article XI

- 1. The provisions of this Convention shall not apply to warships or other ships owned or operated by a State and used, for the time being, only on government non-commercial service.
- 495. An identical period of prescription is stipulated in article 6 of the 1971 Convention on the Establishment of an International Fund for Compensation for Oil Pollution Damage:

#### Article 6

- 1. Rights to compensation under article 4 or indemnification under article 5 shall be extinguished unless an action is brought thereunder or a notification has been made pursuant to article 7, paragraph 6, within three years from the date when the damage occurred. However, in no case shall an action be brought after six years from the date of the incident which caused the damage.
- 2. Notwithstanding paragraph 1, the right of the owner or his guarantor to seek indemnification from the Fund pursuant to article 5, paragraph 1, shall in no case be extinguished before the expiry of a period of six months as from the date on which the owner or his guarantor acquired knowledge of the bringing of an action against him under the Liability Convention.
- 496. Contributory negligence by the injured party is also held to extinguish the total or partial liability of the operator or the acting State in some multilateral conventions. Under the 1963 Vienna Convention on Civil Liability for Nuclear Damage, if the injury is caused as a result of the gross negligence of the claimant or an act or omission of such person with intent to cause damage, the competent court may, if its domestic law so provides, relieve the operator wholly or partly from his obligation to pay damage to such person. However, it is the operator who should prove the negligence of the claimant. Paragraph 2 of article IV of the Convention reads:

## Article IV

2. If the operator proves that the nuclear damage resulted wholly or partly either from the gross negligence of the person suffering the damage or from an act or omission of such person done with intent to cause damage, the competent court may, if its law so provides, relieve the operator wholly or partly from his obligation to pay compensation in respect of the damage suffered by such person.

Paragraph 3 of the same article also provides for exoneration from liability if the injury is caused by a nuclear incident directly due to an act of armed conflict, hostilities, civil war or insurrection. Thus, unless the domestic law of the installation State provides to the contrary, the operator is not liable for nuclear damage caused by a nuclear incident directly due to a grave natural disaster of an exceptional character:

- 3. (a) No liability under this Convention shall attach to an operator for nuclear damage caused by a nuclear incident directly due to an act of armed conflict, hostilities, civil war or insurrection.
- (b) Except in so far as the law of the Installation State may provide to the contrary, the operator shall not be liable for nuclear damage caused by a nuclear incident directly due to a grave natural disaster of an exceptional character.
- 497. Under the 1969 International Convention on Civil Liability for Oil Pollution Damage, war, hostilities, civil war, insurrection or natural phenomena of an exceptional, inevitable and irresistible character are elements providing exoneration from liability, independently of negligence on the part of the claimant. Thus, when the damage is wholly caused by the negligence or other wrongful act of any Government or authorities responsible for the maintenance of lights or other navigational aids, the owner is exonerated from liability. Again the burden of proof is on the shipowner. Paragraphs 2 and 3 of article III of the Convention read:

#### Article III

- 2. No liability for pollution damage shall attach to the owner if he proves that the damage:
- (a) resulted from an act of war, hostilities, civil war, insurrection or a natural phenomenon of an exceptional, inevitable and irresistible character, or
- (b) was wholly caused by an act or omission done with intent to cause damage by a third party, or
- (c) was wholly caused by the negligence or other wrongful act of any Government or other authority responsible for the maintenance of lights or other navigational aids in the exercise of that function.
- 3. If the owner proves that the pollution damage resulted wholly or partially either from an act or omission done with intent to cause damage by the person who suffered the damage or from the negligence of that person, the owner may be exonerated wholly or partially from his liability to such person.
- 498. Under the 1972 Convention on International Liability for Damage caused by Space Objects, if the launching State proves that the damage caused to the claimant State has been wholly or partly the result of gross negligence or of an act or omission of the claimant State or its nationals with intent to cause damage, it will be exonerated from liability. Paragraph 1 of article VI of the Convention reads:

## Article VI

- 1. Subject to the provisions of paragraph 2 of this article, exoneration from absolute liability shall be granted to the extent that a launching State establishes that the damage has resulted either wholly or partially from gross negligence or from an act or omision done with intent to cause damage on the part of a claimant State or of natural or juridical persons it represents.
- 499. If a passenger suffers injuries due to his own wrongful act or neglect or his behaviour not in conformity with the normal conduct of a passenger, he will have no right of action against the railway. The railway in such cases will be relieved wholly or partially from liability. The 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 25 February 1961 relating to the liability of the railway for death of

and personal injury to passengers provides in article 2, paragraphs 3 and 4:

#### Article 2. Extent of liability

- 3. The railway shall be relieved wholly or partly of liability to the extent that the accident is due to the passenger's wrongful act or neglect or to behaviour on his part not in conformity with the normal conduct of passengers.
- 4. The railway shall be relieved of liability if the accident is due to a third party's behaviour which the railway, in spite of taking the care required in the particular circumstances of the case, could not avoid and the consequences of which it was unable to prevent.
- 500. Under the 1952 Convention on Damage caused by Foreign Aircraft to Third Parties on the Surface, if injury is caused solely through the negligence or other wrongful act or omission of the injured person or his servants or agents, the compensation shall be reduced to the extent to which the negligence or other wrongful act contributed to the damage. Article 6 of the Convention reads:

#### Article 6

- 1. Any person who would otherwise be liable under the provisions of this Convention shall not be liable for damage if he proves that the damage was caused solely through the negligence or other wrongful act or omission of the person who suffers the damage or of the latter's servants or agents. If the person liable proves that the damage was contributed to by the negligence or other wrongful act or omission of the person who suffers the damage, or of his servants or agents, the compensation shall be reduced to the extent to which such negligence or wrongful act or omission contributed to the damage. Nevertheless there shall be no such exoneration or reduction if, in the case of the negligence or other wrongful act or omission of a servant or agent, the person who suffers the damage proves that his servant or agent was acting outside the scope of his authority.
- 2. When an action is brought by one person to recover damages arising from the death or injury of another person, the negligence or other wrongful act or omission of such other person, or of his servants or agents, shall also have the effect provided in the preceding paragraph.
- 501. Under article 3, paragraph 3, and article 7, paragraph 5, of the draft convention on liability and compensation in connection with the carriage of noxious and hazardous substances by sea prepared by IMO in 1984, if the owner of the ship or the shipper of noxious substances proves that the damage resulted wholly or partially either from an act or omission done with the intent to cause damage by the person who suffered the damage or from the negligence of that person, the owner or the shipper may be exonerated wholly or partially from his liability to such person.
- Article 3, paragraph 2, of the draft convention provides that no liability shall attach to the owner of the ship or the shipper if he proves that the damage resulted from an act of war, hostilities, civil war, insurrection or a natural phenomenon of an exceptional and irresistible character, or was wholly caused by an act or omission done with the intent to cause damage by a third party. It was proposed that another subparagraph should be included in the article in which exoneration from liability of the owner or the shipper would be provided for if the damage was wholly caused by negligence or other wrongful act of any Government or other authority responsible for the maintenance of lights or other navigational aids. There is, however, no indication in the draft convention whether or not the negligent State is liable for damage. Article 3 does not appear to pro-

vide for exoneration from liability for damage caused by natural disaster.

503. Article 139 of the 1982 United Nations Convention on the Law of the Sea also provides for exoneration from liability of the State for damage caused by any failure of a person whom the State has sponsored to comply with regulations on sea-bed mining, if the State party has taken all necessary and appropriate measures to secure effective compliance under article 153, paragraph 4, and annex III, article 4, paragraph 4. Paragraph 2 (b) of article 153 deals with joint activities undertaken by the Authority, or by natural or juridical persons, or by States parties to exploit sea-bed resources. Paragraph 4 of the same article provides for control by the Authority over activities undertaken by States parties, their enterprises or nationals. (For the text of article 139 of the Convention, see para. 444 above.)

504. Finally, article 3 of the 1976 Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources provides that the operator of an installation shall be exonerated from liability if he proves that the damage resulted from an act of war, hostilities, civil war, insurrection, or a natural phenomenon of an exceptional, inevitable and irresistible character; if the operator proves that the damage resulted wholly or partly either from an act or omission done with intent to cause damage by the person who suffered the damage or from the negligence of that person, he may be exonerated wholly or partly from his liability to such person. Furthermore, the operator of an abandoned well is not liable for pollution damage if he proves that the incident which caused the damage occurred more than five years after the date on which the well was abandoned under the authority and in accordance with the requirements of the controlling State. If the well has been abandoned in other circumstances, the liability of the operator is governed by the applicable national law. Article 3 of the Convention reads in part:

## Article 3

- 3. No liability for pollution damage shall attach to the operator if he proves that the damage resulted from an act of war, hostilities, civil war, insurrection, or a natural phenomenon of an exceptional, inevitable and irresistible character.
- 4. No liability for pollution damage shall attach to the operator of an abandoned well if he proves that the incident which caused the damage occurred more than five years after the date on which the well was abandoned under the authority and in accordance with the requirements of the controlling State. Where a well has been abandoned in other circumstances, the liability of the operator shall be governed by the applicable national law.
- 5. If the operator proves that the pollution damage resulted wholly or partly either from an act or omission done with intent to cause damage by the person who suffered the damage or from the negligence of that person, the operator may be exonerated wholly or partly from his liability to such person.

Moreover, under article 10 of the Convention, rights of compensation shall be extinguished within twelve months of the date on which the injured party knew or should reasonably have known of the damage:

## Article 10

Rights of compensation under this Convention shall be extinguished unless, within twelve months of the date on which the person suffering

the damage knew or ought reasonably to have known of the damage, the claimant has in writing notified the operator of his claim or has brought an action in respect of it. However in no case shall an action be brought after four years from the date of the incident which caused the damage. Where the incident consists of a series of occurrences, the four years' period shall run from the date of the last occurrence.

## (b) Bilateral agreements

505. Exoneration from liability is stipulated in a few bilateral agreements. It is provided for only in the case of injuries resulting from operations of assistance to the other party, or in such circumstances as war, major calamities, etc. Under the 1959 Convention between France and Spain on mutual assistance in case of fire, 353 the party called upon to provide assistance is exonerated from liability for any damage it might cause (see para. 454 above). Again, the 1961 Treaty between Canada and the United States of America relating to the Columbia River Basin<sup>354</sup> provides, in article XVIII, that neither of the contracting parties shall be liable for injuries resulting from an act, an omission or a delay resulting from war, strikes, major calamity, act of God, uncontrollable force or maintenance curtailment. The article reads in part:

## Article XVIII. Liability for damage

- 1. Canada and the United States of America shall be liable to the other and shall make appropriate compensation to the other in respect of any act, failure to act, omission or delay amounting to a breach of the Treaty or of any of its provisions other than an act, failure to act, omission or delay occurring by reason of war, strike, major calamity, act of God, uncontrollable force or maintenance curtailment.
- 2. Except as provided in paragraph 1, neither Canada nor the United States of America shall be liable to the other or to any person in respect of any injury, damage or loss occurring in the territory of the other caused by any act, failure to act, omission or delay under the Treaty whether the injury, damage or loss results from negligence or otherwise.
- 506. Article 13 of the 1970 Treaty between Liberia and the Federal Republic of Germany relating to the use of Liberian ports by the German nuclear ship, the Otto Hahn, 355 provides that liability for a nuclear accident shall be governed by article VIII of the 1962 Convention on the Liability of Nuclear Ships, which exonerates operators of nuclear ships from liability in case of damage resulting directly from an act of war, hostilities, civil war or insurrection (see para. 489 above).
- 507. Article 14 of the same Treaty provides for a tenyear period within which claims for compensation must be brought:

## Article 14

- 1. Rights of compensation under article 13 of this Treaty shall be extinguished if an action is not brought within ten years from the date of the nuclear incident.
- 2. Where nuclear damage is caused by nuclear fuel, radioactive products or waste which were stolen, lost, jettisoned, or abandoned, the period established in paragraph 1 shall also be computed from the date of the nuclear incident causing the nuclear damage, but the period shall in no case exceed a period of twenty years from the date of the theft, loss, jettison or abandonment.
- 3. If the period established in paragraph 1 and the period established in paragraph 2 have not been exceeded, the rights of compensation under article 13 of this Treaty shall be subject to a prescrip-

<sup>353</sup> See footnote 327 above.

<sup>354</sup> See footnote 212 above.

<sup>355</sup> See footnote 19 above.

tion period of three years from the date on which the person who claims to have suffered a nuclear damage had knowledge or ought reasonably to have had knowledge of the damage and of the person liable for the damage.

# (c) Judicial decisions and State practice other than agreements

508. Judicial decisions and official correspondence reveal no incident in which exoneration from liability

has been recognized. In the few cases where the acting-State has not paid compensation for injuries caused, the injured State does not appear to have agreed with such conduct or recognized it to be within the right of the acting State. Even after the injuries caused by the nuclear tests which, according to the United States Government, had been necessary for reasons of security, that Government had paid compensation for one reason or another without seeking to evade its liability.

## CHAPTER VI

## Compensation

509. State practice relates to both the content and the procedure of compensation. Some treaties provide for a limitation of compensation (limited liability) in case of injuries. These treaties relate principally to activities generally considered essential to present-day civilization, such as the transport of goods and transport services by air, land and sea. The signatories to such treaties have agreed to tolerate such activities, with the potential risks they entail, provided the damage they may cause is compensated. However, the amount of the compensation to be paid for injuries caused is generally set at a level which, from an economic point of view, does not paralyse the pursuit of these activities or obstruct their development. Clearly, this is a deliberate policy decision on the part of the signatories to treaties regulating such activities and, in the absence of such treaties, judicial decisions do not appear to have set limits on the amount of compensation. The study of judicial decisions and official correspondence has not revealed any substantial limitation on the amount of compensation, although some sources indicate that it must be "reasonable" and that the parties have a duty to "mitigate damages".

## A. Content

## 1. Compensable injuries

## (a) Multilateral agreements

510. Under a number of conventions, material injuries such as loss of life, loss of or damage to property are compensable injuries. Article 1 of the 1963 Vienna Convention on Civil Liability for Nuclear Damage defines nuclear damage as follows:

## Article I

- 1. For the purposes of this Convention:
- (k) "Nuclear damage" means:
- (i) loss of life, any personal injury or any loss of, or damage to, property which arises out of or results from the radioactive properties or a combination of radioactive properties with toxic, explosive or other hazardous properties of nuclear fuel or radioactive products or waste in, or of nuclear material coming from, originating in, or sent to, a nuclear installation;
- (iii) if the law of the Installation State so provides, loss of life, any personal injury or any loss of, or damage to, property which

arises out of or results from other ionizing radiation emitted by any other source of radiation inside a nuclear installation.

511. The 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 26 February 1961 relating to the liability of the railway for death of and personal injury to passengers provides for the payment of necessary expenses such as the cost of *medical* treatment and transport, and compensation for loss due to partial or total incapacity to work and increased expenditure on the injured person's personal requirements necessitated by the injury. In the event of the death of the passenger, the compensation must cover the cost of transport of the body, burial or cremation. If the deceased passenger had a legally enforceable duty to support other persons who are now deprived of such support, such persons are entitled to compensation for their loss. National law governs the right to compensation for those to whom the deceased was providing support on a voluntary basis. Articles 3 and 4 of the Convention read:

## Article 3. Dumages in case of death of the passenger

- 1. In the case of the death of the passenger the damages shall include:
- (a) any necessary expenses following on the death, in particular the cost of transport of the body, burial and cremation;
  - (b) if death does not occur at once, the damages defined in article 4.
- 2. If, through the death of the passenger, persons towards whom he had, or would have had in the future, a legally enforceable duty to maintain are deprived of their support, such persons shall also be indemnified for their loss. Rights of action for damages by persons whom the passenger was maintaining without being legally bound to do so shall be governed by national law.

## Article 4. Damages in case of personal injury to the passenger

In the case of personal injury or any other bodily or mental harm to the passenger the damages shall include:

- (a) any necessary expenses, in particular the cost of medical treatment and transport;
- (b) compensation for loss due to total or partial incapacity to work, or to increased expenditure on his personal requirements necessitated by the injury.
- 512. Under the 1976 Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources, not only "pollution damage" but also *preventive measures* are compensable (art. 1, para. 6). Preventive measures are defined as "any reasonable measures taken by any

person in relation to a particular incident to prevent or minimize pollution damage with the exception of wellcontrol measures and measures taken to protect, repair or replace an installation" (art. 1, para. 7).

513. A few conventions dealing with nuclear materials include express provisions concerning damage other than nuclear damage caused by a nuclear incident or jointly by a nuclear incident and other occurrences. To the extent that those injuries are not reasonably separable from nuclear damage, they are considered nuclear damage and consequently compensable under the conventions. For example, article IV, paragraph 4, of the 1963 Vienna Convention on Civil Liability for Nuclear Damage provides:

#### Article IV

- 4. Whenever both nuclear damage and damage other than nuclear damage have been caused by a nuclear incident or jointly by a nuclear incident and one or more other occurrences, such other damage shall, to the extent that it is not reasonably separable from the nuclear damage, be deemed, for the purposes of this Convention, to be nuclear damage caused by that nuclear incident. Where, however, damage is caused jointly by a nuclear incident covered by this Convention and by an emission of ionizing radiation not covered by it, nothing in this Convention shall limit or otherwise affect the liability, either as regards any person suffering nuclear damage or by way of recourse or contribution, of any person who may be held liable in connection with that emission of ionizing radiation.
- 514. Similarly, article IV of the 1962 Convention on the Liability of Operators of Nuclear Ships provides:

## Article IV

Whenever both nuclear damage and damage other than nuclear damage have been caused by a nuclear incident or jointly by a nuclear incident and one or more other occurrences and the nuclear damage and such other damage are not reasonably separable, the entire damage shall, for the purposes of this Convention, be deemed to be nuclear damage exclusively caused by the nuclear incident. However, where damage is caused jointly by a nuclear incident covered by this Convention and by an emission of ionizing radiation or by an emission of ionizing radiation in combination with the toxic, explosive or other hazardous properties of the source of radiation not covered by it, nothing in this Convention shall limit or otherwise affect the liability, either as regards the victims or by way of recourse or contribution, of any person who may be held liable in connection with the emission of ionizing radiation or by the toxic, explosive or other hazardous properties of the source of radiation not covered by this Convention.

515. Non-material injuries may also be compensable. Thus it is clearly stated in article 5 of the 1966 Additional Convention to the CIV that under national law compensation may be required for mental, physical pain and suffering and for disfigurement:

## Article 5. Compensation for other injuries

National law shall determine whether and to what extent the railway shall be bound to pay damages for injuries other than those for which there is provision in articles 3 and 4, in particular for mental or physical pain and suffering (previum doloris) and for disfigurement.

- 516. Under article I of the 1963 Vienna Convention on Civil Liability for Nuclear Damage, loss or damage are compensable under the law of the competent court. Hence, if the law of the competent court provides for compensability of *non-material* injury, such injury is compensable under the Convention. Article I, paragraph 1(k) (ii), of the Convention reads:
  - (k) "Nuclear damage" means:

(ii) any other loss or damage so arising or resulting if and to the extent that the law of the competent court so provides . . . ;

## (b) Bilateral agreements

- 517. The provisions concerning compensable injuries in bilateral agreements vary. For example, the 1951 Agreement between Finland and Norway relating to the transfer of the waters of their shared river (the Näätämo), 356 enumerates the injuries that may be compensated. Article 2 (c) of the Agreement provides for compensation for any loss of water power and for the cost of clearing operations, both of which injuries are material (see para, 452 above).
- 518. The 1948 Agreement between Poland and the Soviet Union concerning their frontier<sup>357</sup> provides, in article 14, paragraph 2, that the *material* damage resulting from the fault of a contracting party shall give rise to compensation by the party responsible therefor (see para. 446 above).
- 519. A more general language concerning compensable injuries is used in the 1967 Agreement between the Federal Republic of Germany and Austria relating to the operation of the Salzburg Airport. Article 5, paragraph 1, in addition to referring to injuries to persons and property, also mentions injuries to *interests*, but without defining what constitutes interests (see para. 455 above).
- 520. Article 13 of chapter VI of the 1971 Agreement between Finland and Sweden relating to their frontier rivers provides for compensation for *damage* or *inconvenience*:

## Article 13

Where it has been decided that compensation for damage or inconvenience\* caused by the operations referred to in article 3 is to be paid in a specified annual amount, such decision shall not prevent the Commission from issuing, in connexion with a decision concerning new or amended regulations to combat pollution or if conditions have otherwise changed, such amended regulations as may be required with regard to compensation and the manner in which it is to be paid.

There is nothing in the Agreement to define inconvenience or to indicate whether compensation is confined to material injury or also includes non-material injury. However, in the context of the Agreement, the damage or inconvenience referred to in article 13 may be interpreted as referring to material injury.

521. Compensation for *damage* or *nuisance* is also stipulated in the 1929 Convention between Norway and Sweden concerning their frontier waters, <sup>359</sup> article 6 of which reads:

## COMPENSATION

## Article 6

With regard to compensation for damage or nuisances\* resulting from an undertaking, the law of the country in which the damage or nuisance occurs shall apply. With regard to measures for preventing or reducing the damage or nuisance, the law of the country in which the measures are to be carried out shall apply.

<sup>356</sup> See footnote 227 above.

<sup>357</sup> See footnote 39 above.

<sup>358</sup> See footnote 18 above.

<sup>359</sup> See footnote 36 above.

Again, there is no definition of nuisance in the Convention

# (c) Judicial decisions and State practice other than agreements

522. Judicial decisions and State practice reveal that only material injuries are compensable. Material injuries here refer to physical, tangible or quantitative injuries, as opposed to intangible harm to the dignity of the State. Material injuries which have been compensated in the past include loss of life, personal injury and loss of or damage to property. This has not, however, prevented States from claiming compensation for nonmaterial injuries.

523. State practice shows that in some cases involving potential or actual nuclear contamination or other damage caused by nuclear accidents, which have given rise to great anxiety, reparation has neither been made nor claimed for non-material injury. The outstanding examples are the Palomares incident and the Marshall Islands case. The Palomares incident involved the collision between a United States B-52G nuclear bomber and a KC-135 supply plane during a refuelling operation off the coast of Spain, resulting in the dropping of four plutonium-uranium 235 hydrogen bombs, with a destructive power of 1.5 megatons (75 times the power of the Hiroshima bomb).360 This incident created not only substantial material damage, but also gave rise to fears and anxiety throughout the western Mediterranean basin for two months, until the causes of potential damage had been neutralized. Two of the bombs that fell on land ruptured and discharged their TNT, scattering uranium and plutonium particles near the Spanish coastal village of Palomares, thereby causing imminent danger to the health of the inhabitants and the ecology of the area. Immediate remedial action was taken by the United States and Spain, and it is reported that the United States removed 1,750 tons of mildly radioactive Spanish soil and buried them in the United States.361 The third bomb hit the ground intact, but the fourth bomb was lost somewhere in the Mediterranean. After a two-month search by submarines and growing apprehension among the nations of the Mediterranean area, the bomb was located, but was lost during the operation for nine more days. Finally, after 80 days of the threat of detonation of the bomb, the device was retrieved.

524. Apparently, the United States did not pay any compensation for the apprehension caused by the incident, and there was no formal "open discussion" between Spain and the United States about the legal liability. The accident, however, is unique; if the bomb had not been retrieved, the extent of its damage could not have been measured in monetary terms. The United States could not have left the dangerous "instrument" of its activity in or near Spain and discharged its responsibility by paying compensation.

525. Following the nuclear tests in the atmosphere undertaken by the United States in *Eniwetok Atoll*, in the Marshall Islands, the Japanese Government did not demand compensation for non-material injuries. In a note by the United States Government concerning the payment of damages through a global settlement, the United States Government referred to the final settlement with the Japanese Government for "any and all injuries, losses, or damages arising out of the said nuclear tests". It was left to the Japanese Government to determine which individual injuries deserved compensation:

Following nuclear testing on March 1, 1954, at the Eniwetok testing grounds, the Government of Japan announced that injuries from radioactive fallout had been sustained on that date by members of the crew of a Japanese fishing vessel, the Diago Fukuryu Maru, which at the time of the test was outside the danger zone previously defined by the United States. On September 23, 1954, the chief radio operator, Aikichi Kuboyama, of the fishing vessel died. By an Agreement effected by exchange of notes, January 4, 1955, which entered into force the same day, the United States tendered, ex gratia, "as an additional expression of its concern and regret over the injuries sustained" by Japanese fishermen as a result of the nuclear tests in 1954 in the Marshall Islands, the sum of \$2 million for purposes of compensation for the injuries or damages sustained, and in full settlement of any and all claims on the part of Japan for any and all injuries, losses, or damages arising out of the said nuclear tests. The sum paid was, under the Agreement, to be distributed in such an equitable manner as might be determined by the Government of Japan and included provision for a solatium on behalf of each of the Japanese fishermen involved and for the claims advanced by the Government of Japan for their medical and hospitalization expenses. 362

526. In the *Trail Smelter* case, the tribunal *rejected* the United States proposal that *liquidated damages* be imposed on the operator of the smelter whenever emissions exceeded the predefined limits, regardless of any injuries it might cause. The tribunal stated:

The Tribunal has carefully considered the suggestions made by the United States for a régime by which a prefixed sum would be due whenever the concentrations recorded would exceed a certain intensity for a certain period of time or a certain greater intensity for any twenty minute period.

It has been unable to adopt this suggestion. In its opinion, and in that of its scientific advisers, such a régime would unduly and unnecessarily hamper the operations of the Trail Smelter and would not constitute a "solution fair to all parties concerned".<sup>363</sup>

The tribunal took the view that only actual injuries incurred deserved compensation.

527. It may therefore be assumed that the concept of non-material injury is not accepted in State practice in connection with activities causing extraterritorial injuries. States have not made monetary or other reparation for non-material damage. However, States have sometimes demanded reparation for such damage. In at least one case, a State has demanded compensation for violation of its territorial sovereignty. When the Cosmos 954 crashed on Canadian territory, Canada demanded compensation for the injuries it had sustained by reason of the crash, including violation by the satellite of its territorial sovereignty. Basing its claim on "international precedents", Canada stated:

The intrusion of the Cosmos 954 satellite into Canada's air space and the deposit on Canadian territory of hazardous radioactive debris

<sup>&</sup>lt;sup>360</sup> For further details on this accident, see T. Szuld, *The Bombs of Palomares*, (New York, Viking Press, 1967), and F. Lewis, *One of our H-Bombs is Missing* (New York, McGraw Hill, 1967).

<sup>&</sup>lt;sup>361</sup> "Radioactive Spanish earth is buried 10 feet deep in South Carolina", *The New York Times*, 12 April 1966, p. 28, col. 3.

<sup>&</sup>lt;sup>362</sup> Whiteman, op. cit. (footnote 45 above), vol. 4, p. 565.

<sup>&</sup>lt;sup>363</sup> United Nations, *Reports of International Arbitral Awards*, vol. III, p. 1974.

from the satellite constitutes a violation of Canada's sovereignty. This violation is established by the mere fact of the *trespass\** of the satellite, the harmful consequences of this intrusion being the damage caused to Canada by the presence of hazardous radioactive debris and the interference with the sovereign right of Canada\* to determine acts that will be performed on its territory. *International precedents recognize that a violation of sovereignty gives rise to an obligation to pay compensation.\** 

- 528. In the *Trail Smelter* case, in reply to the United States claim for damages for wrong done in violation of its sovereignty, the tribunal held that it *lacked jurisdiction*. The tribunal found it unnecessary to decide whether the facts proven did or did not constitute an infringement or violation of sovereignty of the United States under international law independently of the Arbitration Convention.<sup>363</sup>
- 529. State practice reveals instances of potential material damage. This category of practice is parallel to the role of injunction in judicial decisions, as in the Nuclear Tests case. There can certainly be no material injury prior to the operation of a particular injurious activity. Nevertheless, in a few instances, negotiations have take place to secure the adoption of protective measures, and even to demand the halting of the proposed activity. Such demands have been based on the gravity of the potential damage entailed. The general feeling seems to be that States must take reasonable protective measures to ensure, outside the limits of their territorial sovereignty, the safety and harmlessness of their lawful activities. Of course, the potential harm must be incidental and unintentional; none the less, the potentially injured States have the right to demand that protective measures be taken.
- 530. State practice regarding liability for reparation of actual damage is more settled. There is clearer acceptance of the explicit or implicit liability of States for their behaviour. In connection with a few incidents, States have also accepted responsibility for reparation of actual damage caused by the activities of private persons in their territorial jurisdiction or under their control. In the River Mura incident, Yugoslavia claimed damages from Austria for the economic loss incurred by two paper mills and by the fisheries, as a result of the extensive pollution caused by the Austrian hydroelectric facilities (see para. 478 above). In the tanker Juliana incident, the flag State, Liberia, offered 200 million yen to the Japanese fishermen in compensation for the damage which they had suffered as a result of the Juliana running aground and washing its oil on the coast of Japan (see para. 479 above).
- 531. Compensation has been made where an activity occurring in the shared domain has required the *relocation of people*. In connection with the United States nuclear tests in the *Eniwetok Atoll*, the compensation entailed payment for temporary usage of land and for *relocation costs* (see para. 84 above).
- 532. In the *Trail Smelter* case, the tribunal awarded the United States damages in respect of *physical damage* to cleared land and uncleared land and buildings by

reason of the reduction in crop yield and in the rental value of the land and buildings and, in one instance, of soil impairment. The denial of damages for other injuries, it appears, resulted mainly from failure of proof. With respect to damage to cleared land used for crops, the tribunal found that damage through reduction in crop yield due to fumigation had occurred in varying degrees during each of the years 1932 to 1936, but found no proof of damage in the year 1937. The properties owned by individual farmers which allegedly had suffered damage had been divided by the United States into three classes: (a) properties of "farmers residing on their farms"; (b) properties of "farmers who do not reside on their farms"; (a, b) properties of "farmers who were driven from their farms"; and (c) properties of large owners of land. The tribunal did not adopt that division, and adopted as the *measure of indemnity* to be applied on account of damage in respect of cleared land used for crops the measures of damage which the American Courts applied in cases of nuisance or trespass of the type involved in the case, namely, the amount of reduction in the value of use or rental value of the land caused by fumigations. 366

- 533. The tribunal found that, in the case of farm land, such reduction in the value of the use was in general the amount of the reduction of the crop yield arising from injury to crops, less cost of marketing the same.<sup>367</sup> In the opinion of the tribunal, the *failure* of farmers to increase their seeded land in proportion to such increase in other localities might also be taken into consideration. This is an example of the *duty to mitigate the injury*.
- 534. With regard to the problem of abandonment of properties by their owners, the tribunal noted that practically all such properties listed appeared to have been abandoned prior to the year 1932. In order to deal with that problem as well as with that of farmers who had been unable to increase their seeded land, the tribunal decided to estimate the damage on the basis of the *statistical data* available concerning the *average acreage* on which it was reasonable to believe that crops would have been seeded and harvested during the period under consideration but for the fumigations.<sup>368</sup>
- 535. With regard to claims for *impairment of the soil* content through increased acidity produced by the sulphur dioxide contained in the waters, the tribunal considered that the evidence put forward in support of that contention was not conclusive, except in the case of one small area in respect of which an indemnity was awarded.<sup>369</sup> The tribunal also awarded an indemnity for reduction in the value of farms in proximity to the frontier line by reason of their exposure to the fumigations.<sup>370</sup>
- 536. With regard to the claim that the fumes had inhibited the growth and reproduction of timber, the tribunal adopted the measure of damages applied in

<sup>&</sup>lt;sup>364</sup> International Legal Materials (Washington, D.C.), vol. 18, p. 907, para. 21.

<sup>&</sup>lt;sup>165</sup> United Nations, *Reports of International Arbitral Awards*, vol. III, p. 1932.

<sup>366</sup> Ibid., pp. 1924-1925.

<sup>367</sup> Ibid., p. 1925.

<sup>168</sup> Ibid.

<sup>169</sup> Ibid.

<sup>170</sup> Ibid., p. 1926.

United States courts, namely, reduction in value of the land itself due to such destruction and impairment:

(b) With regard to damage due to destruction and impairment of growing timber (not of merchantable size), the Tribunal has adopted the measure of damages applied by American courts, viz., the reduction in value of the land itself due to such destruction and impairment. Growing timberland has a value for firewood, fences, etc., as well as a value as a source of future merchantable timber. No evidence has been presented by the United States as to the locations or as to the total amounts of such growing timber existing on January 1, 1932, or as to its distribution into types of conifers—yellow pine, Douglas fir, larch or other trees. While some destruction or impairment, deterioration, and retardation of such growing timber has undoubtedly occurred since such date, it is impossible to estimate with any degree of accuracy the amount of damage. The Tribunal has, however, taken such damage into consideration in awarding indemnity for damage to land containing growing timber.

(c) With respect to damage due to the alleged lack of reproduction, the Tribunal has carefully considered the contentions presented. The contention made by the United States that fumigation prevents germination of seed is, in the opinion of the Tribunal, not sustained by the evidence. Although the experiments were far from conclusive, Hedgcock's studies tend to show, on the contrary, that, while seedlings were injured after germination owing to drought or to fumes, the actual germination did take place.

With regard to the contention made by the United States of damage due to failure of trees to produce seed as a result of fumigation, the Tribunal is of opinion that it is not proved that fumigation prevents trees from producing sufficient seeds, except in so far as the parenttrees may be destroyed or deteriorated themselves. This view is confirmed by the Hedgcock studies on cone production of yellow pine. There is a rather striking correlation between the percentage of good, fair, and poor trees found in the Hedgcock Census studies and the percentages of trees bearing a normal amount of cones, trees bearing few cones, and trees bearing no cones in the Hedgcock cone production studies. In so far, however, as lack of cone production since January 1, 1932, is due to death or impairment of the parent-trees occurring before the date, the Tribunal is of opinion that such failure of reproduction both was caused and occurred prior to January 1, 1932, with one possible exception as follows: From standard American writings on forestry, it appears that seeds of Douglas fir and yellow pine rarely germinate more than one year after they are shed, but if a tree was killed by fumigation in 1931, germination from its seeds might occur in 1932. It appears, however, that Douglas fir and yellow pine only produce a good crop of seeds once in a number of years. Hence, the Tribunal concludes that the loss of possible reproduction from seeds which might have been produced by trees destroyed by fumigation in 1931 is too speculative a matter to justify any award of indemnity

It is fairly obvious from the evidence produced by both sides that there is a general lack of reproduction of both yellow pine and Douglas fir over a fairly large area, and this is certainly due to some extent to fumigations. But, with the data at hand, it is impossible to ascertain to what extent this lack of reproduction is due to fumigations or to other causes such as fires occurring repeatedly in the same area or destruction by logging of the cone-bearing trees. It is further impossible to ascertain to what extent lack of reproduction due to fumigations can be traced to mortality or deterioration of the parenttrees which occurred since the first of January, 1932. It may be stated. in general terms, that the loss of reproduction due to the forest being depleted will only become effective when the amount of these trees per acre falls below a certain minimum. But the data at hand do not enable the Tribunal to say where and to what extent a depletion below this minimum occurred through fumigations in the years under consideration. An even approximate appraisal of the damage is further complicated by the fact that there is evidence of reproduction of lodgepole pine, cedar, and larch, even close to the boundary and in the Columbia River Valley, at least in some locations. This substitution may not be due entirely to fumigations, as it appears from standard American works on conifers that reproduction of yellow pine is often patchy; that when yellow pine is substantially destroyed in a given area, it is generally supplanted by another species of trees; and that lodgepole pine in particular has a tendency to invade and take full possession of yellow pine territory when a fire has occurred. While the other species are inferior, their reproduction is, nevertheless, a factor which has to be taken into account; but here again quantitative data are entirely lacking. It is further to be noted that the amount of rainfall is an important factor in the reproduction of yellow pine, and that where the normal annual rainfall is but little more than eighteen inches, yellow pine does not appear to thrive. It appears in evidence that the annual precipitation at Northport, in a period of fourteen years from 1923 to 1936, averaged slightly below seventeen inches. With all these considerations in mind, the Tribunal has, however, taken lack of reproduction into account to some extent in awarding indemnity for damage to uncleared land in use for timber.

On the basis of the foregoing statements as to damage and as to indemnity for damage with respect to cleared land and uncleared land, the Tribunal has awarded with respect to damage to cleared land and to uncleared land (other than uncleared land used for timber), an indemnity of sixty-two thousand dollars (\$62,000); and with respect to damage to uncleared land used for timber an indemnity of sixteen thousand dollars (\$16,000)—being a total indemnity of seventy-eight thousand dollars (\$78,000). Such indemnity is for the period from January 1, 1932, to October 1, 1937.

There remain for consideration three other items of damage claimed in the United States Statement: (Item c) "Damages in respect of livestock"; (Item d) "Damages in respect of property in the town of Northport"; (Item g) "Damages in respect of business enterprises".<sup>371</sup>

## 537. The United States had *failed* to *prove* damage in respect of livestock:

(3) With regard to "damages in respect of livestock", claimed by the United States, the Tribunal is of opinion that the United States has failed to prove that the presence of fumes from the Trail Smelter has injured either the livestock or the milk or wool productivity of livestock since January 1, 1932, through impaired quality of crop or grazing. So far as the injury to livestock is due to reduced yield of crop or grazing, the injury is compensated for in the indemnity which is awarded herein for such reduction of yield."

# 538. Again, proof of damage to property in the town of Northport was also insufficient:

(4) With regard to "damages in respect of property in the town of Northport", the same principles of law apply to assessment of indemnity to owners of urban land as apply to owners of farm and other cleared land, namely, that the measure of damage is the reduction in the value of the use or rental value of the property, due to fumigations. The Tribunal is of opinion that there is no proof of damage to such urban property; that even if there were such damage, there is no proof of facts sufficient to enable the Tribunal to estimate the reduction in the value of the use or rental value of such property; and that it cannot adopt the method contended for by the United States of calculating damages to urban property.<sup>373</sup>

539. With regard to damages in respect of business enterprises, the United States had claimed that the businessmen had suffered loss of business and impairment of the value of goodwill because of the reduced economic status of the residents of the damaged area. The tribunal found that such damage was too indirect, remote and uncertain to be appraised and not such for which an indemnity could be awarded. In the opinion of the tribunal, the argument that indemnity should be obtained for an injury to or reduction in a man's business due to the inability of his customers or clients to buy—which inability or impoverishment had been caused by a nuisance, even if proved—was too indirect and remote to become the basis, in law, for an award of indemnity.<sup>374</sup>

540. The United States contention of pollution of waterways had not been proved and, since the tribunal considered itself bound by the terms of the Arbitration Convention, it did not consider the United States re-

<sup>&</sup>lt;sup>371</sup> *Ibid.*, pp. 1929-1931.

<sup>&</sup>lt;sup>372</sup> Ibid., p. 1931.

<sup>373</sup> Ibid.

<sup>374</sup> Ibid.

quest for indemnity for money expended in the investigation undertaken concerning the problems created by the smelter. The United States had made this claim in connection with its action for violation of sovereignty. The tribunal, however, appeared to recognize the possibility of granting indemnity for the expenses of processing claims. It agreed that in some cases of international arbitration, damages had been awarded for expenses, not as compensation for violation of territorial sovereignty, but as compensation for expenses incurred by individual claimants in prosecuting their claims for wrongful acts by the offending Governments. For the tribunal, the difficulty lay not so much in the content of the claim as in its characterization as damages for violation of territorial sovereignty. It therefore decided that "neither as a separable item of damage nor as an incident to other damage should any award be made for that which the United States terms 'violation of sovereignty' ".375

- 541. In the Alabama case, the tribunal awarded damages in respect of net freights lost and other undefined damage resulting from Great Britain's failure to exercise "due diligence". However, damages in respect of the costs of pursuit of the confederate cruisers outfitted in British ports were denied because such costs could not be distinguished from the ordinary expenses of the war, as were damages in respect of prospective earnings since they depended on future and uncertain contingencies. 376
- 542. In its claim against the Soviet Union for injuries resulting from the crash of the Soviet nuclear-powered satellite *Cosmos 954* on Canadian territory, Canada stressed the duty to *mitigate damages*:

Under general principles of international law, Canada had a duty to take the necessary measures to prevent and reduce the harmful consequences of the damage and thereby to mitigate damages. Thus, with respect to the debris, it was necessary for Canada to undertake without delay operations of search, recovery, removal, testing and clean-up. These operations were also carried out in order to comply with the requirements of the domestic law of Canada. Moreover, article VI of the Convention [on International Liability for Damage caused by Space Objects] imposes on the claimant State a duty to observe reasonable standards of care with respect to damage caused by a space object.<sup>377</sup>

543. The Canadian claim also indicated that the compensation sought was reasonable, proximately caused by the accident and capable of being calculated with a reasonable degree of certainty:

In calculating the compensation claimed, Canada has applied the relevant criteria established by general principles of international law according to which fair compensation is to be paid, by including in its claim only those costs that are reasonable, proximately caused by the intrusion of the satellite and deposit of debris and capable of being calculated with a reasonable degree of certainty. 378

544. The Atlantic Richfield Corporation (ARCO), which operated the refinery at *Cherry Point*, in the State of Washington, where some 12,000 gallons of crude oil had spilled into the sea in 1972 (see para. 436 above), paid an initial *clean-up bill* of \$19,000 submitted by the

municipality of Surrey to cover its operations. ARCO later agreed to pay another \$11,606.50, to be transmitted by the United States to the Canadian Government, for its costs incurred in connection with the clean-up operation, but refused to reimburse an additional item of \$60 designated "bird loss (30 birds at \$2 a bird)". The payment was made "without admitting any liability in the matter and without prejudice to its rights and legal position". <sup>379</sup>

## 2. FORMS OF COMPENSATION

545. In State practice, compensation for extraterritorial damage caused by activities conducted within the territorial jurisdiction or under the control of States has been paid either in the form of a lump sum to the injured State, so that it may settle individual claims, or direct to the individual claimants. The forms of compensation prevailing in relations between States are similar to those existing in domestic law. Indeed, some conventions provide that national legislation is to govern the question of compensation. When damages are monetary, States have generally sought to select readily convertible currencies.

## (a) Multilateral agreements

546. While references to the forms of compensation are made in multilateral conventions, they are not sufficiently detailed. Attempts have been made in the conventions to make the compensation useful to the injured party in terms of currency and of its transferability from one State to another. Under the 1960 Convention on Third Party Liability for Nuclear Energy, for example, the nature, form and extent of the compensation as well as its equitable distribution has to be governed by national law. Furthermore, the compensation must be freely transferable between the contracting parties. The relevant provisions of the Convention are:

## Article 7

(g) Any interest and costs awarded by a court in actions for compensation under this Convention shall not be considered to be compensation for the purposes of this Convention and shall be payable by the operator in addition to any sum for which he is liable in accordance with this article.

## Article 11

The nature, form and extent of the compensation, within the limits of this Convention, as well as the equitable distribution thereof, shall be governed by national law.

## Article 12

Compensation payable under this Convention, insurance and reinsurance premiums, sums provided as insurance, reinsurance, or other financial security required pursuant to article 10, and interest and costs referred to in article 7 (g), shall be freely transferable between the monetary areas of the Contracting Parties.

547. The 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 26 February 1961 relating to the liability of railways for death of and personal injury to passengers also provides that, for certain injuries, compensation may be awarded in the form

<sup>375</sup> Ibid., pp. 1932-1933.

<sup>&</sup>lt;sup>376</sup> Moore, op. cit. (footnote 306 above), p. 658.

<sup>&</sup>lt;sup>377</sup> International Legal Materials (Washington, D.C.), vol. 18, pp. 905-906, para. 17.

<sup>&</sup>lt;sup>378</sup> *Ibid.*, p. 907, para. 23.

<sup>&</sup>lt;sup>119</sup> The Canadian Yearbook of International Law 1973, vol. 11, pp. 333-334; and Montreal Star, 9 June 1972.

of a lump sum. However, if national law permits, payment of an annuity or, if the injured passenger so requests, compensation, shall be awarded as an annuity. Such forms of damages are also provided for injuries suffered by persons for whose support the deceased passenger was legally responsible, as well as for medical treatment and transport of an injured passenger and for loss due to his total or partial incapacity to work. The relevant provisions of the Convention read:

## Article 6. Form and limit of damages in case of death of, or personal injury to the passenger

1. The damages under article 3 (2) and article 4 (b) shall be awarded in the form of a lump sum; however, if national law permits payment of an annuity, damages shall be awarded in this form if so requested by the injured passenger or the claimants designated in article 3 (2).

#### Article 9. Interest and refund of compensation

- 1. The claimant shall be entitled to claim interest on compensation which shall be calculated at the rate of 5 per cent per annum. Such interest shall accrue from the date of the claim, or, if a claim has not been made, from the date on which legal proceedings are instituted, save that for compensation due under articles 3 and 4, interest shall accrue only from the day on which the events relevant to its assessment occurred, if that day is later than the date of the claim or the date on which legal proceedings were instituted.
- 2. Any compensation improperly obtained shall be refunded.
- 548. The 1962 Convention on the Liability of Operators of Nuclear Ships states the value in gold of the franc, the currency in which compensation must be paid. It also provides that the awards may be converted into each national currency in round figures and that conversion into national currencies other than gold shall be effected on the basis of their gold value. Paragraph 4 of article III of the Convention reads:

## Article III

- 4. The franc mentioned in paragraph 1 of this article is a unit of account constituted by sixty-five and one half milligrams of gold of millesimal fineness nine hundred. The amount awarded may be converted into each national currency in round figures. Conversion into national currencies other than gold shall be effected on the basis of their gold value at the date of payment.
- 549. If agreed between the parties concerned, compensation under the 1972 Convention on International Liability for Damage Caused by Space Objects may be paid in any currency; otherwise, it is to be paid in the currency of the *claimant State*. If the claimant State agrees, the compensation may be paid in the currency of the *State from which compensation is due*. Article XIII of the Convention reads:

## Article XIII

Unless the claimant State and the State from which compensation is due under this Convention agree on another form of compensation, the compensation shall be paid in the currency of the claimant State or, if that State so requests, in the currency of the State from which compensation is due.

## (b) Bilateral agreements

550. Most bilateral agreements regarding activities likely to result in extraterritorial injuries are silent on the forms of compensation. The decision thereon appears to have been left to the relevant organs under the individual agreements, whether local courts, joint commissions or government authorities. At least two

bilateral agreements refer to the forms of compensation.

- 551. In the 1964 Agreement between the United States of America and Ireland concerning the use of Irish ports by the United States nuclear ship, the N.S. Savannah, 380 the United States Government agrees to ensure prompt payment in respect of its liability for nuclear damage under the Agreement. Paragraph 5 (note 1) of the Agreement reads:
- 5. The Government of the United States shall ensure that prompt\* payment is made in respect of the liability referred to in paragraph 1 of this Agreement.
- 552. Article 13 of chapter VI of the 1971 Agreement between Finland and Sweden concerning frontier rivers provides that compensation is to be paid in a *specific annual amount* (see para. 520 above).

# (c) Judicial decisions and State practice other than agreements

- 553. The forms of compensation are referred to in judicial decisions and official correspondence in only a few cases, such as the compensation afforded Japan by the United States for injuries arising out of the Pacific nuclear tests (see para. 525 above) and the compensation required of the United Kingdom in the *Alabama* case. <sup>381</sup> In each case, a lump sum payment was made in order to allow the injured States to pay equitable compensation to the injured individuals.
- 554. In addition to monetary compensation, compensation has occasionally consisted in removing the danger or effecting restitutio in integrum. That was the case, for example, in the Palomares incident, in 1966, when nuclear bombs dropped on Spanish territory and near the coasts of Spain following a collision between a United States nuclear bomber and a supply plane. In a situation where the damage or danger of damage is so grave, the primary compensation is restitution, that is, removing the cause of the damage and restoring the area to its condition prior to the incident. The United States removed the causes of danger from Spain by retrieving the bombs and by removing the contaminated Spanish soil and burying it in its own territory.<sup>382</sup>
- 555. Following the nuclear tests conducted in the Marshall Islands, the United States reportedly spent nearly \$110 million to clean up several of the islands of the Eniwetok Atoll so that they could again become habitable. However, one of the islands of the Runit Atoll, which had been used to bury nuclear debris, was declared off-limits for 20,000 years. 363 A clean-up operation is not restitution, but the intention and the policy behind it are similar. Following the accidental pollution of the Mura River, Austria, in addition to paying monetary compensation for the damage caused to Yugoslav fisheries and paper mills, delivered a certain quantity of paper to Yugoslavia (see para. 478 above).

<sup>380</sup> See footnote 21 above.

<sup>381</sup> Moore, op. cit. (footnote 306 above), p. 568.

<sup>382</sup> The New York Times, 12 April 1966, p. 28, col. 3.

<sup>&</sup>lt;sup>383</sup> International Herald Tribune, 15 June 1982, p. 5, col. 2.

## 3. Limitation on compensation

556. As in domestic law, State practice has provided for limitations on compensation, particularly in connection with activities which, although important to present-day civilization, can be very injurious, as well as with activities capable of causing devastating injuries, such as those involving the use of nuclear materials. The provisions on limitation of compensation have been carefully designed to fulfil two objectives: (a) to protect industries from an unlimited liability that would paralyse them financially and discourage their future development; (b) ensure reasonable and fair compensation for those who suffer injuries as a result of these dangerous activities.

## (a) Multilateral agreements

557. The 1960 Convention on Third Party Liability in the Field of Nuclear Energy is drafted to deal systematically and uniformly only with the question of liability and compensation in the field of nuclear energy. In the preamble to the Convention, the Governments of the signatory States specifically declare themselves desirous of "ensuring adequate\* and equitable\* compensation for persons who suffer damage caused by nuclear incidents while taking the necessary steps to ensure that development of the production and uses of nuclear energy for peaceful purposes is not thereby hindered"\*. Article 7 of the Convention defines the minimum and maximum amounts of compensation:

## Article 7

- (a) The aggregate of compensation required to be paid in respect of damage caused by a nuclear incident shall not exceed the maximum liability established in accordance with this article.
- (b) The maximum liability of the operator in respect of damage caused by a nuclear incident shall be 15,000,000 European Monetary Agreement units of account as defined at the date of this Convention (hereinafter referred to as "units of account"): provided that any Contracting Party, taking into account the possibilities for the operator of obtaining the insurance or other financial security required pursuant to article 10, may establish by legislation a greater or less amount, but in no event less than 5,000,000 units of account. The sums mentioned above may be converted into national currency in round figures.
- (c) Any Contracting Party may by legislation provide that the exception in article 3 (a) (ii) (2) shall not apply: provided that in no case shall the inclusion of damage to the means of transport result in reducing the liability of the operator in respect of other damage to an amount less than 5,000,000 units of account.
- (d) The amount of liability of operators of nuclear installations in the territory of a Contracting Party established in accordance with paragraph (b) of this article as well as the provisions of any legislation of a Contracting Party pursuant to paragraph (c) of this article shall apply to the liability of such operators wherever the nuclear incident occurs.
- (e) A Contracting Party may subject the transit of nuclear substances through its territory to the condition that the maximum amount of liability of the foreign operator concerned be increased, if it considers that such amount does not adequately cover the risks of a nuclear incident in the course of the transit: provided that the maximum amount thus increased shall not exceed the maximum amount of liability of operators of nuclear installations situated in its territory.
  - (f) The provisions of paragraph (e) of this article shall not apply:
  - (i) to carriage by sea where, under international law, there is a right of entry in cases of urgent distress into the ports of such Contracting Party or a right of innocent passage through its territory; or

- (ii) to carriage by air where, by agreement or under international law there is a right to fly over or land on the territory of such Contracting Party.
- 558. Under the 1952 Convention on Damage Caused by Foreign Aircraft to Third Parties on the Surface, if the total amount of claims established exceeds the limit of liability, they shall be reduced in proportion to their respective amounts in respect of claims exclusively for loss of life or personal injury or exclusively for damage to property. But if the claims concern both loss of life or personal injury and damage to property, one half of the total sum shall be allocated preferentially for loss of life or personal injury. The remainder shall be distributed proportionately among the claims in respect of damage to property and the portion not already covered of the claims in respect of loss of life and personal injury. Article 14 of the Convention reads:

#### Article 14

If the total amount of the claims established exceeds the limit of liability applicable under the provisions of this Convention, the following rules shall apply, taking into account the provisions of paragraph 2 of article 11:

- (a) If the claims are exclusively in respect of loss of life or personal injury or exclusively in respect of damage to property, such claims shall be reduced in proportion to their respective amounts.
- (b) If the claims are both in respect of loss of life or personal injury and in respect of damage to property, one half of the total sum distributable shall be appropriated preferentially to meet claims in respect of loss of life and personal injury and, if insufficient, shall be distributed proportionately between the claims concerned. The remainder of the total sum distributable shall be distributed proportionately among the claims in respect of damage to property and the portion not already covered of the claims in respect of loss of life and personal injury.
- 559. The 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 26 February 1961 relating to the liability of the railway for death of and personal injury to passengers provides for limitation of liability. However, if the damage is caused by the wilful misconduct or gross negligence of the railway, the limitation of liability is removed. Articles 7 and 8 read:

## Article 7. Limit of damages in case of damage to or loss of articles

When, under the provisions of this Convention, the railway is liable to pay damages for damage to, or for total or partial loss of any articles which the passenger who has sustained an accident had either on him or with him as hand luggage, including any animals which he had with him, compensation for the damage may be claimed up to the sum of 2,000 francs per passenger.

## Article 8. Amount of dumages in case of wilful misconduct or gross negligence

The provisions of articles 6 and 7 of this Convention or those of the national law which limit compensation to a fixed amount shall not apply if the damage results from wilful misconduct or gross negligence of the railway.

560. Article 10 of the Convention nullifies any agreement between passengers and the railway in which the liability of the railway is *precluded* or has been *limited* to a lower amount than that provided for in the Convention. Articles 10 and 12 read:

## Article 10. Prohibition of limitation of liability

Any terms or conditions of carriage or special agreements concluded between the railway and the passenger which purport to exempt the railway in advance, either totally or partially, from liability under this Convention, or which have the effect of reversing the burden of proof resting on the railway, or which provide for limits lower than those laid down in article 6 (2) and article 7, shall be null and void. Such nullity shall not, however, avoid the contract of carriage which shall remain subject to the provisions of CIV and this Convention.

## Article 12. Bringing of actions not within the provisions of this Convention

No action of any kind shall be brought against a railway in respect of its liability under article 2 (1) of this Convention, except subject to the conditions and limitations laid down in this Convention.

The same shall apply to any action brought against persons for whom the railway is liable under article 11.

561. The preamble to the 1957 International Convention relating to the Limitation of the Liability of Owners of Seagoing Ships clearly indicates the objectives of the contracting parties:

The High Contracting Parties,

Having recognized the desirability of determining by agreement certain uniform rules relating to the limitation of liability of owners of seagoing ships;

Having decided to conclude a Convention for this purpose, . . .

## 562. Article 1 of the Convention reads:

#### Article 1

- 1. The owner of a seagoing ship may limit his liability in accordance with article 3 of this Convention in respect of claims arising from any of the following occurrences, unless the occurrence giving rise to the claim resulted from the actual fault or privity of the owner:
- (a) loss of life of, or personal injury to, any person being carried in the ship, and loss of, or damage to, any property on board the ship;
- (b) loss of life of, or personal injury to, any other person, whether on land or on water, loss of or damage to any other property or infringement of any rights caused by the act, neglect or default of any person on board the ship for whose act, neglect or default the owner is responsible or any person not on board the ship for whose act, neglect or default the owner is responsible: provided however that in regard to the act, neglect or default of this last class of person, the owner shall only be entitled to limit his liability when the act, neglect or default is one which occurs in the navigation or the management of the ship or in the loading, carriage or discharge of its cargo or in the embarkation, carriage or disembarkation of its passengers;
- (c) any obligation or liability imposed by any law relating to the removal of wreck and arising from or in connection with the raising, removal or destruction of any ship which is sunk, stranded or abandoned (including anything which may be on board such ship) and any obligation or liability arising out of damage caused to harbour works, basins and navigable waterways.
- 2. In the present Convention the expression "personal claims" means claims resulting from loss of life and personal injury; the expression "property claims" means all other claims set out in paragraph 1 of this article.

Under paragraph 3 of article 1, the limitation of liability of the seagoing ship will cease if it is proved that the injury was caused by the *negligence* of the shipowner, or of persons for whose conduct he is responsible. The question upon whom lies the burden of proving whether there has been a fault is to be determined by the law of the forum. Paragraph 6 of article 1 reads:

6. The question upon whom lies the burden of proving whether or not the occurrence giving rise to the claim resulted from the actual fault or privity of the owner shall be determined by the *lex fori*.

If an action is brought against the master of a ship or a member of the crew, their liability is still limited even if the accident causing injury is due to their fault. But if the master or the member of the crew is at the same time the owner, co-owner, charter manager or operator of the seagoing ship, the limitation of liability applies only when the act of neglect or default is committed by such persons in their capacity as master or member of the crew. Paragraph 3 of article 6 provides:

- 3. When actions are brought against the master or against members of the crew such persons may limit their liability even if the occurrence which gives rise to the claims resulted from the actual fault or privity of one or more of such persons. If, however, the master or member of the crew is at the same time the owner, co-owner, charterer, manager or operator of the ship the provisions of this paragraph shall only apply where the act, neglect or default in question is an act, neglect or default committed by the person in question in his capacity as master or as member of the crew of the ship.
- 563. The liability of the operator is also limited under article 6 of the 1976 Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources, paragraphs 1, 2 and 3 of article 6 of which read:

#### Article 6

- 1. The operator shall be entitled to limit his liability under this Convention for each installation and each incident to the amount of 30 million Special Drawing Rights until five years have elapsed from the date on which the Convention is opened for signature and to the amount of 40 million Special Drawing Rights thereafter.
- 2. Where operators of different installations are liable in accordance with paragraph 1 of article 5, the liability of the operator of any one installation shall not for any one incident exceed any limit which may be applicable to him in accordance with the provisions of this article and of article 15.
- 3. When in the case of any one installation more than one operator is liable under this Convention, the aggregate liability of all of them in respect of any one incident shall not exceed the highest amount that could be awarded against any of them, but none of them shall be liable for an amount in excess of the limit applicable to him.

Under paragraph 4 of the same article 6, the operator will not be entitled to limit his liability if it is proved that the pollution damage occurred as a result of an act or omission of the operator himself, done deliberately with actual knowledge that pollution damage will result. Two elements are thus required to remove the limitation on liability: one is an act or *omission* of the operator, and the second is *actual knowledge* that pollution damage will result. Hence the negligence of the operator does not, under this Convention, remove the limitation on liability.

## (b) Bilateral agreements

564. Although in most bilateral agreements regarding activities that might have extraterritorial injurious consequences no limitation on liability is provided for, a few of them include provisions on that question. These agreements all relate to the use of the ports of host States by the nuclear ships of other States. In the 1963 Agreement between the Netherlands and the United States of America on public liability for damage caused by the N.S. Savannah, 384 the liability of the United States is limited to \$500 million, as follows:

## Article 4

It is agreed that the aggregate liability of the United States arising out of a single nuclear incident involving the N.S. Savannah, regardless of where damage may be suffered, shall not exceed \$500 million.

565. Similarly, in the two agreements concluded in 1964 and 1965 between the United States of America

<sup>384</sup> See footnote 22 above.

and Italy concerning the N.S. Savannah, the limitation on liability was set at \$500 million, in accordance with United States legislation (see art. VIII of the 1964 Agreement and note I of the 1965 Agreement, in paras. 458 and 459 above). In a similar agreement concluded in 1964 with Ireland, 385 the liability of the United States for any damage that the N.S. Savannah might cause on Irish territory was also limited to \$500 million. Paragraphs 2 and 7 of the Agreement read:

- 2. The aggregate liability of the United States Government in accordance with paragraph 1 of this Agreement shall not exceed \$500 million for any single incident regardless of where damage may be incurred.
- 7. Subject to the \$500 million limitation referred to above, nothing in this Agreement shall affect any right which the Government of Ireland might otherwise have under international law in respect of the operation of N.S. Savannah and any claims relating thereto shall be dealt with in accordance with customary procedures for the settlement of international claims under generally accepted principles of law and equity. In particular, the two Governments will consult together, in the event of a nuclear incident, and in such consultations the question of liability and amount of compensation to those who have suffered loss or damage as a result of such incident shall be subject to the mutual agreement of the two Governments.
- 566. Again, the 1970 Treaty between Liberia and the Federal Republic of Germany<sup>386</sup> sets a limit of DM 400 million on liability for any damage that might be caused by the German nuclear ship, *Otto Hahn*, while visiting Liberian ports. Article 13 of the Treaty reads:

#### Article 13

Liability for a nuclear damage caused by a nuclear incident involving the nuclear fuel of, or radioactive products or waste produced in the ship shall be governed by article II, paragraph I of article III, article IV, article VIII, and paragraphs I and 2 of article X of the Convention [on the Liability of Operators of Nuclear Ships] as well as by the following articles of this Treaty, provided, however, that the liability mentioned in paragraph I of article III of the Convention shall be limited to DM 400 million (four hundred million).

Article 17 of the same Treaty, however, stipulates that the provisions of national legislation and international conventions on limitation of shipowners' liability shall not apply to nuclear damage under that Treaty:

## Article 17

The provisions of national legislation or international conventions on the limitation of shipowners' liability shall not apply to claims established under article 13 of this Treaty.

# (c) Judicial decisions and State practice other than agreements

567. Judicial decisions and official correspondence reveal no limitation on compensation other than that agreed upon in treaties. Some references have been made to equitable, fair and adequate compensation. By a broad interpretation, limitation on compensation may sometimes be compatible with equitable and fair compensation.

## B. Authorities competent to award compensation

568. Article 33 of the Charter of the United Nations provides a wide choice of peaceful modes of dispute settlement from the most informal to the most formal:

## Article 33

- 1. The parties to any dispute, the continuance of which is likely to endanger the maintenance of international peace and security, shall, first of all, seek a solution by negotiation, inquiry, mediation, conciliation, arbitration, judicial settlement, resort to regional agencies or arrangements, or other peaceful means of their own choice.
- 2. The Security Council shall, when it deems necessary, call upon the parties to settle their disputes by such means.
- State practice reveals that these modes of settlement of disputes have been utilized to resolve questions of liability and compensation relating to acts with extraterritorial injurious consequences. International courts, arbitral tribunals, joint commissions as well as domestic courts have decided on those questions. Generally, on the basis of prior agreements among States, the Permanent Court of International Justice, the International Court of Justice and arbitral tribunals have dealt with disputes relating to the utilization of and activities on the continental shelf, in the territorial sea, etc. When there have been ongoing activities, usually among neighbouring States, such as the use of shared waters, for which there are established institutions constituted by States, claims arising from these activities have normally been referred to the joint institution or commission concerned.

#### 1. LOCAL COURTS AND AUTHORITIES

## (a) Multilateral agreements

570. A number of multilateral agreements refer to local courts and authorities as competent authorities to decide on questions of liability and compensation. With regard to activities, primarily of a commercial nature, in which the actors are private entities and the primary liability is that of the operator, local courts have been recognized as appropriate decision makers. For example, the 1960 Convention on Third Party Liability in the Field of Nuclear Energy confers jurisdiction only on the courts of the contracting State in whose territory the nuclear installation of the operator liable is located. When the nuclear incident occurs during transportation, jurisdiction lies, unless as otherwise provided, with the courts of the contracting State in whose territory the nuclear substances involved were at the time of the incident. Article 13 of the Convention indicates in detail how jurisdiction is divided among the domestic courts of the contracting parties, according to the place of occurrence of the nuclear incident:

## Article 13

- (a) Except as otherwise provided in this article, jurisdiction over actions under articles 3, 4, 6(a) and 6(e) shall lie only with the courts of the Contracting Party in whose territory the nuclear incident occurred.
- (b) Where a nuclear incident occurs outside the territory of the Contracting Parties, or where the place of the nuclear incident cannot be determined with certainty, jurisdiction over such actions shall lie with the courts of the Contracting Party in whose territory the nuclear installation of the operator liable is situated.
- (c) Where jurisdiction would lie with the courts of more than one Contracting Party by virtue of paragraphs (a) or (b) of this article, jurisdiction shall lie.
  - (i) if the nuclear incident occurred partly outside the territory of any Contracting Party and partly in the territory of a single Contracting Party, with the courts of that Contracting Party; and

<sup>385</sup> See footnote 21 above.

<sup>186</sup> See footnote 19 above.

- (ii) in any other case, with the courts of the Contracting Party determined, at the request of a Contracting Party concerned, by the Tribunal referred to in article 17 as being the most closely related to the case in question.
- 571. Under article VIII of the 1963 Vienna Convention on Civil Liability for Nuclear Damage, the nature, form and extent of compensation, as well as its equitable distribution, are governed by the competent courts of the contracting parties:

## Article VIII

Subject to the provisions of this Convention, the nature, form and extent of the compensation, as well as the equitable distribution thereof, shall be governed by the law of the competent court.

572. The Convention further provides, in article XI, that jurisdiction lies with the domestic courts of the contracting party in whose territory the nuclear incident occurs and that, if the incident occurs outside the territory of any contracting party, or if the place of the incident cannot be determined, the courts of the installation State of the operator liable have jurisdiction:

## Article XI

- 1. Except as otherwise provided in this article, jurisdiction over actions under article II shall lie only with the courts of the Contracting Party within whose territory the nuclear incident occurred.
- 2. Where the nuclear incident occurred outside the territory of any Contracting Party, or where the place of the nuclear incident cannot be determined with certainty, jurisdiction over such actions shall lie with the courts of the Installation State of the operator liable.
- 3. Where under paragraph 1 or 2 of this article jurisdiction would lie with the courts of more than one Contracting Party, jurisdiction shall lie:
- (a) if the nuclear incident occurred partly outside the territory of any Contracting Party, and partly within the territory of a single Contracting Party, with the courts of the latter; and
- (b) in any other case, with the courts of that Contracting Party which is determined by agreement between the Contracting Parties whose courts would be competent under paragraph 1 or 2 of this article.
- 573. Article X of the 1962 Convention on the Liability of Operators of Nuclear Ships provides that action for compensation shall be brought either before the courts of the licensing State or before the courts of the contracting State or States in whose territory nuclear damage has been sustained:

## Article X

- 1. Any action for compensation shall be brought, at the option of the claimant, either before the courts of the licensing State or before the courts of the Contracting State or States in whose territory nuclear damage has been sustained.
- 2. If the licensing State has been or might be called upon to ensure the payment of claims for compensation in accordance with paragraph 2 of article III of this Convention, it may intervene as party in any proceedings brought against the operator.
- 3. Any immunity from legal processes pursuant to rules of national or international law shall be waived with respect to duties or obligations arising under, or for the purpose of, this Convention. Nothing in this Convention shall make warships or other State-owned or State-operated ships on non-commercial service liable to arrest, attachment or seizure or confer jurisdiction in respect of warships on the courts of any foreign State.
- 574. Under the 1969 International Convention on Civil Liability for Oil Pollution Damage, only the courts of the contracting State or States in whose territory, including the territorial sea, the pollution damage has occurred, or preventive measures have been taken to prevent or minimize damage, are to entertain claims for

compensation. Thus each contracting State has to ensure that its courts possess the necessary jurisdiction. Once a fund has been established in accordance with the requirements of article V of the Convention, the courts of the State where the fund is established have *exclusive* jurisdiction to decide on all matters relating to its apportionment and distribution. Article IX of the Convention reads:

#### Article IX

- 1. Where an incident has caused pollution damage in the territory including the territorial sea of one or more Contracting States, or preventive measures have been taken to prevent or minimize pollution damage in such territory including the territorial sea, actions for compensation may only be brought in the courts of any such Contracting State or States. Reasonable notice of any such action shall be given to the defendant.
- Each Contracting State shall ensure that its courts possess the necessary jurisdiction to entertain such actions for compensation.
- 3. After the fund has been constituted in accordance with article V the Courts of the State in which the fund is constituted shall be exclusively competent to determine all matters relating to the apportionment and distribution of the fund.
- 575. Under article XI of the Convention, the domestic courts also have jurisdiction in respect of ships owned by a contracting State and used for commercial purposes:

## Article XI

- 2. With respect to ships owned by a Contracting State and used for commercial purposes, each State shall be subject to suit in the jurisdictions set forth in Article IX and shall waive all defences based on its status as a sovereign State.
- 576. Similarly, the 1971 International Convention on the Establishment of an International Fund for Compensation for Oil Pollution Damage provides that the domestic courts of the contracting parties are competent to decide on actions against the Fund, and that the contracting States must endow their courts with the necessary jurisdiction to entertain such actions. Article 7 of the Convention reads in part:

## Article 7

- 1. Subject to the subsequent provisions of this article, any action against the Fund for compensation under article 4 or indemnification under article 5 of this Convention shall be brought only before a court competent under article IX of the Liability Convention [International Convention on Civil Liability for Oil Pollution Damage] in respect of actions against the owner who is or who would, but for the provisions of article III, paragraph 2, of that Convention, have been liable for pollution damage caused by the relevant incident.
- Each Contracting State shall ensure that its courts possess the necessary jurisdiction to entertain such actions against the Fund as are referred to in paragraph 1.
- 3. Where an action for compensation for pollution damage has been brought before a court competent under article IX of the Liability Convention against the owner of a ship or his guarantor, such court shall have exclusive jurisdictional competence over any action against the Fund for compensation or indemnification under the provisions of article 4 or 5 of this Convention in respect of the same damage. However, where an action for compensation for pollution damage under the Liability Convention has been brought before a court in a State Party to the Liability Convention but not to this Convention, any action against the Fund under article 4 or under article 5, paragraph 1, of this Convention shall at the option of the claimant be brought either before a court of the State where the Fund has its head-quarters or before any court of a State Party to this Convention competent under article IX of the Liability Convention.
- 4. Each Contracting State shall ensure that the Fund shall have the right to intervene as a party to any legal proceedings instituted in ac-

cordance with article IX of the Liability Convention before a competent court of that State against the owner of a ship or his guarantor.

577. The 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 25 February 1961 relating to the liability of the railway for death of and personal injury to passengers provides that, unless otherwise agreed upon by States, or stipulated in the licence of the railway, the domestic courts of the State in whose territory the accident to the passenger occurs are competent to entertain actions for compensation. Article 15 of the Convention reads:

#### Article 15. Jurisdiction

Actions brought under this Convention may only be instituted in the competent court of the State on whose territory the accident to the passenger occurred, unless otherwise provided in agreements between States, or in any licence or other document authorizing the operation of the railway concerned.

578. Under the 1974 Convention on the Protection of the Environment between Denmark, Finland, Norway and Sweden, the nuisance which an activity entails or may entail in the territory of another contracting State is equated with a nuisance in the State where the activity is carried out. Thus any person who is or may be affected by such a nuisance may bring a claim before the court or administrative authority of that State for compensation. The rules on compensation must not be less favourable to the injured party than those in the State where the activity is carried out. Indeed, the Convention provides for equal access to the competent authorities and for equal treatment of the injured parties, whether local or foreign. The relevant articles of the Convention read:

## Article 2

In considering the permissibility of environmentally harmful activities, the nuisance which such activities entail or may entail in another Contracting State shall be equated with a nuisance in the State where the activities are carried out.

## Article 3

Any person who is affected or may be affected by a nuisance caused by environmentally harmful activities in another Contracting State shall have the right to bring before the appropriate court or administrative authority of that State the question of the permissibility of such activities, including the question of measures to prevent damage, and to appeal against the decision of the court or the administrative authority to the same extent and on the same terms as a legal entity of the State in which the activities are being carried out.

The provisions of the first paragraph of this article shall be equally applicable in the case of proceedings concerning compensation for damage caused by environmentally harmful activities. The question of compensation shall not be judged by rules which are less favourable to the injured party than the rules of compensation of the State in which the activities are being carried out.

## Protocol

. . .

The right established in article 3 for anyone who suffers injury as a result of environmentally harmful activities in a neighbouring State to institute proceedings for compensation before a court or administrative authority of that State shall, in principle, be regarded as including the right to demand the purchase of his real property.

579. Under article 11 of the 1976 Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources, the competent authorities to decide on ques-

tions of liability and compensation are the national courts of either the controlling State or the State in whose territory the damage has occurred. Each contracting party is required to ensure that its courts possess the necessary jurisdiction to entertain actions for compensation. It appears, under the Convention, that the national courts are to apply both the Convention and their domestic law, the former for questions of liability and compensation and the latter for evidentiary and procedural matters. However, only the courts of a State party in which a fund has been constituted are competent to determine all matters relating to the apportionment and distribution of that fund. Article 11 of the Convention reads:

#### Article 11

- 1. Actions for compensation under this convention may be brought only in the courts of any State Party where pollution damage was suffered as a result of the incident or in the courts of the controlling State. For the purpose of determining where the damage was suffered, damage suffered in an area in which, in accordance with international law, a State has sovereign rights over natural resources shall be deemed to have been suffered in that State.
- 2. Each State Party shall ensure that its courts possess the necessary jurisdiction to entertain such actions for compensation.
- 3. After the fund has been constituted in accordance with article 6, the courts of the State Party in which the fund is constituted shall be exclusively competent to determine all matters relating to the apportionment and distribution of the fund.

Furthermore, if a well has been abandoned in circumstances other than those provided in the Convention, the liability of the operator is governed by the applicable domestic law. The relevant passage of article 3, paragraph 4, reads:

- . . . Where a well has been abandoned in other circumstances, the liability of the operator shall be governed by the applicable national law.
- 580. Under article 232 of the 1982 United Nations Convention on the Law of the Sea, States are liable for damage or loss attributable to them arising from measures taken in accordance with section 6 of part XII, relating to the protection and preservation of the marine environment, when such measures are *unlawful* or *exceed those reasonably required*. Accordingly, States are required to endow their courts with appropriate jurisdiction to deal with actions brought in respect of such loss or damage (see para. 246 above).

## (b) Bilateral agreements

- 581. Some bilateral agreements recognize the competence of national courts to decide on questions of liability and compensation. The 1929 Convention between Norway and Sweden relating to their frontier waters<sup>387</sup> provides, in article 6, that the law of the country in whose territory *the damage has occurred* shall govern the question of compensation (see para. 521 above). However, the article does not indicate which court is competent. It seems that it is the courts of the country in which the injury has occurred that are competent to decide on compensation.
- 582. The Agreements concluded by the United States of America with Italy and with the Netherlands provide that the authorities and local courts of the host States

<sup>387</sup> See footnote 36 above.

shall be competent to decide on the liability of the United States in case of injuries caused by the United States nuclear ship, the N.S. Savannah, in their territories. The relevant paragraph of the 1965 Agreement between the United States and Italy concerning liability<sup>388</sup> reads:

... Within the \$500 million limitation in such public laws [Public Laws 85-256 and 85-602] the operator of the ship shall be subject to the jurisdiction of the *Italian court*\*...

583. Article 5 of the 1963 Agreement between the Netherlands and the United States on public liability<sup>389</sup> reads:

#### Article 5

The United States agrees to submit to proceedings before any competent court of the Netherlands\* or before any other body established under Netherlands law for the purpose of considering and determining liability for damage as described in article 1.

Article 1 of the same Agreement provides that the Netherlands courts or a commission established in accordance with Netherlands law shall determine United States liability for certain nuclear damage (see para. 460 above). It is further provided, in article 2, that the United States shall indemnify any person who is held liable under the law of a country other than the Netherlands for damage caused in Netherlands territory:

#### Article 2

The United States shall indemnify any person who on account of any act or omission committed on Netherlands territory is held liable for public liability under the law of a country other than the Netherlands for damage as described in article 1.

# (c) Judicial decisions and State practice other than agreements

584. Judicial decisions and official correspondence contain no indication concerning the competence of local courts and authorities to rule on questions of liability and compensation, except possibly on the distribution of lump sum payments. However, press reports indicate the desire of the United States Government to settle claims against the United States arising out of the nuclear tests in the Marshall Islands; apparently suits seeking over \$4 billion have already been filed in United States courts and others are in progress.<sup>390</sup> Islanders from Bikini, for example, whose largest island had remained radioactive two decades after the last test, were seeking \$450 million. It appears that settlements on compensation have in most cases been reached by negotiation between the authorities of the Governments concerned.

## 2. International courts, arbitral tribunals and joint commissions

## (a) Multilateral agreements

585. In the case of activities not exclusively of a commercial nature, in which the acting entities are primarily States, the competent organs for deciding on questions of liability and compensation are generally arbitral

tribunals. The 1972 Convention on International Liability for Damage Caused by Space Objects provides that, if the parties fail to reach agreement through diplomatic negotiations, the question of compensation shall be submitted to arbitration. Accordingly, a claims commission composed of three members, one appointed by the claimant State, one appointed by the launching State and a chairman, is to be established upon the request of either party. The relevant articles of the Convention read:

#### Article VIII

- 1. A State which suffers damage, or whose natural or juridical persons suffer damage, may present to a launching State a claim for compensation for such damage.
- 2. If the State of nationality has not presented a claim, another State may, in respect of damage sustained in its territory by any natural or juridical person, present a claim to a launching State.
- 3. If neither the State of nationality nor the State in whose territory the damage was sustained has presented a claim or notified its intention of presenting a claim, another State may, in respect of damage sustained by its permanent residents, present a claim to a launching State.

#### Article IX

A claim for compensation for damage shall be presented to a launching State through diplomatic channels. If a State does not maintain diplomatic relations with the launching State concerned, it may request another State to present its claim to that launching State or otherwise represent its interests under this Convention. It may also present its claim through the Secretary-General of the United Nations, provided the claimant State and the launching State are both Members of the United Nations.

## Article XI

- 1. Presentation of a claim to a launching State for compensation for damage under this Convention shall not require the prior exhaustion of any local remedies which may be available to a claimant State or to natural or juridical persons it represents.
- 2. Nothing in this Convention shall prevent a State, or natural or juridical persons it might represent, from pursuing a claim in the courts or administrative tribunals or agencies of a launching State. A State shall not, however, be entitled to present a claim under this Convention in respect of the same damage for which a claim is being pursued in the courts or administrative tribunals or agencies of a launching State or under another international agreement which is binding on the States concerned.

## Article XIV

If no settlement of a claim is arrived at through diplomatic negotiations as provided for in article IX, within one year from the date on which the claimant State notifies the launching State that it has submitted the documentation of its claim, the parties concerned shall establish a Claims Commission at the request of either party.

## Article XV

- 1. The Claims Commission shall be composed of three members: one appointed by the claimant State, one appointed by the launching State and the third member, the chairman, to be chosen by both parties jointly. Each party shall make its appointment within two months of the request for the establishment of the Claims Commission.
- 2. If no agreement is reached on the choice of the chairman within four months of the request for the establishment of the Commission, either party may request the Secretary-General of the United Nations to appoint the chairman within a further period of two months.

## Article XVI

- 1. If one of the parties does not make its appointment within the stipulated period, the chairman shall, at the request of the other party, constitute a single-member Claims Commission.
- 2. Any vacancy which may arise in the Commission for whatever reason shall be filled by the same procedure adopted for the original appointment.

<sup>388</sup> See footnote 20 above.

<sup>389</sup> See footnote 22 above.

<sup>&</sup>lt;sup>390</sup> International Herald Tribune, 15 June 1982, p. 5, col. 2.

- 3. The Commission shall determine its own procedure.
- 4. The Commission shall determine the place or places where it shall sit and all other administrative matters.
- 5. Except in the case of decisions and awards by a single-member Commission, all decisions and awards of the Commission shall be by majority vote.

. . .

#### Article XVIII

The Claims Commission shall decide the merits of the claim for compensation and determine the amount of compensation payable, if any.

586. In part XV of the 1982 United Nations Convention on the Law of the Sea, the parties are encouraged and requested to settle their disputes by peaceful means. The Convention provides for a wide range of possible modes of settlement of disputes, as well as for an elaborate system according to which the competent organs for deciding on a dispute, depending upon the nature of the dispute, are the International Tribunal for the Law of the Sea, or the International Court of Justice, or an arbitral tribunal, Articles 278 to 285 set out the modes of settlement compatible with Article 33 of the Charter. The States parties are required in the first place to settle their disputes by peaceful means (art. 279). They may agree on a peaceful means of their choice to settle disputes concerning the interpretation and application of the Convention (art. 280). Disputes may also be settled in the framework of regional or other agreements in force between the parties, at the option of any party to the dispute (art. 282). The States parties are required to proceed promptly to an exchange of views for the purpose of agreeing on a suitable mode of peaceful settlement (art. 283). Conciliation is also provided for in article 284, with its procedure (annex V, sect. 1). The parties may agree on a different procedure for conciliation. If the parties cannot agree on a procedure of their choice, the Convention lays down compulsory procedures leading to binding decisions, in which case the competent forums are the International Court of Justice, the International Tribunal for the Law of the Sea and ad hoc tribunals (arts. 286-299). Under article 295 of the Convention, the parties, before submitting their disputes to this compulsory procedure, must have exhausted local remedies where this is required by international law. This article may be interpreted as referring to the exhaustion of the remedies available in domestic and administrative courts, as well as negotiation with the competent authorities of the acting State. The provisions of the Convention on the settlement of disputes are as follows:

## Part XV. Settlement of disputes

SECTION 1. GENERAL PROVISIONS

Article 279. Obligation to settle disputes by peaceful means

States Parties shall settle any dispute between them concerning the interpretation or application of this Convention by peaceful means in accordance with Article 2, paragraph 3, of the Charter of the United Nations and, to this end, shall seek a solution by the means indicated in Article 33, paragraph 1, of the Charter.

Article 280. Settlement of disputes by any peaceful means chosen by the parties

Nothing in this Part impairs the right of any States Parties to agree at any time to settle a dispute between them concerning the interpretation or application of this Convention by any peaceful means of their own choice.

## Article 281. Procedure where no settlement has been reached by the parties

- 1. If the States Parties which are parties to a dispute concerning the interpretation or application of this Convention have agreed to seek settlement of the dispute by a peaceful means of their own choice, the procedures provided for in this Part apply only where no settlement has been reached by recourse to such means and the agreement between the parties does not exclude any further procedure.
- 2. If the parties have also agreed on a time-limit, paragraph 1 applies only upon the expiration of that time-limit.

## Article 282. Obligations under general, regional or bilateral agreements

If the States Parties which are parties to a dispute concerning the interpretation or application of this Convention have agreed, through a general, regional or bilateral agreement or otherwise, that such dispute shall, at the request of any party to the dispute, be submitted to a procedure that entails a binding decision, that procedure shall apply in lieu of the procedures provided for in this Part, unless the parties to the dispute otherwise agree.

## Article 283. Obligation to exchange views

- 1. When a dispute arises between States Parties concerning the interpretation or application of this Convention, the parties to the dispute shall proceed expeditiously to an exchange of views regarding its settlement by negotiation or other peaceful means.
- 2. The parties shall also proceed expeditiously to an exchange of views where a procedure for the settlement of such a dispute has been terminated without a settlement or where a settlement has been reached and the circumstances require consultation regarding the manner of implementing the settlement.

#### Article 284. Conciliation

- 1. A State Party which is a party to a dispute concerning the interpretation or application of this Convention may invite the other party or parties to submit the dispute to conciliation in accordance with the procedure under annex V, section 1, or another conciliation procedure.
- 2. If the invitation is accepted and if the parties agree upon the conciliation procedure to be applied, any party may submit the dispute to that procedure.
- 3. If the invitation is not accepted or the parties do not agree upon the procedure, the conciliation proceedings shall be deemed to be terminated.
- 4. Unless the parties otherwise agree, when a dispute has been submitted to conciliation, the proceedings may be terminated only in accordance with the agreed conciliation procedure.

## Article 285. Application of this section to disputes submitted pursuant to Part XI

This section applies to any dispute which pursuant to Part XI, section 5, is to be settled in accordance with procedures provided for in this Part. If an entity other than a State Party is a party to such a dispute, this section applies mutatis mutandis.

## Section 2. Compulsory procedures entailing binding decisions

## Article 286. Application of procedures under this section

Subject to section 3, any dispute concerning the interpretation or application of this Convention shall, where no settlement has been reached by recourse to section 1, be submitted at the request of any party to the dispute to the court or tribunal having jurisdiction under this section.

## Article 287. Choice of procedure

- 1. When signing, ratifying or acceding to this Convention or at any time thereafter, a State shall be free to choose, by means of a written declaration, one or more of the following means for the settlement of disputes concerning the interpretation or application of this Convention:
- (a) the International Tribunal for the Law of the Sea established in accordance with annex VI;
  - (b) the International Court of Justice:
  - (c) an arbitral tribunal constituted in accordance with annex VII;

- (d) a special arbitral tribunal constituted in accordance with annex VIII for one or more of the categories of disputes specified therein.
- 2. A declaration made under paragraph 1 shall not affect or be affected by the obligation of a State Party to accept the jurisdiction of the Sea-Bed Disputes Chamber of the International Tribunal for the Law of the Sea to the extent and in the manner provided for in Part X1, section 5.
- 3. A State Party, which is a party to a dispute not covered by a declaration in force, shall be deemed to have accepted arbitration in accordance with annex VII.
- 4. If the parties to a dispute have accepted the same procedure for the settlement of the dispute, it may be submitted only to that procedure, unless the parties otherwise agree.
- 5. If the parties to a dispute have not accepted the same procedure for the settlement of the dispute, it may be submitted only to arbitration in accordance with annex VII, unless the parties otherwise agree.
- 6. A declaration made under paragraph 1 shall remain in force until three months after notice of revocation has been deposited with the Secretary-General of the United Nations.
- 7. A new declaration, a notice of revocation or the expiry of a declaration does not in any way affect proceedings pending before a court or tribunal having jurisdiction under this article, unless the parties otherwise agree.
- 8. Declarations and notices referred to in this article shall be deposited with the Secretary-General of the United Nations, who shall transmit copies thereof to the States Parties.

#### Article 288. Jurisdiction

- 1. A court or tribunal referred to in article 287 shall have jurisdiction over any dispute concerning the interpretation or application of this Convention which is submitted to it in accordance with this Part.
- 2. A court or tribunal referred to in article 287 shall also have jurisdiction over any dispute concerning the interpretation or application of an international agreement related to the purposes of this Convention, which is submitted to it in accordance with the agreement.
- 3. The Sea-Bed Disputes Chamber of the International Tribunal for the Law of the Sea established in accordance with annex VI, and any other chamber or arbitral tribunal referred to in Part XI, section 5, shall have jurisdiction in any matter which is submitted to it in accordance therewith.
- 4. In the event of a dispute as to whether a court or tribunal has jurisdiction, the matter shall be settled by decision of that court or tribunal.

## Article 289. Experts

In any dispute involving scientific or technical matters, a court or tribunal exercising jurisdiction under this section may, at the request of a party or *proprio inotu*, select in consultation with the parties no fewer than two scientific or technical experts chosen preferably from the relevant list prepared in accordance with annex VIII, article 2, to sit with the court or tribunal but without the right to vote.

## Article 290. Provisional ineasures

- 1. If a dispute has been duly submitted to a court or tribunal which considers that *prima facie* it has jurisdiction under this Part or Part XI, section 5, the court or tribunal may prescribe any provisional measures which it considers appropriate under the circumstances to preserve the respective rights of the parties to the dispute or to prevent serious harm to the marine environment, pending the final decision.
- 2. Provisional measures may be modified or revoked as soon as the circumstances justifying them have changed or ceased to exist.
- 3. Provisional measures may be prescribed, modified or revoked under this article only at the request of a party to the dispute and after the parties have been given an opportunity to be heard.
- 4. The court or tribunal shall forthwith give notice to the parties to the dispute, and to such other States Parties as it considers appropriate, of the prescription, modification or revocation of provisional measures.
- 5. Pending the constitution of an arbitral tribunal to which a dispute is being submitted under this section, any court or tribunal agreed upon by the parties or, failing such agreement within two weeks from the date of the request for provisional measures, the International Tribunal for the Law of the Sea or, with respect to activities in the Area, the Sea-Bed Disputes Chamber, may prescribe, modify or

revoke provisional measures in accordance with this article if it considers that *prima facie* the tribunal which is to be constituted would have jurisdiction and that the urgency of the situation so requires. Once constituted, the tribunal to which the dispute has been submitted may modify, revoke or affirm those provisional measures, acting in conformity with paragraphs 1 to 4.

6. The parties to the dispute shall comply promptly with any provisional measures prescribed under this article.

#### Article 291. Access

- 1. All the dispute settlement procedures specified in this Part shall be open to States Parties.
- 2. The dispute settlement procedures specified in this Part shall be open to entities other than States Parties only as specifically provided for in this Convention.

## Article 292. Prompt release of vessels and crews

- 1. Where the authorities of a State Party have detained a vessel flying the flag of another State Party and it is alleged that the detaining State has not complied with the provisions of this Convention for the prompt release of the vessel or its crew upon the posting of a reasonable bond or other financial security, the question of release from detention may be submitted to any court or tribunal agreed upon by the parties or, failing such agreement within 10 days from the time of detention, to a court or tribunal accepted by the detaining State under article 287 or to the International Tribunal for the Law of the Sea, unless the parties otherwise agree.
- 2. The application for release may be made only by or on behalf of the flag State of the vessel.
- 3. The court or tribunal shall deal without delay with the application for release and shall deal only with the question of release, without prejudice to the merits of any case before the appropriate domestic forum against the vessel, its owner or its crew. The authorities of the detaining State remain competent to release the vessel or its crew at any time.
- 4. Upon the posting of the bond or other financial security determined by the court or tribunal, the authorities of the detaining State shall comply promptly with the decision of the court or tribunal concerning the release of the vessel or its crew.

## Article 293. Applicable law

- 1. A court or tribunal having jurisdiction under this section shall apply this Convention and other rules of international law not incompatible with this Convention.
- 2. Paragraph 1 does not prejudice the power of the court or tribunal having jurisdiction under this section to decide a case ex aequo et bono, if the parties so agree.

## Article 294. Preliminary proceedings

- 1. A court or tribunal provided for in article 287 to which an application is made in respect of a dispute referred to in article 297 shall determine at the request of a party, or may determine proprio motu, whether the claim constitutes an abuse of legal process or whether prima facie it is well founded. If the court or tribunal determines that the claim constitutes an abuse of legal process or is prima facie unfounded, it shall take no further action in the case.
- 2. Upon receipt of the application, the court or tribunal shall immediately notify the other party or parties of the application, and shall fix a reasonable time-limit within which they may request it to make a determination in accordance with paragraph 1.
- 3. Nothing in this article affects the right of any party to a dispute to make preliminary objections in accordance with the applicable rules of procedure.

## Article 295. Exhaustion of local remedies

Any dispute between States Parties concerning the interpretation or application of this Convention may be submitted to the procedures provided for in this section only after local remedies have been exhausted where this is required by international law.

## Article 296. Finality and binding force of decisions

- 1. Any decision rendered by a court or tribunal having jurisdiction under this section shall be final and shall be complied with by all the parties to the dispute.
- 2. Any such decision shall have no binding force except between the parties and in respect of that particular dispute.

#### Section 3. Limitations and exceptions to Applicability of section 2

Article 297. Limitations on applicability of section 2

- 1. Disputes concerning the interpretation or application of this Convention with regard to the exercise by a coastal State of its sovereign rights or jurisdiction provided for in this Convention shall be subject to the procedures provided for in section 2 in the following cases:
- (a) when it is alleged that a coastal State has acted in contravention of the provisions of this Convention in regard to the freedoms and rights of navigation, overflight or the laying of submarine cables and pipelines, or in regard to other internationally lawful uses of the sea specified in article 58;
- (b) when it is alleged that a State in exercising the aforementioned freedoms, rights or uses has acted in contravention of this Convention or of laws or regulations adopted by the coastal State in conformity with this Convention and other rules of international law not incompatible with this Convention; or
- (c) when it is alleged that a coastal State has acted in contravention of specified international rules and standards for the protection and preservation of the marine environment which are applicable to the coastal State and which have been established by this Convention or through a competent international organization or diplomatic conference in accordance with this Convention.
- 2. (a) Disputes concerning the interpretation or application of the provisions of this Convention with regard to marine scientific research shall be settled in accordance with section 2, except that the coastal State shall not be obliged to accept the submission to such settlement of any dispute arising out of:
  - (i) the exercise by the coastal State of a right or discretion in accordance with article 246; or
  - (ii) a decision by the coastal State to order suspension or cessation of a research project in accordance with article 253.
- (b) A dispute arising from an allegation by the researching State that with respect to a specific project the coastal State is not exercising its rights under articles 246 and 253 in a manner compatible with this Convention shall be submitted, at the request of either party, to conciliation under annex V, section 2, provided that the conciliation commission shall not call in question the exercise by the coastal State of its discretion to designate specific areas as referred to in article 246, paragraph 6, or of its discretion to withhold consent in accordance with article 246, paragraph 5.
- 3. (a) Disputes concerning the interpretation or application of the provisions of this Convention with regard to fisheries shall be settled in accordance with section 2, except that the coastal State shall not be obliged to accept the submission to such settlement of any dispute relating to its sovereign rights with respect to the living resources in the exclusive economic zone or their exercise, including its discretionary powers for determining the allowable catch, its harvesting capacity, the allocation of surpluses to other States and the terms and conditions established in its conservation and management laws and regulations.
- (b) Where no settlement has been reached by recourse to section 1 of this Part, a dispute shall be submitted to conciliation under annex V, section 2, at the request of any party to the dispute, when it is alleged that:
  - (i) a coastal State has manifestly failed to comply with its obligations to ensure through proper conservation and management measures that the maintenance of the living resources in the exclusive economic zone is not seriously endangered;
  - (ii) a coastal State has arbitrarily refused to determine, at the request of another State, the allowable catch and its capacity to harvest living resources with respect to stocks which that other State is interested in fishing; or
  - (iii) a coastal State has arbitrarily refused to allocate to any State, under articles 62, 69 and 70 and under the terms and conditions established by the coastal State consistent with this Convention, the whole or part of the surplus it has declared to exist.
- (c) In no case shall the conciliation commission substitute its discretion for that of the coastal State.
- (d) The report of the conciliation commission shall be communicated to the appropriate international organizations.
- (e) In negotiating agreements pursuant to articles 69 and 70, States Parties, unless they otherwise agree, shall include a clause on measures

which they shall take in order to minimize the possibility of a disagreement concerning the interpretation or application of the agreement, and on how they should proceed if a disagreement nevertheless arises.

#### Article 298. Optional exceptions to applicability of section 2

- 1. When signing, ratifying or acceding to this Convention or at any time thereafter, a State may, without prejudice to the obligations arising under section 1, declare in writing that it does not accept any one or more of the procedures provided for in section 2 with respect to one or more of the following categories of disputes:
  - (a) (i) disputes concerning the interpretation or application of articles 15, 74 and 83 relating to sea boundary delimitations, or those involving historic bays or titles, provided that a State having made such a declaration shall, when such a dispute arises subsequent to the entry into force of this Convention and where no agreement within a reasonable period of time is reached in negotiations between the parties, at the request of any party to the dispute, accept submission of the matter to conciliation under annex V, section 2; and provided further that any dispute that necessarily involves the concurrent consideration of any unsettled dispute concerning sovereignty or other rights over continental or insular land territory shall be excluded from such submission;
    - (ii) after the conciliation commission has presented its report, which shall state the reasons on which it is based, the parties shall negotiate an agreement on the basis of that report; if these negotiations do not result in an agreement, the parties, or to any such dispute which is to be settled in accordance with a bilateral or multilateral agreement binding upon those parties;
    - (iii) this subparagraph does not apply to any sea boundary dispute finally settled by an arrangement between the parties, or to any such dispute which is to be settled in accordance with a bilateral or multilateral agreement binding upon those parties;
- (b) disputes concerning military activities, including military activities by government vessels and aircraft engaged in non-commercial service, and disputes concerning law enforcement activities in regard to the exercise of sovereign rights or jurisdiction excluded from the jurisdiction of a court or tribunal under article 297, paragraph 2 or 3;
- (c) disputes in respect of which the Security Council of the United Nations is exercising the functions assigned to it by the Charter of the United Nations, unless the Security Council decides to remove the matter from its agenda or calls upon the parties to settle it by the means provided for in this Convention.
- 2. A State Party which has made a declaration under paragraph 1 may at any time withdraw it, or agree to submit a dispute excluded by such declaration to any procedure specified in this Convention.
- 3. A State Party which has made a declaration under paragraph 1 shall not be entitled to submit any dispute falling within the excepted category of disputes to any procedure in this Convention as against another State Party, without the consent of that party.
- 4. If one of the States Parties has made a declaration under paragraph 1 (a), any other State Party may submit any dispute falling within an excepted category against the declarant party to the procedure specified in such declaration.
- 5. A new declaration, or the withdrawal of a declaration, does not in any way affect proceedings pending before a court or tribunal in accordance with this article, unless the parties otherwise agree.
- 6. Declarations and notices of withdrawal of declarations under this article shall be deposited with the Secretary-General of the United Nations, who shall transmit copies thereof to the States Parties.

## Article 299. Right of the parties to agree upon a procedure

- 1. A dispute excluded under article 297 or excepted by a declaration made under article 298 from the dispute settlement procedures provided for in section 2 may be submitted to such procedures only by agreement of the parties to the dispute.
- 2. Nothing in this section impairs the right of the parties to the dispute to agree to some other procedure for the settlement of such dispute or to reach an amicable settlement.

## (b) Bilateral agreements

587. Under a number of bilateral agreements, separate bodies are set up as authorities competent to decide on questions of compensation. Under the 1971 Agreement between Finland and Sweden concerning frontier rivers, the Frontier River Commission is competent to take decisions on questions of compensation arising from activities within the scope of the Agreement. Article 2 of chapter 7 reads:

#### Article 2

The Frontier River Commission may also take decisions otherwise than in connexion with applications for permission on questions of compensation arising from measures falling within the scope of this Agreement.

Compensation for damage and inconvenience resulting from the measures referred to in chapter 3, article 21, shall, in the absence of agreement, be fixed by the Frontier River Commission.

Article 3 of the same chapter states that the applicable law is that of the country in whose territory the injury has taken place:

#### Article 3

Save as otherwise provided in this Agreement, the law of the State in which the property used is situated or in which loss, damage or inconvenience otherwise occurs shall apply in respect of the grounds for compensation, the right of the owner of property used or damaged to demand payment and the manner and time of payment of compensation.

588. The 1974 Agreement between the United States of America and Canada relating to certain rocket launches<sup>391</sup> provides that, if the claims arising from those launches are not settled through negotiation, the two Governments may establish a Claims Commission, as provided for in article XV of the Convention on International Liability for Damage Caused by Space Objects. The relevant paragraph of the Agreement (note I) reads:

In the event that a claim arising out of these launches is not settled expeditiously in a mutually acceptable manner, the Government of the United States and the Government of Canada shall give consideration to the establishment of a Claims Commission\* such as that provided for in article XV of the Convention on International Liability for Damage Caused by Space Objects with a view to arriving at a prompt and equitable settlement.

# (c) Judicial decisions and State practice other than agreements

589. Most judicial decisions in this matter have been rendered by the Permanent Court of International Justice, by the International Court of Justice or by arbitral tribunals on the basis of agreement between the parties or of a prior treaty obligation. At least one arbitral tribunal, that called upon to adjudicate in the *Trail Smelter* case, provided in its award for an arbitration mechanism in the event that the States parties were unable to agree on the modification or amendment of the régime proposed by one side (see para. 296 above).

## 3. APPLICABLE LAW

## (a) Multilateral agreements

590. International law is applicable in disputes arising from activities conducted solely by States. Domestic

laws, on the other hand, are applicable in disputes arising from activities mainly of a commercial nature, which are substantially dominated by private entities. The 1972 Convention on International Liability for Damage Caused by Space Objects regulates space activities at present controlled by States and provides that international law and the principles of justice and equity are the applicable law in accordance with which compensation and such reparation in respect of the damage as will restore the person, natural or juridical, State or international organization, shall be accorded. Article XII of the Convention reads:

## Article XII

The compensation which the launching State shall be liable to pay for damage under this Convention shall be determined in accordance with international law and the principles of justice and equity, in order to provide such reparation in respect of the damage as will restore the person, natural or juridical, State or international organization on whose behalf the claim is presented to the condition which would have existed if the damage had not occurred.

- 591. Similarly, article 293 of the 1982 United Nations Convention on the Law of the Sea provides that, when a court (that is, the International Court of Justice or the International Tribunal for the Law of the Sea) or a tribunal having jurisdiction, in accordance with section 2 of part XV of the Convention, to rule in a dispute concerning the application or interpretation of the Convention, it shall apply the provisions of the Convention and other rules of international law not incompatible with the Convention. However, if the parties to a dispute agree, the court or tribunal can adjudicate ex aequo et bono. (See para. 586 above for the text of section 2 of part XV, including article 293.)
- 592. On the other hand, the 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 26 February 1961 relating to the liability of the railway for death of and personal injury to passengers, which regulates an essentially commercial activity, provides in article 6, paragraph 2, for the application of national law:

Article 6. Form and limit of damages in case of death of or personal injury to the passenger

- 2. The amount of damages to be awarded under paragraph 1 shall be determined in accordance with national law. However, in the event of the national law providing for a maximum limit of less than 200,000 francs, the limit per passenger shall, for the purposes of this Convention, be fixed at 200,000 francs in the form of a lump sum or of an annuity corresponding to that amount.
- 593. Similarly, the 1962 Convention on the Liability of Operators of Nuclear Ships provides in article VI for the application of *national law*:

## Article VI

Where provisions of national health insurance, social insurance, social security, workmen's compensation or occupational disease compensation systems include compensation for nuclear damage, rights of beneficiaries under such systems and rights of subrogation, or of recourse against the operator, by virtue of such systems, shall be determined by the law of the Contracting State having established such systems. However, if the law of such Contracting State allows claims of beneficiaries of such systems and such rights of subrogation and recourse to be brought against the operator in conformity with the terms of this Convention, this shall not result in the liability of the operator exceeding the amount specified in paragraph 1 of article III.

<sup>391</sup> See footnote 84 above.

- 594. Under article 5, paragraph 5, of the 1957 International Convention relating to the Limitation of the Liability of Owners of Seagoing Ships, claims for liability and compensation are to be brought before the appropriate national courts of the contracting parties and the *national law* is applicable as far as the *procedure* of bringing such claims is concerned, and also as to the *time limits* within which such actions shall be brought or prosecuted. The provision reads:
- 5. Questions of procedure relating to actions brought under the provisions of this Convention and also the time limit within which such actions shall be brought or prosecuted shall be decided in accordance with the national law of the Contracting State in which the action takes place.

The Convention provides further, in article 1, paragraph 6, that the *national law* shall determine the question upon whom lies the burden of proving whether or not the accident causing the injury resulted from a fault (see para. 562 above).

## (b) Bilateral agreements

595. In bilateral agreements, the law applicable in a dispute is contained in the provisions of these agreements themselves. In a few agreements, however, the domestic law of one of the parties is recognized as the applicable law. For example, article 6 of the 1929 Convention between Norway and Sweden concerning frontier waters<sup>392</sup> provides that the question of compensation shall be governed by the law of the country in whose territory the damage has occurred (see para. 521 above).

596. In agreements concluded with Italy and with the Netherlands concerning its nuclear ship, the N.S. Savannah, the United States of America recognized the competence of the Italian and Netherlands courts to adjudicate on the liability of the United States for any injuries that the ship might cause in the territories of those two countries. Article 1 of the 1963 Agreement concluded with the Netherlands<sup>393</sup> provides for the application of the existing principles of law (see para. 460 above).

597. Article VIII of the 1964 Agreement concluded with Italy<sup>394</sup> recognizes the competence of the Italian courts and the applicability of Italian law (see para. 458 above).

598. By the 1965 Agreement with Italy, 395 which nullified the 1964 Agreement following the operation of the N.S. Savannah by a private entity, the United States recognized the competence of the Italian courts to adjudicate in matters of liability for damage, but without indicating what would be the applicable law. The new Agreement provides as follows:

In view of the inapplicability of the Agreement of November 23, 1964 to the new situation, the Embassy proposes that the following shall constitute the agreement between the two Governments in the new situation.

Within the limitation of liability set by United States Public Law 85-256 (Annex A), as amended by 85-602 (Annex B) in any legal action

or proceeding brought in personam against the operator to the N.S. Savannah in an Italian court, the United States Government will provide compensation by way of indemnity for any legal liability which an Italian court may find for any damage to people or goods deriving from a nuclear incident in connection with, arising out of or resulting from the operation, repair, maintenance or use of the N.S. Savannah, in which the N.S. Savannah may be involved within Italian territorial waters, or outside of them on a voyage to or from Italian ports if damage is caused in Italy or on ships of Italian registry. Within the \$500 million limitation in such public laws, the operator of the ship shall be subject to the jurisdiction of the Italian court and shall not invoke the provisions of Italian law or any other law relating to the limitation of shipowner's liability.

599. The 1965 Agreement provides that the operator of the ship shall not invoke the provisions of Italian law to limit its liability. This language, together with the reference to the jurisdiction of Italian courts, may be taken to imply that Italian law is applicable in the same way as in the 1964 Agreement. In the two abovementioned agreements, the limitation of liability set by United States Public Law 85-256, as amended by Public Law 85-602, is also applicable. Similarly, the Agreement concluded with the Netherlands in 1963 on public liability for damage provides in article 6 that the terms "persons indemnified", "public liability" and "nuclear incident" have the same meaning as in the definitions of those terms in section 11 of the United States Atomic Energy Act of 1954, as amended:

#### Article 6

As used in this Agreement and its annex, the terms "persons indemnified", "public liability" and "nuclear incident" have the same meaning as in the definitions of those terms found in Section 11 of the United States Atomic Energy Act of 1954, as amended (U.S. Code, Title 42, Section 2014).

600. Hence it appears that, in addition to the provisions of treaties and of the domestic laws of both "potentially injured States"—Italy and the Netherlands—certain laws of the acting State, the United States, are also applicable. Under article 1 of the Agreement with the Netherlands, the *principles of law* are also relevant and applicable (see para. 460 above) and, under article 2, the United States will be liable for injuries caused to any person in Netherlands territorial waters under the law of a country other than the Netherlands (see para. 583 above). The legislation of a third State may thus also be applicable, although it is unclear what court would be competent to apply that legislation.

601. Under the Agreement concluded with Ireland in 1964, 396 again in connection with the N.S. Savannah, the United States submits to the jurisdiction of Irish courts for injuries its ship may cause in Irish territory or outside that territory during a voyage to or from Ireland. The Agreement provides for the applicability of the limitation on liability prescribed by United States law, of the terms of the Agreement itself, as well as of Irish domestic law.

## (c) Judicial decisions and State practice other than agreements

602. Under Article 38 of the Statute of the Permanent Court of International Justice as well as of the International Court of Justice, the function of the Court is to

<sup>392</sup> See footnote 36 above.

<sup>393</sup> See footnote 22 above.

<sup>&</sup>lt;sup>194</sup> See footnote 20 above.

<sup>395</sup> Ibid.

<sup>396</sup> See footnote 21 above.

decide such disputes as are submitted to it in accordance with international law, the sources of which are:

- (a) international conventions, whether general or particular, establishing rules expressly recognized by the contesting States;
- (b) international custom, as evidence of a general practice accepted as law:
- (c) the general principles of law recognized by civilized nations;
- (d) subject to the provisions of Article 59, judicial decisions and the teachings of the most highly qualified publicists of the various nations, as subsidiary means for the determination of rules of law.
- 603. Under this article, if the parties agree, the Court has the competence to decide their case ex aequo et bono. It is within this legal framework that international courts have adjudicated on issues of extraterritorial injuries and liability.
- 604. The decisions of arbitral tribunals have also been based on the treaty obligations of the contracting parties, on international law, and occasionally on the domestic law of States. In the *Trail Smelter* case, the tribunal examined the decisions of the United States Supreme Court as well as other sources of law and reached the conclusion that, "under the principles of international law, as well as of the *law of the United States*,\* no State has the right to use or permit the use of its territory in such a manner as to cause injury by fumes in or to the territory of another . . .". 397
- 605. In their official correspondence, States have invoked international law and the general principles of law, as well as treaty obligations. Canada's claim for damages for the crash of the Soviet satellite, Cosmos 954, was based on treaty obligations as well as the "general principles of law recognized by civilized nations" (see para, 399 above), Regional principles or standards of behaviour have also been considered relevant in relations between States. The principles accepted in Europe concerning the obligation of States whose activities may be injurious to their neighbours to negotiate with them were invoked by the Netherlands Government in 1973 when the Belgian Government announced its intention to build a refinery near its frontier with the Netherlands (see para. 112 above). Similarly, in an official letter to Mexico concerning the protective measures taken by that country to prevent flooding, the United States Government referred to the "principle of international law\* which obligates every State to respect the full sovereignty of other States" (see para. 248 above).
- 606. In their decisions, domestic courts, in addition to citing domestic law, have referred to the applicability of international law, the principles of international comity, etc. For example, the German Constitutional Court, having to render a provisional decision concerning the flow of the waters of the Danube in the *Donauversinkung* case (1927), raised the question of the bearing of acts of interference with the flow of the waters of an international watercourse on international law. It stated that "only considerable interference with the natural flow of international rivers can form the basis for claims under *international law*" (see para. 160 above). Again, in the *Roya* case (1939), the Italian Court of Cassation referred to international obligations. It stated that a

State "cannot disregard the international duty"... not to impede or to destroy... the opportunity of the other States to avail themselves of the flow of water for their own national needs" (see para. 154 above). Finally, in its judgment in the United States v. Arjona case (1887), the United States Supreme Court invoked the law of nations, which "requires every national Government to use 'due diligence' to prevent a wrong being done within its own dominion to another nation . . ." (see para. 187 above).

## C. Enforcement of judgments

607. If the rights of injured parties are to be effectively protected, it is essential that decisions and judgments awarding compensation should be enforceable. State practice has established the principle that States must not impede or claim immunity from judicial procedures dealing with disputes arising from extraterritorial injuries resulting from activities undertaken within their jurisdiction. States have thus agreed to enforce the awards rendered by the competent organs concerning disputes arising from such injuries.

## (a) Multilateral agreements

608. Multilateral agreements generally contain provisions relating to this last step in the protection of the rights of injured parties. They provide that, once a final judgment on compensation has been rendered, it shall be enforced in the territories of the contracting parties and that the parties may not invoke jurisdictional immunity. For example, the 1960 Convention on Third Party Liability in the Field of Nuclear Energy provides, in article 13, paragraphs (d) and (e), that final judgments rendered by a court competent under the Convention are enforceable in the territory of any of the contracting parties, and that, if an action for damages is brought against a contracting party, as an operator liable under the Convention, such party may not invoke jurisdictional immunity:

## Article 13

- (d) Judgments entered by the competent court under this article after trial, or by default, shall, when they have become enforceable under the law applied by that court, become enforceable in the territory of any of the other Contracting Parties as soon as the formalities required by the Contracting Party concerned have been complied with. The merits of the case shall not be the subject of further proceedings. The foregoing provisions shall not apply to interim judgments.
- (e) If an action is brought against a Contracting Party under this Convention, such Contracting Party may not, except in respect of measures of execution, invoke any jurisdictional immunities before the court competent in accordance with this article.
- 609. Similar provisions are contained in the 1952 Convention on Damage caused by Foreign Aircraft to Third Parties on the Surface, which provides that a final judgment pronounced by a competent court shall be enforceable in the territory of any contracting State once the formalities prescribed by the laws of that State have been complied with. Article 20 of the Convention reads in part:

<sup>197</sup> See footnote 23 above.

## Article 20

- 4. Where any final judgment,\* including a judgment by default, is pronounced by a court competent in conformity with this Convention, on which execution can be issued according to the procedural law of that court, the judgment shall be enforceable upon compliance with the formalities prescribed by the laws of the Contracting State,\* or of any territory, State or province thereof, where execution is applied for:
- (a) in the Contracting State where the judgment debtor has his residence or principal place of business or,
- (b) if the assets available in that State and in the State where the judgment was pronounced are insufficient to satisfy the judgment, in any other Contracting State where the judgment debtor has assets.
- 5. Notwithstanding the provisions of paragraph 4 of this article, the court to which application is made for execution may refuse to issue execution if it is proved that any of the following circumstances exist:
- (a) the judgment was given by default and the defendant did not acquire knowledge of the proceedings in sufficient time to act upon it;
- (b) the defendant was not given a fair and adequate opportunity to defend his interests;
- (c) the judgment is in respect of a cause of action which had already, as between the same parties, formed the subject of a judgment or an arbitral award which, under the law of the State where execution is sought, is recognized as final and conclusive;
  - (d) the judgment has been obtained by fraud of any of the parties;
- (e) the right to enforce the judgment is not vested in the person by whom the application for execution is made.
- 6. The merits of the case may not be reopened in proceedings for execution under paragraph 4 of this article.
- 7. The court to which application for execution is made may also refuse to issue execution if the judgment is contrary to the public policy of the State in which execution is requested.
- 8. If, in proceedings brought according to paragraph 4 of this article, execution of any judgment is refused on any of the grounds referred to in subparagraphs (a), (b) or (d) of paragraph 5 or paragraph 7 of this article, the claimant shall be entitled to bring a new action before the courts of the State where execution has been refused. The judgment rendered in such new action may not result in the total compensation awarded exceeding the limits applicable under the provisions of this Convention. In such new action the previous judgment shall be a defence only to the extent to which it has been satisfied. The previous judgment shall cease to be enforceable as soon as the new action has been started.

The right to bring a new action under this paragraph shall, notwithstanding the provisions of article 21, be subject to a period of limitation of one year from the date on which the claimant has received notification of the refusal to execute the judgment.

9. Notwithstanding the provisions of paragraph 4 of this article, the court to which application for execution is made shall refuse execution of any judgment rendered by a court of a State other than that in which the damage occurred until all the judgments rendered in that State have been satisfied.

The court applied to shall also refuse to issue execution until final judgment has been given on all actions filed in the State where the damage occurred by those persons who have complied with the time limit referred to in article 19, if the judgment debtor proves that the total amount of compensation which might be awarded by such judgments might exceed the applicable limit of liability under the provisions of this Convention.

Similarly such court shall not grant execution when, in the case of actions brought in the State where the damage occurred by those persons who have complied with the time limit referred to in article 19, the aggregate of the judgments exceeds the applicable limit of liability, until such judgments have been reduced in accordance with article 14.

10. Where a judgment is rendered enforceable under this article, payment of costs recoverable under the judgment shall also be enforceable. Nevertheiess the court applied to for execution may, on the application of the judgment debtor, limit the amount of such costs to a sum equal to ten *per centum* of the amount for which the judgment is rendered enforceable. The limits of liability prescribed by this Convention shall be exclusive of costs.

- 11. Interest not exceeding four *per centum* per annum may be allowed on the judgment debt from the date of the judgment in respect of which execution is granted.
- 12. An application for execution of a judgment to which paragraph 4 of this article applies must be made within five years from the date when such judgment became final. It should be noted that, under paragraph 7 of this article, the execution of a judgment may be refused if that judgment is contrary to the *public policy* of the State where it is to be enforced, and, under paragraph 5, if the judgment has been obtained by fraud or was given by default, the defendant not having acquired knowledge of the proceedings in sufficient time to act.
- 610. Under the 1966 Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 26 February 1961 relating to the liability of the railway for death of and personal injury to passengers, the final judgments rendered by competent courts are enforceable in any other contracting State. Article 20 of the Convention provides:

#### Article 20. Execution of judgments. Security for costs

1. Judgments entered by the competent court\* under the provisions of this Convention after trial, or by default, shall, when they have become enforceable under the law applied by that court, become enforceable in any of the other Contracting States\* as soon as the formalities required in the State concerned have been complied with. The merits of the case shall not be the subject of further proceedings.

The foregoing provisions shall not apply to interim judgments nor to awards of damages in addition to costs, against a plaintiff who fails in his action.

Settlements concluded between the parties before the competent court with a view to putting an end to a dispute, and which have been entered on the record of that court, shall have the force of a judgment of that court.

- 2. Security for costs shall not be required in proceedings arising out of the provisions of this Convention.
- 611. Article XII of the 1963 Vienna Convention on Civil Liability for Nuclear Damage contains similar language:

## Article XII

- 1. A final judgment\* entered by a court having jurisdiction under article X1 shall be recognized within the territory of any other Contracting Party,\* except:
  - (a) where the judgment was obtained by fraud;
- (b) where the party against whom the judgment was pronounced was not given a fair opportunity to present his case; or
- (c) where the judgment is contrary to the public policy of the Contracting Party within the territory of which recognition is sought, or is not in accord with fundamental standards of justice.
- 2. A final judgment which is recognized shall, upon being presented for enforcement in accordance with the formalities required by the law of the Contracting Party where enforcement is sought, be enforceable as if it were a judgment of a court of that Contracting Party.
- 3. The merits of a claim on which the judgment has been given shall not be subject to further proceedings.
- 612. Under article 12 of the 1976 Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources, a judgment given by a competent court, which is enforceable in the State of origin where it is not subject to ordinary forms of review, shall be recognized in the territory of any other State party. If, however, the judgment is obtained by fraud, or if the defendant was not given reasonable notice and a fair opportunity to present his case, the judgment is not enforceable. The article provides further that a judgment recognized as

valid shall be enforceable in the territory of any State party once the "formalities" required by that State have been complied with, but that those formalities may neither reopen the case nor raise the question of applicable law. Article 12 reads:

#### Article 12

- 1. Any judgment given by a court with jurisdiction in accordance with article 11, which is enforceable in the State of origin where it is no longer subject to ordinary forms of review, shall be recognized in any State Party, except:
  - (a) where the judgment was obtained by fraud; or
- (b) where the defendant was not given reasonable notice and a fair opportunity to present his case.
- 2. A judgment recognized under paragraph 1 of this article shall be enforceable in each State Party as soon as the formalities required in that State have been complied with. The formalities shall not permit the merits of the case to be reopened, nor a reconsideration of the applicable law.
- 613. Article 13 of the same Convention provides that, if the operator is a State party, it will still be subject to the national court of the controlling State or the State in whose territory the damage has occurred, and must waive all defences based on its status as a sovereign State:

#### Article 13

Where a State Party is the operator, such State shall be subject to suit in the jurisdictions set forth in article 11 and shall waive all defences based on its status as a sovereign State.

614. The 1969 International Convention on Civil Liability for Oil Pollution Damage similarly provides that final judgments rendered in a contracting State are enforceable in any other contracting State. Article X of the Convention reads:

## Article X

- 1. Any judgment given by a Court with jurisdiction in accordance with article IX which is *enforceable in the State of origin\** where it is no longer subject to ordinary forms of review shall be recognized in *any Contracting State,\** except:
  - (a) where the judgment was obtained by fraud; or
- (b) where the defendant was not given reasonable notice and a fair opportunity to present his case.
- 2. A judgment recognized under paragraph 1 of this article shall be enforceable in each Contracting State as soon as the formalities required in that State have been complied with.\* The formalities shall not permit the merits of the case to be reopened.
- 615. The Convention provides further, in paragraph 2 of article XI, that States shall waive all defences based on their status as sovereign States:

## Article XI

- 2. With respect to ships owned by a Contracting State and used for commercial purposes, each State shall be subject to suit in the jurisdictions set forth in article IX and shall waive all defences based on its status as a sovereign State.\*
- 616. In the 1972 Convention on International Liability for Damage caused by Space Objects, the language on enforceability of awards is different. Under article XIX, a decision of the Claims Commission shall be final and binding if the parties have so agreed; otherwise, the Commission shall render a recommendatory award, which the parties shall consider in good faith. The enforceability of awards thus depends entirely upon the agreement of the parties. Article XIX of the Convention reads in part:

#### Article XIX

- 1. The Claims Commission shall act in accordance with the provisions of article XII.
- 2. The decision of the Commission shall be final and binding if the parties have so agreed; otherwise the Commission shall render a final and recommendatory award, which the parties shall consider in good faith. The Commission shall state the reasons for its decision or award.
- 617. Finally, the 1978 Kuwait Regional Convention for Co-operation on the Protection of the Marine Environment from Pollution authorizes States to invoke sovereign immunity only in the case of operations conducted by ships owned by a State or used only on government service. Article XIV of the Convention reads:

#### Article XIV. Sovereign immunity

Warships or other ships owned or operated by a State, and used only on government non-commercial service, shall be exempted from the application of the provisions of the present Convention. Each Contracting State shall, as far as possible, ensure that its warships or other ships owned or operated by that State, and used only on government non-commercial service, shall comply with the present Convention in the prevention of pollution to the marine environment.

## (b) Bilateral agreements

618. Explicit or implicit references to the enforcement of judgments regarding liability for extraterritorial injuries have been made in bilateral agreements. Articles 18 and 19 of the 1970 Treaty between Liberia and the Federal Republic of Germany concerning the German nuclear ship, the *Otto Huhn*, <sup>398</sup> provide that final judgments of Liberian courts regarding nuclear injuries caused by the ship shall be recognized in the Federal Republic of Germany:

## Article 18

- 1. Any definite judgment passed by Liberian courts on a nuclear incident caused by the ship shall be recognized in the Federal Republic of Germany if, under paragraph 1 of article X of the Convention, jurisdiction lies with the Liberian Courts.
  - 2. Recognition of a judgment may be refused only if
  - (a) the judgment was obtained by fraud;
- (b) a legal proceeding between the same parties and on account of the same subject matter is pending before a court in the Federal Republic of Germany and if application was first made to this court;
- (c) the judgment is contrary to a definite decision passed by a court in the Federal Republic of Germany on the subject matter between the same parties;
- (d) the operator of the ship did not enter an appearance in the proceeding and if the document instituting the proceeding was served on him not effectively according to the laws of the Republic of Liberia, or not on him personally in the Republic of Liberia or not by granting him German legal assistance or not in due time for the operator of the ship to defend himself, or if the operator can prove that he was unable to defend himself because, without any fault on his part, he did not receive the document for the institution of the legal proceeding or received it too late.
  - 3. In no event will the merits of any case be subject to review.

## Article 19

Any judgments passed by Liberian courts, which are recognized according to article 18 of this Treaty and which are enforceable under Liberian law, shall be enforceable in the Federal Republic of Germany as soon as the formalities required by the law of the Federal Republic of Germany have been complied with.

<sup>398</sup> See footnote 19 above.

619. Under article VIII of the 1964 Agreement concerning the use of Italian ports by the United States nuclear ship, the N.S. Savannah, <sup>399</sup> the United States agreed not to plead the defence of sovereign immunity and to submit to the jurisdiction of Italian courts in the event of a nuclear accident involving the Savannah (see the second paragraph of article VIII, cited in para. 458 above). That article appears to refer only to the initial jurisdiction of the Italian courts and does not constitute a waiver of immunity from the execution of judgments. However, it may be assumed that the United States agreed to give effect voluntarily to any judgment rendered against it. It may also be plausibly maintained that the language is general enough to include not only initial jurisdiction but also execution.

620. In the similar Agreement concluded in 1964 with Ireland, 400 the United States agreed not to plead sovereign immunity in any legal action or proceeding brought *in personam* against the United States in an Irish court regarding nuclear injuries involving the

Savannah. Paragraph 3 of the Agreement (note 1) provides:

- 3. Subject to the provisions of this Agreement in any legal action or proceeding brought *in personam* against the United States, in an Irish court, on account of any nuclear incident caused by the ship in Irish waters, or occurring outside Ireland during a voyage of the ship to or from Ireland and causing damage in Ireland, the United States Government
  - (a) shall not plead sovereign immunity;
- (b) shall not seek to invoke the provisions of Irish law or any other law relating to the limitation of shipowner's liability.

This paragraph too may be interpreted to imply United States consent to satisfy any judgment rendered by Irish courts.

# (c) Judicial decisions and State practice other than agreements

621. The issue of enforcement of awards and judgments by arbitral tribunals and courts has not been raised in judicial decisions. In their official correspondence, States have usually arrived at compromises and in most cases have complied with the solutions agreed upon. The content of such correspondence has been examined in the preceding chapters.

## ANNEXES

[See following pages]

<sup>399</sup> See footnote 20 above.

<sup>400</sup> See footnote 21 above.

## ANNEX I

## Multilateral treaties

|   | Source   | Paragraphs and footnotes in the study   |
|---|--|---|
| Disarmam  | ent  |   |
| Treaty Banning Nuclear Weapon Tests in the Atmosphere, in Outer Space and under Water (Moscow, 5 August 1963)   | United Nations, <i>Treaty Series</i> , vol. 480, p. 43   | 20, fn. 11, 23, 52  |
| Treaty on the Prohibition of the Emplacement of Nuclear Weapons and other Weapons of Mass Destruction on the Sea-bed and the Ocean Floor and in the Subsoil thereof (London, Moscow and Washington, 11 February 1971) | Ibid., vol. 955, p. 115  | 52, 267   |
| Convention on the Prohibition of Military or any other Hostile Use of Environmental Modification Techniques (New York, 10 December 1976)  | United Nations, Juridical Yearbook 1976 (Sales No. E.78.V.5), p. 125   | 20, fn. 12, 52, 268   |
| Outer spa   | ce   |   |
| Treaty on Principles governing the Activities of States in the Exploration and Use of Outer Space, including the Moon and other Celestial Bodies (London, Moscow and Washington, 27 January 1967)                     | United Nations, <i>Treaty Series</i> , vol. 610, p. 205  | 140   |
| Convention on International Liability for Damage caused by Space Objects (London, Moscow and Washington, 29 March 1972)   | <i>Ibid.</i> , vol. 961, p. 187  | 24, 305, 440-443, 492, 498,<br>549, 585, 590, 616   |
| Law of the  | Sea  |   |
| Convention on the Continental Shelf (Geneva, 29 April 1958)   | <i>Ibid.</i> , vol. 499, p. 311  | 229   |
| United Nations Convention on the Law of the Sea (Montego Bay, Jamaica, 10 December 1982)  | Official Records of the Third United Nations Conference on the Law of the Sea, vol. XVII (United Nations publication (Sales No. E.84.V.3), p. 151, document A/CONF.62/122) | 25, 43, 56, 57, 75, 101, 126, 130, 201-207, 244, 252, 253, 306, 349, 388, 444, 503, 580, 586, 591 |
| Navigation  |  |   |
| International Convention relating to the Limitation of the Liability of Owners of Seagoing Ships (Brussels, 10 October 1957)  | N. Singh, ed., International Maritime<br>Law Conventions, vol. 4, Maritime<br>Law (London, Stevens, 1983), p. 2967   | 561, 562, 594   |

| International Convention for the Safety of Life at Sea (London, 17 June 1960)   | United Nations, <i>Treaty Series</i> , vol. 536, p. 27  | 83   |  |
|---|---|--|--|
| Convention on Limitation of Liability for Maritime Claims (London, 19 November 1976)  | IMCO publication, Sales No. 77.04.E   | 409  |  |
| Transpor  | rt  |  |  |
| Convention for the Unification of certain Rules relating to Damage caused by Aircraft to Third Parties on the Surface (Rome, 29 May 1933) [did not come into force]   | M. O. Hudson, <i>International Legislation</i> (Washington, D.C.), vol. VI (1932-1934), p. 334, No. 329                                     | 374  |  |
| Convention on Damage caused by Foreign Aircraft to Third Parties on the Surface (Rome, 7 October 1952)  | United Nations, Treaty Series, vol. 310, p. 181   | 24, 345, 374, 408, 416-420,<br>485, 500, 558, 609                      |  |
| Additional Convention to the International Convention concerning the Carriage of Passengers and Luggage by Rail (CIV) of 25 February 1961, relating to the Liability of the Railway for Death of and Personal Injury to Passengers, and Protocol B (Bern, 26 February 1966), and Protocol I (Bern, 22 October 1971) | United Kingdom, Treaty Series No. 20 (1973), Cmnd. 5249   | 415, 486, 487, 499, 511, 515, 547, 559, 560, 577, 592, 610             |  |
| Telecommunic  | cations   |  |  |
| International Radiotelegraph Convention (Washington, 25 November 1927)  | League of Nations, Treaty Series, vol. LXXXIV, p. 97  | 27, 58   |  |
| International Telecommunication Convention (Madrid, 9 December 1932)  | Ibid., vol. CLI, p. 4   | 27, 59   |  |
| International Convention concerning the Use of Broadcasting in the Cause of Peace (Geneva, 23 September 1936)   | Ibid., vol. CLXXXVI, p. 301   | 27, 59   |  |
| Uses of nuclear energy  |   |  |  |
| Convention on Third Party Liability in the Field of Nuclear Energy (Paris, 29 July 1960) and Additional Protocol (Paris, 28 January 1964)   | United Nations, <i>Treaty Series</i> , vol. 956, pp. 251 and 335  | 23, 42, 344, 407, 428, 491, 546, 557, 570, 608                         |  |
| Convention on the Liability of Operators of Nuclear Ships (Brussels, 25 May 1962)   | IAEA, International Conventions relating<br>to Civil Liability for Nuclear Damage,<br>Legal Series No. 4, rev. ed. (Vienna,<br>1976), p. 34 | 23, 50, 342, 421, 439, 488,<br>489, 506, 514, 548, 573,<br>593         |  |
| Vienna Convention on Civil Liability for Nuclear Damage (Vienna, 21 May 1963)   | United Nations, <i>Treaty Series</i> , vol. 1063, p. 265  | 23, 42, 50, 198, 343, 425, 426, 490, 496, 510, 513, 516, 571, 572, 611 |  |
| Convention relating to Civil Liability in the Field of Maritime Carriage of Nuclear Material (Brussels, 17 December 1971)   | <i>Ibid.</i> , vol. 974, p. 255   | 424  |  |

| Environme   | Source<br>ent   | Paragraphs and footnotes in the study                   |  |  |
|---|---|---|--|--|
|   | Atmosphere  |   |  |  |
| Convention on Long-range Transboundary Air Pollution (Geneva, 13 November 1979)   | E/ECE/1010  | 71, 99, 129   |  |  |
| Vienna Convention for the Protection of the Ozone Layer (Vienna, 22 March 1985)   | UNEP, Nairobi, 1985   | 266   |  |  |
| Marine enviro   | ONMENT  |   |  |  |
| International Convention for the Northwest Atlantic Fisheries (Washington, 8 February 1949)   | United Nations, <i>Treaty Series</i> , vol. 157, p. 157   | 26, 51, 303, 304  |  |  |
| North-East Atlantic Fisheries Convention (London, 24 January 1959)  | Ibid., vol. 486, p. 157   | 231   |  |  |
| International Convention for the Conservation of Atlantic Tunas (Rio de Janeiro, 14 May 1966)   | <i>Ibid.</i> , vol. 673, p. 63  | 26, 51, 72, 265, 308                                    |  |  |
| Agreement for Co-operation in dealing with Pollution of the North Sea by Oil (Bonn, 9 June 1969)  | <i>Ibid.</i> , vol. 704, p. 3   | 264   |  |  |
| International Convention relating to Intervention on the High Seas in Cases of Oil Pollution Casualties (Brussels, 29 November 1969)                          | Ibid., vol. 970, p. 211   | 52, 102, 132, 175, 243, 309,<br>410                     |  |  |
| International Convention on Civil Liability for Oil Pollution Damage (Brussels, 29 November 1969)   | <i>Ibid.</i> , vol. 973, p. 3   | 25, 129, 347, 422, 423, 493,<br>494, 497, 574, 614, 615 |  |  |
| International Convention on the Establishment of an International Fund for Compensation for Oil Pollution Damage (Brussels, 18 December 1971)                 | Ibid., vol. 1110, p. 57   | 495, 576  |  |  |
| Convention on the Prevention of Marine Pollution by Dumping of Wastes and Other Matter (London, Mexico, Moscow and Washington, 29 December 1972)              | <i>Ibid.</i> , vol. 1046, p. 120  | 25, 51, 257, 387  |  |  |
| International Convention for the Prevention of Pollution from Ships (London, 2 November 1973)   | United Nations, Juridical Yearbook 1973 (Sales No. E.75.V.1), p. 81   | 25, 128   |  |  |
| 1973 Protocol relating to Intervention on the High Seas in case of Marine Pollution by Substances other than Oil (London, 2 November 1973)                    | <i>Ibid.</i> , p. 91  | 242   |  |  |
| Denmark, Finland, Norway and Sweden: Convention on the Protection of the Environment (Stockhom, 19 February 1974)   | United Nations, <i>Treaty Series</i> , vol. 1092, p. 279  | 41, 134, 135, 170-172, 199,<br>429, 578                 |  |  |
| Convention on the Protection of the Marine Environment of the Baltic Sea Area (Helsinki, 22 March 1974)   | UNEP, Selected Multilateral Treaties in<br>the Field of the Environment,<br>Reference Series 3 (Nairobi, 1983),<br>p. 405 | 122, 123, 126, 198, 386                                 |  |  |
| Convention on the Prevention of Marine Pollution from Land-based Sources (Paris, 4 June 1974)   | <i>Ibid.</i> , p. 430   | 100, 116-121, 174, 254-256                              |  |  |
| Convention for the Protection of the Mediterranean Sea against Pollution (Barcelona, 16 February 1976)  | <i>Ibid.</i> , p. 448   | 124-126, 198, 386                                       |  |  |
| Convention on Civil Liability for Oil Pollution Damage resulting from Exploration for and Exploitation of Seabed Mineral Resources (London, 17 December 1976) | Ibid., p. 474   | 348, 384, 504, 512, 563, 579, 612, 613                  |  |  |

| Kuwait Regional Convention for Co-operation on the Protection of the Marine Environment from Pollution (Kuwait, 24 April 1978)   | United Nations, <i>Treaty Series</i> , vol. 1140, p. 133   | 70, 302, 385, 617          |
|--|--|----------------------------|
| Convention for the Protection and Development of the Marine Environment of the Wider Caribbean Region (Cartagena de Indias, Colombia, 24 March 1983)   | UNEP, Nairobi, 1983; International Legal Materials (Washington, D.C.), vol. 22, 1983, p. 221   | 60, 103, 270, 389          |
| IMO, Draft convention on liability and compensation in connection with the carriage of noxious and hazardous substances by sea (13 January 1984)   | LEG/CONF.6/3   | 346, 430, 501, 502         |
| RIVERS AND L   | AKES   |                            |
| Belgium, France and Luxembourg: Protocol to Establish a Tripartite Standing Committee on Polluted Waters (Brussels, 8 April 1950)  | United Nations, <i>Treaty Series</i> , vol. 66, p. 285   | 74, 173, 200, 259          |
| Convention on the Protection of Lake Constance against Pollution (Steckborn, Switzerland, 27 October 1960)   | Switzerland, Recueil officiel des lois et des ordonnances, 1961, vol. 2, p. 923, No. 43 [for English text, see AEU/TFP/ENV/75.11, annex 1] | 25, 74, 133, 258, 301, 307 |
| France, Federal Republic of Germany and Luxembourg: Protocol concerning the Establishment of an International Commission to Protect the Moselle against Pollution (Paris, 20 December 1961)  | United Nations, <i>Treaty Series</i> , vol. 940, p. 211  | 261                        |
| Agreement on the International Commission for the Protection of the Rhine against Pollution (Bern, 29 April 1963)  | <i>Ibid.</i> , vol. 994, p. 3  | 262                        |
| Additional Agreement to the Agreement signed at Bern on 29 April 1963 (Bonn, 3 December 1976)  | Official Journal of the European Com-<br>munities, vol. 20, No. L 240,<br>19 September 1977, p. 35   | fn. 201                    |
| European Agreement on the Restriction of the Use of Certain Detergents in Washing and Cleaning Products (Strasbourg, 16 September 1968)  | European Treaty Series, No. 64,<br>Strasbourg, 1971  | 131                        |
| Convention for the Protection of the Rhine against Chemical Pollution (Bonn, 3 December 1976)  | United Nations, <i>Treaty Series</i> , vol. 1124, p. 375   | 127, 263                   |
| Frontier ques  | stions   |                            |
| France, Federal Republic of Germany and Switzerland: Exchange of letters constituting an agreement concerning the establishment of an Intergovernmental Commission on Contiguity problems in frontier regions (Paris, 22 October 1975) | United Nations, <i>Treaty Series</i> , vol. 1036, p. 367   | 260                        |
|  |  |                            |

## ANNEX II

## Bilateral treaties

## **ABBREVIATIONS**

| BFSP              | British and Foreign State Papers   |
|-------------------|--|
| ILM               | International Legal Materials  |
| Ríos y Lagos      | OAS, Ríos y Lagos Internacionales (Utilización para fines agrícolas e industriales), 4th ed., rev. (OAS/SER.1/VI, CIJ-75 Rev.2)                                  |
| Rüster and Simma  | B. Rüster and B. Simma, eds., International Protection of the Environment. Treaties and Related Documents (Dobbs Ferry, N.Y., Oceana Publications)               |
| Legislative Texts | United Nations, Legislative Texts and Treaty Provisions concerning the Utilization of International Rivers for Other Purposes than Navigation (Sales No. 63.V.4) |

Paragraphs and footnotes
Source in the study

## Nuclear and space activities

## Belgium and France

| Convention on radiological protection with regard to the installations of the Ardennes nuclear power station (Paris, 23 September 1966)               | United Nations, <i>Treaty Series</i> , vol. 588, p. 227  | 28, 82, 313   |
|---|--|---------------|
| France and USSR   |  |               |
| Exchange of notes constituting an agreement concerning the prevention of the accidental or unauthorized use of nuclear weapons (Moscow, 16 July 1976) | United Nations, <i>Treaty Series</i> , vol. 1036, p. 299 | 28, 283       |
| United States of America and Canada   |  |               |
| Exchange of notes constituting an agreement concerning liability for loss or damage from certain rocket launches (Ottawa, 31 December 1974)           | <i>Ibid.</i> , vol. 992, p. 97                           | 140, 392, 548 |

## Frontier waters

## Argentina and Chile

Act of Santiago concerning hydrologic basins (26 June 1971)

Ríos y Lagos, pp. 495-496; English text in Yearbook . . . 1974, vol. II (Part Two), p. 324, document A/CN.4/274, para. 327

| Argentina and Uruguay   |  |                                      |
|---|--|--------------------------------------|
| Treaty concerning the La Plata River and its maritime limits (Montevideo, 19 November 1973)   | INTAL, Derecho de la Integración, (Buenos Aires), vol. VII, No. 15, March 1974, p. 225; ILM, vol. 13, 1974, p. 251 | 451                                  |
| Belgium and Netherlands   |  |                                      |
| Treaty concerning the improvement of the Terneuzen and Ghent Canal and the set-<br>tlement of various related matters (Brussels, 20 June 1960)  | United Nations, <i>Treaty Series</i> , vol. 423, p. 19   | 177                                  |
| Belgium and United Kingdom  |  |                                      |
| Arrangement regarding water rights on the boundary between Tanganyika and Ruanda-Urundi (London, 22 November 1934)  | League of Nations, <i>Treaty Series</i> , vol. CXC, p. 103   | 143                                  |
| Bulgaria and Turkey   |  |                                      |
| Agreement concerning co-operation in the use of the waters of rivers flowing through the territory of both countries (Istanbul, 23 October 1968)  | United Nations, <i>Treaty Series</i> , vol. 807, p. 117  | 81                                   |
| Canada and United States of America   |  |                                      |
| Treaty relating to co-operative development of the water resources of the Columbia River Basin (Washington, 17 January 1961)  | Ibid., vol. 542, p. 245  | 285                                  |
| Czechoslovakia and Poland   |  |                                      |
| Agreement concerning the use of water resources in frontier waters (Prague, 21 March 1958)  | <i>Ibid.</i> , vol. 538, p. 89   | 141, 179                             |
| Finland and Norway  |  |                                      |
| Agreement between the Governments of Finland and Norway on the transfer from the course of the Näätämo (Neiden) river to the course of the Gandvik river of water from the Garsjöen, Kjerringvatn and Förstevannene lakes (Oslo, 25 April 1951) | Legislative Texts, p. 609, No. 168; Rüster and Simma, vol. X, p. 5011  | 314, 452, 517                        |
| Finland and Sweden  |  |                                      |
| Agreement concerning frontier rivers (Stockholm, 15 December 1971)  | United Nations, <i>Treaty Series</i> , vol. 825, p. 191  | 208-210, 433, 520, 552, 587          |
| Finland and Russian Socialist Federal Soviet Republic   |  |                                      |
| Convention concerning the maintenance of river channels and the regulation of fishing on watercourses forming part of the frontier between Finland and Russia (Helsingfors, 28 October 1922)  | League of Nations, <i>Treaty Series</i> , vol. XIX, p. 183   | 61, 141                              |
| Finland and USSR  |  |                                      |
| Agreement concerning frontier watercourses (Helsinki, 24 April 1964)  | United Nations, <i>Treaty Series</i> , vol. 537, p. 231  | 178, 450                             |
| Germany and Denmark   | -  |                                      |
| Agreement for the settlement of questions relating to watercourses and dikes on the German-Danish frontier (Copenhagen, 10 April 1922)  | League of Nations, <i>Treaty Series</i> , vol. X, p. 201   | 105, 183, 214, 246, 394,<br>412, 434 |

| , 212, |  |
|--------|--|
| 7      | Documents of the thirty-seventh session-Addendum |
|        |  |

134

Paragraphs and footnotes

|  | Source  | in the study                               |
|--|---|--|
|  |   |  |
| Hungary and Austria  |   |  |
| Treaty concerning the regulation of water economy questions in the frontier region (Vienna, 9 April 1956)  | United Nations, <i>Treaties Series</i> , vol. 438, p. 123               | 148, 282                                   |
| Norway and Sweden  |   |  |
| Convention on certain questions relating to the law on watercourses (Stockholm, 11 May 1929)   | League of Nations, Treaty Series, vol. CXX, p. 263                      | 77, 107, 141, 181, 182, 212, 521, 581, 595 |
| Poland and German Democratic Republic  |   |  |
| Agreement concerning navigation in frontier waters and the use and maintenance of frontier waters (Berlin, 6 February 1952)  | United Nations, <i>Treaty Series</i> , vol. 304, p. 131                 | 284, 315                                   |
| Romania and Yugoslavia   |   |  |
| General Convention concerning the hydraulic system (Belgrade, 14 December 1931)  | League of Nations, Treaty Series, vol. CXXXV, p. 31                     | 106, 138, 176, 272, 317                    |
| USSR and Finland   |   | •  |
| Additional Protocol concerning compensation for loss and damage and for the work to be carried out by Finland in connection with the implementation of the Agreement of 29 April 1959 between the Government of the Union of Soviet Socialist Republics, the Government of Finland and the Government of Norway concerning the regulation of Lake Inari by means of the Kaitakoski hydroelectric power station and dam (Moscow, 29 April 1959) | United Nations, <i>Treaty Series</i> , vol. 346, p. 209                 | 453  |
| United Kingdom and United States of America  |   |  |
| Treaty relating to boundary waters and questions arising along the boundary between Canada and the United States (Washington, 11 January 1909)   | BFSP, 1908-1909, vol. 102, p. 137;<br>Legislative Texts, p. 260, No. 79 | 76, 139, 213, 393, 505                     |
| Yugoslavia and Austria   |   |  |
| Convention concerning water economy questions relating to the Drava (Geneva, 25 May 1954)  | United Nations, <i>Treaty Series</i> , vol. 227, p. 111                 | 149  |
| Agreement concerning water economy questions in respect of the frontier sector of the Mura and of its frontier waters (Mura Agreement) (Vienna, 16 December 1954)  | <i>Ibid.</i> , vol. 396, p. 75  | 274  |
| Yugoslavia and Greece  |   |  |
| Agreement concerning hydro-economic questions (Athens, 18 June 1959)   | Ibid., vol. 363, p. 133   | 79   |

| Yugoslavia and Hungary   |  |  |
|--|--|--|
| Agreement concerning fishing in frontier waters (Belgrade, 25 May 1957)  | Legislative Texts, p. 836, No. 229; Rüster and Simma, vol. IX, p. 4572 | 447  |
| Yugoslavia and Romania   |  |  |
| Agreement concerning questions of water control on water control systems and watercourses on or intersected by the State frontier, together with the Statute of the Yugoslav-Romanian Water Control Commission (Bucharest, 7 April 1955) | Legislative Texts, p. 928, No. 253; Rüster and Simma, vol. IX, p. 4531 | 141, 180                                   |
| Environme  | ent  |  |
| Canada and United States of America  |  |  |
| Exchange of notes constituting an agreement relating to joint pollution contingency plans for spills of oil and other noxious substances in waters of mutual interest (Ottawa, 19 June 1974)   | United Nations, <i>Treaty Series</i> , vol. 951, p. 287                | 316  |
| Agreement relating to the exchange of information on weather modification activities (Washington, 26 March 1975)   | Ibid., vol. 977, p. 385  | 29, 104, 245, 391                          |
| Agreement to track air pollution across Eastern North America (Acid Rain Research) (Ottawa, 23 August 1983)  | ILM, vol. 22, 1983, p. 1017  | 286  |
| United States of America and Mexico  |  |  |
| Agreement on co-operation for the protection and improvement of the environment in the border area (La Paz, Baja California, Mexico, 14 August 1983)   | Ibid., p. 1025   | 30, 63, 78, 136, 176, 211, 273, 390        |
| Nuclear sh   | nips   |  |
| Liberia and Federal Republic of Germany  |  |  |
| Treaty on the use of Liberian waters and ports by the N.S. Otto Hahn (Bonn, 27 May 1970)   | Rüster and Simma, vol. I, p. 482                                       | 46, fn. 25, 146, 462, 506, 507, 566, 618   |
| Netherlands and United States of America   |  |  |
| Agreement on public liability for damage caused by the N.S. Savannah (The Hague, 6 February 1963)  | United Nations, Treaty Series, vol. 487, p. 113                        | 46, 147, 215, 460, 564, 583, 596, 599, 600 |
| Operational agreement on arrangements for a visit of the N.S. Savannah to the Netherlands (The Hague, 20 May 1963)   | <i>Ibid.</i> , p. 123  | fn. 22, 83, 184, 215, 461                  |
| United States of America and Ireland   |  |  |
| Exchange of notes constituting an agreement relating to public liability for damage caused by the N.S. Savannah (Dublin, 18 June 1964)   | <i>Ibid.</i> , vol. 530, p. 217  | 46, 457, 52, 565, 601                      |

136

|   | Source   | Paragraphs and footnotes in the study |  |  |
|---|--|---------------------------------------|--|--|
| United States of America and Italy  |  |                                       |  |  |
| Agreement on the use of Italian ports by the N.S. Savannah (Rome, 23 November 1964)   | <i>Ibid.</i> , vol. 532, p. 133  | 46, 83, 184, 215, 458, 565, 597, 619  |  |  |
| Exchange of notes constituting an agreement concerning liability during private operation of the N.S. Savannah (Rome, 16 December 1965)   | Ibid., vol. 574, p. 139  | fn. 20, 459, 565, 582, 598, 599       |  |  |
| Pipelines   | S .  |                                       |  |  |
| Federal Republic of Germany and Norway  |  |                                       |  |  |
| Agreement relating to the transmission of petroleum by pipeline from the Ekofisk field and neighbouring areas to the Federal Republic of Germany (Bonn, 16 January 1974)  | Ibid., vol. 1016, p. 91  | 29, 275-278, 351, 431                 |  |  |
| Norway and United Kingdom   |  |                                       |  |  |
| Agreement relating to the transmission of petroleum by pipeline from the Ekofisk field and neighbouring areas to the United Kingdom (Oslo, 22 May 1973)   | <i>Ibid.</i> , vol. 885, p. 5  | 29, 350, 431, 432                     |  |  |
| Frontier reg  | Frontier regime  |                                       |  |  |
| Czechoslovakia and Hungary  |  |                                       |  |  |
| Treaty concerning the regime of State frontiers (Prague, 13 October 1956)   | <i>Ibid.</i> , vol. 300, p. 125  | 311                                   |  |  |
| Hungary and Czechoslovakia  |  |                                       |  |  |
| Convention relating to the settlement of questions arising out of the delimitation of the frontier between the Kingdom of Hungary and the Czechoslovak Republic (Frontier Statute) (Prague, 14 November 1928)   | League of Nations, Treaty Series, vol. CX, p. 425                      | 150                                   |  |  |
| Hungary and Romania   |  |                                       |  |  |
| Treaty concerning the regime of the Hungarian-Romanian State frontier and co-<br>operation in frontier matters (Budapest, 13 June 1963)   | United Nations, <i>Treaty Series</i> , vol. 576, p. 275                | 62                                    |  |  |
| Hungary and USSR  |  |                                       |  |  |
| Treaty concerning the regime of the Soviet-Hungarian State frontier (Moscow, 24 February 1950)  | Legislative Texts, p. 823, No. 226; Rüster and Simma, vol. IX, p. 4493 | 81, 448                               |  |  |
| Netherlands and Federal Republic of Germany   |  |                                       |  |  |
| Treaty concerning the course of the common frontier, the boundary waters, real property situated near the frontier, traffic crossing the frontier on land and via inland waters, and other frontier questions (Frontier Treaty) (The Hague, 8 April 1960) | United Nations, <i>Treaty Series</i> , vol. 508, p. 149                | 150, 449                              |  |  |

| Norway and USSR  |   |                        |
|--|---|------------------------|
| Agreement concerning the regime of the Norwegian-Soviet frontier and procedure for the settlement of frontier disputes and incidents (Oslo, 29 December 1949)                      | <i>Ibid.</i> , vol. 83, p. 291                        | 44, 64, 145, 280       |
| Poland and USSR  |   |                        |
| Convention concerning juridical relations on the State frontier (Moscow, 10 April 1932)  | League of Nations, Treaty Series, vol. CXLI, p. 349   | 137, 144               |
| Agreement concerning the regime of the Polish-Soviet State frontier (Moscow, 8 July 1948)  | United Nations, <i>Treaty Series</i> , vol. 37, p. 25 | 80, 312, 448, 518      |
| USSR and Finland   |   |                        |
| Agreement concerning the regime of the Soviet-Finnish frontier (Moscow, 9 December 1948)   | <i>Ibid.</i> , vol. 217, p. 135                       | 281, 318               |
| USSR and Poland  |   |                        |
| Treaty concerning the regime of the Soviet-Polish State frontier and co-operation and mutual assistance in frontier matters (Moscow, 15 February 1961)                             | <i>Ibid.</i> , vol. 420, p. 161                       | 141                    |
| Contiguity re  | ations  |                        |
| France and Spain   |   |                        |
| Convention on mutual assistance between French and Spanish fire and emergency services (Madrid, 14 July 1959) and Additional Agreement to the Convention (Madrid, 8 February 1973) | <i>Ibid.</i> , vol. 951, p. 135                       | 454, 505               |
| Federal Republic of Germany and Austria  |   |                        |
| Agreement concerning the effects on the territory of the Federal Republic of Germany of construction and operation of the Salzburg airport (Vienna, 19 December 1967)              | <i>Ibid.</i> , vol. 945, p. 87                        | 45, 143, 455, 456, 519 |
| Agreement concerning co-operation with respect to land use (Vienna, 11 December 1973)  | Ibid., vol. 966, p. 301                               | 31, 62                 |

## ANNEX III

## Judicial decisions and State practice other than agreements

|   | Paragraphs and footnotes in the study          |
|---|--|
| Permanent Court of International Justice  |  |
| Lotus, judgment No. 9 of 7 September 1927, P.C.I.J., Series A, No. 10   | fn. 10   |
| Territorial jurisdiction of the International Commission of the River Oder, judgment No. 16 of 10 September 1929, P.C.I.J., Series A, No. 23  | 227  |
| Railway traffic between Lithuania and Poland, advisory opinion of 15 October 1931, P.C.I.J., Series A/B, No. 42, p. 166   | 108  |
| International Court of Justice  |  |
| Corfu Channel, merits, judgment of 9 April 1949, I.C.J. Reports 1949, p. 4  | 22, 47, 66-68, 152, 185, 337,<br>401, 402, 465 |
| Asylum case, judgment of 20 November 1950, I.C.J. Reports 1950, p. 286  | fn. 10   |
| Fisheries (United Kingdom v. Norway), judgment of 18 December 1951, I.C.J. Reports 1951, p. 116   | 35, 193, 228, 237                              |
| Nottebohm (Liechtenstein v. Guatemala), second phase, judgment of 6 April 1955, I.C.J. Reports 1955, p. 4   | fn. 10   |
| North Sea Continental Shelf (Federal Republic of Germany/Denmark; Federal Republic of Germany/Netherlands), judgment of 20 February 1969, I.C.J. Reports 1969, p. 3   | 10, 35, 108, 165, 193, 229, 237, 338           |
| Legal consequences for States of the continued presence of South Africa in Namibia (South West Africa) notwithstanding Security Council resolution 276 (1970), order No. 2 of 26 January 1971 (request for advisory opinion), I.C.J. Reports 1971, p. 6 | 476  |
| Nuclear Tests (Australia v. France; New Zealand v. France), Interim measures of protection, orders of 22 June 1973, I.C.J. Reports 1973, pp. 99 and 135   | 33, 166, 322, 529                              |
| Nuclear Tests (Australia v. France; New Zealand v. France), judgments of 20 December 1974, I.C.J. Reports 1974, pp. 253 and 457   | 6, fn. 8                                       |
| Fisheries Jurisdiction (United Kingdom v. Iceland; Federal Republic of Germany v. Iceland), judgments of 25 July 1974, I.C.J. Reports 1974, pp. 3 and 175   | 35, 54, 111, 155, 193, 230, 237, 321, 327, 336 |
| Continental Shelf (Tunisia/Libyan Arab Jamahiriya), judgment of 24 February 1982, I.C.J. Reports 1982, p. 18  | 35, 232-234                                    |
| International arbitration  Source   |  |
| Alabama claims (United States of America/United Kingdom), award of 14 September 1872  J. B. Moore, History and Digest of the International Arbitrations to which the United States has been a Party (Washington, D.C., 1898), vol. I, p. 653            | 49, 396, 397, 541, 553                         |

139

| Island of Palmas case (Netherlands/United States of America), award of 4 April 1928  | United Nations, Reports of International Arbitral Awards, vol. II (Sales No. 1949.V.1), p. 829  | 476   |
|--|---|---|
| Trail Smelter case (United States of America/Canada), awards of 16 April 1938 and 11 March 1941  | Ibid., vol. III (Sales No. 1949.V.2),<br>p. 1905  | 20, 32, 47, 65, 96, 153, 194,<br>195, 294-296, 325, 326,<br>330, 335, 340, 400, 437,<br>481, 526, 528, 532-540,<br>589, 604 |
| Arbitration between Petroleum Development (Trucial Coast) Ltd. and the Sheikh of Abu Dhabi, award of 28 August 1951  | The International and Comparative Law Quarterly, London, vol. 1, 1952, p. 247   | 35  |
| Lake Lanoux case (Spain/France), award of 16 November 1957   | United Nations, Reports of International Arbitral Awards, vol. XII (Sales No. 63.V.3), p. 281. For summary in English, see International Law Reports 1957 (London, 1961), p. 101; see also Yearbook 1974, vol. II (Part Two), pp. 194-199, document A/CN.4/5409, paras. 1055-1068 | 34, 47, 109, 110, 152,<br>156-158, 162, 192, 219,<br>237, 334, 403, 404   |
| Gut Dam Claims (United States of America/Canada), award of 27 September 1968   | International Legal Materials (Washington, D.C.), vol. 8, 1969, p. 118  | 471, 472  |
| Delimitation of the continental shelf between the United Kingdom of Great Britain and Northern Ireland and the French Republic, award of 30 June 1977  | United Nations, Reports of International Arbitral Awards, vol. XVIII (Sales No. E/F 80.V.7), p. 3   | 235   |
| National co  | urts  |   |
| France   |   |   |
| Jeand'heur v. Galeries belfortaises (1927), Dalloz, Recueil périodique et critique, 1930   | 9 (Paris), part 1, p. 57  | fn. 275, fn. 300  |
| Germany  | (   |   |
| Donauversinkung (Württemberg and Prussia v. Baden) (1927), Entscheidungen des Reappendix 2, p. 18; Annual Digest of Public International Law Cases, 1927-1928, p                                     |   | 22, fn. 13, 160, 606  |
| Italy  |   |   |
| Société énergie électrique du littoral méditerranéen v. Compania imprese elettriche lig<br>(Rome), vol. 64, part 1, col. 1036; Annual Digest and Reports of Public Internatio<br>p. 120, case No. 47 | guri (the Roya case) (1939), Il Foro Italiano nal Law Cases, 1938-1940 (London), vol. 9,  | 34, 36, 154, 606  |
| Switzerlan   | ND  |   |
| Solothurn v. Aargau (1900), Recueil officiel des arrêts du Tribunal fédéral suisse, 190  | 00, vol. 26, part 1, p. 444   | 221   |
| Aargau v. Solothurn (1915), Recueil, 1915, vol. 41, part 1, p. 126   |   | 221   |

#### UNITED KINGDOM

| Rylands v. Fletcher (1866), | The Law Reports, | Court of Exchequer, | vol. I, p. 26 | 5; (1868), The Law | Reports, House of Lords, | 363, fn. 269, 364, 377 |
|-----------------------------|------------------|---------------------|---------------|--------------------|--------------------------|------------------------|
| vol. III, p. 330            | •                | •                   | •             |                    | •                        |                        |

#### United States of America

| Connecticut v. Massachusetts (1931), United States Reports, vol. 282, p. 660                      | 225, 226         |
|---|------------------|
| Escola v. Coca Cola Bottling Co. (1944), California Reports, 2d Series, vol. 24, p. 453           | 370              |
| Georgia v. Tennessee Copper Company (1907), United States Reports, vol. 206, p. 230               | 161              |
| Goldberg v. Kollsman Instrument Corp. (1963), New York Supplement, 2d Series, vol. 240, p. 592    | fn. 281          |
| Illinois v. Milwaukee (1972), United States Reports, vol. 406, p. 91                              | 191              |
| Kansas v. Colorado (1902), ibid., vol. 185, p. 125  | 159              |
| Kansas v. Colorado (1907), ibid., vol. 206, p. 46   | 34, 226          |
| Missouri v. Illinois (1906), ibid., vol. 200, p. 496  | 34, 162          |
| Natural Resources Defense Council v. Morton (1972), Federal Reporter, 2d Series, vol. 458, p. 827 | fn. 234          |
| Nebraska v. Wyoming (1945), United States Reports, vol. 325, p. 589                               | 223              |
| New Jersey v. New York (1931), ibid., vol. 283, p. 336  | 224              |
| Prentiss v. National Airlines (1953), Federal Supplement, vol. 112, p. 306                        | fn. 287          |
| United States v. Arjona (1887), United States Reports, vol. 120, p. 479                           | 36, 186-188, 606 |
| Washington v. Oregon (1936), ibid., vol. 297, p. 517  | 222              |
| Wilderness Society v. Hickel (1970), Federal Supplement, vol. 325, p. 422                         | 320, fn. 234     |
| Wilderness Society v. Morton (1973), Federal Reporter, 2d Series, vol. 479, p. 842                | 320, fn. 234     |

### State practice\*

#### HARMFUL OR DANGEROUS ACTIVITIES IN FRONTIER AREAS

Source

#### Czechoslovakia-Austria

Construction of a nuclear reactor at Dukovany, Czechoslovakia, near the Austrian border (1975)

G. Handl, "An international legal perspective of abnormally dangerous activities in frontier areas: The case of nuclear power plant siting", Ecology Law Quarterly (Berkeley, Cal.), vol. 7, 1978, p. 28

112

<sup>\*</sup> The source State is cited first.

14

| France-Switzerland  |   |                                    |
|---|---|------------------------------------|
| French military target practice near the Swiss border (1892)  | P. Guggenheim, "La pratique suisse (1956)", Annuaire suisse de droit international, 1957 (Zürich), vol. 14, p. 168  | 238, 332                           |
| Hungary-Austria   |   |                                    |
| Mine-laying by the Hungarian authorities near the Austrian border (1949)  | Handl, loc. cit., pp. 23-24   | 474                                |
| Italy-Switzerland   |   |                                    |
| Explosion of a munitions plant at Arcisate, Italy, near the Swiss border (1948)   | Guggenheim, loc. cit., p. 169   | 477                                |
| Mexico-United States of America   |   |                                    |
| Correspondence concerning the construction of a highway, in Mexico, across the Smugglers and Goat Canyons, and consequent danger of flooding in the United States (1957-1959) | M. M. Whiteman, ed., Digest of International Law (Washington, D.C.), vol. 6, pp. 260-262  | 36, 93, 163, 220, 239, 292,<br>466 |
| Switzerland-Austria   |   |                                    |
| Planned construction of a nuclear reactor at Rüthi, Switzerland, near the Austrian border (1972-1976)   | Handl, loc. cit., pp. 28-30   | 112                                |
| Switzerland-Liechtenstein   |   |                                    |
| Accidental shelling of Liechtenstein territory by Swiss artillery (1968)  | Annuaire suisse de droit international,   | 475                                |
| United States of America-Mexico   | 1969-1970 (Zürich), vol. 26, p. 158   |                                    |
| Correspondence concerning the construction of the Rose Street Canal at Douglas, Arizona, and consequent danger of flooding at Agua Prieta, in Mexico (1951-1955)              | Whiteman, op. cit., vol. 6, pp. 262-265   | 94, 95, 248, 299, 467, 605         |
| POLLUTION AND ENVIRONME   | ENTAL PROTECTION  |                                    |
| Austria-Yugoslavia  Pollusian of the Mana Binar has Acataian hadron leasting last (1956, 1959)  | TT. 11 4460 . P. 1 197 . C  | 450 500 555                        |
| Pollution of the Mura River by Austrian hydroelectric plants (1956-1959)  | Handl, "State liability for accidental transnational environmental damage by private persons", American Journal of International Law (Washington, D.C.), vol. 74, 1980, pp. 545-546 | 478, 530, 555                      |
| Belgium-Netherlands   |   |                                    |
| Establishment of a refinery at Lanaye, Belgium, near the Netherlands frontier (1972-1973)   | Belgian Parliament, Questions et<br>Réponses, 19 July 1973  | 113, 605                           |
| Canada-United States of America   |   |                                    |
| Oil prospection in the Beaufort Sea (1976-1977)   | International Canada (Toronto), vol. 7, 1976, pp. 84-85   | 357, 437, 483                      |

| Documents of                            |
|---|
| of the                                  |
| cuments of the thirty-seventh session—. |
| session-Addendun                        |
| 3                                       |

|   | Source  | Paragraphs and footnotes in the study  |
|---|---|--|
| France-Belgium  |   |  |
| Atmospheric pollution at the mouth of the Yser (1973)   | Belgian Parliament, Questions et Réponses, 29 May 1973                                  | 297  |
| Pollution of the Thure, an affluent of the Sambre (1973)  | <i>Ibid.</i> , 4 July 1973  | 298  |
| France-Federal Republic of Germany  |   |  |
| Adoption by a plant in Lorraine, situated near the German border, of German standards of air quality (1980)                                     | International Environment Reporter (Washington, D.C.), vol. 3, No. 9, 10 September 1980 | 190  |
| France-Switzerland  |   |  |
| Pollution of Swiss territory by a French chemical plant situated near the border (1973)   | Annuaire suisse de droit international (Zürich), 1974, vol. 30, p. 147                  | 482  |
| Liberia-Japan   |   |  |
| Oil slick caused by the foundering of the Liberian tanker <i>Juliana</i> on the Japanese coast (1971)   | The Times, London, 1 October 1974, p. 8, col. 1   | 479. 530   |
| United States of America-Canada   |   |  |
| Oil spill at Cherry Point, State of Washington, and pollution of Canadian beaches (1972)  | The Canadian Yearbook of International Law, 1973, (Vancouver), vol. 11, p. 333          | 436, 480, 544  |
| United States of America-Mexico   |   |  |
| Correspondence concerning the polluting activities of the Peyton Packing Company and the Casuco Company situated near the Mexican border (1961) | Whiteman, op. cit., vol. 6, pp. 256-259   | 32, 324, 339, 437  |
| Nuclear and outer s   | SPACE TESTING   |  |
| United Kingdom-Japan  |   |  |
| Nuclear tests in the Christmas Islands (1957)   | Whiteman, op. cit., vol. 4, pp. 596-600   | 33, 90, 91, 189, 216, 218, 236, 239, 247   |
| United States of America-Canada   |   |  |
| "Cannikin" underground nuclear testing in the Aleutian island of Amchitka (1971)  | International Canada (Toronto), vol. 2, 1971, p. 97                                     | 240, 247, 468  |
| United States of America-Japan and Marshall Islands Territory   |   |  |
| Nuclear testing at Eniwetok Atoll (1954-1958)   | Whiteman, op. cit., vol. 4, pp. 565-596   | 33, 84-90, 189, 216, 217, 236, 239, 247, 288-290, 328, 329, 331, 333, 469, 470, 508, 525, 531, 553, 555, 584 |

164, 293

| United States of America-Scientific community                                      |  |                                     |
|--|--|-------------------------------------|
| West Ford Test: launching of 20 kilograms of copper needles into outer space (1961 | The New York Times, issues of 30 July 1961, p. 48, col. 1; 3 February 1962, p. 5, col. 1; 10 May 1962, p. 16, col. 4; 13 May 1963, p. 1, col. 5; 21 May 1963, p. 3, col. 1; 23 September 1963, p. 28, col. 2 | 319                                 |
| Air and spa  | CE CRASHES   |                                     |
| USSR-Canada  |  |                                     |
| Crash of the Soviet satellite Cosmos 954 on Canadian territory (1978)              | International Legal Materials (Washington, D.C.), vol. 18, p. 907  | 33, 48, 399, 464, 527, 541-543, 605 |
| United States of America-Spain   |  |                                     |
| Palomares air crash (1966)   | T. Szuld, <i>The Bombs of Palomares</i> (New York, 1967)   | 523, 554                            |
| Health Pi  | ROTECTION  |                                     |
| Mexico-United States of America  |  |                                     |

Whiteman, op. cit., vol. 6, p. 266

Campaign for the eradication of foot-and-mouth disease (1947-1949)

## RELATIONS BETWEEN STATES AND INTERNATIONAL ORGANIZATIONS (SECOND PART OF THE TOPIC)

[Agenda item 9]

### DOCUMENT A/CN.4/L.383 and Add.1-3

## The practice of the United Nations, the specialized agencies and the International Atomic Energy Agency concerning their status, privileges and immunities

### Supplementary study prepared by the Secretariat

[Original: English] [27 May, 27 June and 28 July 1985]

#### CONTENTS

| Principal instruments cited in the present study   |            | Pag<br>148<br>149 |
|--|------------|-------------------|
|  | Paragraphs |                   |
| Introduction   | 1-4        | 151               |
| A. Summary of practice relating to the status, privileges and immunities of the United Nations                   |            |                   |
| Chapter  |            |                   |
| I. JURIDICAL PERSONALITY OF THE UNITED NATIONS   | 1-10       | 152               |
| Section 1. Contractual capacity  | 1-4        | 152               |
| (a) Recognition of the contractual capacity of the United Nations  | 1          | 152               |
| (b) Choice of law; settlement of disputes and system of arbitration  | 2-4        | 152               |
| Section 2. Capacity to acquire and dispose of immovable property   | 5-8        | 160               |
| (b) Acquisition and disposal of immovable property   | 5-8        | 160               |
| Section 4. Legal proceedings brought by and against the United Nations   | 9          | 160               |
| (c) Claims of a private law nature made against the United Nations and steps taken avoid or mitigate such claims |            | 160               |
| Section 6. Treaty-making capacity  | 10         | 160               |
| (a) Treaty-making capacity of the United Nations   |            | 160               |
| II. PRIVILEGES AND IMMUNITIES OF THE UNITED NATIONS IN RELATION TO ITS PROPERTY, FUN                             | DS         |                   |
| AND ASSETS   |            | 161               |
| Section 7. Immunity of the United Nations from legal process   | 11-14      | 161               |
| (a) Recognition of the immunity of the United Nations from legal process   |            | 161               |
| Section 8. Waiver of the immunity of the United Nations from legal process                                       | 15-21      | 162               |
| (a) Practice relating to waiver by the United Nations of its immunity from legal proces                          |            | 162               |
| (c) Interpretation of the phrase "any measure of execution"  |            | 162               |
| Section 10. Immunity of United Nations property and assets from search and from a                                |            |                   |
| other form of interference   | -          | 163               |
| Section 11. United Nations name, emblem and flag   | 23-25      | 163               |
| (a) United Nations name and emblem   | 23-24      | 163               |
| (b) United Nations flag  | 25         | 163               |
| Section 14. Direct taxes   | 26-36      | 163               |
| (a) Definition of direct taxes   | 26-34      | 163               |
| (iii) Taxes on United Nations financial assets   | 26-28      | 163               |
| (iv) Taxes in respect of the occupation or construction of United Nations premise                                | s 29-32    | 164               |
| (v) Hotel taxes  | 33-34      | 164               |
| (b) Practice in respect of "charges for public utility services"   | 35-36      | 165               |

| Chapter  | Paragraph                | s Pag    |
|--|--------------------------|----------|
| Section 15. Customs duties   |                          | 16       |
| (b) Imposition of customs duties, prohibitions and restrictions  |                          | 16       |
| Section 16. Publications   |                          | 16       |
| (a) Interpretation of the term "publications" and problems re  |                          | 16       |
| (b) United Nations copyright and patents   |                          | 16       |
| Section 17. Excise duties and taxes on sales; important purchases  |                          | 16       |
| (a) Excise duties and taxes on sales forming part of the price to  |                          | 16       |
| (b) Important purchases  |                          | 16       |
| III. PRIVILEGES AND IMMUNITIES OF THE UNITED NATIONS IN RESPE  |                          | 16       |
| Section 18. Treatment equal to that accorded to Governmen telegrams and other communications   |                          | 16       |
| IV. Privileges and immunities of officials of the United Nation  |                          | 17       |
| Section 22. Categories of officials to which the provisions of articl  | es V and VII of the Con- |          |
| vention on the Privileges and Immunities of the United Nations   | •••                      | 17       |
| Section 23. Immunity of officials in respect of official acts  |                          | 17<br>17 |
| (c) Cases of detention or questioning of United Nations of   |                          | 17       |
| public bodies  |                          | 17       |
| Section 24. Exemption from taxation of salaries and emoluments   |                          | 17       |
| (b) Position in the United States of America   |                          | 17       |
| (f) National taxation on non-exempt income   |                          | 17       |
| Section 25. Immunity from national service obligations   |                          | 17       |
| Section 26. Immunity from immigration restrictions and alien regis   |                          | 17       |
| (b) Practice in respect of the United States of America  Section 29. Importation of furniture and effects                                      |                          | 17<br>17 |
|  |                          | 1 /      |
| Section 30. Diplomatic privileges and immunities of the Secre senior officials   |                          | 17       |
| Section 31. Waiver of the privileges and immunities of officials   |                          | 17       |
| V. PRIVILEGES AND IMMUNITIES OF EXPERTS ON MISSION FOR THE UPERSONS HAVING OFFICIAL BUSINESS WITH THE UNITED NATIONS                           |                          | 17       |
| Section 33. Persons falling within the category of experts on Nations  |                          | 17       |
| Section 35. Privileges and immunities of persons having official   |                          |          |
| Nations  |                          | 18       |
| VI. UNITED NATIONS LAISSEZ-PASSER AND FACILITIES FOR TRAVEL  |                          | 18       |
| Section 36. Issue of United Nations laissez-passer and their recog   |                          |          |
| travel documents   |                          | 18       |
| Section 39. Issue of visas for holders of United Nations <i>laissez-pas</i> Section 41. Diplomatic facilities for the Secretary-General and of |                          | 18       |
| travelling on official business  |                          | 18       |
| B. Summary of practice relating to the status, privileges and image and of the International Atomic Energy                                     |                          |          |
| I. JURIDICAL PERSONALITY OF THE SPECIALIZED AGENCIES AND C<br>ATOMIC ENERGY AGENCY   |                          | 18       |
| Section 1. Contractual capacity  | 1-14                     | 18       |
| (a) Recognition of the contractual capacity of the specialized as  |                          | 18       |
| (b) Choice of law; settlement of disputes and system of arbitrat   |                          | 18       |
| Section 2. Capacity to acquire and dispose of immovable property   |                          | 18       |
| Section 3. Capacity to acquire and dispose of movable property   |                          | 18       |
| (a) Recognition of the capacity of the specialized agencies and dispose of movable property  |                          | 18       |
| (b) Licensing and registration of land vehicles, vessels and aircr   |                          | 18       |
| Section 4. Legal proceedings brought by and against the specialize   |                          | 18       |
| Section 5. International claims brought by and against the speciali  | •                        | 18       |

| Chapter |  | Paragraphs      | Page       |
|---------|--|-----------------|------------|
|         | Section 6. Treaty-making capacity  | 40-41<br>40     | 186<br>186 |
|         | (b) Registration, or filing and recording of agreements on the status, privileges and immunities of the specialized agencies and of IAEA   | 41              | 186        |
| II.     | PRIVILEGES AND IMMUNITIES OF THE SPECIALIZED AGENCIES AND OF THE INTERNATIONAL ATOMIC ENERGY AGENCY IN RELATION TO THEIR PROPERTY, FUNDS AND ASSETS  | 42-112          | 187        |
|         | Section 7. Immunity of the specialized agencies and of IAEA from legal process   | 42-54           | 187        |
|         | Section 8. Waiver of the immunity from legal process of the specialized agencies and of IAEA   | 55-57           | 189        |
|         | Section 9. Inviolability of the premises of the specialized agencies and of IAEA and exercise of control by the specialized agencies and by IAEA over their premises   | 58-65           | 189        |
|         | Section 10. Immunity of the property and assets of the specialized agencies and of IAEA from search and from any other form of interference  | 66              | 190        |
|         | Section 11. Name and emblem of the specialized agencies and of IAEA: United Nations flag   | 67-68           | 191        |
|         | Section 12. Inviolability of archives and documents  | 69              | 191        |
|         | Section 13. Immunity from currency controls  | 70-71           | 191        |
|         | Section 14. Direct taxes   | 72-76           | 191        |
|         | Section 15. Customs duties   | 77-87           | 192        |
|         | (a) Imports and exports by the specialized agencies and by IAEA "for their official  | .,              | .,_        |
|         | use"   | 77-78           | 192        |
|         | (b) Imposition of customs duties, prohibitions and restrictions  | 79-83           | 193        |
|         | (c) Sales of articles imported by the specialized agencies and by IAEA   | 84-87           | 193        |
|         | Section 16. Publications   | 88-91           | 193        |
|         | Section 17. Excise duties and taxes on sales; important purchases  | 92-112<br>92-97 | 194<br>194 |
|         | (b) Important purchases  | 92-97<br>98-104 | 194        |
|         | (c) Remission or return of taxes paid  | 105-112         | 196        |
|         | David and the second se |                 |            |
| 111.    | PRIVILEGES AND IMMUNITIES OF THE SPECIALIZED AGENCIES AND OF THE INTERNATIONAL ATOMIC ENERGY AGENCY IN RESPECT OF COMMUNICATION FACILITIES   | 113-120         | 197        |
|         | Section 18. Treatment equal to that accorded to Governments in respect of mails, telegrams and other communications  | 113-115         | 197        |
|         | Section 19. Use of codes and dispatch of correspondence by courier or in bags  | 116-120         | 197        |
| IV.     | PRIVILEGES AND IMMUNITIES OF OFFICIALS OF THE SPECIALIZED AGENCIES AND OF THE INTERNATIONAL ATOMIC ENERGY AGENCY   | 121-205         | 198        |
|         | Section 22. Categories of officials to which the provisions of articles VI and VIII of the Convention on the Privileges and Immunities of the Specialized Agencies apply   | 121-130         | 198        |
|         | Section 23. Immunity of officials in respect of official acts  | 131-136         | 199        |
|         | Section 24. Exemption from taxation of salaries and emoluments   | 137-176         | 200        |
|         | Section 25. Immunity from national service obligations   | 177-180         | 203        |
|         | Section 26. Immunity from immigration restrictions and alien registration  | 181-185         | 204        |
|         | Section 27. Exchange facilities  | 186-187         | 204        |
|         | Section 28. Repatriation facilities in time of international crisis  | 188-189         | 204        |
|         | Section 29. Importation of furniture and effects   | 190-193         | 204        |
|         | Section 30. Diplomatic privileges and immunities of the executive heads and other senior officials of the specialized agencies and of IAEA   | 194-198         | 205        |
|         | Section 31. Waiver of the privileges and immunities of officials   | 199-201         | 205        |
|         | Section 32. Co-operation with the authorities of member States to facilitate the proper administration of justice  | 202-205         | 206        |
| v.      | PRIVILEGES AND IMMUNITIES OF EXPERTS ON MISSION FOR THE SPECIALIZED AGENCIES AND THE INTERNATIONAL ATOMIC ENERGY AGENCY AND OF PERSONS HAVING OFFICIAL BUSINESS WITH THE SPECIALIZED AGENCIES AND THE INTERNATIONAL ATOMIC ENERGY AGENCY   | 206-210         | 206        |
|         | Section 33. Persons falling within the category of experts on mission for the specialized agencies and IAEA  | 206-209         | 206        |
|         | Section 34. Privileges and immunities of experts on mission for the specialized agencies and IAEA  | 210             | 206        |
| 1/1     |  | 211.225         | 307        |
| V1.     | United Nations Laissez-Passer and facilities for travel  Section 36. Issue of United Nations laissez-passer and their recognition as valid travel  | 211-225         | 207        |
|         | documents  | 211-212         | 207        |

| Chapter |                          |   | Paragraphs | Page |
|---------|--------------------------|---|------------|------|
|         |                          | Freedom of movement of personnel of the specialized agencies and of IAEA; ability of persona non grata doctrine   | 213-218    | 207  |
|         | Section 38.              | Issue of visas for holders of United Nations laissez-passer   | 219-220    | 207  |
|         | Section 39.              | Certificates issued by the specialized agencies and by IAEA   | 221-224    | 207  |
|         | Section 40.<br>specializ | Diplomatic facilities for the executive heads and other senior officials of the ted agencies and of IAEA while travelling on official business                    | 225        | 208  |
| VII.    | SELLEMENT                | OF DISPUTES   | 226-237    | 208  |
|         | Section 41.              | Settlement of disputes  | 226-229    | 208  |
|         | Section 42.              | Settlement of disputes regarding alleged abuses of privileges   | 230-233    | 208  |
|         |                          | Reference to the International Court of Justice of differences arising out of the tation of the Convention on the Privileges and Immunities of the Specialized is | 234-237    | 209  |
| VIII.   | ANNINIS M                | D HINAL PROVISIONS  | 238-247    | 209  |
|         |                          | Annexes to the Convention on the Privileges and Immunities of the Specialized s   | 238-239    | 209  |
|         | Section 45.              | Supplemental agreements   | 240-243    | 209  |
|         | ized Ag                  | Accession to the Convention on the Privileges and Immunities of the Special-<br>encies by Member States of the United Nations and by member States of the         | 244 247    | 210  |
|         | speciali                 | zed agencies  | 244-247    | 210  |

#### PRINCIPAL INSTRUMENTS CITED IN THE PRESENT STUDY

#### Conventions and Agreements

Convention on the Privileges and Immunities of the United Nations (New York, 13 February 1946) (United Nations, *Treaty Series*, vol. 1, p. 15, and *ibid.*, vol. 90, p. 327)

Convention on the Privileges and Immunities of the Specialized Agencies (New York, 21 November 1947) (United Nations, *Treaty Series*, vol. 33, p. 261)

Agreement on the Privileges and Immunities of the International Atomic Energy Agency (Vienna, 1 July 1959) (United Nations, *Treaty Series*, vol. 374, p. 147)

Vienna Convention on Diplomatic Relations (Vienna, 18 April 1961) (United Nations, *Treaty Series*, vol. 500, p. 95)

Vienna Convention on Consular Relations (Vienna, 24 April 1963) (United Nations, *Treaty Series*, vol. 596, p. 261)

Convention on Special Missions (New York, 8 December 1969) (United Nations, Juridical Yearbook 1969 (Sales No. E.71.V.4), p. 125)

Vienna Convention on the Representation of States in their Relations with International Organizations of a Universal Character (Vienna, 14 March 1975) (United Nations, *Juridical Yearbook 1975* (Sales No. E.77.V.3), p. 87)

#### Headquarters Agreements

#### United Nations Organization (New York)

Headquarters Agreement concluded between the United Nations Organization and the United States of America (Lake Success, 26 June 1947) (United Nations, *Treaty Series*, vol. 11, p. 11)

United Nations Organization (Geneva) (Office of the United Nations at Geneva)

Interim Arrangement on Privileges and Immunities of the United Nations concluded between the Secretary-General of the United Nations and the Swiss Federal Council (Bern, 11 June 1946, and New York, 1 July 1946) (United Nations, *Treaty Series*, vol. 1, p. 163), modified by an exchange of letters of 5 and 11 April 1963 between the Swiss Federal Council and the United Nations and thereafter entitled "Agreement on the Privileges and Immunities of the United Nations concluded between the Swiss Federal Council and the Secretary-General of the United Nations" (United Nations, *Juridical Yearbook 1963* (Sales No. 65.V.3), pp. 43-44)

#### **ECA**

Agreement relating to the Headquarters of the United Nations Economic Commission for Africa (Addis Ababa, 18 June 1958) (United Nations, *Treaty Series*, vol. 317, p. 101)

#### FCL A

Agreement regulating the conditions for the operation, in Chile, of the Headquarters of the United Nations Economic Commission for Latin America in Chile (Santiago, 16 February 1953) (United Nations, *Treaty Series*, vol. 314, p. 49)

#### **ECWA**

Agreement relating to the Headquarters of the United Nations Commission for Western Asia (Baghdad, 13 June 1979) (United Nations, *Treaty Series*, vol. 1144, p. 213)

#### **ESCAP**

Agreement relating to the Headquarters of the Economic Commission for Asia and the Far East in Thailand (Geneva, 26 May 1954) (United Nations, *Treaty Series*, vol. 260, p. 35)

#### **FAO**

Agreement regarding the Headquarters of the Food and Agriculture Organization of the United Nations (Washington, D.C., 31 October 1950) (United Nations, Legislative Texts and Treaty Provisions concerning the Legal Status, Privileges and Immunities of International Organizations, vol. II (Sales No. 61.V.3), p. 187)

Agreement relating to the Headquarters of the International Atomic Energy Agency (Vienna, 11 December 1957) (United Nations, Treaty Series, vol. 339, p. 110)

Articles of Agreement of the International Bank for Reconstruction and Development (Bretton Woods, 27 December 1945) (IBRD, Articles of Agreement of the International Bank for Reconstruction and Development (Washington, D.C., 1980), art. VII)

Agreement relating to the Headquarters of the International Civil Aviation Organization (Montreal, 14 April 1951) (United Nations, Treaty Series, vol. 96, p. 155)

#### IDA

Articles of Agreement of the International Development Association (Washington, D.C., 24 September 1960) (IDA, Articles of Agreement of the International Development Association (Washington, D.C., 1974), art. VIII)

Articles of Agreement of the International Finance Corporation (Washington, D.C., 25 May 1955) (IFC, Articles of Agreement of the International Finance Corporation (Washington, D.C., 1986), art. VI)

Articles of Agreement of the International Monetary Fund (Bretton Woods, 27 December 1945) (IMF, Articles of Agreement of the International Monetary Fund (Washington, D.C., 1978) art. IX)

#### ITU

Agreement between the Swiss Federal Council and the International Telecommunication Union concerning the legal status of that organization in Switzerland (Geneva, 22 July 1971) (United Nations, Juridical Yearbook 1971 (Sales No. E.73.V.1), p. 26)

#### UNEP

Agreement relating to the Headquarters of the United Nations Environment Programme (Nairobi, 26 March 1975) (United Nations, Treaty Series, vol. 962, p. 91)

#### UNESCO

Agreement relating to the Headquarters of the United Nations Educational, Scientific and Cultural Organization and to its privileges and immunities on French territory (Paris, 2 July 1954) (United Nations, Treaty Series, vol. 357, p. 3)

#### UNIDO

Agreement between the United Nations and Austria regarding the Headquarters Seat of the United Nations Industrial Development Organization (Vienna, 19 January 1981) (United Nations, Juridical Yearbook 1981 (Sales No. E.84.V.1), p. 11), confirmed by an exchange of letters dated 20 December 1985

See the Headquarters Agreement of the United Nations Office at Geneva, rendered applicable to the Universal Postal Union by an exchange of letters of 5 February/22 April 1948 between UPU and the Swiss Federal Council

#### WHO

Agreement between the Swiss Federal Council and the World Health Organization regulating the legal status of that organization in Switzerland (Bern, September 1955) (Switzerland, Recueil systématique du droit fédéral, Bern, 1970 (0.192.120.281))

East

#### LIST OF ABBREVIATIONS

| CFA<br>ECA<br>ECLA<br>ECWA<br>ECAFE<br>FAO | Communauté financière africaine Economic Commission for Africa Economic Commission for Latin America Economic Commission for Western Asia Economic Commission for Asia and the Far East Food and Agriculture Organization of the United Nations International Atomic Energy Agency |
|--|--|
| IBRD<br>World Bank                         | International Bank for Reconstruction and Development  |
| ICAO ´                                     | International Civil Aviation Organization  |
| ICC  | International Chamber of Commerce  |
| I.C.J.                                     | International Court of Justice   |
| IDA  | International Development Association  |
| IDB  | Inter-American Development Bank  |
| IFAD                                       | International Fund for Agricultural Development  |
| IFC  | International Finance Corporation  |
| ILO  | International Labour Organisation  |
| IMF  | International Monetary Fund  |
| ITU  | International Telecommunication Union  |
| UNDP                                       | United Nations Development Programme   |
| UNEF                                       | United Nations Emergency Force   |
| UNEP                                       | United Nations Environment Programme   |
| UNESCO                                     | United Nations Educational, Scientific and Cultural Organization   |
| UNICEF                                     | United Nations Children's Fund   |
| UNIDO                                      | United Nations Industrial Development Organization   |
| UNITAR                                     | United Nations Institute for Training and Research   |
| UNRRA                                      | United Nations Relief and Rehabilitation Administration  |
| UNRWA                                      | United Nations Relief and Works Agency for Palestine Refugees in the Near  |
| UNTSO                                      | United Nations Truce Supervision in Palestine  |
| VAT  | Value-added tax  |

UPU Universal Postal Union
WFP World Food Programme
WHO World Health Organization
WIPO World Intellectual Property Organization

\* \*

I.C.J. Reports

ICJ, Reports of Judgments, Advisory Opinions and Orders

\* \*

### NOTE CONCERNING QUOTATIONS

Unless otherwise indicated, quotations from works in languages other than English have been translated by the Secretariat.

#### Introduction

- 1. At its thirty-fifth session, in 1983, the International Law Commission requested the Secretariat "to revise the study prepared in 1967 on 'The practice of the United Nations, the specialized agencies and the International Atomic Energy Agency concerning their status, privileges and immunities" and to update that study in the light of the replies to the further questionnaire sent on 13 March 1978 by letter of the Legal Counsel of the United Nations addressed to the legal counsels of the specialized agencies and IAEA in connection with the status, privileges and immunities of those organizations, except in matters pertaining to representatives of States, and which complemented the questionnaire on the same topic sent out on 5 January 1965".
- 2. The present study was prepared by the Secretariat in accordance with that request. It constitutes a supplement to Part Two of the 1967 study, entitled "The organizations", and closely follows its structure and format. The table of contents is based upon that of Part Two of the 1967 study. Part A, entitled "Summary of practice relating to the status, privileges and immunities of the United Nations", is based upon the provisions of the Convention on the Privileges and Immunities of the United Nations, adopted by the General Assembly on 13 February 1946, and part B, entitled "Summary of practice relating to the status, privileges and immunities of the specialized agencies and the International Atomic Energy Agency", upon those of the Convention on the Privileges and Immunities of the Specialized Agencies,

- adopted by the General Assembly on 21 November 1947.4
- 3. The present study summarizes the major features of the practice followed since 1966 by the United Nations, the specialized agencies and IAEA in respect of their status, privileges and immunities. An attempt has been made to avoid, as far as possible, repeating those features of the practice indicated in the 1967 study which remained valid in 1985 (in particular, the sections for which the Secretariat had received no additional information have been omitted); hence the need to read this supplement in conjunction with the previous study. Part A was prepared on the basis of material taken largely from the records of the Office of Legal Affairs of the United Nations Secretariat. Part B was prepared on the basis of replies to the questionnaire transmitted to the heads of the specialized agencies and IAEA by letter of the Legal Counsel dated 13 March 1978. In the course of preparing the present study, the Legal Counsel addressed a further letter, dated 24 October 1984, to the heads of the specialized agencies and IAEA, inviting them to submit any information additional to that submitted in response to the earlier letter.
- 4. As in the 1967 study, most of the international agreements and national enactments mentioned in the present supplement are contained in the two volumes of the United Nations Legislative Series entitled Legislative Texts and Treaty Provisions concerning the Legal Status, Privileges and Immunities of International Organizations.<sup>5</sup> In addition, legislative texts and treaty provisions concerning the legal status of the United Nations and the specialized agencies and IAEA may be found in the successive issues of the United Nations Juridical Yearbook, beginning in 1962.

<sup>&</sup>lt;sup>1</sup> Yearbook . . . 1967, vol. II, p. 154, document A/CN.4/L.118 and Add.1 and 2.

<sup>&</sup>lt;sup>2</sup> Yearbook . . . 1983, vol. II (Part Two), pp. 80-81, document A/38/10, para. 277 (e).

<sup>&#</sup>x27;Yearbook...1967, vol. II, p. 154, document A/CN.4/L.118 and Add.1 and 2. Part One of the 1967 study, entitled "The representatives of Member States", concerned matters relevant to the first part of the topic, which was the subject of the 1975 Vienna Convention on the Representation of States in their Relations with International Organizations of a Universal Character.

<sup>\*</sup> The provisions of the Agreement on the Privileges and Immunities of IAEA are the same as or closely similar to those contained in the Convention on the Privileges and Immunities of the Specialized Agencies.

Junited Nations publications, Sales Nos. 60.V.2 and 61.V.3.

## A. SUMMARY OF PRACTICE RELATING TO THE STATUS, PRIVILEGES AND IMMUNITIES OF THE UNITED NATIONS

#### CHAPTER I

### Juridical personality of the United Nations

#### Section 1. Contractual capacity

### (a) Recognition of the contractual capacity of the United Nations

1. Questions are raised from time to time concerning the legal personality and status of subsidiary bodies of the Organization such as UNDP, UNRWA or UNICEF. In response to an inquiry regarding the legal status of WFP, which was established by concurrent resolutions of the United Nations General Assembly and the FAO Conference, the Office of Legal Affairs, in an unpublished memorandum of 24 March 1969, examined the contractual capacity of WFP and its juridical status as follows:

We concur with your view that WFP possesses the legal capacity to acquire and dispose of movable property, enter into contracts, and be sued. We further agree that WFP possesses, under the General Regulations, the authority to enter into project agreements, that, subject to the considerations set forth below in this letter, its capacity to enter into international agreements is not to be restrictively interpreted, and that WFP must be deemed to possess various implied powers, in addition to those expressly conferred by the United Nations and FAO. As regards the entry by WFP into agreements with Governments relating to the administration of contributions under the 1980 Food Aid Convention, . . . such an agreement has been concluded by the United Kingdom and WFP through an exchange of letters.

We do, however, have reservations concerning the view...that the possession by an entity of the capacity to perform the legal acts referred to in your memorandum necessarily signifies that it enjoys an independent juridical personality. While the converse of this proposition-that a body enjoying independent juridical personality necessarily possesses legal capacity—is manifestly valid, it seems to us that whether a body possessing legal capacity may also be deemed to enjoy an independent juridical personality depends in each case upon the relevant terms of its constituent instrument. These views are based upon the practice observed by the United Nations with respect to various subsidiary bodies. For example, UNDP, while enjoying the capacity to enter into international agreements in its own name and the competence to perform other legal acts, is not considered to possess a juridical personality separate and distinct from the United Nations. International agreements entered into by UNDP are registered ex officio by the Secretariat under article 4 of the Regulations concerning the registration and publication of treaties and international agreements. Similarly, UNDP is entitled to the privileges and immunities of the United Nations by virtue of its status as a subsidiary body of the Organization, and this entitlement, therefore, subsists with respect to all Governments, whether or not they have entered into a basic agreement with UNDP stipulating that the Convention on the Privileges and Immunities of the United Nations shall apply to UNDP.

### (b) Choice of law; settlement of disputes and system of arbitration

2. The attribution to properly constituted arbitral bodies of jurisdiction for the settlement of disputes concerning contract claims has not been considered as implying a choice as to the applicable law. In the rare cases where the problem has arisen, the determination of the law applicable to the contract has been left to the parties

to the dispute. One such case may be cited in this connection, namely, that of Starways Limited v. United Nations Organization (1969), which is summarized in the memorandum of the Office of Legal Affairs reproduced in the present study (see p. 155 below).

- 3. More generally, the determination of the applicable law has been left to the arbitrators. The overwhelming majority of commercial contracts which have been entered into by the United Nations have been performed without the occurrence of any serious difficulties. The number of disputes presented to arbitration, therefore, is not great and few formal written opinions have been rendered. See also the following cases: Balakhany (Chad) Limited v. Food and Agriculture Organization of the United Nations (1972),2 Aerovías Panama, S.A. v. United Nations (1965)3, Lamarche v. Organisation des Nations Unies au Congo (1965).4 Very few cases regarding commercial contracts to which the United Nations was a party have come before municipal courts; in instances in which the United Nations was the plaintiff the most frequent issue was the capacity of the Organization to institute proceedings. In one case it was held that a subsidiary organ of the United Nations bringing an action arising out of a contract was obliged to comply with venue requirements.5
- 4. An inquiry received from the Institute of International Law in 1976 provided the Office of Legal Affairs with the occasion for a comprehensive review of the questions of the law applicable to contracts concluded by the United Nations with private parties and the procedures for settling disputes arising out of such contracts. In its reply to a questionnaire submitted by the Institute, the Office of Legal Affairs provided the following information:
- 1. Do the Constitution, the internal rules of your organization, or international conventions (headquarters agreements, etc.) provide any indication as to the law applicable to contracts concluded with private parties?

<sup>&#</sup>x27;Arbitral award of 24 September 1969, under the rules of the American Arbitration Association, rendered by Howard H. Bachrach, sole arbitrator. For a summary of the case, see United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), pp. 233-234.

<sup>&</sup>lt;sup>2</sup> Arbitral award of 29 June 1972 (arbitrator, Barend van Marwijk-Kooy). See United Nations, *Juridical Yearbook 1972* (Sales No. E.74.V.1), p. 206.

<sup>&</sup>lt;sup>3</sup> Arbitral award of 14 January 1965, rendered under the rules of the American Arbitration Association.

<sup>4</sup> Arbitral award of 6 August 1965, rendered under the rules of ICC.

<sup>&</sup>lt;sup>3</sup> United Nations Korean Reconstruction Agency v. Glass Production Methods (1956) (Federal Supplement, vol. 143, 1957, p. 248).

<sup>&</sup>lt;sup>6</sup> United Nations, *Juridical Yearbook 1976* (Sales No. E.78.V.5), pp. 159-176.

The legal capacity of the Organization to contract is derived from Article 104 of the United Nations Charter' and granted express recognition in section 1 (a) of the Convention on the Privileges and Immunities of the United Nations (hereinafter referred to as the 'General Convention'').2 This capacity has been fully acknowledged in practice. Recognition of United Nations capacity in this sphere has been given both by State organs on which the Organization has needed to rely in connection with the performance of its contracts and by official bodies, private firms and individuals with whom the United Nations has wished to enter into contractual relations. The United Nations has exercised its contractual capacity both through officials of the Secretariat acting on behalf of the Secretary-General, in his capacity as chief administrative officer of the Organization, and through subsidiary bodies established for particular purposes by one of the principal organs. Subsidiary organs, such as UNICEF and UNRWA, which have been entrusted by the General Assembly with a wide range of direct functions, have regularly entered into commercial contracts in their own names.

In addition, legal capacity to contract has been given express recognition in the statutes and regulations of United Nations organs, e.g. in the Regulations of UNEF,4 in the Agreement with Thailand concerning the Headquarters of ECAFE,5 and in the Regulations for the United Nations Force in Cyprus.6.

So far as is known, no State has placed any express limitation upon its recognition of the contractual capacity of the United Nations. The Organization may therefore use its contractual powers, subject to the limitations imposed by its own structure and the authority given by resolutions adopted by its organs, for the same purposes as any other legal entity recognized by particular municipal systems.

Neither the United Nations Charter, the General Convention nor any of the regulations granting legal capacity to contract to subsidiary organs of the Organization have specified the law applicable to contracts concluded with private parties.

Regarding the application of article III, section 7 (b), of the Headquarters Agreement between the United Nations and the United States of America (the "Headquarters Agreement"), we wish to make the following observations.

Article III, section 7 (b) of the Headquarters Agreement provides:

"Except as otherwise provided in this agreement or in the General Convention, the federal, State and local law of the United States shall apply within the headquarters district."

Whether such a provision applies to contracts concluded in the Headquarters district must be interpreted in conjunction with both the Charter of the United Nations and the General Convention.

This principle of interpretation may be drawn from the Headquarters Agreement itself. In section 26° thereof, it is stated that the provisions of that Agreement and those of the General Convention are to be treated as complementary. Section 2710 of the Agreement further states that the provisions of that Agreement are to be construed so as to enable the United Nations to discharge its responsibilities and fulfil its purposes.

The Organization executes a great number of its contracts at Headquarters. However, a substantial number of contracts are also executed either by the United Nations itself or through its subsidiary organs under a variety of conditions and in numerous countries. If the Headquarters Agreement were interpreted to apply the federal, State and local law of the United States to contracts signed at Headquarters, there would arise a dichotomy of practice in the interpretation of the law to be applied to such contracts. Contracts signed at Headquarters would be governed by United States law, whereas contracts signed elsewhere (including all other places in the United States) would be governed by general principles of law or by the law specified in the contract.11 This could result in confusion and in difficulties not consonant with the proper and efficient performance of the functions of the Organization. The position of the Organization has been that the place of the signing of a contract can at most be considered only as one of many factors in a determination of the law of the contract. For this reason, it has never recognized article III, section 7 (b), of the Headquarters Agreement as imposing local law upon contracts concluded at Headquarters.

Consistent with this interpretation, the Organization relies on general principles of law in the interpretation of contracts concluded by it with private parties. Neither the United States nor any other State has placed any express limitation upon this interpretation of the relevant law to be applied in the construction of contracts entered into by the United Nations.

The application of general principles of law to contracts entered into by the United Nations may be seen, paraphrasing Judge Jessup, as an invocation of conflict rules and principles themselves.12 It may be presumed that the forum called upon to adjudicate a dispute arising out of a contract between the United Nations and a private party would be guided by these principles. In such a case, the selection of the relevant law governing the contract, as suggested by Professor Cavers, would not be the result of the automatic operation of a rule or principle of selection but of a search for a just decision in the principal case.13

II. (a) What are the purposes of the principal contracts which your organization concluded with private parties? Could you classify the different types of contracts involved?

The United Nations has entered into a variety of contracts of a private law character. At the Headquarters of the United Nations, these include, for example, contracts for maintenance, for purchase of office equipment, for the leasing of premises, for printing, etc.14 They further include contracts between private parties and the United Nations for materials, supplies, equipment, studies, etc., when the Organization acts as the executing agency for contracts with private parties under agreements concluded with Governments by other United Nations agencies, e.g. UNDP. In addition, the United Nations

<sup>&#</sup>x27; Article 104 of the Charter states:

The Organization shall enjoy in the territory of each of its Members such legal capacity as may be necessary for the exercise of its functions and the fulfilment of its purposes."

<sup>&</sup>lt;sup>2</sup> United Nations, Treaty Series, vol. 1, p. 15. Article 1, section 1 (a) of the General Convention states in part:

<sup>&</sup>quot;The United Nations shall possess juridical personality. It shall have the capacity:
"(a) to contract; . . . "

<sup>&</sup>lt;sup>3</sup> Yearbook . . . 1967, vol. II, p. 207, document A/CN.4/L.118 and Add.1 and 2.

<sup>\*</sup>ST/SGB/UNEF/1. Article 27 of those Regulations provided: "The Commander shall enter into contracts and make commitments for the purpose of carrying out his functions under these regulations."

<sup>&</sup>lt;sup>5</sup> United Nations, Treaty Series, vol. 260, p. 35, art. II, sect. 2.

<sup>\*</sup> ST/SGB/UNFICYP/1, art. 22. Reproduced in United Nations, Juridical Yearbook 1964 (Sales No. 66.V.4), p. 181. The regulations came into force on 10 May 1964.

See Yearbook . . . 1967, vol. II, p. 208, document A/CN.4/L.118 and Add.1 and 2.

<sup>\*</sup> United Nations, Treaty Series, vol. 11, p. 11.

<sup>&</sup>quot;Section 26

<sup>&</sup>quot;The provisions of this agreement shall be complementary to the provisions of the General Convention. In so far as any provision of this agreement and any provisions of the General Convention relate to the same subject matter, the two provisions shall, wherever possible, be treated as complementary so that both rovisions shall be applicable and neither shall narrow the effect of the other; but in any case of absolute conflict, the provisions of this agreement shall prevail.

<sup>&</sup>quot; "Section 27

<sup>&</sup>quot;This agreement shall be construed in the light of its primary purpose to enable the United Nations at its headquarters in the United States fully and efficiently to discharge its responsibilities and fulfil its purposes."

<sup>&#</sup>x27; The Organization does not consider it to be inconsistent with these principles for it to rely upon the law imposed by the contract itself. In a limited number of cases a contract may specify a particular law of applicability. In such cases, the contract may be construed consistently with such a provision. Clauses of the latter description have now almost ceased to be used (see Yearbook. 1967, vol. II, p. 208, document A/CN.4/L.118 and Add.1 and 2). More often the contract is silent on the law of applicability. In such cases, general principles

<sup>12</sup> P. C. Jessup, Transnational Law (New Haven, Conn., Yale University Press, 1956), p. 94. It should be noted also that Jessup refers to the Serbian Loans case (1929) before the Permanent Court of International Justice, concerning which he states: "It (the Court) noted that some rules of private international law are to be found in treaties and are thus transformed into 'true international law'." (Ibid., p. 95.)

<sup>13</sup> Ibid., p. 99.

<sup>14</sup> United Nations, Repertory of Practice of United Nations Organs, vol. V, Articles 92-111 of the Charter (Sales No. 1955.V.2 (vol. V)), p. 332.

contracts with private parties, both individuals and institutional or corporate entities, for work on a short-term basis supplemental to its work at Headquarters, e.g. research, editing, translation. It also concludes contracts with private individuals for their services as consultants or experts. These contracts are, then, classifiable as contracts for materials and equipment and contracts for services.

Of course, the Organization concludes a large number of contracts for the services of staff members.13 However, such contracts are considered by the Organization to be governed by its internal administrative law as established by the Staff Regulations,16 Staff Rules17 and Administrative Instructions18 of the United Nations.

11. (b) Do the contracts concluded between your organization and private parties generally (or occasionally, in which case on what occasions? specify the law or legal system which governs them?

Generally speaking, United Nations contracts (both those of a commercial nature and employment contracts) have not specified the law considered to be applicable to such agreements.<sup>19</sup> In the case of employment contracts, the contract itself has formed part of a growing system of international administrative law, independent of given systems of municipal law. The references to municipal law contained in employment contracts have therefore been specific rather than general (e.g. social security laws). Very occasionally, they have been introduced for the purpose of providing a convenient yardstick for measuring compensation or separation benefits.20 As indicated above, clauses of the latter description have now almost ceased to be used. In any case, at no time did they amount to a choice of an actual system of municipal law to govern the entire terms of an employment contract. An internal appellate system has been established to consider disputes of a serious nature regarding employment contracts of staff members. The United Nations Administrative Tribunal has referred both to the internal administrative law of the Organization and to general principles of law in interpreting employment contracts. It has largely avoided references to municipal systems.

It the case of commercial contracts, express reference has rarely been made to a given system of municipal law. The standard practice is for the contract to contain no choice-of-law clause as such; provision is made, however, for the settlement of disputes by means of arbitration when agreement cannot be reached by direct negotiations.21 For example, in the case of contracts concluded with parties resident

15 The Organization is empowered to contract for the services of staff members, through the Secretary-General, under Article 101, paragraph 1, of the United Nations Charter, which states:

'The staff shall be appointed by the Secretary-General under regulations established by the General Assembly.'

"The General Assembly established the Staff Regulations of the United Nations in accordance with Article 101 of the United Nations Charter by resolution 590 (VI) of 2 February 1952. They have thereafter been amended from time to time by that body

17 ST/SGB/Staff Rules/1/Rev.4 and Corr.1, 1977.

18 Administrative instructions are internal Secretariat documents that contain or deal with Staff Regulations and Staff Rules, interpretations of those regulations and rules, the Secretary-General's policy for implementation of those regulations and rules, instructions and procedures and statements of established policy. They are the principal means by which the Secretary-General communicates with the staff on matters of financial, administrative and personnel policies. The authority for the issuance of administrative instructions is the circular ST/SGB/100, dated 14 April 1954. See ST/A1/226 and Amend.1.

"See Yearbook . . . 1967, vol. II, p. 208, document A/CN.4/L.118 and Add,1 and 2.

20 Ibid. For examples of employment contracts containing clauses of this nature, see the decisions in the following cases: Hilpern v. United Nations Relief and Works Agency for Palestine Refugees in the Near East (1955, 1956, 1956) and Radicopoulos v. United Nations Relief and Works Agency for Palestine Refugees in the Near East (1957), in United Nations, Judgments of the Administrative Tribunal, Nos. 1-70, 1950-1957, (Sales No. 58.X.1), judgments Nos. 57, 63, 65 and 70.

21 Although informal negotiations between the parties are preferred before the contractual provision on arbitration is invoked, no specific provision is made for such process in contracts concluded by the Organization at Headquarters. However, under the governing procedures for contracts concluded by the Geneva Office of the United Nations, provision is made for such negotiations through the prior use of a designated expert. Article 25 of the Cahier des clauses el conditions générales applicables aux marchés de fournitures (MUN/251/68-GE.68-6632) provides:

#### "Expert opinions

"1. If any dispute arises as to the interpretation of execution of the contract, the parties shall arrange to obtain an expert opinion prior to the institu-

in the United States, reference may be made to arbitration in accordance with the procedures established by the American Arbitration Association, by the Inter-American Arbitration Association in respect of contracts with Latin American suppliers, or by ICC in many of the remaining cases. No further reference is made in the contract to the legal system to be applied.22

In an opinion of the General Legal Division of the Office of Legal Affairs of the United Nations, drafted in reply to an inquiry of the Legal Office of FAO, the question of specifying a legal system in United Nations contracts was addressed. That opinion, dated 10 December 1962, stated:

"You have deliberately omitted from your proposed standard form any provision making the law of a particular State applicable to the particular contract, and have expressly provided instead in your article 17 that the rights and obligations of the parties would be governed by the agreement and by generally recognized principles of law, to the exclusion of any given municipal law. The idea behind your practice is not to volunteer to make the laws of any particular State applicable to our contract. However, our feelings on this point are reflected in the contract by means of a complete absence of any provision on this matter rather than an express provision such as you have included in your own contract.

"We have felt that it would be preferable as a matter of normal practice to deal with the question only if and when it actually arises. and in the light of the circumstances of the case, rather than in advance by means of a provision in the contract on this point. However, some of our contracts have included a provision making a particular law applicable because of the importance of such a provision to the other party. When the proposal is to make New York law applicable, our familiarity with that law makes it relatively easy for us to accept such a proposal. We try to avoid laws with which we are unfamiliar, but have on occasion had to accept. In such cases, the law is usually that of the country of residence of the other party or in which he has a place of business, and I cannot recall any instance in which the laws of a third State (that is, other than the country of residence of the other party or New York State) has been made the applicable law.

The view expressed in that opinion reflects what has continued to be the approach of the United Nations with regard to specifying any governing law or legal system.

II. (c) If the contracts concluded between your organization and private parties generally specify the law or legal system which govern them, do they refer to international legal rules (international law, the internal law of the organization or general principles of law) or to a national legal system (which one?)? In the latter case, is the law considered "frozen" as of a particular date or is there no limitation of this sort? Is there reference in the contract to the national law on a subsidiary basis? Is there reference to a combination of the national law and general principles of law? Or does the applicable law vary according to the contract? In the last alternative, do you make the distinction on the basis of the importance of the contract. its subject matter, the fact that it is concluded and carried out in one country or in several, the public or private status of the other party to the contract, etc.? What are the effects of such distinctions? Please provide examples of the clauses used.

The contracts in question do not specifically refer to any international legal rules. However, it has been the practice of the United

tion of any judicial proceedings. The more expeditious of the two parties shall notify the other party in writing of the subject of the dispute and shall propose the name of an expert. The other party shall, within 10 days, signify whether or not it agrees to the appointment of that expert and, if it does not so agree, shall make a counter-proposal, to which a reply shall be given within 10 days after notification thereof. This exchange of correspondence shall be by registered letter with acknowledgement of receipt.

'2. If the two parties fail to agree, the expert shall be appointed, at the request of the more expeditious party, by the President of the International Chamber of Commerce

The expert shall have full powers to require the submission to him of any documents, of whatever kind, and to seek such explanations from the parties as he deems necessary in order to determine the nature and cause of the dispute. His function shall be to draw up and communicate to the parties, within one month after the date of his appointment, a report analysing the origin and nature of the dispute which has arisen and to propose a settlement.

'4. The costs of the expert opinion shall be apportioned equally between

the two parties.'

<sup>22</sup> See Yearbook . . . 1967, vol. 11, p. 209, document A/CN.4/L.118 and Add.1 and 2.

Nations to interpret the contracts concluded by it on the basis of general principles of law, including international law, and upon the standards and practice established by its internal law, including its Financial Regulations,<sup>23</sup> principles of delegation of authority under the United Nations Charter and the internal rules and procedures promulgated thereunder.

The law considered to apply would be for matters of substance all applicable laws to which it may refer which were in force at the time of the conclusion of the contract.<sup>24</sup> For matters of adjective or procedural law which might arise in connection with the resolution of a dispute, the applicable law would be that law in effect at the time of the resolution of the dispute.

No reference is generally made to municipal law either as a primary or subsidiary basis.

No reference is made to the application of any combination of municipal law and general principles of law. However, from a practical point of view special attention is given in the making of contracts that such contracts be in general conformity with the law of the place where the contract is made and is to be executed and the national law of the private parties with which the contract is concluded. Similar considerations of a general character may be given to municipal laws in the resolution of disputes which may therefrom arise. However, in no case does the United Nations consider the law of any national system to be binding upon it either in the execution of contracts or in the settlement of disputes arising therefrom.

As a result, while no reference will be made to applicable law in its contracts, the United Nations may give special attention to laws of national jurisdictions and may on occasion consult with local authorities as to the current status of municipal laws as a matter of comity.

II. (d) What is the most recent trend in the contractual practice of your organization?

The most recent trend in United Nations contractual practice is to avoid wherever possible reference to any specific law of application, especially any system of national law, and to consider the governing law of the contract to be found in general principles of law, including international law, as well as in the terms of the contract itself.

III. Is there any case law or established practice concerning the law applicable to contracts concluded by your organization? If so, please give examples and the transcripts of the main decisions taken in this respect.

The established practice concerning the law applicable to contracts is, as stated otherwise herein, to reject any specific reference to municipal laws and to rely on general principles of law in the interpretation of contracts with private parties.

Section 1 (c) of the General Convention refers expressly to the capacity of the United Nations "to institute legal proceedings". This capacity has been widely recognized by judicial and other State authorities.<sup>25</sup> United Nations practice in respect of the receipt of

private law claims, specifically claims in contract, and steps taken to mitigate or avoid such claims, is not extensive.

One example of judicial action may be cited with reference to the resolution of disputes in respect of contracts to which the United Nations was a party.

In Balfour, Guthrie & Co. Ltd. v. United States (1950),26 the United Nations brought an action arising out of the loss of and damage to a cargo of milk which had been shipped on behalf of UNICEF on a United States vessel; the United Nations action was joined with that of six other shippers. The Court, having regard to the terms of Article 104 of the United Nations Charter which, as a treaty ratified by the United States of America, formed part of the law of the United States, stated that: "No implemental legislation would appear to be necessary to endow the United Nations with legal capacity in the United States." It noted further: "The President, however, has removed any possible doubt by designating the United Nations as one of the organizations entitled to enjoy the privileges conferred by the International Organizations Immunities Act", under section 2 (a) of that Act. These privileges included "to the extent consistent with the instrument creating them" the capacity "to institute legal proceedings".

In addition, a number of arbitrations have been conducted in which the United Nations was a party. In Starways Limited v. United Nations (1969), the United Nations had contracted with Sabena airlines for the charter of several DC-4 aircraft to be stationed in the Democratic Republic of the Congo, in connection with the United Nations mission in the Congo. One such aircraft belonged to and was operated by Starways Limited, a subcontractor of Sabena. The aircraft was destroyed by fire on 17 September 1961, having been attacked by rebel forces hostile to the United Nations mission. A claim was brought and submitted to arbitration. The arbitration agreement stipulated that the question of contractual liability was excluded from the terms of reference.

However, of special interest is the fact that the applicable law was stipulated to be that of the (former) Belgian Congo. The Arbitration Agreement stipulated:

"Except for the conduct of the case and the procedure indicated in this agreement, the law applied by the Arbitrator shall be the codes and legislation of the Belgian Congo which remained in force in the Democratic Republic of the Congo pursuant to article 2 of the Loi fondamentale of 19 May 1960."

It should, however, be pointed out that the applicable law here was established by agreement between the parties. It was neither stipulated by the contract nor automatically applied as a matter of conflict of law principles.

IV. Is the establishment of contracts (for supplies, etc.) preceded by calls for tenders on a competitive basis? What rules govern such procedures?

The Office of General Services of the United Nations Secretariat is responsible for procuring equipment, supplies and services in accordance with the prescribed Financial Regulations and Rules of the United Nations<sup>24</sup> (regulation 10.5 and rules 110.16-110.24). Financial regulation 10.5 provides that tenders for such equipment, supplies and other requirements are normally to be invited by advertisement. Contracts may be entered into only by duly authorized officers of the Organization. That authority normally rests with the Assistant Secretary-General, Office of General Services, or that officer's authorized delegate (rule 110.16). A Committee on Contracts has been established to render advice to the Assistant Secretary-General, Office of General Services, in matters involving, but not limited to, single requisitions of \$10,000 or more; contracts involving income to the Organization of \$5,000 or more, and proposals for modifications and renewals of contracts (rule 110.17).

Normally, contracts are let after competitive bidding. Tenders are invited by advertising through publication or distribution of formal invitations to bid. However, in cases where the nature of the work involved precludes invitation of tenders and where proposals are called,

<sup>&</sup>lt;sup>21</sup> ST/SGB/Financial Rules of 2 July 1975, and successive revisions.

<sup>&</sup>quot;Attention must be called, however, to the special circumstances governing an employment contract of a staff member of the United Nations. In addition to the letter of appointment, which forms the primary basis for the contractual relationship, the Staff Rules and Regulations may also form part of the basis of the contract. In that regard, the Administrative Tribunal of the United Nations, in particular in its judgment No. 19 in Kaplan v. Secretary-General of the United Nations (1953), has distinguished between the contractual and statutory elements in the relation between staff members and the Organization:

<sup>&</sup>quot;All matters being contractual which affect the personal status of each staff member, e.g. nature of his contract, salary, grade;

<sup>&</sup>quot;All matters being statutory which affect in general the organization of the international civil service, and the need for its proper functioning, e.g. general rules that have no personal reference.

<sup>&</sup>quot;While the contractual elements cannot be changed without the agreement of the two parties, the statutory elements on the other hand may always be changed at any time through regulations established by the General Assembly, and these changes are binding on staff members." (United Nations, Judgments of the Administrative Tribunal, Nos. 1-70, 1950-1957 (Sales No. 58.X.1), p. 74.)

See also judgment No. 202 in Queguiner v. Secretary-General of the Inter-Governmental Maritime Consultative Organization (1975). (Ibid., Nos. 167-230, 1973-1977 (Sales No. E.78.X.1), p. 317).

<sup>&</sup>lt;sup>23</sup> Yearbook . . . 1967, vol. II, p. 216, document A/CN.4/L.118 and Add.1 and 2.

<sup>26</sup> Federal Supplement, vol. 90, 1950, p. 831.

<sup>&</sup>lt;sup>27</sup> Arbitral award of 24 September 1969, under the rules of the American Arbitration Association, rendered by Howard H. Bachrach, sole arbitrator. For a summary of the case, see United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), pp. 233-234.

<sup>34</sup> See footnote 23 above.

a comparative analysis of such proposals is kept on record (rule 110.18). Contracts may be awarded without advertising or formal invitations to bid when the contract involves a commitment of less than \$2,500 in the case of United Nations Headquarters, the United Nations Office at Geneva and UNIDO, Vienna, and \$1,000 in the case of regional commissions, provided that the award is made in conformity with designated specifications (rule 110.19). Other exceptions concern cases where prices are fixed by national legislation (rule 110.19 (b)), where standardization of supplies or equipment has received prior approval of the Committee on Contracts, or where the subject of the contract is considered to be a matter of special priority or urgency for the Organization (rule 110.19 (c) to (g)), or where the Assistant Secretary-General, Office of General Services, determines that competitive bidding would not give satisfactory results (rule 110.19 (h)). All bids are publicly opened at the time and place specified in the invitation to bid (rule 110.20). Contracts are awarded to the lowest acceptable bidder. However, when the interests of the Organization so require, all bids may be rejected (rule 110.21). Written contracts or purchase orders are required to be made for every purchase beyond specified amounts. Those amounts vary according to the agency of the United Nations executing the contract (rule 110.22).

V. Do you find it useful to draw up as detailed contracts as possible, for instance by establishing standard models, in order to avoid disputes?

Contracts are generally on a fixed price basis with firm specifications describing the work to be done. The performance of the contractor is controlled, where applicable, by progress reports and results. The policy of the Organization is to avoid "open-ended" contracts with regard to time and costs. On the other hand, certain types of work call for payment on a time/rate basis. Time/rate contracts are appropriate when the work to be executed is of a measurable quality. For example, when a contract stipulates a certain amount of drilling work, payments may be based on a fixed rate per foot or per type of operation.

Contracts are amended only when there exist legitimate and agreed reasons for so doing, i.e. an extension of work to be executed, a change in the scope of the work or a change in emphasis resulting in a change in or extension of time or personnel. All amendments involving financial modifications must be submitted to the Committee on Contracts. This body endorses the general terms of the amendments and ensures that they are consistent with those set out in the original contract.

The final text of a contract may be subject to review and approval by the Office of Legal Affairs, the Office of the Controller and the substantive division. This is not always the case, and is generally not true of routinely recurring contracts. Contracts are generally signed by the Chief, Purchase and Transportation Service, on behalf of the United Nations. Copies of the contract are then forwarded to the contracting body. The contractor retains its copy (copies) and returns the others to the Organization.

The degree to which a detailed contract proves to be useful varies according to a number of conditions, including its nature and the purposes underlying it. Contracts may take a variety of forms, including purchase orders, letters of agreement and formal contracts. In principle, the "boiler plate" or standardized "general conditions" clauses of the contract are uniformly applied to contracts entered into by the United Nations with private parties. However, the Geneva Office of the United Nations has developed its own set of "general conditions" in contract making. In the case of contracts of a less important nature, or those in which certain provisions of the "general conditions" would be inapplicable, such conditions may be partially deleted or included in abbreviated form.

In March 1975, there was held a meeting of the Agency Contract Specialists Group of the Working Group on Administration and Finance Matters of the United Nations and its specialized agencies. That meeting was held in pursuance of a decision taken by the Group at its fifteenth session, in the course of which it considered a draft standard contract form prepared by the United Nations for use of all the specialized agencies.<sup>29</sup> At the meeting of the Group, the United Nations representative stated that, from a study of the comments received from the specialized agencies, it was obvious that, with the exception of "General conditions", a single standard form for use by all the agencies was difficult to design in view of the different cir-

VI. Would you consider that the elaboration of international substantive rules and uniform laws in the field of contract (for instance by means of international conventions) and the widest possible use of the result in international commercial relations could play a useful role in the emergence of an international legal system applicable to the contracts here under consideration?

Yes, this Organization would consider that such elaboration and use could play a useful role in the question under consideration.

#### PROCEDURES FOR SETTLING DISPUTES

VII. Do the Constitution, the internal rules of your organization or international conventions (headquarters agreements, etc.) make any provision as to procedures for settlement of, and the body competent to deal with, disputes arising from contracts concluded between your organization and private parties?

Section 29 of the General Convention states that:

"The United Nations shall make provision for appropriate modes of settlement of:

"(a) Disputes arising out of contracts or other disputes of a private law character to which the United Nations is a party;".

In order to provide a suitable means of settlement of any disputes of a private law character, the United Nations has regularly made provision in its contracts for recourse to arbitration.<sup>32</sup>

VIII. Do the contracts in question generally (or occasionally, in which case on what occasions?) contain a provision designating the body to which any disputes arising under such contracts are to be referred?

The contracts in question generally contain provisions designating arbitration as the manner in which any disputes are to be resolved.

In addition, special attention has been given to the preferred place of arbitration and its designation in such contracts. In 1964, the Office of Legal Affairs advised the Office of General Services regarding proposal that the United Nations standard bid form and United Nations contracts should specify that the place of arbitration would be New York. An extract from that opinion is reproduced below:

"There would naturally be practical advantages from our point of view should arbitrations be held in New York. On the other hand, there is the consideration that a requirement to this effect might dissuade parties either not resident or not represented in New York from bidding for United Nations contracts, and such a possibility should be avoided. To provide therefore in the standard bid form that arbitration should be in New York would not seem to us to be entirely advisable.

"On the other hand, when it is apparent at the time of contracting that a strong conflict of interest would exist between the United Nations and the contracting party in respect to the place of arbitration, it would be advisable to include agreement on the place of

cumstances. The United Nations representative therefore proposed that, rather than a rigid standard form contract, a model contract outline would be more suitable as a tool for the standardization of contracts within the United Nations system. A contracting officer might then use the elements of the model outline as appropriate to the special requirements of the contract. The components of the proposed model included: (a) a model contract cover page, and (b) a model schedule of contractual provisions. The production of such standard models for contracting within the United Nations system is an example of the current trend within the system towards establishing universally applied norms. This effort towards creating standard approaches to contract-making has been conducted within the United Nations for some time, and guidelines have been issued in an effort to maintain uniform approaches to contracts whenever possible. It

<sup>&</sup>lt;sup>10</sup> DP/WGAFM/WP.3/R.1, para. 3.

<sup>&</sup>lt;sup>31</sup> Two manuals are currently published by the Office of General Services of the United Nations Secretariat for the purpose of supplying guidelines in contract-making procedures, namely, Manual of Procedures for Purchase and Standards Section (New York, 1971) and Manual of Procedures for Contracts Section (New York, 1971). The latter manual is designed to supply models for contracts entered into by the United Nations as the executing agency for UNDP projects.

<sup>&</sup>lt;sup>32</sup> See Yearbook . . . 1967, vol. II, p. 296, document A/CN.4/L.118 and Add.1 and 2.

<sup>29</sup> DP/WGAFM/R.15.

arbitration in the disputes clauses. In such cases, should the United Nations consider it advisable that arbitration in the particular case should be in New York, it would be advisable to try to reach agreement on the inclusion of the words 'Any arbitration hereunder shall take place in New York unless otherwise agreed by the parties' in the arbitration clause of the contract."<sup>33</sup>

IX. If the contracts in question do contain a provision on machinery for the settlement of disputes, to what type of body is such competence ascribed: international or national arbitration bodies, an international administrative tribunal or a national court? Please provide examples of such provisions.

Competence for the resolution of disputes that may arise out of such contracts is ascribed to arbitration bodies. There have been various clauses on arbitration used by the United Nations in contracting. These provisions designate the structuring of the arbitration either under the rules of the American Arbitration Association or ICC, or they may designate the structuring of an ad hoc panel with final recourse, in case of dispute, to the President of the United Nations Administrative Tribunal. Three examples of these clauses are set out below:

- (a) "Any controversy or claim arising out of or in connection with the interpretation or enforcement of this Agreement or any breach thereof shall be settled by arbitration in New York City in accordance with the then obtaining rules of the American Arbitration Association. The parties hereto agree to be bound by any arbitration award rendered as a result of such arbitration as the final adjudication of any such controversy or claim."
- (b) "Any dispute arising out of the interpretation or application of the terms of this contract shall, unless it is settled by direct negotiations, be referred to arbitration in accordance with the rules then obtaining of ICC. The United Nations and the contractor agree to be bound by an arbitration award rendered in accordance with this section as the final adjudication of any such dispute."
- (c) "Any dispute arising out of or in connection with this contract shall, if attempts at settlement by negotiation have failed, be submitted to arbitration in New York by a single arbitrator agreed to by both parties. Should the parties be unable to agree on a single arbitrator within 30 days of the request for arbitration, then each party shall proceed to appoint one arbitrator and the two arbitrators thus appointed shall agree on a third. Failing such agreement, either party may request the appointment of a third arbitrator by the President of the United Nations Administrative Tribunal. The arbitrator shall rule on the costs which may be divided between the parties. The decision rendered in the arbitration shall constitute the final adjudication of the dispute."

The choice as to which of the three arbitration provisions is selected in any given case is based to some degree on the particular exigencies of each contract, on the convenience of the parties and their familiarity with the rules and procedures referred to, and on the prospective costs which might be involved in case of invocation of this procedure for the settlement of disputes.

X. Have the bodies in question had to meet frequently? Has their activity given rise to difficulties?

The United Nations has had recourse to arbitral proceedings in only a limited number of cases. The arbitral awards which have been made have been very largely based on the particular facts relating to the contract concerned and have not raised points of general legal interest regarding the status, privileges and immunities of the Organization. Very few cases regarding commercial contracts to which the United Nations was a party have come before municipal courts; in instances in which the United Nations was the plaintiff the most frequent issue was the capacity of the Organization to institute proceedings. In one case, it was held that a United Nations subsidiary organ bringing an action arising out of a contract was obliged to comply with venue requirements.

Such difficulties as have arisen regarding the contractual capacity of the Organization have usually followed a dispute over the execution of

a particular contract. On several occasions, it has been alleged by the other party that the United Nations lacked juridical personality and thus could not enforce its contractual rights before a local court. These arguments, in which the legal personality of the Organization was denied as part of a denial of its capacity to institute legal proceedings, do not appear to have been raised in any commercial dispute in which the United Nations took action as a plaintiff, although they have been presented in correspondence. In *United Nations* v. B. (1952), Tand in *UNRRA* v. *Daan* (1950), however, arguments denying the legal personality of the two organizations were presented by former staff members when action was brought to recover sums paid to them in error under their contracts of employment; these arguments were rejected by the courts. It may also be noted that, in a dispute which arose in 1952 with a private firm with which the United Nations had entered into a commercial contract, the firm sought to halt arbitration proceedings by means of a court order on the grounds that the Organization's immunity from suit and execution rendered its contracts unenforceable. In correspondence, the Office of Legal Affairs denied this argument, relying on precedents with respect to State immunities and the firm's acceptance of an arbitral procedure for the settlement of disputes. The request of a motion to stay arbitration was subsequently dropped by the firm concerned.39

In 1958, following a dispute as to the execution of a commercial contract, UNRWA sought to enter into arbitration with the other party. The other party having declined to appoint an arbitrator in accordance with the terms of the contract, UNRWA requested the President of the Court of Arbitration of ICC to appoint one. The latter appointed Professor Henri Batiffol of the Faculty of Law of the University of Paris. The section of Professor Batiffol's award dealing with the question of the competence of the arbitrator included the following passage which is of general interest regarding the capacity of an international organization, or of its subsidiary organs, to enter into contracts and to secure their enforcement:

"... Whereas UNRWA, an organ of the United Nations, derives from the treaties under which it was constituted, especially the Convention on the Privileges and Immunities of the United Nations of 13 February 1946, juridical personality and the capacity to contract; and whereas the stipulation of an arbitration clause, implied by such capacity, thus derives its legal basis from an instrument of public international law and is valid under that law without any need in that respect for reference to a national law, as would be the case for a contract between private parties, who to this day are subject to the authority of a State and hence to a national legal system, whether by reason of their nationality or domicile, the location of their property or their place of business or employment;

"Whereas, although certain legal systems do allow the signatory to an arbitration clause to petition an ordinary court of justice to supervise the arbitration proceedings or even, if the court deems it appropriate, to act as a substitute for the arbitrator, such substitution presupposes that the action is brought under a national system which has made provision for that possibility and regulated its consequences; and whereas, inasmuch as the present case does not involve an action brought under a national legal system but is governed by public international law, which has not made provision for such a possibility and does not, moreover, possess any organization of its own capable of regulating the consequences, the stipulated arbitration clause must be read according to its terms, which preclude recourse to the ordinary courts in case of disputes to which it refers, that being the only solution compatible with the immunity from jurisdiction of international agencies;

"Whereas the refusal of the respondent company to co-operate in the appointment of the arbitrator and in drawing up a settlement cannot bar the implementation of the arbitration clause; whereas, although national legal systems make varying assessments of the respective roles of damages and of performance in kind in the event of non-performance of a contract attributable to the debtor, all such systems recognize, in varying degrees, the right to require performance in kind wherever possible; and whereas, inasmuch as international law, on which the present arbitration clause is based,

<sup>33</sup> Ibid., p. 209.

<sup>34</sup> See however the award rendered by H. Batiffol (below).

<sup>35</sup> See United Nations Korean Reconstruction Agency v. Glass Production Methods (1956) (Federal Supplement, vol. 143, 1957, p. 248).

<sup>&</sup>lt;sup>36</sup> Yearbook . . . 1967, vol. II, p. 207, document A/CN.4/L.118 and Add.1 and 2.

<sup>&</sup>lt;sup>37</sup> Pasicrisie belge, 1953, part 3, p. 66, decision of the Brussels Civil Court of

<sup>31</sup> Annual Digest and Reports of Public International Law Cases, 1949 (London, 1955), vol. 16, p. 337.

<sup>&</sup>lt;sup>39</sup> Yearbook . . . 1967, vol. II, p. 208, document A/CN.4/L.118 and Add.1 and 2.

makes no provision on this subject, it is necessary to adhere to the general principle of the binding effect of contracts and to consider whether implementation of the arbitration clause in accordance with the terms thereof is possible despite the refusal of the respondent to co-operate;

"Whereas appointment of the arbitrator despite respondent's failure to act is possible, at least where the contract, as in the present case, provided for recourse to a third party for the purpose of such appointment in case of disagreement between the parties; whereas no distinction is to be made between disagreement concerning the person to be appointed and disagreement concerning the desirability of an appointment; and whereas the wording of article 12 ('should the parties not agree within 30 days as to the choice of the arbitrator, the appointment will be made by the President of the Court of Arbitration of the International Chamber of Commerce') covers both eventualities, in accordance with the genuine will of the parties, which was to submit to arbitration any dispute arising from the contract;

"Whereas the refusal of the respondent to co-operate in drawing up a settlement can be made good by the submission to arbitration of the draft settlement proposed to the respondent, whereupon the arbitrator will decide whether the proposed wording adequately and correctly sets out the subject of the dispute, having regard to the documents produced and particularly the correspondence between the parties; and whereas such replacement of the contract by a judgment, which is admissible, *inter alia*, in case of refusal to fulfil a promise of sale, is purely and simply the performance, upon a ruling by the judge, of the original contract, such ruling standing, in these circumstances, in lieu of a settlement;

"Whereas in the present case the complainant requested the President of the Court of Arbitration of the International Chamber of Commerce, in accordance with article 12 of the general conditions annexed to the contract, to appoint the arbitrator; whereas that request was acted upon; whereas, the complainant having submitted to the appointed arbitrator the draft settlement proposed by the complainant to the respondent, the arbitrator found, in the light of the documents produced, that the said draft adequately and correctly set out the subject of the dispute; whereas the arbitrator was therefore validly seized of the dispute and is competent to take cognizance of it."

The arbitrator found in favour of UNRWA as regards the merits of the dispute.

X1. Is the attribution of jurisdiction to be considered as implying a choice as to the applicable law? Or is the question left to the appreciation of the bodies in question? Are there any decisions by these bodies on the subject?

The attribution of jurisdiction for the settlement of disputes or contract claims to properly constituted arbitral bodies has not been considered as implying a choice as to the applicable law. The determination of the applicable law of the contract is left to the arbitrators. The number of disputes presented to arbitration for settlement is not great and few formal written opinions have been rendered. Reference may however be made to the opinion of Professor Batiffol (see sect. X above) and to the following cases: Balakhany (Chad) Limited v. Food and Agriculture Organization of the United Nations (1972);<sup>41</sup> Aerovías Panamá, S.A. v. United Nations (1965);<sup>42</sup> Lamarche v. Organisation des Nations Unies au Congo (1965).<sup>43</sup>

XII. (a) In case of an action by a private party against your organization on the basis of a contract, do you generally rely on such immunity from jurisdiction as the organization may enjoy or do you agree to waive such immunity?

The United Nations normally does not waive its immunity except in cases of third party liability covered by insurance. Rather than waive immunity, it submits to arbitration. However, as to its immunity, it may be noted that, as stated in section 2 of the General Convention:

"The United Nations, its property and assets, wherever located and by whomsoever held, shall enjoy immunity from every form

of legal process except in so far as in any particular case it has expressly waived its immunity."

Similar provisions are contained in the majority of other international agreements relating to the privileges and immunities of the United Nations. "Article I, section I, of the Agreement with Switzerland expresses the privilege as one derived from international law:

"The Swiss Federal Council recognizes the international personality and legal capacity of the United Nations. Consequently, according to the rules of international law, the Organization cannot be sued before the Swiss courts without its express consent."45

Immunity from legal process is not one of the privileges granted to the Organization under the Headquarters Agreement with the United States of America. Until the United States became a party to the General Convention, "6 the Organization's immunity from suit in that country had been based on national enactments. 47 Title I, section 2 (b), of the International Organizations Immunities Act provides:

"International organizations, their property and their assets, wherever located, and by whomsoever held, shall enjoy the same immunity from suit and every form of judicial process as is enjoyed by foreign Governments, except to the extent that such organizations may expressly waive their immunity for the purpose of any proceedings or by the terms of any contract."

One judicial decision may be noted relating to the immunity of the United Nations. In Curran v. City of New York (1947), 49 the plaintiff brought an action against the City of New York, the Secretary-General of the United Nations and others, to set aside grants of land and easements by the City to the United Nations for its headquarters site, exemption of the site from taxation and the allocation of funds by the City for the improvement of nearby streets. The Secretary-General moved to dismiss the action against him on grounds of his immunity from suit and legal process. The United States Attorney for the Eastern District of New York informed the court that the State Department recognized and certified the immunity of the United Nations and of the Secretary-General. The City of New York sought to dismiss the complaint on the ground that it failed to state a sufficient cause of action. The court held that the complaint should be dismissed and stated:

"The Department of State, the political branch of our Government, having without any reservation or qualification whatsoever, recognized and certified the immunity of the United Nations, and the defendant Lie to judicial process, there is no longer any question for independent determination by this court."

On a number of occasions, most notably in the case of actions involving United Nations immunities brought before United States courts, the United Nations has entered an *amicus curiae* brief. The majority of these cases, however, were in the early years of the

<sup>40</sup> Ibid., p. 208.

<sup>&</sup>lt;sup>41</sup> Arbitral award of 29 June 1972 (arbitrator, Barend van Marwijk Kooy) (United Nations, *Juridical Yearbook 1972* (Sales No. E.74.V.1), pp. 206-207).

<sup>42</sup> Award of the arbitration dated 14 January 1965, under the rules of the American Arbitration Association.

<sup>&</sup>lt;sup>43</sup> Award of the arbitration dated 6 August 1965, under the rules of ICC.

<sup>&</sup>quot;For the regional commissions, see section 7 of the Agreement concerning the Headquarters of ECLA (United Nations, Treaty Series, vol. 314, p. 49) and section 6 of the Agreement concerning the Headquarters of ECAFE (ibid., vol. 260, p. 35). In the case of the Agreement concerning the Headquarters of ECA (ibid., vol. 317, p. 101), no immunity from legal process is provided for the Commission itself, expressis verbis, although the Headquarters of the Commission are declared inviolable (sect. 2), its officials are granted immunity in respect of offical acts (sect. 11 (a)), and the Executive Secretary himself and his immediate assistants are granted diplomatic privileges and immunities (sect. 13). the Agreement and the General Convention are to be treated as complementary, however, in so far as their provisions relate to the same subject matter (sect. 17).

<sup>&</sup>quot; United Nations, Treaty Series, vol. 1, p. 163.

<sup>44</sup> The United States of America acceded to the General Convention on 29 April 1970.

<sup>&</sup>quot;Even prior to the accession of the United States to the General Convention, the United Nations had taken the position that its immunity from suit formed part of general international law, and thus part of the law of the United States, even in the absence of any legislation and, moreover, that the Organization's immunity from suit was derived from Articles 104 and 105 of the United Nations Charter, a treaty to which the United States was a party and which similarly formed part of the law of the land. United States courts have preferred to rely on national legislation, however, in upholding the Organization's immunity. See Yearbook . . . 1967, vol. 11, p. 223, footnote 49, document A/CN.4/L.118 and Add.1 and 2.

<sup>&</sup>quot;United Nations, Legislative Texts and Treaty Provisions concerning the Legal Status, Privileges and Immunities of International Organizations, vol. 1 (Sales No. 60.V.2), p. 129.

<sup>&</sup>lt;sup>49</sup> New York Supplement, 2nd series, 1948, vol. 1977, p. 206. The United Nations was not a defendant as such. It may be assumed, however, that the Secretary-General was named in his representative capacity.

<sup>39</sup> Yearbook . . . 1967, vol. II, p. 223, document A/CN.4/L.118 and Add. 1 and 2.

Organization's history. The established practice at the present time is to assert the immunity from suit of the United Nations in a written communication to the Ministry of Foreign Affairs of the State concerned. When time permits, this communication is sent through the Permanent Representative of the State concerned at United Nations Headquarters. In the written communication, the Ministry of Foreign Affairs is requested to take the necessary steps to inform the appropriate office of government (usually the Ministry of Justice or the Attorney-General's Office) to appear or otherwise move the court to dismiss the suit on the grounds of the Organization's immunity. When a summons or notification of appearance has been received, this is returned to the Ministry of Foreign Affairs. In cases brought by former staff members, the United Nations has usually referred in its note to the Ministry of Foreign Affairs to the fact that an alternative means of recourse exists for the staff member in the internal appellate machinery maintained by the Organization for its staff.5

In some instances local courts have taken decisions denying the immunity of the Organization or of its subsidiary organs despite the non-waiver of immunity.<sup>92</sup>

The case of Bergaveche v. United Nations Information Centre (1958)<sup>33</sup> concerned an employee of the United Nations Information Centre in Buenos Aires. In 1954, when his fixed-term contract was not renewed, he brought an action before the local labour court for termination indemnities. The United Nations Information Centre did not submit to the jurisdiction and requested the Ministry of Foreign Relations to notify the court of its immunity from suit. The court dismissed the action on the grounds that under the terms of Article 105 of the Charter and of the General Convention it lacked jurisdiction.

In response to a fresh submission by Mr. Bergaveche, another labour court gave a decision on 7 February 1956, in which it assumed jurisdiction by virtue of the fact that Argentina was not a party to the General Convention. Argentina acceded to the Convention on 31 August 1956 and in April 1957 the Ministerio Público advised the labour court that the action should be dismissed since the United Nations and its agencies enjoyed immunity from suit under the Convention and the Convention had become law in Argentina. The court therefore dismissed the action on 23 April 1957. On appeal, it was argued that, since the employment of Mr. Bergaveche had ended in 1954, the Statute adopted in 1956 could not be applied retroactively to his case, or, if retroactivity was intended, this could not affect rights under labour legislation already acquired. In its decision of 19 March 1958, the court held that the appellant's argument did not succeed since the statute concerned was a procedural one which was immediately applicable in the case of both pending and future proceedings.34

In an internal memorandum prepared by the Office of Legal Affairs in 1948 it was stated, with reference to section 2 of the General Convention, that, since the words "except in so far as in any particular case it shall have waived its immunity" must refer to the immediately preceding words ("shall enjoy immunity from every form of legal process"),

"it would appear that by this article permission is given to the United Nations to waive its immunity only in so far as legal process in any particular case is concerned, and such waiver cannot extend to any measure of execution".

This conclusion was said to be in accordance with a number of municipal decisions, notably those given by English and United States courts, in respect of the waiver of State immunities. The memorandum continued:

"According to the reports of the Preparatory Commission of the United Nations, article 2 of the General Convention was based on similar articles in the constitutions of international organizations. Some of their constitutional instruments, such as that of UNRRA, provide that the member Governments accord to the administration the facilities, privileges, exemptions and immunities which they accord to each other, including immunity from suit and legal process except with the consent of or so far as is provided for in any contract entered into by or on behalf of the administration.

"A similar provision is contained in article IX, section 3, of the Articles of Agreement of IMF" providing for waiver of immunity for the purposes of any proceedings or by the terms of any contract, thereby differentiating between the two forms of waiver. Apparently, it was not the intention of the Preparatory Commission or the General Assembly to extend waiver this far in so far as the United Nations was concerned, or such a provision would have been included, rather than just the words 'legal process'. In fact the words used in the original draft of this section were: 'The Organization, its property and its assets wherever located and by whomsoever held shall enjoy immunity from every form of judicial process except to the extent that it expressly waives its immunity for the purpose of any proceedings or by the terms of any contract.'

"This wording was changed by the Legal Committee of the Preparatory Commission to read in the more restrictive fashion in which it now stands. It must be concluded, therefore, that it was not the intention of the Preparatory Commission, or of the General Assembly, to extend the right of waiver to waiver in future by the terms of a contract.

"Since permission is given by the General Convention to the United Nations to waive its immunity in any particular case in so far as legal process is concerned, it is to be supposed that the authority to carry out such a waiver is placed with the Secretary-General, since the Secretary-General is responsible for the administration of the United Nations. It would not be possible to expect the Secretary-General to ask further authority from the General Assembly in each instance that legal process is to be served upon the United Nations; also the fact that the General Assembly found it necessary to write in a limitation upon the extent of any waiver, in so far as execution was concerned, would indicate that the General Assembly intended to transfer this authority to the Secretary-General, since if it were itself the waiving authority, there would be no necessity for making a limitation for its own right of waiver. This argument might be countered by stating that it is specifically provided in the General Convention that the Secretary-General may waive immunity in so far as officials and experts of the United Nations are concerned (sections 20, 23, 29). However, such a provision would be necessary in this instance since otherwise it might be supposed that the official or expert was entitled to waive his own immunity. In the case of the United Nations, the Secretary-General is 'the chief administrative officer of the Organization' and therefore such a clarification concerning his ability probably did not appear to be necessary to the Preparatory Commission or the General Assembly."

In practice, the Secretary-General has determined in all cases whether or not the immunity of the Organization should be waived.

XII. (b) Does your attitude regarding waiver of immunity depend on the jurisdiction seized of the case and the law which would be applied by it?

The only situation in which the Organization might normally waive its immunity would be one involving third party liability insurance.

One example of this situation would be a contract of insurance for motor vehicles. By resolution 23 (1), section E, of the General Assembly, the Secretary-General was instructed to ensure that drivers of the United Nations and all members of the staff who owned or drove motor cars should be properly insured against third party risk. In a 1949 memorandum, the Office of Legal Affairs stated:

"As it is really not feasible to take out insurance without permitting the insurance carrier the right to defend any suits which might be brought against the United Nations, the Secretary-General clearly has the power to waive immunity of the United Nations for the purpose of permitting such suits to be brought."

<sup>&</sup>lt;sup>51</sup> *Ibid.*, pp. 223-224.

<sup>&</sup>lt;sup>32</sup> Ibid., p. 224. A number of these decisions, for the most part rendered by courts of first instance, involved UNRWA. They are briefly described in the annual report of the Secretary-General on the activities of the Organization during the period 1 July 1953 to 30 June 1954 (Official Records of the General Assembly, Ninth Session, Supplement No. 1 (A/2663), pp. 106-107), and in Repertory of Practice of United Nations Organs, Supplement No. 2, vol. III (United Nations publication, Sales No. 63.V.7), pp. 518-519. Further information is contained in the annual reports of the Director of UNRWA for the years 1953/54 to 1957/58 (Official Records of the General Assembly, Ninth Session, Supplement No. 17 (A/2717), annex G, para. 11 (i); ibid., Tenth Session, Supplement No. 15 (A/2978), annex G, para. 19; ibid., Eleventh Session, Supplement No. 14 (A/3931), annex G, para. 26.

<sup>33</sup> Argentina, Cámara Nacional de Apelaciones del Trabajo de la Capital Federal, decision of 19 March 1958.

<sup>&</sup>lt;sup>34</sup> See Yearbook . . . 1967, vol. 1I, p. 224, document A/CN.4/L.118 and Add.1 and 2.

<sup>&</sup>quot; IMF, Articles of Agreement of the International Monetary Fund (Washington, D.C., 1978).

<sup>\*</sup> Yearbook . . . 1967, vol. 11, p. 225, document A/CN.4/L.118 and Add.1 and 2.

<sup>57</sup> Ibid., p. 226.

XII. (c) What is your position when the contract at issue does not provide for procedures for the settlement of disputes?

There are very few cases in which provision is not made for arbitration. If the other party prefers, the United Nations does not insert the arbitration clause but includes a provision that no waiver of immunity is intended. Current practice requires that all contracts provide for arbitration to be specified as the method of dispute settlement. However, were the situation to arise, the United Nations would not generally waive its immunity from jurisdiction but would seek resolution of the dispute through a forum other than national courts, most usually arbitration.

#### GENERAL QUESTION

XIII. Do you consider present practice satisfactory? In what way do you think it should be directed or developed?

In general, the present practice is deemed to be satisfactory.<sup>58</sup>

### Section 2. Capacity to acquire and dispose of immovable property

- (b) Acquisition and disposal of immovable property
- 5. The expansion of the United Nations Secretariat in New York has led to the rental under leasehold agreements of space in several buildings in the vicinity of the Headquarters district. Under the terms of a Supplemental Agreement of 9 February 1966 between the United States of America and the United Nations, as amended by an exchange of notes of 8 December 1966, and a Second Supplemental Agreement of 28 August 1969, as amended by an exchange of notes of 9 March and 25 May 1970, these premises were to be considered as included under the terms of the original Headquarters Agreement of 1947, with consequent privileges and immunities. A Third Supplemental Agreement was concluded on 10 December 1980.
- 6. With the consolidation of United Nations premises in two new buildings built by the United Nations Development Corporation and completed in 1983, a proposal for a fourth supplemental agreement was submitted to the United States in November 1984.
- 7. In addition to the expansion of facilities at Headquarters, in New York, a number of property transactions have occurred at the seats of the regional commissions or with respect to the establishment of major centres of United Nations activities, such as Nairobi and Vienna. In Vienna, the United Nations leases the Vienna International Centre from the Austrian Government for one Austrian schilling per annum pursuant to an Agreement between the United Nations and Austria signed on 19 January 1981." In Nairobi, the Government of

Kenya has provided the United Nations with a 100-acre site upon which the Organization has constructed its headquarters for UNEP and other offices. An Agreement between the Kenyan Government and the United Nations was signed on 11 December 1980 for the use of land by the United Nations.12 In Baghdad, the Government of Iraq has leased the premises for the permanent headquarters of ECWA to the United Nations for one dinar per annum. An Agreement to this effect between the United Nations and the Iraqi Government was concluded on 13 June 197913 and confirmed on 30 June 1983. Finally, a memorandum was signed on 2 November 1981 by the Rector of the United Nations University and the Governor of Tokyo regarding the donation of land to the University for its permanent headquarters site.

8. The United States Foreign Missions Act was enacted on 24 August 1982<sup>14</sup> and became effective on 1 October 1982. The Act is intended to regulate the operation in the United States of foreign missions and public international organizations and the official missions to such organizations, including the permissible scope of their activities and the location and size of their facilities. Application of the provisions of the Act to international organizations is subject to a determination by the Secretary of State. Except for the question of automobile liability insurance, no such determinations have thus far been made with respect to the United Nations.

### Section 4. Legal proceedings brought by and against the United Nations

- (c) Claims of a private law nature made against the United Nations and steps taken to avoid or mitigate such claims
- 9. The Standard Basic Assistance Agreement of UNDP<sup>15</sup> (which, in the case of countries that have signed the Agreement, replaces the Special Fund and technical assistance agreements and which is now in widespread use) provides, in article X, paragraph 2, that:

Assistance under this Agreement being provided for the benefit of the Government and people of ————, the Government shall bear all risks of operations arising under this Agreement. It shall be responsible for dealing with claims which may be brought by third parties against UNDP or an Executing Agency, their officials or other persons performing services on their behalf, and shall hold them harmless in respect of claims or liabilities arising from operations under this Agreement. The foregoing provision shall not apply where the Parties and the Executing Agency are agreed that a claim or liability arises from the gross negligence or wilful misconduct of the above-mentioned individuals.

#### Section 6. Treaty-making capacity

- (a) Treaty-making capacity of the United Nations
- 10. On 16 December 1982, by its resolution 37/112, the General Assembly decided that an international con-

<sup>&</sup>quot;The above reply was communicated to the Rapporteur of the Institute of International Law on the question of contracts concluded by international organizations with private parties, for the purposes of the preparation of his report to the fourth committee of the Institute. On 6 September 1977, the Institute adopted a resolution on the question, which is report on its Oslo session in 1977 (Annuaire de l'Institut de droit international, 1977, vol. 57, t. II, p. 264).

<sup>&</sup>lt;sup>7</sup> United Nations, Treaty Series, vol. 554, p. 309.

<sup>\*</sup> Ibid., vol. 581, p. 363.

<sup>9</sup> Ibid., vol. 687, p. 409.

<sup>&</sup>lt;sup>10</sup> United Nations, Juridical Yearbook 1980 (Sales No. E.83.V.1),

United Nations, Juridical Yearbook 1981 (Sales No. E.84.V.I), p. 11.

<sup>12</sup> United Nations, Treaty Series, vol. 962, p. 89.

<sup>13</sup> Ibid., vol. 1144, p. 213.

<sup>&</sup>lt;sup>14</sup> Public Laws 97-241 of 24 August 1982, *United States Statutes at Large, 1982* (1984), vol. 96.

<sup>15</sup> See DP/107.

vention should be concluded on the basis of the draft articles on the law of treaties between States and international organizations or between international organizations adopted by the International Law Commission

at its thirty-fourth session. On 13 December 1984, by its resolution 39/86, the General Assembly decided to convene a plenipotentiary conference in 1986 for this purpose.

#### CHAPTER II

### Privileges and immunities of the United Nations in relation to its property, funds and assets

### Section 7. Immunity of the United Nations from legal process

- (a) Recognition of the immunity of the United Nations from legal process
- The United States of America became a party to the Convention on the Privileges and Immunities of the United Nations on 29 April 1970. This accession strengthened the legal position of the United Nations with regard to immunity from legal process in the United States, which until that time had been based on domestic legislation and general international law derived, in particular, from Articles 104 and 105 of the United Nations Charter. This action was all the more significant for the Organization as it came at a time when the doctrine of sovereign immunity was undergoing a rapid evolution. A more restrictive doctrine was being developed in many countries, culminating in the enactment of national legislation such as the United States Foreign Sovereign Immunities Act of 1976.16 Although not directly applicable to international organizations, the changing doctrine of sovereign immunity and in particular the more restrictive approach to the commercial activity of foreign sovereigns will inevitably have an impact on the way national courts view the activities of international organizations. The United Nations, however, has continued to enjoy unrestricted immunity from legal process and has experienced no particular difficulties in this regard, unlike other organizations which do not enjoy the same legal protection under agreements in force.
- 12. In the *Menon* case (1973),<sup>17</sup> the estranged wife of a non-resident United Nations employee challenged the refusal of family court judges to order the United Nations to show cause why her husband's salary should not be sequestered to provide support for herself and her minor child. Her application was dismissed by a decision of the New York County Supreme Court, special term. The court declared that the law specifically exempted a sovereign from the jurisdiction of the United States courts, unless the sovereign consented to submit itself. The court further held that the United Nations "holds sovereign status and may extend that protection over its agents and employees . . " and that "the sovereign status of the United Nations, concerning its personnel and its financial agents, is beyond this or

- 13. In the Manderlier v. United Nations and Belgian State case (1966), 19 before the Brussels court of first instance, the plaintiff had instituted proceedings with a view to obtaining compensation from the United Nations or the Belgian Government, or from both jointly, for damage he claimed to have suffered "as the result of abuses committed by the United Nations troops in the Congo". The court dismissed the proceedings in so far as they pertained to the United Nations on the grounds that the Organization enjoyed immunity from every form of legal process under section 2 of the Convention of 13 February 1946 on the Privileges and Immunities of the United Nations. By its decision of 15 December 1969, the Brussels Court of Appeals pointed out that the immunity from legal process granted to the United Nations under the Convention on the Privileges and Immunities of the United Nations was in no way conditional upon the respect by the Organization of other obligations imposed by the same Convention, more particularly by article VIII, section 29, and that, although it was true that article 10 of the Universal Declaration of Human Rights stated that everyone was entitled to a hearing by a tribunal, the Declaration was not legally binding and could not alter the rule of positive law constituted by the principle of immunity from every form of legal process formulated in the Convention.20
- 14. With regard to the argument invoked by the plaintiff that Article 105 of the United Nations Charter limited the privileges of immunity to the minimum necessary to enable the United Nations to fulfil its purposes, the Brussels court of first instance replied that section 2 of the Convention conferred on the United Nations a general immunity from legal process and that, since the Convention and the Charter had equal status, the former, which was dated 13 February 1946, could not limit the scope of the latter, which dated from 26 June 1945. This judgment was upheld by the Brussels Appeals Court in its decision of 15 September 1969. The

the family court authority to challenge". The *Means* v. *Means* case (1969)<sup>18</sup> also concerns the immunity of a United Nations staff member from attachment of salary.

<sup>10</sup> Public Law 94-583 of 21 October 1976, United States Statutes at Large, 1976 (1978), vol. 90.

<sup>&</sup>lt;sup>17</sup> See United Nations, *Juridical Yearbook 1973* (Sales No. E.75.V.1), p. 198.

<sup>&</sup>lt;sup>18</sup> See United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), p. 243. In that case the court stated that the United Nations had sovereign immunity and therefore its "monies which it is in the process of transmitting to its own employees cannot be interfered with *en route* unless and to the extent the sovereign consents...".

<sup>&</sup>quot;See United Nations, Juridical Yearbook 1966 (Sales No. E.68.V.6), p. 283.

<sup>&</sup>lt;sup>20</sup> See United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), pp. 236-237.

Court added that, in acceding to the Convention of 13 February 1946, the signatories to the Charter had defined the necessary privileges and immunities and that the courts would be exceeding their authority if they were to arrogate to themselves the right of determining whether the immunities granted to the United Nations by that Convention were or were not necessary.<sup>21</sup>

## Section 8. Waiver of the immunity of the United Nations from legal process

- (a) Practice relating to waiver by the United Nations of its immunity from legal process
- 15. The accession by the United States of America to the Convention on the Privileges and Immunities of the United Nations has been referred to above (para. 11). The practice of the United Nations with regard to waiver of immunity has been maintained.
- 16. In 1969, the Office of Legal Affairs advised the Personnel Service that the Secretary-General's delegation of authority to the Administrator of UNDP and to the Executive Director of UNICEF could not be viewed as including authority to permit staff members to waive the privileges and immunities of the United Nations. The Office of Legal Affairs explained in its opinion that "the Secretary-General's authority with respect to the Organization's privileges and immunities (of which those applicable to officials are, of course, only a part) is not essentially a personnel matter and, without an express provision on this point, no such delegation could be inferred from the delegation of powers relating to administration of the Staff Regulations and Rules on appointment and selection of staff". The opinion concluded: "In our view, the authority has not been formally delegated and, moreover, it should not be."
- 17. In practice, the Secretary-General has determined in all cases whether or not the immunity of the Organization should be waived. In the instances where the Secretary-General judged it proper to waive the immunity of the United Nations from legal process, he was guided by a general sense of justice and equity.
- In 1949, a suit was commenced by a private individual against the United Nations for damages arising out of an automobile accident in New York in which a United Nations vehicle was involved. Under the terms of the insurance policy held by the Organization, the insurers were ready to defend the action in court. Before they could do so, however, it was necessary for the United Nations to waive its immunity. In an internal memorandum, the Office of Legal Affairs recommended that this should be done "for the purpose of allowing this particular suit to go to trial and that, as a matter of policy, it also be prepared to waive its immunity in any other case of a similar nature, subject to each such case being first reviewed by the Office of Legal Affairs to make sure that it has no complication such as might merit special treatment". The memorandum continued:

The question arises as to how this immunity may be waived. By its resolution 23 (1), section E, of 13 February 1946 [concerning third

party accident insurance for vehicles of the Organization and of staff members], the General Assembly instructs the Secretary-General "to ensure that the drivers of all official motor cars of the United Nations and all members of the staff who own or drive motor cars shall be properly insured against third party risk".

Under this resolution, the Secretary-General has clear authority to take whatever steps he may deem necessary to implement its terms. As it is really not feasible to take out insurance without permitting the insurance carrier the right to defend any suits which might be brought against the United Nations, the Secretary-General clearly has the power to waive the immunity of the United Nations for the purpose of permitting such suits to be brought.

This memorandum is only intended to deal with the waiver of the Organization's immunity in insurance cases. The question as to the circumstances in which the United Nations might be prepared to waive its immunity in other cases is complex, but as this question has no bearing on insurance cases, which are in a class by themselves, the necessity for discussing the waiver of immunity as a whole does not arise at this time.

In accordance with the conclusions reached in this memorandum, it is proposed that the Office of Legal Affairs should authorize the insurance carrier to defend this particular suit on behalf of the United Nations, thereby, of course, resulting in the United Nations waiving its immunity for this particular case and that the Office of Legal Affairs take similar action in all other insurance cases where it considers it would be within the spirit of the relevant General Assembly resolution so to do.<sup>22</sup>

- The policy of waiving immunity to permit the resolution of insurance claims against the United Nations has been continued. In Gibson v. United Nations liability insurance, where the question was raised whether the United Nations would exceptionally permit the Security Mutual Insurance to plead immunity in response to summons and complaint in action involving a claim for damages by a child who fell off a slide in the United Nations playground, the Office of Legal Affairs, following earlier practice, explained that, "in maintaining insurance for third party liability claims, the United Nations intended that the claims be defended and liability would be determined like liability of any other insured, including, if need be, by court adjudication. United Nations practice is . . . to authorize and request the insurance companies to enter a voluntary appearance on the United Nations behalf in defence of the action."
- 20. The foregoing policy, however, has not been applied with regard to risks in respect of operations arising under basic agreements between Governments and the United Nations on technical assistance. Under these agreements, the Government concerned assumes responsibility for the handling of claims arising from those programmes of assistance. In view of these arrangements, the Office of Legal Affairs advised the Insurance Unit in 1975 that, if the vehicle involved in an accident was part of a programme of this nature, "waiver of United Nations immunity from legal process in the United States would not be in accordance with the principles incorporated in the said international agreements or the practices observed by the United Nations pursuant to those agreements".

### (c) Interpretation of the phrase "any measure of execution"

21. In 1968, the Office of Legal Affairs had occasion to reaffirm its position with regard to measures of ex-

 $<sup>^{22}</sup>$  Yearbook . . . 1967, vol. II, p. 226, document A/CN.4/L.118 and Add.1 and 2.

ecution. In response to a request for advice from UNIDO regarding a hypothetical situation where a court of law, in execution of a judgment against a staff member, attempted to attach the salary of the staff member, the Office of Legal Affairs stated:

There is no doubt that such a proceeding with respect to UNIDO is null and void. In the first place, service of the court order upon UNIDO is a legal process from which UNIDO is immune. This is in virtue of section 2 of the Convention on the Privileges and Immunities of the United Nations and section 9 (a) of the UNIDO Headquarters Agreement. Secondly, the proceeding would be tantamount to a seizure of the assets of UNIDO from which UNIDO is exempt under section 3 of the Convention on the Privileges and Immunities of the United Nations. It should be noted that any such court order would be directed to UNIDO and the "salary" to be seized is, before it is actually paid to the staff member, a part of the assets of UNIDO. 14

#### Section 10. Immunity of United Nations property and assets from search and from any other form of interference

22. In 1974, the Secretary-General was advised by UNDP that a UNDP project account had been blocked by a judicial decision in a Member State as a result of a claim arising out of an accident involving a project vehicle in which a government employee assigned to the project had been injured. In an aide-mémoire, prepared by the Office of Legal Affairs and handed to the Permanent Representative of the State concerned, attention was drawn to section 3 of the Convention on the Privileges and Immunities of the United Nations. The UNDP account was unblocked shortly thereafter.

#### Section 11. United Nations name, emblem and flag

#### (a) United Nations name and emblem

23. The decision on the use of the United Nations name and emblem has generally proceeded on an ad hoc basis, although in general conformity with certain "rules of procedure" enunciated in a memorandum by the Office of Legal Affairs dated 5 April 1972. The memorandum suggested, inter alia, that United Nations associations with national coverage might use the United Nations emblem side by side with the national insignia of the country concerned, while those with local coverage might be permitted to use the emblem, although not next to the insignia of the local body. Organizations authorized by the Secretary-General to use the words "for the support of the United Nations" or "for the United Nations" in their titles would also be permitted to use the United Nations emblem, although not side by side with the insignia of the Organization, and only with the addition of the words "Our hope for mankind" beneath the emblem. In addition, authorization might be granted for use of the United Nations emblem with suitable words showing support for the United Nations in press notices by commercial bodies, when the notice was found to contain a genuine expression of support for the United Nations and not to imply endorsement of a particular product or firm.

24. The Office of Legal Affairs has continued to take whatever action may be necessary to protect the use of the United Nations name and emblem, particularly in regard to commercial exploitation. The legal basis of the protection of the flag and emblem was set out in a letter to the International Olympic Committee in 1973.<sup>25</sup> The Office of Legal Affairs also provided advice to the Office of Technical Co-operation on whether non-United Nations bodies established or maintained with the participation of the Organization might use the emblem of the United Nations on their stationery. The use of the emblem by such bodies was deemed inappropriate. The Office of Legal Affairs has also objected to the use of pictures or the name of the United Nations for commercial purposes.

#### (b) United Nations flag

25. In an internal memorandum from the Legal Counsel to the Chef de Cabinet of the Secretary-General dated 13 November 1969, it was agreed that the United Nations flag might be displayed on vessels of the Lake Nasser Development Centre Project. In the operative segment of his reasoning, the Legal Counsel stated:

As the Centre is a UNDP project and in view of the circumstances referred to by the Acting Project Manager, it seems to us that use of the flag may be authorized as an exceptional measure.

#### Section 14. Direct taxes

#### (a) Definition of direct taxes

#### (iii) Taxes on United Nations financial assets

26. On 11 July 1977, the Legal Counsel wrote in the following terms to the Permanent Representative of the United States to the United Nations:

I wish to call to your attention a matter of very serious concern both to the Secretary-General of the United Nations and to the United Nations Joint Staff Pension Fund and its participants. Those participants are in the employ of nearly all the intergovernmental organizations which make up the United Nations family of organizations. This matter relates to the recognition by the competent authorities of your country of the exemption of the United Nations under, *inter alia*, the Convention on the Privileges and Immunities of the United Nations, from the stock transfer tax levied in one of the States of your country in relevant transfers executed on behalf of all United Nations assets, in particular the United Nations Joint Staff Pension Fund.

It is our position that the exemption from taxation of the United Nations extends to all funds of the Organization, whatever their form or purpose. This position derives from Article 105 of the Charter of the United Nations and is supported by and is a logical interpretation of section 7 (a) of the Convention on the Privileges and Immunities of the United Nations, to which your country is a party. It is further supported by practice in other Member States where similar taxes are levied on non-United Nations institutions or individuals. While in 1967 the State legislature amended the stock transfer tax law to exempt international organizations from the provisions of that law, we have unfortunately not been able to obtain from the State authorities effective recognition of this exemption as applied to stock transfers executed by or on behalf of the United Nations Joint Staff Pension Fund. A considerable portion of the assets of the Fund have been and are being invested through the Stock Exchange, and the imposition of the stock transfer tax in regard to such transactions imposes an unwarranted and very heavy burden on the Fund, and thus on the contributors to the Fund, including States Members of the United Nations.

<sup>23</sup> See footnote 11 above.

<sup>&</sup>lt;sup>24</sup> United Nations, *Juridical Yearbook 1968* (Sales No. E.70.V.2), pp. 215-216.

<sup>&</sup>lt;sup>25</sup> See United Nations, *Juridical Yearbook 1973* (Sales No. E.75.V.1), pp. 136-137.

Consequently, the Organization attaches the greatest importance to the recognition of its rights in the present matter, rights which derive from international law and the treaty obligations of your country. It is the position of the United Nations that the practice of the State concerned must be conformed to the international obligations of your country and that the stock transfer tax law must be interpreted in this light.

Before considering recourse to the other remedies available to us under international law, we are seeking your assistance and that of the Department of State in intervening with the appropriate State tax authorities in an effort to seek an effective and full recognition of the exemption granted to the Organization under section 7 (a) of the Convention and under the law of the State concerned.<sup>26</sup>

- 27. In 1980, the Legal Counsel was advised by the United States Permanent Mission to the United Nations that, following a study of the matter by the New York State Commission on Taxation and Finance, a ruling had been made to the effect that the United Nations Joint Staff Pension Fund was exempt from the New York stock transfer tax.
- 28. A somewhat similar problem has concerned the imposition of withholding taxes on cash dividends paid on securities, including securities forming part of the assets of the United Nations Joint Staff Pension Fund. While the Organization has obtained exemption from the tax in some countries parties to the Convention on the Privileges and Immunities of the United Nations, certain other countries have refused to take such measures. In an internal memorandum dated 28 October 1969 to the Office of the Treasurer, the Office of Legal Affairs analysed the status of such a tax:
- . . . the tax in question is a tax on the dividends and, as such, would be a direct tax levied on income and assets of the owner of the securities. The fact that it is withheld at the source does not convert it into a tax against the corporation as such.

On the above basis there would appear to be no doubt that the United Nations would be entitled to exemption from the tax in question. The Government of Japan acceded without reservation to the Convention on the Privileges and Immunities of the United Nations on 18 April 1963. Section 7 (a) provides that the "United Nations, its assets, income, and other property shall be exempt from all direct taxes . . .".

- (iv) Taxes in respect of the occupation or construction of United Nations premises
- 29. In a memorandum addressed in 1971 to the Assistant Secretary-General, General Services, the Legal Counsel responded to the question whether the United Nations could be relieved of the increases in rent resulting from increases in real estate taxes payable by landlords of premises leased by the United Nations. He stated in part:

It is well established that immunity or exemption from tax on the part of a mission or an international organization does not affect the tax liability of the owner of premises leased to the mission or international organization.

The fact that part or all of the landlord's tax liability is passed on to the Government or international organization as rent does not change the character of the tax from a tax on the property payable by the owner to a tax on the immune organization; the tenant's obligation to pay the amount of the tax is, from the legal viewpoint, part of his rent obligation to the landlord.

So far, therefore, as concerns the increases in rent payable by virtue of the tax escalation clauses in the various United Nations leases to which your memorandum refers, I can see no basis on which the landlord could claim exemption from the proportionate part of the

<sup>26</sup> United Nations, *Juridical Yearbook 1977* (Sales No. E.79.V.1), pp. 238-239.

property tax represented by the United Nations leased premises; nor can I see any basis on which the United Nations could relieve itself of the obligation to pay the landlord the share of the tax increase specified in the lease as additional rent.

30. This position was subsequently confirmed in further correspondence between the Office of Legal Affairs and the Assistant Secretary-General, General Services. In a memorandum dated 2 December 1974, it was stated, regarding property taxes levied on the common gardens of 3-5 Sutton Place, that the exemption under section 7 of the Convention

applies only to taxes imposed directly on the United Nations or on United Nations property; . . . there is no basis in the Convention for asserting that the Organization should as a matter of right be reimbursed for increased costs resulting from tax payments collected by the State from a non-exempt owner who, pursuant to a private law agreement, passes the charge on to the United Nations.

- 31. Another longstanding question concerns the taxexempt status of UNITAR premises. On 27 October 1964, the United Nations, on behalf of UNITAR, purchased for \$450,000, from the Ninth Federal Savings and Loan Association of New York City, the building and ground lease with all the estate and rights of the Association. Since the assumption of the lease, the United Nations has paid the real property taxes on the building in conformity with article IV, paragraph I, of the lease, which provides that "the tenant shall also pay from time to time, when and as the same become due and payable, all such taxes, duties, assignments for local improvements, and all other governmental impositions extraordinary as well as ordinary".
- 32. Despite its assumption of this obligation under the lease, the United Nations has consistently claimed that it should be exempted from payment of these taxes. There exist a number of bases on which such an exemption might be claimed, including the Convention on the Privileges and Immunities of the United Nations and the Headquarters Agreement. However, the basis upon which such a claim has been pressed in the past is the statutory exemption provided by section 416 of the New York real property tax law.27 That law grants a tax exemption to "real property owned by the United Nations". All prior attempts to secure an exemption under the terms of the law have been to no avail. The argument used to frustrate the exemption in the past appears to have been based on the fact that the United Nationsinterest in the property was construed to be a leasehold. The reasoning, succinctly put, has been that a leasehold interest must be considered as less than a fee simple ownership in the premises and furthermore that a lease is not real property but rather personal property.

### (v) Hotel taxes

33. The exemption from direct taxes has also been applied to hotel charges paid directly by the United Nations. Such was the view of the Office of Legal Affairs in a memorandum of 9 January 1969 concerning taxes on hotel charges for room and board of United Nations personnel in Korea, where it was concluded that

The United Nations has a legal basis (in the Convention on the Privileges and Immunities of the United Nations, which is applicable to personnel in Korea under the UNICEF and UNDP agreements) for insisting on exemption from direct taxes on the Organization itself.

<sup>27</sup> Laws of the State of New York, 1958, vol. 11, p. 2134.

. . .

Where the taxes, however, are on hotel charges which are in turn paid by the staff members themselves, rather than the Organization, the Convention does not require exemption.

- 34. In another memorandum, dated 22 November 1976, where the question was whether the exemption from New York hotel occupancy tax was applicable to non-local short-term staff members whose remuneration consisted of monthly allowance and subsistence allowance, the Office of Legal Affairs wrote:
- 3. The exemption from hotel occupancy tax was accorded to the Organization by a letter from the Office of the Controller of the City of New York dated 17 July 1946. A subsequent letter, dated 6 March 1953, purported to clarify the scope of the exemption while at the same time imposing certain conditions upon the Organization, namely, rent must be paid directly by the United Nations or the employee must be reimbursed directly or on the basis of a per diem expense allowance.
- 4. The material question is whether the Organization derives the benefit to which it is entitled or whether on purely technical and administrative grounds it is deprived of that benefit. It does not seem reasonable to suppose that the applicability of the exemption accorded to the Organization should depend on a particular salary structure to be determined not by the Organization but by the authority granting the exemption.
- 6. . . . When it can be shown that the burden of the hotel tax falls on the Organization directly or indirectly, the exemption should be invoked regardless of the particular administrative arrangements in force.

### (b) Practice in respect of "charges for public utility services"

- 35. In a case which arose in 1968, the Office of Legal Affairs drew an important distinction between charges for municipal services billed according to real estate evaluation and not according to the services actually rendered, and concluded that the former constituted a direct tax. In a memorandum of 27 February 1968, the Office of Legal Affairs noted *inter alia* that,
- ... under the fifth clause of the proposed lease, the obligation to pay for waste removal and "any other service" falls on the tenant, i.e. UNDP. It would appear that waste removal and the other "services" are in fact services rendered by the municipality concerned. The Office of Legal Affairs has always held the view that, where services furished by municipalities are charged not according to the value of the services but according to property evaluation or other independent criteria, the payment thus made constitutes a tax. Under section 7 (a) of the Convention on the Privileges and Immunities of the United Nations, the Organization and its subsidiary organs, such as UNDP, are exempted from such taxes. In our opinion, the Resident Representative of UNDP should seek exemption from these charges if they are billed according to real estate evaluation and not according to the service actually rendered.<sup>23</sup>
- 36. The question has also arisen whether taxes on air travel characterized as user charges for specific services, and levied in accordance with a national legislation, would constitute a direct tax under section 7 (a) of the Convention. Here the Office of Legal Affairs, in a letter dated 20 June 1973 addressed to the permanent representative of the State concerned, 29 examined (see below) the nature of the tax and the definition of charges for public utility services, and concluded that such a tax on air travel would in fact come under section 7 (a):

### I. Convention on the Privileges and Immunities of the United Nations

- 7. The exemption is sought on the basis of the Convention on the Privileges and Immunities of the United Nations, which has been acceded to by your country. Section 7 (a) of the Convention provides:
  - "The United Nations, its assets, income and other property shall be:
  - "(a) exempt from all direct taxes; it is understood, however, that the United Nations will not claim exemption from taxes which are, in fact, no more than charges for public utility services."

It is submitted that the taxes for which exemption is sought are within the purview of the exemption provided in section 7 (a).

- 8. There can be no doubt that the charges in question constitute direct taxes. This appears clearly, *inter alia*, from the reports and proceedings quoted in the Treasury Counsel's opinion. The fact that they are characterized as "user taxes" does not remove them from the category of direct taxes; it merely describes their incidence.
- 9. The question, therefore, is whether these taxes are "no more than charges for public utility services". In this connection it should be noted that the term "public utility services" is much narrower than the term "public services" and has been interpreted most restrictively in the application of the Convention. The taxes here in question cannot, for a number of reasons, be considered as coming within the quoted phrase.
- 10. In the first place, the term "public utility services" has a restricted connotation applying to particular supplies or services rendered by a Government or by a corporation under government regulation, for which charges are made at a fixed rate according to the amount of supplies furnished or services rendered.
- 11. In the second place, the "charges", in accordance with established practice in applying the Convention, must be for services that can be specifically identified, described, itemized and calculated according to some predetermined unit. While "transportation" is an accepted public utility, it is the fare for such transportation (exclusive of taxes) that is a charge for that public utility service. For example, in the case of a government-owned bus company it is the fare, and not any tax added thereto for any purpose, such as the construction of highways, that would qualify as a charge for public utility services.
- 12. Moreover, the purpose of the tax clearly indicates that it is more than a charge for public utility services. It appears from the 1970 Act that the Trust Fund into which the taxes in question are to be paid is to be utilized primarily for the capital expenditures incurred in establishing and developing a national system of airports. The Act sets out, as the reason for its adoption,

"That the nation's airport and airway system is inadequate to meet the current and projected growth in aviation.

"That substantial expansion and improvement of the airport and airway system is required to meet the demands of interstate commerce, the postal service, and the national defence.

- ··. . . · '
- 13. The Act also specifies that the assets in the Trust Fund be available to meet expenditures incurred under title I of the Act, which provides for the preparation and implementation of a "national airport system plan for the development of public airports" and those "which are attributable to planning, research and development, construction, or operation and maintenance" of air traffic control, navigation and communication for the airways system.
- 14. The expenditures in question are clearly intended to be largely of a capital nature, and would, if the airways system were in private hands, be financed from capital funds raised by the sale of stocks or bonds, and not from current revenues. Since the system is government-owned, these capital expenditures would normally be borne by the general tax revenues either immediately or gradually, as bonds issued for the purpose are repaid.
- 15. While it is true that public utility charges normally do include an element for return on or repayment of capital, this is generally merely incidental to the portion of the charges designed to cover current expenditures for labour and materials. Moreover, the capital in question would be that already invested in the infrastructure used to provide the services for which the charges are rendered, rather than that required for the future expansion of the system, the cost of which must in the first instance be borne either by existing stockholders

<sup>&</sup>lt;sup>28</sup> United Nations, *Juridical Yearbook 1968* (Sales No. E.70.V.2), p. 184.

<sup>&</sup>lt;sup>29</sup> United Nations, *Juridical Yearbook 1973* (Sales No. E.75.V.1), pp. 133-136.

through retained earnings or by new investors in equity or debt securities.

- 16. While some of the revenues produced by the taxes here under consideration may be used for current operation and maintenance, and thus are of the type for which a utility could normally charge its customers, this is clearly not the principal destination of these taxes. It cannot therefore be said that these taxes are "no more" than public utility charges, as specified by section 7 (a) of the Convention for taxes as to which no exemption is to be claimed by the United Nations.
- 17. If such exemption were not claimed by and granted to the United Nations, then the Organization would, in effect be forced to use its resources to build up the aeronautical infrastructure of one of its members, that is, of a host State, in which of necessity a significant proportion of flights by its staff members originate or terminate.
- 18. It is not disputed that the United Nations through its staff members travelling on official business will benefit from the proposed national airport system, but that is not the criterion specified in section 7 (a) of the Convention. Staff members also benefit from police and fire protection, public health and sanitary measures, the work of the meteorological office and the countless other protective and supportive services of a modern government. These are financed by taxes paid by the nationals and residents of the country, except to the extent that certain persons are exempted from such contributions for various policy reasons, such as international civil servants whose taxation by national authorities would merely burden the coffers of the organization employing them. It appears to the United Nations Secretariat that the aeronautical facilities and charges here under consideration, which later would burden directly the Organization itself, fall within the category of services and taxes covered by the above principle.
- 19. The United Nations has therefore consistently taken the position that taxes that are not merely substitutes for charges for current services are covered by the general exemption granted by section 7 (a) of the Convention. This issue is discussed in a Secretariat study on relations between States and intergovernmental organizations, which is cited in the Treasury Counsel's opinion. The citation concerns a letter from the Legal Counsel of the United Nations, reading in pertinent part as follows:
  - "...l am sure it is not necessary to refer to the fact that the public utilities supervised by such governmental bodies in any of a large number of countries are principally gas and electricity, water and transport. For example, Quemner, *Dictionnaire juridique*, gives the following entry:
  - "'Public utilities, public services corporation—services publics concédés (transports, gaz, électricité, etc.).'"
  - "I think it is clear that the Convention had specifically in mind the payment by the United Nations of water and electricity charges on the grounds that the costs as billed are no more than the *quid pro quo* for commodities or services received. . .

"The authorities in international law generally seem to make a distinction as to whether the services rendered by a municipality or other public agency are special ones for which a special charge is made, with definite rates payable by the individual in his character as a consumer and not as a general taxpayer according to fixed principles of real property taxation."

This reasoning is equally applicable, mutatis mutandis, with respect to the taxes in question, as is the argument set forth in the memorandum of the Office of Legal Affairs of 27 February 1968.<sup>b</sup>

21. The position taken by the United Nations as to the interpretation of the Convention has generally been accepted by its Members, and indeed the effectiveness of a multilateral instrument of this type requires that the parties thereto accept such uniformity of interpretation. The summary of international practice in part V of the Treasury Counsel's opinion, which asserts that in a number of countries aviation-related taxes are imposed on international organizations, does not indicate either the nature of these taxes, which in some instances are purely public utility charges (such as those discussed in the note quoted in the previous paragraph), or whether any genuine taxes

are imposed on the United Nations by States parties to the Convention.

#### II. INTENT OF THE LEGISLATIVE AUTHORITIES

- 22. The Treasury Counsel's opinion demonstrates that the legislative authorities of your country intended that the taxes here in question be charged to all users of the civil aviation system, including international organizations. However, it is by no means clear that in so doing those authorities expressed an intention "to abrogate or restrict the application" of any relevant treaties; therefore, such a purpose should not be implied.
- 23. As pointed out in the opinion, your country has in the past granted and at present still grants exemptions from various excise taxes to diplomatic, consular and international personnel and organizations, on various bases and for different reasons: as a customary courtesy, on the basis of reciprocity, because of the requirements of customary international law, because of provisions of domestic legislation or administrative rulings, etc. While the legislative authorities evidently decided that these considerations should not limit the imposition of the taxes here in question on normally protected persons and organizations, there is no evidence that it was aware that in some instances exemptions are required by treaties or that it in any way wished to abrogate or limit such treaties. Indeed, it appears more than likely that the impact of that treaty on the legislation then under consideration was never explicitly taken into account.

#### III. CONCLUSION

24. On the basis of the foregoing considerations, the Secretariat of the United Nations trusts that the Government of your country will agree that the United Nations is, by virtue of section 7 (a) of the Convention on the Privileges and Immunities of the United Nations, entitled to exemption from the taxes imposed under the Act of 1970. Consequently it is hoped that the Government will find it possible to review and reverse the position taken by the Department of the Treasury concerning the liability of the United Nations in the payment of those taxes.

#### Section 15. Customs duties

### (b) Imposition of customs duties, prohibitions and restrictions

- 37. A decree adopted in a Member State provided that "foreign missions and international organizations, as abstract individuals, are not exempted from the rules of prohibition of import imposed on the products of foreign companies subject to the decisions of boycott, no matter whether these products are new or used, nor if the import is temporary or for transit passage". A vehicle consigned for UNTSO official use in the country which fell within the terms of the circular was refused import clearance and held by the customs officials. The Legal Counsel, in a letter of 9 August 1971 to the permanent representative of the country concerned, contended that the circular was contrary to the provision of section 3 of the Convention (immunity of United Nations property from requisition and confiscation) and also section 7 (b), exempting articles imported or exported by the United Nations for official use from prohibitions and restrictions. Moreover, such restrictions "would obviously deny to the United Nations the facility to obtain for the official purposes of UNTSO vehicles and equipment under the most favourable contractual terms".
- 38. Upon learning that the customs officials intended to sell the detained vehicle at auction, the Secretary-General, in an *aide-mémoire* dated 13 October 1971, reiterated the arguments put forward in the letter of the Legal Counsel, concluding that, because of the para-

 $<sup>^</sup>a$  Yearbook . . . . 1967, vol. 11, p. 247, document A/CN.4/L.118 and Add.1 and 2.

<sup>&</sup>lt;sup>b</sup> United Nations, Juridical Yearbook 1968 (Sales No. E.70.V.2), p. 184.

mount importance of the above-mentioned provisions on privileges and immunities, it would appear necessary, should a difference arise in their interpretation or application, to have recourse to the International Court of Justice pursuant to section 30 of the Convention. The dispute was ultimately satisfactorily resolved.

#### Section 16. Publications

- (a) Interpretation of the term "publications" and problems relating to the distribution of publications
- 39. A new press law in a Member State required that all periodical publications should record the name of the editor. In a memorandum of 16 January 1970 to the External Relations Division of the Office of Public Information, the Office of Legal Affairs stated:

The purpose of the provision referred to above of the press law in question is obviously to identify the author of any periodical publication so as to hold him responsible under the law of the Member State concerned. In the distribution of United Nations publications in that State, the Director of the United Nations Information Centre would be performing a United Nations function in his capacity as a United Nations official. He cannot be held accountable to the Government concerned or, for that matter, to any other authority external to the United Nations, in virtue of Article 105 of the United Nations Charter and section 18 (a) of the Convention on the Privileges and Immunities of the United Nations. The said provision of the law in question obviously has no application with respect to United Nations publications, including those issued by the Information Centre.

Accordingly, the Director of the Centre should take the necessary steps to request recognition of the exemption from the application of the law in question.<sup>30</sup>

- 40. The question of censorship of United Nations films under the censorship laws of a Member State was dealt with by the Office of Legal Affairs in a memorandum addressed to the Office of Public Information on 7 January 1970.<sup>31</sup> In its memorandum, the Office of Legal Affairs stated *inter alia*:
- 2. The United Nations is not in a position to submit its films to censorship, since it would be contrary to the United Nations Charter and to the Convention on the Privileges and Immunities of the United Nations to which the Member State concerned acceded without reservations. The position of the United Nations in this regard derives, in general terms, from Article 105 of the Charter and, more specifically, from sections 3, 4 and 7 (c) of the Convention on the Privileges and Immunities of the United Nations. These sections of the Convention provide as follows:
  - "Section 3. The premises of the United Nations shall be inviolable. The property and assets of the United Nations, wherever located and by whomsoever held, shall be immune from search, requisition, confiscation, expropriation and any other form of interference, whether by executive, administrative, judicial or legislative action.
  - "Section 4. The archives of the United Nations, and in general all documents belonging to it or held by it, shall be inviolable wherever located.
    - "Section 7. The United Nations . . . shall be
  - "(c) exempt from customs duties and prohibitions and restrictions on imports and exports in respect of its publications."
- 3. As you will appreciate, a demand to censor United Nations films would constitute interference as prohibited in section 3 of the

- Convention. As regards section 4, United Nations films are part of United Nations documentation, and censorship therefore would be in violation of this section, which provides for inviolability of documentation "wherever located". United Nations films are also covered by the exemption under section 7 (c), since they are a part of United Nations publications.
- 4. Furthermore, if a Government were to demand, in particular, the right to censor United Nations material and if that demand were complied with, the question would arise of a contravention of Article 100 of the Charter, under which a Member State is required to refrain from influencing the Secretariat in the discharge of its responsibilities and the latter is prohibited from receiving instructions from any authority external to the Organization.
- 5. The concrete case described in your memorandum concerns United Nations films proposed for screening in commercial cinemas in the Member State concerned by the United Nations Information Center. The question was raised whether a distinction could be drawn between United Nations films intended "for screening in commercial cinemas" and films "shown at public or private group-screenings".
- 6. It is our opinion that no such distinction can be made in relation to the Convention on the Privileges and Immunities of the United Nations. The establishment of the Information Centre on the territory of the Member State concerned was, as is always the case, effected in accordance with resolutions of the General Assembly under which both Member States and the Secretary-General are to further the public information work of the United Nations as spelled out in General Assembly resolutions 13 (I) of 13 February 1946, 595 (VI) of 4 February 1952 and 1405 (XIV) of 1 December 1959.
- 7. In particular, resolution 595 (VI) approved the "Basic principles underlying the public information activities of the United Nations", as suggested by Sub-Committee 8 of the Fifth Committee on Public Information. Under paragraph 8 of the basic principles, it is anticipated that the Department of Public Information of the United Nations Secretariat should "promote and where necessary participate in the production and distribution of documentary films, film strips, posters and other graphic exhibits on the work of the United Nations". Concerning the mode of distribution, paragraph 10 of the annex to the basic principles states:
  - "Free distribution of materials is necessary in the public information activities of the United Nations. The Department should, however, as demands increase and whenever it is desirable and possible, actively encourage the sale of its materials. Where appropriate, it should seek to finance production by means of revenue-producing and self-liquidating projects."
- 8. It is thus a long-established principle that distribution of United Nations public information material may take place through commercial channels. It follows that there is no foundation for distinguishing between various forms of distribution as long as the activities are performed within the scope of the above-mentioned General Assembly resolutions.

#### (b) United Nations copyright and patents

- 41. In a memorandum of 19 September 1966 to the Bureau of Operations and Programming of UNDP, the Office of Legal Affairs discussed United Nations patent practice and policy:<sup>32</sup>
- 1. The practice of the United Nations with respect to work financed by it which is susceptible to patent or copyright is to retain for itself the proprietary rights in the work, including the right to take out any copyright or right in such work. Provisions reflecting this practice may be found in Staff Rules 112.7 and 212.61, in United Nations special service agreements and in other agreements relating to projects in which the United Nations is Executing Agency for UNDP (Special Fund).
- 2. The foregoing practice is a manifestation of a general policy aimed at the widest possible dissemination and use of work performed

<sup>&</sup>lt;sup>30</sup> United Nations, *Juridical Yearbook 1970* (Sales No. E.72.V.1), p. 167.

<sup>&</sup>lt;sup>31</sup> United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), pp. 205-206.

<sup>&</sup>lt;sup>a</sup> See Official Records of the General Assembly, Sixth Session, Annexes, agenda item 41, document A/C.5/L.172, annex.

<sup>&</sup>lt;sup>12</sup> United Nations, *Juridical Yearbook 1966* (Sales No. E.68.V.6), pp. 225-226.

under the auspices of or financed by the Organization and is thus directed not so much towards acquiring a source of revenue in the form of royalties from the use of patent rights but to ensuring the general availability of techniques developed by the Organization or under its aegis. In retaining the rights in question, the Organization prevents any single individual or entity from taking out a patent or copyright over a work and from acquiring exclusive rights to control its exploitation and use. The Organization itself does not normally take out patents or copyrights. It achieves its purpose by publication or disclosure of the work, which has the effect of yielding it into the public domain.

- 3. It has of course been recognized that cases may arise in which it is necessary or appropriate to grant to an outside entity or person the right to take out a patent or a copyright on work performed under the auspices of the United Nations, such as when the provision of a financial incentive to others is required to encourage the development or exploitation of a work.
- 4. The foregoing practice and policy would seem to be as valid for UNDP as for the United Nations, if not more so.
- 5. In this connection, it may be noted that patent rights are assets in the same way as other intangible assets and thus constitute property from the point of view of both the Organization and the Special Fund. There is no provision in the Financial Regulations of the Special Fund which relates specifically to such assets, but the Financial Rules of the United Nations contain provisions dealing with the disposal of property in general, e.g. United Nations Financial Rules 110.32 (c) and 110.33 (a) (ii). The Financial Regulations of the Special Fund (SF/2/Rev.1) stipulate (art. 22.2) that the appropriate provisions of the United Nations Rules should apply in regard to any matter not specifically governed by the Special Fund Regulations.

### Section 17. Excise duties and taxes on sales; important purchases

### (a) Excise duties and taxes on sales forming part of the price to be paid

42. With the introduction of VAT in a number of European countries and Israel, the definition of that tax became a matter of importance. As early as 1972, the Office of Legal Affairs, in examining the character of VAT, concluded that:

VAT can be regarded as a direct tax to the extent to which it is readily identifiable, i.e. not incorporated in the price but, for example, shown separately on the invoice and assessed against the purchases as offered to the manufacturer or seller.

It was noted, however, that it was difficult to persuade Governments to accept such an argument, since VAT "is commonly regarded as a more sophisticated form of sales tax, which indeed it often replaces. Since sales taxes and turnover taxes are in general dealt with under "the important purchases" provision, there is a tendency to argue that VAT should be dealt with under the "remission or return" arrangements for important purchases."

43. Following the decision that VAT was to be regarded as an indirect tax, the question arose as to what constituted an important purchase so as to qualify for a remission. Studies have shown that, in countries where VAT has been introduced, the United Nations and its agencies have been granted exemption on both goods and services or reimbursement with respect to all transactions above a low threshold price. For example, in the case of UNESCO and the United Nations Information Centre in Paris, the minimum value of an important purchase has been set at 250 francs; while in the case of IAEA and UNIDO, the minimum has been 20,000 Austrian schillings, although there were negotiations with the Austrian authorities to lower it to 1,000 schil-

lings. The details of exemptions or reimbursements vary in different agreements. The following exchange of notes between the United Kingdom and the United Nations concerning reimbursements for VAT on goods and services may be cited as an example.

NOTE NO. 1, addressed to the Secretary-General of the United Nations by the Secretary of State for Foreign and Commonwealth Affairs

> London 16 May 1974

Your Excellency,

I have the honour to refer to the Convention on the Privileges and Immunities of the United Nations adopted by the General Assembly on 13 February 1946 and to correspondence between the Government of the United Kingdom of Great Britain and Northern Ireland and the United Nations regarding the application in the United Kingdom of article II, section 8, of the Convention in view of alterations in the tax system of the United Kingdom.

I now have the honour to propose that section 8 should be interpreted and applied in the United Kingdom so as to accord the United Nations a refund of car tax and value added tax on the purchase of new motor cars of United Kingdom manufacture, and of value added tax paid on the supply of goods or services necessary for its official activities and which are supplied on a recurring basis or involve considerable quantities of goods or considerable expenditure.

If the foregoing proposals are acceptable to the United Nations, I have the honour to propose that this note, together with Your Excellency's reply in that sense, shall constitute an agreement between the Government of the United Kingdom of Great Britain and Northern Ireland and the United Nations which shall enter into force on the date on which the United Kingdom legislation giving effect to the agreement comes into operation, which date will be notified to the United Nations.

For the Secretary of State (Signed) J. N. O. Curle

NOTE No. 2, addressed by the Secretary-General of the United Nations to the Secretary of State for Foreign and Commonwealth Affairs

> London 14 June 1974

Sir,

I have the honour to refer to your note of 16 May 1974, which reads as follows:

#### [See note No. 1 above]

I have the honour to inform you that the foregoing proposals are acceptable to the United Nations, which therefore agree that your note and the present reply shall constitute an agreement between the United Nations and the Government of the United Kingdom which shall enter into force on the date on which the United Kingdom legislation giving effect to the agreement comes into operation.

For the Secretary-General (Signed) Michael Popovic

#### (b) Important purchases

- 44. A question which has arisen in this connection has been gasoline taxes forming part of the price to be paid. In an opinion of 26 February 1974, the Office of Legal Affairs wrote:
- ... It has been the consistent position of the Office of Legal Affairs that a petrol tax forming part of the price to be paid is to be considered as falling under the terms of section 8 of the Convention and that the question whether or not a rebate should be granted should be determined by reference to the importance, quantitatively or financially, of the purchase. In the case of petrol, which is a recurring purchase, the amounts involved would normally qualify as important.

The United Nations is furthermore normally exempted from excise duty on gasoline required for its operations in the territories of Member States.<sup>33</sup>

- 45. Similarly, in an earlier memorandum, dated 26 January 1972, the Office of Legal Affairs dealt with the question whether the United Nations might claim exemption from "production duties" levied on gasoline by a Member State and discussed in detail the nature of such "production taxes":
- 1. You have asked for our views on a statement by the authorities of a Member State that the granting to UNTSO of exemption from "production duties" on gasoline is not legally justified.
- 2. Section 7 of the Convention on the Privileges and Immunities of the United Nations provides that the United Nations shall be "exempt from all direct taxes; it is understood, however, that the United Nations will not claim exemption from taxes which are, in fact, no more than charges for public utility services".
- 3. With regard to the definition of the term "direct" taxes, the principle is that the Convention should be uniformly applied in all Member States and therefore the characterization given to that term by municipal law or municipal officials cannot be controlling if the nature and incidence of the tax affect the United Nations and increase the financial expenses of the Organization to the advantage of a Member State. The interpretation of the term "direct" in accordance with the stated principle is intended to achieve equality in the implementation of the Convention among Member States within the spirit and the provision of Article 105 of the Charter and to relieve the Organization from undue financial burdens.

- 4. It is foreseen, however, that the authorities of the Member State concerned may maintain that excise duties on gasoline are indirect taxes which form part of the price of sale and from which the Convention does not accord to the United Nations automatic exemption. Even assuming for the purpose of argument that excise duties on gasoline constitute an indirect tax, the Organization is entitled to request that the Government make administrative arrangements for the remission or return of the amount of the excise duty under section 8 of the Convention, which provides:
  - "While the United Nations will not, as a general rule, claim exemption from excise duties and from taxes on the sale of movable and immovable property which form part of the price to be paid, nevertheless when the United Nations is making important purchases for official use of property on which such duties and taxes have been charged or are chargeable, Members will, whenever possible, make appropriate administrative arrangements for the remission or return of the amount of duty or tax."
- 5. Where the United Nations purchases goods or commodities on a recurring basis in the territory of a Member State, such purchases constitute "important" purchases on which the United Nations is entitled to request the remission or return of the amount of duties. Particularly in the case of purchases of gasoline, the amount of duty and the proportion that amount bears to the total purchase price is sufficient to consider the purchases as "important" and the tax as an undue burden upon the Organization. Moreover, whether characterized as "direct" or "indirect", all taxes which are important enough to make their remission or return administratively possible fall within the provisions of Article 105 of the Charter, which clearly contemplates the exemption of the United Nations from the financial burden of taxation.
- 6. It may be mentioned, incidentally, that the United Nations is normally exempted from excise duties on gasoline required for its operations in the territories of Member States.<sup>34</sup>

### CHAPTER III

## Privileges and immunities of the United Nations in respect of communication facilities

## Section 18. Treatment equal to that accorded to Governments in respect of mails, telegrams and other communications

46. Following the adoption in 1966 of a Convention between the Latin American States, Canada and Spain, signed at Mexico on 16 July, 35 which granted special franking privileges to the correspondence of diplomatic missions of the members of the Postal Union of the Americas and Spain, the Secretary-General, in a letter of 24 August 1971 to the permanent representatives of the States concerned to the United Nations, claimed those privileges for the United Nations under section 9 of the Convention on the Privileges and Immunities of the United Nations.

<sup>&</sup>lt;sup>33</sup> United Nations, *Juridical Yearbook 1974* (Sales No. E.76.V.1), p. 147.

<sup>&</sup>lt;sup>34</sup> United Nations, *Juridical Yearbook 1972* (Sales No. E.74.V.1), p. 158.

<sup>&</sup>lt;sup>35</sup> Postal Union of the Americas and Spain, "Convention, Final Protocol and Regulations of Execution between the United States of America and other Governments", *Treaties and Other International Acts Series 6354* (Washington, D.C., 1969).

#### CHAPTER IV

### Privileges and immunities of officials of the United Nations

- Section 22. Categories of officials to which the provisions of articles V and VII of the Convention on the Privileges and Immunities of the United Nations apply
- 47. While the formal categories established in General Assembly resolution 76 (I) of 7 December 1946 have remained unchanged, the Secretary-General found it necessary in 1973 to draw the attention of Member States to instances where the General Assembly had appointed or participated in the appointment of members of subsidiary bodies and where he considered that it would be appropriate to apply the provisions of section 17 of article V ("Officials") of the Convention.
- 48. The Secretary-General proposed that such cases be determined according to two criteria: (a) the official in question must be engaged on a full-time or substantially full-time basis to the point where he is effectively precluded from accepting other employment; (b) the official must be a member of a body responsible directly to the General Assembly. On the basis of these criteria, the Secretary-General proposed that the inspectors serving the United Nations Joint Inspection Unit and the Chairman of the Administrative Committee on Administrative and Budgetary Questions be included within the purview of articles V and VII ("United Nations laissez-passer") of the Convention. The General Assembly endorsed that proposal in its resolution 3188 (XXVIII) of 18 December 1973.
- 49. It was recognized that that action would serve as a precedent in any similar cases that might arise in the future. Similar action has since been taken with respect to the Chairman and Vice-Chairman of the International Civil Service Commission, the President of the Third United Nations Conference on the Law of the Sea and the Co-ordinator of International Assistance for the Reconstruction of Viet Nam.
- 50. The provisions of the UNDP Standard Basic Assistance Agreement extend the protection of article V of the Convention on the Privileges and Immunities of the United Nations to "persons performing services on behalf" of UNDP, a category which includes operational experts, volunteers, consultants and juridical as well as natural persons and their employees.<sup>36</sup>
- 51. While the United Nations has generally enjoyed the understanding and co-operation of Member States, problems have arisen from time to time with regard to recognition of the status of locally recruited officials, and it has been necessary to reaffirm, clarify and restate the policy of the United Nations as established in resolution 76 (I), of the General Assembly.
- 52. A proposal by a Member State in 1973 that its nationals should not enjoy privileges and immunities on its territory was not agreed to by the Office of Legal Af-

fairs on the grounds that "it could not be considered to be in accord with the Convention on the Privileges and Immunities of the United Nations", to which the Member State was a party. In its memorandum addressed to the Technical Assistance Recruitment Service, the Office of Legal Affairs noted that the Convention provided in article V for privileges and immunities to be accorded to "officials of the United Nations" and that "it is required under the Convention, therefore, that nationals of the Member State concerned who are officials of the United Nations be accorded privileges and immunities in accordance with the Convention".

### Section 23. Immunity of officials in respect of official acts

#### (a) General

- 53. In 1980, concerned by reports alleging that the privileges and immunities of officials of the United Nations and the specialized agencies had been encroached upon or ignored, the General Assembly requested the Secretary-General to submit to it a report on such cases. A report entitled "Respect for the privileges and immunities of officials of the United Nations and the specialized agencies" is now submitted annually by the Secretary-General to the General Assembly. The report, which is prepared by the Office of Legal Affairs, is introduced in the Fifth Committee by the Legal Counsel.
- 54. In connection with the submission of that report by the Secretary-General to the thirty-sixth session of the General Assembly, in 1981,<sup>37</sup> the Legal Counsel made the following statement in which he outlined the general views of the Organization with regard to the question of immunity of international officials:

The first question concerns what I might call the character of the immunity of international officials and the nature of its violation. The law of international immunities, which is based principally on the Charter, the conventions on privileges and immunities and other instruments referred to in paragraph 3 of the Secretary-General's report (A/C.5/36/31), distinguishes between diplomatic and functional immunities. The very great majority of officials of the United Nations and specialized agencies are accorded functional rather than diplomatic immunities. This distinction is significant not only from the point of view of the scope and content of the immunity but also because of the fundamentally different character of the two types of immunity. While diplomatic immunity attaches to the person, the functional immunity of international officials is organizational rather than personal. This is made clear by the conventions on privileges and immunities: section 20 of the Convention on the Privileges and Immunities of the United Nations provides that: "Privileges and immunities are granted to officials in the interests of the United Nations and not for the personal benefit of the individuals themselves". An identical provision is contained in the Convention on the Privileges and Immunities of the Specialized Agencies.

This distinction is important to an understanding of the nature of the violation of immunities reported by the Secretary-General in his annual report. It is essential to understand that the various cases referred to in the report involve a breach of the organizations' rights.

<sup>36</sup> See DP/107, annex I, art. IX.

<sup>37</sup> A/C.5/36/31.

Thus, for example, if we may refer to violations involving immunity from legal process, the type of case most frequently cited in the report of the Secretary-General, the substance of the Secretary-General's protest in such cases is not that a particular staff member was subjected to legal process but that the Secretary-General was prevented from exercising his right under the international instruments in force to determine independently whether or not an official act was involved. The position of the Secretary-General in this regard is set out in paragraphs 7-9 of the report. Where a determination is made that no offficial act is involved, the Secretary-General has, by the terms of the Convention on Privileges and Immunities, both the right and the duty to waive the immunity of any official.

As the Secretary-General has stated in his report, Member States have on the whole respected and complied with the Secretary-General's right of functional protection, which was clearly enunciated by the International Court of Justice in its advisory opinion of 11 April 1949 in the Bernadotte case (Reparation for injuries suffered in the service of the United Nations, *I.C.J. Reports 1949*, p. 174) and which now forms part of generally accepted international law. It is not the intent of the provisions regarding immunity from legal process or the principle of functional protection to place officials above the law but to ensure, before any action is taken against them, that no official act is involved and that no interest of the Organization is prejudiced.

A second and related question concerns who is entitled to privileges and immunities. It has been suggested by some delegations that locally recruited staff members are not officials of the United Nations and the specialized agencies for the purposes of privileges and immunities and that as local recruits they are first and foremost nationals of that country and subject to its laws. On this point, I should like to clarify and reaffirm the meaning of the term "officials" as it is used in the conventions on privileges and immunities. Section 17 of the Convention on the Privileges and Immunities of the United Nations states that the Secretary-General shall specify the categories of officials to which articles V and VII of the Convention shall apply. The Convention on the Privileges and Immunities of the Specialized Agencies and the Agreement on the Privileges and Immunities of IAEA contain similar provisions. In 1946, the General Assembly adopted resolution 76 (1), in which it approved the granting of the privileges and immunities referred to in articles V and VII of the Convention on the Privileges and Immunities of the United Nations to "all members of the staff of the United Nations, with the exception of those who are recruited locally and are assigned to hourly rates". The specialized agencies and IAEA have taken similar action. Consequently, all staff members, regardless of rank, nationality or place of recruitment, whether professional or general service, are considered as officials of the organizations for the purposes of privileges and immunities, except those who are both locally recruited and employed at hourly rates. United Nations locally recruited staff such as clerks, secretaries and drivers are in nearly every case paid in accordance with established salary or wage scales and not at hourly rates, and they are therefore covered by the terms of General Assembly resolution 76 (I).

# 55. In a memorandum of 1968, the General Counsel of UNRWA provided the following rationale for section 18 (a) of the Convention on the Privileges and Immunities of the United Nations:

... The extreme importance of this provision lies in the fact that, when acting in their official capacity, the acts of the official are in effect the acts of the United Nations itself, and the nationality of the official is totally irrelevant. Without this immunity, officials would be liable to be sued or prosecuted for acts done in their official capacity; they would be liable to be forced to appear as witnesses in court to give evidence on official matters; they would be liable to arrest and interrogation by State authorities on matters arising out of their official duties. Removal of such protection would place officials in a situation where they could be subjected to external pressures and influence directly contrary to Article 100 of the Charter. . . . Admittedly, there can be borderline cases in which it may be disputed whether the act is "official" or "non-official" and, as the employer, the agency must reserve the right to make this decision."

# 56. The exclusive competence of the Secretary-General to decide what constitutes an "official" act was the subject of a letter from the Office of Legal Affairs to the

Permanent Representative of the United States following a decision rendered by the Criminal Court of the City of New York in the case of *People of the State of New York v. Mark S. Weiner* (1976).<sup>39</sup> In that case, a United Nations security officer appeared as a complainant on behalf of the United Nations in a matter relating to his official duties. The Office of Legal Affairs took issue with certain *obiter dicta* made by the judge:

First and foremost, in the view of the United Nations Secretariat, it is exclusively for the Secretary-General to determine the extent of the authority, duties and functions of United Nations officials. These matters cannot be determined by or be subject to scrutiny in national courts. It is clear that, if such court could overrule the Secretary-General's determination that an act was "official", a mass of conflicting decisions would be inevitable, given the many countries in which the Organization operates. In many cases it would be tantamount to a total denial of immunity.

Likewise, the Secretariat cannot accept that what is otherwise an "official act" can be determined by a local court to have ceased to have been such an act because of alleged excess of authority. This, again, would be tantamount to a total denial of immunity. It may be noted, in addition to what is said in the paragraphs that follow, that the Secretariat has its own disciplinary procedures in cases where an official has acted in excess of his authority, and also the power to waive the immunity, particularly where the course of justice would otherwise be impeded. The Secretariat realizes that cases of conflict may arise as to whether an act was "official" or whether an official had overstepped his authority, but the Convention on the Privileges and Immunities of the United Nations expressly provides procedures for waiver of immunity, or for the settlement of disputes by the International Court of Justice. These are the appropriate procedures for settlement, not the overruling of the Secretary-General's determinations by national courts.40

57. In a letter to the Legal Liaison Officer of UNIDO in 1977,<sup>41</sup> the Office of Legal Affairs drew a distinction between acts to be considered as service-related for the purposes of staff regulations and rules and acts performed by officials "in their official capacity" within the meaning of the Convention, in cases involving traffic violations or traffic accidents:

This is in reply to your letter of 25 November 1977 on the question of the status of staff members when travelling directly from their home to the Organization and vice versa. Your inquiry and this reply relate solely to the question of immunity from legal process in connection with traffic violations or traffic accidents involving staff members travelling directly between their homes and the Organization. This reply also assumes that the staff member does not have diplomatic immunities by virtue either of his rank or under the particular host country agreement.

As indicated in my letter of 29 September, travel between home and office is not in itself considered to be an official act within the meaning of section 18 (a) of the Convention on the Privileges and Immunities of the United Nations, which provides for immunity from legal process in respect of acts performed by officials "in their official capacity".

To avoid confusion stemming from the phrase "on duty", I would emphasize the difference between the basis for the immunity for official acts under the Convention and the basis for various entitlements under the Staff Regulations and Rules.

The immunity of an official from legal process in respect of acts performed in his official capacity (i.e. on behalf of the United Nations) must be distinguished from service-related benefits under the Staff Regulations and Rules, such as compensation for injuries at-

<sup>&</sup>lt;sup>38</sup> United Nations, *Juridical Yearbook 1968* (Sales No. E.70.V.2), p. 213.

<sup>39</sup> Reports of Cases decided in the Court of Appeals of the State of New York, 2d Series, 1976, vol. 378, p. 966.

<sup>&</sup>lt;sup>40</sup> United Nations, *Juridical Yearbook 1976* (Sales No. E.78.V.5), pp. 237-238.

<sup>&</sup>lt;sup>41</sup> United Nations, *Juridical Yearbook 1977* (Sales No. E.79.V.1), pp. 247-248.

tributable to United Nations service or travel entitlements for service-related trips, including home leave travel. An injury may be compensable as service-related under appendix D to the Staff Rules without having been incurred by the staff member acting in his official capacity; the fact that a staff member's travel expenses are paid by the United Nations does not render his journey or his actions on the journey "official actions". Driving is, of course, official action by United Nations chauffeurs and such staff members may engage the United Nations liability as well as their own, and hence they are covered by the United Nations automobile liability insurance. Their immunity (and that of the United Nations) is frequently waived for the purpose of litigating damages, but the practice with respect to their immunity from charges of traffic violation is highly flexible.

As far as the General Assembly is concerned, one of its very first actions in the field of privileges and immunities was directed towards the prevention of abuse of privileges and immunities in connection with traffic accidents. Resolution 22 (1) E instructed the Secretary-General to ensure that staff members be properly insured against third-party risks, an instruction which finds its implementation in Staff Rule 112.4.

The functional and non-personal nature of the privileges and immunities of United Nations officials is made clear by the language of the Convention on the Privileges and Immunities of the United Nations and Staff Regulation 1.8.<sup>a</sup> The Secretary-General's position with respect to suggestions of immunity has always been that he and he alone may decide what constitutes an official act, when to invoke immunity and when to waive immunity.

There is no precise definition of the expressions "official capacity", "official duties" or "official business". These are functional expressions and must be related to a particular context. Indeed, it is doubtful whether a definition would be desirable since it would not be in the interest of the Organization to be bound by a definition which may fail to take into account the many and varied activities of United Nations officials.

Finally, there are certain pragmatic considerations which must be taken into account. While Headquarters practice does not exclude invoking immunity in certain traffic cases, a reverse practice in which immunity is automatically raised would give rise to considerable difficulties with the police and in the courts, not to mention the political consequences at a time when the general public and legislative bodies are opposed to privileges and immunities.

The practical handling of this question at Headquarters has not given rise to any difficulties, probably because of the firm position taken by the Secretary-General from the very beginning. Staff members are expected to obey local laws and regulations and, as the Secretary-General stated in a 1949 press release: "If there is any infringement of any laws, traffic violations for example, a Secretariat member is in the same group—unless on official business—as the average citizen who may pass a red light . . . He just pays his fine, and many already have."

58. In 1978, in a letter to the Legal Liaison Officer of the United Nations Office in Geneva,<sup>42</sup> the Office of Legal Affairs restated the policy of the Organization with respect to testimony by United Nations officials in domestic courts as follows:

I refer to your letter of 7 February asking advice on how to handle a summons addressed to a United Nations official for the purpose of eliciting testimony about salaries, pension, career prospects, etc. of a staff member victim of an automobile accident which is the subject of a suit for damages. You particularly ask whether United Nations officials can take an oath in court consistently with their obligations under the Staff Regulations.

We have a longstanding United Nations policy with respect to requests for staff members to appear as witnesses in court proceedings, in cases in which the United Nations as such has no interest, to testify on matters within their knowledge as United Nations officials or to provide information contained in United Nations files. Our policy is based on the Secretary-General's duty under section 20 of the Convention on the Privileges and Immunities of the United Nations "to waive the immunity of any official in any case where, in his opinion, the immunity would impede the course of justice and can be waived without prejudice to the interests of the United Nations".

The United Nations authorizes officials to appear and to testify on specific matters within their official knowledge provided (1) that there is no reasonable effective alternative to such testimony for the orderly adjudication or prosecution of the case; and (2) that no significant United Nations interest would be adversely affected by the waiver. The authority to waive the immunity and to authorize the testimony has been delegated to the Legal Counsel.

Occasions for the authorization and waiver are limited to cases in which the subject matter within the official's knowledge may be made public without giving rise to any problem as regards, for example, privileged papers or controversial political issues. Most frequently, where testimony by officials is required for criminal cases where cross examination is anticipated, we have had prior consultation with the attorneys requesting the appearance concerning the area of questioning.

We have on frequent occasions received summonses or subpoenae in connection with matrimonial and personal injury cases where United Nations salary entitlements and allowances are relevant. Our usual practice is to reply stating that the United Nations is immune but that information may be provided in relation to specific questions on a voluntary basis. Frequently, letters or documentary material are sufficient. In some instances, Personnel officers have appeared in judicial or quasi-judicial proceedings to provide information on United Nations salaries and emoluments. In cases where the staff member is a party to the dispute and the opposing party needs information about his United Nations emoluments, we sometimes provide the information to the staff member and require him to transmit the material required in the court proceedings so as to relieve the United Nations of the need to waive. In other words, our effort is to provide the information other than by court appearance if possible.

When staff members are authorized to appear and to testify on a particular subject matter, they are implicitly authorized to take whatever oath or to make whatever affirmation is necessary for the testimony to be admissible. Given the conditions for the waiver and authorization, the oath to testify truthfully would not, in our view, give rise to a conflict with the staff member's obligations under the Staff Regulations.

## (c) Cases of detention or questioning of United Nations officials; testifying before public bodies

The arrest and detention of United Nations officials has been the subject of annual reports to the General Assembly by the Secretary-General since 1981. At the same time, the Secretary-General has instituted a number of administrative reforms in order to improve the response of the Organization to cases of arrest and detention. These reforms have been embodied in a circular by the Secretary-General entitled "Security, safety and independence of the International Civil Service"43 and in an administrtive instruction entitled "Reporting of arrest or detention of staff members and other agents of the United Nations and members of their families",44 both of which were issued on 10 December 1982. These documents outline the procedures to be followed in the event of arrest or detention and clarify the nature and scope of the privileges and immunities of officials in the light of the United Nations Charter, the Convention on the Privileges and Immunities of the United Nations and the Staff Regulations and Rules.

a Reading as follows:

<sup>&</sup>quot;The immunities and privileges attached to the United Nations by virtue of Article 105 of the United Nations Charter are conferred in the interests of the Organization. These privileges and immunities furnish no excuse to the staff members who enjoy them for non-performance of their private obligations or failure to observe laws and police regulations. In any case where these privileges and immunities arise, the staff member shall immediately report to the Secretary-General, with whom alone it rests to decide whether they shall be waived."

<sup>&</sup>lt;sup>42</sup> United Nations, Juridical Yearbook 1978 (Sales No. E.80.V.1), pp. 191-192.

<sup>43</sup> ST/SGB/198.

<sup>44</sup> ST/AI/299.

### Section 24. Exemption from taxation of salaries and emoluments

#### (b) Position in the United States of America

- 60. The accession by the United States to the Convention on the Privileges and Immunities of the United Nations on 29 April 1970 did not materially alter the position of principle adopted by the United States with regard to exemption from taxation of its nationals or permanent residents. The United States accession was accompanied by a reservation to the effect that:
- ... paragraph (b) of section 18 regarding immunity from taxation ... shall not apply with respect to United States nationals and aliens admitted for permanent residence.
- 61. In accepting this reservation, the Secretary-General was guided by the fact that the tax equalization system effectively placed all staff members in a position of equality and that in this way the principle underlying section 18 (b) of the Convention was preserved.
- 62. In a letter to the Permanent Mission of the United States to the United Nations in 1975, the Office of Legal Affairs explained how the United States reservation was a formality in the light of the tax equalization system:

In accordance with section 18 (b) of the Convention on the Privileges and Immunities of the United Nations, all members of the United Nations Secretariat stationed at Headquarters in New York, with the exception of those who are recruited locally and are assigned to hourly rates, are exempt from taxation on the salaries and emoluments paid to them by the United Nations. The only exception at Headquarters results from the special situation in which officials of the United Nations who are nationals or permanent residents of the United States of America find themselves. When acceding to the Convention on the Privileges and Immunities of the United Nations on 29 April 1970, the United States Government reserved its position with respect to section 18 (b) in the case of nationals and permanent residents of the United States. Those officials therefore continued to be subject to the tax levied by the United States authorities on the salaries and emoluments paid to them by the United Nations. In establishing the Tax Equalization Fund (resolutions 973 (X) and 1099 (XI)), the General Assembly did all that could be done in practice to remedy the inequality which would otherwise have existed between officials who are subject to taxation and those who are exempt, and between the United States and the other Member States. Under this arrangement, United Nations officials at all levels are subject to assessment by the Organization in lieu of payment of national taxes, the total amount of the assessment being credited to the Member States; taxes paid by nationals and permanent residents of the United States are refunded to them and the refunds are charged against the sums standing to the credit of the United States in the Tax Equalization Fund.

63. From time to time, the tax exemption of locally recruited officials is queried by national tax authorities. Following representations made by the United Nations, recognition is usually given to the provisions of section 18 (b) of the Convention. In the rare instances where such recognition is withheld, the United Nations has, where possible, applied the provisions of the Tax Equalization Fund to reimburse the staff member for any taxes paid.

#### (f) National taxation on non-exempt income

64. In recent years the question has arisen whether national tax authorities may take into account United

<sup>45</sup> United Nations, *Juridical Yearbook 1975* (Sales No. E.77.V.3), p. 192.

Nations salaries and emoluments when setting the tax rate on non-exempt income. The United Nations has not considered it legally correct for a State party to the Convention on the Privileges and Immunities of the United Nations to take into account United Nations salaries in establishing tax rates on non-exempt private income. In the Organization's opinion, the exemption provided for in section 18 (b) of the Convention precludes any tax assessment based directly or indirectly on the exempted income. That position was set forth in a memorandum dated 16 October 1969 from the Office of Legal Affairs to the Director of the Accounts Division, Office of the Controller:<sup>46</sup>

- 1. You raise the question whether a Member State party to the Convention on the Privileges and Immunities of the United Nations is entitled to enforce a law providing that United Nations emoluments of staff members are to be taken into account in establishing the rate of tax on their non-exempt private income. Our view is that a party to the Convention is not entitled to make use of United Nations emoluments for any tax purposes.
- 2. The same position has been taken by UNESCO. It may also be mentioned that, in a case decided by the Court of Justice of the European Communities in December 1960 [Court of Justice of the European Communities, Reports of Cases before the Court, Luxembourg, 1960, p. 559], the Court held that article 11 (b) of the Protocol on the Privileges and Immunities of the European Coal and Steel Community, a which mutatis mutandis is identical with section 18 (b) of the Convention on the Privileges and Immunities of the United Nations, prevented the Belgian Government from taking the official salary of an official of the Coal and Steel Community into account in setting the rate of tax on non-exempt income. It may be convenient to summarize the more important lines of reasoning in the correspondence and the judgment referred to above.
- 3. Literal meaning of the Convention. Section 18 (b) of the Convention on the Privileges and Immunities of the United Nations provides that officials of the United Nations "shall be exempt from taxation on the salaries and emoluments paid to them by the United Nations". If the rate of tax on non-exempt income is set by taking account of exempt income from the United Nations, then the exempt income is part of the legal basis for the tax. If that is the case, then there is "taxation on the [United Nations] salaries and emoluments", which is forbidden by the Convention. The Court of Justice of the European Communities held that the literal meaning of the same language in the Protocol on the Privileges and Immunities of the European Coal and Steel Community prevented the exempt income from being taken into account.
- 4. Purposes of the immunity: independence of the staff. The principal purpose of the immunities which staff members enjoy under the Convention on the Privileges and Immunities of the United Nations is to protect and ensure the independent exercise of their functions with the Organization (Article 105 of the Charter). Accordingly, their official salaries are intended to be wholly exempt from national jurisdiction; but if they are taken into account in setting the tax on other income, they must be reported on in national tax returns, there are various governmental controls and administrative steps which apply to them, and a means exists by which the independence of the staff may be impaired.
- 5. Purposes of the immunity: independence and efficiency of the Organization. The United Nations must have complete freedom to select the best possible staff. If, however, official salaries are to be taken into account in setting taxes on non-exempt income, there may be a serious deterrent to persons considering service with the United Nations. This is particularly true with short-term service, where United Nations compensation is often substantial, but will be far less attractive if it has the effect of putting earnings during the rest of the year into a much higher tax bracket.
- 6. Inequities among international officials. The Court of Justice of the European Communities found that there would be a serious in-

<sup>&</sup>lt;sup>a</sup> See United Nations, Treaty Series, vol. 261, p. 242.

<sup>&</sup>lt;sup>46</sup> United Nations, Juridical Yearbook 1969 (Sales No. E.71.V.4), pp. 226-228.

equity between two officials who had the same gross salaries from a Community and the same private income from outside sources, if the Government of one of them took the Community salary into account in setting tax rates and the other did not. It may be pointed out that the effect of the judgment of the Court of Justice of the European Communities was probably to free all the officials of all the Communities (European Economic Community, European Atomic Energy Community (Euratom), European Investment Bank as well as the Coal and Steel Community), in all the countries members of those Communities, from having their official salaries used as the basis of their private taxes. Some of those countries are the ones that have sought to take United Nations salaries into account in setting the rates on private incomes. It would be obviously unjust to United Nations officials if they-who are protected by exactly the same treaty language as officials of the Communities-suffered a tax disadvantage from which the latter were free.

- 7. Analogy to diplomatic immunities. The best analogy to the immunity of United Nations salaries is that of diplomatic salaries in the receiving State; full exemption is required, although for somewhat different reasons, in both cases, No State, as far as we are aware, has ever tried to take the salaries of diplomats into account in setting taxes on their non-official incomes, and some of the countries that have tried to do so with United Nations officials have clear statutory provisions preventing it in the case of diplomats.
- 8. False analogy to double taxation arrangements. The attempt to take exempt United Nations salary into account for tax purposes seems to have originated in misapplication of a device found in some double taxation agreements. But the situation under discussion, where there is on the one hand a complete exemption and on the other taxable income, is completely different from that dealt with in double taxation arrangements, where both States have the undoubted legal right to tax the full income at their usual rates but wish, for reasons of policy and fairness, to avoid doing so. United Nations salaries are exempt, and it is not a matter of option for Governments bound by the Convention to decide whether to tax them or not.
- 9. Conclusion. For the foregoing reasons we are of the opinion that it is not legally correct for a State party to the Convention on the Privileges and Immunities of the United Nations to take account of United Nations salaries in establishing tax rates on non-exempt private income. We also agree with you that such a State should not ask for nor be informed about United Nations salaries and, if a case arises which is not merely a low-echelon discussion between an individual staff member and subordinate officials but rather a dispute between the United Nations and a Member State, we could consider submitting the matter to the General Assembly with the object of securing a request for an advisory opinion from the International Court of Justice. If the General Assembly took such action, the advisory opinion would, under section 30 of the Convention, be binding.
- 65. A similar position has been taken with regard to the filing of an annual income tax return in respect of United Nations salaries and emoluments. In a *note verbale* of 9 January 1973 to the Permanent Representative of a Member State, the Secretary-General stated:
- ... in accordance with the principle of exemption, United Nations salary and emoluments are considered as nonexistent for income tax purposes. United Nations officials are in consequence not required to submit a return unless the income from non-United Nations sources is in excess of the specified amount, nor may United Nations income be taken into account in determining the rate of tax on any additional income. Thus, in the opinion of the Secretary-General, United Nations officials of the nationality of the State concerned would be obliged to submit an income tax return only in so far as they may have other income in excess of the specified amount referred to in the first paragraph above.

The note [of the Permanent Representative] states that a fine is payable when a national passport is extended or renewed and the holder did not file a return. Since, for the reasons explained, United Nations officials are not, in the view of the Secretary-General, under an obligation to file a return where their sole source of income is from the United Nations, and since their need for a passport is directly related to their United Nations employment, the Secretary-General would express the wish that the authorities concerned would take the necessary steps to waive this fine, at least in the case of officials whose

income from non-United Nations sources is below the specified amount.<sup>47</sup>

### Section 25. Immunity from national service obligations

- 66. Section 18 (c) of the Convention on the Privileges and Immunities of the United Nations, which provides that United Nations officials are immune from national service obligations, has not given rise to any difficulties, largely because appendix C to the United Nations Staff Rules makes detailed provision for cases in which the staff members concerned may perform military service with the consent of the Secretary-General. Five member States have made reservations or declarations regarding the application of section 18 (c) when acceding to the Convention.
- 67. In an internal memorandum of 24 December 1975, the Office of Legal Affairs gave an opinion regarding the applicable law relating to military service of a staff member who had requested special leave from the Organization to complete such service:
- 1. Under article V, section 18 (c), of the Convention on Privileges and Immunities of the United Nations, officials of the Organization are immune from national service obligations. The Member State of which the staff member concerned is a national has acceded to the Convention without declaration or reservation. The Member State in question would therefore be obligated to recognize the immunity of an official under the terms of article V, section 18 (c). The staff member has a contract with the Organization which qualifies him as an official under the terms of article V, section 17, of the Convention.
- 2. Under section (c) of appendix C of the Staff Rules, a staff member who has completed one year of satisfactory probationary service or who holds a permanent or regular appointment may, if called by the Government of a Member State for military service, be granted special leave without pay by the Organization for the duration of that service. This is true even though section (a) of appendix C recognizes that staff members who are nationals of those Member States having acceded to the Convention on the Privileges and Immunities of the United Nations are immune from such service. Section (l) of appendix C furthermore states that the Secretary-General may apply the provisions of that appendix where a staff member volunteers for military service or requests a waiver of his immunity under article V, section 18 (c) of the Convention.
- 3. The Secretary-General, therefore, has discretionary authority to grant special leave in the case of the staff member in question, even though the staff member is exempt from national service obligation. The staff member may not waive his own immunity. Such immunity may be waived only by the Secretary-General in conformity with article V, section 20, of the Convention.<sup>48</sup>
- 68. Section 18 (c) has been held by the Office of Legal Affairs not to be applicable to jury duty. Practice at Headquarters in New York is to give special leave with full pay for 10 days and annual leave or special leave with pay thereafter where jury duty is compulsory and cannot be excused on other grounds. In practice, the United States authorities have, where necessary, interceded on behalf of the Organization to obtain a waiver of jury duty.

<sup>&</sup>lt;sup>47</sup> United Nations, *Juridical Yearbook 1973* (Sales No. E.75.V.1), p. 168.

<sup>&</sup>lt;sup>48</sup> United Nations, *Juridical Yearbook 1975* (Sales No. E.77.V.3), pp. 190-191.

### Section 26. Immunity from immigration restrictions and alien registration

### (b) Practice in respect of the United States of America

69. Prior to the accession of the United States to the Convention, the legal basis of United States practice resided in the United Nations Charter, the Headquarters Agreement concluded between the United Nations and the United States of America, and the United States International Organizations Immunities Act. In reply to an inquiry in 1969, the Office of Legal Affairs provided the following information:

Article 105, paragraph 2, of the Charter of the United Nations provides: "Representatives of the Members of the United Nations and officials of the Organization shall . . . enjoy such privileges and immunities as are necessary for the independent exercise of their functions in connection with the Organization."

The General Assembly, in accordance with Article 105, paragraph 3, of the Charter, proposed to the Members of the United Nations the Convention on the Privileges and Immunities of the United Nations, which sets forth in detail the obligations of Members under Article 105, paragraph 2, of the Charter. Under Article V, section 18 (d), of the Convention, officials of the United Nations shall "be immune, together with their spouses and relatives dependent on them, from immigration restrictions and alien registration".

Apart from the Charter and the Convention, the Agreement between the United Nations and the United States on the Headquarters of the United Nations provides in article IV, section 11:

"The federal, State or local authorities of the United States shall not impose any impediments to transit to or from the headquarters district of: (1) representatives of Members or officials of the United Nations, or of specialized agencies as defined in Article 57, paragraph 2, of the Charter, or the families of such representatives or officials."

The requirement of reasonable evidence to establish that the persons claiming the rights granted by section 11 come within the classes described in that section is specifically envisaged in section 13 (c) of the Agreement.

From the point of view of the United Nations, the United States statutory provision for entry of officers and employees of the United Nations [United States, International Organizations Immunities Act, 22 USCA, section 288 (a); 8 USCA, section 1101 (a) (15) G (iv)] implements the United States obligations as a Member of the United Nations and as the host country for the United Nations Headquarters. The procedure followed by the United Nations for securing entry for family members of officials is as follows. The official himself completes a United Nations form entitled "Request for Visa". In making this request, the staff member accepts responsibility for keeping the United Nations Office of Personnel informed of members of his family residing in the United States. On the basis of this request, the United Nations itself requests (if considered proper) the issuance of a visa. 50

### Section 29. Importation of furniture and effects

70. The Organization and its officials have in general encountered few difficulties in regard to the implementation of section 18 (g) of the Convention. Questions have been raised from time to time regarding the meaning of the term "effects" and the entitlement to duty-free importation of officials who are assigned to their own country after having served in a third country.

After consultations with the authorities concerned, such questions have been resolved satisfactorily.

### Section 30. Diplomatic privileges and immunities of the Secretary-General and other senior officials

71. The most important issue which has arisen in recent years in regard to diplomatic privileges and immunities of senior officials is whether, under section 19 of the Convention, Member States are under an obligation to accord such privileges and immunities to their own nationals residing in their own countries. A number of States have taken the position that international law as codified in the 1961 Vienna Convention on Diplomatic Relations does not so oblige them, whereas the United Nations and the specialized agencies have taken the position that section 19 of the Convention on the Privileges and Immunities of the United Nations allows of no discrimination based on nationality.

## 72. In a letter addressed to the Permanent Representative of the United States of America in 1971, the Legal Counsel stated:

I am directed by the Secretary-General to bring to your personal attention an important question bearing upon the status of the highest officials of the United Nations under section 19 of the Convention on the Privileges and Immunities of the United Nations, 1946, to which the United States acceded on 29 April 1970. The said section 19 of the Convention reads as follows:

"Section 19. In addition to the immunities and privileges specified in section 18, the Secretary-General and all Assistant Secretaries-General shall be accorded in respect of themselves, their spouses and minor children, the privileges and immunities, exemptions and facilities accorded to diplomatic envoys, in accordance with international law."

By a letter dated 4 May 1971, Mr. Albert F. Bender, Counsellor, on instruction of the Department of State of the United States, informed me of the position of the Department with respect to the application of section 19 of the aforesaid Convention, in the following terms:

"The Department of State notes that section 19 of the Convention provides that certain United Nations officials shall be accorded the privileges and immunities, exemptions and facilities accorded to diplomatic envoys 'in accordance with international law'. On the basis of international practice, the Department of State has decided that United States nationals are not entitled by section 19 to tax or customs privileges or to immunity from civil or criminal process except with respect to official acts."

After careful consideration of the above-quoted position, we find ourselves unable to agree with the conclusion of the Department of State and, under instruction of the Secretary-General, I set forth the view of the Secretariat of the United Nations with the request that the Department of State reconsider its position in the matter.

It appears to us that the above-quoted interpretation of section 19 of the Convention on the Privileges and Immunities of the United Nations by the Department of State—in excluding United States nationals from the enjoyment of certain specified privileges and immunities otherwise available to non-United States nationals in similar status in the United Nations—is at variance with the plain meaning of the words of the section; is contrary to the intention of the General Assembly, which adopted the Convention on 13 February 1946, as may be seen from the preparatory work on the Convention; and is inconsistent with the principle of an international civil service based on the United Nations Charter in which there is no inequality by reason of nationality.

In the first place, by the plain meaning of the words used, section 19 of the Convention on the Privileges and Immunities of the United Nations accords the Secretary-General and all Assistant Secretaries-General in respect of themselves, their spouses and minor children, the privileges and immunities, exemptions and facilities accorded to

<sup>&</sup>lt;sup>49</sup> United Nations, Legislative Texts and Treaty Provisions concerning the Legal Status, Privileges and Immunities of International Organizations, vol. 1 (Sales No. 60.V.2), p. 128.

<sup>&</sup>lt;sup>50</sup> United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), p. 226.

diplomatic envoys, in accordance with international law. It contains no exception excluding nationals from the benefits envisaged in the section. All Assistant Secretaries-General, without exception, are granted the benefits. As regards the phrase in section 19 "in accordance with international law", it is meant to indicate the scope of "privileges and immunities, exemptions and facilities". The phrase obviously does not qualify the words "the Secretary-General and all Assistant Secretaries-General" so as to exclude some of them from "the privileges and immunities, exemptions and facilities accorded to diplomatic envoys", because international law prevailing at the time of the adoption of the Convention could not have regulated a category of persons which had not previously existed. From these considerations, one cannot but conclude that, by its plain meaning, section 19 of the Convention cannot be read as envisaging any exception excluding nationals from the benefits provided therein.

Furthermore, reference to the *travaux préparatoires* of the Convention shows that the intention of the General Assembly, in adopting the Convention at the first part of its first session, was not to exclude nationals from the benefits provided in section 19 of the Convention. This intention is clearly manifested by the fact that the General Assembly deliberately deleted from the draft convention on privileges and immunities submitted by the Preparatory Commission a clause providing for such an exclusion (but with respect to only one form of immunity). Article 6 of the draft convention read as follows:

#### "Article 6

"1. All officials of the Organization shall:

"(a) be immune from legal process with respect to acts performed by them in their official capacity;

٠...

"2. In addition to the Secretary-General, all Assistant Secretaries-General, their spouses and minor children shall be accorded the privileges and immunities, exemptions and facilities accorded to diplomatic envoys, their spouses and minor children in accordance with international law, but shall not be entitled to invoke immunity from legal process as regards matters not connected with their official duties, before the courts of the country of which they are nationals."

The final clause in the above-quoted paragraph 2 was clearly intended to exclude nationals, but only with respect to one form of immunity, from the benefits provided for the Secretary-General and all Assistant Secretaries-General, etc. in the paragraph. This exclusion clause was deleted by the General Assembly and the paragraph thus amended became section 19 of the Convention. I submit that the deletion of the exclusion clause related above is significant, in that it shows conclusively that the authors of the Convention intended that the benefits of section 19 should be enjoyed by all the persons therein referred to, without distinction as to nationality.

The interpretation of the intention of the General Assembly finds corroboration, a contrario, by reference to section 15 of the same Convention on the Privileges and Immunities of the United Nations. This section 15 effectively excludes any representative, as against the authorities of the State of which he is a national, of any Member State from all the privileges and immunities provided in the Convention for representatives of Members in article IV of the Convention. The section reads as follows:

"Section 15. The provisions of sections 11, 12 and 13 are not applicable as between a representative and the authorities of the State of which he is a national or of which he is or has been the representative."

(Sections 11, 12 and 13 provide for various privileges and immunities for representatives of Member States.)

Thus it will be seen that, where the General Assembly, at the time of the preparation of the Convention, intended to provide for an exclusion on account of nationality, it did so by inserting an express provision to that effect. And it may be noted that section 15 follows in substance paragraph 3 of article 5 of the draft convention on privileges and immunities submitted by the Preparatory Commission. This legislative history corroborates our view that no exclusion on the grounds of nationality was intended by the authors of the Convention in respect of the high officials of the United Nations

referred to in section 19 of the Convention, and that that section therefore admits of no interpretation that justifies any such exclusion.

In the third place, while I have shown, by the foregoing, that by its plain meaning section 19 of the Convention on the Privileges and Immunities of the United Nations admits of no interpretation that entails, among high officials of the United Nations, a distinction on the grounds of nationality, and that the travaux préparatoires of the Convention show that the authors of the Convention deliberately took action to remove such a distinction from the draft text, our objection to the position of the Department of State is, above all, motivated by a desire to uphold a principle that, we believe, is vital to the effective functioning of an international organization. This is the principle that the staff of the United Nations are "international officials responsible only to the Organization". Under Article 100 of the Charter: "Each Member of the United Nations undertakes to respect the exclusively international character of the responsibilities of the Secretary-General and the staff and not to seek to influence them in the discharge of their responsibilities." The same article provides that the staff of the United Nations "shall not seek or receive instructions from any Government or from any authority external to the Organization". And "they shall refrain from any action which might reflect on their position as international officials responsible only to the Organization". Thus it is this Charter concept that the staff of the United Nations are in the position of "international officials responsible only to the Organization" that precludes any distinction, or any discrimination, among the staff on the basis of nationality. Any distinction or discrimination not tolerated by the Charter itself runs counter to the Charter and we deem it our duty to strive for its rectification.

Accordingly, the Secretary-General wishes me to request that you may be good enough to convey our view as stated herein above to the competent authorities in Washington so that the same treatment provided for in section 19 of the Convention on the Privileges and Immunities of the United Nations may be accorded to all persons described therein, without distinction as to nationality. The number of persons involved is only a handful but the principle is of paramount importance to the Organization.

I should add, before concluding, that the Department of State has declared that it based its decision to discriminate against United States nationals "on the basis of international practice". This assertion is so obviously without foundation that I have thus far tended to disregard it. In point of fact, the practice of States Members of the United Nations is contrary to the position taken by the Department of State; no State to our knowledge has evinced an attitude with respect to section 19 of the Convention similar to that position; certainly no State on acceding to the Convention has made a reservation to that section.

# 73. The same matter was taken up in relation to United Kingdom legislation in a letter from the Legal Counsel to the Adviser for International Organizations Affairs of ILO in 1975 as follows:<sup>51</sup>

This is further to your letter of 28 January 1975 in which you refer to "the Specialized Agencies of the United Nations (Immunities and Privileges) Order 1974'''3 and "the United Nations and International Court of Justice (Immunities and Privileges) Order 1974''''3 of the United Kingdom (Statutory Instruments 1974 Nos. 1260 and 1261). You have mentioned article 15, paragraph 2, of the two orders, which deny the diplomatic privileges provided for high officials in section 21 of the Convention on the Privileges and Immunities of the Specialized Agencies and section 19 of the Convention on the Privileges and Immunities of the United Nations to "any person who is a citizen of the United Kingdom and Colonies or a permanent resident of the United Kingdom". You have asked for the matter to be considered at the forthcoming session of the Preparatory Committee of the Administrative Committee on Co-ordination.

In the first place, I would like to stress that the privileges granted by sections 19 and 20 of the Convention concerning the specialized agencies, like those under section 18 of the Convention concerning the United Nations, are and must be enjoyed by all officials, regardless of nationality. Sections 21 and 19 of the respective conventions,

<sup>&</sup>lt;sup>a</sup> United Nations, Report of the Preparatory Commission of the United Nations, PC/20, 23 December 1945, p. 73.

b Ibid.

<sup>&</sup>lt;sup>31</sup> United Nations, *Juridical Yearbook 1975*, (Sales No. E.77.V.3), pp. 184-186.

<sup>&</sup>lt;sup>52</sup> United Nations, *Juridical Yearbook 1974* (Sales No. E.76.V.1), p. 7.

<sup>53</sup> *Ibid.*, p. 11.

however, refer to "the privileges and immunities, exemptions and facilities accorded to diplomatic envoys, in accordance with international law", and two interpretations of that phrase seem to be possible.

In the ordinary diplomatic context, States are not required to extend full diplomatic privileges to persons who are their nationals or permanent residents, even if they have consented to receive such persons in a diplomatic capacity. Article 38 of the 1961 Vienna Convention on Diplomatic Relations provides in paragraph 1:

"Except in so far as additional privileges and immunities may be granted by the receiving State, a diplomatic agent who is a national of or permanently resident in that State shall enjoy only immunity from jurisdiction, and inviolability, in respect of official acts performed in the exercise of his functions."

Similar provisions are to be found in article 71 of the 1963 Vienna Convention on Consular Relations and article 40 of the 1969 Convention on Special Missions. Substantially the same wording is used in the draft articles on the representation of States in their relations with international organizations prepared by the International Law Commission in 1971<sup>34</sup> and now being considered by the plenipotentiaries at the United Nations Conference on the Representation of States in their Relations with International Organizations meeting in Vienna. Article 37, paragraph 1, of that draft reads as follows:

"Except in so far as additional privileges and immunities may be granted by the host State, the head of mission and any member of the diplomatic staff of the mission [to an international organization] who are nationals of or permanently resident in that State shall enjoy only immunity from jurisdiction and inviolability in respect of official acts performed in the exercise of their functions."

Article 68 makes the same provision in respect of delegations to organs and conferences, and article V of the annex to the draft (relating to observers) repeats it yet again.

Those articles have not yet been discussed by the current Vienna Conference, which will end only on 14 March 1975. If, however, provisions like those in the Commission's draft articles are adopted, it will appear that the States participating do not consider that international law requires a host State to accord full diplomatic privileges to diplomatic envoys accredited to an international organization who are nationals of or permanently resident in that State. If that view is taken, sections 21 and 19 of the respective conventions on privileges and immunities would tend to be interpreted in the same way in regard to high officials. As a practical matter, it does not seem that arguments to the effect that high officials of organizations ought to be treated more favourably than diplomatic envoys sent to those organizations would meet with much support.

On occasion in the past, the United Nations Secretariat, without success, has taken a position against discrimination on grounds of nationality in the application of section 19 of the Convention on the Privileges and Immunities of the United Nations. After the United States became a party to that Convention, the question arose as to the privileges to be accorded to high officials of United States nationality. In May 1971, the United States informed us that it would not extend diplomatic privileges to them. We replied, requesting the Department of State to change its position, and arguing:

- (i) that the plain meaning of "all Assistant Secretaries-General" was obvious; and that "in accordance with international law" referred only to the scope of the privileges to be accorded rather than to the persons entitled to them;
- (ii) that the original draft of the Convention prepared by the Preparatory Commission had contained a limitation that high officials could not invoke immunity as regards matters not connected with their official duties before courts of their country of nationality, but that this limitation had been rejected by the General Assembly at its first session:
- (iii) that section 15 of the Convention on the Privileges and Immunities of the United Nations expressly provided that immunities were not applicable as between a representative and the State of which he was a national or which he represented, while there was no such express limitation in the case of high officials; and

(iv) that the status of "international officials" provided in the Charter implied a prohibition of discrimination among them on grounds of nationality.

The United States, however, upon reconsideration maintained its position and did not accept our arguments. We have not pursued the matter further, nor has the Secretariat protested the new United Kingdom Order relating to the United Nations.

You have referred to the Italian instrument of accession to the Convention on the Privileges and Immunities of the Specialized Agencies, which was transmitted to us in 1952 but has not been registered owing to objections. That instrument contained two reservations, of which the second related to the immunities of high officials under section 21, but the first and more important related to the immunities of the organizations themselves under section 4. Both reservations were objected to, and thus, even if it comes to seem futile to insist on the diplomatic immunities of high officials in their own countries, no change of attitude is necessary in respect of the instrument of accession as a whole.

I shall of course be glad to provide any further information that the Preparatory Committee may desire.

74. The question has been the subject of discussion in the Administrative Committee on Co-ordination, which has maintained the position outlined in the two letters cited above (p. 159, para. 72). For their part, neither the United Kingdom nor the United States has accepted the United Nations position and the matter remains unresolved.

### Section 31. Waiver of the privileges and immunities of officials

75. In a memorandum to the Office of Personnel Services in 1969, the Office of Legal Affairs advised that the Secretary-General's delegation of authority in personnel matters to the Administrator of UNDP did not include authority to waive the privileges and immunities of a staff member, which was vested exclusively in the Secretary-General. Regarding the conditions under which a staff member might be permitted to waive immunity, the policy formulated and maintained by the Secretary-General, pursuant to expressions of intention and understanding by the General Assembly, was against permitting staff members in the professional category to do so for the purpose of acquiring permanent residence status in a Member State; permission has, however, been granted to staff members who were stateless, de facto or de jure, and to general service staff.

76. In a letter of 11 February 1976 addressed to the Permanent Representative of the United States, 55 the Legal Counsel registered the Organization's concern with regard to remarks made by a judge in the Criminal Court of the City of New York. The question at issue was the exclusive competence of the Secretary-General to determine whether in any given instance a staff member had performed an official act and whether immunity should be waived. The letter stated:

I have the honour to refer to a decision rendered in the Criminal Court of the City of New York, on 19 January 1976, in the case of People of the State of New York v. Mark S. Weiner (published 20 January 1976 under New York County, Criminal Court, Trial Term, Part 17.36 In this case a United Nations security officer is appearing on

<sup>&</sup>lt;sup>54</sup> For the text of the draft articles, see Official Records of the United Nations Conference on the Representation of States in their Relations with International Organizations, Vienna, 4 February-4 March 1975, vol. II (United Nations publication, Sales No. E.75.V.12), p. 5.

<sup>&</sup>quot; United Nations, Juridical Yearbook 1976 (Sales No. E.78.V.5), pp. 236-239.

<sup>36</sup> See p. 155 above, footnote 39.

behalf of the United Nations as complainant, in a matter relating to his official duties, and the judge's decision contains a number of remarks which bear upon the privileges and immunities of the United Nations and which give rise to the most serious concern on the part of the Organization. This concern compels me to bring the matter to your attention and to place on record the position of the Secretary-General on the major legal issues involved.

#### Facts of the case

Before turning to the legal issues, it is necessary to give a brief account of the facts surrounding the case.

On Friday, 14 November 1975, at approximately 0300 a.m., the defendant in the case in question sprayed red paint on the wall dividing the circular driveway to the Secretariat building at the entrance to the Headquarters Division at 43rd Street. He was immediately detained by United Nations security officers, who also called in police officers from the 17th precinct of New York City Police Department. The defendant was then arrested, charged with criminal mischief (a class A misdemeanour under section 145.00 of New York Penal Law) and he was taken to the 17th precinct station in the custody of the officers of the New York City Police Department.

As already indicated, one of the United Nations security officers who detained the defendant is the chief witness and complainant on behalf of the Secretariat. The security officer was therefore directed by his supervisors to appear voluntarily, as and when requested by the Court, and to testify as to his personal knowledge of facts and circumstances relevant to the complaint and the charge.

There have been four hearings in the case, all of which were held before the same judge. Responding to pleadings by counsel for the defendant, the court, at the hearing held on 25 November 1975, requested the Secretariat to submit a legal memorandum on the question of the court's jurisdiction over acts against United Nations property situated within the Headquarters district. On 9 December, I, as United Nations Legal Counsel, wrote to the judge stating the Secretariat's view on the jurisdictional issue' and, at the hearing held on 12 December 1975, the judge indicated that he did not intend to sustain the objections made against the court's jurisdiction.

At the hearing held on 12 December, counsel for the defendant raised objections to the admission of the testimony by the United Nations security officer, who was present, on the grounds of the security officer's immunity from jurisdiction for official acts. As a result of this objection, the court requested the Secretariat to submit a further legal memorandum on the extent of the immunity from jurisdiction possessed by the security officer in connection with his appearance as a witness for the prosecution in the criminal proceeding against the defendant. The judge ruled that, for the court to proceed with the case, the Secretariat should state in a memorandum its view on whether the security officer had acted in his official capacity and whether he—were he to appear as a witness—would be immune from contempt of court citations, perjury charges or "cross complaints".

Pursuant to this request, on 8 January 1976, the officer-in-charge of the Office of Legal Affairs wrote to the judge stating the Secretariat's position on the extent of the immunity from jurisdiction enjoyed by United Nations officials appearing voluntarily as witnesses in criminal proceedings.<sup>34</sup>

In his written ruling on 19 January 1976, referred to at the outset of this letter, the judge denied the motion by the defence to dismiss for lack of jurisdiction and ordered a hearing held on 9 February 1976.

At the hearing on 9 February, the District Attorney proposed adjournment of the case in contemplation of dismissal. However, this was refused by the defendant and his attorney, both of whom insisted on a full hearing. The judge fixed such a hearing for 27 February 1976, at 9.30 a.m.

#### Legal position of the Secretariat

The Secretariat has no comments on the actual decision of the judge to deny the motion to dismiss for lack of jurisdiction in his ruling of 19 January. Its concern, however, is raised by some of the reasoning advanced on the matter of the security officer's privileges and im-

munities. In effect, it would seem, the judge was arguing that it was in the last instance for him, and not for the Secretary-General, to determine whether the security officer was acting in an official capacity and, furthermore, whether the guard had exceeded his authority through the use of excessive force, such excess, in the judge's view, rendering inapplicable the guard's immunity for official act. While the judge's remarks are in the nature of *obiter dicta*, their circulation in published form, without the Secretariat's contrary views being on record, could have a most serious effect upon the position of United Nations officials in countries throughout the world.

First and foremost, in the view of the United Nations Secretariat, it is exclusively for the Secretary-General to determine the extent of the authority, duties and functions of United Nations officials. These matters cannot be determined by or be subject to scrutiny in national courts. It is clear that, if such courts could overrule the Secretary-General's determination that an act was "official", a mass of conflicting decisions would be inevitable, given the many countries in which the Organization operates. In many cases it would be tantamount to a total denial of immunity.

Likewise, the Secretariat cannot accept that what is otherwise an "official act" can be determined by a local court to have ceased to have been such an act because of alleged excess of authority. This again would be tantamount to a total denial of immunity. It may be noted, in addition to what is said in the paragraphs that follow, that the Secretariat has its own disciplinary procedures in cases where an official has acted in excess of his authority, and also the power to waive immunity, particularly where the course of justice would otherwise be impeded. The Secretariat realizes that cases of conflict may arise as to whether an act was "official" or whether an official had overstepped his authority, but the Convention on the Privileges and Immunities of the United Nations expressly provides procedures for waiver of immunity, or for the settlement of disputes by the International Court of Justice. These are the appropriate procedures for settlement, not the overruling of the Secretary-General's determinations by national courts.

In the present case, the Secretary-General at no point waived the immunity of the security officer concerned, under section 18 (a) of the Convention on the Privileges and Immunities of the United Nations and also section 288 d (b) of the United States International Organizations Immunities Act.<sup>a</sup> The authority granted in section 20 of the Convention to waive the immunity of any official is enjoyed exclusively by the Secretary-General, and waiver cannot be effected instead by the Court. That this is a reasonable understanding of the Convention is borne out not only by the specification in section 20 of the conditions under which the Secretary-General may waive, but also by the provisions in article VIII for the settlement of disputes regarding all differences arising out of the interpretation or application of the Convention. As already mentioned, the Convention foresees that disputes are not to be settled by the courts of a Member State party to the Convention, but that differences between the United Nations on the one hand and a Member State on the other hand are to be decided by an advisory opinion of the International Court of Justice. The fact that such a procedure is available conclusively demonstrates the weakness of the assumption by the judge that national courts may determine the extent of immunity from jurisdiction enjoyed by a United Nations official acting in his official capacity as directed by the Secretary-General.

I trust that the foregoing will serve to explain the very real concern which the Secretariat feels over the reasoning of the judge, and its need to place its absolute reservations to that reasoning on record. The Secretariat cannot accept an approach which would submit the official acts of its officials to the scrutiny of national courts throughout the world. To do so, as already pointed out, would be tantamount to

<sup>&</sup>lt;sup>57</sup> See United Nations, *Juridical Yearbook 1975* (Sales No. E.77.V.3), pp. 157-159.

<sup>&</sup>lt;sup>58</sup> See United Nations, *Juridical Yearbook 1976* (Sales No. E.78.V.5), pp. 234-236.

a The opinion of the judge is inaccurate and misleading in not referring to these sources of immunity, which were made plain in the Secretariat's letter to him of 8 January 1976. The judge instead refers in his opinion to Articles 104 and 105 of the United Nations Charter and the Headquarters Agreement of 1947. The Charter articles are worded only in the most general terms, which are subsequently spelt out in specific detail in the Convention on the Privileges and Immunities of the United Nations, and the Headquarters Agreement does not deal with the privileges and immunities of United Nations officials. The judge is further in error when he cites the decision in *United States ex relatione Casanova v. Fitzpatrick* (1963) (Federal Supplement, vol. 214, 1963, p. 425) as a precedent, as that case related to a member of a permanent mission and turned on the interpretation of section 15 of the Headquarters Agreement, not upon the Convention on the Privileges and Immunities of the United Nations which is here involved.

stripping officials of their immunity. The Organization frequently operates in areas of tension and conflict, in which immunity for official acts is essential if United Nations officials are to function at all.

Finally, I trust you will agree that it is crucial that testimony by United Nations security officers be admitted and accepted as competent by criminal courts in cases that involve the safety of United Nations personnel or property. The absolute need for such testimony, both by officials and by members of permanent missions in relation to complaints made by such missions, has been constantly stressed by

United States representatives in the Committee on Relations with the Host Country. The Secretariat, however, would be most reluctant to instruct its officials to testify if it is accepted that the particular court before which they are to appear may strip them of the proper immunities accorded to them by international and national law.

I very much hope that, in the light of the above, we may arrive at a mutual understanding on the procedures and issues to be taken into account when United Nations officials are called upon to testify as witnesses in courts in the United States.

#### CHAPTER V

### Privileges and immunities of experts on mission for the United Nations and of persons having official business with the United Nations

### Section 33. Persons falling within the category of experts on mission for the United Nations

- 77. The scope and meaning of the category of "experts on mission" in relation to the members of a treaty organ, as distinct from a subsidiary organ, was the subject of a memorandum from the Office of Legal Affairs to the Director of the Division of Human Rights dated 15 September 1969, as follows:<sup>59</sup>
- 1. I have received your memorandum inquiring about the status, privileges and immunities of the members of the Committee on the Elimination of Racial Discrimination and members of ad hoc conciliation commissions established under article 12 of the International Convention on the Elimination of All Forms of Racial Discrimination. In our opinion, members of the Committee and members of the conciliation commissions are to be considered experts on mission for the United Nations within the meaning of sections 22, 23 and 26 of the Convention on the Privileges and Immunities of the United Nations and section 11 of the Headquarters Agreement with the United States, and are entitled to the privileges, immunities and facilities therein laid down.
- 2. The International Convention on the Elimination of All Forms of Racial Discrimination, opened for signature on 7 March 1966, does not expressly provide for the status of the members of the Committee. Nevertheless the Convention gives indications from which that status can be inferred.
- 3. There is a group of organs which, although their establishment is provided for in a treaty, are so closely linked with the United Nations that they are considered organs of the Organization. These include the former Permanent Central Opium Board (established by an Agreement of 1925b but made a United Nations organ by General Assembly resolution 54 (I) of 19 November 1946 and the protocol of amendment annexed thereto), the former Drug Supervisory Body (established by a Convention of 1931c but made a United Nations organ by the same resolution and protocol), the International Bureau for Declarations of Death (established by the Convention on the Declaration of Death of Missing Persons,d adopted by a United Nations conference on 6 April 1950), the Appeals Committee established under the Protocol for Limiting and Regulating the Cultivation of the Poppy Plant, the Production of, International and Wholesale Trade in, and Use of Opiume (adopted by a United Nations conference on 23 June 1953), and the International Narcotics Control Board (estab-

lished under the Single Convention on Narcotic Drugs, f adopted by a United Nations conference on 30 March 1961). Other similar organs are provided for in United Nations conventions which have not yet entered into force. Except for the mode of their creation, these organs are in the same position as recognized subsidiary organs of the United Nations. The Committee established under the Convention on the Elimination of All Forms of Racial Discrimination falls in the same category.

- 4. That Convention, which in article 8 (para. 1) establishes the Committee, was adopted by the General Assembly in resolution 2106 (XX) of 21 December 1965. On the organs referred to in the preceding paragraph, only the Permanent Central Opium Board and the Drug Supervisory Body share with the Committee on the Elimination of Racial Discrimination the distinction of having been made United Nations organs by a treaty which is at the same time a decision of the General Assembly. In the other cases, it has been necessary for the Assembly to decide to undertake the functions conferred on the United Nations by treaties adopted at a conference, and thereby to confer the status of United Nations organs on the bodies in question. Where the treaty itself is also a decision of the Assembly, however, no such separate decision on assumption of functions and conferenct of status is required.
- 5. The mode of creation of the Committee, the nature of its functions, their similarity to those of subsidiary organs, and the continuing administrative and financial ties which bind it to the United Nations remove all doubt that it is a United Nations organ, and it is thus without significance that the Third Committee rejected a proposal of the name "United Nations Committee on Racial Discrimination".8 As none of the other organs referred to in paragraph 3 above has the words "United Nations" in its name, that decision is not a strong basis for argument.
- 6. The purpose of the Convention, and consequently of the Committee, is, according to the preamble, to advance certain principles of the United Nations Charter. One of the main functions of the Committee (art. 9) is to make annual reports to the General Assembly, and that function is like the typical activity of subsidiary organs. Another main function of the Committee is consideration of allegations by a party that another party is not giving effect to the provisions of the Convention (art. 11), and the Committee may also be given competence by a declaration of a party to consider claims of violation submitted by individuals or groups of individuals (art. 14). Under article 15 and General Assembly resolution 2106 B (XX), the Committee has functions relating to petitions from inhabitants of Trust and Non-Self-Governing Territories. These functions seem to be of a judicial or quasi-judicial character; that character, however, does not prevent the Committee from being a United Nations organ. The various narcotics bodies referred to in paragraph 3 above perform quasi-judicial functions, and the Appeals Committee established under the 1953 Opium Protocol is of a fully judicial nature. Functions of these types can also be performed by subsidiary organs; the International Court of Justice, in its advisory opinion of 13 July 1954 on the Effect of Awards of Compensation made by the United Nations Administrative Tribunal

<sup>&</sup>lt;sup>a</sup> United Nations, Treaty Series, vol. 660, p. 195.

b League of Nations, Treaty Series, vol. L1, p. 337.

c Ibid., vol. CXXXIX, p. 301.

<sup>&</sup>lt;sup>d</sup> United Nations, Treaty Series, vol. 119, p. 99.

e Ibid., vol. 456, p. 56.

<sup>&</sup>lt;sup>39</sup> United Nations, *Juridical Yearbook 1969* (Sales No. E.71.V.4), pp. 207-210.

f Ibid., vol. 520, p. 151.

<sup>&</sup>lt;sup>8</sup> Official Records of the General Assembly, Twentieth Session, Annexes, agenda item 58, document A/6181, paras. 104 (a) and 110 (a) (i).

(I.C.J. Reports 1954, p. 47) has recognized the legal capacity of the General Assembly to establish judicial bodies for the fulfilment of its purposes.

- 7. Under article 10, the secretariat of the Committee is provided by the Secretary-General of the United Nations, and the meetings of the Committee are normally held at United Nations Headquarters. These are important connections with the Organization, and they ensure that the bulk of the expenses of the Committee, which will be for servicing meetings and for the secretariat, will be borne by the regular budget of the United Nations. Article 8, paragraph 6, of the Convention provides that: "States Parties shall be responsible for the expenses of the members of the Committee while they are in performance of Committee duties." The travel and subsistence costs of members, however, are a minor fraction of the total expenses of the Committee, and the payment of part of the expenses of an organ by some means other than the regular budget of the United Nations does not prevent that body from being a United Nations organ. As regards the expenses of the Permanent Central Opium Board, the Drug Supervisory Body and the International Narcotics Control Board, there are special arrangements for the assessment of contributions from States not members of the United Nations which take part in activities concerning narcotic drugs. It may be added that in practice the members of the Committee will be paid their travel and subsistence costs from a suspense account alimented by the United Nations Working Capital Fund, as the contributions of the parties are not paid in advance of expenditure. Recognized subsidiary organs can also be financed by other means than the regular budget (e.g. UNIDO, UNRWA etc., which depend upon voluntary contributions, and UNCTAD, to which contributions are made by participating States which are not members of the United Nations). In view of all these facts, the rejection by the Third Committee of a proposal to have all the expenses of the Committee borne by the regular budget of the United Nationsh is not significant.
- 8. The General Assembly rejected a proposal that it should itself elect the members of the Committee' and provided in article 8 of the Convention that the members should be "elected by States Parties from among their nationals". This does not prevent the Committee from being a United Nations organ. Two members of the Drug Supervisory Body were appointed by WHO, the International Bureau for Declarations of Death is appointed by the Secretary-General, and the Appeals Committee under the Protocol of 1953 is appointed by the President of the International Court of Justice or the Secretary-General; thus the status of United Nations organs does not require any particular mode of election. The same is true of ordinary subsidiary organs. Thus, for example, under General Assembly resolution 1995 (XIX) of 30 December 1964, the Trade and Development Board is elected by the United Nations Conference on Trade and Development, and the membership of other subsidiary organs has been left to be decided by the President of the General Assembly (e.g. the Special Committee on Principles of International Law concerning Friendly Relations and Co-operation among States) or by the Secretary-General (e.g. the tribunals for Libya and Eritrea).
- 9. What has been said above concerning the Committee applies with equal force to ad hoc conciliation commissions established under article 12 of the Convention. Those commissions, like the Committee itself, are part of the machinery for the execution of the Convention and for the settlement of disputes about its application and interpretation; and the Convention aims at applying principles of the Charter. The secretariat of the Committee, provided by the Secretary-General, also serves commissions (art. 12, para. 5), and their meetings "shall normally be held at United Nations Headquarters ..." (art. 12, para. 4), with the result that the bulk of the expenses of commissions will be borne by the United Nations. The fact that commissions have judicial or quasi-judicial functions, that members are appointed by the Chairman of the Committee, and that the expenses of their members are to be shared by the parties to the dispute does not prevent them from being United Nations organs.
- 10. Members of the Committee and members of commissions serve "in their personal capacity" (art. 8, para. 1 and art. 12, para. 2), and are therefore not representatives of Governments. It follows that they have the same status, privileges and immunities as those of members of other United Nations organs who serve in a personal capacity, that is, those of experts on mission.

### Section 35. Privileges and immunities of persons having official business with the United Nations

78. Although difficulties have arisen from time to time with regard to the entry into the United States of America of representatives of non-governmental organizations, these matters have usually been resolved following the intervention of the Secretariat. In 1982, a major difficulty arose in connection with the participation of certain non-governmental organizations and their representatives at the second special session of the General Assembly devoted to disarmament. The issues that arose concerned the proper interpretation of section 11, paragraph 4, of the Headquarters Agreement in connection with that special session and the limitations, if any, that could properly be placed upon the number of representatives of each non-governmental organization attending the session. A note prepared by the Office of Legal Affairs set out the views of the United Nations as follows:

The Office of Legal Affairs has never had the occasion to seek a general definition of what constitutes, under the Headquarters Agreement, an invitation to United Nations Headquarters requiring the host State to grant admission to the invitee. Nor is this a matter which has been considered by the General Assembly, although immigration procedures are on the agenda of the Committee on Relations with the Host Country and it is open to any member of that Committee to raise at any time with the Committee a particular case or cases or the question of a general definition. No member of the Committee has asked for a meeting in connection with admission to the United States for the present special session on disarmament.

This is a matter which it has been found best to deal with on a pragmatic basis in the context of the particular meeting concerned, and there would appear to be no reason to believe that a general definition would necessarily obviate difficulties. In the past, since the conclusion of the Headquarters Agreement in 1947, there have been very few occasions where differences over admission between the United Nations and the United States have arisen which could not be resolved. Such occasions have in the past not turned on the issue of what constitutes an invitation but on assertions by the host State that the invitee would abuse or had previously abused the privilege of admission by engaging in activities other than those for which admission was ostensibly sought.

Without seeking to be comprehensive in any way, and in the present context relating to non-governmental organizations, the Office of Legal Affairs considers that an invitation under the Headquarters Agreement to the special session on disarmament is clearly involved where a non-governmental organization has been invited by name by the General Assembly. This applies to the organizations listed in annex III of the report of the Preparatory Committee for the second special session of the General Assembly devoted to disarmament.60 The Preparatory Committee further refers in paragraph 28 of its report in a general way to other "non-governmental organizations concerned with disarmament", without naming them. Obviously, interpretations of this phrase can differ. In the view of the Office of Legal Affairs, to qualify for an invitation in terms of the Headquarters Agreement, these other organizations would have to be recognized by the United Nations, for instance under the procedures for consultative status with the Economic and Social Council, with the Centre for Disarmament or with the Department of Public Information.

When an organization is entitled to participate in a United Nations meeting, its participation is necessarily through a reasonable number of representatives of the organization concerned, and not of all its members. It is manifestly unreasonable to expect the host State to accept that it is under an obligation to grant admission to the entire populations of States because the General Assembly has asked "all States" to attend a meeting, or that all members of organizations and liberation movements having invitations to participate in the Assembly have a right of admission to the host State. It is within the

h Ibid., paras. 109 and 110 (f) (i).

i Ibid., paras. 104 (c) and 110 (a) (vi).

<sup>\*\*</sup> Official Records of the General Assembly, Twelfth Special Session, Supplement No. 1 (A/S-12/1).

discretion of the host State to decide to what extent it is prepared to grant visas to large numbers of numbers of an invited group, although the United Nations would insist that a reasonable number of representatives of the group should be admitted to follow the proceedings and, if so invited, to address the meetings concerned. So far, in connection

with the present special session, there have been no instances of which the Office of Legal Affairs is aware where a particular representative of a non-governmental organization whose name has been communicated by the Secretariat as invited has been denied a visa, although there have been delays in granting visas.

#### CHAPTER VI

### United Nations laissez-passer and facilities for travel

### Section 36. Issue of United Nations laissez-passer and their recognition by States as valid travel documents

- 79. The issue of *laissez-passer* continues to be carefully regulated and restricted to officials travelling on official business. A notable exception to this rule, which derives from the Convention on the Privileges and Immunities of the United Nations, concerned a United Nations external auditor of Pakistan nationality.
- 80. External auditors, who under the Convention are considered experts on mission and not officials of the Organization, are entitled to a United Nations certificate but not to a *laissez-passer*. Since Pakistani passports do not authorize travel in Israel and since in the performance of their duties it was necessary for them to visit UNRWA offices in territory under Israeli occupation, the Legal Counsel on 30 October 1968 wrote that:

The decision to issue the *laissez-passer* is taken in the very special circumstances of the present case and is based solely on the agreement of the Government (of Israel), which is asked to recognize the *laissez-passer*. It is not to be considered a precedent with respect to any case in which such agreement of the Government concerned has not been expressly obtained.

### Section 39. Issue of visas for holders of United Nations laissez-passer

- 81. In 1973, the question arose of the legal right of a member of the staff of a regional commission to obtain a visa from the host country in order to return to his duty station. In a memorandum of 13 November 1973 to the Regional Commission Section of the Department of Economic and Social Affairs, the Office of Legal Affairs stated:
- 2. The Convention on the Privileges and Immunities of the United Nations to which the country concerned is a party provides in article V, section 18, that "officials of the United Nations shall be immune, together with their spouses and relatives dependent on them, from immigration restrictions and alien registration". This provision has been taken to mean that States parties to the Convention are bound to issue visas to officials of the United Nations without any restrictions. In addition, the Convention, in article VII, section 25, provides for a speedy handling of applications for visas from the holders of United Nations laissez-passer when such applications are accompanied by a

certificate that the applicants are travelling on the business of the United Nations.

- 3. The headquarters agreement for the regional commission concerned provides that the appropriate authorities shall impose no impediment to transit to or from the headquarters of the commission of, among others, officials of the commission and their families.
- 4. In view of the foregoing, there can be no doubt that from a legal point of view an official of the commission concerned, regardless of his nationality, has the right to return to his duty station and to the issuance of any visa which may be required for entry into the host country.<sup>61</sup>

#### Section 41. Diplomatic facilities for the Secretary-General and other senior officials while travelling on official business

82. Following the reorganization of the top echelon of the United Nations Secretariat in 1967, the stickers or inserts to be attached to the *laissez-passer* issued to Under-Secretaries-General or Assistant Secretaries-General, which dated from 1955, were revised. According to a memorandum from the Legal Counsel dated 28 May 1968, *laissez-passer* issued to Under-Secretaries-General and officials of equivalent rank would bear the following stamp or notation:

### [Diplomatic]

The bearer of this *laissez-passer* is an Under-Secretary-General and, under section 19 of article V of the Convention on the Privileges and Immunities of the United Nations, is entitled to the privileges and immunities, exemptions and facilities accorded to diplomatic envoys in accordance with international law.

83. Similarly, *laissez-passer* issued to Assistant-Secretaries-General and officials of equivalent rank would bear the following stamp or notation:

#### [Diplomatic]

The bearer of this *laissez-passer* is entitled under section 19 of article V of the Convention on the Privileges and Immunities of the United Nations to the privileges and immunities, exemptions and facilities accorded to diplomatic envoys in accordance with international law.

<sup>&</sup>lt;sup>81</sup> United Nations, *Juridical Yearbook 1973* (Sales No. E.75.V.1), p. 168.

### B. SUMMARY OF PRACTICE RELATING TO THE STATUS, PRIVILEGES AND IMMUNITIES OF THE SPECIALIZED AGENCIES AND OF THE INTERNATIONAL ATOMIC ENERGY AGENCY

#### CHAPTER I

### Juridical personality of the specialized agencies and of the International Atomic Energy Agency

#### Section 1. Contractual capacity

- (a) Recognition of the contractual capacity of the specialized agencies and of IAEA
- 1. The capacity of the specialized agencies and of IAEA to enter into contracts continues to be recognized. There have been no decisions of courts or of arbitral tribunals.
- 2. The question of the juridical personality of UPU in Switzerland arose in 1926 on the occasion of the acquisition of a building to house the International Bureau of UPU. The Swiss Federal Council and Federal Tribunal were asked to consider whether UPU, or the International Bureau representing it, could under Swiss law acquire a building. The answer was in the affirmative and this was later expressly recognized by statutory provision. When UPU became a specialized agency of the United Nations, the Federal Council declared the Interim Arrangement on Privileges and Immunities of the United Nations in Switzerland applicable by analogy to UPU as from 1 January 1948, and the legal capacity of UPU was thus confirmed.
  - (b) Choice of law; settlement of disputes and system of arbitration
- 3. The practice of the specialized agencies is, for the most part, not to provide in contracts for the applicability of a particular national law.
- On this question, and in the absence of any provision on the matter in the contract, reference was made to the position of FAO in an arbitral award of 1972 (Balakhany (Chad) Limited v. Food and Agriculture Organization of the United Nations). That position was not the subject of a ruling, since choice of law was not at issue. The arbitrator stated that, as to the law applicable to the contract, the respondent (the organization) had declared that the contract "deliberately contained no choice of law"; as an international organization, FAO "considered that such contract should not be governed by any particular system of municipal law but [exclusively] by generally accepted principles of law". WHO maintains that this is self-evident and applies to contracts between the organization and outside entities. such as contracts for the purchase of supplies and equipment or service contracts, which are governed by the principles of private law; there is no generally accepted international law of contract, international contracts being generally subject to the proper law of the con-

- tract. In the case of leases contracted by IBRD as a tenant or lessor, contracts are executed in accordance with local usage and the *lex situs* normally governs, even if it is not expressly stipulated in the lease. Contracts between IBRD and consulting firms do not normally contain express stipulations on the applicable law although, inasmuch as these contracts are governed by municipal law, such stipulations could be made.
- 5. In the event of a dispute requiring settlement by arbitration, the positions of FAO and WHO do not preclude an arbitrator from referring to a particular system of law in order to ascertain the intention of the parties with respect to certain contractual provisions. For example, contracts concluded by FAO for services sometimes contain a clause requiring the party supplying the services to observe certain provisions of the local law; reference would have to be made to the law concerned if a question arose as to the application of that clause.
- 6. In some cases, contracts made by specialized agencies include express reference to a specific system of municipal law. From time to time, ICAO contracts contain a provision stating that the interpretation of the contract shall be construed in conformity with the laws of the Province of Quebec. While many contracts between IMF and local suppliers make no mention of any kind as to the law applicable, some contracts concluded by IMF for goods and services specify that they will be governed by the law of the site of the main office of the commercial company, usually that of the District of Columbia. In addition, certain financial obligations are subject to the law of the State of New York. On occasion, WHO contracts specify that they are subject to application and interpretation in accordance with a particular system of municipal law, where technical reasons make this desirable, e.g. building contracts and civil engineering contracts. In FAO practice, on rare occasions, as in contracts for the rental of premises, reference is made to interpretation of the contract in accordance with a system of municipal law in case of arbitration. Moreover, in FAO contracts for the provision of certain services, particularly those to be performed on headquarters premises (cleaning, catering, etc.), the concessionaire is specifically required by the contract to apply to his personnel all relevant local laws, regulations and collective agreements governing such matters as conditions of work and social security. IAEA contracts do not refer to a complete system of municipal law as applicable to the contractual relations between the parties, but occasionally an understanding between the parties is recorded in the contract as to the settlement of a particular problem in accordance with national law.

<sup>&</sup>lt;sup>1</sup> Arbitral award of 29 June 1972 (United Nations, *Juridical Yearbook 1972* (Sales No. E.74.V.1), pp. 206-207).

- 7. The practice of IFC has varied from case to case, guided always by the paramount consideration of enforceability of IFC contracts. Consequently, whenever the circumstances of the investment have made it desirable, IFC contracts have made specific reference to a given system of municipal law. Most frequently this has been the case where the parties have wished to stipulate that the governing law shall be a municipal law different from the law of the jurisdiction where the enterprise in question is located.
- 8. The general practice of UPU is that private law contracts concluded by the International Bureau include explicit references to Swiss law (rental, leasing and publishing contracts, etc.), but there are exceptions to this practice. It may happen that contracts concluded with various Swiss or foreign commercial firms make no reference to a positive law.
- The majority of contracts entered into by specialized agencies and IAEA continue to provide for the settlement of disputes by arbitration, after recourse to direct negotiation. Considerable variety exists as to the form or mode of such arbitration. Several agencies (FAO, IFAD, ITU and IAEA) include in their arbitration clauses reference to arbitration under the rules of ICC. For example, as a general rule, FAO makes every attempt to reach an amicable settlement of a dispute, failing which it would seek to have the matter resolved in accordance with the arbitration procedures set out in the contract. FAO has recently submitted two cases to the ICC Court of Arbitration; one was withdrawn upon a settlement being reached, while the other is still pending. In some cases of FAO contracts with United States firms, the rules of the American Board of Arbitration have been declared applicable. WHO contracts provide for arbitration, the form or mode of which is to be agreed between the parties; failing that, the dispute is to be settled under the ICC rules. The WHO standard agreement with consulting firms for pre-investment projects provides for three stages of dispute settlement: negotiation, conciliation by a conciliator jointly nominated by the parties, and arbitration under the ICC rules. IAEA contracts may include one of two models for arbitration: (a) submission of the dispute to arbitration under ICC rules, which have been used in recent years as appropriate; or (b) submission of the dispute to three arbitrators, one appointed by each party and the third by the two appointed arbitrators. Failing agreement on the appointment of the third arbitrator, the Secretary-General of the United Nationsmay be requested to appoint him. IBRD contracts with consulting firms normally provide for arbitration by ad hoc arbitrators or by reference to the ICC rules. When IBRD lease contracts contain provision for dispute settlement, which is not always the case, they may refer to settlement by arbitration or submission of disputes to the jurisdiction of the local courts. Contracts between IMF and local construction companies and certain suppliers have provided for the submission of disputes to arbitration. In certain of its borrowing agreements, IMF has agreed to settlement of disputes by arbitration. No such disputes have arisen. Agreements providing for IFC investments do not refer to arbitration. Occasionally, however, IFC has agreed that disputes thay may arise in respect of a contract for ser-

- vices required by it for its operations, or in respect of certain arrangements between creditors of a company, be submitted to arbitration. In such cases, *ad hoc* arrangements are made. It may be recalled that, under certain jurisdictions, disputes between shareholders or partners in a company may not be submitted to the ordinary courts of the country and can only be adjudicated by arbitrators.
- 10. Thus far, no lawsuit relating to the private law contracts of UPU has been brought before a Swiss or foreign court. On the other hand, disputes concerning the interpretation of the Acts of UPU,<sup>2</sup> or liability resulting from the application of the Acts, are subject to the special arbitration procedure provided for in article 127 of the General Regulations of UPU.
- 11. IBRD and IDA have maintained their distinct body of practice as regards those contractual transactions that constitute their field of activity. IBRD practice varies depending upon the type of contract involved.
- 12. The practice of IBRD as a lender has been fully explained in several publications. With respect to loans made by IBRD, international arbitration is the method for settling disputes. Standard provisions for arbitration are found in section 10.04 of the IBRD General Conditions applicable to loan and guarantee agreements dated 15 March 1974.
- As a borrower, IBRD practice varies depending on the custom in the particular market in which the issuance of bonds takes place and on the character of the lender. As to the custom in a particular market, IBRD bonds issued in continental markets are expressly governed by the law of the relevant market, while IBRD bonds issued in the United States of America, the United Kingdom or Canada contain no stipulation of applicable law. It may be assumed, however, that in both cases the law of the relevant market applies. As to the character of the lender, loans made by the Swiss Government to IBRD are governed by international law. Loans raised by IBRD from certain national institutions, such as the Deutsche Bundesbank, while governed by municipal law, contain no express stipulation of applicable law. Under annex VI, paragraph 1, of the Convention on the Privileges and Immunities of the Specialized Agencies (hereinafter referred to as the "specialized agencies Convention"), IBRD, as a borrower, enjoys no general immunity from suit. It may be sued by its creditors in "a court of competent jurisdiction in the territories of a member . . . in which the Bank has an office, has appointed an agent for the purpose of accepting service or notice of process, or has

<sup>&</sup>lt;sup>2</sup> Organic texts concerning the structure, operation and legal status of UPU (United Nations, *Treaty Series*, vol. 611, p. 7), revised at Hamburg in 1984 (International Bureau of UPU, *Acts of the Universal Postal Union*, Bern, 1985).

<sup>&#</sup>x27;See A. Broches, "International legal aspects of the operations of the World Bank", Recueil des cours, Académie de droit international, 1959-III, vol. 98 (Leyden, Sijthoff, 1960), p. 301. See also G. R. Delaume, Legal Aspects of International Lending and Economic Development Financing (Dobbs Ferry, N.Y., Oceana Publications, 1967), pp. 81-85, 88-91 and 108-110, and Transnational Contracts, Applicable Law and Settlement of Disputes (Dobbs Ferry, N.Y., Oceana Publications, 1986), paras. 1.12 and 2.12.

issued or guaranteed securities". Moreover, in certain markets, especially those in Europe, bonds issued by IBRD expressly provide for the jurisdiction of local courts.

14. IDA has so far made credits available only to Governments of member States. Relations arising under the relevant credit agreements are governed by international law. The IDA General Conditions applicable to development credit agreements, dated 15 March 1974, contain substantially the same provisions as to enforceability and arbitration (sections 10.01 and 10.03) as the corresponding provisions of the IBRD General Conditions.

### Section 2. Capacity to acquire and dispose of immovable property

- 15. The specialized agencies and IAEA report that no problems have been encountered with regard to their capacity to acquire and dispose of immovable property, as provided for in section 3 (b) of the specialized agencies Convention. It may be noted, however, that the deposit by the Government of Indonesia in 1972 of its instrument of accession to the Convention was accompanied by a reservation to that provision. The reservation reads: "The capacity of the specialized agencies to acquire and dispose of immovable property shall be exercised with due regard to national laws and regulations." In 1973, the Indonesian Government informed the Secretary-General, in reference to that reservation, that it would grant to the specialized agencies the same privileges and immunities as it had granted to IMF and IBRD.5
- 16. Instances of the use, acquisition or disposal of immovable property by the specialized agencies and IAEA are given below.

#### (i) *ILO*

17. In 1967, ILO concluded a contract with the Property Foundation for International Organizations, which acted on behalf of the Swiss authorities, by which ILO transferred to the Foundation the ownership of the land on which the former ILO building was located and the Foundation transferred to ILO the ownership of the land on which the present ILO building was to be constructed. The ownership of the former ILO building was also transferred to the Foundation in return for monies to be used by ILO in the construction of the new building. In 1975, ILO concluded a contract with the Etat de Genève by which the organization transferred to the Etat de Genève the ownership of land in the vicinity of the old ILO building in exchange for a piece of land

adjoining the site of the new ILO building. The transfers were made in the form required by Swiss law and registered. No fee or charges were paid by ILO.

#### (ii) FAO

18. FAO has not acquired title (freehold or leasehold) to real estate. Land and buildings with appurtenances for headquarters and regional offices, and more recently for the offices of FAO representatives in various countries, have generally been made available to FAO directly by the host Government on the basis of an agreement with that Government, or have been rented by FAO from the owner.

### (iii) UNESCO

19. In 1973, UNESCO acquired a property at San Isidro (Villa Ocampo) and another property at Mar del Plata (Villa Victoria), in Argentina. At Villa Ocampo, which it had acquired by deed of gift, UNESCO has established an Iberoamerican Research and Study Centre for Scientific and Cultural Translation. A head-quarters agreement is to be concluded between UNESCO and Argentina to that effect.

#### (iv) ICAO

20. ICAO owns its regional office building in Paris (in co-ownership with the French Government).

#### (v) WHO

21. WHO has acquired immovable property which it has either bought or had donated to it exempt from duties. So far it has disposed of only one such property, namely, a villa situated in Florence, Italy, which had been bequeathed to it by a private individual. As the legatee, WHO disposed of the property after the death of the testator in 1975.

#### (vi) IBRD/IDA/IFC

22. IBRD has purchased, sold, rented and leased property at its headquarters and in various member countries. It has concluded leases with other international organizations (e.g. with IMF), Governments and private entities. IFC leases necessary office space in various member States. For example, it holds a long-term lease on a house in London for use as the residence of the IFC Special Representative in Europe. In addition, IFC acquired two parcels of forest land in Paraguay in foreclosure proceedings initiated by IFC against a company that had not serviced a loan IFC had made to it.

#### (vii) IMF

23. IMF has purchased and sold property in Washington, D.C. and adjoining areas; it has also leased property from private parties, in Washington, D.C. and Geneva, and from the World Bank in Paris.

#### (viii) UPU

24. Since 1926, UPU has owned the three buildings which have housed its headquarters. The first building was sold to a commercial firm. The one that it occupied from 1953 to 1970 was transferred to the UPU provi-

<sup>\*</sup> Delaume, Legal Aspects of International Lending ..., pp. 171-175, and Transnational Contracts ..., para. 11.03.

<sup>&</sup>lt;sup>5</sup> United Nations, Multilateral Treaties deposited with the Secretary-General: Status as at 31 December 1982 (Sales No. E.83.V.6), chap. III.2.

<sup>&</sup>lt;sup>6</sup> See International Labour Office, Official Bulletin, vol. L. No. 3, July 1967, p. 335. For a detailed study on the question, see B. Knapp, "Questions juridiques relatives à la construction d'immeubles par les organisations internationales", Annuaire suisse de droit international, vol. XXXIII (Zurich, Société suisse de droit international, 1977), p. 51.

dent scheme and forms part of its assets. When the present UPU building in Bern was under construction, UPU had to pay compensation to the owners of three neighbouring buildings who had objected to the construction of the building because its height exceeded the maximum permitted under the district plan in force at the time.

#### (ix) ITU

25. ITU has constructed an office tower with right of superficies.

#### (x) IAEA

26. IAEA has no title to immovable property. The buildings that it occupies are either rented or occupied free of charge.

### Section 3. Capacity to acquire and dispose of movable property

(a) Recognition of the capacity of the specialized agencies and of IAEA to acquire and dispose of movable property

27. The specialized agencies and IAEA have generally not encountered problems concerning their capacity to acquire and dispose of movable property. FAO, however, has found that it is often difficult to maintain direct title to vessels and aircraft in view of the fact that such title would require registration under the laws of a particular country. The specific problem that arises, as far as the registration of vessels is concerned, is that, under the legislation of many countries, intergovernmental organizations are not among the entities recognized as entitled to register vessels. The normal practice when vessels are assigned to a project is for FAO to conclude an agreement with the Government receiving assistance, under which title is transferred to the Government or a government agency and the vessel is registered accordingly. Title is normally transferred back to FAO upon completion of the assignment. Where necessary, the vessel remains registered in the name of the Government or agency for a short period following such completion, pending its reassignment by FAO to another project. Under some agreements of the kind referred to, title is retained by FAO, but registration is in the name of the government agency. However, FAO has occasionally registered vessels in a member State.

### (b) Licensing and registration of land vehicles, vessels and aircraft

28. Most specialized agencies and IAEA have licensed or registered land vehicles with the appropriate local authorities and in accordance with local law. In 1974,

ILO acquired ownership of a seagoing vessel for training purposes. The vessel was registered in Bangladesh in accordance with local law. The practice of FAO with regard to registration of vessels has been described above (para. 27).

### Section 4. Legal proceedings brought by and against the specialized agencies and IAEA

29. The capacity of the specialized agencies and IAEA to institute legal proceedings before national tribunals has not been questioned. However, these organizations have seldom instituted such legal proceedings.

FAO has instituted legal proceedings or filed claims in legal proceedings on a number of occasions. In 1969, FAO filed a proof of debt with a United States court in proceedings for an arrangement under the United States Bankruptcy Act. More recently, a proof of claim and release has been filed on behalf of FAO with a United States court in a class action of stockholders of a United States corporation. In November 1974, FAO retained local legal counsel and initiated action in Italian courts to obtain payments due under a mortgage loan which had been bequeathed to FAO by an Italian citizen; the matter is still pending in the Italian courts. In March 1981, FAO retained local legal counsel and filed a claim together with other major creditors in a bankruptcy proceeding in the State of Vermont, in the United States. FAO and the other creditors agreed to a plan of reorganization which was confirmed by the Vermont bankruptcy court and is at present being carried out under the supervision of the court. In June 1981, FAO retained local legal counsel and initiated legal action in the courts of the Philippines for damages arising from the loss of a marine cargo shipment in June 1980. The shipment had been insured and action was instituted by FAO against both the shipping company and the insurers. The shipping company made a counterclaim for the cost of salvage operations. The case is still pending before a court of first instance. In November 1981, FAO, acting jointly with the United Nations, retained local legal counsel and initiated legal action in the United Republic of Tanzania following the crash of an aircraft in which four staff members of FAO died. The action is for third party compensation on behalf of the dependants of the deceased FAO staff members. FAO entitlements with respect to any award or settlement would be limited to legal costs and the amounts charged to the FAO Staff Compensation Plan Reserve Fund, for compensation payments made by FAO to the deceased staff members' dependants. The case is still pending.

31. IMF instituted legal proceedings in the District of Columbia for recovery of compensation for water damage caused to IMF premises. The suit was settled in June 1984. IMF has also filed an opposition to a trademark application made by a private Canadian company before the Canadian Registrar of Trademarks. The case is still pending.

<sup>&#</sup>x27;The UPU provident scheme is not affiliated with the United Nations Joint Staff Pension Fund. It is independent of the Fund and constitutes a foundation within the meaning of articles 80 et seq. of the Swiss Civil Code. It enjoys the same exemptions, immunities and privileges as UPU with regard to its activity on behalf of staff members of the International Bureau of the Union and within the limits of the Agreement between the United Nations and Switzerland, which is applicable by analogy to UPU (see p. 182, para. 2, above).

<sup>&</sup>lt;sup>8</sup> United States Code, 1982 Edition, vol. IV, 1983, title 11.

<sup>&</sup>lt;sup>9</sup> International Bank for Reconstruction and Development v. Charles H. Tompkin Co. (No. 83-1045 (D.D.C.)).

- 32. In 1972, WHO instituted legal proceedings, jointly with a staff member, before the Supreme Court of the Philippines, against the decision of a judge in a court of first instance, not to quash a search warrant. The Supreme Court declared null and void the search warrant in question.<sup>10</sup>
- 33. UNESCO was involved as a third party in a case brought before a French tribunal between the heirs of a former UNESCO official, victim of a road accident, and the insurance company of the author of the accident (a private company). In another case, where a building society claimed payment of additional costs, UNESCO was defendant in the proceedings brought before an arbitration tribunal. The decisions in both cases were favourable to UNESCO.
- 34. IFC has brought legal proceedings before the municipal courts of Argentina, Brazil, Chile, Costa Rica, Indonesia, Paraguay and Spain, the majority of which were bankruptcy proceedings.
- 35. With regard to steps to avoid or mitigate liability, other than the purchase of insurance, the practice varies. UPU and IFC have taken no particular measures. WHO, so far as possible, endeavours to contract out of liability, particularly in cases where liability arises out of aid or assistance provided to Governments. Only where this is not possible is recourse made to insurance. IAEA has either disclaimed liability (such disclaimer being effective in relation to the other party to the agreement) or has tried to obtain hold-harmless undertakings, from the other parties to the agreement, against third party liability.
- 36. Under the FAO Staff Compensation Plan for service-incurred injury, staff members or their survivors may be required to sue third parties as a prior condition for receipt of compensation. Provision is made for compensation thus recovered to be set off against FAO liability under the Compensation Plan. In three cases, actions that were brought in Senegal, Canada and Algeria substantially reduced FAO liability.
- 37. FAO practice in technical assistance projects is normally to include "hold-harmless" clauses, worded in substantially the same way as in article X, paragraph 2, of the UNDP standard basic technical assistance agreement," in agreements with Governments receiving technical assistance. Such agreements usually contain a provision whereby the Government undertakes to provide adequate insurance for counterpart personnel. Moreover, subcontractors are normally required to make provision for third-party liability insurance, in addition to the insurance of their own staff. With regard to projects executed by IBRD for UNDP, the Bank in-

The practice adopted by IMF is often to clarify the limits on its liability in the course of its varied financial and other activities. For example, following a decision in 1975 to sell gold for the benefit of developing countries, the Executive Board in May 1976 adopted a fouryear gold sales programme in which one sixth of the Fund's gold was to be sold at public auctions. Under the "Terms and Conditions" for such auctions, title to gold purchases passes to the purchaser upon delivery made to the carrier designated by him. After passage of title, all risk of loss or damage, from any cause whatsoever, is to be borne by the purchaser. In 1952, the Fund established a gold transaction service to assist members and certain international organizations in their gold transactions by trying to match prospective purchasers and sellers of gold. One of the terms on which the service was provided was that the Fund would not become a party to any contract of purchase or sale and would incur no liability or obligation in connection with the transactions.

### Section 5. International claims brought by and against the specialized agencies and IAEA

39. In the period under review, neither any specialized agency nor IAEA has instituted international claim proceedings against another subject of international law, or has been a respondent in an international claim proceeding.

#### Section 6. Treaty-making capacity

- (a) Treaty-making capacity of the specialized agencies and of IAEA
- 40. The specialized agencies and IAEA have experienced no special problems concerning their treaty-making capacity. A number of them have entered into agreements with non-member States as well as member States.
  - (b) Registration, or filing and recording of agreements on the status, privileges and immunities of the specialized agencies and of IAEA
- 41. Most of the agreements entered into by the specialized agencies and IAEA concerning their status, privileges and immunities have been registered or filed and recorded with the United Nations Secretariat. Agreements not registered or filed and recorded relate to arrangements with Governments hosting conferences, seminars, meetings, etc., outside the head-quarters or established regional offices of an agency. Such agreements often take the form of an exchange of letters.

corporates in the project documents the clauses on privileges and immunities of the basic agreements between the country concerned and UNDP.

<sup>&</sup>lt;sup>10</sup> World Health Organization and Dr. L. Verstuyft v. Benjamin Aquino (1972) (United Nations, Juridical Yearbook 1972 (Sales No. E.74.V.1), pp. 209-211).

<sup>11</sup> DP/107, annex I.

#### CHAPTER II

### Privileges and immunities of the specialized agencies and of the International Atomic Energy Agency in relation to their property, funds and assets

### Section 7. Immunity of the specialized agencies and of IAEA from legal process

- 42. Most specialized agencies and IAEA state that their immunity from legal process has been fully recognized by the competent national authorities.
- 43. ILO reports that in 1966 a private person filed a claim against the organization in Costa Rica. Following assertions of immunity by ILO to the Government of Costa Rica, the Government informed the court of the privileges and immunities of ILO. The matter is considered closed. IMF was the subject of a claim filed by a private person in 1974 before the United States Equal Employment Opportunity Commission, and later before a United States federal district court.12 In both instances, the immunity of IMF was upheld. In 1975, IMF was notified to appear in a hearing before the District of Columbia Minimum Wage and Safety Board. The District's counsel, following assertion of immunity by IMF, concluded that the Fund was immune from prosecution. UPU has in one case invoked its immunity from jurisdiction in order not to appear as a witness in a criminal proceeding.
- 44. FAO reports that, since 1978, legal proceedings have been instituted against it in nine cases before national tribunals, seven in the host country and two in other countries. Of the nine cases, six were brought by private citizens or private companies, while three were instituted by parastatal corporations. Such proceedings were instituted notwithstanding the existence of applicable international agreements according FAO immunity from legal process. Except in two of the proceedings that had been instituted before local courts in the host country by a parastatal corporation, FAO did not appear before local courts in any of the cases referred to. However, in every case it formally advised the Government that legal proceedings had been instituted against FAO in a local court; called the attention of the Government to the specific provisions of the international agreement that provided for the organization's immunity from legal process; and requested that the judicial authority concerned be informed.
- 45. With respect to the two proceedings instituted before local courts in the host country, Italy, by a parastatal corporation, FAO, on the advice of the Ministry of Foreign Affairs, made a limited appearance before the local court, for the sole purpose of invoking its immunity from legal process as provided for under the relevant provision of its Headquarters Agreement. In 1982, the Supreme Court of the host country rendered a judgment denying immunity. Since then, FAO, following the instructions of its governing council, has declined to appear in court even on a limited

- basis. It is the present policy of FAO not to appear in proceedings before courts of the host country.
- 46. FAO has continued to contest jurisdiction in actions brought against it in local courts. It has always raised the question of jurisdiction with the Government of the country concerned and in two cases (para. 44, above) the jurisdiction of the court was also contested in a local court.
- 47. The facts relating to six of the nine proceedings instituted against FAO, have been reported by the organization. They are described in paragraph 48.
- 48. The following cases were brought before courts of the host country, Italy:
- (a) A parastatal corporation which, on behalf of persons employed in the performing arts, collects contributions from employers to pension and social security benefits, instituted proceedings against FAO for failing to make such contributions on emoluments paid to a citizen who had been engaged from time to time by FAO over a period of years as a non-staff member under a series of special service agreement contracts. The claim was brought to the attention of the permanent representative of the host country. FAO did not consider negotiation or settlement, since it was fundamental for the organization to remain independent of application of local labour laws. This position was communicated to the Italian Government.13 The corporation obtained a court judgment in October 1982 against FAO for payment of contributions on the payments made to the nonstaff member. There have been no further developments since the judgment.
- (b) Two legal proceedings were instituted against FAO by a parastatal corporation that manages a pension fund for directors and managers of private industry. The actions were for rental arrears and for eviction. The dispute concerned the applicability or nonapplicability of rent control laws to FAO tenancy of part of an office building owned by the corporation. The two proceedings resulted in a number of court decisions on procedural and substantive questions. In one case, a judgment by a local tribunal did not recognize FAO immunity from legal process. FAO submitted the issue of its immunity to the Supreme Court of the host country. The Supreme Court held that FAO did not enjoy immunity from legal process in the proceedings in question. There followed a judgment to the effect that the corporation did not have the right to evict FAO and another judgment in favour of the corporation for pay-

<sup>12</sup> Kissi v. de Larosière (No. 82-1267 (D.D.C.)).

<sup>&</sup>quot;FAO noted a 1969 opinion of an Italian court of first instance whereby the agency's plea of immunity from jurisdiction was accepted in a case brought against it by a former staff member. In the court's opinion, "such immunity could only be recognized with regard to public law activities, i.e., in the case of an international organization, with regard to the activities by which it pursues its specific activities". United Nations, Juridical Yearbook 1969 (Sales No. E.71.V.4), pp. 238-239.

ment by FAO of rental arrears. The Court, in the latter judgment, considered the relevant section of the Headquarters Agreement with the host country, providing for "immunity from every form of legal process". The Court concluded that FAO immunity from legal process extended only to matters relating to activities undertaken in carrying out the purpose and functions of the organization, i.e. acts jure imperii, and not to transactions of a private law nature that might arise out of other activities, i.e. *iure gestionis*. Measures of execution have not been sought against FAO. Negotiations between FAO and the corporation are proceeding with a view to an extrajudicial settlement of the question of rental arrears. Nevertheless, the interpretation placed upon the relevant section of the Headquarters Agreement by the Court is of great concern to the FAO governing bodies, which disagree with the interpretation and maintain that the provisions of the headquarters agreement should be given their full literal meaning. Otherwise, it is considered, FAO would be open to litigation detrimental to effective implementation of its programmes. Other organizations of the United Nations system with immunities covered by analogous provisions are likely to be in a similar position. The FAO governing bodies will in 1985 consider whether an advisory opinion of the International Court of Justice should be sought as to the interpretation of the relevant provisions of the Headquarters Agreement with the host country.

- (c) While negotiations with respect to a contractual dispute were continuing between FAO and a contractor who had been engaged by FAO for services to be performed at FAO headquarters, a subcontractor of the contractor instituted legal proceedings against FAO with respect to the subject matter of the dispute. The subcontractor is not a party to the organization's contract with the principal contractor and thus is not bound by the arbitration clause in the contract.
- 49. Legal proceedings were instituted against FAO in Honduras for compensation for personal injury and property damage caused by an FAO vehicle engaged on a UNDP project. It was the position of FAO that the Government was responsible for dealing with the claim under the "hold-harmless" clause in the UNDP standard basic technical assistance agreement with the Government (see p. 186, para. 37, above). The Government declined to assume responsibility on the grounds that the vehicle had not been used for project purposes at the time of the accident. The Government declined to intervene in court on behalf of FAO, and maintained that the organization should itself invoke its immunity. FAO did not appear in court and in September 1984 a judgment was entered in favour of the plaintiff. FAO declined thereafter to receive service of the judgment. An extrajudicial settlement was finally arranged with the collaboration of the Ministry of Foreign Affairs.
- 50. Legal proceedings were instituted against FAO in Bangladesh by a locally employed FAO staff member who had been separated upon expiry of his appointment. Summons addressed to FAO by the local court were returned by the organization to the Ministry of Foreign Affairs under a *note verbale* recalling FAO immunity from legal process. In August 1984, judgment of court, holding the plaintiff's termination void and il-

legal and providing for re-employment of the plaintiff and an award of damages, was communicated to FAO. FAO advised the Government of the judgment. Execution of the judgment has not been sought.

- IBRD, IDA and IFC do not enjoy general immunity from suit. Their immunity is limited to actions brought by member States or persons acting for or deriving claims for such States. Actions by other persons may be brought in a court of competent jurisdiction in the territory of a member State in which the organization has an office, has appointed an agent for the purpose of accepting service or notice of process, or has issued or guaranteed securities. No cases have been reported by IBRD, IDA or IFC in which their limited immunity has not been recognized, IBRD and IDA state that actions have been brought against them on rare occasions (no more than eight in number) in conformity with the relevant annexes of the specialized agencies Convention. Such cases have been settled amicably, discontinued or dismissed. One case is of special interest. In 1972, a complaint filed in a United States federal district court was served on IBRD, IDB and Uruguay. The plaintiff alleged that a contract for consulting services had been broken by Uruguay and brought action for damages. IBRD and IDB moved to dismiss the action on the grounds that the court lacked jurisdiction, that the case should the transferred to the court of first instance of the District of Columbia and that the complaint did not state a cause of action against either Bank. The motion was granted. The action against Uruguay was dismissed on the grounds that the court lacked jurisdiction in light of the choice of forum clause in the contract, in favour of the courts of Uruguay, a clause deemed not unreasonable and therefore enforceable.14
- 52. As to the provision in section 4 of the specialized agencies Convention concerning immunity "from every form of legal process", most of the specialized agencies and IAEA report no special difficulties of interpretation, although it would seem that occasions requiring such interpretation have seldom arisen. IMF has taken the view that the term is to be interpreted broadly, as extending to the exercise of all forms of judicial power. The Fund has received notices of attachment of funds due to taxpayers and bankrupt persons, as well as subpoenas requiring staff members to appear as witnesses. It has asserted its immunity from judicial process and the inviolability of its archives. Such immunity has been recognized whenever it has been invoked. Apart from cases where legal proceedings have been instituted against FAO, the organization has successfully invoked its immunity from "every form of legal process" whenever it has been ordered, by a national court or other authority, to disclose information (concerning in particular salaries) relating to a staff member, or where national courts have sought to attach the salary due to staff members before salary payment has been made (see also sections 23 and 32 below).
- 53. IBRD reports that on three occasions attempts were made by self-styled creditors of member States of the World Bank to attach funds allegedly held by the

<sup>&</sup>lt;sup>14</sup> Republic International Corporation v. Amco Engineers (1975) (Federal Reporter, 2nd Series, 1975, vol. 516, p. 161).

Bank on behalf of those members. IBRD claimed immunity on several grounds, arguing that: (a) under article VII, section 3, of its Articles of Agreement, the proceeds of its loans to member States were its property and, as such, could not be attached prior to the delivery of a final judgment against it (as distinguished from a judgment against a member); (b) under the terms of the second sentence of the same provision, the self-styled creditors, who "derived" their claims from member States, were as such barred from bringing action against the Bank; (c) under article III, sections 1 (a), 5 (b) and 5 (c) of the same Articles of Agreement, loan agreements between IBRD and member States were governed by international law and were intended for public purposes that could not be compromised by private self-styled creditors, especially since the use of IBRD resources and withdrawals of loan proceeds were subject to strict conditions. In one case the action was discontinued, one case appears to be at a standstill, and the last is still pending. It is to be noted that the United States Foreign Sovereign Immunities Act of 1976 provides expressly that the property of international organizations designated by the President of the United States (IBRD, IDA and IFC are among the organizations designated) "shall not be subject to attachment or any other judicial process impeding the disbursement of funds to, or on the order of, a foreign State as the result of an action brought in the courts of the United States or of the States".15 This provision does not add to the immunities derived by IBRD from its Articles of Agreement; it simply puts an end to the speculation that has sometimes arisen that loans made by IBRD might be a source of attachable funds.16

54. There have been no cases in which the question of immunity from measures of execution has been addressed. FAO reports, however, that, at the eighty-sixth session of the FAO Council, in November 1984, the representative of the host country, Italy, made a declaration on the agency's immunity from legal process and measures of execution in the host country. The declaration is incorporated in the Council's report and includes the following statement:

The other point that deserves the utmost attention is the clear distinction which exists between, on the one hand, the general concept of immunity from legal process, with which I dealt at the beginning of my speech, and, on the other hand, the concept of immunity from measures of execution. While the former concept has some limits (i.e. it applies only to acts jure imperii as opposed to acts jure gestionis), the immunity from measures of execution enjoyed by FAO under the [host country] legal system is full and complete. It is true that there has never been any test case to prove that the courts would uphold such immunity, but it is not difficult to understand that the reason why no one has ever tried to attach FAO property (for instance [a para-statal corporation] which had already obtained a court judgment condemning FAO to pay) is exactly the legal impossibility to carry out measures of execution against the organization. In this connection, too, however, it is important to realize that, if someone attempted to carry out measures of execution against FAO (by initiating an ad hoc proceeding before the competent "judge of the execution" in accordance with the code of civil procedure), the organization

would have to appear before the judge in order to point out the existence of its immunity under . . . the Headquarters Agreement. 17

FAO considers that the words "every form of legal process", used in its Headquarters Agreement, also covers immunity from measures of execution. On the other hand, the statement cited above indicates a limitative interpretation of these words.

### Section 8. Waiver of the immunity from legal process of the specialized agencies and of IAEA

- 55. There have been a few cases of agencies waiving their immunity from legal process in particular cases. ILO reports that in 1980 testimony was sought by Canadian provincial authorities in relation to prosecution of a third party under the Occupational Health and Safety Act. ILO waived the immunity of the Director of its Ottawa Branch Office. IMF has waived its immunity for the purpose of leases. Bearer notes associated with certain IMF borrowing agreements provide waiver by IMF of its immunity from judicial process and submission to designated national courts with respect to both actions and execution. UPU has recognized the jurisdiction of Swiss tribunals in litigation cases, but no suits have been brought in such tribunals.
- 56. FAO reports that, by initiating legal proceedings under national law, it has implicitly waived its immunity with respect to counter-claims that might be raised by the defendant in the proceedings. A counter-claim was made against FAO when it initiated action against a shipping company in the Philippines. A statement by FAO concerning certain pesticides at public hearings of the United States Environmental Protection Agency brought FAO within the scope of a rule of practice which made statements made at the hearing subject to the availability for cross-examination of persons making such statements. In another case, when invoking immunity with respect to proceedings brought against it before a national court, FAO informed the Government of the country concerned that it did not consider that its liability was involved (the claim related to the actions of a person who was not a staff member), but undertook to make further investigations.
- 57. The majority of contracts entered into by the specialized agencies and IAEA provide for settlement of disputes by arbitration (see p. 182, section 1 (b) above).

# Section 9. Inviolability of the premises of the specialized agencies and of IAEA and exercise of control by the specialized agencies and by IAEA over their premises

- 58. The inviolability of the premises of the specialized agencies and IAEA has in general been recognized. The specialized agencies and IAEA have for the most part remained immune from search and from any other form of interference.
- 59. IMF reports that on a number of occasions local police have attempted unsuccessfully to serve subpoenas

<sup>15</sup> Public Law 94-583 of 21 October 1976 (United States Statutes at Large, 1976 (1978, vol. 90), section 1611 a).

<sup>&</sup>lt;sup>16</sup> Delaume, "Public debt and sovereign immunity revisited: some considerations pertinent to H.R. 11315", *The American Journal of International Law* (Washington, D.C.), vol. 70, No. 3, July 1976, p. 529, and *Transnational Contracts*..., paras. 12.02 and 12.04.

<sup>17</sup> FAO, Report of the Council of FAO, Eighty-sixth Session, Rome, 19-30 November 1984 (CL 86/REP), annex J.

and arrest warrants on the Fund's premises. IMF has taken the position that its premises may not be entered for such purposes without its express consent, ILO has in some cases authorized the police to enter its premises in Switzerland but there has been no waiver of immunity. The WHO Headquarters Agreement provides that no agent of the Swiss public authority may enter the organization's premises without the express consent of WHO or at its request. The premises are thus inviolable. The same appears to be the case in the WHO regional offices. IBRD and IDA report that there have been no problems with respect to the immunity of their premises. Some of their member countries, however, in which IBRD or IDA have offices, have not adhered to the specialized agencies Convention. IBRD and IDA rely, in such countries, on the pertinent provisions of their Articles of Agreement.

- 60. FAO reports that in 1984 it rejected work done by a contractor on FAO headquarters premises. The contractor requested a local court to designate an expert to provide a technical evaluation of the work. A copy of the court order was transmitted to FAO with a note verbale by the permanent representative of the host country. FAO returned the court order with a note verbale pointing out that, in addition to the organization's immunity from legal process, its headquarters were inviolable under section 7 of the FAO Headquarters Agreement with the host country. FAO could not therefore accept the court order. The matter was amicably settled with the contractor in February 1985 and no further court action was taken.
- Difficulties have on occasion arisen with regard to 61. the inviolability of the premises of regional offices of certain agencies. In 1967, the police entered ILO premises in Lagos and arrested a member of the local staff. Following intervention by the Director of the ILO regional office, the staff member was promptly released and the Nigerian Government indicated that steps to avoid a recurrence had been taken. In 1973, an ILO office in Santiago was searched by the police. The matter was referred to the Chilean Government, which sent ILO a satisfactory reply. There have lately been some cases of arrest of locally recruited officials on the premises of the ILO office in Addis Ababa. ILO has referred the matter to the Minister of Foreign Affairs, drawing attention to the provisions of the Agreement concerning the ILO regional office in Ethiopia. WHO has also reported violations of its premises in some of its field offices. In Rio de Janeiro, in the zone office of the WHO regional office for the Americas and in the Pan American Foot-and-Mouth Disease Centre, difficulties arose in connection with claims made, under local labour laws, by locally recruited staff, which had repercussions on the jurisdictional immunity of WHO and the inviolability of its premises and property. The difficulties were resolved to the satisfaction of the organization through negotiations involving the good offices of the Ministry of Foreign Affairs of Brazil.
- 62. As to the authority of the specialized agencies and IAEA to adopt regulations superseding municipal law within their premises, FAO points out that, under section 6 (a) of its Headquarters Agreement, the Italian

- Government recognizes the extraterritoriality of the "headquarters seat", "which shall be under the control and authority of FAO". Section 6 (b) provides that, "except as otherwise provided in this agreement", the laws of the Italian Republic apply within the headquarters seat, and section 6 (c) provides that the Italian courts have jurisdiction over acts done and transactions taking place at the headquarters seat. These provisions ensure that the extraterritoriality of the headquarters seat does not lead to private acts and transactions performed there being in what might be termed a legal vacuum. FAO considers that it has exclusive authority to regulate all matters within its competence, namely, matters connected with the carrying out of its purposes and functions. Under article 5, paragraph 2, of the UNESCO Headquarters Agreement, the organization has the right to make internal regulations applicable throughout its headquarters premises in order to enable it to carry out its work. The Staff Regulations and Staff Rules of the organization, in particular, have been drafted in conformity with this provision.
- 63. WHO reports that municipal law is not applicable on its premises and that it has the right to adopt regulations applicable thereto. Such regulations have been adopted in connection with parking in the WHO head-quarters underground garages and in connection with security measures for the protection of persons and property in case of fire, flood, earthquake and loss and theft of property. It is pointed out, however, that Swiss municipal law has been taken into account in formulating certain regulations, such as fire regulations.
- 64. Although IAEA is not empowered to adopt regulations superseding municipal law, section 8 (a) of its Headquarters Agreement authorizes it "to make regulations, operative within the headquarters seat, for the purpose of establishing therein any conditions necessary for the full execution of its functions". The effect of such regulations is to exclude the application, within the headquarters seat, of any Austrian laws inconsistent therewith, and their texts are to be notified to the Government from time to time.
- 65. Other specialized agencies report that they do not have such a right and that it has not been contemplated. IMF adds that it does not have such a right, except with regard to the adoption of administrative and personnel regulations.

## Section 10. Immunity of the property and assets of the specialized agencies and of IAEA from search and from any other form of interference

66. Only one case has been reported involving difficulty with recognition of the immunity of a specialized agency's property and assets from search and from any other form of interference. A problem arose in 1976 when a consignment purchased by WHO for assistance to a member State was sold by the customs authorities in Kenya. WHO initiated consultations, which are continuing, to recover the value of the consignment with the authorities of the State concerned.

### Section 11. Name and emblem of the specialized agencies and of IAEA: United Nations flag

- 67. In connection with the display or use by the specialized agencies and IAEA of the flag, official emblem or seal of their own organization, of the United Nations or of a member State, no major problems are reported to have arisen. Minor problems are reported to have arisen from time to time by FAO regarding the appropriate place on which to display the United Nations flag on FAO vessels. In the case of WHO, doubts arose on one occasion as to the circumstances in which the WHO flag should be flown at half-mast. Such issues have not given rise to difficulties.
- 68. Certain agencies have had occasion to protect their name, emblem or flag from unauthorized use, through adoption of resolutions, codes or other measures. WIPO, for instance, has done so in accordance with the Paris Convention for the Protection of Industrial Property (Stockholm, 14 July 1967).18 ILO has brought such measures to the attention of certain States in order to avoid the unauthorized use of its name, and it reports that the competent authorities in the countries concerned have always lent their support. No legal proceedings have been necessary. Problems have arisen involving the unauthorized use of the name and official emblem of WHO by firms (mostly pharmaceutical) in connection with their publicity or promotional material for their products. The practice of WHO in such cases is to write to such firms requesting them to desist from any such use, and in the great majority of cases the firms have complied with such requests. UPU has been obliged to intervene on numerous occasions to prevent the misuse of its name, emblem or flag for philatelic or commercial purposes. IMF recently filed an opposition to a trademark application in Canada in which the applicant sought to register the abbreviation "IMF". IMF has also asserted against private parties its exclusive entitlement to the use of the name "International Monetary Fund". On occasion, IFC has taken steps relating to the use of the initials "IFC" by others.

#### Section 12. Inviolability of archives and documents

69. No controversies regarding recognition of the inviolability of the archives and documents of the specialized agencies and of IAEA have been reported. IMF, however, notes that its staff members on mission carry an IMF briefcase for papers and documents. On a few occasions, customs officials have insisted on searching the briefcase even when informed of the inviolability of the organization's archives, and documents including codes have been examined. No documents, however, have been confiscated. IMF has protested these actions, and assurances have been received that such incidents would be avoided. Similarly, there have been some incidents of interference with IMF documents sent by private courier.

### <sup>18</sup> United Nations, *Treaty Series*, vol. 828, p. 305.

#### Section 13. Immunity from currency controls

- 70. Most of the specialized agencies and IAEA have encountered no legal problems regarding immunity from currency controls. In 1965, ILO informed the Government of Brazil that its office in Brazil should, by virtue of section 7 of the specialized agencies Convention, be exempt from a tax of 1 per cent on all exchange operations to which it had been subjected. The exemption was obtained. In 1979 and the following years, representations were made to the authorities in Ethiopia by agencies operating in the country (including ILO) concerning restrictive exchange laws.
- 71. Each investment agreement entered into by IFC requires that arrangements satisfactory to IFC be made for the remission to IFC or its assigns of all monies payable in respect of the investment. Particular agreements have been made between IFC and States with regard to repatriation rights and privileges in respect of investments made, or caused to be made, by IFC in enterprises in the States concerned. Problems have been encountered in India, Nigeria, Zaire and Zambia, where the receipt by IFC of dividends usually takes several months due to a shortage of foreign exchange. In Peru, there is delay, from time to time, in the repatriation of dividends in excess of varying stated percentages of the funds invested. In Brazil, interim dividends may not be repatriated until the company declaring the dividend closes its books for its fiscal year and its accounts are audited. Such interim dividends are invested in treasury bills. Interest on such investments may be repatriated with the approval of the central bank of the State concerned. Since 1978, similar problems have continued to arise from time to time. The specifics of each case have not been indicated by IFC in light of the rapidly evolving foreign exchange situation of many of the countries in respect of which such problems have arisen. IFC notes, however, that it must be borne in mind that paragraph 2 of annex XIII, relating to IFC, of the specialized agencies Convention, 19 provides that subsection 7 (b) thereof, concerning transfers of funds, gold or currency, shall apply to IFC subject to article III, section 5, of the IFC Articles of Agreement.

#### Section 14. Direct taxes<sup>20</sup>

72. Few controversies appear to have arisen concerning the immunity of the specialized agencies and IAEA from direct taxes, and when such controversies have arisen, they have normally been resolved satisfactorily. For example, income from FAO investments has sometimes been taxed, but the amounts withheld have been refunded. In addition, FAO reports that the Peruvian Government had imposed a tax on air fares and sojourn abroad with respect to residents of Peru, with no exception being made for residents travelling on behalf of the United Nations or its specialized agencies. The matter was taken up by UNDP and as a result exemption from the tax has now been accorded in the case of travel for the United Nations or its specialized agencies.

<sup>19</sup> Ibid., vol. 327, p. 326.

<sup>20</sup> See also sections 17 and 24 below.

In the case of WHO, a controversy arose in its Regional Office for South-East Asia when the Indian Government, in October 1971, imposed a tax on all international travel for which the fare was paid or was payable in local currency, as well as a tax on internal travel. WHO claimed exemption from these taxes by virtue of its local agreement with the Government of the host country. With regard to the tax on international travel, WHO obviated the difficulty by purchasing tickets in foreign currency, but it had to pay the other tax pending the conclusion of negotiations. After prolonged correspondence between WHO and the Indian authorities, the latter decided, in May 1972, to exempt the organization from these taxes. WHO claimed reimbursement of the taxes it had had to pay on internal travel and, after some correspondence, it obtained satisfaction.

- 73. WHO also states that a controversy arose in 1975 when the organization was required to pay certain taxes in Italy on the sale of a villa that had been bequeathed to it. The matter is still under negotiation with the Italian authorities. In 1975, a request for refund of stamp duty paid on the lease of the ILO Branch Office in London received a negative reply from the British Government on the grounds that exemption from such duty was not provided for in the specialized agencies Convention.
- 74. As provided in section 9 (a) of the specialized agencies Convention, the specialized agencies pay taxes corresponding to "charges for public utility services". Questions have arisen regarding the interpretation of that phrase. In 1966, the ILO Area Office in Beirut was requested to pay municipal taxes, inter alia, on telephone and electricity bills, as well as taxes on air tickets bought in the country. Discussions were held between UNRWA, on behalf of all United Nations agencies operating in the country, and the Lebanese Government. ILO reports that United Nations agencies seem to have been subsequently exempted from most of these taxes.
- 75. In 1975, the city of Bern requested UPU to contribute to the financing of the construction of a road adjacent to the land on which the present headquarters of UPU stands. This request was based on a practice whereby, in certain cases, the residents concerned participated in the cost of building a road. UPU claimed exemption under its Headquarters Agreement and the general practice followed at the headquarters of the other specialized agencies of the United Nations. Switzerland maintained its position, referring to longstanding practice in the country with respect to such contributions and to Swiss legal theory, and did not agree that the contribution represented a tax from which UPU was exempt under its Headquarters Agreement. In its view, what was involved was not a tax but a "preferential charge" (charge de préférence, Vorzuglasten), comparable to the utility charges for water, electricity, gas, etc. UPU questioned such a view in a letter dated 27 January 1977, which was accompanied by an opinion by the United Nations Legal Counsel. There have been no further developments in the matter.
- 76. ILO reports three cases arising in the host country. An annual autoroute tax came into effect in Switzerland in 1985. The Swiss Government decided that there would be no immunity from payment for residents en-

joying financial privileges and immunities on the grounds that the tax was in the nature of a payment for services rendered. No disagreement with this view was expressed by United Nations agencies in Geneva, including ILO. In 1981/82, the Geneva authorities required a university tax to be paid by international officials whose child or spouse attended the University of Geneva. For these purposes, officials were treated in the same way as Swiss citizens domiciled in a canton other than Geneva and subjected to a more modest tax than that applying to other non-Swiss residents. The Swiss authorities took the view that this was in effect a charge for services rendered and, while the United Nations agencies, including ILO, did not fully accept this view, they did not challenge the decision in view of the small amount involved and as a gesture of goodwill towards the University of Geneva. In 1983, payment of a similar tax was required, after non-collection of the tax from international officials for many years, for secondary schools in Geneva, except in the case of nationals of countries with reciprocal exemption agreements with Switzerland. The Swiss Government justified the tax as a charge for services rendered. United Nations agencies, including ILO, expressed serious reservations as to the charge, which was not however withdrawn.

#### Section 15. Customs duties

- (a) Imports and exports by the specialized agencies and by IAEA "for their official use"
- The question whether a given item has been imported or exported "for ... official use" has rarely given rise to difficulties. Where difficulties have arisen, they have usually been resolved by communication with the appropriate officials. WHO reports that the customs authorities in the United Kingdom refused to allow a consignment addressed to it to enter the country without an attestation that the consignment was the property of WHO and intended for use in connection with the functions of the organization. The attestation was provided and the consignment was allowed to enter the country. ILO reports that, in a case involving machinery purchased with funds provided by a private foundation for use at a Turkish institution beneficiary of a project of which the ILO was executing agency, it was considered that the element of official use was absent and that there were no grounds for exemption.
- FAO states that in early 1982 the Ministry of Finance of the host country, Italy, initiated the practice of issuance of import licences, which affected equipment and materials required for FAO purposes. The Ministry interpreted the relevant section of the FAO Headquarters Agreement in such a manner as to justify the Government's making a determination in each case whether the equipment and materials imported by FAO were for official use and the quantities reasonable. Such a procedure implied the right of the Government to deny duty-free importation of equipment and material. Although in no case has an import licence been categorically refused, inordinate delays have been experienced which in some cases have caused extra costs to FAO in the form of demurrage or the necessity to buy supplies locally at higher prices. The matter was the sub-

ject of discussion at the eighty-sixth session of the FAO Council, in November 1984. At that time the representative of the host country stated that his Government recognized that the organization was entitled, in accordance with the relevant section of the Headquarters Agreement, to import and export all the equipment and materials it required for official purposes without limitation as to quantity or nature, and that further efforts were being made to convince the Ministry of Finance to accept this interpretation. It would appear that the problem is now in the process of resolution.

### (b) Imposition of customs duties, prohibitions and restrictions

- 79. As a general rule, customs duties, prohibitions and restrictions have not been imposed on official imports and exports. In one case involving FAO, a Government placed restrictions on the importation and transportation of ammunition for harpoon guns. FAO took no action in the matter, considering the restrictions to be reasonable as normal security measures for the control of explosives.
- 80. Most specialized agencies and IAEA have found it unnecessary to enter into any standard arrangements with respect to non-imposition or automatic refund of customs duties.
- 81. The practice of WHO, however, has been to include in all host agreements provisions granting to the organization complete exemption from customs, statistical and similar duties on all goods for its official use, imported or exported. Where customs duties are levied in the form of purchase or turnover taxes, these are reimbursed to WHO under administrative arrangements concluded with the States concerned.
- 82. For all materials imported by UPU, arrangements have been made for formalities to take place at Bern, with reception at the frontier. This procedure was also followed for the import of certain materials used for the construction of the building which now houses the head-quarters of UPU.
- 83. IBRD, IDA and IFC have occasionally made requests for refunds of customs duties and have successfully secured them. Similarly, in the case of IAEA, if customs duties should be charged by error, refund is secured within a period of approximately six weeks. However, no refund is possible regarding customs duties which have already been paid by the importer. FAO has taken no action is cases where there was a possibility that the purchase price of equipment procured in the field from local suppliers included import duties.

### (c) Sales of articles imported by the specialized agencies and by IAEA

84. Some specialized agencies, such as IBRD, IDA and IFC, have on occasion made arrangements with Governments on an *ad hoc* basis regarding sales of imported articles. In the case of FAO, in the execution of field projects, imported equipment is sometimes sold within a country, also under an *ad hoc* arrangement with the Government concerned. Expendable goods—such as food aid provided by WFP, or fer-

tilizers supplied under the FAO International Fertilizer Supply Scheme—are sometimes supplied to Governments as grants, but with a view to sale. The agreements concerned normally provide that the Government concerned shall grant exemption from, or bear the cost of, any customs duties, levies and charges on such commodities and that the proceeds from sales shall be deposited in a special account, to be used for a related development activity to be agreed between the Government and FAO.

- Some agencies, however, have made standing arrangements with Governments for the resale of imported items under certain agreed conditions. The practice followed by WHO is that, in the case of the import of articles, whether for official use or otherwise, customs exemption is subject to an understanding that articles imported under customs franchise will not be sold in the country into which they were imported except under conditions agreed with the Government of the country. In all WHO offices, arrangements have been made concerning the resale of articles imported duty free, whether by WHO or its staff members. No difficulties have been encountered except in the WHO Regional Office for Western Asia, in the Philippines, where in two cases staff members had to pay customs duties on cars that had been imported duty free but had been sold three years later, in accordance with the relevant section of the agreement with the host country. It was considered that, by requiring payment of these duties, the authorities of the host country were departing from the practice followed in that regional office since the conclusion of the agreement. The matter is still pending.
- 86. The standing arrangement of ICAO with the host country, Canada, in regard to sales of imported articles is incorporated in its Heaquarters Agreement. Section 7 of that Agreement reads:

When goods are purchased under appropriate certificates from manufacturers or wholesalers who are licensed under the Excise Tax Act, the organization should be eligible to claim for the remission or refund of the excise tax and/or consumption or sales tax for goods imported or purchased in Canada for the official use of the organization as a body, provided, however, that any article which is exempted from these taxes, other than publications of the organization, shall be subject thereto at existing rates if sold or otherwise disposed of within a period of one year from the date of purchase, and the vendor shall be liable for such tax.

87. IAEA has entered into an arrangement with Austria allowing duty-free disposal of goods imported by the agency two years after their import. This two-year period also applies to automobiles. IAEA has also made arrangements with Italy in connection with the International Centre for Theoretical Physics at Trieste, whereby automobiles may be sold free of tax after four years.

#### Section 16. Publications

88. Most of the specialized agencies and IAEA have encountered no problems as to the interpretation of the term "publications". The term has in practice been understood to include films, photographs, prints and recordings (prepared as part of an organization's public information programme and exported or imported for

exhibition or broadcasting), as well as books, periodicals and other printed material. Since the exemption accorded in respect of publications is an exemption from customs duties, prohibitions and restrictions, no import or export licences have been required. However, in some cases it is required that customs clearance forms accompany the material.

- The FAO Headquarters Agreement contains a provision similar to that in section 19 (b) of the specialized agencies Convention, but with the explanation, in section 19 (c) of the Agreement, that the term "articles" includes "publications, still and moving pictures, and film and sound recordings". Similar provisions are included in FAO regional office agreements. FAO reports that, while no controversies have arisen concerning the scope of the term "publications", it has encountered difficulties in the application of the relevant provisions and also of the UNESCO Agreement for Facilitating the Circulation of Visual and Auditory Materials of an Educational, Scientific and Cultural Character.21 For example, some countries impose import duties on the organization's publications and documents, and the circulation of books, film strips and microfiches is sometimes hampered by restrictions or by long delays in customs clearance.
- 90. IBRD and IDA state that there have sometimes been difficulties with regard to the requirement that customs clearance forms should accompany materials in transit. The difficulties that have arisen relate to films, especially at the United States-Canadian border, where officials do not always act uniformly. Some prints have been lost while crossing international borders. This is thought to be due to the fact that authorities at certain ports of entry have been instructed to keep film from entering and they fail to discriminate between IBRD film, which is immune, and other types of film which are not.
- 91. In connection with the dispatch and receipt of films, FAO states that, with the exception of those forwarded by FAO pouch, films are covered by a blanket export/import permit which has to be reissued to the organization each year by the host country. Exhibits sent outside the host country, however, are subject to specific permits.

### Section 17. Excise duties and taxes on sales; important purchases

- (a) Excise duties and taxes on sales forming part of the price to be paid
- 92. Section 10 of the specialized agencies Convention provides that the specialized agencies will not, as a general rule, claim exemption "from excise duties and from taxes on the sale of movable and immovable property, which form part of the price to be paid". The words "excise duties" and "taxes" are deemed to include turnover taxes, designated differently in various States but often known as VAT. A number of agencies report that the terms of their respective headquarters agreements contain different language, often providing

for exemptions broader than those envisaged in section 10 of the Convention. IMF has claimed exemption from payment of manufacturer's excise taxes, stamp taxes, recording taxes, air ticket taxes and head taxes, as well as sales taxes.

93. The experience of FAO has been that the interpretation of the terms "excise duties and . . . taxes . . . which form part of the price to be paid", in section 10 of the specialized agencies Convention, varies from country to country. No uniform definition or interpretation has so far been devised or applied. The provisions of sections 19 (b) of the FAO Headquarters Agreement are fairly comprehensive. They read:

Regarding indirect taxes, levies and duties on operations and transactions, FAO shall enjoy the same exemptions and facilities as are granted to Italian governmental administrations. In particular, but without limitation by reason of this enumeration, FAO shall be exempt from the registration tax (imposta di registro); the general receipts tax (imposta generale sull'entrata) on wholesale purchases, on contractual services and on tenders for contractual supplies (prestazioni d'opera, appalti), on leases of lands and buildings; from the mortgage tax; and from the consumption taxes on electric power for lighting, on gas for lighting and heating, and on building materials.

In 1972, the host country, Italy, introduced VAT to replace the turnover tax. For many years, FAO had unsuccessfully sought exemption from that tax on all transactions in respect of goods and services procured by the organization pursuant to section 19 (b) of the Headquarters Agreement. FAO made a similar claim in 1972, when VAT was introduced. Finally, a decree was issued on 2 July 1975 which, read in conjunction with the basic legislation of 1972, expressly stipulated exemption on all transactions exceeding 100,000 Italian lire. As small purchases of goods and services could be grouped together, FAO decided not to insist on exemption for invoices of less than 100,000 Italian lire.

94. Section 16 (a) of the Agreement regarding the African Regional Office of FAO<sup>22</sup> at Accra, Ghana, contains somewhat different provisions on the subject of exemption from indirect taxes:

FAO shall be exempt from levies and duties on operations and transactions, and from excise duties, sales and luxury taxes and all other indirect taxes when it is making important purchases for official use by FAO of property on which such duties or taxes are normally chargeable. However, FAO will not, as a general rule, claim exemption from excise duties, and from taxes on the sale of movable and immovable property which form part of the price to be paid, and cannot be identified separately from the sale price.

95. ITU pays neither duties nor taxes, but pays the turnover tax included in the price of items purchased. Article 2, paragraph 2, of the Arrangement made for the execution of the Agreement between the Swiss Federal Council and ITU concerning the legal status of the organization in Switzerland<sup>23</sup> provides:

With regard to federal turnover tax, however, whether included in the price or patently transferred, the exemption shall apply only to purchases intended for the Union's official use and provided that the amount invoiced for one and the same purchase exceeds 100 Swiss francs.

96. In section 6 of the UPU Headquarters Agreement, the reference is not to "excise duties and . . . taxes on

<sup>&</sup>lt;sup>21</sup> United Nations, Treaty Series, vol. 197, p. 3.

<sup>&</sup>lt;sup>22</sup> See Report of the FAO Conference, Tenth Session, Rome, 31 October-20 November 1959, resolution 75/59, annex D.

<sup>&</sup>lt;sup>23</sup> United Nations, *Juridical Yearbook 1971* (Sales No. E.73.V.1), p. 33.

the sale of movable and immovable property which form part of the price to be paid", as in section 10 of the specialized agencies Convention, but instead to "indirect taxes or sales taxes included in the price of movable or immovable property". The application of this provision of the Agreement has not given rise to difficulties. It has been agreed that UPU would not ask for reimbursement of indirect taxes amounting to less than 100 Swiss francs.

97. None of the specialized agencies nor IAEA has encountered difficulties in determining whether or not excise duties and taxes "form part of the price to be paid" on the sale of property. In most cases, such duties and taxes are readily identifiable and are stated separately from the purchase price.

#### (b) Important purchases

- 98. The question of what constitutes an important purchase for the purposes of section 10 of the specialized agencies Convention has arisen in connection with a number of organizations. In the case of ILO, the amount of what constitutes an important purchase is at present 105 Swiss francs; for IAEA, the minimum total sum of the invoice on which VAT remission may be claimed is 1,000 Austrian schillings, exclusive of VAT.
- WHO notes that the term "important purchase" is nowhere specifically defined, nor has it acquired any standard and uniform interpretation, although prima facie any purchase made by the organization for its official use may be deemed "important". WHO takes the view that what constitutes an "important purchase" can in practice be stated only in monetary terms. In an exchange of letters concerning the interpretation and implementation of the 1955 Headquarters Agreement between WHO and the host country (Denmark) of a regional bureau, the term "minor purchases" was defined as purchases the amount of which did not exceed 200 Danish kroner. By implication, purchases in excess of such a sum are "important", and WHO is thus entitled to remission or return of the amount of duty or tax paid on such purchases. The principle of tax reimbursement also applies in Switzerland in respect of purchases exceeding 100 Swiss francs; in the Congo, where the amount involved may not be less than CFA 10,000; and in France, where purchases entail the collection of a turnover tax of at least 250 French francs. Thus it would seem that purchases made in Switzerland, the Congo and France exceeding the amounts mentioned above constitute "important purchases". An exchange of letters between WHO and the United Kingdom on the application of section 10 of the specialized agencies Convention to goods and services purchased by WHO in the United Kingdom uses the expression "considerable quantities of goods or services" and interprets the expression as goods or services the aggregate cost of which is at least £50 sterling per claim. Applications for refunds will be considered where the aggregate cost exceeds such an amount.
- 100. For a number of agencies, section 10 of the specialized agencies Convention has not given rise to questions of interpretation. In the case of ICAO, IBRD, IDA, IFC and IMF, no distinction is made as to the

amount of purchases or whether purchases are important or not. FAO and UNESCO report that their host country agreements do not limit exemption from indirect taxation to important purchases.

- 101. IAEA reports that, in the case of goods delivered for the IAEA commissary, the turnover tax is reimbursed for foodstuffs, alimentary products and tobacco products. Reimbursement of turnover tax for other goods is made only if such goods have been exempted from import duties in accordance with the provisions of its Headquarters Agreement and the relevant supplemental agreements and if appropriate evidence thereof can be furnished.
- 102. UNESCO notes that the UNESCO commissary is an integral part of the organization's secretariat and is operated under the authority of the Director-General in accordance with the regulations of the commissary and the appropriate procedures established for the various services of the organization. The employees of the commissary are not governed by the organization's staff regulations and consequently are not entitled to the privileges, immunities and facilities accorded to UNESCO staff members under its Headquarters Agreement. The finances of the commissary are governed by the financial regulations of UNESCO and the commissary's own financial regulations and rules. All staff members of UNESCO and all other employees of the organization at headquarters are eligible to participate in and benefit from the facilities of the commissary, subject to payment of a deposit in an amount determined by the Director-General on the recommendation of the general assembly of the commissary. Assimilated personnel, such as retired employees of UNESCO, staff members of the United Nations and the specialized agencies stationed in Paris, or staff members of permanent delegations officially accredited to UNESCO may, at the discretion of the Director-General and subject to payment of the required deposit, be permitted to make purchases at the commissary. At the discretion of the Director-General, temporary permission may be granted to persons temporarily at headquarters such as field staff members, consultants and members of delegations to the General Conference. Goods sold at the commissary are not acquired duty free or imported tax free. The commissary is simply a co-operative shop.
- As noted above (p. 194, para. 93), the Headquarters Agreement between FAO and Italy does not limit exemption from indirect taxation to "important" purchases. The FAO commissary, which is part of the organization, was established on the basis of an exchange of letters between the Italian Government and FAO, pursuant to section 27 (j) (ii) of the Headquarters Agreement, under which officers of FAO have the right to import, free of duty "through the medium of FAO, reasonable quantities . . . of foodstuffs and other articles for personal use and consumption". The organization is responsible for ensuring the appropriate administration and distribution of the duty-free items provided. The entitlements of the staff are set out in the FAO Administrative Manual. The Government of the host country, Italy, establishes yearly quotas with respect to various categories of foodstuffs and other items on the basis of the number of entitled staff at FAO headquarters.

104. Since the beginning of 1984, FAO has experienced difficulties and delays in the issuance by the Italian Government of duty-free import licences for the benefit of the staff, and there was delay in the issuance of licences for 1984. It was stated, moreover, that, after 1 January 1985, FAO staff members of Italian nationality would no longer be granted the duty-free import privileges that had been extended to them, at the initiative of the host Government, since 1971. The dutyfree privileges accorded to non-Italian staff, especially in respect of tobacco, alcoholic beverages and petrol, were also called into doubt. As a result of the withdrawal or reduction of these privileges, FAO would incur extra costs in the form of upward adjustments of staff remuneration, since some duty-free privileges are taken into account in the calculation of such remuneration. The FAO Council, at its eighty-sixth session, in November 1984, expressed concern about the matter, particularly with regard to the additional costs to all member States if the privileges accorded to FAO staff and taken into account in establishing the levels of staff remuneration were reduced, and opposed reduction of the privileges accorded to non-Italian staff since the organization's transfer to Rome in 1951. The Council urged the Italian Government to take into consideration the financial and other implications of any reduction of privileges on the FAO budget and the considerable benefits to the local economy deriving from the presence of the organization in Italy. The Council unanimously adopted resolution 4/86 to that effect.24 As of this date, the Italian Government has not withdrawn the duty-free privileges accorded to FAO officials of Italian nationality, and the import quotas for various items are currently under discussion.

### (c) Remission or return of taxes paid

105. The specialized agencies and IAEA have made administrative arrangements with most of the States in which they are active for the remission or return of the amount of duties not payable. Except for occasional delay in the receipt of such remission or return, these administrative arrangements appear to work satisfactorily. Some examples are given below.

In 1974, the ILO Branch Office in Brussels was transformed into a Liaison Office with the European Communities and the BENELUX countries. Treatment in regard to exemption from VAT is the same as that formerly accorded to the Branch Office. The amount of tax is deducted directly by suppliers from all invoices except those for stationery and office supplies amounting to less than 5,000 Belgian francs. Although France is not party to annex I of the specialized agencies Convention, concerning ILO, in accordance with a decision by the Ministry of Foreign Affairs (March 1967), the Paris Branch Office of ILO obtains reimbursement of VAT paid on all purchases of goods and services except those relating to the construction and maintenance of premises, provided that the tax amounts to 250 French francs or more.

107. WHO has made administrative arrangements, in the form of an exchange of letters, with the Swiss auth-

orities under which the Federal Tax Administration reimburses the organization the sums levied as taxes on purchases exceeding 100 Swiss francs made by the organization for its official use. To facilitate implementation of the arrangements, statements serving as a basis for reimbursement are submitted periodically (every month or at longer intervals) to the Swiss federal tax authorities.

108. ITU submits a half-yearly statement, including a copy of all invoices of purchases exceeding 100 Swiss francs, to the Swiss federal tax authorities for remission of taxes paid. No difficulties have arisen concerning this arrangement.

109. IAEA has made administrative arrangements with the Government of the host country, Austria, to submit every six months a list of invoices paid by the organization and for which a claim for refund of turnover tax is made. The arrangements work satisfactorily, although sometimes with delay.

110. IBRD, IDA and IFC report that, except in Belgium, where they have been granted full exemption from payment of VAT, they have made administrative arrangements with member States under which they pay the tax but are reimbursed upon presentation of the relevant invoices to the appropriate authorities.

111. IMF notes that no special arrangements in this regard have been made and that it has encountered only occasional problems, principally with taxes on air transportation tickets.

The UNESCO Headquarters Agreement provides for the exemption of the organization from indirect taxes which form part of the cost of goods sold and services rendered. Prior to 1967, regardless of the importance of the purchase or transaction and of the provisions of the Headquarters Agreement, the exemption was obtained not by later reimbursement of the tax levied but at the time of the purchase or transaction, when the supplier was authorized upon receipt of a written declaration from the organization to exempt the sale or transaction from domestic taxes. This procedure had been set out in an exchange of letters and had produced excellent results. In 1967, however, the Government of the host country, France, decided to change this procedure, but without questioning the terms of the agreement whereby the organization received a reimbursement of the turnover taxes for all purchases, remuneration of services or transactions effected for its official use; in particular, for construction work and improvements at headquarters. In July 1967, following a decision of the UNESCO Executive Board and an exchange of letters constituting an agreement, an arrangement was established whereby the organization would be reimbursed for all indirect taxes concerning transactions undertaken for its official use which formed part of the cost of goods sold to it, services rendered to it and transactions involving movable or immovable property, including construction work. For this purpose, UNESCO sends to the Ministry of Foreign Affairs each month a request for reimbursement of tax, enclosing the suppliers' invoices relating to expenditure incurred during the preceding month and a statement of the expenditure. Each month, the Ministry of Economy and Finance makes an advance to the organization in

<sup>&</sup>lt;sup>24</sup> FAO, Report of the Council of FAO, Eighty-sixth Session, Rome 19-30 November 1984, para. 206.

anticipation of the amount of such taxes. This advance is reconciled each month with the amount actually spent. Thus the arrangement currently in force may be considered as tax exemption by reimbursement of the amounts levied, in which the reimbursement is effectively made before the expenditure is incurred.

#### CHAPTER III

### Privileges and immunities of the specialized agencies and of the International Atomic Energy Agency in respect of communication facilities

## Section 18. Treatment equal to that accorded to Governments in respect of mails, telegrams and other communications

113. For the most part, the specialized agencies have reported no particular difficulties in the application of section 11 of the specialized agencies Convention, which provides that the agencies shall enjoy treatment equal to that accorded to Governments in respect of mails, telegrams and other communications. As regards telegrams and telephone calls, however, the discrepancy subsists between section 11 and annex IX of the Convention, relating to the International Telecommunication Union, which does not provide for such equal treatment.<sup>25</sup> As of 1 June 1985, eight Governments had declared their inability to comply fully with the provisions of section 11 until such time as all other Governments had decided to co-operate in granting such treatment to the agencies.<sup>26</sup>

The ITU Plenipotentiary Conference held in Nairobi in 1982 adopted resolution No. 40 entitled "Possible revision of article IV, section 11, of the Convention on the Privileges and Immunities of the Specialized Agencies".27 By that resolution, the ITU Plenipotentiary Conference of 1982 resolved inter alia to confirm the decisions of the Plenipotentiary Conferences of Buenos Aires (1952), Geneva (1959), Montreux (1965) and Malaga-Torremolinos (1973) not to include the heads of the specialized agencies among the authorities listed in annex 2 of the International Telecommunication Convention28 as entitled to send government telegrams or to request government telephone calls, and expressed the hope "that the United Nations will agree to reconsider the matter and, bearing in mind the above decision, will make the necessary amendment to article IV, section 11, of the Convention on the Privileges and Immunities of the Specialized Agencies". The Plenipotentiary Conference instructed the ITU Administrative Council to take the necessary steps with the appropriate organs of the United Nations with a view to reaching a satisfactory solution.

115. IBRD, IDA and IFC report that difficulties have sometimes been encountered when claiming preferential

### Section 19. Use of codes and dispatch of correspondence by courier or in bags

116. None of the specialized agencies nor IAEA reports having experienced any problem concerning the interpretation of the terms "correspondence" and "other official communications" appearing in the first paragraph of section 12 of the specialized agencies Convention. The specialized agencies and IAEA also state that they are not aware of any censorship by State authorities being applied to their official correspondence and communications.

117. Generally, recognition has always been given to the rights and related immunities and privileges referred to in the second paragraph of section 12 of the specialized agencies Convention, namely, "the right to use codes and to dispatch and receive correspondence by courier on in sealed bags, which shall have the same immunities and privileges as diplomatic couriers and bags". FAO, however, reports that in one case a Government refused to recognize diplomatic immunity with respect to the FAO pouch. The matter was later resolved following intervention by UNDP. IMF states that it has taken measures to ensure that its property, correspondence, etc. are clearly identifiable as pertaining to the Fund, together with a clearly displayed statement of the Fund's privileges and immunities under its Articles of Agreement.

118. Most specialized agencies and IAEA have not formally adopted "appropriate security arrangements" as envisaged in the third paragraph of section 12 of the Convention. In accordance with airport security regulations, FAO pouches arriving from certain points have been subject to X-ray examination.

119. A number of WHO agreements, however, are subject to the condition that they shall not derogate from or abridge the right of the Government of the host country to take the precautions necessary to protect the security of the State. State authorities are none the less

cable rates. As this might have been due to lack of adequate identification, for several years after 1965 IBRD issued credit/identification cards to staff members going on mission. This proved to be administratively unworkable and was discontinued. It should be noted that, in States not parties to the specialized agencies Convention but parties to the Articles of Agreements of IBRD, IDA and IFC, the relevant provisions of the respective Articles of Agreement apply.

<sup>25</sup> United Nations, Treaty Series, vol. 79, p. 326.

<sup>&</sup>lt;sup>26</sup> Multilateral Treaties deposited with the Secretary-General . . . (see p. 184, footnote 5 above) and Supplement (Sales No. E.84.V.4), chap. III.2.

<sup>&</sup>lt;sup>27</sup> ITU, International Telecommunication Convention, Nairobi, 1982 (Geneva), p. 293.

<sup>28</sup> Ibid., p. 147.

obliged, whenever they deem it necessary to adopt measures for the protection of security, to approach WHO as rapidly as circumstances allow to determine by mutual agreement the measures required to ensure such security. Likewise, WHO is required to collaborate with the authorities of the host countries to avoid any prejudice to security that might be occasioned by the organization's activities.

120. Section 40 of the ICAO Headquarters Agreement provides that nothing in the Agreement "shall be construed as in any way diminishing, abridging, or weakening the right of the Canadian authorities to safeguard the security of Canada, provided the organization shall be immediately informed in the event that the Canadian Government shall find it necessary to take any action against any person enumerated in the Agreement".

#### CHAPTER IV

### Privileges and immunities of officials of the specialized agencies and of the International Atomic Energy Agency<sup>29</sup>

Section 22. Categories of officials to which the provisions of articles VI and VIII of the Convention on the Privileges and Immunities of the Specialized Agencies apply

121. Most specialized agencies and IAEA send to member States, on a periodic basis, a list of officials to which the provisions of article VI ("Officials") and ar-

<sup>29</sup> On 17 December 1980, the General Assembly, on the recommendation of the Fifth Committee, adopted resolution 35/212 entitled "Respect for the privileges and immunities of officials of the United Nations and the specialized agencies". That resolution reads in part:

"The General Assembly,

". .

"Recalling that, under Article 105 of the Charter, officials of the Organization shall enjoy in the territory of each of its Member States such privileges and immunities as are necessary for the independent exercise of their functions in connection with the Organization, which is indispensable for the proper discharge of their duties.

"Realizing that staff members of the specialized agencies enjoy similar privileges and immunities,

"Mindful of the Convention on the Privileges and Immunities of the United Nations of 13 February 1946 and of the Convention on the Privileges and Immunities of the Specialized Agencies of 21 November 1947,

"Concerned about reports alleging that the privileges and immunities of officials of these organizations have been encroached upon.

"1. Appeals to all Member States to respect the privileges and immunities accorded to officials of the United Nations and the specialized agencies by the Convention on the Privileges and Immunities of the United Nations of 13 February 1946 and by the Convention on the Privileges and Immunities of the Specialized Agencies of 21 November 1947;

"2. Requests the Secretary-General to bring the present resolu-

"2. Requests the Secretary-General to bring the present resolution to the attention of all organs, organizations and bodies of the United Nations system with the request to furnish information on cases in which there are clear indications that the status of the staff members of such organizations has not been fully respected;

"3. Requests the Secretary-General to submit, on behalf of the Administrative Committee on Co-ordination, a report to the General Assembly containing any cases in which the international status of the staff members of the United Nations or of the specialized agencies has not been fully respected."

On 18 December 1981, the General Assembly, on the recommendation of the Fifth Committee, adopted resolution 36/232 entitled "Respect for the privileges and immunities of officials of the United Nations and the specialized agencies and related organizations". That resolution reads in part:

"The General Assembly,

". .

"Recalling the Convention on the Privileges and Immunities of the United Nations of 13 February 1946, the Convention on the Privileges and Immunities of the Specialized Agencies of ticle VIII (Laissez-passer) of the specialized agencies Convention apply. For example, IBRD, IDA and IFC periodically notify the Secretary-General of the United

- 21 November 1947, the Agreement on the Privileges and Immunities of the International Atomic Energy Agency of 1 July 1959 and the agreements between the United Nations and the specialized agencies and related organizations and the respective host Governments,
- "Noting also the position consistently upheld by the United Nations in the event of the arrest and detention of United Nations-staff members by governmental authorities,
- "Mindful of Article 100 of the Charter of the United Nations, under which each Member State has undertaken to respect the exclusively international character of the responsibilities of the Secretary-General and the staff and not to seek to influence them in the discharge of their responsibilities,

"Mindful also of the fact that, under the same Article of the Charter, the Secretary-General and the staff shall not, in the performance of their duties, seek or receive instructions from any Government or from any other authority external to the Organization,

"Recalling that the International Court of Justice has held that international organizations have the power and responsibility to protect members of their staff,

"Recalling also the obligations of the staff in the conduct of their duties to observe the laws and regulations of Member States,

"Reaffirming the relevant staff regulations,

"Aware of the absolute necessity that staff members be enabled to discharge their tasks as assigned to them by the Secretary-General without interference on the part of any Member State or any other authority external to the Organization,

"Realizing that staff members of the specialized agencies and related organizations enjoy similar privileges and immunities in accordance with the instruments mentioned in the second preambular paragraph above,

"1. Appeals to any Member State which has placed under arrest or detention a staff member of the United Nations or of a specialized agency or related organization to enable the Secretary-General or the executive head of the organization concerned, in accordance with the rights inherent under the relevant multilateral conventions and bilateral agreements, to visit and converse with the staff member, to apprise himself of the grounds for the arrest or detention, including the main facts and formal charges, to enable him also to assist the staff member in arranging for legal counsel and to recognize the functional immunity of a staff member asserted by the Secretary-General or by the appropriate executive head, in conformity with international law and in accordance with the provisions of the applicable bilateral agreements between the host country and the United Nations or the specialized agency or related organization concerned;

organization concerned;
"2. Requests the Secretary-General and the executive heads of the organizations concerned to ensure that the staff observe the obligations incumbent upon them, in accordance with the relevant staff rules and regulations, the Convention on the Privileges and

Nations and the Governments of all States that have acceded to the Convention, each for its own organization, of the categories of officials to which the provisions of articles VI and VIII of the Convention shall apply. Each such list contains the names of all executive directors, alternate executive directors and all officials of each organization. IBRD, IDA and IFC additionally make specific notification to member States in individual cases as required.

- 122. IMF has a similar practice whereby a list of the members of the Executive Board, officers and staff is sent to States parties to the Convention with a letter of transmittal.
- 123. ITU sends a list of all staff members on 1 January of each year to the Governments of member States.
- 124. ICAO and IAEA issue once a year a staff list which is circulated to the Governments of member States.
- 125. In the case of FAO, upon request, the Government of member States are annually sent computer listings containing the names of their nationals who worked for FAO during the previous year. This information can be furnished more than once a year if requested. In accordance with instructions contained in the FAO Manual, appointments are communicated to certain Governments for information or clearance.
- 126. UPU transmits to the Department of Foreign Affairs of the Swiss Confederation a list of all new staff members engaged on a permanent or temporary basis.
- 127. The practice of IAEA is to inform the Government of Austria of every arrival and departure of agency staff immediately.
- 128. The Twelfth World Health Assembly approved in May 1959, by resolution WHA 12.41, <sup>30</sup> granting of the privileges and immunities referred to in articles VI and VIII of the specialized agencies Convention to all WHO officials, with the exception of those recruited locally and paid at hourly rates. In practice, therefore, WHO officials who enjoy the benefit of these articles of the Convention are those who occupy posts subject to international recruitment and locally recruited staff who are not paid at hourly rates.

Immunities of the United Nations, the Convention on the Privileges and Immunities of the Specialized Agencies and the Agreement on the Privileges and Immunities of the International Atomic Energy Agency:

- Agency;
  "3. Requests the Secretary-General to bring the present resolution to the attention of all specialized agencies and related organizations of the United Nations system, with the request that they furnish information to him on cases where there are clear indications that the principles expressed in paragraph 1 above or the status of the staff members of such an organization have not been fully respected:
- respected;
  "4. Requests the Secretary-General to submit to the General Assembly at each regular session, on behalf of the Administrative Committee on Co-ordination, an updated and comprehensive annual report relating to cases in which the Secretary-General or the competent executive head has not been able to exercise fully his responsibility in respect of the protection of staff members of the United Nations or of a specialized agency or related organization in accordance with the multilateral conventions and applicable bilateral agreements with the host country."
- <sup>30</sup> WHO, Handbook of Resolutions and Decisions of the World Health Assembly and the Executive Board, vol. I, 1948-1972 (Geneva, 1973), p. 356.

- 129. With regard to technical assistance experts, Governments have for the most part recognized their status as staff members. ILO, however, reports that Indonesia declared by decree, in 1981, that no privileges or immunities might be enjoyed by persons employed on projects financed from funds other than foreign grants. Discussions on the matter are proceeding between the Indonesian Government and the United Nations agencies.
- 130. When technical assistance is provided by IMF, at the request of Governments of member States, through employment of outside experts on a contractual basis, the Fund requests the Government concerned for written assurance that the expert will be accorded at least the same privileges and immunities as are granted to staff members under the Fund's Articles of Agreement. In some instances, problems have arisen when the Government concerned does not provide such assurance promptly and the programme is thereby delayed.

### Section 23. Immunity of officials in respect of official acts

- 131. The legal controversies that have arisen concerning the immunity of officials in respect of their official acts have related mainly to the question of what constitutes "acts performed by them in their official capacity", as provided in section 19 (a) of the specialized agencies Convention. In the few cases that have arisen, neither the specialized agencies nor IAEA have accepted or referred to municipal law definitions of "official acts".
- 132. ILO reports having experienced some problems concerning recognition of immunity for locally recruited personnel in respect of their official acts. One case concerned a motor vehicle accident in Bangladesh involving an ILO driver on official duty. The ILO position was that immunity should be claimed from criminal jurisdiction while ensuring that civil claims would be met. However, the UNDP Resident Representative on the spot considered that such a position would not be appropriate. In that connection, ILO drew attention to the constant practice in Switzerland, where immunity from jurisdiction was always recognized in cases of traffic offences committed in the exercise of official functions.
- 133. FAO states that one controversy of a legal nature has arisen with respect to section 19 (a) of the Convention. The FAO project manager and another staff member working on a project in Kenya had been summoned to appear as witnesses in criminal proceedings against a person who had been assigned to the project and had been charged with an offence relating to the use of counterpart funds contributed by Governments participating in the project. Since the organization had waived the immunity from legal process of the officials concerned, the controversy between the Government and FAO was one of principle only. It concerned the question whether or not section 19 (a) was applicable. The Government considered that there was no need for it to request a waiver of immunity since the officials were not being required to testify with respect to "acts performed by them in their official capacity". In its opi-

nion, where a privileged person saw a crime being committed, this could not be said to be an act performed by him in the course of his official duties, since it was not part of his official duties to witness the commission of crimes. The organization's position, with which the Legal Counsel of the United Nations concurred, was that immunity from legal process under section 19 (a), which was granted to officials in the interests of the organization and not for the personal benefit of the individuals concerned extended to all forms of legal process which were in any way connected with the performance by an official of his official functions, regardless whether proceedings had been brought against him or against a third party. The Government concerned reserved its position on the organization's understanding of "official acts". In a recent case, two officials of FAO made a special appearance and successfully invoked their immunity before a court of the host country when requested to give evidence concerning the salary of a staff member in a case involving a lease. While not disputing the immunity, the Government subsequently informed the organization of its view that the principle of secrecy should cover acts of the organization other than purely administrative acts relating to payments made to staff members. No request for waiver of immunity was made. FAO also reports that there have been problems arising out of the arrest of staff members where the Government concerned has maintained that the arrest arose out of non-official acts.31

- 134. WHO reports that controversies have arisen in some of its regional offices where staff members have been subject to prosecution or civil suits on grounds determined unilaterally by the authorities of the host country as not connected with the exercise by the staff member concerned of official duties. While WHO would not invoke immunity in circumstances where this would not be justified under the terms of the specialized agencies Convention or of the headquarters agreement, it considers that the organization must be in a position, if appropriate, to invoke immunity in cases where it considers that the staff member was acting in exercise of official duties.
- 135. UPU reports that, while its officials have generally not encountered difficulties, in 1967 an adviser travelling on home leave with his family was detained and held without valid reason for over three months in a country to which the aircraft in which he was travelling had been re-routed because of atmospheric disturbances. UPU representations to the authorities of the country concerned emphasized that travel on home leave was equivalent to a mission and that the authorities had therefore detained the official and his family illegally.
- 136. UNESCO reports that one of its senior officials was arrested in his home country and condemned to three years of imprisonment despite several protests and

requests for his release by the Director-General and the Executive Board. The immunity of this official was not waived and no request to that effect was ever addressed to UNESCO.

### Section 24. Exemption from taxation of salaries and emoluments

- 137. Some States do not accord staff members of the specialized agencies or IAEA exemption from taxation of salaries and emoluments.
- 138. Some countries do not exempt locally recruited ILO staff from taxation. In such cases, taxes are paid under protest by the officials concerned and are reimbursed by ILO. The matter has been raised at various times with the Governments concerned, either by ILO or on an inter-organizational basis.
- 139. In the case of the IFC Regional Mission in the Middle East, based in Cairo, IFC staff assigned from headquarters, Egyptian or foreign, are exempt from taxation of IFC earnings, but not staff appointed locally.
- 140. As regards FAO, a number of countries have on occasion assessed taxes on the FAO-derived income of their citizens who are deemed to have maintained residence in their countries (e.g. Australia, Canada). Citizens and alien residents of the United States of America are subject to taxation whether or not they actually reside in the United States. There have also been isolated cases where the tax authorities of other countries (Brazil, France, Libyan Arab Jamahiriya, Sudan, Turkey, Uganda, United Republic of Tanzania, Venezuela, Yemen, etc.) have levied taxes on FAOderived income. FAO-derived income was taken into account in the United Kingdom for determination of tax relief under the Income and Corporation Taxes Act of 1970.32 Representations were made to the United Kingdom authorities that it was contrary to the specialized agencies Convention to take FAO-derived income into account in any manner for assessment of income tax on other sources of income. It seems that the United Kingdom no longer takes FAO-derived income into account for the purposes of the 1970 Act. There have been certain instances where local Italian authorities have taken FAO-derived income into account in determining the rate of taxation applicable to other sources of income. Representations have been made to the Italian Government, on the grounds that such action is contrary to the provision in the Headquarters Agreement which, in effect, corresponds to section 19 (b) of the specialized agencies Convention. The cases in question have not yet been settled.
- 141. WHO reports that in States that are parties to the specialized agencies Convention the salaries of WHO officials are exempt from taxation, pursuant to section 19 (b) of the Convention. Where a Government taxes the salary of a WHO official and he is unable to obtain exemption, WHO reimburses the official the amount of the tax if the terms of his appointment provide for such reimbursement. However, in the host country,

<sup>&</sup>lt;sup>31</sup> Information regarding the arrest of staff members is contained in the annual report of the Secretary-General to the General Assembly in connection with the agenda item entitled "Personnel questions. Respect for the privileges and immunities of officials of the United Nations and the specialized agencies and related organizations". See e.g. the report submitted to the General Assembly at its thirty-ninth session (A/C.5/39/17).

<sup>32</sup> The Public General Acts, 1970, 1971, part I, chap. 10, p. 111.

Switzerland, the authorities have refused to grant exemption to short-term consultants employed by WHO for less than three months. WHO does not therefore reimburse such consultants of Swiss nationality for any taxes paid to the Swiss authorities, whether federal or cantonal.

- 142. IBRD, IDA and IFC report that a number of member States have not adhered to the relevant provisions of the specialized agencies Convention. In such cases, the relevant provisions of the Articles of Agreement of the agency concerned apply.
- 143. UPU reports that the tax authorities in certain countries have sought to tax the income of UPU staff members who are nationals of those countries. UPU has in such cases successfully obtained the relevant tax exemptions, but the staff members concerned have had to forego participation in the social security system of their country.
- 144. ITU states that the United States does not provide for exemption of its citizens from taxation. ITU refunds tax to staff members who are United States nationals and in turn charges the United States Government for the amount in question.
- 145. UNESCO states that the United States terminated, as of 1 January 1982, an agreement providing for reimbursement of United States income tax levied on salaries and emoluments of United States nationals employed by the organization. Negotiations are under way to conclude a new agreement.
- 146. IAEA states that Indonesia, the Federal Republic of Germany, the Republic of Korea and Turkey have made reservations concerning the application of the tax exemption clause of the IAEA Agreement on Privileges and Immunities to their nationals when present in their respective countries. The Agency reimburses its staff members for taxes paid, pursuant to its Staff Regulations and Rules. Such reimbursement is subject to the limitations contained in the relevant staff rule. Arrangements have been made with the United States under which United States taxes paid by IAEA staff members of United States nationality on IAEA emoluments are reimbursed by the Agency, which is in turn reimbursed by the United States.
- 147. As noted in the present section, certain Governments impose income tax on the salaries and emoluments of their nationals or permanent residents employed as staff members of the specialized agencies or IAEA. In the case of non-nationals, however, Governments as a general rule exempt staff members from payment of such taxes. Staff members may, however, be subject to the payment of other taxes.
- 148. FAO reports that as a general rule in many countries staff members are not exempt from payment of capital gains taxes, real property taxes, sales taxes and VAT. In the host country, only non-Italian staff members with diplomatic status under the FAO Head-quarters Agreement, namely, staff members with the grade P-5 and above, are exempt from payment of VAT on invoices of 100,000 Italian lire or over. Professional non-Italian staff members are exempt from the registration tax on leases. Since 1979, the Italian Government has maintained that non-Italian professional staff

- members with non-diplomatic status are not entitled to exemption from the road circulation tax, and government exemption vignettes are no longer issued to such staff members. FAO is of the view that this is inconsistent with the Headquarters Agreement and with a related exchange of letters between FAO and the Italian Government. FAO provides such staff members with a certificate stating that they are exempt from payment of the road circulation tax; notifications of impositions of fines are returned by FAO to the Ministry of Foreign Affairs.
- 149. UNESCO staff members with the grade P-5 and above who are not French nationals are exempt within the host country from the annual occupiers' tax (taxe d'habitation) in respect of their residential premises. They are also in practice exempt from the annual television tax and from value added and sales taxes in respect of certain goods imported by them for their personal use. Apart from these exemptions, and the exemption from direct taxation enjoyed by all staff members irrespective of grade in respect of UNESCO salaries and emoluments, UNESCO staff are usually required to pay all other taxes.
- 150. UPU staff members with the grade P-5 and above and having diplomatic status are not subject to turnover tax, duties or taxes on liquor and tobacco, and goods imported free of duty. Other non-Swiss staff members are exempt from such duties and taxes on first installation in Switzerland and when transferred. Non-Swiss staff members are also exempt from tax on insurance premiums.
- 151. IAEA staff members with the grade P-5 and above are exempt, under certain conditions, from VAT.
- 152. There has been no uniform interpretation of the terms "salaries and emoluments paid to them [staff members] by the specialized agencies" (section 19 (b) of the Convention). ILO considers that the terms include anything of financial value derived from ILO, with the exception of pension benefits.
- 153. FAO interprets "salaries and emoluments" as including base salary and allowances (family, language, non-resident and rent), plus overtime bonuses and separation payments.
- 154. The WHO Headquarters Agreement and its related arrangements extend tax exemption to WHO indemnities, capital sums due from the pension fund or any other provident fund, and all WHO indemnities for sickness or accident.
- 155. The question has arisen at IBRD and IDA whether a pension to a former employee or his beneficiary is an "emolument". IBRD experience provides no comprehensive answer. It is understood that in Austria such pensions are exempt from taxation because they are deemed to fall within the meaning of "salaries and emoluments". The view in the Netherlands is that such pensions are not exempt from taxation and a 1977 Supreme Court decision held that a pension paid by the United Nations Joint Staff Pension Fund to a former official of the International Court of Justice resident in the Netherlands was not exempt from income tax. The Netherlands tax authorities have also held that the pension of a widow of a deceased IBRD staff member was

not exempt from taxation. In the United States, pensions are not considered as a part of the "salaries and emoluments" referred to in article VII, section 9 (b), of the IBRD Articles of Agreements, even in the case of non-nationals who intend to remain in the United States and receive their pensions there.

- 156. Benefits under the IMF staff retirement plan payable to the estate of a deceased IMF staff member have not been considered as coming under the heading of "salaries and emoluments". The Executive Board of IMF, under the terms of a 1960 decision, does not consider the United States social security tax as falling within the category of reimbursable taxes on salaries and allowances.
- 157. UPU considers the term "salaries and emoluments" to include the total sum paid by the Union to staff members in service.
- 158. IAEA reports that at least one member State considers all benefits paid to a staff member, including tax reimbursement, as taxable emoluments.
- As to what constitutes "emoluments", FAO has found that some States give the term a broad interpretation. For example, the United States considers as emoluments paid to a staff member the moving expenses paid by FAO in connection with the appointment, transfer, home leave and repatriation travel of the staff member. India does not consider the honorariums paid to consultants as emoluments but as taxable income. The question has arisen whether lump-sum withdrawals from the United Nations Joint Staff Pension Fund should be considered as emoluments due and received from FAO and, as such, be exempt from taxation. FAO, however, has not yet decided how these withdrawals are to be interpreted. Accordingly, it has not sought tax exemption on such sums, nor does it reimburse taxes paid on them.
- 160. There is no uniform definition of the types of taxes included under the term "taxation in respect of . . . salaries and emoluments".
- 161. ILO staff employed by the ILO London Branch Office, are subject to the national social security scheme and contribute thereto. ILO (like other employers benefiting from immunities) is considered liable for payment of the employer's contribution, but such liability is not enforced. Non-payment of the employer's contribution does not diminish the rights of the employee provided the employee's contribution is paid. In 1973, locally recruited ILO staff in Cairo were obliged to contribute to the national social security scheme. ILO declined to pay the employer's contribution for its staff, maintaining that the benefits under ILO schemes to which it contributed were equivalent or superior to those provided under the national social security legislation. In Romania, locally recruited ILO staff are subject to the compulsory national social security scheme.
- 162. ILO reports that, in 1975, the Swiss authorities informed the international organizations with head-quarters in Geneva that the reason previously accepted for exemption from AVS (Assurance vieillesse survivants) (Old age and survivors insurance), namely, that it was "excessively burdensome", no longer existed and that they intended to subject officials of international

organizations of Swiss nationality to such insurance. A reply was sent to the Swiss authorities on behalf of all United Nations agencies in Geneva. The matter is still under discussion. As for the application of such insurance to Swiss UPU officials in Bern, UPU reports that, after lengthy negotiations, it was agreed that Swiss staff members would be free to pay or not to pay the insurance. If they chose not to pay, they would be able to rejoin the insurance scheme later if they wished, provided that they did so on a permanent basis and that the contributions due from self-employed persons were paid. ILO reports that questions had arisen earlier (in connection with the imposition of school taxes (1968) and certain communal taxes) as to the imposition of taxes, solely on persons otherwise enjoying immunity from taxation, for services provided free to the population as a whole. ILO took the position that, while taxes corresponding to services should be paid, when such taxes were imposed exclusively on tax-exempt officials they constituted a method of circumventing the general tax exemption provided for in the Headquarters Agreement. As far as school taxes are concerned, however, the relevant provisions imposing such taxes were modified in 1971, putting on an equal footing exempt officials and other persons domiciled in Geneva. For example, WHO notes that, since 1971, its officials living in the canton of Geneva have been exempt from "school taxes". See however section 14 of part B of the present study.

- 163. FAO practice in regard to United States nationals who are required to make social security payments is to reimburse them in part (that is, to reimburse the difference between the amount of the contribution required of an official of FAO and the amount that he would have had to pay if he worked for an employer subject to United States taxation). Under FAO rules, States and/or city taxes levied on FAO-derived income are reimbursed only to United States nationals and to non-national residents stationed in the United States. However, exceptions may be made from time to time with respect to Canadian provincial taxes or to State taxes in the United States imposed on United States staff members or on alien residents stationed outside the United States who are regarded as residents of a State on the grounds, for example, of owning property there.
- 164. ICAO reports that Canadian provincial health taxes are considered as a tax on "salaries and emoluments".
- 165. IBRD, IDA and IFC report that federal and State income taxes and social security taxes are considered as taxes on "salaries and emoluments". IMF does not consider social security taxes as a tax on "salaries and emoluments" but reimburses staff members obliged to pay the tax in the amount of the difference between the amount the staff member pays and the lesser amount he would pay if the employer contributed.
- 166. Tax exemption for UPU salaries and emoluments covers income tax, social security deductions (old age and survivors' insurance and disability insurance), the ecclesiastical contribution and the national defence tax.
- 167. For WHO, taxes levied on salaries and emoluments are the only type of taxes considered as covered by the relevant provision of the specialized

agencies Convention. WHO does not include other types of taxes, such as personal income tax on income not derived from WHO, social security contributions, the ecclesiastical contribution, local taxes and school taxes.

- 168. In the case of IAEA, the relevant income tax exemption provision covers personal income tax on emoluments paid by the Agency, but no other taxes.
- 169. ILO and ITU report that in Switzerland the spouses of tax-exempt international officials who are taxable on their own income may either not declare their international income and pay tax essentially as if they were unmarried or declare their international income and pay tax at a rate which takes it into account. The procedure followed in such cases, and in cases of officials having taxable income in addition to their international income, is currently under consideration with the Swiss authorities.
- 170. FAO reports that certain countries exempt FAO staff members from even filing a tax return, but that in other countries tax returns are filed and are required to show income derived from FAO or to provide evidence of FAO employment and consequent tax exemption.
- 171. UNESCO reports that, under French law, the salaries of French staff members of international organizations should be taken into account in calculating the rate of tax applicable to the income (including accessory income or professional earnings) of the spouse. Following representations from UNESCO, the French Government ceased to apply this provision to UNESCO. Thus salaries and emoluments of UNESCO staff members, whether of French or other nationality, are not taxed directly or indirectly in France.
- 172. UPU staff members who are not Swiss nationals are not required to file an income tax return. Staff members of Swiss nationality are required to do so but it is UPU that certifies information as to UPU salaries and emoluments.
- 173. As regards ICAO, only Canadian staff members have to file federal and provincial income tax returns.
- 174. IFC reports that occasional difficulties arise in relation to taxation, for example where a staff member is designated by IFC to serve as director of a local development finance company in which IFC has invested and where the local development finance company pays the director's remuneration. That remuneration is due to IFC but, because it is paid to (or through) the individual, municipal law frequently requires the payee to deduct income tax, which must then be subject to a complex reclaim procedure.
- 175. IBRD, IDA, IFC and IMF report that, in calculating income tax payable, the United Kingdom formerly took into account the remuneration and emoluments received from those agencies in calculating the rate of tax on the income of its nationals, but later decided to exclude from such calculation the remuneration and emoluments received by its nationals who were officials of the United Nations and non-resident in the country. IBRD has requested confirmation that its officials will be accorded similar treatment, but a reply has not yet been received. IMF states that this decision

has not been interpreted to include the Fund. Again with respect to IBRD, the policy of Malaysia is to impose income tax on non-IBRD income at rates fixed by taking into account the amount of non-taxable IBRD income. No effort is currently being made by the Bank to have this policy changed.

176. Concerning the issue of calculation of rate of tax, the Supreme Court of the Netherlands held in 1972<sup>33</sup> that no account should be taken of tax-exempt United Nations salary in determining the rate of tax on the non-exempt income of a United Nations official. The Court of Justice of the European Communities, in a decision of 16 December 1960,<sup>34</sup> held that Communty salaries might not be taken into account in determining marginal tax rates.

### Section 25. Immunity from national service obligations

- 177. WHO reports that staff members other than temporary staff engaged for conference and other short-term service, or temporary consultants, may on application be granted leave of absence to fulfil national service obligations. Such absence, charged to annual leave and thereafter to leave without pay, extends for a period not exceeding one year in the first instance, subject to extension on request. Upon application, within 90 days after release from national service, the staff member is restored to active duty in the organization, usually with the same status as at the time he left for national services. These provisions have so far been made use of only by Swiss nationals. The organization has not had requests for such leave from staff members of other nationalities.
- 178. UPU states that only Swiss staff members have to fulfil national service obligations. All other staff are exempt while in the employment of the organization. If the dates for national service obligations in respect of officials of Swiss nationality, are not suitable, a request for postponement is addressed to the Swiss authorities. UPU and ITU have prepared a list of officials whom they would wish to have exempted from national service. UPU notes that the Swiss authorities have been increasingly restrictive in the granting of such dispensation, and that currently no Swiss staff member is exempt from national service obligations.
- 179. FAO states that, at its request, the Italian Government has granted temporary deferment of national service requirements for Italian staff. The organization has not been called upon to take action with respect to staff members of other nationalities.
- 180. IMF states that, while it has taken no action under section 20 of the specialized agencies Convention, it has adopted a liberal leave policy for individuals responding to national service obligations. IAEA has not compiled a list of officials for exemption from

<sup>&</sup>lt;sup>33</sup> Netherlands, Supreme Court, "Beslissingen in Belastingzaken", Nederlandse Belastingrechtspraak [Fiscal jurisprudence], case No. 25, June 1972, Deventer-Amsterdam, 1972.

<sup>&</sup>lt;sup>34</sup> Court of Justice of the European Communities, Reports of cases before the Court (Luxembourg, 1960), p. 559.

national service obligations. However, special paid leave is granted to staff members who have to serve in Austria. IAEA notes that Luxembourg and Switzerland have made reservations to the relevant provisions of its Headquarters Agreement.

### Section 26. Immunity from immigration restrictions and alien registration

- 181. ILO reports that in 1968 the Chilean immigration authorities refused admission to an official of the International Labour Office because of his political activities in Chile prior to his service with the Office. On the personal intervention of a high official of the Chilean Government, the official was admitted to the territory on a provisional basis. Shortly thereafter, the Government decided to authorize him to remain in the country on the understanding that it did so out of respect for its international obligations concerning the admission of staff members of the International Labour Office on official mission, but required the official to sign a statement promising strict observance of the requirements of article 1.2 of the Staff Regulations of the Office, namely, that staff members "shall not engage in any political . . . activity".
- 182. FAO states that since 1965 minor problems have arisen. These have been resolved after consultations. It is reported in this connection that one country withholds immunity with respect to officials' children over the age of 18.
- WHO states that, so far as can be ascertained, only three cases have been encountered in this regard. The first was in 1965, when the Egyptian Government claimed that non-nationals residing in the country at the time of appointment continued to be subject to alien registration while in the service of WHO. The matter was finally settled to the organization's satisfaction. The second case, which was similar to the first, arose in 1968 but was also settled to the organization's satisfaction. The third case arose in Geneva in 1978, when the son of a staff member of the organization residing away from his parents and who was seeking employment in Geneva was ordered to leave Switzerland by the Swiss authorities. The matter was settled to the satisfaction of WHO when the person concerned took up residence with his parents.
- 184. UPU notes that in its experience the expression "relatives dependent on them" applies to children and sometimes to others (father, mother, sister, brother) for whom a dependency allowance is payable under the UPU Staff Regulations.
- 185. IAEA reports that Austria does not impose restrictions on immigration nor does it have a system of alien registration. There exists, however, an obligation on both the lessee and the lessor of an apartment or house to register with the local police whenever residence in Austria is taken up or changed. However, after amendment of the Law of Registration (Meldegesetz) in 1979, staff of international organizations, including IAEA, holding valid identity cards issued by the Federal Ministry of Foreign Affairs are exempt from such obligation.

#### Section 27. Exchange facilities

- 186. Nearly all the specialized agencies and IAEA state that no problems have arisen under paragraph (d) of section 19 of the specialized agencies Convention concerning exchange facilities. FAO, however, reports that staff members in the field sometimes have difficulties in converting accumulated local currency when leaving a country at the end of an assignment.
- 187. UNESCO states that, according to the Banque de France, French exchange control is applicable to all natural and legal persons (whatever their nationality) who are resident—as defined under the French exchange control—in one of the countries of the *Zone franc* and that accordingly such persons may have only internal accounts in France. Inasmuch as this requirement would apply to certain UNESCO officials, it could be interpreted as incompatible with article 22 (e) of the UNESCO Headquarters Agreement, which reads:

Officials governed by the provisions of the Staff Regulations of the organization:

(e) shall, with regard to foreign exchange, be granted the same facilities as are granted to members of diplomatic missions accredited to the Government of the French Republic.

The issue has not yet been resolved with the French authorities.

### Section 28. Repatriation facilities in time of international crisis

- 188. Most of the specialized agencies and IAEA have had no recourse to the provisions of section 19 (e) of the specialized agencies Convention relating to repatriation facilities in time of international crisis. FAO states that adequate arrangements have been made whenever necessary with host countries for the evacuation of FAO staff and their families, on occasion within the framework of the United Nations emergency evacuation scheme.
- 189. IMF reports that IMF staff members have on a limited number of occasions of official travel experienced a threat of physical injury because of civil disturbances. Arrangements for possible repatriation have been made on an *ad hoc* basis, at times in co-operation with the host country. In addition, IMF personnel are instructed to follow the advice of the "designated official" of the United Nations.

### Section 29. Importation of furniture and effects

190. No major difficulties have arisen concerning the interpretation of the words "furniture and effects" and "at the time of first taking up their post" in section 19 (f) of the specialized agencies Convention. The term "furniture and effects" has been generally construed to include a car, which may be imported by the staff member within a period of from three to 18 months of arrival, depending on the country concerned. IBRD and IDA state that a limited number of problems have arisen in connection with the importation of cars. In one in-

stance, IBRD decided as a matter of practical policy not to insist on its rights.

- 191. FAO reports that in one country the phrase "at the time of first taking up their post in the country" has been understood as not applying to non-nationals already residing in the country when appointed to FAO. FAO also reports that the Italian authorities have recognized small motorcycles as part of "effects", while a large motorcycle is considered a substitute for a car.
- 192. FAO notes that import privileges are generally granted on the understanding that the articles are imported for the personal use of the official and his dependants, and not for gift or sale. FAO staff members who leave Italy for a field assignment of one year or more must request duty-free importation of their personal effects within six months of their re-entry on duty at FAO Headquarters in Rome. Furniture and vehicles imported by FAO officials should be re-exported on transfer or termination of employment. In some cases, arrangements have been made to permit the transfer of property (by gift, loan or sale) to other persons enjoying similar import privileges and/or sale of the articles on the local market subject to payment of import duties (full or reduced rate). Currency conversion, however, is the responsibility of the staff member, in compliance with local regulations.
- 193. WHO notes that, when a person ceases to be an international official, he automatically ceases to enjoy privileges and immunities under the specialized agencies Convention. On his return to his home country, he is treated like any other national. The question whether furniture and effects imported into a country on such an occasion are subject to taxation depends upon the municipal regulations of the country concerned. IBRD, IDA, IFC and IMF report that no problems have arisen in connection with the removal of effects by a staff member at the end of a tour of duty.

## Section 30. Diplomatic privileges and immunities of the executive heads and other senior officials of the specialized agencies and of IAEA

- 194. The privileges and immunities, exemptions and facilities of the executive heads of specialized agencies and of IAEA have been fully recognized. In the case of UPU, full recognition is granted provided that the executive head, the Director-General, is not of Swiss nationality. A minor problem arose with respect to a request by IMF for a diplomatic parking space at National Airport in Washington D.C. for the Managing Director's car. The request was denied on the grounds that the car provided to the Managing Director did not have diplomatic licence plates.
- 195. Section 20 of the Agreement on the Privileges and Immunities of IAEA provides that diplomatic privileges and immunities be accorded not only to the Director-General but also to Deputy Directors-General. The United Kingdom has made a reservation concerning that provision with respect to citizens of the United Kingdom and colonies.
- 196. UPU states that senior officials of UPU not of Swiss nationality are accorded diplomatic privileges and

- immunities on the basis of a 1947 decision of the Swiss Federal Council, later confirmed by letter when the present Headquarters Agreement came into force. UPU also notes that a senior Swiss staff member, who because of his nationality does not enjoy the privileges and immunities accorded to other staff members of similar rank, was unable to purchase a car from a third country at the diplomatic price because he did not enjoy diplomatic status in Switzerland.
- 197. ITU reports that five elected officials (above D-2 level) enjoy the same privileges and immunities as the Executive Head of ITU; and that non-Swiss staff at the P-5 level and above enjoy limited diplomatic privileges.
- 198. IMF states that many resident representatives, as well as some staff members of IMF offices in Paris and Geneva, have been granted diplomatic privileges as a matter of courtesy.

### Section 31. Waiver of the privileges and immunities of officials

- 199. Most specialized agencies and IAEA state that they have received no requests for waiver of immunity in respect of acts performed or words spoken or written by staff members in their official capacity. FAO reports that in cases of traffic violations FAO would usually accede to requests for waiver of immunity. FAO officials are aware of this policy and usually endeavour to settle fines imposed by police authorities as well as any third-party liability claims without invoking immunity.
- 200. Where proceedings have been instituted against a third person and an official of WHO is requested to appear as a witness, the organization generally allows a written deposition, which may be used as evidence, but is reluctant to extend the waiver of immunity to appearance in court or to oral examination and crossexamination. The Director-General has exceptionally waived the immunity of staff members involved in proceedings. In such cases, the Director-General was satisfied that such waiver was in the interests of justice and that the interests of the organization would not be adversely affected. WHO notes that the general policy of non-waiver of immunity does not mean that the organization would not, in appropriate cases, be prepared to assume responsibility for compensation in respect of injury or damage caused by a staff member in the exercise of official functions. Governments of countries receiving technical assistance from FAO are generally under the obligation, except in cases of gross negligence or wilful misconduct on the part of the official who caused the injury or damage, to assume responsibility for third-party claims.
- 201. In the experience of WHO, UNESCO and IAEA, waiver of immunity has generally been sought in respect of private matters, as in questions relating to family law. WHO notes that, as such cases do not engage the official responsibility of the staff member concerned, waiver has always been granted. UNESCO reports that it has granted waiver of immunity in two divorce cases. IAEA, however, on the occasions when it has been requested to waive immunity, has declined to do do.

## Section 32. Co-operation with the authorities of member States to facilitate the proper administration of justice

202. Most specialized agencies and IAEA report that they have little experience of co-operation with the appropriate authorities of member States to facilitate the proper administration of justice, to secure observance of police regulations, and to prevent the occurrence of abuse in connection with the privileges, immunities and facilities accorded by the specialized agencies Convention.

203. FAO notes that the FAO Staff Regulations provide that privileges and immunities are granted in the interests of the organization and furnish no excuse to staff members for non-performance of their private obligation or failure to observe laws and police regulations. The FAO Manual enumerates types of conduct that may give rise to disciplinary measures, inter alia conduct detrimental to the name of the organization, serious violation of any applicable national law, conduct tending to endanger lives or property, and neglect or avoidance of just claims for debts or comparable obligations. Officials are reminded from time to time of their responsibility for enrolment of their domestic staff in the social security system and for the regular payment of contributions thereto. Where the organization has invoked its immunity in the case of orders for disclosure of information, it requests the staff member concerned to communicate the information required and is prepared to certify the accuracy of information relating to the staff member's salary.

204. WHO and IAEA state that they maintain close liaison with the host country authorities, particularly in cases of violations of traffic regulations. When police reports in such cases are received, they are transmitted to the staff members concerned, whose attention is drawn to their obligations to respect local laws and regulations. IAEA co-operates with the local authorities by directly providing the necessary information, without waiving the immunity of the staff member.

205. IMF has acceded in exceptional cases to court requests for the appearance of staff members to provide evidence on matters connected with their official duties. However, it has refused to recognize court orders summoning the organizations as such to appear in court. IMF has also declined to observe court orders aimed at attaching its funds or requiring it to make deductions from salaries or terminal emoluments of staff members with a view to settling debts that the latter may have contracted. However, the Fund will, in accordance with its Staff Rules, make deductions from staff members' salaries or terminal emoluments and make payments to third parties where indebtedness has been established by a final judgment or is admitted by the staff member.

### CHAPTER V

Privileges and immunities of experts on mission for the specialized agencies and the International Atomic Energy Agency and of persons having official business with the specialized agencies and the International Atomic Energy Agency

## Section 33. Persons falling within the category of experts on mission for the specialized agencies and IAEA

206. FAO regards the following as "experts" within the meaning of paragraph 2 of annex II to the specialized agencies Convention: (a) experts participating in committees of the organization in their individual capacity; (b) experts not staff members of the organization (in other words, not subject to its staff regulations and rules or responsible to the Director-General) performing services for the organization either on a contractual basis or on the basis of an agreement with a Government or of designation by a governing body; (c) staff of the External Auditor's Office, while on the business of FAO.

207. WHO considers persons appointed in an advisory capacity to the organization or to a Government for temporary periods, and who are not staff members, to be "experts".

208. IAEA considers safeguard inspectors, project examiners and persons other than officials travelling on mission for the Agency to be experts.

209. It should be noted that the annexes to the specialized agencies Convention, each of which concerns a particular agency, do not all contain a reference to the privileges and immunities to be accorded to experts on mission.

### Section 34. Privileges and immunities of experts on mission for the specialized agencies and IAEA

210. With regard both to the specialized agencies for which the relevant annexes of the specialized agencies Convention make reference to the privileges and immunities to be accorded to experts on mission and to IAEA (article XVI of the IAEA Headquarters Agreement, and article VII of the Agreement on the Privileges and Immunities of IAEA), the granting of privileges and immunities to the experts concerned has raised virtually no problems or difficulties. There have been no cases where waiver of immunity has been requested. WHO states that it would waive the immunity of experts in private matters not related to their official duties, in conformity with its practice concerning staff members. ILO, however, reports that in one case an ILO expert was arrested (see section 42 below).

#### CHAPTER VI

### United Nations laissez-passer and facilities for travel

### Section 36. Issue of United Nations *laissez-passer* and their recognition as valid travel documents

- 211. The specialized agencies and IAEA state that practice varies from country to country regarding recognition of the *laissez-passer* as a valid travel document. IMF, IBRD, IDA and IFC state that a number of countries which have not adhered to the specialized agencies Convention recognize the *laissez-passer*. IMF and UPU emphasize the usefulness of the *laissez-passer* for the official travel of an official where his national passport is not recognized. Several countries require a national passport in addition to the *laissez-passer* before permitting entry, while others recognize the *laissez-passer* without production of a national passport. FAO notes that visas in *laissez-passer* are accepted.
- 212. IAEA reports however that, while the *laissez-passer* has been recognized, bilateral agreements between States providing for waiver of visa requirements in the case of their nationals are not applicable to the *laissez-passer*, which does not specify nationality.

## Section 37. Freedom of movement of personnel of the specialized agencies and of IAEA; inapplicability of persona non grata doctrine

- 213. Difficulties have seldom arisen in the exercise of the right of transit for persons (officials, experts on mission or other persons), other than representatives of States, in connection with the performance of official functions, e.g. attendance at meetings, or travel on mission. FAO reports one difficulty that arose in the case of experts who arrived in a country without visas and were required to remain at the airport until the question could be settled.
- 214. UNESCO reports one case in which a staff member on mission to the country of which he was a national was prevented from leaving the country. Following a number of representations by the Director-General and the Executive Board to the Government concerned, the staff member was permitted to leave, after a period of more than one year.
- 215. IBRD states that although several problems (delay or denial of visas, restriction of transit, etc.) have arisen since 1965, these have involved member States not parties to the specialized agencies Convention with respect to the World Bank (annex VI of the Convention).
- 216. There have been few cases in which officials of the specialized agencies or IAEA have been declared persona non grata or in which expulsion proceedings have been initiated. FAO states that on a few occasions it has spontaneously withdrawn an official when difficulties arose in his relations with the national authorities.
- 217. WHO reports occasions when expulsion proceedings have been taken against officials taking part in

- technical projects. In the majority of cases, such action was taken on purely political grounds and was not justified. Where there was a manifestly improper motivation, WHO has requested the official to protest the expulsion proceedings and has assigned the official elsewhere.
- 218. IAEA states that occasional difficulties have been experienced in obtaining visas for persons of certain nationalities who are required to attend meetings convened by the Agency.

### Section 38. Issue of visas for holders of United Nations laissez-passer

- 219. For the most part, neither the specialized agencies nor IAEA have encountered problems with respect to the speedy issuance of visas. A few problems are reported by some organizations. FAO states that a project in a country was prejudiced by delays in obtaining transit permits in a neighbouring country. IBRD reports substantial delay in one State in obtaining visas for staff members of a certain nationality. ITU has experienced substantial delays in obtaining visas for official travel from certain countries.
- 220. As a general rule, no charge is made by States for the issue of a visa for a *laissez-passer* or a national passport accompanied by a *laissez-passer*. WHO reports, however, that a number of countries generally impose a charge for visas sought on national passports for official travel, notwithstanding the presentation of a certificate showing the official nature of the travel.

### Section 39. Certificates issued by the specialized agencies and by IAEA

- 221. There is no standard definition by the specialized agencies and IAEA of the term "experts" and "other persons" in section 29 of the specialized agencies Convention. FAO issues such certificates to subcontractor personnel employed on field projects, persons employed under special services agreements, and experts "employed on missions" within the meaning of annex II, paragraph 2, of the Convention.
- 222. WHO considers all persons appointed by WHO in an advisory capacity, and who are not staff members, as within the purview of the terms "expert" and "other persons".
- 223. IBRD, IDA and IFC employ these terms only with reference to consultants. IMF and UPU use them to refer to technical assistance experts who are engaged on a contractual basis and are not members of the regular staff. ITU uses the terms of refer to subcontractors and individuals employed under special service agreements. IAEA refers to persons attending IAEA advisory group meetings as experts.
- 224. All the specialized agencies and IAEA report that adequate recognition is usually given to certificates

issues to experts and other persons travelling on the business of these organizations who are not holders of United Nations *laissez-passer*.

Section 40. Diplomatic facilities for the executive heads and other senior officials of the specialized agencies and of IAEA while travelling on official business

225. No problems are reported concerning the application of section 30 of the specialized agencies Con-

vention, which provides that executive heads and other senior officials travelling on the United Nations *laissez-passer* on official business are to be accorded the same travel facilities as are accorded to officials of comparable rank in diplomatic missions. UPU notes that only the Director-General's *laissez-passer* is marked "diplomatic", thus giving him the status of ambassador. Other UPU senior officials carrying the red *laissez-passer* (D-2 and above) do not seem to enjoy any greater facilities than staff members carrying the blue *laissez-passer*.

#### CHAPTER VII

### Settlement of disputes

### Section 41. Settlement of disputes

226. The modes of settlement of disputes have in the case of the specialized agencies and IAEA included negotiation, conciliation and arbitration (see p. 182 above, section 1 (b)). FAO has had recourse to arbitration in three cases: once by ICC and twice by an individual arbitrator chosen by the parties. In one of the two latter cases the result was unsatisfactory, since the parties were in dispute as to the intepretation of the arbitrator's findings. FAO is at present endeavouring to reach agreement on the settlement. In general, however, the parties have been satisfied with the fairness of settlements.

227. FAO notes that a problem has arisen in a case brought against it in the host country, Italy, by a parastatal corporation (see p. 187 above, section 7) relating to a lease. FAO informed the Corte di Cassazione that no denial of justice would ensue since the dispute could be settled by arbitration, as provided in the relevant lease. The court, however, considered the standard arbitration clause used in FAO contracts inoperative, since (a) the parties could not by mutual consent extend the organization's immunity as established under international law, nor (b) could they limit the court's jurisdiction pursuant to article 2 of the Italian Code of Civil Procedure. Such limitation could be made only with respect to a contractual dispute between foreigners or between a foreigner and a citizen who was neither resident nor domiciled in Italy. The conclusions of the Corte di Cassazione, in the view of FAO, were inconsistent with a statement on the organization's immunity from legal process and measures of execution in Italy made by the representative of Italy at the eightysixth session of the FAO Council, reading in part as follows:

The first point concerns the validity of arbitration clauses (which FAO might include in all the contracts it executes in Italy) aimed at avoiding that any dispute arising from the contract be subjected to the jurisdiction of Italian courts.

This matter was dealt with briefly and incidentally, as an obiter dictum, in the 1982 judgment of the Corte di Cassazione. The court stated that the particular arbitration clause contained in the FAO/INPDAI contract (which clause had not even been invoked by FAO) was not valid under Italian law, and that therefore it could not possibly derogate from the jurisdiction of Italian courts.

The matter, however, deserves much more attention. Italy has become a party to the Convention on the Recognition and Enforcement of Foreign Arbitral Awards (New York, 10 June 1958).35 Not only has the said Convention been approved by the Italian Parliament (thus acquiring force of law in Italy), but the Corte di Cassazione has maintained in numerous judgments that arbitration clauses providing for foreign arbitral awards in accordance with the New York Convention do have the power to derogate from the jurisdiction of Italian courts. In practice, therefore, FAO could very well make use of such clauses in the contracts it executes in Italy, and would thus be no longer subjected to Italian courts in any dispute arising from its contracts. It goes without saying, however, that, if the other contractor attempted to ignore the arbitration clause and initiated legal proceedings against FAO before an Italian court, the organization would have to appear before the judge in order to demonstrate to him the existence of a valid arbitration clause; otherwise, the proceedings would continue in absentia until the issuance of a final judgment. Under the Italian laws of civil procedure, it is not conceivable that anyone else but FAO should appear before the court to protect its own interests. In particular, the Italian Government could not defend the interests of FAO before a court, but could at most put at the disposal of FAO, with no charge, the Avvocatura dello Stato, which is a body of lawyers by which the Italian State itself is represented and defended in court disputes.36

- 228. WHO has on two occasions settled, by conciliation, disputes that had arisen between it and firms carrying out UNDP-supported projects. The settlements were satisfactory to both parties.
- 229. IMF states that a small number of disputes have been settled through negotiations to the satisfaction of both parties. IMF has also agreed in its contracts to the submission of disputes to arbitration. One matter of contention is the applicability of section 31 of the specialized agencies Convention to staff members. IMF takes the view that the provision is not applicable.

### Section 42. Settlement of disputes regarding alleged abuses of privileges

- 230. A few cases have arisen in respect of alleged abuses of privileges.
- 231. ILO reports that in Ecuador, in 1971, the apartment of an ILO expert was searched by the army and the

<sup>35</sup> United Nations, Treaty Series, vol. 330, p. 3.

<sup>36</sup> See p. 189 above, footnote 17.

expert arrested. No charge, however, was made against the expert, who was promptly released. ILO took the position, transmitted to the Resident Representative to be used in his representations to the Government, that, while it recognized that the immunity of officials was limited to official acts, measures such as search of a residence created a sense of insecurity; ILO experts were often required to meet leaders of labour and cooperative movements and documents in their possession could be the property of the organization and inviolable. ILO considers it highly desirable that problems involving experts be resolved in consultation with the specialized agency concerned, as envisaged in article VII of the specialized agencies Convention.

- 232. WHO reports a case arising at its Regional Office in the Philippines where an abuse of privileges was alleged and a search warrant obtained from a judge, who ordered the seizure of dutiable items in the baggage of a WHO staff member. The warrant was quashed by the Supreme Court of the Philippines, which held that, if the judge has reason to suspect abuse of diplomatic immunity, he should have forwarded his findings to the Department of Foreign Affairs for action under article VII of the specialized agencies Convention.<sup>37</sup>
- 233. The instruments of accession to the specialized agencies Convention tendered for deposit by the Governments of Bulgaria, the Byelorussian SSR, Cuba, Czechoslovakia, the German Democratic Republic, Hungary, Mongolia, Poland, Romania, the Ukrainian SSR and the USSR were accompanied by reservations to the effect that these States did not consider themselves bound by sections 24 (concerning settlement of disputes regarding alleged abuses of privileges and immunities) and 32 (concerning reference to the International Court of Justice of differences arising out of the interpretation or application of the Convention). <sup>38</sup> The Government of the United Kingdom has notified the Secretary-General that it is unable to accept certain reservations made by those States because in its view they are not of the kind

which intending parties to the Convention have the right to make.

# Section 43. Reference to the International Court of Justice of differences arising out of the interpretation of the Convention on the Privileges and Immunities of the Specialized Agencies

- 234. There have been no instances of reference of differences relating to the interpretation of the specialized agencies Convention to the International Court of Justice in accordance with section 32 of the Convention.
- 235. FAO, however, as noted above (p. 187, para. 48 (b)), reports that, in 1985, its governing bodies would consider whether or not the organization should request an advisory opinion from the International Court of Justice on the interpretation of article VIII, sections 16 and 17, of the FAO Headquarters Agreement.
- 236. UPU notes that the possibility of recourse to the International Court of Justice should not be excluded if the existing differences between the host State and UPU in the case concerning contributions for road construction are not satisfactorily resolved (see p. 191 above, section 14).
- 237. As noted above (para. 233), the instruments of accession tendered for deposit by 11 States were accompanied by reservations regarding sections 24 and 32 of the specialized agencies Convention. In addition, the instruments of accession tendered by the Governments of China and Indonesia were accompanied by reservations concerning section 32 of the Convention.<sup>39</sup> The United Kingdom Government had notified the Secretary-General that it is unable to accept certain reservations made by these 13 States, for the reasons already noted (ibid.). The Netherlands Government has notified the Secretary-General of its opinion that the reservation made by one State to section 32, and similar reservations that other States have made or may make in the future, are incompatible with the objectives and purposes of the Convention. It does not however wish to raise a formal objection to these reservations.

### CHAPTER VIII

### Annexes and final provisions

### Section 44. Annexes to the Convention on the Privileges and Immunities of the Specialized Agencies

238. No legal controversies appear to have arisen with respect to the provisions of the annexes to the specialized agencies Convention. Most specialized agencies report that the privileges, immunities, exemptions and facilities granted by the pertinent annexes to the Convention have been generally accorded by States that are not parties to an annex. IBRD, IDA and IFC note that the relevant provisions of their Aricles of Agreement would apply even where certain member States are not parties to the pertinent annex.

239. As to problems that might arise by reason of States being parties to different revised texts of an annex, FAO and WHO report that no such difficulties or problems have arisen.

#### Section 45. Supplemental agreements

240. Some specialized agencies have entered into agreements additional to the specialized agencies Convention. ILO states that many of the agreements relating to ILO offices in the field contain provisions under which the Government of the host country would

<sup>&</sup>quot;See United Nations, *Juridical Yearbook 1972* (Sales No. E.74.V.1), p. 209.

<sup>&</sup>lt;sup>38</sup> Multilateral treaties deposited with the Secretary-General... (see p. 184 above, footnote 5) and Supplement... (p. 197 above, footnote 26).

<sup>39</sup> Ibid.

- grant ILO and its staff privileges and immunities not less favourable than those granted to any other intergovernmental organization and its staff in the country.
- 241. Agreements between WHO and member States receiving assistance in the framework of technical cooperation extend to subcontractors engaged by WHO a measure of privileges and immunities with respect to jurisdiction, taxation and customs duties.
- 242. IMF states that it receives assurances from member States requesting technical assistance that they will grant experts the same privileges and immunities as would be granted to staff members. Several countries have granted additional privileges beyond those provided by the specialized agencies Convention.
- 243. Agreements concerning the status, privileges and immunities of the specialized agencies and of IAEA continue to be included in the United Nations *Juridical Yearbook*.
- Section 46. Accession to the Convention on the Privileges and Immunities of the Specialized Agencies by Member States of the United Nations and by member States of the specialized agencies
- 244. As of 1 June 1985, ninety States were parties to the specialized agencies Convention in respect of one or more of the specialized agencies. As noted previously (p. 209, para. 233), the instruments of accession tendered for deposit by eleven States were accompanied by reservations regarding the application of sections 24

- and 32, and those tendered for deposit by two States (see p. 209, para. 237, above) were accompanied by reservations regarding the application of section 32. Eight States have made declarations regarding the application of section 11. One State has made a declaration regarding the application of section 3 (b). One State has submitted notification of its inability to accept the reservations made by thirteen States concerning sections 24 and/or 32. Another State has submitted notification to the effect that present and future reservations concerning section 32 are incompatible with the objectives and purposes of the Convention, but that it did not wish to raise a formal objection and did not oppose entry into force of the Convention between itself and the States making such reservations.
- 245. States that are not parties to the specialized agencies Convention or that have not extended its application to all agencies have for the most part agreed to apply the provisions of the Convention to agencies operating in their territory. Such agreements concern technical assistance projects or conference agreements concluded for meetings held outside established head-quarters or offices. In the case of IBRD, IDA, IFC and IMF, if a member State is not a party to the specialized agencies Convention, it is their Articles of Agreement that apply.
- 246. No cases have been reported of the withdrawal of privileges and immunities previously granted to an organization.
- 247. The Agreement on the Privileges and Immunities of IAEA, which is open to all 112 Member States of the Agency, had 56 States parties as of 1 June 1985.

### كينية الحصول على منشورات الامم المتحدة

يمكن العمول على منشورات الامم المبتحدة من المكتبات ودور التوريع في حبيع امحاء العالم · استطم عنها من المكتبة التي تتعامل معها أو اكتب الى : الامم المنتحدة ،قسم البيع في يبويورك او في جنيف ·

#### 如何购取联合国出版物

联合国出版物在全世界各地的书店和经售处均有发售。请向书店询问或写信到纽约或日内瓦的联合国销售组。

### HOW TO OBTAIN UNITED NATIONS PUBLICATIONS

United Nations publications may be obtained from bookstores and distributors throughout the world. Consult your bookstore or write to: United Nations, Sales Section, New York or Geneva.

#### COMMENT SE PROCURER LES PUBLICATIONS DES NATIONS UNIES

Les publications des Nations Unies sont en vente dans les librairies et les agences dépositaires du monde entier. Informez-vous auprès de votre libraire ou adressez-vous à : Nations Unies, Section des ventes, New York ou Genève.

#### КАК ПОЛУЧИТЬ ИЗДАНИЯ ОРГАНИЗАЦНИ ОБЪЕДИНЕННЫХ НАЦИИ

Издания Организации Объединенных Наций можно купить в книжных магазинах и агентствах во всех районах мира. Наводите справки об изданиях в вашем книжном магазине или пишите по адресу: Организация Объединенных Наций, Секция по продаже изданий, Нью-Йорк или Женева.

#### COMO CONSEGUIR PUBLICACIONES DE LAS NACIONES UNIDAS

Las publicaciones de las Naciones Unidas están en venta en librerías y casas distribuidoras en todas partes del mundo. Consulte a su librero o diríjase a: Naciones Unidas, Sección de Ventas, Nueva York o Ginebra.